

Sumarium Records Center Stap: Nyanza Chemical PTIAK: 3.6 OTHER: 264487

SDMS DocID

264487

W92194F

FINAL REMEDIAL INVESTIGATION REPORT

VOLUME | OF IV

NYANZA OPERABLE UNIT III - SUDBURY RIVER STUDY MIDDLESEX COUNTY, MASSACHUSETTS

NUS Corporation

EPA Work Assignment No. 02-1L15 Contract No. 68-W8-0117 NUS Project No. 0214

May 1992



W92194F

FINAL REMEDIAL INVESTIGATION REPORT

REMEDIAL INVESTIGATION/FEASIBILITY STUDY

NYANZA OPERABLE UNIT III MIDDLESEX COUNTY, MASSACHUSETTS

NUS Corporation

EPA Work Assignment No. 02-1L15 EPA Contract No. 68-W8-0117 NUS Project No. 0214

May 1992

Walter J. Martin Project Manager

GeorgeJD. Gardner Program Manager

PAGE

E-1

1-1 1-1 1-2 1-2 1-3 1-7 1-10

1-10 1-17

2-1 2-2 2-2

2-9

2-11 2-16 2-16 2-16 2-17

2-17

2-18

2-18

3-1 3-1 3-5 3-7 3-8 3-10 3-10 3-11 3-12

3-12

TABLE OF CONTENTS FINAL REMEDIAL INVESTIGATION REPORT NYANZA III - SUDBURY RIVER STUDY

E.0	EXECU	rive sum	(ARY
1.0	INTRO	DUCTION	
	1.1	Purpose	of the Report
	1.2	Site Bac	ckground
		1.2.1	Site Industrial History
		1.2.2	Study Area Description
		1.2.3	Nyanza Previous Investigations
	1.3	Specific	Cobjectives of the Remedial
		Investiç	gation
	1.4	Summary	of Remedial Investigation Activities
	1.5	Report (Organization
2.0	STUDY	AREA IN	ZESTIGATION
	2.1	Sudbury	River Study
		2.1.1	Sudbury River Surface Water Sampling
			and Analysis
~		2.1.2	Sudbury River Sediment Sampling and
			Analysis
		2.1.3	Biota Sampling and Analysis
		2.1.4	Biological Survey
	2.2	Eastern	Wetlands Sampling and Analysis
		2.2.1	Physiography
		2.2.2	Eastern Wetlands Surface Water
			Sampling and Analysis
		2.2.3	Eastern Wetlands Sediment Sampling
			and Analysis
	2.3	Chemical	Cold Spring Book Culverts/Raceway
		Sampling	and Analysis
		2.3.1	Visual Inspection/Mapping
3.0	GENER	AL CHARAG	CTERISTICS OF THE SITE STUDY AREA
	3.1	Introduc	ction
	3.2	Surface	Features
		3.2.1	Surface Water Use
		3.2.2	Land Use
	3.3	Climate	
	3.4	General	Geology
		3.4.1	Soils
		3.4.2	Bedrock
	3.5	Surface	Water Hydrology
		3.5.1	Hydrological Setting

W92194F

SECTION

TABLE OF CONTENTS FINAL REMEDIAL INVESTIGATION REPORT NYANZA III - SUDBURY RIVER STUDY

BECTION

4.0

.

PAGE

3.6	Sudbury	River	3-13
	3.6.1	Hydrology	3-13
	3.6.2	Sediments/Bathymetry	3-14
	3.6.3	Wetlands	3-16
	3.6.4	Ecology	3-17
3.7	Eastern	Wetlands	3-22
	3.7.1	Hydrology	3-22
	3.7.2	Sediments	3-22
	3.7.3	Ecology	3-24
3.8	Chemica	l Brook Culvert	3-24
	3.8.1	Hydrology	3-24
	3.8.2	Sediments	3-25
	3.8.3	Ecology	3-25
NATU	RE AND EXT	TENT OF CONTAMINATION	4-1
4.1	Introduc	ction	4-1
	4.1.1	Monthly Water Quality Sampling Results	4-1
	4.1.2	Summary Tables	4-42
4.2	Potentia	al Sources of Contamination	4-51
	4.2.1	Nyanza Site-Related Contaminants	4-51
	4.2.2	Contamination Documented at the Nyanza	4-52
		Site	
	4.2.3	Off-Site Contaminant Sources	4-52
4.3	Results	from Analysis of Data: Surface	4-59
	Water/Se	ediment	
	4.3.1	Background Sample Locations	4-59
	4.3.2	Site Drainage System and Tributaries	4-60
	4.3.3	Sudbury River	4-63
4.4	Result s	from Analysis of Data: Fish Tissue	4-75
	4.4.1	Summary Tables	4-75
	4.4.2	Inorganic Contaminants	4-75
	4.4.3	Organic Contaminants	4-76
4.5	Benthic	Survey Results	4-77
	4.5.1	Grain Size/TOC	4-84
4.6	Inverte	brate Tissue Analysis	4-85
4.7	Eastern	Wetlands Soll Boring Program	4-86
	4.7.1	Background	4-86
	4.7.2	Samples	4-86
	4.7.3	Surficial Geology	4-87
	4.7.4	Analvtical Results	4-87

ii

TABLE OF CONTENTS FINAL REMEDIAL INVESTIGATION REPORT NYANZA III - SUDBURY RIVER STUDY

.....

-

SECTION

<u>PAGE</u>

	4.8	Summary Contami	and Discussion of Distribution of nants in the Study Area	4-93
		4.8.1	Inorganic Contaminants in the Study Area	4-94
		4.8.2	Organic Contaminants in the Study Area	4-96
5.0	CONTA	MINANT F	ATE AND TRANSPORT	5-1
	5.1	Sedimen	it Transport in Aquatic Systems	5-1
	5.2	Persist	ence (T1/2)	5-3
	5.3	Partiti	oning Coefficients	5-3
	5.4	Bioconc	entration Potential (BCF)	5-4
	5.5	Fate an	d Transport Modeling	5~6
	5.6	Sedimen	t Transport	5-6
		5.6.1	Introduction	5-6
		5.6.2	Reach 1	5-7
		5.6.3	Reach 2	5-9
		5.6.4	Reach 3	5-9
		5.6.5	Reach 4	5-10
		5.6.6	Reach 5	5-10
		5.6.7	Reach 6	5-11
		5.6.8	Reach 7	5-11
		5.6.9	Eastern Wetlands	5-12
		5.6.10	Chemical Brook Culvert/Outfall Creek	5-12
		5.6.11	Raceway	5-13
	5.7	Chemica	1-Specific Fate and Transport	5-13
		5.7.1	Chromium	5-14
		5.7.2	Lead	5-15
		5.7.3	Mercury	5-16
		5.7.4	Polynuclear Aromatic Hydrocarbons	5-16
		5.7.5	Chlorinated Benzenes	5-17
		5.7.6	Pesticides	5-18
		5.7.7	Polychlorinated Biphenyls	5-19
		5.7.8	Chemical Fate and Transport Summary	5-20
6.0	BASEL	INE PUBL	IC HEALTH RISK ABSESSMENT	6-1
	6.1	Objecti	ves	6-1
	6.2	Hazard	Identification	6-2
		6.2.1	Selection of Study Area Chemicals of Concern	6-2
		6.2.2	Toxicity Profiles	6-25
	6.3	Dose-Re	sponse Assessment	6-37
		6.3.1	Toxicity Criteria (Dose-Response	6-38
			Parameters)	
		6.3.2	Criteria/Standards ·	6-43

W92194F

•

TABLE OF CONTENTS FINAL REMEDIAL INVESTIGATION REPORT NYANZA III - SUDBURY RIVER STUDY

.

SECTION

PAGE

	6.4	Exposur	e Assessment	6-45
		6.4.1	Sources of Contamination	6-45
		6.4.2	Contaminant Transport and Migration	6-47
		6.4.3	Receptor Identification and Exposure	6-48
			Routes	
	6.5	Risk Ch	aracterization	6-59
		6.5.1	Methodology for Estimation of	6-59
			Carcinogenic Risks	
		6.5.2	Methodology for Estimation of	6-61
			Noncarcinogenic Risks	
		6.5.3	Risk Assessment Results	6-62
	6.6	Summary	of Risk Assessment Results	6-157
	6.7	Uncerta	inty Analysis	6-173
7.0	ECOLO	OGICAL RI	SK ASSESSMENT	7-1
	7.1	General	Approach	7-1
	7.2	Biologi	.cal Description	7-2
		7.2.1	Vegetation	7-2
		7.2.2	Animals	7-7
		7.2.3	Sensitive, Threatened, or Endangered	7-7
			Species	
		7.2.4	Indicator Species for Evaluating	7-7
			Food Chain Effects	
	7.3	Selecti	on of Chemicals of Concern (COCs)	7-9
		7.3.1	Potential Chemicals of Concern	7-9
		7.3.2	Selection of Surface Water COCs	7-9
		7.3.3	Selection of Sediment COCs	7-16
		7.3.4	Selection of Biota COCs	7-18
	7.4	Exposur	e Assessment	7-18
		7.4.1	Fate and Transport of COCs	7-19
		7.4.2	Exposure Pathways	7-19
		7.4.3	Future Exposures	7-21
		7.4.4	Surface Water Exposure Assessment	7-24
		7.4.5	Sediment Exposure Assessment	7-26
		7.4.6	Biota Exposure Assessment	7-27
		7.4.7	Uncertainty Analysis, Exposure	7-46
			Assessment	
	7.5	Hazard	Assessment	7-48
		7.5.1	Benchmark Toxicity Values - Surface	7-51
			Water COCs	
		7.5.2	Benchmark Toxicity Values - Sediment	7-51
			COCs	
		7.5.3	Benchmark Toxicity Values - Biota COCs	7-54

TABLE OF CONTENTS FINAL REMEDIAL INVESTIGATION REPORT NYANZA III - SUDBURY RIVER STUDY

SECTION

PAGE

	7.5.4 Uncertainty Analysis, Hazard	7-56
	Assessment	
7.6	Risk Characterization	7-60
	7.6.1 General Approach	7-60
	7.6.2 Cumulative Toxicity	7-61
	7.6.3 Risk from Surface Water COCs	7-61
	7.6.4 Risk from Sediment COCs	7-62
	7.6.5 Risk from Biota COCs	7-76
	7.6.6 Uncertainty Analysis, Risk	7-95
	Characterization	
7.7	Ecological Risk Assessment Summary and	7-98
	Conclusions	
CONC	LUSION	8-1
8.1	Contaminants in Water and Sediment	8-1
8.2	Contaminants in Fish	8-2
8.3	Risk Assessment	8-2

~ REFERENCES

8.0

APPENDICES

APPENDIX

VOLUME

A	RIVER STUDY SURFACE WATER AND SEDIMENT	II
	ANALYTICAL RESULTS	
В	MONTHLY SAMPLING - ANALYTICAL RESULTS	II
С	FISH TISSUE - ANALYTICAL RESULTS	II
D	EASTERN WETLAND SAMPLING - ANALYTICAL RESULTS	II
Ε	FISH SURVEYS - U.S. FISH AND WILDLIFE SERVICE	III
F	INORGANIC METALS IN CADDISFLY TISSUE - U.S. EPA	III
	MEMORANDUM	
G	WETLANDS ASSESSMENT FOR THE SUDBURY RIVER BETWEEN	III
	ASHLAND AND SAXONVILLE. MASSACHUSETTS - NUS CORP.	
н	CHEMICAL BROOK CULVERT INSPECTIONS - NUS CORP.	III
	MEMORANDUM	
I	"BATHYMETRY AND SEDIMENT DISTRIBUTION IN	III
-	RESERVOIRS 1 AND 2. FRAMINGHAM, MASSACHUSETTS"	
	U.S.G.S. OFR 91-508	
-	WITHITE AND ANTWAL COPOIDE ITEM	ттт
J	WILDLIFE AND ANIMAL SPECIES LIST	<u>+++</u>
K	SUPPLEMENTAL INFORMATION FOR THE RISK ASSESSMENT	III
	CALCULATIONS	
L	SUMMARY REPORT - MONTHLY WATER QUALITY DATA	III

TABLE OF CONTENTS (Continued) FINAL REMEDIAL INVESTIGATION REPORT NYANZA III - SUDBURY RIVER STUDY

TABLES

NUMBER

3 .

P	A	G	E

1-1	SEDIMENT SAMPLING PARAMETERS	1-12
1-2	SURFACE WATER SAMPLING PARAMETERS	1-16
2-1	SURFACE WATER FIELD DATA SUMMARY	2-4
2-2	MONTHLY MONITORING FLOW DATA	2-8
2-3	FISH SAMPLING SUMMARY	2-13
3-1	POPULATION OF TOWNS ALONG THE SUDBURY RIVER	3-2
3-2	SUDBURY RIVER WATERSHED LAND USES, 1951 - 1981	3-9
3-3	CRITICAL MEAN CURRENT VELOCITY	3-15
3-4	SUDBURY RIVER FISH	3-19
3-5	ENDANGERED, THREATENED AND SPECIES OF SPECIAL	3-21
	CONCERN ALONG THE SUDBURY RIVER	
4-1	SURFACE WATER AND SEDIMENT ANALYTICAL RESULTS	4-2
4-2	COMPARISON OF PARAMETERS IN SURFACE WATER TO	4-22
	STANDARDS	
4-3	SUMMARY OF FISH ANALYTICAL RESULTS: FILLET BY SPECIES	4-24
4-4	SUMMARY OF FISH ANALYTICAL RESULTS: FILLET BY REACH	4-43
4-5	BENTHIC SAMPLES SUMMARY DATA-PHASE I	4-78
4-6	GRAIN SIZE. TOC. DISSOLVED OXYGEN IN SEDIMENT	4-79
4-7	SUMMARY OF ANALYTICAL RESULTS/EASTERN WETLANDS	4-89
	BORING PROGRAM	
5-1	SUMMARY OF HYDRAULIC DATA	5-8
6-1	CHEMICALS OF CONCERN	6-3
6-2	TOXICITY CRITERIA FOR CHEMICALS OF CONCERN SELECTION	6-10
6-3	SELECTION OF CHEMICALS OF CONCERN FOR SEDIMENTS	6-15
6-4	SELECTION OF CHEMICALS OF CONCERN FOR SURFACE WATER	6-18
6-5	SELECTION OF CHEMICALS OF CONCERN FOR FISH (FILLETS)	6-26
6-6	TOXICITY CRITERIA FOR COC RISK CHARACTERIZATION	6-39
6-7	REGULATORY REQUIREMENTS FOR CHEMICALS OF CONCERN	6-42
6-8	SUMMARY OF INPUT PARAMETERS FOR EXPOSURE SCENARIOS	6-53
6-9	RISK ASSESSMENT RESULTS/SEDIMENT EXPOSURE SCENARIOS	6-64
6-10	RISK ASSESSMENT RESULTS/SURFACE WATER EXP. SCENARIOS	6-66
6-11	FISH INGESTION EXPOSURE SCENARIOS/SUDBURY RESERVOIR	6-69
6-12	FISH INGESTION EXPOSURE SCENARIOS/SOUTHVILLE POND	6-70
6-13	FISH INGESTION EXPOSURE SCENARIOS/CEDAR SWAMP POND	6-71
6-14	SEDIMENT EXPOSURE SCENARIOS REACH NO. 2	6-74
6-15	SURFACE WATER EXPOSURE SCENARIOS REACH NO. 2	6-76
6-16	FISH INGESTION EXPOSURE SCENARIOS/MILL POND	6-79
6-17	RISK ASSESSMENT RESULTS/SEDIMENTS/EASTERN WETLANDS	6-81
6-18	RISK ASSESSMENT RESULTS/SEDIMENTS/WETLANDS DRILLING	6-83
<	DICK DECUT DE (CUDE) OF MARED (EA CREEN, MERT ANDE	6-95

W92194F

TABLE OF CONTENTS (Continued) FINAL REMEDIAL INVESTIGATION REPORT NYANZA III - SUDBURY RIVER STUDY

TABLES

NUMBER

.

<u>PAGE</u>

6-20	RISK RESULTS/SEDIMENTS/CHEMICAL BROOK CULVERT	6-90
6-21	RISK RESULTS/SEDIMENTS/OUTFALL CREEK	6-93
6-22	RISK RESULTS/SURFACE WATER/OUTFALL CREEK	6-95
6-23	RISK ASSESSMENT RESULTS/SEDIMENTS/RACEWAY	6-98
6-24	RISK RESULTS FOR SURFACE WATER/RACEWAY	6-100
6-25	RISK ASSESSMENT RESULTS/SEDIMENT/COLD SPRING BROOK	6-103
6-26	RISK RESULTS/SURFACE WATER/COLD SPRING BROOK	6-105
6-27	RISK ASSESSMENT RESULTS/SEDIMENT/REACH NO. 3	6-108
6-28	RISK RESULTS/SURFACE WATER/REACH NO. 3	6-110
6-29	RISK RESULTS/FISH INGESTION/RESERVOIR NO. 2	6-113
6-30	RISK RESULTS/SEDIMENT/REACH NO. 4	6-115
6-31	RISK RESULTS/SURFACE WATER/REACH NO. 4	6-117
6-32	RISK RESULTS/FISH INGESTION/RESERVOIR NO. 1	6-120
6-33	RISK RESULTS/SEDIMENT/REACH NO. 5	6-122
6-34	RISK RESULTS/SURFACE WATER/REACH NO. 5	6-124
6-35	RISK RESULTS/SEDIMENT/REACH NO. 6	6-127
6-36	RISK RESULTS/SURFACE WATER/REACH NO. 6	6-129
6-37	RISK RESULTS/FISH INGESTION/SAXONVILLE IMPOUNDMENT	6-132
6-38	RISK RESULTS/SEDIMENT/REACH NO. 7	6-134
6-39	RISK RESULTS/SURFACE WATER/REACH NO. 7	6-136
6-40	RISK RESULTS/SEDIMENT/REACH NO. 8	6-138
6-41	RISK RESULTS/SEDIMENT/REACH NO. 9	6-141
6-42	RISK RESULTS/SEDIMENT/REACH NO. 10	6-144
6-43	RISK RESULTS/FISH INGESTION/FAIRHAVEN BAY	6-146
6-44	RISK RESULTS/SEDIMENT/BORDERING WETLANDS	6-148
6-45	RISK RESULTS/SEDIMENT/BORDERING WETLANDS: RECREATIONAL	6-150
6-46	RISK RESULTS/SEDIMENT/BORDERING WETLANDS:RESIDENTIAL	6-153
6-47	RISK RESULTS/SEDIMENT/HEARD POND	6-155
6-48	SUMMARY OF RISK RESULTS	6-158
7-1	SUMMARY OF STUDY AREA PLANT AND ANIMAL SPECIES	7-4
7-2	SURFACE WATER COCS	7-10
7-3	SEDIMENT COCS	7-11
7-4	BIOTA COCS	7-13
7-5	ENVIRONMENTAL PERSISTENCE FOR FINAL COCS (ALL MEDIA)	7-20
7-6	SUMMARY OF EXPOSURE PATHWAYS	7-23
7-7	REPORTED MEAN AND MAX. CONCENTRATIONS OF SURFACE	7-25
	WATER COC (FILTERED INORGANICS), BY LOCATION	
7-8	REPORTED MEAN AND MAX. CONCENTRATIONS OF SEDIMENT	7-28
	COC, BY LOCATION	

.

TABLE OF CONTENTS (Continued) FINAL REMEDIAL INVESTIGATION REPORT NYANZA III - SUDBURY RIVER STUDY

TABLES

NUMBER

<u>PAGE</u>

7-9	PARTITIONING COEFFICIENTS FOR ORGANIC SEDIMENT COCS	7-33
7-10	INTERSTITIAL WATER CONCENTRATIONS OF ORGANIC	7-34
	SEDIMENT COC	
7-11	DETECTED MEAN AND MAX. CONCENTRATION OF BIOTA COC	7-37
7-12	SPECIFIC FISH LENGTH, WEIGHT, AND AGE DATA	7-40
7-13	PREDICTED BAFS FOR FOOD CHAIN ORGANISMS	7-45
7-14	BENCHMARK TOXICITY VALUES FOR SURFACE WATER COC	7-52
7-15	BENCHMARK TOXICITY VALUES FOR SEDIMENT COC	7-55
7-16	BENCHMARK TOXICITY VALUES FOR BIOTA COC	7-57
7-17	RISK ESTIMATES-SURFACE WATER COC	7-63
7-18	RISK ESTIMATES-INORGANIC COC IN SEDIMENT	7-66
7-19	RISK ESTIMATES-ORGANIC COC IN SEDIMENT	7-72
7-20	RISK ESTIMATES-BIOTA COCS	7-78
7-21	TOTAL MERCURY CONCENTRATIONS IN WHOLE BODY FISH	7-80
7-22	ESTIMATED TISSUE CONCENTRATIONS OF BIOTA COC FOR	7-88
	LEVEL 1,2, AND 3 TROPHIC LEVELS	
7-23	ESTIMATED DIETARY CONCENTRATIONS OF BIOTA COC FOR	7-90
	REPRESENTATIVE PREDATORS	

FIGURES

NUMBER PAGE STUDY AREA MAP E-3 ES-1 STUDY AREA MAP 1-4 1-1 1-A SUDBURY RIVER STUDY-SAMPLING LOCATIONS, REACH 1-7 VOL. IV SUDBURY RIVER STUDY-SAMPLING LOCATIONS, REACH 8-10 VOL. IV 1-B VOL. IV EASTERN WETLANDS SAMPLING LOCATION 1-D LOCATION OF WELL-DEVELOPED URBAN ZONES AND MAJOR 3-1 3-3 HIGHWAYS 3-2 REMEDIAL INVESTIGATION STUDY AREA LOCATION MAP 3-6 3-3 CROSS SECTION A-A', EASTERN WETLANDS 3-23 PHASE 1 & 2 SEDIMENT SAMPLE RESULTS, REACH 1-7 VOL. IV 2-X PHASE 1 & 2 SEDIMENT SAMPLE RESULTS, REACH 8-10 VOL. IV 2-B VOL. IV PHASE 1 & 2 SEDIMENT SAMPLE RESULTS, ALONG 2-C REACHES 2 & 3 2-D PHASE 1 & 2 SEDIMENT SAMPLE RESULTS, EASTERN VOL. IV WETLANDS PHASE 1 & 2 SURFACE WATER SAMPLE RESULTS VOL. IV 3-A 4-1 LOCATION OF LANDFILLS 4-54 4-2 LOCATION OF POTENTIAL CONTAMINATION SOURCES . 4-56

W92194F

PAGE

TABLE OF CONTENTS (Continued) FINAL REMEDIAL INVESTIGATION REPORT NYANZA III - SUDBURY RIVER STUDY

FIGURES

MERCURY AND CHROMIUM AVERAGE CONCENTRATIONS IN 4-3 4-95 SEDIMENTS 7-1 ECOLOGICAL RISK ASSESSMENT COMPONENTS 7-3 7-2 FOOD WEB FOR SELECTED REPRESENTATIVE ORGANISMS 7-8 7-3 PROCESS FOR SELECTION OF MEDIA-SPECIFIC COCS 7-14 7-4 EXPOSURE PATHWAYS 7-22 7-5 MEAN & MAX. TOTAL MERCURY CONCENTRATIONS IN SEDIMENT 7-84 AND LARGEMOUTH BASS (NORMALIZED FOR AGE) 7-6 MEAN & MAX. TOTAL MERCURY CONCENTRATIONS IN SEDIMENT 7-85 AND LARGEMOUTH BASS

W92194F

NUMBER

EXECUTIVE

•

LO EXECUTIVE SUMMARY

A Remedial Investigation Feasibility Study (RI/FS) of Nyanza Operable Unit III, the Sudbury River, Middlesex County, Massachusetts is being conducted by the NUS Corporation under the ARCS I contract for the Region I U.S. Environmental Protection Agency (EPA). The RI was conducted from October 1989 through February 1992. Results of the RI are presented in this report. The Feasibility Study report will be presented in a separate document.

The RI was conducted to assess contaminants in the Sudbury River basin in the vicinity and downriver of the Nyanza Chemical Waste Dump Site in order to: determine the nature and extent of contamination in the River; evaluate public health and ecological risks associated with observed contamination; provide data to develop general response objectives and cleanup standards focusing on sediment cleanup; and provide data to develop and evaluate remedial alternatives to mitigate the defined risks.

The Nyanza Chemical Waste Dump Site occupies a 35-acre parcel of land located in the town of Ashland, Massachusetts. Beginning in 1917 the property has been occupied by several companies involved in the manufacture of various dye and chemical products. Nyanza, Inc., for whom the Site is named, was the most recent dye lanufacturing company to occupy the Site. Nyanza, Inc. appears to have ceased operations at the Site in 1978. Documentation indicates that several types of organic and inorganic chemical wastes were disposed in various locations at the Site. These wastes included partially treated process wastewater and sludge, solid process wastes, solvent distillation residues, off specification dyes, intermediate dye products acids and nonrecyclable products.

As early as 1967, the Commonwealth of Massachusetts began enforcement activities in connection with Nyanza Inc's wastewater and sludge disposal practices. Later investigations conducted by contractors for the EPA detected elevated levels of mercury and other chemicals linked to the Nyanza Site in sediments, biota, and water in the Sudbury River.

In 1982, the Nyanza Site was classified by the EPA as a National Priority List Site. An RI/FS for the Site was completed in 1985 leading to the signing of the Record of Decision (ROD) in September 1985. The ROD divided the Agency's remedial response into Operable Units for the purpose of addressing distinct problems. Operable Unit I dealt with the consolidation of on-Site and off-Site metallic sludge deposits, and the construction of a cap to secure Site related sludges on-Site. The construction was completed in November 1991.

115 94F

Operable Unit II is addressing groundwater contamination at the Site. An RI/FS has been completed. Results indicate that a groundwater plume containing organic contaminants has migrated 0.5 miles downgradient of the Site in a northeasterly direction to the vicinity of the Sudbury River. The FS proposed remedial alternatives and the EPA has selected an Interim Groundwater Cleanup Plan to mitigate this plume.

Operable Unit III was established in 1987 to address surface water and sediment contamination in the Sudbury River and its tributaries (Figure ES-1). Field investigation activities for the RI began in 1989 and include the following:

- o Collection of surface water and sediment samples throughout the Study Area to determine the nature and extent of contamination in each medium. Sampling activities were conducted during two major events.
- Collection of fish and invertebrate biota throughout the Study Area in two major events.
- Completion of a soil boring program in the Eastern Wetlands area adjacent to the Nyanza Site for the purpose of delineating the nature and extent of contamination in this area.
- Collection of water quality samples on a monthly basis to define seasonal water quality fluctuations. Data was evaluated to determine if aquatic organisms are seasonally more vulnerable to toxic contaminants.

The RI resulted in the following conclusions:

- 1) Nature and Extent of Contamination
 - Chlorobenzene, dichlorobenzene, trichloroethene, and dichloroethene were detected in sediments and soils within the Eastern Wetlands at concentrations in the range of 10⁴ parts per billion (ppb). These compounds were detected immediately downstream of the Wetlands, but were not detected in the Sudbury River.
 - o The highest concentrations of mercury in surface sediments were detected in the Eastern Wetlands. The maximum concentration detected is 424 parts per million (ppm). Mercury concentrations generally decrease in the Sudbury River to the confluence of the Assabet River in Concord, where concentrations were comparable to background levels detected upstream of the Nyanza Site.

FINAL



~~

- Mercury and chromium were detected in River sediments throughout the Study Area downgradient of the Nyanza Site. Concentrations of these metals are higher in depositional areas (reservoirs and impoundments) and are lower in channelized river areas. The highest concentrations of mercury in the River sediments occur in Reservoir 2 where over 90 percent of the samples collected detected mercury at an average of 17.43 ppm. The highest concentration detected in Reservoir 2 was 54.6 ppm.
- o Mercury and chromium were detected in surface water samples in the Eastern Wetlands and several River water samples, but at low concentrations.
- Fish samples collected throughout the Study Area revealed Ο the presence of mercury and methylmercury in the tissues Yellow Largemouth Bass and Perch. While of concentrations of these compounds depended mainly on age of the fish, capture location was also a factor. Fish from Reservoir 2 showed collected the highest concentrations of mercury, and fish collected from upstream locations (background), showed less, as did those from Fairhaven Bay. Pesticides and PCBs were also bioaccumulated by fish.
- Site specific organic compounds used in the Nyanza dye manufacturing process, including aniline, napthalamines, benzidine and related compounds, were not detected in the Study Area.
- A variety of non Site-related contaminants were detected in River sediments including pesticides, polynuclear aromatic hydrocarbons, volatile organic contaminants, phthalate compounds, and metals.
- Many point and non-point sources of contaminants, which may contribute to contamination in the Sudbury River, are present throughout the Study Area.
- 2) Fate and Transport of Contaminants

An analysis of the fate and transport of contaminants found in the Study Area has revealed the following:

 Contaminant transport is provided mainly by the physical transport of contaminated sediment in surface water bodies. Contaminants which adsorb to particulate matter will be transported in flowing waters and deposited in bed sediments.

- o Contaminants in sediments are expected to continue to impact surface waters in the Eastern Wetlands if chemical conditions allow transfer. Heavy metals such as mercury and chromium are not readily desorbed from sediments in the Sudbury River as evidenced by TCLP and ACOE extraction analyses which model environments which would leach contaminants from sediments.
- Contaminated finer grained sediments tend to be wide spread in the reservoirs and impoundments where low water velocities allow deposition of the sediment.
- 3) Public Health Risk Assessment
 - o There are two categories of individuals which are susceptible to health effects of contaminants found in the Study Area. These individuals are those who regularly consume fish caught in the Study Area and those who use parts of the Study Area for wading and swimming.
 - Hazard quotients and hazard indices calculated for recreational and residential sediment exposure scenarios do not exceed unity. The exception is in the Eastern Wetlands, where unity is approached if the receptor is a small child.
 - Accidental ingestion hazard quotients and hazard indices exceed those for dermal absorption. The hazard index for bordering wetlands exceeds unity when maximum concentrations are considered and the receptor is a small child.
 - Maximum or average mercury concentrations detected in fish tissue throughout the Study Area exceed the Food and Drug Administration action level for mercury in fish. The only exception is fish caught in Southville Pond (in a Background area).
 - Adverse noncarcinogenic health effects are anticipated for sports and subsistence fishermen who consume fish caught in the Study Area.
- 4) Ecological Risk Assessment
 - o The toxicity hazards associated with Site-related contaminants are minimal compared to the hazards of contamination bioaccumulation in the food chain.

- o Mercury is the only Site-related contaminant which contributes significantly to bioaccumulation in the food chain.
- Mean estimated whole body concentrations of total mercury in Largemouth Bass and Yellow Perch collected throughout the Study Area exceed values (established by the Fish and Wildlife Service) that are considered to be protective of predators that consume fish.

LIST OF ACRONYMS AND ABBREVIATIONS

-

_--

AAWQC	Acute Ambient Water Quality Criteria
ABS	Absorption rate
ACL	Alternate Concentration List
ACOE	U.S. Army Corps of Engineers
ACR	Acute to Chronic Ratio
ACRe	Estimated Acute to Chronic Ratio
ADC	Average Detected Concentration
am	Before noon
AOC	Area of Concern
АРА	Air pathway analysis
 ARARs	Applicable or relevant and appropriate requirements. A law or regulation that should be applied in a NPL site cleanup.
ARCS	Alternative Remedial Contracting Strategy
ASTM	American Society of Testing and Materials
ATSDR	Agency for Toxic Substances and Disease Registry
AVG	Number of years over which an exposure is averaged
AWQC	Ambient Water Quality Criteria
BAF	Bioaccumulation Factor
BCF	Bioconcentration Factor
BDL	Below Detection Limits
BNAS	Base-neutral and acid compounds
BOD	Biochemical Oxygen Demand
Btu	British Thermal Unit
JW	Receptor body weight

C	Measured/estimated media concentration
°C	degree Centigrade (Celsius)
CAA	Clean Air Act
CAWPC	Chronic Ambient Water Quality Criteria
сс	cubic centimeters
CERCLA	The Comprehensive Environmental Response, Compensation and Liability Act of 1980. Amended by SARA in 1986. Also called the Superfund law.
CF	Volumetric Conversion Factor
cf	cubic feet
cfm	cubic feet per minute
cfs	cubic feet per second
CLP	(EPA) Contract Laboratory Program
cm	Centimeter
CNS	Central Nervous System
COC	Chemicals of Concern
COD	Chemical Oxygen Demand
CR	Contact Rate or Cancer Risks
cm/sec	centimeters per second
CSF	Carcinogenic Slope Factor
AWD	Clean Water Act
су	cubic yard
D	Deposition rate
dB	decibel
DCB	Dichlorobenzene
DCE	Dichloroethene
DDD	1,11-(2,2-Dichloroethylidene)-bis/4-chlorobenzene/
DDE	1,1 ¹ -(2,2-Dichloroethenylidene)-bis/4~chlorobenzene/

DDT	l,ll-(2,2,2-Trichloroethylidene)-bis/4- chlorobenzene/
DEP	Massachusetts Department of Environmental Protection (formerly DEQE)
DEQE	Massachusetts Department of Environmental Quality Engineering (now the DEP)
dia	diameter
DIUF	De-ionized ultra-filtered water
DOD	U.S. Department of Defense
DOE	U.S. Department of Energy
DOI	U.S. Department of Interior
DOT	U.S. Department of Transportation
DPH	Massachusetts Department of Public Health
DQO	Data quality objectives
DWPC	Massachusetts Division of Water Pollution Control
ئ ے	Entrainment rate
ED	Exposure duration
EEC	Estimated Environmental Concentration
EF	Exposure frequency
EPA	U.S. Environmental Protection Agency
ER	Exposure rate
ERi	Ratio of risk for each chemical/total risk
ER-L	Effects Range-Low
ER-M	Effects Range-Median
ESD	EPA's Environmental Services Division
٥ _F	degree Fahrenheit
FDA	U.S. Food and Drug Administration
FID	Flame Ionizing detector
FI	Fraction of consumed fish

foc	Fraction of organic carbon in sediment/soil
FOL	Field Operations Leader
fpm	feet per minute
fps	feet per second
FS	Feasibility Study. See RI/FS.
FSP	Field Sampling Plan
ft	feet
9	gram
GAC	Granulated Activated Carbon
gal	gallon
gpd	gallons per day
gph	gallons per hour
gpm	gallons per minute
GWPS	Groundwater Protection Strategy
НА	Health Advisory
HASP	Health and Safety Plan
HEAST	EPA Health Efficts Assessment Summary Tables
HI	Hazard Index
HNUS	Halliburton NUS Environmental Corporation
hp	horsepower
HPLC	High Pressure Liquid Chromotography
HQ	Hazard Quotient
hr	hour
I.D.	inside diameter
IDW	Investigation-derived Waste
in.	inch
in/hr	inches per hour

ŢR	Ingestion Rate
IRIS	Integrated Risk Information System
٥ _K	degree Kelvin
K	Hydraulic conductivity
Kcal	Kilocalorie
Kd	Sediment or soil/water partitioning coefficient
kg	kilogram
kHz	Kilohertz
km	kilometer
Кос	Partitioning coefficient for chemical/organic carbon
Kow	Partitioning coefficient for octanol in water at STP
1	liter
1b	pound
EC	Lowest effects concentration
LEL	Lower Explosive Limit
ln ft	linear foot
LOEC	Lowest-observed effects concentration
LOAEL	Lowest-observed adverse effects level
m	meter
u	micro (prefix)
MCL	Federal Safe Drinking Water Act maximum contaminant level. The primary MCL is health-based; the secondary is aesthetic-based.
MCLG	Federal Safe Drinking Water Act maximum contaminant level goal.
MDC	Massachusetts Metropolitan District Commission
mg	milligram
igd	million gallons per day

-

mg/kg	milligram per kilogram	
mg/L	milligram per liter	\smile
mi	mile	
mL	milliliter	
MOM	Management of Migration	
mph	miles per hour	
msl	mean sea level	
NAAQS	National Ambient Air quality Standards	
NCP	National Contingency Plan	
NEPA	National Environmental Policy Act	
NF	Net flux	
NIOSH	National Institute of Occupational Safety and Health	
NOAA	National Oceanic and Atmospheric Administration	
NOAEL	No-observed adverse effects concentration	
NOEL	No-observed effects concentration	•
NPDES	National Pollution Discharge Elimination System	
NPL	(EPA's) National Priority List. A list of sites identified for remediation under CERCLA.	
NUS	NUS Corporation (Halliburton NUS Environmental Corporation as of July 1991)	
0.D.	outside diameter	
0&M	Operations and Maintenance	
OS	Oversight	
OSHA	Occupational Safety and Health Administration	
OSWER	(EPA's) Office of Solid Waste and Emergency Response	
ORD	(EPA's) Office of Research and Development	
AVO	Organic Vapor Analyzer	
OVM	Organic Vapor Meter	

oz	ounce
HAG	Polynuclear aromatic hydrocarbons
PCB	Polychlorinated Biphenyl
PC	Permeability constant
PCDF	Polychlorinated Dibenzofuran
pCi	picocurie
PCOC	Potential Chemical of Concern
PDI	Pre-Design Investigation
PEL	Permissible Exposure Level
PH	hydrogen-ion concentration
PID	photo-ionizing detector
pm	afternoon
POTW	Publicly Owned Treatment Works
dqc	parts per billion
PPE	Personal Protection Equipment
ppm	parts per million
psf	pounds per square foot
psi	pounds per square inch
PVC	Polyvinyl Chloride
QA/QC	Quality Assurance/Quality Control
RA	Remedial Action
RAF	Absorption factor for soils/sediment
RD/RA	Remedial Design/Remedial Action
RAS	Routine Analytical Service
RCO	Regional Contracting Officer
RCRA	Resource Conservation and Recovery Act
_ D	Remedial Design

~

- RfD Reference Dose
- Ri Risk factor for a specific chemical in an environmental medium

RI Remedial Investigation

- RI/FS Remedial Investigation/Feasibility Study. The first comprehensive phase in cleaning up an NPL site. An RI involves field work to determine the nature and extent of site contamination. The FS analyses that data and recommends strategies to clean it up.
- ROD (EPA's) Record of Decision. Documents the selection of a cost-effective Superfund remedy.
- RPM Regional Project Manager
- RPO Regional Project Officer
- RQD Rock Quality Designation
- SAP Sampling and Analysis/Quality Assurance Plan
- SAS Special Analytical Service
- SARA Superfund Amendments and Reauthorization Act of 1986. Amended CERCLA. Also known as the Superfund law.
- SCBA Self-Contained Breathing Aparatus
- SCR Soil/sediment contact rate
- SCS (U.S. Department of Agriculture) Soil Conservation Service

SDWA Safe Drinking Water Act

Site, the The National Priority List site, as defined by EPA.

SMO Sampling Management Office

SMP Site Management Plan

SOW Statement of Work

SPHEM Superfund Public Health Evaluation Manual

sq ft square feet

sq in square inch

sq yd square yard

	SSA	Contact rate
_	JVOC	Semi-volatile Organic Compound
	T _{1/2}	Half-life
	TAL	(CLP) Target Analtye List for metals
	TBC	To Be Considered
	TCE	Trichloroethene
	TCL	(CLP) Target Compound List for Organics
	TCLP	Toxicity Characteristic Leaching Process
	TLV	Threshold Limit Value
	тос	Total Organic Carbon
	TSCA	Toxic Substances Control Act
	TSD	Treatment, Storage, or Disposal under RCRA
	UEL	Upper Exposure Limit
	UF	Uncertainty factor
/	ug/kg	microgram per kilogram
	ug/L	microgram per liter
	USAF	U.S. Air Force
	USFWS	U.S. Fish and Wildlife Service
	USGS	U.S. Geological Survey
	USN	U. S. Navy
	VOA	Volatile Organic Analysis
	VOC	Volatile Organic Compound
	WP	Work Plan
	WWTP	Wastewater treatment plant



1.0 INTRODUCTION

1.1 <u>Purpose of the Report</u>

This report presents the results of a detailed investigation of the Sudbury River and associated surficial drainage routes from the Nyanza Site (Nyanza) in the Towr of Ashland, Massachusetts, to the confluence of the Sudbury with the Assabet River in Concord. This Remedial Investigation was prepared in accord with the Final Work Plan, dated August 1989, and the Sampling and Analysis Plan, dated April 1990, for the United States Environmental Protection Agency (EPA) Region I, under Work Assignment No. 02-1L15, Contract 68-W8-0117. The Feasibility Study is under development and will be submitted to the EPA under separate cover.

The Nyanza Site was included on the original National Priorities List (NPL) in 1982. To expedite remediation, the Remedial Investigation/Feasibility Study (RI/FS) for Nyanza was originally List (NPL) in 1982. divided into separate units. Operable Unit I, for which an RI/FS has been completed, a Record of Decision (ROD) signed, and most of the remedial construction activities completed, addressed surficial deposits of sludges and sediments containing elevated levels of heavy metals, including arsenic, chromium, mercury, cadmium, and lead. A ROD has also been signed for Operable Unit II, referred to as "Nyanza II - Groundwater Study." This study addresses groundwater contamination within the former Nyanza property boundary and determines the presence of offsite migration. Operable Unit III, the focus of this Remedial Investigation, is referred to as "Nyanza III - Sudbury River Study." It addresses contamination in the Sudbury River by discharges of wastewater and sludge from the Nyanza Site.

The Nyanza III RI was conducted to determine the nature and extent of chemical contamination of surface water, sediment, and biota, and to gather sufficient information to determine the need for and extent of remedial action. To achieve these objectives, data was collected:

- 1. to assess the nature and distribution of contaminants in surface water, sediments, and fish
- 2. to perform a public health and environmental risk assessment
- 3. to develop response objectives
- 4. to support the evaluation of remedial alternatives

The data and conclusions presented in this report will be used to develop the Feasibility Study (FS), an evaluation of possible remedial alternatives to accomplish the response objectives. A detailed analysis of each alternative will be provided in the FS.

W92194F

The Nyanza III - Sudbury River Study Remedial Investigation has been performed consistent with the requirements of the National Contingency Plan (NCP) and the Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA) of 1980, as amended by the Superfund Amendment and Reauthorization Act (SARA) of 1986. This Remedial Investigation report has been prepared in accord with the latest U.S. Environmental Protection Agency RI/FS Guidance (EPA, 1988) and Data Quality Objective (DQO) Guidance (EPA, 1987).

1.2 <u>Site Background</u>

This section describes the setting of the Site and Study Area, and outlines the Site history and previous investigations within the Study Area. In accord with the National Environmental Policy Act, Section 1500.4, Reducing Paperwork, this document minimizes redundancy of printed information through use of previous studies and reports by reference in lieu of reprinting information readily available from other sources.

1.2.1 Site Industrial History

The Nyanza Site was occupied from 1917 through 1978 by several companies involved in manufacturing textile dyes and dye intermediates. During that period large volumes of chemical wastes were disposed into burial pits, structures including "The Vault" a below ground containment structure, and lagoons at various locations throughout the Hill section and what is now the lower industrial area. Wastes include partially treated process wastewater, chemical sludge from the wastewater treatment process, solid process wastes (chemical precipitate and filter cakes), solvent recovery distillation residue, acids, numerous organic and inorganic chemicals (including mercury) and off-specification products. Process chemicals that could not be recycled or reused (phenol, nitrobenzene, and mercuric sulfate) were also disposed onsite or discharged to the Sudbury River.

Nyanza, Inc., the most recent dye manufacturer at the location, operated from 1965 until 1978, when it ceased business due to financial difficulties. The property was purchased by Edward Camille, a private citizen, in 1978. In 1981, most of the property was acquired by MCL Development Corporation, which leases a large portion of the Site to Nyacol Products, Inc. Mr. Camille and three other small property owners currently operate or lease facilities in the lower industrial area to various light industries and commercial concerns. These include Ashland Industrial Fuel Corporation, Middlesex Equipment, Ashland Excavating Co., A Auto Body, Environmental Restoration Engineering Co., and others.

FINAL

1.2.2 Study Area Description

The Nyanza Superfund Site is located on Megunko Road in Ashland, Massachusetts, in Middlesex County, approximately 22 miles west of Boston. The Site was the location of chemical dye manufacturing facilities for 61 years and is currently occupied by several small industrial enterprises.

The Nyanza Operable Unit III was established as a study of the Sudbury River drainage basin (Figure 1-1) in May 1989. For purposes of this report, the term "Nyanza Site" or "Site" will refer to the 35-acre former Nyanza, Inc., Site.* The term "Study Area" will refer to the 31 miles of the Sudbury River drainage basin from the headwaters at Cedar Swamp Pond in Westborough to the confluence of the Sudbury and Assabet Rivers which form the Concord River in Concord, Massachusetts.

Also included in the Study Area are the "Raceway" (a canal serving the historic mill district in Ashland); Eastern Wetland and Trolley Brook, which drains this ponded wetland (both are recipients of surface water to the east of the Nyanza Site); Cold Spring Brook; and Chemical Brook Culvert. Chemical Brook Culvert is a recipient of flow from Trolley Brook and Chemical Brook, which are drainage routes from the Site. Refer to Figures 1-A and 1-B in Volume IV of this report for depictions of these areas.

The Sudbury River flows in a northerly direction through rolling, hilly terrain. Most of the surrounding area is suburban residential, consisting of several closely spaced urban centers connected by arterial commuting routes to Boston. The Great Meadows National Wildlife Refuge (a total of approximately 3,000 acres) in Sudbury, Wayland, Concord, and Lincoln, Massachusetts borders approximately 6 miles of the Sudbury River (Figure 1-1).

For the purposes of this report, the Study Area has been separated into ten sections termed "Reaches", which are defined by, among other parameters, changes in river configuration, impounding structures, and stream junctures. Tributaries, such as Chemical Brook Culvert, Trolley Brook and Eastern Wetland, and Cold Spring Brook are discussed separately. Reaches are numbered 1 through 10, which follow in order from upstream to downstream. The limits of each Reach are outlined below and in Figures 1-A and 1-B.

^{*} The Site consists of a 35-acre parcel on Megunko Road in Ashland, Massachusetts formerly owned and operated by Nyanza, Inc. (the Nyanza Property) as well as all areas that have come to be contaminated with hazardous substances emanating from the Nyanza Property.



.

Reach 1 extends from Cedar Swamp in Westborough, upstream of the Nyanza Site, to the small dam (referred to as the Pleasant Street Impoundment) downstream of the Pleasant St. river crossing and upstream of Mill Pond in Ashland. This area has been determined to be upgradient of Site-related surficial drainage and the groundwater plume. Reach 1 is considered a background Reach.

FINAL

Reach 2 extends from the Pleasant Street impoundment to the Union Street (Route 135) bridge in Ashland. This Reach is directly impacted by Site discharges at and downstream of Mill Pond, and is geographically the most proximate of the Reaches to the Site. Mill Pond is the only impoundment where sediment carried by the River may be deposited in this Reach. Other depositional areas in Reach 2 are typically along the River's edge, behind obstructions or inside the River bends. There are fewer of these depositional areas in Reach 2 than in the reservoirs immediately downstream. Present in the geographical region, but not included in Reach 2, are Eastern Wetlands, the Chemical Brook and Cold Spring Brook Culverts, and the Raceway, which is a canal paralleling the River through Ashland. These features are not included in Reach 2 discussions and the data sets will be discussed individually later in this section.

Reach 3 includes Reservoir No. 2 and ends at Dam No. 2, which divides Reservoirs 2 and 1. Reach 3, being a relatively deep, slow moving body of water, is a primary sedimentation area directly downstream of Reach 2.

Reach 4 includes Reservoir No. 1 and is bounded by Dam No. 1 at the Winter Street crossing in Framingham. Reach 4 has characteristics similar to Reach 3, the dams act as physical barriers to upstream fish migration, although some downstream migration of contaminants may occur.

Reach 5 extends from Dam No. 1 to the Massachusetts Turnpike (Interstate 90) overpass where the River widens. The Sudbury River is typically narrow in this vicinity, with higher flow velocities than the impounded reservoirs; some channelization has occurred through developed areas. As a result, the River in Reach 5 has less depositional potential than Reservoirs 1 and 2; however, some localized depositional areas are present due to the low flow velocity and low gradient areas present.

Reach 6 extends from the Massachusetts Turnpike to the Saxonville Dam at the Central Street crossing. Reach 6 includes the depositional area of ponded water behind the Saxonville Dam.

Reach 7 extends from the Saxonville Dam downstream to the Route 20 overpass in Wayland. Downstream of the Saxonville Dam (the last impoundment), a low stream gradient (less than 1 foot per mile)

W92194F

results in a slow, meandering river with high depositional potential.

Reach 8 extends from the Route 20 overpass in Wayland to the area of the Route 117 overpass, prior to the inlet at Fairhaven Bay. This Reach includes the Great Meadows National Wildlife Refuge. The River's meandering course and extensive bordering swamps and marshes characterize this Reach as having high depositional potential.

Reach 9 is Fairhaven Bay, a large pond-like feature in the River. Fairhaven Bay is also a potential depositional area.

Reach 10 extends from the outlet of Fairhaven Bay to the confluence of the Sudbury and Assabet Rivers.

The RI also addresses contamination in several locations within the Study Area which are not included in the ten Reaches of the Sudbury River:

- o The Eastern Wetlands are treated separately because they actually constitute a headwaters area of a small tributary of the River. They also receive surface runoff directly from the Nyanza Site (Figure 1-D).
- Chemical and Trolley Brooks were the primary surficial drainage routes from the Site and Eastern Wetlands before remedial activities were conducted under Operable Unit I. Chemical Brook was remediated under Operable Unit I. The Brooks merged and discharged through a subsurface culvert (Chemical Brook Culvert) which discharged to the Raceway, downstream of the Concord Street overpass in Ashland, and to the Sudbury River (Figure 1-A, Detail).
- o The Raceway is a man-made canal which channelizes a portion of the River flow from a flow-control gate at Mill Pond into a culvert which passes beneath a large mill building. The Raceway changes to an open canal downstream of the mill. The Outfall Creek carrys water from the Chemical Brook culvert discharge and in turn, discharges to the open part of the raceway. The raceway rejoins Reach 2 of the Sudbury River downstream of the confluence with the Outfall Creek.
- A second culvert, Cold Spring Brook Culvert, is also located within Reach 2. It discharges to Cold Spring Brook, a tributary of the Sudbury River (Figure 1-A, Detail) near the Main Street crossing in Ashland. This culvert was referred to in previous reports as the Trolley Brook Wetlands Culvert.

W92194F

1-6

FINAL

- The Sudbury Reservoir in Southboro, Massachusetts was sampled as a background location. The reservoir is located on a tributary which discharges to the main stem of the Sudbury River, and therefore, can be used as a background location for comparison with elevated downstream levels of inorganics and organics (Figure 1-A). Data sets from this Reservoir are included with Reach 1 data sets to comprise background data sets.
- Heard Pond, located in Reach 7, is considered separate from the other features of Reach 7. The Pond is, on occasion, reportedly flooded by the River, however, the two are not otherwise linked by surface water.

1.2.3 Nyanza Previous Investigations

This section briefly describes previous investigations relevant to the Nyanza III RI/FS. Several investigations were conducted within the Nyanza III - Sudbury River Study Area by various state and federal agencies from 1972 to the present. Most of the previous studies concentrated on the Nyanza Site; however, several studies addressed, in a cursory fashion, assessments of the Sudbury River downstream of the Site. The following paragraphs present a brief overview of the investigations most relevant to this study.

1.2.3.1 <u>Previous Investigations - Nyanza Site</u>

Investigation of the Nyanza Site began in 1972, when the Massachusetts Departments of Public Health (DPH) and Water Pollution Control (DWPC) cited Nyanza, Inc., for several waste disposal violations. Investigations and interim groundwater remedial activities have continued to the present, with a source control remedial action (Operable Unit I) currently in progress. A summary of major events in the investigative history of the Nyanza Site includes the following:

- o <u>1972-1977</u>: Massachusetts DPH and DWPC cited Nyanza for several waste disposal violations.
- o <u>1974</u>: Camp Dresser and McKee, Inc. (CDM) completed an environmental site investigation for Nyanza, Inc. The study focused on the identification of sources of contamination found on- and off-site. A plan to control groundwater contamination was developed but not implemented.
- o <u>1980</u>: Massachusetts Department of Environmental Quality Engineering (DEQE) (currently called the Massachusetts Department of Environmental Protection [DEP]) performed a preliminary site assessment (SA). DEQE also released

W92194F

a report summarizing findings of previous studies.

- o <u>1981</u>: Connorstone Engineering/Carr Research Labs performed a site characterization study for MCL Development Corporation, the new Nyanza Site owners.
- o <u>1982</u>: The Nyanza Site was listed on the Superfund National Priorities List (NPL); CDM developed a Remedial Action Master Plan (RAMP) for EPA, emphasizing on-site source control remedial actions.
- o <u>1984-1985</u>: NUS performed an RI/FS on Operable Unit I (on-site surficial soils, sediments, and sludge deposits); a ROD based on this RI/FS, recommending source removal and stabilization/landfilling activities, was signed in 1985.
- o <u>1986</u>: CDM conducted additional on-site investigations to locate source areas and support a remedial design of the ROD-specified actions.
- <u>1987</u>: EPA Region I Environmental Services Division (ESD) and DEQE performed a sludge removal action for a buried concrete tank, "the vault", at the Nyanza Site. The vault was a major source of organic chemical groundwater contamination. E.C. Jordan began RI/FS activities on Operable Unit II (a groundwater assessment study). Planning was initiated for Operable Unit III (Sudbury River Study).
- o <u>1989</u>: Source control measures were implemented at the Nyanza Site by the U.S. Army Corps of Engineers using Tricil Environmental Response, Inc.

These investigations and remedial activities indicated that on-site soils, sludges, and sediments were contaminated primarily with heavy metals, including mercury, chromium, lead, and cadmium. These surficial deposits were remediated as part of the Operable Unit 1 remedial action. However, a contaminated groundwater plume containing elevated levels of various organic chemicals trichloroethylene, chlorobenzene) has (nitrobenzene, been identified migrating in a northeasterly direction from the Site; it may be a source of organic compound contamination to the Sudbury River. Organic contaminants may also reach the River through surficial runoff from secondary sources such as the Eastern Wetland and Trolley Brook. A more detailed evaluation of these contaminant source transport pathways and their relationship to contamination in Sudbury River water, sediments, and biota is presented in Section 4.

W92194F
1.2.3.2 <u>Previous Investigations - Sudbury River</u>

Previous Sudbury River contamination studies included sediment, surface water, and biota investigations at various locations from upstream of the Nyanza Site, downstream to the Concord River.

Investigations of Sudbury River contamination related to the Nyanza Site have included the following:

- <u>JBF</u> <u>Scientific</u>, <u>Inc.</u>: JBF performed initial investigations addressing elevated levels of mercury in Sudbury River water, sediments, and biota in 1972. Elevated levels of mercury were detected, and were qualitatively linked to uncontrolled sludge disposal at the Nyanza property.
- U.S. Fish and Wildlife Service (USFWS): USFWS began monitoring fish in the Sudbury River for elevated mercury levels in 1977. Sampling events continued through 1987. Several species of fish were analyzed to determine the potential for bioaccumulation of mercury. Some sampling events also included sediment sampling. Analyses of organic contaminants included concentrations in sediments of polychlorinated biphenyls (PCBs), pesticides, and polynuclear aromatic hydrocarbons (PAHs).
- DEQE/MDC Metropolitan District Commission: DEQE performed several environmental investigations of assessments of the Sudbury River from 1980 to 1987. These studies included fish, surface water, and sediment sampling and analysis for metals, with limited additional organic analyses. The DEQE data were incorporated into an MDC Sudbury Reservoir Water Treatment Plant Draft Environmental Impact Report (1982). In addition, DEQE was involved in a cooperative biota/sediment monitoring effort with USFWS.

These studies have indicated that sediments and fish in the Sudbury River contain elevated levels of mercury and that sediments contain elevated concentrations of other metals, most notably lead and chromium. The highest elevated levels of inorganics (metals) and organic compounds were generally found in the River downstream of the Site, from Reach 2 to the Saxonville Dam, located in Reach 6. Mercury concentrations in approximately 25 percent of the fish tested exceeded the Food and Drug Administration (FDA) action level of 1 mg/kg (ppm). As a result, the DPH issued a Health Advisory in 1986 recommending that fish caught in the Sudbury River not be eaten. Organic analyses in previous investigations have been limited and do not provide a definitive basis for assessment of organic compound contamination.

A detailed discussion of the data generated during this RI concerning inorganic and organic chemical contamination in fish, surface water, and sediment in the Sudbury River is included in Section 4.0.

1.3 <u>Specific Objectives of the Remedial Investigation</u>

The objectives of the Nyanza III - Sudbury River Study RI are:

- o Defining of the nature and extent of contamination in the sediments, surface water, and biota of the Sudbury River
- To the extent possible, determining the distribution of contamination discharged from the Nyanza Site versus contamination attributable to other sources
- Assessing the public health and ecological risks associated with the elevated levels of inorganic and organic contaminants observed in the sediments, surface water, and biota

The Feasibility Study will:

- Develop general response objectives and Site clean-up standards focusing on possible sediment remediation goals appropriate to the observed contamination
- Develop and evaluate remedial alternatives focusing on possible sediment cleanup to mitigate or eliminate the defined risks

1.4 <u>Summary of Remedial Investigation Activities</u>

Section 2 of this RI report presents a detailed description of the field investigation activities.

The large scale river system addressed by the Nyanza III - Sudbury River Study dictated a two-phased approach to meeting RI objectives. The phased approach provided the greatest flexibility as well as the collection of the most applicable data.

<u>Phase I</u>

Phase I data collection focused on preliminary characterization of the nature and distribution of contamination in the water, sediment, and biota in the River system. Phase I also included a one-year, monthly sampling program to identify trends and fluctuations in the water chemistry of the River system. Following Phase I, a Technical Directive Memorandum was issued which identified the preliminary findings of the study, as well as a

W92194F

FINAL

preliminary qualitative assessment of the risks to public health and the ecology of the system.

FINAL

Phase I included sampling 50 sediment station locations. Sediment samples were collected at the sediment surface at all the 50 stations and at a depth interval from six to twelve inches at 14 of these stations.

All 64 sediment samples were analyzed for Target Compound List (TCL) metals and methyl mercury. Twenty-two of these samples were also analyzed for additional parameters, as specified in Table 1-1. With the exceptions of SW3-101 and SW3-103, these 22 samples were collected at locations corresponding with the 22 surface water locations. SW3-101 and SW3-103 were not co-located with sedient samples because a laboratory was not available to analyze sediment samples when these water samples were collected.

Unfiltered surface water samples were collected from 22 of the 50 sediment stations. In addition, filtered surface water samples were collected from 11 of the 22 stations. Surface water samples were analyzed for the parameters presented in Table 1-2.

Biota sampling during Phase I included the collection of three species of fish from four impoundments of the River. These included Reservoirs No. 1 and 2, the Saxonville Dam impoundment, and the Sudbury Reservoir (background location). Biota sampling also included a qualitative assessment of benthic organisms at River sediment sampling locations.

Phase I work included a one-year periodic surface water sampling program (January through December 1990) which involved monthly collection of surface water samples from five locations. These samples were analyzed for metals and water quality parameters.

<u>Phase II</u>

Phase II of the investigation included additional sampling to further define the nature and extent of contamination. The goal of this phase was to fill data gaps identified following the completion of Phase I and to provide a more complete database to identify risks.

Phase II samples were analyzed for the parameters listed in Table 1-1 and 1-2, Section 2.0. A detailed discussion of analytical parameters is presented in the Phase II Final Sampling and Analysis Plan (SAP) Amendment No. 2, October 1990.

The Eastern Wetlands were sampled during Phase II activities to provide information on the horizontal and vertical extent of contaminants in the wetland sediments. A total of 23 locations

W92194F

LOCATION

SUDBURY RIVER

щ
21

					NYANZ		BLE UNIT 3							
		•		MIDD	LESEXC	UUNIT, N	ASSACHU	SEIIS						
	STATION	MORGANIC	METHYL-	ORGANICS O	ROANICS	METALS	METALS	BIZE	pH & En	тос	BOD	COD	TKN	HEX." CHROM, VI
PHASE I	803 - 101	×	X			<u> </u>							t/-	
	SD3 - 102	X	X	X	X	x	X	X	X	X	x	x	X	
	SD3 - 103	X	X ·											
	SD3 - 104	X	x											x
	SD3 - 111	X	x											
	SD3 - 115	X	x	X	x	X	X	x	X	x	x	X	X	
	SD3-118	X	X											
	8D3 - 117	X	X	X	X	x	x	X	X	X	x	x	х	X
	SD3 - 1 18	×	X	X	X	×	x	X	x	X	x	x	X	
	SD3-118+	×	X											
	SD3 - 119	×	x	X	X	x	X	X	X	X	x	x	X	
	SD3 - 120	X	×	X	X	X	X	x	X	x	x	x	X	x
	SD3 - 120+	X	x					x						
	5D3 - 121	X	X											x
	SD3 - 122	X	x	X	X	X	x	X	x	X	x	x	x	x
	SD3 - 123	X	X											
	SD3 - 124	X	X	x	X	x	x	X	x	X	x	x	X	x
	SD3 - 125	x	×	X	X	x	x	x	X	x	x	X	X	
	8D3 - 125+	X	x											
	SD3 - 126	X	×											
	SDJ - 127	x	x											
	SD3 - 128	X	×											
	SD3 - 129	X	x	X	x	x	X	X	X	X	x	x	X	
	SD3 - 130	X	X	X	X	x	x	x	x	x	X	X	X	
	SD3-130+	×	X											
	SD3 - 131	×	X											
	SD3 - 132	X	X	X	x	x	x	x	x	x	x	x	X	
	SDJ - 132+	x	x											
	8D3-133	x	x			x	x							
	8D3 - 134	x	x											
	503 - 135	X	X	x	x	x	x	x	x	x	x	x	x	
	803-135+	X	X											
	SD3 - 130	x	X	X	X	x	x	x	x	x	x	x	x	
	803 - 137	x	Â				n							
	803-138	x												
	503 - 138+	x x	x	x	X			x			x	x	x	
	803-139			x	Ŷ	x	x	Ŷ	x	x	Ŷ	x	x	
	803 - 140	Ŷ	x x	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	x	Ŷ	Ŷ	
	801-140+	Ŷ	Ŷ	~	^	~	~	~	~	~	~	~	~	
	503-141	Ŷ	Ŷ											
	801-1414	Ŷ	Ŷ											
	801-142	Ŷ	Ŷ											
	801-147	Ŷ	Ŷ											
	901 - 14J	Ç	Ŷ											
	SDJ - 144	÷	÷											
	003-143 003 144	, , , , , , , , , , , , , , , , , , ,	÷	v	v	Y	v	v	v	¥	¥	¥	¥	
	803-14 8	÷	Ŷ	~	~	^	~	~	~	^	^	^	^	
	801.44	÷	÷	Y	~	¥	~	v	Y	¥	¥	¥	¥	
	803-148		÷	Ŷ	÷	× ×	Ş	÷	÷.	Ŷ	Ŷ	Ŷ	÷	
	aus - 149	. 5.	~	<u>^</u>	<u>^</u>	<u> </u>	····· ··· · · · · · · · · · · · · · ·	·^	^	<u> </u>	···· • • ····	^	· · · ^	

.

TABLE 1-1 BEDIMENT SAMPLING PARAMETERS

FINAL

TABLE 1-1

SEDIMENT SAMPLING PARAMETERS

NYANZA OPERABLE UNIT 3

MIDDLESEX COUNTY, MASSACHUSETTS PAGE TWO

W92194F

LOCATION STATION INORGANIC MERCURY ORGANICS ORGANICS METALS METALS SIZE TOC BOD COD TKN SUDBURY RIVER PHASE I 603-150 X	HEX." CHROM. V
SUDBURY RIVER PHASE I 603 - 150 X<	
SUDBURY RIVER PHASE # SD3 - 152 X	
603-153 X X X X X X SD3-153+ X X X X X X SD3-153+ X X X X X X SD3-154+ X X X X X X SD3-154+ X X X X X X SD3-156 X X X X X X SD3-157 X X X X X X SD3-158 X X X X X SD3-162 X X X X SD3-163 X X X X	
SD3-153+ X X X X X X SD3-154 X X X X X X X SD3-154 X X X X X X X SD3-154 X X X X X X X SD3-154+ X X X X X X SD3-156 X X X X X X SD3-157 X X X X X X SD3-158 X X X X X SD3-162 X X X X X SD3-163 X X X X	
8D3 - 154 X X X X X X X 8D3 - 154 X X X X X X X 8D3 - 154 X X X X X X X SD3 - 156 X X X X X X X SD3 - 156 X X X X X X SD3 - 157 X X X X X SD3 - 158 X X X X SD3 - 162 X X X SD3 - 163 X X X	
BD3 - 154+ X X X X X X X SD3 - 156 X X X X X X X SD3 - 156 X X X X X X X SD3 - 157 X X X X X X SD3 - 158 X X X X SD3 - 162 X X X SD3 - 163 X X X SD3 - 163 + X X X	
SD3-156 X X X X X X SD3-157 X X X X X X SD3-158 X X X X X SD3-162 X X X X SD3-163 X X X SD3-163+ X X X	
SD3-157 X X X X X X X X X SD3-158 X X X X SD3-162 X X X SD3-163 X X X SD3-163 X X X	
SD3 - 158 X X SD3 - 162 X X SD3 - 163 X X SD3 - 163 + X X	
SD3-162 X X X SD3-163 X X X SD3-163+ X X X	
SD3-163 X X X SD3-163+ X X X	
SQ3-163+ X X X	
SD3-185 X X X	
SD3-185+ X X X	
SD3-166 X X X	
SD3~166+ X X X	
SD3-168 X X X	
SD3 - 189 X X X	
SO3-170 X X X	
SD3-170+ X X X	
SD3-171 X X X X	
S03 - 172 X X X	
S03-173 X X X	
803-185 X X X	
8D3-196 X X X	
5D3-106+ X X X	
8D3-187 X X X	
8D3-186 X X X	
8D3-188+ X X X	
8D3-189 X X X	
SD3 - 190 X X X	
5D3 - 191 X X X	
SD3-192 X X X	
SD3-192+ X X X	
8D3-193 X X X	
\$D3-194 X X X	

Ч

TABLE 1-1

SEDIMENT SAMPLING PARAMETERS

NYANZA OPERABLE UNIT 3

PAGE THHEE		····· · ···			METHYL-	tci	eitë	ACOE	TCLP	GRAIN	oH & Eh			•••••		HEX. *
LOCATION			STATION	INORGANIC	MERCURY	ORGANICS	ORGANICS	METALS	METALS	SIZE		TOC	BOD	COD	TKN	CHROM. VI
SUDBURY P	RIVER	PHASE I	SD3 - 195	- x		•	·			×	· · · · · · · · · · · · · · · · · · ·	X				
			803 - 195+	X						X		X				
			SD3 - 196	X						X		X				
		•	SD3 - 197	X						X		X				
			8D3 - 190	X						X		x				
			503 - 199	x						X		X				
			SD3 - 199+	X						X		X				
			8D3 - 202	X		X	X			X		X	x	x		
			8D3 - 203	X		X	X			X		X	X	X		
			5D3 - 204	X		x	X			X		X	x	x		
			5113-207	÷						, X		X				
			SD1 - 200	Ŷ						÷		Ŷ				
			503-210	Ŷ						Ŷ		Ŷ				
			503 - 242	~		¥	x			Ŷ		Ŷ				
			503 - 242+			x	x			x		Ŷ				
			6D3 - 243			x	X			x		x				
			SDJ - 243+			X	X			X		x				
BOBDERI	NG	PHĂŜE I	\$0j - 113	. X	x											
WETLAN	DS		8D3 - 114	x	x											
		PHASE #	SD3 - 155	X						x		x				
			8D3 - 155+	x						X		X				
			\$D3 - 159	X						х		x				
			8D3 - 159+	X						X		x				
			SDJ - 161	X						х		X				
			8D3 - 161+	X						X		x				
			SD3 - 167	X						х		X				
			\$DJ 167+	X						x		x				
RACEWA	AY .		SDJ - 244	X		x	x			X		X	• • • • • • • • • • • • • • • • • • • •			
			503-244+	X		X	X			X		X				
			SD3 - 245	X		x	X			х		X				
			5D3 - 245+	X		X	X			x		×				
0010 000		DUACE :	ébi wi		~											
BROOK	(PHASE	\$D3 160	x	^	x				x		x				
FASTER	IN	PHÁSĚ Ì	803-108	·· 🖬 ·	¥		¥	¥	Y Y		<u>-</u>		x	Ý	¥	
WETLAND	DS		SD3 - 106+	Ŷ	x	-	~	~	~	^	^	~	^	~	^	
			503 - 107	x	x											
			8D3 - 107+	X	x											
			8D3 - 108	x	X											
			SD3 - 108+	X	X											
			SD3 - 109	X	X											
			8D3 - 109+	x	x							x	x	x	x	
			8D3 - 110	×	x	x	X	X	x	X		x	x	X	x	
			8D3 - 110+	×	x											
		PHASE #	SD3 - 211 #	X												

FINAL

TABLE 1-1 SEDIMENT SAMPLING PARAMETERS

NYANZA OPERABLE UNIT 3

MIDDLESEX COUNTY, MASSACHUSETTS PAGE FOUR

METHYL-TCL BITE ACOE TCLP GRAIN pH & Eh HEX.* BOD LOCATION STATION INORGANIC MERCURY ORGANICS ORGANICS METALS METAL8 SIZE ----- TOC COD TKN CHROM, VI PHASE # 803-212 # EASTERN X X WETLAND9 803-213# X 803-214 # X SD3-215 # X SD3 - 216 # X X SD3-217 # SD3 - 218 SD3-219# SD3 - 220 # SD3-221 € X X 503-222 4 X X X 8DJ - 223 🌒 x SD3-224 # 8D3 - 225 d SD3 - 226 d X X SD3 - 227 🔮 X X SD3 - 228 6 X X X SD3 - 229 6 X X ¥ SD3 - 230 6 X X x SD3 - 231 🔮 X X 1 803-232 6 X X 8D3 - 233 🌒 X X SD3-234 6 X X x SD3-235 6 X X X SD3 - 238 🔮 X X X SD3 - 237 🔮 X X x SDJ - 236 🌒 X X X SD3 - 239 🌒 X X х X SD3 - 240 d X X X X SD3-241 6 x X X SD3 - 339 🌒 x CHEMICAL PHASE I SD3-112 X X X X X X X X X х х х BROOK PHASE II SD3 - 200 X X х x х х CULVERT SD3-201 X Х х X X х х X Х 8D3 - 205 х x х х SDJ - 206 X X X х х х

Notes: Station 5D3 - 151 not collected - Cedar Swamp Pond sediment - 100% organic material

Station SD3 - 164 delated - sample slots were used for sediment sampling at Phase II caddisfly sampling locations

Station 8D3 - 200 Grain Size was not collected due to low evallable sediment volume

* - Hexavalent Chromium collected at Phase I locations during Phase II sampling round

TCL ORGANICES include volatile, PCB/Pesticides, and BNA fractions analysis

SITE ORGANICS include Benzidine, Aniline, oONitroctorobenzene, 1 - Nephylemine, 2 - Nephthylamine, o - Tolidine, o - Dianisidine

ACOE = Army Corps or Engineers Elutriate Test

TCLP = Toxicity Characteristic Leeching Procedure

TOC = Total Organic Carbon

BOD = Biochemical Oxygen Demand

COD = Chemical Oxygen Demand

TKN = Total Kjeldahl Nitrogen HEX CHROM VI = Hexavalent Chromium # = 2 feet total depth - two profiles collected

@ = 6 leet total depth - four profiles collected X = Analysis was performed

+ = 6 to 12 inch profile

1 і б

TA	BL	E	1	-2
	_		•	

SURFACE WATER SAMPLING PARAMETERS NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

		STATION	TCL	SITE OBGANICS	TCL	FILTERED	METHYL-	Alkalinity-TOC-I	BOD-COD-TDS-TSS-TKN	nH	00	HARDNESS
								unfiltered	filtered	Held	field	(calculated)
SUDBURY RIVER	PHASE I	SW3-101	X		x -		<u>x</u>	×		- <u>x</u> -	<u> </u>	X
		SW3-102M	x	x	x		x	i x		x	X	X
		SW3-107	x	x	x	x	x	X	x	X	X	X
		SW3 - 106M	x	x	x		x	x		x	X	x
		SW3 - 109	X	X	X		x	x		x	x	x
		SW3-110	X	x	x		X	X		x	X	X
		SW3 - 111	x	X	x		X	x		x	x	x
		SWJ- 112	x	x	x	x	x	X	x	x	X	x
		SW3-113M	x	X	X		x	x		x	х	x
		8W3 - 114	X	X	X	x	x	x	x	x	x	x
		SWJ-115	x	x	X		x	X		X	х	x
		SWJ - 116	X	x	X		x	X		x	x	x
		SW3-117	x	x	x	X	x	X	x	x	х	x
		SW3 - 118	X	x	x	X	×	X	x	X	X	x
		SW3 - 119M	X	X	X	X	x	X	x	x	X	X
		SW3 - 120	X	x	X	X	x	X	x	X	х	x
		5W3 - 121	X	x	X	X	x	X	x	x	X	x
		5W3 - 122	X	x	x	X	x	X	X	X	x	x
	PHASE N	6W3-123	x	X	X	x		X	x	x	x	x
		SW3 - 124	X	x	X	X		X	X	х	X	x
		8W3 - 125	X	x	X	x		[X	x	X	x	x
		8W3 ~ 126	X	X	X	X		X	X	X	X	x
		SW3-127	X	X	X	X		X	X	X	X	x
		5WJ - 128	X	X	X	X		X	X	X	X	x
		SW3 - 129	X	X	X	X		X	X	X	X	X
		SW3 - 130	X	X	X	X		X	X	X	X	X
		5W3 - 131	X	X	X	X		X	X	X	X	X
		8WJ - 132	X	X	X	X		X	X	x	X	X
		SW3 - 133	X	X	X	X		X	X	X	X	X
		8W3 - 134	X	X	X	X				X	X	X
	• - • • • • • • •	8W3-135	X	X	×	×				<u>×</u>	. <u>X</u>	X
COLD SPRING BROOK	PHASE I	8W3 - 103	X	X	X		X	×		X	X	×
EASTERN	PHASE I	8W3 - 104	x	x	x	x	x	x	x	x	x	x
WETLANDS		8W3-105M	. <u>X</u>	<u>×</u>	×	<u>×</u>	<u>×</u>	×	<u> </u>	X	<u>×</u>	<u>×</u>
CHEMICAL BROOK	PHASE I	8W3-106	x	X	X	X	×	×	X	x	X	x
RACEWAY	PHASE N	8W3-138	x	x	x	×				x	x	x
								ι				

TCL Organics include volatiles, PCB/Pesticides and BNA fractions analyses

SITE ORGANICS Include Benzidine, Aniline, e- Mitchlorobenzene, 1 - Naphthylamine, 2 - Naphthylamine, o- Tolidine, o- Dianisidine

TOC = Total Organic Carbon

BOD = Biochemical Oxygen Demand

COD = Chemical Oxygen Demend

TOS = Total Dissolved Solids TSS = Total Suspended Solids TKN = Total Kjeldahl Nitrogen DO = Dissolved Oxygen X = analysis was performed M = Monthly sampling locations 1

were sampled to a total depth of approximately six feet, nine samples to a total depth of approximately two feet, and one sample to a depth of six inches in a small stream feeding the south end of the wetlands. These samples were analyzed for the parameters presented in Table 1-1.

Additional sampling was conducted at five Phase I sediment sampling locations which exhibited elevated levels of total chromium. These additional samples were collected to determine the presence of the hexavalent chromium.

Biota assessments during Phase II operations consisted of two major sampling events. Caddis fly larvae were collected by EPA from five locations along the River; these were analyzed for body burden concentrations of mercury, chromium, and lead. Two species of fish were collected by USFWS and analyzed for mercury and methyl mercury as well as other organic and inorganic contaminants. Phase II fish collection was performed at all Phase I fish collection locations (with the exception of the Sudbury Reservoir) as well as at additional background and downstream locations.

1.5 <u>Report Organization</u>

The general outline of this RI report follows that presented in the Interim Final USEPA Guidance for Remedial Investigations and Feasibility Studies Under CERCLA (1988). However, this format includes several modifications to better reflect the nature of the project.

Because of the Study Area expanse, the investigation report has been separated into discussions of: the Sudbury River, the Eastern Wetlands, the two tributaries and the Raceway. The Sudbury River has been further divided into reaches as defined in Section 1.2.1. The length of each reach was based on its geographic location and potential applicability of individual remedial action alternatives. Each area can be evaluated separately.

The order of this report has been divided by content. An Executive Summary and a list of abbreviations and acronyms specific to the Nyanza Operable Unit III investigation have been included as reader's aids. Sections 1.0 through 7.0 contain substantive information about the Study. The Introduction and other background information is contained in Section 1.0. The remaining Sections are summarized below:

o Section 2.0 Study Area Investigation

Section 2.0 of this report details the investigation activities performed as a part of Phase I and Phase II, including the surface water, sediments, biota sampling (fish and benthic organisms), the monthly water sampling, and the wetlands evaluation.

o Section 3.0 General Characteristics of the Study Area

The River system, including the wetlands, culverts, and the Raceway, is described in detail. This description includes the surface water hydrology, the bathymetry, and the ecology and surficial geology of the area.

o Section 4.0 Nature and Extent of Contamination

Section 4.0 describes the extent of contamination in the Study Area. The Sudbury River watershed has a long history of industrialization. This industrialization, as well as the population centers and transportation routes within the watershed, are likely to contribute to the contamination of the River. Therefore, Section 4.0 also presents a discussion regarding those contaminants which are likely to be specifically related to the Nyanza Site (Nyanza Site contaminants) and a discussion regarding contaminants from other sources (other Study Area contaminants).

o Section 5.0 Contaminant Fate and Transport

Section 5.0 discusses the various pathways, transport mechanisms, and fate of contaminants in the ecosystem. These include physical transport of contaminated media, transport of contaminants through the food chain by ecological receptors, and the degradation of contaminants in the environment.

o Section 6.0 Public Health Risk Assessment

The Public Health Risk Assessment, presented in Section 6.0, follows guidelines in EPA national and regional documents and presents quantifications of potential risks to human health by the contaminants found in the River system.

o Section 7.0 Ecological Risk Assessment

The Ecological Risk Assessment, presented in Section 7.0, provides analyses of food chain contaminant transfer and additional assessments of risks to receptor organisms. Particular discussion will focus on the potential affect to top predators within the food chain.



2.0 STUDY AREA INVESTIGATION

A number of investigations and environmental monitoring activities have been conducted in the Study Area prior to this investigation, as summarized in Section 1.2.3. These investigations identified the Nyanza Site as a source area of contamination affecting the Sudbury River. This Sudbury River RI builds on these studies to determine the nature and extent of contamination in the River system and to identify the risks to human health and the environment. This section presents a description of each of the field investigation tasks performed in the Study Area to meet the objectives of the RI.

Between September 1989 and December 1989, the following field activities were conducted under Phase I.

<u>Phase I</u>

- Sampling and analysis of surface water and sediment from selected River locations
- Sampling and analysis of surface water and sediment from selected Chemical Brook Culvert locations
- Monthly water sampling from selected River and Trolley Brook locations to define seasonal fluctuations in water chemistry
- o Sampling and analysis of fish from selected River locations
- Surveying benthic biota (population density count) in the Study Area
- o Assessing wetlands in the Study Area
- Surveying bathymetry and sediment thickness in Reservoirs
 No. 1 and No. 2 by the United States Geological Survey
 (USGS)
- Sampling and analysis of sediment within the Eastern Wetlands

A number of data gaps were identified at the completion of a preliminary evaluation of Phase I. The most significant finding was the presence of elevated levels of metals in sediments at the down-river limit of sampling. A Phase II work plan was then developed. This work was conducted from September 1990 through June 1991 and included:

- Expanding the Study Area to delineate the downstream extent of mercury contamination in river sediments
- Sampling and analysis of surface water and sediment from selected River locations
- o Sampling and analysis of sediment in the Eastern Wetlands
- Sampling and analysis of sediment from selected locations within the bordering wetlands of the River
- Sampling and analysis of sediment from selected locations within the Raceway which parallels the River
- Sampling and analysis of additional fish from selected River locations
- Sampling and analysis of caddis fly larvae at selected River locations
- Inspecting the Chemical Brook Culvert by remote video camera to identify areas of sediment deposition
- Sampling and analysis of sediments from the Chemical Brook Culvert

The discussion of the Study Area investigation has been separated into three sections, as described in Section 1.5. These include the Sudbury River; the Eastern Wetlands; and the source tributaries and Raceway. Each of the tasks, the sampling station locations, and the analytical parameters, are described in the following sections.

2.1 <u>Sudbury River Study</u>

2.1.1 Sudbury River Surface Water Sampling and Analysis

The surface water sampling and analysis program and sampling procedures described in Section 4.3.1.4 of the Final Work Plan (August 1989), Final Work Plan Amendment No. 2 (October 1990), and Final Work Plan Amendment No. 3 (June 1991), Sections 4.1.5 and 4.2.2, respectively of the Final Sampling and Analysis Plan (SAP, August 1989), and the Final SAP Amendment No. 2 (October 1990) and the Final SAP Amendment No. 3 (June 1991) address specific sampling methods. Quality control samples including field duplicates, trip blanks, and rinsate blanks were collected and analyzed for each sampling round as specified in EPA guidance (User's Guide to the Contract Laboratory Program, December 1988) and in the Final SAP and the SAP Amendment Memoranda. The analytical results for surface water sampling are discussed in Section 4.3 and are

presented in Appendix A, the Data Summary Analytical Database.

Surface water sampling was conducted for the following reasons: to determine the presence of contaminants; to evaluate surface water quality and potential migration of contaminants from sediments to surface water; to gather data to evaluate the appropriateness of surface water remedial action; and to provide data for the selection and design of remedial alternatives.

2.1.1.1 Phase I and Phase II Surface Water Sampling

The Phase I sampling program included an initial round of 18 unfiltered surface water samples collected from locations numbered SW3-101 through SW3-122, as presented in Figure 1-A. In addition, filtered surface water samples were collected from nine of these 18 locations. Phase I surface water sampling locations included areas upstream of the Nyanza Site, and extended down-river to the vicinity of Heard Pond in Wayland (Reaches 1 through 7). Sample locations were chosen to correspond to sediment sample locations and to provide representative information for each geographical portion of the River.

Surface water samples collected during Phase I were not analyzed for hexavalent chromium (chromium VI), as proposed in the Final Work Plan, since a Contract Laboratory Program (CLP) laboratory could not be identified to perform this analysis.

The Phase II surface water sampling included the collection of 11 unfiltered samples from stations numbered SW3-123 through SW3-133, as presented in Figure 1-A. These included additional background samples from Cedar Swamp Pond (in the headwaters area of the River), to the area directly downstream of the Saxonville Dam. Water samples were also collected at Phase II benthic sample stations and at "fill in" sample stations to provide complete coverage in Mill Pond and the Raceway.

Water quality parameters measured in the field during Phase I and II surface water sampling included pH, temperature, specific conductivity, and dissolved oxygen. Stream flow was also measured at each location. The surface water field data summaries are presented in Table 2-1.

All samples were submitted to the CLP for analysis. All surface water samples were analyzed for the parameters presented in Table 1-2. Phase II analytical parameters were selected to target selected metals and compounds known to have been discharged from the Site (Site contaminants), or were frequently detected during Phase I sampling.

TABLE 2-1 SURFACE WATER FIELD DATA SUMMARY NYANZA OPERABLE UNIT 3 **MIDDLESEX COUNTY, MASSACHUSETTS**

CAMPLE IDENTIFICATION	1 000 101 0101	1 ANNA 100 0001	1 01/2 100 0100	1 000 100 0000	10000 100 0101	0140 400 0005		
SAMPLEIDENTIFICATION	543=101-0/01	8WJ-102-0201	SWJ-102-0102	8W3-102-0203	SW3-102-0104	SW3-102-0205	SW3-102-0206	SW3-102-0207
UATE SAMPLED	11/30/09	11/01/09	01/08/90	02/00/90	03/06/90	04/04/90	05/08/90	06/04/90
		1 100			7.00			
		700	0.04	0.40	7.90		7.30	6.82
TEMPERATURE	0 135	0.150	0.10	1	0.22	0.22	0.258	0.21
DEMPERATURE	50	12.3	00	1.0	1.2	5.0	13.0	16.0
DISSOLVED OXTGEN	12.5	89	14.5	13.5	12.8	12.2	9.5	9.2
VELOCITY R/sec		NO DATA	} 0.1	04 _	NOT RECORDED	3.0	0.4	1.0
CAMPLE IDENTIFICATION				1 000 100 0000	1.0000 400 0040			
SAMPLE IDENTIFICATION	WJ=102=0100	WJ-102-0109	SW3-102-0110	SW3-102-0311	5WJ-102-0212	SW3-102-0113	SW3-103-0101	SW3-104-0101
DATE SAMPLED		08/07/90	09/15/90	10/01/90	11/08/90	12/06/90	11/01/89	11/27/89
рH	6 76	6 60	6 68	6.75	6.17	7.45	7.90	7 10
CONDUCTIVITY	0 227	0 236	0 231	0.271	0.191	0.211	0.16	0.22
TEMPERATURE	23.4	22.6	22.5	14.6	8.9	19	18.5	20
DISSOLVED OXYGEN	77	71	72		14.3	121	7.4	2.0
VELOCITY Blac	0.7	1 10	NOTTAKEN	0.0	10	12	0.0	
	1 01	1	1 HOL MEN	1. ••		L! <u>*</u>	0.0	I NO DATA
	1 8W3 - 105-0101	1 8W1 - 105-0302	1 991-105-0101	SW3-105-0104	SW1-105-0105	SW3-105-0106	SW3-105-0107	SW/1-105-010
DATE CAMPLED	11/77/89	01/08/00	02/05/00	01/06/00	04/04/00	05/08/00	040400	07/00/00
	· · · · · · · · · · · · · · · · · · ·		02/05/20		04/04/80		00/04/30	07/05/80
рH	7 20	6 76	8 25	9 20		6.28	6.93	6.68
CONDUCTIVITY	2 5	0 22	0 215	0 245	0.14	0.22	0.24	0.287
TEMPERATURE	30	25	05	22	7.0	15.0	23.8	22.7
DISSOLVED OXYGEN	• • •	10.5	64	10.25	10.0	9.1	12.5	54
VELOCITY ft/sec	NO DATA	04	00	NOT RECORDED	07	0.3	0.4	NODATA
	1	1 .	1	1	1	1	10.4	
SAMPLE IDENTIFICATION] #W3 - 105 - 0109	1 8W3-105-0110	1 8W3-105-0111	1 SW3-105-0112	SW3-105-0113	SW3-106-0101	SW3-107-0201	SW3-108-020
DATE SAMPLED	06/07/90	09/15/90	10/01/90	11/06/90	12/06/90	11/09/90	11/09/90	11/06/89
σH	6.69	6.91	6.73	6.50	7.32			7 50
CONDUCTIVITY	0 319	0 569	0 584	0 287	0 172	0.18	0.11	0 165
TEMPERATURE	24.2	215	18.5	84	1.8	90	90	80
DISSOLVED OXYOEN	4.75	1	80	13.2	10.0	10.0	5.0 to 0	21.0
VELOCITY #/sec	0.6	NOT TAKEN	0.0	NOT RECORDED	0.2	0.3	0.0	0.0
TELOVITT NAME			J	THOTHEOUNDED	l	0.0	U.I	L0.0
			•					
SAMPLE IDENTIFICATION	SW3-108-0202	8W3-108-0203	SW3-108-0204	SW3-108-0205	SW3-108-0106	SW3-108-0107	SW3-108-0108	SW3-108-030
DATE CAMPLED	0.0000	00.05.000	0.000000	04/04/00	05 000 000	000000	07/00.000	00/07/00

	SAMPLE IDENTIFICATION	5W3 - 100 - 0202	8W3-108-0203	SW3-108-0204	SW3-108-0205	SW3-108-0106	SW3-108-0107	SW3-108-0108	SW3-108-030
	DATE SAMPLED	01/08/90	02/05/90	03/06/90	04/04/90	05/08/90	06/04/90	07/09/90	08/07/90
	pH	6 92	6 33	6 60		7.00	6.75	6.80	6.64
ļ	CONDUCTIVITY	0211	0 22	0 22	0 215	0.219	0.23	0.23	0.221
	TEMPERATURE	18	00	-0 07	6.0	13.5	17.0	26.6	22.8
	DISSOLVED OXYGEN	130	12.4	13 2	15 3	9.8	9.0	7.4	4.7
	VELOCITY ft/sec	00	00	NOT RECORDED	10	NOT RECORDED	0.3	0.0	0.0

2

W92194F

FINAL

TABLE 2-1 SURFACE WATER FIELD DATA SUMMARY NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS PAGE 2

- 108 - 03 10 SV	V3-108-0411	SW3-108-0212	SW3-108-0113	SW3-109-0501	SW3-110-0401	SW3-111-0801	SW3-112-1201
9/15/90	10/01/90	11/08/90	12/06/90	11/02/89	11/02/89	11/01/89	11/08/89
I							
8 52	6 67	6 4 3	7 45	6.70	5.40	7.20	9.60
0 198	0 276	0 207	0.175	0.16	0.16	0.165	0.12
22 5	16 6	8.4	48	11.0	16.0	12.9	7.5
8 15	69	12 6	10.4	9.2	10.0	8.6	8.6
TTAKEN	0.1	0.3	0.4	0.0	NO DATA	0.1	0.0
	108-0310 SV N15/90 8 52 D 198 22 5 8 15 T TAKEN	108-0310 SW3-108-0411 N15/90 10/01/90 8 52 6 67 0 198 0 276 22 5 16 6 8 15 6 9 T TAKEN 0.1	108-0310 SW3-108-0411 SW3-108-0212 N/5/90 10/01/90 11/08/90 8 52 6 67 6 43 0 198 0 276 0 207 22 5 16 6 8.4 8 15 6 9 12 6 T TAKEN 0.1 0.3	108-0310 SW3-108-0411 SW3-108-0212 SW3-108-0113 N/5/90 10/01/90 11/08/90 12/06/90 8 52 6 67 6 43 7 45 0 198 0 276 0 207 0.175 22 5 16 6 8.4 4.8 8 15 6 9 12.6 10.4 T TAKEN 0.1 0.3 0.4	108-0310 SW3-108-0411 SW3-108-0212 SW3-108-0113 SW3-109-0501 10/01/90 11/08/90 12/06/90 11/02/89 11/02/89 8 52 6 67 6 43 7 45 6.70 6.70 0 198 0 276 0 207 0.175 0.16 11.0 815 6.99 12.6 10.4 9.2 T TAKEN 0.1 0.3 0.4 0.0 0.0	108-0310 SW3-108-0411 SW3-108-0212 SW3-108-0113 SW3-109-0501 SW3-110-0401 N/5/90 10/01/90 11/06/90 12/06/90 11/02/89 11/02/89 8 52 6 67 6 43 7 45 6.70 5.40 0 198 0 276 0 207 0.175 0.16 0.16 22 5 16 6 8.4 4.8 11.0 16.0 8 15 6.9 12.8 10.4 9.2 10.0 T TAKEN 0.1 0.3 0.4 0.0 NO DATA	108-0310 SW3-108-0411 SW3-108-0212 SW3-108-0113 SW3-109-0501 SW3-110-0401 SW3-111-0801 N/5/90 10/01/90 11/06/90 12/08/90 11/02/89 11/02/89 11/02/89 11/01/89 8 52 6 67 6 43 7 45 6.70 5.40 7.20 0 198 0 278 0 207 0.175 0.16 0.16 0.165 22 5 18 6 8.4 4 8 11.0 16.0 12.9 8 15 6 9 12 6 10 4 9.2 10.0 8.6 T TAKEN 0.1 0.3 0.4 0.0 NO DATA 0.1

SAMPLE IDENTIFICATION DATE SAMPLED	8W3 - 113 - 0101 11/29/89	8W3 - 113 - 0202 01/06/90	8W3-113-0103 02/06/90	SW3-113-0104 03/06/90	SW3-113-0205 04/04/90	SW3-113-0406 05/08/90	SW3-113-0507 06/04/90	SW3-113-0408 07/09/90
pH	571	6 96	7 50	8 02		6.25	7.19	6.93
CONDUCTIVITY	0 123	0 25	40	0 18	0 21	0.22	0.23	0 238
TEMPERATURE	30	20	17	19	6.0	13.5	19.0	23.4
DISSOLVED OXYGEN	12.4	13 6	13.4	92	12.2	9.2	9.8	6.6
VELOCITY fl/sec	NO DATA	00	00	NOT RECORDED	0.0	0.0	0.0	0.0

SAMPLE IDENTIFICATION	8W3 - 113 - 0409 06/07/90	SW3 - 113 - 0410 09/15/90	8W3-113-0911 10/01/90	SW3-113-0512 11/07/90	SW3-113-0113 12/06/90	SW3-114-0101 11/16/89	SW3-115-0201 11/28/89	SW3-116-0201 11/26/89
На	6 85	6 93	7 11	6 70	7.82			
CONDUCTIVITY	0 229	0 218	0 242	0.187	0.202		250	
TEMPERATURE	24.2	22 8	20 1	93	5.6		3.0	
DISSOLVED OXYGEN	56	56	12.2	15.4	11.5		13.6	11.5
VELOCITY Neec	00	NOT TAKEN	0.0	0.0	0.0	NOT TAKEN	NO DATA	0.0

SAMPLE IDENTIFICATION	SW3-117-0201	\$W3-118-0301	\$W3-119-0301	SW3-119-0202	SW3-119-0103	SW3-119-0104	SW3-119-0205	SW3-119-0406
DATE SAMPLED	11/15/09	11/15/89	11/13/89	01/09/90	02/08/90	03/06/90	04/04/90	05/08/90
			7.40		e 40	7.00		5.84
рн	7 20	/ 30	7.40	וע.ס	0.49	7.20	0.0	0.04
CONDUCTIVITY	0 225	025	021	1	0.21	0.29	0.2	0.23
TEMPERATURE	01	10 0	70	10	1.0	1.0	0.0	13.0
DISSOLVED OXYGEN	10 5	10 5	10.2	12.6	13.2	11.2	11.2	8.1
VELOCITY fl/sec	0 1	00	0.0	00	0.0	NOT RECORDED	0.0	0.0

SAMPLE IDENTIFICATION	8W3-119-0207	\$W3-119-0208	SW3-119-0309	SW3-119-0510	SW3-119-1211	SW3-119-0312	SW3-119-0313	SW3-120-0101
DATE SAMPLED	06/04/90	07/09/90	08/07/90	09/15/90	10/01/90	11/07/90	12/06/90	11/13/89
			ľ	Í		1		
(pH	73	693	6 75	6 96	7.01	7.14	8.03	7.20
CONDUCTIVITY	0 2 1	0 302	0 265	0 254	0.29	0.216	0.204	0.265
TEMPERATURE	20 0	23 6	24 3	20 9	19.5	9.3	5.1	8.0
DISSOLVED OXYGEN	71	61	58	64	11.1	14.5	10.7	116
VELOCITY N/sec	00	00	00	00	0.5	0.0	0.0	0.0

FINAL

TABLE 2-1 SURFACE WATER FIELD DATA SUMMARY NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS PAGE 3

SAMPLE IDENTIFICATION	8W3-121-0101	SW3-122-0401	SW3-123-0111	SW3-124-0411	SW3-125-0111	SW3-126-0111	SW3-127-0111	SW3-128-0111
DATE SAMPLED	11/14/89	11/14/89	10/04/90	10/03/90	10/09/90	10/09/90	10/08/90	10/03/90
pH	7 60	7.10	7.10	6 83	6.82	6.99	6.69	7.16
CONDUCTIVITY	0 2 1	023	0.266	0 326	0.246	0.229	0.262	0.34
TEMPERATURE	12 0	12.0	20.4	14.0	18.1	17.6	20.7	15.0
DISSOLVED OXYGEN	74	10 0	5.5	12.0	6.1	9.0	12.4	15.2
VELOCITY fl/sec	00	NOT TAKEN	NOT RECORDED	0.0	0.1	0.0	0.0	0.5

SAMPLE IDENTIFICATION	8W3-129-0111 10/03/90	8W3-130-0111 10/05/90	8W3-131-0111 10/09/90	SW3-132-0111 10/08/90	SW3-132-0111 10/09/90
	7.06	7.00	6.98	7 11	8.07
	0 335	0 254	0 231	0 24	0.21
TEMPERATURE	14.7	158	18.8	192	17.9
DISSOLVED OXYGEN	14 8	126	10.1	10.8	9.3
VELOCITY flyee	13	NOT RECORDED	0.3	25	0.3

٠

í

.

1

2.1.1.3 Monthly Surface Water Sampling and Analysis

Water samples were collected from SW3-102, the Pleasant Street crossing; SW3-105, at the Chemical Brook Culvert intake; SW3-108, at the Route 135 crossing; SW3-113, in Reservoir No. 2; and SW3-119, immediately upstream of the Saxonville Dam. Water samples were collected from five of the original Phase I sample locations on the first week of every month from January to December 1990. Sample locations were selected in several upgradient and several downgradient points relative to the Nyanza Site. These samples were analyzed for the parameters listed in the Sampling Summary Table 1-2. The analytical results for monthly surface water sampling are discussed in Section 4.3 and Appendix L and are presented in Appendix B, the Data Summary Analytical Database.

During monthly sample collection, stream flow, pH, conductivity, dissolved oxygen, and temperature were measured using standard field techniques and apparatus. No sediment samples were collected in conjunction with the monthly sampling program.

Stream Flow Measurements

Stream flow measurements were conducted monthly at location SW3-102 (Pleasant Street overpass), at the end of Phase I sampling, and during the surface water monthly sampling (January through December 1990) to determine stream flow volumes. The collection of stream flow measurements was interrupted during the months of January through March 1990 due to severe icing conditions. Additionally, stream flow data from the Saxonville Dam was acquired from the USGS. The data presented in Table 2-2 indicates a net increase in stream flow volume from the upstream location in Reach 1 (SW3-102) to the Saxonville Dam in Reach 4. Stream flow data from the Dam No. 1 (outlet of Reservoir No. 1) is available but it cannot be correlated in sufficient detail to augment the database.

The stream flow measurement at the SW3-102 location was conducted using a vertical shaft Teledyne Gurley Current Meter. The crosssection River profile was measured at a minimum of three vertical cross-section points. Velocity measurement readings were taken at 0.6 feet of the total depth below the surface at each vertical transect location (0.6-depth method). The velocity readings were multiplied by the respective estimated area represented by the transect point and were added to compute total discharge at each location. Results of the stream flow measurements are included in Section 3-6.

MONTHLY MONITORING FLOW DATA NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS NUS USGS FLOW FLOW (FT3/SEC) (FT3/SEC) SAXONVILLE DATE LOCATION SW3-102 DAM 12/7/89 229 54 186 572 4/5/90 5/9/90 21 196 70 205 6/5/90 7/10/90 8 45 8/8/90 36 276 10/2/90 4 41 11/8/90 66 155 12/6/90 77 430

TABLE 2-2

2.1.2 Sudbury River Sediment Sampling and Analysis

The sediment sampling and analysis program and sampling procedures are described the Final Work Plan and Work Plan Amendments and detailed in the Final Sampling and Analysis Plan and Amendments.

Quality control samples including field duplicates, trip blanks, and rinsate blanks were collected and analyzed for each sampling round as specified in EPA guidance (User's Guide to the Contract Laboratory Program) and in the Final SAP and the SAP Amendment Memoranda. The analytical parameters are presented in Table 1-1; analytical results for sediment sampling are discussed in Section 4.4 and are presented in Appendix A, the Summary Analytical Database.

Sediment sampling was conducted to determine the presence and distribution of inorganic and organic contaminants in the sediments, to gather data to evaluate the appropriateness of sediment remedial actions, and to provide data for the selection and design of remedial alternatives.

Phases I and II were purposely biased towards areas of low current velocity since silt- and clay-sized particles would be deposited in these areas. This sediment size readily adsorbs metal contaminants associated with Site-related discharges. Therefore, the extent of contamination may be determined through sampling and analysis of these sediments.

2.1.2.1 <u>River Sediment Phase I Sampling and Analysis</u>

The Phase I sampling program included a comprehensive initial sampling round of 50 sediment stations numbered SD3-101 to SD3-150 (Figure 1-A). Sediment samples were collected from depths of zero to six inches below the surface. At 14 of these locations, profile sediment samples were also collected at a depth of from six to 12 inches to provide information on vertical distribution of inorganic and organic contaminant concentrations. Sediment sampling locations include the River and Reservoir areas upstream and downstream of the Nyanza Site, as far as Heard Pond in Wayland. Samples were collected from expected depositional areas, such as near stream confluences, inside River bends, etc. Specific sample location rationale is discussed in detail in Sections 4.3.1.4 and 4.3.1.5 of the Final Work Plan (August 1989).

All 64 sediment samples were analyzed for TCL metals and methyl mercury. Twenty-two of these samples were also analyzed for other parameters, as specified in Table 1-1. These 22 samples were collected from locations that corresponded with the 22 surface water sampling locations, with the exceptions of SW3-102 and 103.

Analytical parameters were chosen to provide confirmation regarding the extent of contamination and contaminant transport/fate information in the ecosystem.

2.1.2.2 River Sediment Phase II Sampling and Analysis

The Phase II sediment sampling program included the collection of 59 surface sediment sample stations (0-6 inch depth), numbered SD3-152 through SD3-204, in the Sudbury River. Sediment profile samples were also collected at depths ranging from six to 12 inches at 19 of these locations to provide a vertical distribution of contaminants. In addition, sediment samples were collected at six locations to collect information on levels of hexavalent chromium in areas where elevated levels of total chromium had been detected in Phase I. Sediment sampling locations were selected in order to:

- o fill in data gaps in Phase I sampling
- o determine the down-river extent of elevated mercury levels in sediments
- determine if selected wetland areas adjacent to the River have been impacted by elevated metals concentrations
- provide additional background sediment samples where background fish samples were collected
- o provide additional sediment samples where Phase II benthic organism samples were collected

Several sediment sampling locations were changed from those proposed in the SAP Amendment No. 2. Changes were made because of poor sample recovery and to locate sampling stations at Phase II caddis fly larvae sample locations which were planned after preparation of the SAP.

2.1.2.3 Bordering Wetlands Sediment Sampling and Analysis

Sediments from wetland areas adjacent to the Sudbury River were sampled during Phase I at two stations (SD3-113 and 114) and Phase II at four stations (SD3-155, 159, 161, and 167), as presented in Figure 1-2A. The sediments were collected to determine if these wetlands have been impacted by contamination. The Phase I locations were sampled at depths ranging from zero to six inches while the Phase II stations were sampled at two intervals, zero to six inches and from six to 12 inches. The sediment samples were analyzed for the parameters presented in Table 1-1; the analytical results are presented in Appendix A. Locations were selected to geographically represent bordering wetlands and lands subject to flooding along the River.

W92194F

2-10

2.1.3 Biota Sampling and Analysis

2.1.3.1 Phase I Biota Sampling and Analysis

The Phase I biota sampling and analysis program and sampling procedures are described in Section 4.3.1.6 of the Final Work Plan (WP), August 1989) and Section 4.1.7 of the Field Sampling and Analysis Plan (SAP, August 1989).

Fish Sampling and Analysis

Phase I sampling of biota included collection of three different fish species for tissue analysis: Yellow perch (Perca flavescens), brown bullhead (<u>Ictalurus</u> <u>nebulosus</u>), and largemouth bass (Micropterus salmoides) at four selected sampling locations, FH3-101 through FH3-104 (see Figure 1-A). Sampling was conducted from September through December 1989. Severe weather conditions during sampling at the Sudbury Reservoir forced the discontinuation of the fish sampling in early December. The three species were selected to represent a range of trophic levels, including game fish. The four locations were selected to provide samples from a control reservoir upstream from the Site (Sudbury Reservoir, location FH3-104) and from three increasingly distant downstream locations.

The sample locations included FH3-102, which encompasses the Reservoir No. 2 impoundment (Reach 3); FH3-103, which includes the Reservoir No. 1 impoundment (Reach 4); and FH3-101, which includes the Saxonville Dam impoundment (Reach 6). Upstream migration of the sampled species within this portion of the River is prevented by the four spillway dams, however, these spillways do not prevent downstream migration of fish resulting in some mixing of populations.

Fish were collected at the locations using standard collection techniques. Techniques included electroshocking, gill netting, and hook and line. Electroshocking, using boat-mounted equipment, involved stunning the fish with an electric current and collecting the species of interest with dip nets. Gill netting involved setting monofilament gill nets at several locations within the sampling area so the fish became entangled in the nets. Nets were recovered and the fish of interest were retained as samples. Two types of hook and line apparatus were used during the sampling event: conventional rod and reel and weighted trout lines.

Samples of both muscle tissue (fillet) and offal were analyzed. All samples were analyzed for TAL metals, semi-volatile organics, and pesticides. Approximately twenty-five percent of the samples were also analyzed for methyl mercury. However, because of the low number of fish collected at some locations, the TAL metals, semi-

- W92194F

volatile organics, and pesticides fractions were prioritized over the methyl mercury analysis. Therefore, when the required sample weight for the metals, semi-volatile organics, and pesticides was achieved, the remaining samples (if available) were submitted for methyl mercury analysis.

The Final Work Plan specified that ten individuals per species of similar size be collected at each location for edible tissue (fillet) analysis, resulting in a total of 120 samples. To the extent that conditions permitted, fish size was to be consistent at each location. However, the sampling team often did not have an abundant catch and was not able to collect the full requirement of samples per species from each location. In an effort to maintain the required number of samples per location, most of the fish caught were needed to meet minimum weight requirements for analysis. This resulted in a wider variation in the size and weight of fish than was originally anticipated.

The less than anticipated catch, possibly resulted from cooler seasonal temperatures where fish become less mobile, or to effects on population density from contamination. Severe weather conditions during sampling at the Sudbury Reservoir (FH3-104), with eventual icing of the reservoir, forced the discontinuation of the biota sampling in early December.

Table 2-3 presents a summary of all Phase I fish collected, including the number of samples per species from each location, the types of analyses, and the fish weight and length. A range of weights or lengths is presented for those samples which were composited. The analytical results for fish sampling are discussed in Section 4.3.4 and are presented in Appendix C, the Data Summary Analytical Database.

Benthic Macroinvertebrate Survey

In Phase I, an assessment of the benthic macroinvertebrate community was conducted on samples from 45 of the sediment sampling locations, as presented in Table 1-1. Sample collection and preparation methods are presented in the SAP (April 1990). At each location approximately five gallons of sediment was collected and sieved for benthic organisms. Samples were sorted and specimens from the sample were collected. Sorting continued until 100 specimens were collected or the sample was exhausted. Any remaining sample was preserved in ethanol and sent to the laboratory for sorting. Results of this survey are presented in Section 4.5 of this report.

FINAL

TABLE 2-3

FISH SAMPLING SUMMARY – PHASE I NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

FINAL

~									
LOCATION	SPECIES	NO, OF	BNA/PEST	METHYL	BNA/PEST	METHYL	WEIGHT	LENGTH	TOTAL
AND SAMPLE		FISH	METALS	Ha	METALS	Ha	OF FISH	OF FISH	FISH IN
IDENT NO			FILLET	FILIET	OFFAL	OFFAL	(orams)	(ft)	SPECIES
						UTAL	(grams)		JF LUILS
EH2 101 BE200		~	v	М	v	A I	000 047	70 90	
FH3-101-BF302	BULLHEAD	2	T N	IN N	T N	EN	200-247	.7902	-
FH3-101-BF303	BULLHEAD	1	Y	N	Y	N	317	0.94	5
FH3-101-BF304	BULLHEAD	1	Y	N	N	Y	320	0.98	BULLHEAD
FH3-101-BF305	BULLHEAD	1	Y	N	N	Y	513	1.06	
FH3-101-BO306	BULLHEAD	4	N	N	Y	Y	COMPOSITE	COMPOSITE	Ξ
				•					
FH3-101-MF287	BASS	1	Y	Y	Y	N	916	1.33	
FH3-101-MF288	BASS	1	Y	Y	N	~	642	1.21	4
FH3-101-MF289	BASS	1	Y	Y	N		556	1.12	BASS
EH3-101-MO307	BASS	3	Ň	Ň	Ŷ	Y	COMPOSITE	COMPOSITE	=
	0,.00	Ŭ		••	•	•			-
EH2 101 VE200		4	v	N	v	NI	200	1.01	I
15H2 101 VE201	V DEDOU	1	I V	IN N	T V	IN N	110 150		
FH3-101-1F291	T. PERCH	2	Y	N	Y	N	116-159	./U/6	
FH3-101-YF293	Y. PERCH	1	Y	N	N	Y	292	0.91	
FH3-101-YF294	Y. PERCH	1	Y	Y	Y	Y	296	0.92	
FH3-101-YF295	Y. PERCH	2	Y	N	Y	N	172-192	. 79 –. 8 1	11
FH3-101-YF297	Y. PERCH	3	Y	Y	Y	Y	217-234	.8691	Y. PERCH
EH3-101-YE298	Y PERCH	3	Y	Y	Y	Y	188 - 209	82-83	
EH3-101-YE299	Y PERCH	4	v v	Ý	v.	Ý	76-180	62-79	
EH3_101_VE300	V PEPCH		Ý	, N	, v	, v	93-155	70 86	
EH2 101 VE201		4	, v	IN N	, v	, v	122 155	.7000	
	T. PERCH	4	Y NI		Y N		133-133	./3//	-
FH3-101-10308	Y. PERCH	10	<u>N</u>	<u> </u>	<u> </u>	Y	COMPOSITE	COMPOSITE	<u> </u>
		_							
'3-102-BF279	BULLHEAD	2	Ŷ	N	Y	N	237-274	.88-89	
.3-102-BF280	BULLHEAD	1	Y	N	Y	N	452	1.03	
FH3-102-BF281	BULLHEAD	1	Y	N	Y	N	374	0.98	
FH3-102-BF282	BULLHEAD	1	Y	N	Y	N	370	1.01	7
FH3-102-BF2B3	BULLHEAD	2	Y	N	Y	N	301-369	92-1.0	BULLHEAD
EH3-102-BE284	BUILHEAD	1	v	N	v	N	585	1 19	
EH3-102-BE286	BUILLIEAD		, v	N	, v	N	300	1.10	
	BULLHEAD		, i		, i	N N			-
FH3=102-B0205	BULLHEAD	0	N	N	Ť	N	COMPOSITE	COMPOSITE	5
		_				••			
FH3-102-MF243	BASS	1	Y	Y	Ŷ	N	2723	2.85	
FH3-102-MF244	BASS	1	Ŷ	Y	N	Y	1724	1.62	
FH3-102-MF245	BASS	1	Y	Y	Y	N	1461	1.51	
FH3-102-MF246	BASS	1	Y	N	N	Y	922	1.30	
FH3-102-MF247	BASS	1	Y	N	Y	N	624	1.24	12
FH3-102-MF248	BASS	1	Y	N	N	Y	535	1.11	BASS
FH3-102-MF249	BASS	1	Y	N	Ŷ	N	534	1 16	
EH3-102-ME251	BASS	1	Ý	N	Ň	Y	1918	1.66	
EH3-102-ME252	BASS		Ý	N	, i	Ň	840	1 30	
	DASS		, L		1 61	, in the second	663	1.30	
FFI3 - 102 - MF 233	DASS	1	Y N	IN .	rN .	Y	303	1.97	
FH3-102-MF258	BASS	1	Y	Y	N	N	394	1.04	_
FH3-102-MO267	BASS	10	N	N	Y	Y	COMPOSITE	COMPOSITE	
FH3-102-YF254	Y PERCH	7	Y	Y	Y	Y	72-114	.6274	
FH3-102-YF255	Y PERCH	16	Y	Y	Y	Y	68-110	.6070	
FH3-102-YF256DU	Y PERCH	16	Y	Y	Y	Y	68-110	.6070	
FH3-102-YE257	Y PERCH	A	Y	Y	Y	Y	65 - 180	5881	9
FH3-102-VE259	Y PERCH	÷	v v	×.	v.	×.	56-70	58-63	Y PERCH
EH1_102_VE776		0	, v	, N	, NI	, N	73_162	64_ 97	
ELD 102 TE2/0		3	T V	rw Ni	r¶ 1.1	i V Ni	10-100		
		1	T	N N	N .	IN N	24/	U.34	
-43-102-YF278	Y. PERCH	1	Y	N	N	N	342	1.03	
3-102-YO275	Y. PERCH	16	N	N	Y	Y	COMPOSITE	COMPOSIT	

Notes:

FH3-101 = SAXONVILLE IMPOUNDMENT

FH3-102 = RESERVOIR No. 2

FH3-103 = RESERVOIR No. 1 FH3-104 = SUDBURY RESERVOIR

TABLE 2-3 FISH SAMPLING SUMMARY - PHASE I NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS PAGE TWO

- --

LOCATION	SPECIES	NO OF	BNA/PEST	METHYL	BNA/PEST	METHYL	WEIGHT	LENGTH	TOTAL
AND SAMPLE		FISH	METALS	Ha	METALS	Ha	OF FISH	OF FISH	FIGHIN
			FULET	FILLET	OFFAI	OFFAI		(4)	SPECIES
IDENT. NO.					UFFAL	OFFAL	(grams)		SPECIES
EH2 102 BE210			v	v	v	N.	E10	1 20	
FH3 103 BF210	BULLIEAD		v	I NI	1 N		510	1.30	į
	BULLHEAD	· · ·	r V	IN N	in .	T NI	636	1.13	
	BULLHEAD		r V	IN N	, T	IN N	500	1.20	
FH3-103-BF222	BULLHEAD		r V	N	T NI	N	683	1.12	40
FH3-103-BF223	BULLHEAD		Ť	Ť	N	¥ N	703	1.16	12
FH3-103-BF226	BULLHEAD	1	Ŷ	N	Y	N	568	1.11	BULLHEAD
FH3-103-BF227	BULLHEAD	1	Y	N	N	Ŷ	392	1.05	
FH3-103-BF228	BULLHEAD	1	Y	N	N	Ŷ	593	1.15	
FH3-103-BF236	BULLHEAD	1	Y	N	N	N	678-546	1.19-1.11	
FH3-103-BF237DU	BULLHEAD	1	Y	N	N	N	678-546	1.19-1.11	
FH3-103-BF238	BULLHEAD	1	Y	N	N	N	665	1.13	
FH3-103-BO234	BULLHEAD	8	N	N	Y	Y			
EH3-103-ME201	BASS	1	Y	v	v	N	1116	1 22	
EH3_103_ME202	BASS	1	Ý	Ý	N	Ŷ	040	1.35	
FH3_103_MF202	RASS	, 1	`	N	v v	N	243 0195	1.50	
EH2 103-ME204	BASS	1	v v	, in the second	, v	N	2100	1.70	1
EH2 102 ME204	DASS	1	v	r V	Y Y	IN N	2002	1.70	i
FH3-103-MF20400	BASS		r V	T NI	T	N	2002	1.70	10
FH3-103-MF200	BASS	1	Ť	IN N	N	Y Y	2092	1.70	D400
FH3-103-MF207	DASS		T N	IN N	Y	Ť	618	1.10	BASS
FH3-103-MF215	BASS	1	Y	N	N	Ŷ	646	1.20	1
FH3-103-MF216	BASS	1	Y	N	Y	N	528	1.17	1
FH3-103-MF217	BASS	1	Y	N	N	Y	439	1.08	i
FH3-103-MF220	BASS	1	Y	N	N	Y	722	1.25	
FH3-103-MO211	BASS	10	N	N	Y	Y	COMPOSITE	ECOMPOSITE	
FH3-103-YF212	Y. PERCH	4	Y	Y	Y	N	151-232	.78~.86	
FH3-103-YE213	Y PERCH	5	Ý	Ŷ	Ň	Ŷ	117-240	70-90	
EH3-103-YE214	Y PERCH	7	Ý	Ý	Ŷ	Ň	75-118	60 - 72	i i
EH3_103_YE224	Y PERCH	, ,	Ý	Ý	N	v v	161-235	78 - 89	1
EH3-103-VE24	Y PEPCH	5	Ý	Ý	N	, v	122-266	74 - 93	
EH2_103_VE229	V PEOCH	2	Ý	Ň	N N	N	100-177	.7455	12
FH3 103 - 1F223	V PEDCH	3	, v	N N	v v		100-111	.000	VBEDCH
FH3-103-1F230		3	, i	in N	Y	N V	39-133	.00/3	T. PENUN
FFI3-103-1F231	T. PERCH	3	Ť	N	N	T N	140-166	./6~ //	
FH3-103-TF232	T PERCH	3	Ť	N	N	•	114-120	.6873	
FH3-103-YF233	Y. PEHCH	3	Y	N	Ŷ	N	89-200	.6782	!
FH3-103-YF239	Y. PERCH	3	Y	N	N	N	266 - 292	.8894	
FH3-103-YF240DU	Y PERCH	3	Y	N	N	N	266 - 292	.8894	
FH3-103-YO235	Y PERCH	10	<u>N</u>	<u>N</u>	Y	Y			
EH2. 104 - ME280	BACC		~	v	v	v	660	1.16	1
	BACC	1	v	Ý	Ý	v	600	1 10	
	DASS		T V	, , , , , , , , , , , , , , , , , , ,	T N	, T	022	1 10	
FTIJ-104-MF202	DASS		1	Ť	T N	N	8/4	1.33	
FH3-104-MF263	BASS	1	•	¥.	¥.	N	1253	142	
FH3-104-MF264	BASS	1	Y	Y	Y	N	1336	1 45	
HI3-104-MF265	BASS	1	Y	Y	Y	N	683	1.23	13
FH3-104-MF266	BASS	1	Y	Y	N	Y	557	1.10	BASS
FH3-104-MF268	BASS	1	Y	N	N	Y	872	1.28	
FH3-104-MF270	BASS	1	Y	N	N	Y	98 5	1.35	
FH3-104-MF271	BASS	1	Y	Y	N	Y	1442	1 49	
FH3-104-MF272DU	BASS	1	Y	Y	N	N	1442	1.49	
FH3-104-MO273	BASS	5	N	N	Y	N	COMPOSITE	COMPOSITE	
FH3-104-MO274	BASS	5	N	N	N	Y	COMPOSITE	COMPOSITE	

Notes:

FH3-101 = SAXONVILLE IMPOUNDMENT

FH3-102 = RESERVOIR No. 2

FH3-103 = RESERVOIR No. 1 FH3-104 = SUDBURY RESERVOIR

.

2.1.3.2 <u>Biota Phase II Sampling and Analyses</u>

Phase II biota sampling and analyses included additional fish sampling and benthic organism collection. These surveys were completed by EPA and the U.S. FWS. NUS was responsible for data validation and interpretation of the fish data and interpretation of the benthic organism data.

Fish Sampling and Analysis

Sampling of biota during Phase II (July 1990) included collection of two fish species for tissue analysis. This activity was performed by the U.S. FWS in cooperation with EPA. The species collected included Yellow perch (<u>Perca flavescens</u>) and largemouth bass (<u>Micropterus Salmoides</u>) at seven selected locations throughout the Sudbury River. These included two upstream reference sites, FH3-204 (Cedar Swamp Pond) and FH3-205 (Southville Pond); and five downstream sites including FH3-206 (Mill Pond), FH3-202 (Reservoir No. 2), FH3-203 (Reservoir No. 1), FH3-201 (Saxonville Reservoir), and FH3-207 (Fairhaven Bay). No yellow perch were observed or captured for analysis in Southville Pond.

Fish were captured using an electroshocking boat at all locations. Additionally, experimental gill nets were used at Cedar Swamp Pond. Fish were weighed, measured, and scales were removed for age determination; whole fish were packaged for selected contaminant analyses. Fish analyses included TAL metals and TCL organics (including methyl mercury). The analytical results are presented in Appendix C (See FWS report in Appendix E).

<u>Caddis Fly Larvae Sampling and Analyses</u>

The Phase II benthic organism sampling included the collection of caddis fly larvae (aquatic insects) at six locations in the Sudbury River. The caddis fly larvae sampling locations correspond to surface water and sediment sample locations numbered SD3-128, 129, 130, 131, 132, and 133. These organisms were collected for mercury and methyl mercury analysis since a portion of their life cycle involves the filtering of River water to acquire food. Therefore, the contaminant content in the body burden of these organisms may be indicative of the contaminants which are available to all benthic organisms in the River system and may also serve as bioconcentrators and bioaccumulators of contaminants.

The caddis fly larvae sampling program was performed by the Region I Environmental Services Division (ESD) Biology Section. The ESD report is presented in Appendix F and the analytical results of the survey are described in Section 4.6.

2.1.4 Biological Survey

This section provides a summary of the wildlife and wetlands assessment conducted along the Sudbury River.

2.1.4.1 <u>Sudbury River Wetlands Assessment</u>

In October 1989, a field assessment was performed of bordering wetlands in the vicinity of the Site and in the Sudbury River immediately downstream from the Site. Wetland vegetation and hydrology were observed at 27 key locations along approximately ten miles of the River between Ashland and the Massachusetts Turnpike north of Framingham. The wetlands assessment report is provided in Appendix G.

A brief qualitative assessment was made of the wildlife potential in the Sudbury River Study Area. The findings of this study are presented in Section 3.6.4.3.

2.2 <u>Eastern Wetlands Sampling and Analysis</u>

The Phase I and II Eastern Wetlands sampling and analysis program and sampling procedures are described in Section 4.3.1.5 of the Final Work Plan (WP), (August 1989) and Sections 4.1.6 and 4.2.4, respectively, of the Field Sampling and Analysis Plan (SAP, August 1989) and the Final Sampling and Analysis Plan Amendment No. 2 (October 1990). The analytical results for the Eastern Wetlands sampling are discussed in Section 4.7 and are presented in Appendix D, the Data Summary Analytical Database.

2.2.1 Physiography

The Eastern Wetlands is an area of approximately ten acres east of the abandoned Trolley bed at the Nyanza Site. The wetland area studied in this survey includes two open water ponds (approximately two to three acres) which are separated by an earthen dike. Surficial drainage from the Nyanza Site enters the wetland from the western side of the old Trolley embankment through a granite slab culvert as illustrated in Figure 1-D. Surface water flow between the two ponds is constricted through a narrow opening in the dike. Standing water was present over approximately 30 percent of the wetland areas to a depth of from one to three feet.

The wetland area presently drains north through Trolley Brook to its confluence with Chemical Brook. Trolley Brook and Chemical Brook drain in a northeasterly direction through the Chemical Brook Culvert and discharge through the Outfall Creek into the Raceway near Concord Street in Ashland (Figure 1-A, Detail).

2.2.2 Eastern Wetlands Surface Water Sampling and Analysis

Phase I surface water sampling included the collection of two surface water samples, numbered SW3-104 and SW3-105, as illustrated in Figure 1-A. The samples were submitted to CLP for analysis for the parameters presented in Table 1-2. Station SW3-105 was also included in the monthly surface water sampling. No additional surface water sampling and analysis was performed in the Eastern Wetlands during Phase II.

FINAL

2.2.3 Eastern Wetlands Sediment Sampling and Analysis

The Phase I sediment sampling in the Eastern Wetlands included the collection of five samples to obtain initial information on potential contamination; the Phase II study provided detailed information on the horizontal and vertical distribution of contaminants in the sediments, as described below.

2.2.3.1 Phase I Eastern Wetlands Sediment Sampling and Analysis

The Phase I sediment sampling in the Eastern Wetlands included five sediment sample stations numbered SD3-106 through SD3-110, as presented in Figure 1-A. In addition, subsurface sediment samples were collected at a depth of from six to 12 inches at all five stations to provide information on vertical distribution of contaminants at these locations. The 10 sediment samples were submitted to a CLP laboratory and analyzed for TAL metals and methyl mercury.

2.2.3.2 Phase II Eastern Wetlands Sampling and Analysis

Unconsolidated sediments in the Eastern Wetlands area were sampled in Phase II to provide information on the horizontal and vertical distribution of contaminants in the wetland. The sampling program included advancing a total of 32 sediment soil borings. Twentythree samples were collected from areas of free-standing water, with four discrete sample intervals. Sample intervals included: zero to two, two to four, four to five and five to six feet below sediment surface at most locations. Nine additional samples were collected from two discrete one-foot sample intervals in areas adjacent to the open water or from those areas which appeared to be affected only by seasonal flooding. Each discrete sample interval collected within the Eastern Wetlands was a grab sample.

A total of 108 sediment samples were submitted for CLP analyses, which included TCL volatile organics, BNA, Pesticides, PCBs, sitespecific organic compounds, and TAL metals.

2.3 <u>Chemical/Cold Spring Brook Culverts/Raceway Sampling and</u> <u>Analysis</u>

2.3.1 Visual Inspection/Mapping

A reconnaissance was made of the Chemical Brook and Cold Spring Brook Culverts to determine the applicability of viewing the interior of the culverts (Figure 1-A). The focus of this task was to determine the integrity of the structures and establish whether sediment deposition was occurring. Such deposits could be a continuing source of contamination. Based on the reconnaissance, a televiewing inspection was performed in the Chemical Brook Culvert in January 1991. It was determined that televiewing the Cold Spring Brook Culvert was not feasible. A memoranda discussing the reconnaissance results are presented in Appendix H.

2.3.1.1 <u>Chemical Brook Culvert Inspection/Sampling & Analysis</u>

The visual inspection of the Chemical Brook Culvert included measuring water and sediment depths in the stormwater lines and manholes. General information concerning the stormwater lines such as construction materials, line diameter, and the presence of feeder lines was recorded.

The televiewing inspection of the Chemical Brook Culvert indicated it was constructed over a long period of time based on the various construction techniques and materials encountered. Sections of the system had also been repaired or replaced in a piecemeal fashion. Materials included section(s) of cast iron pipe, corrugated steel pipe, cement pipe, and a granite slab/block construction. The bottom of the slab/block construction was open to the environment. The Culvert is a stormline system and several storm drain feeder lines were present throughout the length of the pipe. The televiewing inspection was ended at Front Street in Ashland, Massachusetts, where debris blocked further advancement of the video camera. It appears that catch basins and areas of large debris deposition, such as tree branches, may act as minor sediment depositional areas. A review of the video inspection is presented in Appendix H.

Surface water sampling included the collection of one surface water sample at the Culvert outfall, numbered SW3-106, as shown in Figure 1-A. The sample was submitted for CLP analysis for parameters presented in Table 1-2. Station SW3-105, which is associated with the Eastern Wetland, is located at the culvert's intake. Samples collected from SW3-105 can be used for comparison with contaminant concentrations within the culvert. Results are presented in Section 4.8.

The Chemical Brook Culvert sediment sampling included the collection of five surface sediment samples, numbered SD3-112 (Phase I) and SD3-200, 201, 205, and 206 (Phase II). Station SD3-112 was at the Culvert outfall; Stations SD3-200 and 201 were collected along the Culvert in catch basins where sediment had accumulated; and Stations SD3-205 and 206 were collected in the Outfall Creek. Results are presented in Section 4.8.

2.3.1.2 Cold Spring Brook Culvert

No surface water samples were collected in the Cold Spring Brook Culvert (formerly designated the Trolley Brook Wetlands Culvert). Inspections of aerial photography, supported by field confirmation, indicated that the Culvert was not a surficial drainage route from the Eastern Wetlands as originally presumed. During both Phases I and II sampling rounds, there was no water available for sampling at the Culvert outfall. However, during Phase I, surface water (Station SW3-103) was sampled at the apparent conjunction of the Culvert with Cold Spring Brook.

No sediment samples were collected within the Cold Spring Brook Wetlands Culvert. During both Phases I and II sampling rounds, surface water was not being discharged from the Culvert outfall; this may be due to a blockage in the Culvert or to a leak prior to the discharge point. However, during Phase I operations, one sediment sample (Station SD3-105) was collected at the confluence of the Culvert with Cold Spring Brook. It was analyzed for the parameters presented in Table 1-1.

2.3.1.3 <u>Raceway Surface Water Sampling</u>

One surface water sample (SW3-136) was collected in the Raceway for the parameters listed in Table 1-2.

Two profile sediment sample locations, Stations SD3-244 and SD3-245, were established within the Raceway during a supplemental Phase II sample round. Two-depth profile samples (zero to six and six to 12 inches) were collected at each location. The sediment samples were collected for the parameters presented in Table 1-1.

FINAL

 $\overline{}$

3

3.0 GENERAL CHARACTERISTICS OF THE SITE STUDY AREA

3.1 <u>Introduction</u>

The Sudbury River Study Area is situated within the Metrowest area of eastern Massachusetts, (the western portion of the Boston metropolitan area). The Sudbury River and its tributaries in the Study Area flow through several communities that range in size from approximately 4,000 to 65,000 residents (Table 3-1). The area is classified as semi-rural to urban-suburban.

The population in the area has grown rapidly since World War II because of its proximity and access to the Boston metropolitan area, the City of Worcester, and the economic expansion of technology-related industries along major transportation routes, including Interstate Routes 495 and 90 (Figure 3-1). Individual communities have felt the effect of population and employment growth as a result of major highway construction within the last 30 years. These once rural communities, with an agricultural and small-scale manufacturing economic base, have become suburban.

Because of the dramatic population growth since 1950, land use patterns in the Study Area have changed substantially. The portion of the Study Area that falls within the Town of Framingham is the most developed due to the growing number of business and manufacturing concerns along I-495 and Route 9. Residential land use in the Study Area has approximately tripled in acreage, with a corresponding decrease in the amount of land used for agricultural purposes.

Within the Sudbury River drainage basin, vegetation on undeveloped marshy land and land formerly used for farming is generally in the brushy thicket stage. Many areas support fair to good quality stands of white pine, pitch pine, and mixed hardwoods.

3.2 <u>Surface Features</u>

The Study Area includes approximately 33 miles of the Sudbury River and the surficial drainage routes from the Nyanza Site in Ashland, Massachusetts. The surficial drainage routes include the Eastern Wetlands and Trolley Brook which drain the eastern/southeastern portion of the Site, and Chemical Brook which drains the northwestern onsite wetland. Both brooks combine and form the Chemical Brook Culvert which eventually drains to the Sudbury River (Figures 2-C and 2-D).

TABLE 3-1 POPULATION OF TOWNS ALONG THE SUDBURY RIVER NYANZA III - SUDBURY RIVER STUDY

Town	Population Count
Westborough	14,133
Southborough	6,628
Ashland	12,066
Framingham	64,969
Wayland	11,874
Sudbury	14,358
Lincoln	7,666
Concord	17,076

Source: U.S. Bureau of the Census as of April 1, 1990



<u>Site Vicinity</u>

The Nyanza Site is drained by two small Brooks, Chemical Brook and Trolley Brook. The Chemical Brook source area is the wetland located in the northwestern portion of the Site (Western Wetland); the Brook flows eastward along the railroad tracks which delineate the Site's northern border. The Brook has a watershed of about 83 acres (NUS, 1985). Chemical Brook was remediated as part of construction activities associated with Operable Unit I.

Trolley Brook flows northeast through a wetland along the eastern border of the Site, following the berm of an abandoned trolley line. The Eastern Wetland is approximately ten acres, of which approximately three acres is open water. Trolley Brook joins Chemical Brook near the railroad tracks. The combined flow discharges to the Sudbury River through an underground culvert (Chemical Brook Culvert) that has an outfall downstream of Concord Street, about 2,000 feet northeast of the northern limit of the Site. A visual and remote video inspection was performed on the Chemical Brook Culvert and is described in Section 2.3 of this report.

A second culvert, referred to as Cold Spring Brook Culvert, appears to be comprised of a groundwater seepage area, parking lot runoff, and storm-event runoff from a small wetland area east of the Site; it is not a surficial drainage route of the Eastern Wetlands. In a previous report (Technical Directive Memorandum, March, 1990), this second culvert was designated the Trolley Brook Wetlands Culvert, since it appeared that the Culvert was a surficial drainage from the wetlands. However, inspection of aerial photographs, with field confirmation, indicated that the culvert is separated from the Eastern Wetlands by uplands. Therefore, the name was determined to be inappropriate and changed to the Cold Spring Brook Culvert. Historical information provided by private citizens and visual observations were used to plot the approximate location of this Culvert as it drains easterly to Cold Spring Brook.

Sudbury River Description

The headwaters of the Sudbury River is Cedar Swamp Pond, in Westborough, Massachusetts. The River flows approximately eight miles eastward through Hopkinton and Ashland, and then approximately 25 miles northward through Framingham, Sudbury, and Wayland, to the confluence with the Assabet River in Concord, Massachusetts. The Concord River begins at this point and flows northward 16 miles to the Merrimack River in Lowell, Massachusetts.
The Merrimack River flows east, discharging to the Atlantic Ocean (Figure 3-2). The average hydraulic gradient of the upper Sudbury River is approximately five feet per mile. However, the average gradient of the River between the Saxonville Dam and the Assabet River is less than one foot per mile.

The watershed area of the Sudbury River is approximately 165 square miles. The Sudbury River watershed from Cedar Swamp to Dam No. 1 in Framingham consists primarily of uplands. Gentle to steep hills and ridges of bedrock and glacial till are found in the watershed. The largest surface water bodies associated with the River are located in this area and include Reservoirs Nos. 1, 2, and 3, and the Sudbury Reservoir. The overflow from the Sudbury Reservoir drains into Reservoir No. 3 and subsequently into Reservoir No. 1 in Framingham. The River also receives water from three other reservoirs: Whitehall, Hopkinton, and Ashland.

The topography of the watershed from Dam No. 1 to the Assabet River in Concord is gently rolling and consists of stratified glacial drift interspersed with many drumlins. In this area of the basin, the Sudbury River flows through extensive wetlands which overlie stratified glacial drift. The River floodplain is nearly three miles wide in the Wayland vicinity (Reach 7). The largest surface water bodies of interest in this portion of the watershed are the Saxonville Pond impoundment, Heard Pond, and Fairhaven Bay.

The stretch of the River from Westborough to Saxonville is characterized by a series of dams. The Sudbury River had more than a dozen small and medium-sized mills above Saxonville. Many of the impoundments associated with these mills exist today behind intact or partially collapsed dams. These include the Cordaville Dam, Mill Pond Dam (a.k.a. Myrtle Street Dam), Fenwick Street Dam, and the Saxonville Dam (a.k.a. Colonna Dam) (Figure 1-A). Several other dams were built in the late 1800s to form Reservoir Nos. 1 and 2. A USGS gauging station is located at the Saxonville Dam. A second gauging station is maintained by the MDC at the outlet of Reservoir No. 1 (Dam No. 1) in Framingham Center (Figure 1-1).

3.2.1 Surface Water Use

Sections of the Sudbury River are part of the MDC's backup water supply system. Reservoir Nos. 1 and 2, on the south branch of the Sudbury River, have not been used for drinking water since 1930 and are maintained only as an emergency water supply. Sudbury Reservoir and Framingham Reservoir No. 3, which comprise the North Sudbury System, are used to supplement water supplies during summertime high demand periods (DWPC 1988). The Sudbury Reservoir and Framingham Reservoir No. 3 are on a separate tributary of the Sudbury River and do not receive surface water flow from the Site.

W92194F

FINAL



NYANZA III - SUDBURY RIVER STUDY REMEDIAL INVESTIGATION STUDY AREA LOCATION MAP

3-6

FIGURE 3-2



W92194F

The Sudbury River in Ashland is classified as Class B water, according to Massachusetts Surface Water Quality Standards. Class B water is designated for uses of protection and propagation of fish, other aquatic life and wildlife, and for primary and secondary contact recreation. Surface waters/wetlands in the vicinity of the Nyanza Site are accessible to the public.

Major discharges to the Sudbury River originate at the Raytheon Company and the Wayland and Marlborough Easterly Wastewater Treatment Plant (WWTP) which discharges to Hop Brook, a tributary of the Sudbury River in Wayland. The WWTP has been at its present location since 1896. In 1973, it was upgraded to an advanced treatment facility incorporating nitrification and phosphorus removal. The Raytheon Company wastewater treatment discharge to the Sudbury River consists of electroplating process wastewater and sanitary wastewater (DWPC 1988).

Portions of the Sudbury River are used for numerous recreational purposes. These include boating, fishing, hiking, and other passive recreational activities. Limited access to many parts of the River (MDC restrictions, fishing advisories, and private lands) prevent full utilization of its recreational potential. Water use in the Reservoirs is restricted by MDC policy which prohibits boating, fishing, swimming, and skating. However, it should be noted that this restriction is being reviewed by MDC.

3.2.2 Land Use

Land uses along the Sudbury River shoreline are varied. The land largely undeveloped in the vicinity of Cedar Swamp in is Westborough. The upper portion of the River flows through the residential sections of Southborough and then through Ashland Center, where it is bordered by mixed zoning areas including residential, commercial, and industrial properties. The River flows into Reservoir No. 2, Reservoir No. 1 and then into Framingham, where river-front property is characterized by urban land uses, downriver to the Saxonville area of Framingham. The river shoreline regains a rural character downstream from the Saxonville Dam. From the Framingham/Wayland town line northward approximately nine miles to Route 117, the River is bordered by the Great Meadows National Wildlife Refuge (Reach 8, Figure 1- λ). Heard Pond, which is in this area, is reported to be occasionally inundated by River flooding. Downstream of the refuge, bordering land uses include agriculture and forests, to the Route 27 overpass in Wayland where residential development and a country club dominate river-front usage.

The River is bordered by meadows and wetlands downstream from Wayland, with occasional residential usage. The River flows into Fairhaven Bay which is bordered by forested hills and the historic

W92194F

FINAL

Baker Farm on the Concord/Lincoln town line. Downstream from the Bay, the River is bordered by wetlands, residences, and a canoe rental establishment near the confluence with the Assabet River in Concord, Massachusetts.

Land use patterns in the Sudbury River watershed have changed significantly, as would be expected from the dramatic growth of the area since 1950 (Table 3-2). During this period, the total acreage in forest and wetlands has not changed appreciably. Residential land uses have tripled in acreage with an associated decrease in the amount of land used for agricultural purposes.

The residential character of the communities within the watershed is largely low density, single family housing units. However, in recent years, significant multi-family construction has occurred. Typically, business, commercial, and industrial uses are concentrated in town centers and along major transportation routes that pass through the watershed (Massachusetts Turnpike (Interstate 90), Interstate 495, and Routes 2, 9, 30).

3.3 <u>Climate</u>

The Study Area is situated in a temperate-humid weather zone. Climatic characteristics include changeable weather, large daily and annual ranges in temperature, seasonal variability between years, and uniform distribution of precipitation throughout the year.

The regional National Oceanic and Atmospheric Administration (NOAA) Climatological Station is located in Worcester, approximately 15 miles west of the Study Area. Data from this station, from the 1951-1980 record period, were used to describe the general climate of the Study Area.

The Study Area lies within the central climatological division of Massachusetts where the Atlantic Ocean has a limited effect. July is the warmest month with a normal mean temperature of 70 degrees Fahrenheit. The three coldest months (December, January, and February) have an average temperature of approximately 25 degrees Fahrenheit. Prevailing wind direction from November through June is from the northwest, whereas southwesterly winds prevail from July through October.

Precipitation occurs an average of 80 days per year, with mean monthly normal precipitation ranging from a low of 3.3 inches in February to a high of 4.4 inches in both August and November. The mean annual precipitation is 47 inches. The mean annual snowfall is approximately 48 inches, which is equivalent to about five

FINAL

TABLE 3-2 SUDBURY RIVER WATERSHED LAND USES, 1951 - 1981 NYANZA III - SUDBURY RIVER STUDY

	<u>1951⁽¹⁾</u>	<u>1971⁽¹⁾</u>	<u>1981</u> ⁽²⁾
Forest (acres)	101,907	97,058	95,116
(% of Total)	55.6%	52.9%	51.9%
Wetland	14,796	14,066	14,066
	8.1%	7.78	7.78
Agriculture	51,020	23,327	16,960
	27.8%	12.7%	9.28
Residential	12,244	33,428	41,067
	6.7%	18.28	22.48
C/I/M/T*	1,387	8,045	8,715
	4.0%	4.4%	4.8%
Open/Recreational	2,043	7,473	7,473
	1.0%	4.18	4.18

- * Includes commercial, industrial, mining, and transportation land use.
- (1) MacConnell, William P. <u>Remote Sensing of Twenty Years of</u> <u>Change in Middlesex (Worcester) Counties, 1951 - 1971</u>, 1974.
- (2) Parsons, Brinkerhaff, Quado and Douglas, Inc., <u>Sudbury</u> <u>Reservoir Water Treatment Plant</u>, <u>Draft Environmental Impact</u> <u>Report</u>.

inches of water. The average annual evapotranspiration rate is estimated at about 21 inches and the average annual runoff is 22 inches (NUS, 1985a).

3.4 <u>General Geology</u>

The topography of the watershed of the Sudbury River is characterized by a flat meandering river valley within a kame and kettle landscape created during the Pleistocene glaciation. This valley is rimmed by bedrock and glacial till uplands which form the surface watershed divide.

3.4.1 Soils

Soils in the Sudbury River watershed originated primarily from glacial materials. Upland areas, (elevations greater than 200 feet above Mean Sea Level), are covered with two to three feet of loam underlain by glacial till. Cobbles and boulders are common, as are bedrock outcrops on steeper slopes.

Valley soils (elevations below 200 feet Mean Sea Level) were deposited predominantly by glacial meltwater streams. More recent alluvial and organic sediments are results of modern river deposition.

The soils derived from glacial meltwater streams are coarse and well drained. These occur along the main River valley and along major tributaries in Framingham. Soils in Wayland, Sudbury, Lincoln, and Concord were deposited in the many glacial lakes which occupied large portions of the River valley. These poorly drained soils are characterized by finer grained, silt-sized particles and are often overlain by an organic muck or peat layer which may be up to 20 feet thick (SCS, 1973).

Within the watershed there are two broad groups, or associations, of soils. The soils in the Paxton-Hollis-Canton Association typify the upland areas of the watershed. These soils, derived from glacial till, are stony, fine sandy loam at ground surface and are well drained. Exceptions occur in the Paxton soils which have a permeable layer at depths of from two to four feet and Hollis soils where bedrock may occur at or just below ground surface. Paxton soils comprise about 50 percent of this association, Hollis soils about 15 percent, and Canton soils about 10 percent. The remainder of the association consists of a variety of less extensive fine grained soils.

FINAL

The Hinckley-Windsor-Muck Association is the second major category of soils found in the watershed of the Sudbury River. The Association constitutes the majority of soils present within the watershed, particularly in the River floodplain and along major tributaries. These soils were formed from materials pre-sorted by The Hinckley soils, comprising 50 percent of glacial meltwaters. the association, and Windsor soils, 30 percent, are coarse and very approximately percent of Muck soils, permeable. 10 the association, are derived from organic matter deposited on top of the glacial lake sediments and are found throughout the watershed. A number of other soils, minor in extent, comprise the remaining 20 percent.

3.4.2 Bedrock

The relief and configuration of the pre-glacial bedrock surface determines the resulting thickness of overlying glacial deposits and the amount of water potentially available in those deposits. Over the last 60 million years, streams and rivers have incised a valley along faulted and fractured zones of weakness in the bedrock (SCS, 1973). Pleistocene glaciation further eroded the watershed by scouring and eroding the bedrock surface and deepening the valleys.

The bedrock which underlies the Nyanza Site has been described by the USGS (Nelson, 1974) as a quartz monzonite, locally referred to as the Milford Granite. This pink to light gray granite sequence is believed to be of Lower Paleozoic to Precambrian in age and generally is unfoliated. Bedrock throughout the central portions of the Study Area, from Route 9 north to Sudbury, is characterized by more mafic granitic rocks and minor felsic metavolacanic sequences. These units are grouped within the Milford granite sequence.

North of Sudbury, the area is underlain by the Nashoba sequence. This sequence is locally represented by coarse grained biotite granite, a muscovite-biotite granite, Andover granite, which is the Straw Hollow Diorite, and other undifferentiated intrusives. A minor amphibolite/gneiss sequence is present south of Concord.

Nelson (1974) has identified four faults which are within a two mile radius of the Nyanza Site. Ebasco (1974) has observed some fractures in the granite at several locations. These locations include the bottom of dry lagoons on the Site, outcrops about one mile from the Site and samples of bedrock recovered from borings. Slight weathering was observed along several of the fractures and is interpreted as an indication of groundwater migration along the fractures. Other regional faults may affect regional hydrogeology.

FINAL

3.5 <u>Surface Water Hydrology</u>

This section describes the surface water runoff and the infiltration characteristics from the Nyanza Site to the Sudbury River. This information is required to assess the potential pathways and receptors of contaminated surface water and sediment.

3.5.1 Hydrological Setting

The Nyanza Site is located in the drainage basin of the Sudbury River, a tributary to Concord River, which is, in turn, a tributary to the Merrimack River, as shown on Figures 1-1 and 3-1.

The Nyanza Site is drained by two small Brooks, Chemical Brook and Trolley Brook, as shown on Figure 1-A and 1-D. The western wetland in the northwestern corner of the Nyanza Site forms the headwaters of the intermittent stream, Chemical Brook. Chemical Brook flows easterly along the northern boundary of the Site, parallel to the Conrail Railroad tracks. Chemical Brook has a watershed of about 83 acres. The western wetland and Chemical Brook were remediated in 1990 as part of construction activities associated with Operable Unit I.

Trolley Brook watershed comprises approximately 185 acres, originates on Megunko Hill in the southern portion of the Site and flows along the western embankment of the abandoned trolley bed and into the wetland (Eastern Wetland) along the eastern border of the Site. Trolley Brook flows under the trolley bed through a culvert, then along the western side of the trolley bed, under Megunko Road and merges with Chemical Brook at a culvert near the Conrail railroad tracks. The combined flow discharges through Chemical Brook Culvert, approximately 2,000 feet northeast of the Sudbury River Raceway. The culvert discharges to the Outfall Creek, which flows in a northeasterly direction to discharge into the Raceway (Figures 1-A and 1-D).

The Raceway converges with the Sudbury River approximately 700 feet downstream from the Culvert Outfall Creek. The Raceway is an old mill spillway diverted at Mill Pond and roughly parallels the River for approximately 2,600 feet. From Mill Pond, the Raceway enters a culvert and passes beneath reconditioned mill buildings on Main Street. The Raceway emerges again near the intersection of Main and Pleasant Streets and flows easterly, converging with the River near the Front Street crossing.

The recharge to Chemical and Trolley Brooks is from surface runoff/precipitation and from shallow groundwater, depending on the season (Ebasco, 1991). The wetlands act as water storage areas during the dry summer months.

W92194F

Chemical Brook is fed by groundwater after heavy rain and during periods of high groundwater in the spring melt. During these episodes, the wetlands act as groundwater discharge areas. Chemical Brook and its source area are often dry in summer months; this area then becomes a groundwater recharge area. Eastern Wetland, however, is an area of groundwater discharge and almost always contains flowing water (Ebasco, 1990).

Approximately 0.5 mile east (downstream) from the Chemical Brook Culvert is the Cold Spring Brook confluence and the upstream end of Reservoir No. 2. From this point, the River flows north into the southwestern arm of Reservoir No. 1. Additionally, the outlet of the Sudbury Reservoir flows east into Reservoir No. 3 which flows into the northwestern arm of Reservoir No. 1. Reservoir No. 3 and the Sudbury Reservoir are on a separate tributary of the Sudbury River and do not receive surface water flow from the Nyanza Site. The Sudbury River flows out of the northeastern end of Reservoir No. 1. The average annual discharge below Reservoir No. 1 is 118 cubic feet/second (U.S. Geological Survey, 1989).

3.6 <u>Sudbury River</u>

3.6.1 Hydrology

A detailed description of the Sudbury River is provided in Section 3.2. The headwaters of the Sudbury River is in Cedar Swamp, east of Westborough, Massachusetts. The river flows eastward through Hopkinton and Ashland, and then northward through Framingham, Sudbury, Wayland, Lincoln and Concord. The Sudbury joins the Assabet River in Concord, Massachusetts, forming the Concord River.

The watershed area of the Sudbury River is approximately 165 square miles. The largest surface water bodies of the River include Reservoir Nos. 1, 2, and 3; Sudbury Reservoir; Saxonville Impoundment; and Fairhaven Bay. The overflow from the Sudbury Reservoir drains into Reservoir Nos. 3 and 1 in Framingham. The River also receives flows from three other reservoirs which are all upstream of the Site: Whitehall, Hopkinton, and Ashland.

The stretch of the River from Westborough to Saxonville is characterized by a series of dams. The Sudbury River had more than a dozen small- and medium-sized mills above Saxonville. These include the Cordaville Dam, Mill Pond Dam (Myrtle Street), Fenwick Street Dam, and the Saxonville Dam (a.k.a. Colonna Dam). Several other dams were built in the late 1800s to form Reservoir Nos. 1 and 2.

3.6.2 Sediments/Bathymetry

Α survev of Reservoir Nos. 1 and 2 was performed in August/September 1989 by the USGS to provide bathymetric and sediment thickness maps for refining sediment sampling locations (Appendix I). More than seven miles of ground-penetrating radar profiles were obtained along selected segments. Depth of water ranged from zero to more than 20 feet. Soft sediment thickness ranged from less than one foot to more than three feet. Depth of water and sediment thickness were verified by physical measurements along selected transects.

Sudbury River sediments were characterized consistent with the American Society for Testing and Materials (ASTM) standard sieve analysis (ASTM D-421 and D-422). There are four primary divisions of soils: clay, silt, sand, and gravel. Coarse-grained particles include sands and gravels, while fine-grained particles include silts and clays. Sediment particles may be classified as cohesive or noncohesive. Cohesive sediments include clays, some silts, and associated organic materials. These fine-grained sediments can adsorb contaminants and form bonds between particles that result in flocculation during overland or stream flow (Novotny and Chester, 1981). Noncohesive sediments, mostly particles of sand and gravel size, are transported as individual particles. Their sorption potential is small compared to cohesive sediments.

The chart below presents characteristics of the different substrate materials.

<u>Material</u>	<u>Particle Size (mm)</u>	<u>Sorption Potential</u>
Clay	0.001-0.002	high
Silt	0.006 - 0.074	moderate
Sand	0.074 - 2	low
Gravel	2 - 75	low

The three major sources of suspended sediments in bodies of water are shore erosion, river inflow (plant and animal material), and logging, activity such as erosion from mining, human and These sediments are dispersed and settled by an agriculture. ongoing process involving wave action and currents until the sediments are either transported out of the system or are buried by further sedimentation and compaction, thus becoming a permanent part of the bottom sediments. Sediments can be resuspended by increased water velocities during storm/runoff events. However, smaller sized particles may be resuspended by lower water velocities. In some instances, a hard-packed clay may require a higher suspending velocity than gravel. Approximate velocities required to initiate movement of particles along a stream bed (traction) are presented in Table 3-3 (Hynes, 1970).

W92194F

TABLE 3-3 CRITICAL MEAN CURRENT VELOCITY (cm/sec) TO INITIATE TRACTION IN SEDIMENT NYANZA III - SUDBURY RIVER STUDY

Type of Bed	<u>Clear Water</u>	<u>Muddy Water</u>
Fine-grained clay	30	50
Sandy clay	30	50
Hard clay	60	100
Fine Sand	20	30
Coarse sand	30 - 50	45 - 7 0
Fine gravel	60	80
Medium gravel	160 - 80	80 -100
Coarse gravel	100 - 140	180

_____ W92194F

Areas with fine-grained sediments typically exhibit higher Total Organic Carbon (TOC) values than those with coarse-grained sediments. This is primarily due to the adsorptive nature of these small particles. Fine sediments that are high in TOC bind with a wide range of organic and inorganic compounds. Results from grain size and TOC analyses of sediments collected in the Study Area are presented in Appendix A.

The grain size analysis indicated that nearly 50 percent of the sediment samples from the River were predominantly silts and clays. Nearly 25 percent of the sediments were predominantly gravel or sands while the remaining 25 percent were predominantly gravel or organic material. Generally, silts and clays were found in the wider stretches of the River and the Reservoirs where flow velocities were lower, allowing fine sediments to settle out.

Sands and gravels were found in the narrower stretches of the River where flow velocities were generally higher. Selection of sampling locations was biased toward areas where finer grained sediments were deposited and could be recovered during sampling. The finer grained sediments would absorb more mercury and other contaminants than the coarser grained sediments, and thus indicate the extent of contaminant migration within the River system.

The average depth of the Sudbury River is approximately two to three feet with the exception of Fairhaven Bay, which is about six feet deep. Typical stream depths vary from reach to reach depending on the regional gradient and channel cross-sectional geometry. In addition to water depths varying as a result of the geometry and slope of the River channel, stream flow and water level elevation vary on a seasonal basis and, in part, as a function of the amount of water released at individual dams. Generally, however, Reservoirs 1 and 2 range from 10 to 25 feet deep. The USGS bathymetry report presented in Appendix I includes more detail on depths of water in Reservoirs 1 and 2.

3.6.3 Wetlands

The wetlands associated with the Sudbury River system were assessed in 1989 and reported under separate cover (Appendix F). The wetlands assessment contains a description of the wetlands and associated plant cataloging.

The Nyanza Site related contaminants appear to have had little or no visible effect on the vegetation or hydrology of the wetlands. However, wetlands adjacent to and on the Nyanza Site have been heavily disturbed by machinery and equipment during Site remediation. The absence of visible damage to vegetation is not an indication of the presence of the systemic effect of contamination

W92194F

in the wetlands. Most plant species are not killed or inhibited by levels of contamination which may be harmful to wildlife or human health.

3.6.4 <u>Ecology</u>

The Sudbury River provides a diversity of habitat types. The headwaters at Cedar Swamp is an extensive wetland with a variety of wetland types, e.g., heath bog, shrub/scrub, forested, and open water. Several abandoned mill ponds are typically followed by riffle/run rapids. Channelization has been established through some urbanized areas (Ashland and Framingham) and the impoundments of Mill Pond, Reservoir Nos. 2 and 1, and the Saxonville Impoundment have altered the natural flow and ecology of the valley. The floodplains of the Great Meadows National Wildlife Refuge provide a diverse habitat, including open water, marshes, and uplands.

The physiology, typical of much of the Sudbury River, has developed a stream community in the large water bodies that is similar to a community in standing water. The bottoms of the sluggish areas of the River include fine silts and muck with a vegetative surface layer. Occasional riffle/runs, which are highly oxygenated, are generally found, in the areas directly downstream of the impoundments and in some of the channelized urban areas. Riffle/runs were not observed beyond the quickwater after the Saxonville Dam. The stream bottom of these riffles is typically comprised of fine sand, gravel, and cobbles.

3.6.4.1 <u>Vegetation</u>

The wetlands located upriver of Reservoir Nos. 2 and 1 can generally be described as streamside wooded swamps dominated by red maple (Acer rubrum), speckeled alder (Alnus rugosa), silky dogwood (Cornus amomum), and buckthorn (Rhamnus frangula). The Reservoirs consist of a series of open water impoundments with limited wetland types. The shorelines of the Reservoirs are often bordered by mixed hardwood forests and planted stands of Northern white cedar (Thuia occidentalis) or red pine (Pinus resinosa) and usually do not have marsh wetlands along the edges. Where residences border the reservoirs, limited areas of lawn and shrubbery have been planted to the waters' edge. Downstream, as the River flows through the Great Meadows National Wildlife Refuge, extensive shrub swamp floodplains of buttonbush (<u>Cephalanthus occidentalis</u>), purple loosetrife (Lythrum salicaria), and shallow and deep marshes of cattail (Typha latifolia) appear. Marshes of broad- and narrowleaved aquatic vegetation such as pickerelweed (Pontederia cordata) and sedge (Carex sp.) and large, vigorous stands of wild rice (Zizania aquatica), and wild millet (Echinochola sp.) are also present.

W92194F

Most of the shallow water zone of the large water bodies of the River and the low gradient sections of the River (Reaches 1 and 7 through 10) have dense growth of submerged and floating-leafed plants including coontail (<u>Ceratophyllum demersum</u>), fanwort (<u>Cabomba carolinianal</u>), watermilfoil (<u>Myriophyllum sp.</u>), bladderwort (<u>Utricularia sp.</u>), slender pondweed (<u>Potemogeton pusillus</u>), ribbonleaf pondweed (<u>P. epihydrus</u>), and bushy pondweed (<u>Najas flexilus</u>). Waterlillies, (<u>Nuphar variegatum</u> and <u>Nymphaea</u> oderata) were found in coves and other low-flow areas.

The low gradient and slower flow sections of the Sudbury River through Wayland and Sudbury are bordered by stands of emergent vegetation such as cattails (<u>Typha latifolia</u>). Downstream of the Route 20 bridge, sections of the River channel support moderate to dense growth of ribbonleaf pondweed and tapegrass (<u>Vallisneria</u> <u>americana</u>). Floating duckweed (<u>Lemna</u> sp.) and water chestnut (<u>Trapa natans</u>) become abundant towards the River's confluence with the Assabet River.

3.6.4.2 <u>Fisheries</u>

The Sudbury River provides sport fishing for a variety of common warm-water game and pan species. Table 3-4 presents several species of fish that were found in the Sudbury River which are common in Massachusetts and some are typically consumed by anglers.

The Massachusetts Division of Fisheries and Wildlife annually stocks trout in Hopkinton, Southborough, and Ashland. Much of the Sudbury River is unsuitable for trout and most other sport fish (with the exception of those listed above), due to low dissolved oxygen levels in some areas of the River (DWPC, 1988)

3.6.4.3 <u>Wildlife</u>

Although the area surrounding the wetlands along the Sudbury River typically is highly developed, human activity within the wetlands themselves appears to be minimal. The variety of food sources observed in the wetlands could support many species of mammals, birds, and reptiles.

TABLE 3-4 SUDBURY RIVER FISH NYANZA III - SUDBURY RIVER STUDY

Largemouth Bass Yellow Perch Brown Bullhead Yellow Bullhead Black Crappie Bluegill White Perch Pumpkinseed Chain Pickerel Redfin Pickeral Rainbow Trout American Eel Brook Trout Common Carp Creek Anubsucker Golden Shiner

Micropterus salmoides Perca flavescens Ictalurus nebulosus Ictalurus natalis Pomoxis nigromaculatus Lepomis macrochirus Morone americana Lepomis qibbosus <u>Esox niger</u> Esox americanus Oncorhynchus mykiss <u>Anguilla rostrata</u> <u>Salvelinus fontinailis</u> <u>Cvprinus carpio</u> Semotilus atromaculatus Notenigonus crysolencas The wetlands associated with the Great Meadows National Wildlife Refuge in Reach 8 have high wildlife value. Excellent vegetative conditions exist for seed crops and wildlife plant food; the cattail marshes provide excellent migrating, nesting, and breeding habitat.

Thirteen species of fish presented in Table 3-4 were observed throughout the Sudbury River System. Several others were reported to be present. The index of fish species present is qualitative. Sampling operations targeted specific species of fish and other species may not have been noticed.

Wildlife observations provide an additional indicator of habitat type and current use. No formal wildlife inventory was conducted, although species observed during the course of the vegetation assessment were recorded. Mammal observations include woodchuck (Marmota monax) burrows, raccoon (Procyon lotor) tracks, and beaver (<u>Castor canadensis</u>). Squirrels (<u>Sciurus</u> <u>carolinensis</u>) and chipmunks (Tsmias striatus) were also sighted in trees in the wetlands. Birds present include Blue Jays (Cvanocitta cristata), Crows (Corvus brachyrhynchos), Hawks (Buteo sp.), Black-capped Chickadees (Parus atricapillus), Blue Heron (Ardea herodias), Osprey (Pandion haliaetus), and Woodpeckers. Waterfowl, including Mallards (<u>Anas platyrhynchos</u>) and Canadian Geese (Branta canadensis) were frequently observed on the impoundments. Reptiles observed included Painted Turtles (Chrysemys picta) and Green Frogs (<u>Rana_clamitans</u>).

The number of animal species observed are fewer than expected and was limited by the brief time spent at each site. During the wetlands assessment, most time was spent noting vegetation and hydrology and most visits were made during daylight hours. A comprehensive faunal study would require extended site visits during different times of the year both at night and during daylight. Lists of wildlife observed or expected to exist along the Sudbury River (including mammals, birds, fish, reptiles, and amphibians) are presented in Appendix J.

Currently endangered, threatened, and species of special concern exist in the Study Area according to the Natural Heritage and Endangered Species Program as presented in Table 3-5. Three verified vernal pools exist in Reach 6. This review was performed by the Natural Heritage and Endangered Species Program and concerns only rare and endangered species of animals and plants for which the program maintains Site-specific records.

FINAL

TABLE 3-5 ENDANGERED, THREATENED AND SPECIES OF SPECIAL CONCERN ALONG THE SUDBURY RIVER NYANZA III - SUDBURY RIVER STUDY

<u>Common Name</u>	<u>Scientific Name</u>	State Status
Reach 1		
Golden-winged Warbler	<u>Vermivora chrysotera</u>	endangered
Reach 4		
Certified Vernal Pool Englemann's Umbrella- sedge	<u>Cyperus_engelmannii</u>	sc
Reach 6		
Three Certified Vernal Pools		

Reach 7

Least Bittern	<u>Ixobychus exilis</u>	threatened
American Bittern	<u>Botaurus lentiginosus</u>	SC
King Rail	<u>Rallus elegans</u>	threatened
Common Moorhen	-	
River Bulrush	<u>Scirpus fluviatilis</u>	SC

Reach 8

Blue-spotted salamander	<u>Ambystoma laterale</u>	SC
Common Moorhen	Gallinula chloropus	SC

Reach 9

Eastern Box Turtle	<u>Terrapene carolina</u>	SC
Linear-leaved Milkweed	Asclepias verticillata	threatened
Climbing Fumitory	Adlumia fungosa	threatened
Swamp Oats	Sphenopholis pensylvanica	threatened

SC = Species of special concern
Source: Natural Heritage and Endangered Species Program
(Appendix J)

3.7 <u>Eastern Wetlands</u>

A description of the Eastern Wetlands is presented in Sections 2.2 and 3.5. The Wetlands are bordered by the old trolley line on the west, by Megunko Road and an oil storage depot on the north, and by uplands on the east and south. The Wetlands receive surficial drainage from the Site and waste effluent discharge from various manufacturing operations in the area (Ebasco, 1991).

Historical aerial photography (1938 through 1981) indicates changes in the extent of the Eastern Wetlands. The original wetland area (1938) was bisected by the old trolley line in the northeast corner of the Site and extended across Megunko Road into the present industrial area. Subsequent photos show filling of the area across Megunko Road (1952) and industrial construction in 1963, 1970, and 1979. As a result of filling and construction activities, it appears that the remaining wetlands east of the old trolley bed (Eastern Wetlands) have expanded, with more open water visible from 1963 to 1981. The Trolley Brook wetland (wetland area west of the old trolley line bed) was remediated under Operable Unit I in 1990 (Ebasco 1991).

3.7.1 Hydrology

Eastern Wetlands is a palustrine wetland system located east of the old trolley line railbed on the eastern border of the Nyanza Site. The wetland receives with surface water drainage from the eastern portion of the Nyanza Site and surrounding uplands. Surface water from the Site enters the wetland from a feeder stream in the south/southwest portion of the wetland. The stream passes under through a stone culvert and the old trolley line. Another stream feeds the wetland from uplands south and offsite. Surface water discharge from the northwestern corner of the wetland into Trolley Brook which flows northward to the confluence with Chemical Brook. It appears that the Wetlands has been bisected by an earthen dike (approximately two feet above the surface water level). A limited hydrological connection through the dike is provided by a small open culvert.

3.7.2 Sediments

Sediments within the Eastern Wetlands were typically comprised of six to 12 inches of vegetation or a peat layer followed by several feet of silty sands or fine to coarse sands. See Section 3.6.2 for a discussion of the classification system and sediment types. Grain size results are presented in Appendix A. A cross-section profile of selected soil borings within the Wetlands is illustrated in Figure 3-3 (See Figure 1-D for cross - section profile location).

W92194F

18/06/6

021411-6.046

Ş



NYANZA - OPERABLE UNIT III. ASHLAND. MA

VERTICAL SCALE IN FEET VERTICAL EXAGGERATION 50x



3.7.3 Ecology

The Eastern Wetlands has been divided into two areas by the earthen dike. The northern portion of the Wetlands consists of a fairly large (approximately two to three acres), shallow (approximately one to three feet deep) pond. The dominant wetland class consists of open water with emergent marsh habitats on the eastern and southeastern fringe of the pond. The shallow emergent habitat consists of robust, persistent emergent vegetation dominated by cattails.

The southern portion of the Wetlands has a smaller (less than one acre), shallow pond. However, in this area the emergent marsh habitats comprise a relatively larger portion of this wetland complex with common reed (Phragmites sp.) and purple loosestrife as the dominant components. The phragmites and purple loosestrife represent invasive species which typically out-compete native vegetation and form nearly monotypic plant communities. In addition, both species have limited wildlife food value and can reduce structural diversity within the Wetlands. The eastern and southern areas also have scrub-shrub and forested habitats as significant components of this Wetlands system. Common woody species within the scrub-shrub habitat include Alder-buckthorn (Rhamnus frangula) and silky dogwood. Red maple is the dominant overstory species within the forested habitats with northern red (<u>Quercus rubra</u>) and cottonwood (Populus deltoides) oak as associated species.

Few wildlife species were observed in the area during sampling operations in November 1989 and December 1990, possibly due to the sampling activities. Observations included Mallard ducks and unidentified small fish, crayfish and tadpoles.

3.8 <u>Chemical Brook Culvert</u>

3.8.1 Hydrology

A detailed description of The Chemical Brook Culvert is presented in Section 3.2 and shown in Figures 1-A and 1-D. Trolley Brook, which drains the Eastern Wetlands; and Chemical Brook, which drains the western wetland area, converge and discharge into the Chemical Brook Culvert. The culvert flows northeast approximately 2,000 feet where it discharges to the Raceway and eventually to the Sudbury River.

In 1985, the DEQE undertook an Interim Response Measure to construct a culvert from the Chemical Brook/Trolley Brook Conrail tracks culvert crossing, parallel to the tracks, to Cherry Street and Tilton Avenue. One foot of clean fill was also placed in the area of the construction.

W92194F

A remote video inspection was performed within the culvert; the results of this inspection are presented in Section 2.3.1.

3.8.2 Sediments

A visual and video inspection were performed at the Chemical Brook Culvert. These inspections indicated little deposition of sediments in the Culvert from the railroad tracks to the intersection of Main and Front Street near the end point of the video inspection. Typically, less than two inches of sediment was present in the junction box or catch basins, with less sediment present in the corrugated steel or cement Culvert sections. Portions of the Culvert have been replaced in a piecemeal fashion with newer construction. It appears that the Culvert is flushed relatively clean during high energy storm events. Exceptions to this may be in the older portions of the culvert (the rectangular These sections were not accessible for slab stone sections). visual inspection and water depth interfered with any determination of sediment presence or depth during the video inspection. Also, due to the stone construction of the culvert, the section from Front Street to the discharge point at the Raceway could not be inspected with the video system. Therefore, it is possible sediments were present in greater quantity in the sections of the Culvert with the older stone construction than in the currogated steel or cement sections. The base of the stone lined Culvert is an open stream bed, which allows water to infiltrate to the water Sediment may accumulate between cobbles or in potholes table. which may be present.

3.8.3 Ecology

The ecology of the Chemical Brook Culvert is not considered a important component of this system. The Culvert's primary function is to carry storm water and spring runoff from the Eastern Wetlands to the Sudbury River.

. 4

4.0 NATURE AND EXTENT OF CONTAMINATION

4.1 <u>Introduction</u>

This section discusses the nature and extent of contamination in water, sediment and fish collected in the Sudbury River and associated wetland systems. Data is organized by Reach; other areas, such as Cold Spring Brook and the Eastern Wetlands are discussed separately. Contaminant levels found within each Reach are discussed and compared to background levels. Results of Phase I and II sampling events, including the Eastern Wetlands drilling program are included. This section also presents a discussion of other potential sources of contamination. Contaminants associated with Nyanza Site sources which were identified by previous investigations are termed Site-related contaminants. Contaminants associated with "point" and "non-point" off-site sources are termed Study Area contaminants.

Discussions focus on Site-related contaminants. These include (but are not limited to) mercury, chromium, 1,2-dichlorobenzene (1,2-DCB), chlorobenzene, and TCE. Of these, only mercury (and associated methylmercury) is considered unique to the Nyanza Site discharges. Later sampling and analysis events were limited to inorganic contaminants due to low concentrations of organics detected and the relative importance of mercury.

Other organic compounds and metals have been reported to be associated with Site; however due to the widespread presence of other potential contributors of similar contaminants, no clear conclusion can be drawn pertaining to the origin of these other contaminants. Lead is of particular interest. Although it is a documented Site-related contaminant, lead is ubiquitous in the Study Area. Widespread use of lead in transportation, agriculture and the chemical industry has resulted in elevated levels of lead in the Sudbury River in no distinct pattern of occurrence. This type of widespread occurrence is discussed in Section 4.2.

The analytical database is presented in Appendices λ through D. The data were assimilated into summary tables (Tables 4-1 through 4-3) presented in this section. Other analytical data is summarized elsewhere as appropriate.

4.1.1 Monthly Water Quality Sampling Results

A discussion of the monthly sampling results is included as Appendix L, which describes potential environmental effects caused by fluctuations of each parameter analyzed.

TABLE 4-1

SUMMARY OF SURFACE WATER AND SEDIMENT ANALYTICAL RESULTS NYANZA III REMIEDIAL INVESTIGATION SUDBURY RIVER STUDY

	1						SEDIMEN	IT - INORGANI	CS						
					_		U	NITS: mg/kg							
INORGANICS	REACH 1,	BACKO	ERVOIR, REBEI				I	REACH 2				F	EACH 3		
	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE	AVERAGE REPORTED	MAXMUM	MINIMUM
ALUMINUM	11 OF 11	8200 81	8200 81	10000	2070	12 OF 12	7906 00	7008.33	16600.00	3010.00	23 OF 23	12210.00	12210.00	21500.0	5580.0
ANTIMONY	1 OF \$	8.45	0 05			O OF O					1 OF 22	18.00	4,11	18.0	18.0
ARBENIC	80F 11	10 64	8 74	21.1	25	10 OF 12	6 11	8 95	14.90	2 55	17 OF 23	9.14	7.61	21.5	2.1
BARUM	11 OF 11	60 60		178	13.3	12 OF 12	71.48	71 48	209.00	13.90	23 OF 23	69.65	69.65	116.0	16.2
BERYLLIUM	8 OF 10	0 83	0 85	18	0 37	7 OF 10	1.03	0 87	190	0.10	18 OF 23	1.84	1.52	4.3	0.5
CADMIUM	00#0					0 OF 0]	6 OF 23	11.13	4.96	10.0	1.3
CALCIUM	110#11	2101 01	2101 01	4020	588	12 OF 12	3097 42	3097.42	10500 00	620 00	22 OF 23	2699.57	2008.54	4910 0	607.0
CHROMIUM	# OF 11	24 20	22 40	66 2		11 OF 12	38 32	34.00	216 00	5.70	22 OF 23	305.62	292.82	2620.0	13.9
COBALT	10 OF 11	11 00	10 65	18.8	37	10 OF 11	13 15	13 66	27 70	2.90	7 OF 23	18.76	11.27	28.9	9.0
COPPER	8 OF 11	66 76	73 02	340.4		\$OF 12	50 03	40 19	164 00	10.80	21 OF 23	200.73	184.47	454.0	13.3
IRON	11 OF 11	22124.00	22124 08	110000	4110	12 OF 12	16252 50	16252 50	34800 00	6500 00	23 OF 23	20973.33	20973 33	47400 0	7980.0
LEAD	11 0F 11	85 09	85.00	240	6	12 OF 12	58 07	50.07	295.00	19.90	23 OF 23	137.97	137.97	265.0	8.0
MAGNEBIUM	11 OF 11	1877 82	1877 82	3200	762	12 OF 12	1774 75	1774.75	3720.00	825.00	23 OF 23	3054.79	3054.79	5030.0	1480.0
MANGANESE	11 OF 11	434 61	434 51	1840	85 P	12 OF 12	887 87	867.87	4100.00	76.40	23 OF 23	391.42	391.42	847.0	68.0
MERCURY	20F 0	1 05	0 27	1 50	05	BOF 0	6 81	3.61	30.60	0.22	21 OF 22	17.43	15.98	54.6	0.2
NICKEL	5 OF 11	20 98	11 36	61	32	4 OF 12	9.67	7.47	19.10	5.45	11 OF 23	48.65	26.43	88.8	14.0
POTASSIUM	11 OF 11	872 64	872 84	1365	224	12 OF 12	585.02	565.02	1280.00	175.00	17 OF 23	1407.89	1076.28	2260.0	412.0
BELENNM	20F7	3 00	1 04	3.1	20	0 OF 8					0 F 23	1.82	0.01	4.0	0.3
BILVER	0 OF 9					00F 8					0 OF 23				
SODIUM	9 OF 11	214 02	188.17	300	72	12 OF 12	407.87	407.87	1480.00	48.60	5 OF 23	317.25	164.40	425.0	201.0
VANADIUM	10 OF 11	2171	30.10	48.45	10 2	11 OF 12	20.48	10.00	45.80	6.80	22 OF 23	38.11	35.02	67.9	11.7
ZINC	11 0F 11	133 00	133 80	639	13 8	11 OF 12	136.62	128.44	330.00	26.65	21 OF 23	192.20	177.08	435.0	27.0
THALLEUM	0 OF 0					0 OF 0					0 OF 22				

(

Frequency: Average Detected: Average Reported:

The arithmatic average of only detected concentrations The arithmatic average of the detected concentrations and one half the detection limit

Number of detections in the number of samples validated

for undetested values reported

1

٠

.

.

FINAL

t

TABLE 4-1

SUMMARY OF SURFACE WATER AND SECTIMENT ANALYTICAL RESULTS

NYANZA IN REMIEDIAL INVESTIGATION SUDBURY RIVER STUDY

1

W92194F PAGE 2

٦ ۵

							BEDIMEN	T - INORGANI	CS						
		-			•·· · · •	······	UN	118. mg/kg							
INORGANICS			NEACH 4				f	EACH 5				1	REACH		
	FREQUENCY	AVENAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINING	FREQUENCY	ÁVERAGE DETECTED	REPORTED AVERAGE	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUI
ALUMINUM	13 OF 13	12821 54	12021 54	22000 00		8.0F 8	7023 33	7023 33	20700.00	3140 00	21 OF 21	11636.00	11638 40	19300.00	4260.0
ANTIMONY	0 OF 12					O OF C					0 OF 17				
ARSENIC	0 OF 13	13 00	10 00	32 30	40	30F 6	8 05	3 83	9.20	3 65	20 OF 21	12.03	11.10	24.00	0.7
BARIUM	13 OF 13	00 01	00 9 1	135 00	212	80F 8	73 18	73.18	225.00	27.20	21 OF 21	85.84	65.80	143.00	21.7
BERYLLIUM	7 OF 12	1 11	0 80	3 60	0 54	1 OF 6	0 25	0 48	0 25	0.25	2 OF 17	1.85	0.49	2.00	1.7
CADMIUM	0 OF 13	8 88	401	14 80	44	0 OF 0					8 OF 17	7.50	3 63	13.60	3.4
CALCIUM	13 0F 13	8323 40	2323 40	4410 00	015	OF 6	2273 33	2273 33	4070 00	1170 00	21 OF 21	3676.80	3677.60	7185.00	834.0
CHROMIUM	12 OF 13	78.28	71 01	224 00	11	3 OF 8	30 82	17 95	60 60	13.65	20 OF 21	69 84	66.60	261.00	11.0
COBALT	8 OF 13	18 88	14 00	34 60	10 8	1 OF 0	8 80	4.45	6 90	8.90	11 OF 17	20.19	18 18	35.60	0.9
COPPER	10 OF 13	140 01	115 85	332 00	20.0	30F 8	67 40	41 67	158 00	10.90	19 OF 21	127.32	113.50	303.00	27.0
IRON	13 OF 13	20176 46	20178-48	37100.00	6870	6 OF 6	10400 67	10486 87	18000 00	6960 00	21 OF 21	20672.38	20588.10	44200.00	7920.0
LEAD	13 OF 13	83 88	83 86	219 00	51	8 OF 8	237 99	237 99	809.00	24 95	21 OF 21	265.65	285.70	876.00	2.4
MAGNEBIUM	13 OF 13	2342 31	2342 31	3500 00	1270	0 OF 0	1838 33	1838 33	3300 00	1320.00	21 OF 21	3402.14	3401.90	4580.00	1010.0
MANGANESE	13 OF 13	540 23	548 23	1030 00	115	0 0F 0	699 83	699 63	1935 00	194 00	21 OF 21	569 26	589 30	1415.00	126.0
MERCURY	11 OF 12	3 📾	3 38	7 30	0 81	50F 6	1 17	0.98	4.10	0 30	14 OF 20	5.00	3.30	17.60	0.3
NICKEL	11 OF 13	36 21	20.00	63 00	0 2	1 OF 6	16 10	6.79	18.10	16.10	13 OF 21	28.05	20 30	78 40	11.0
POTASSIUM	7 OF 13	801 71	604.34	1340 00	556	3 OF 6	824 17	812 50	1298.50	555.00	14 OF 21	1053.57	708 30	1650.00	263.0
BELENIUM	8 OF 12	1 00	174	4 00	0.45	20F8	0 43	0.48	0 44	0.41	7 OF 19	3.13	1.37	6.10	1.7
BILVER	00F8					10F 6	8 40	1 90	6 40	6.40	0 OF 9				
BODIUM	10F13	1230 00	211 92	1230 00	1230	2 OF 8	599 73	260 83	1110.00	69.45	9 OF 21	484.00	268 60	1720.00	113.0
VANADIUM	13 0F 13	34 83	34 83	87 10	12 6	0 OF 0	17 05	17.05	30.70	9.40	17 OF 21	39.33	55.90	83.00	10.0
ZINC	12 OF 13	173 88	101.05	327 00	28 J	50F 8	233 48	199 88	765.00	72 30	20 OF 21	306.00	292.30	564.00	30.1
THALLIUM	0 OF 10					0 OF 0					0 OF 0				

Frequency:

Number of detections in the number of samples validated

Average Detected: The arithmetic everage of only detected concentrations Average Reported:

The arithmetic average of the detected concentrations and one half the detection limit for undetected values reported

TABLE 4-1 SUMMARY OF SURFACE WATER AND SEDMENT ANALYTICAL RESULTS NYANZA III REMIEDIAL INVESTIGATION SUDBURY RIVER STUDY PAGE 3

							BEDIMEN	T - INORGAN	CS CS						
INORGANICS		 P	EACH 7					REACH 6			REACH 9				
	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXMUM	MINIMUM	FREQUENCY	AVERAGE	AVERAGE	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE	AVERAGE REPORTED	MAXMUM	MINIMUM
	13 OF 13	7730 82	7736 90	16800 00	1840 00	SOF 6	6638 67	6636.00	6310.00	4310.00	3 OF 3	9606.67	9806.00	11700.00	7020.00
ANTIMONY	0 OF 12					0 OF 1					0 OF 0				
ARSENIC	10 OF 13	11 04	10 13	40 60	2 30	8 OF 8	12 13	12.13	30 00	4 20	3 OF 3	32.03	32 00	64.60	15.30
BARIUM	13 OF 13	82 42	92 42	272 00	15 80	8 OF 8	113 42	113 41	222 00	26.60	3 OF 3	118.70	118.00	151.00	55.10
BERYLLIUM	2 OF 10	0 57	0 57	0 81	0 52	0 OF 1					0 OF 0				
CADMIUM	2 OF 12	11.50	2 21	17.80	5 10	1 OF 3	4 30	2.08	4.30	4.30	2 OF 2	7.05	7.05	8.60	5.50
CALCIUM	13 OF 13	2635 15	2835 15	6330 00	500 000	8 OF 6	4121 87	4121.00	5210 00	2770.00	3 OF 3	5150.00	5150.00	5620.00	4160.00
CHROMIUM	10 OF 13	01 50	47 96	208 00	10 30	50F 8	26 30	22 00	36 40	14 20	3 OF 3	50.27	50.26	78.00	17.80
COBALT	4 OF 12	8.40	7 05	10 40	7 80	3 OF 4	6 33	7 20	6.90	7.30	3 OF 3	14 20	14.20	16 60	9.30
COPPER	8 OF 13		85 94	278 00	18 80	8 OF 8	67 62	67 61	98.60	14.40	3 OF 3	63.33	63.33	135.00	17.10
IRON	13 OF 13	14626 92	13820 92	96000 00	2870 00	0 OF 8	15581 67	15581 00	32000 00	6710 00	3 OF 3	17533.33	17533.30	18000 00	16700.00
LEAD	13 OF 13	117.44	117 84	526 00	2 60	0 OF 6	86 77	86 70	100 00	6 60	3 OF 3	115.27	115 28	164 00	4.80
MAGNEBIUM	13 OF 13	2558 31	2306 31	4670 00	411 00	80F 8	1568 33	1568.30	1980.00	1130.00	3 OF 3	1863.33	1883.30	2090.00	1700.00
MANGANESE	13 OF 13	300 20	300 20	1430 00	57 60	0 OF 8	1680 17	1660.10	5660.00	123.00	3 OF 3	429 33	429.33	516.00	348.00
MERCURY	8 OF 11	2 08	1 52	8 90	0 21	5 OF 5	1 62	1.62	2 10	0 88	2 OF 2	3.15	3 15	3.90	2.40
NICKEL	5 OF 13	22 44	11 20	44 10	e 70	2 OF 5	11 80	7.08	13.10	10.50	2 OF 2	24.95	24.95	26.10	21.80
POTABBILM	0 OF 13	568 67	521 36	1210 00	1 15 00	0 OF 0	380.00	360.00	437.00	303.00	3 OF 3	577.67	577.66	759.00	363.00
SELENIUM	5 OF 13	2.48	1 18	7 20	0 85	3 OF 6	1.43	0 75	1.60	1.00	0 OF 0				
BILVER	0 OF #					0 OF 1					0.OF 0				
BODIUM	0 OF 13	211 20	174 40	374 00	75 20	0 OF 6	404 00	404 00	614 00	142 00	3 OF 3	237.33	237.33	379.00	138.00
VANADIUM	12 OF 13	21 04	19 55	47 00	6 30	5 OF 5	9 72	9.72	12.80	6.60	3 OF 3	18.40	16 40	23.10	11.40
ZINC	11 OF 13	198 30	170.08	\$46.00	25 50	9 OF 9	187 37	187 36	329.00	38 20	3 OF 3	197.13	197.13	319.00	20.40
THALLINM	00F7					OOF 1					OOF 0				

(

 Frequency:
 Number of detections in the number of samples validated

 Average Detected:
 The arithmetic average of only detected concentrations

 Average Reported:
 The arithmetic average of the detected concentrations and one half the detection limit for undetected values reported

FINAL

(

TABLE 4-1

BUMMARY OF SURFACE WATER AND BEDIMENT ANALYTICAL REBULTS NYANZA IN REMIEDIAL INVESTIGATION

SUDBURY RIVI PAGE 4	ERSTUDY												N/504.05		
	T		AVENAUE				OF DIMEN	AVERAUE	~				AVENAGE		
	(BEDIMEN		6						
								1119: mg/kg			r				
INDRGANICS	1		REACH 10				1	EASTERN WET	LANDS		•	CHEMICAL BRO	OK CULVERT		
	FREQUENCY	AVERAGE DETECTED	REPORTED AVERAGE	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE DETECTED	REPORTED AVERAGE	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE DETECTED	REPORTED AVERAGE	MAXMUM	MINIMUN
ALUMNUM	50F 8	8790 00	6780	17800 00	1840 00	10 OF 10	7218 00	7216	11050 00	4350 00	20F 2	3520.00	3520.00	3690.00	3350.00
ANTIMONY	0 OF 5					1 OF 10	8 70	8 07	8.70	6 70	0 OF 2				
ARBENIC	5 OF 5	5 00	5 99	12 20	2 80	10 OF 10	8 59	6 59	12.70	2.30	20F2	4 30	4.30	6.60	2.00
BAFIUM	50F 6	33 61	33 61	64 30	16 00	8 OF 10	38 13	33 615	57.70	25.50	2 OF 2	18.65	18.65	25 90	11.40
BERYLLIUM	0 OF 8					10 OF 10	2 21	2 2 1	4.00	0.52	2 OF 2	0.45	0.45	0.56	0.34
CADMIUM	0 OF 5					0 OF 10					0 OF 2				
CALCIUM	SOF 8	1620.00	1620	2480 00	870 00	10 OF 10	6941 50	8941.5	18400.00	714.00	2 OF 2	1048 00	1048.00	1130.00	966.00
CHROMIUM	4 OF 5	12 13	11 24	17 10	7 30	€OF 10	138 79	123 68	462 00	16 80	20F2	82.30	62.30	135.00	29.80
COBALT	5 OF 8	4 24	4 24	8 40	2 80	3 OF 10	15 83	7 41	31.20	6 80	2 OF 2	4 40	4.40	4.60	4.20
COPPER	2 OF 5	20.15	14 115	31.00	25 30.	#OF 10	75 54	63.73	120.00	19 60	2 OF 2	63.35	63.35	92.50	34.20
IRON	5 OF 8	7102 00	7102	10800 00	3090 00	10 OF 10	11871.00	11671	38200.00	1980.00	2 OF 2	20505.00	20505 00	33200.00	7610.00
LEAD	5 OF 5	27 44	22 48	33 00	7 80	10 OF 10	72 58	72 58	142 00	8 50	2 OF 2	38.25	38 25	57 50	19.00
MAGNESIUM	5 OF 5	1277 20	1277 2	2510 00	355 00	10 OF 10	858 55	958 55	2010 00	195 00	2 OF 2	1172.00	1172 00	1890.00	454.00
MANGANESE	5 OF 5	178 04	178 04	336 00	84 80	0OF 10	317 17	289.11	1320.00	28 40	20F2	90 60	90.80	119.00	62.20
MERCURY	2 OF 5	0 37	0 101	0 53	0 20	8 OF 10	44 64	35 878	152 00	0 23	2 OF 2	5.60	6.60	7.10	6.50
NICKEL	5 OF 5	7 37	7 37	11 10	4 80	5 OF 10	16 98	10.1425	40.40	5.40	0 OF 2				
POTASSIUM	5 OF 5	527 40	527 4	1360.00	100 00	●OF 10	293 47	314.12	656.00	92.20	2 OF 2	314.50	314.50	326.00	303.00
SELENIUM	0.04 5					0 OF 10	2 55	1.43	6.50	0.54	0 OF 2				
SILVER	0 OF 5					0 OF 10					00F2				
SODIUM	50F 5	127 01	127 01	267 00	42 70	8 OF 10	1096 03	911 495	4830 00	20.20	2 OF 2	161.50	161.50	165.00	158.00
VANADIUM	5 OF 5	0 18	9 15	15 80	4 90	10 OF 10	17 67	17.57	37.40	5.70	2 OF 2	12.00	12.00	18.00	6.00

78 25

65 3925

10.50

164.00

2 OF 2

0 OF 2

92.95

92.95

4-5

Frequency: Average Detected: 5 OF 8

0 OF 8

ZINC

THALLNM

Number of detections in the number of samples validated

30 35

63 40

20 80

8 OF 10

0 OF 10

The arithmetic everage of only detected concentrations

38 35

The arithmetic average of the detected concentrations and one half the detection limit Average Reported: for undetected values reported

FINAL

103.00

82.90

W92194F

TABLE 4–1 SUMMARY OF SURFACE WATER AND SEDIMENT ANALYTICAL RESULTS

NYANZA III REMIEDIAL INVESTIGATION

SUDBURY RIVER STUDY

PAGE 5

Į	ł						BEDIMEN	T - INORGANI	cs						
					,			llTS: mg/kg						· · · · · · · · · · · · · · · · · · ·	
INORGANICS		c	NUTFALL CREE	ĸ			I	RACEWAY					COLD SPRING	8ROOK	
	FREQUENCY	AVERAGE DETECTED	AVERAOE REPORTED	MAXIMUM	MINIMUM	FREQUENCY	ÁVERAGE DETECTED	AVERAGE REPORTED	MAXMUM	MINIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXMUM	MINIMUM
ALUMINUM	3 OF 3	5676 67	5676 67	11900.00	2480 00	4 OF 4	22850 00	22650.00	29200	15400	20F 2	9225 .00	9225.00	10700.00	7750.00
ANTIMONY	0 OF 3					0 OF 4					0 OF 2				
ARBENIC	3 OF 3	2 83	2 83	4 10	2 20	4 OF 4	24 98	24 98	37.2	18 65	1 OF 2	4.30	3.50	4.30	4.30
BARIUM	3 OF 3	31 56	31 66	56 50	12 85	40F 4	44 10	44 10	72.1	32.7	2 OF 2	59.35	59.00	85.50	53.20
BERYLLIUM	10F3	2 20	1 07	2 20	2 20	40F 4	7 08	7 08	10.15	4	10F 2	1.90	1.00	1,90	1.90
CADMIUM	0.0F.3					2 OF 4	4 30	2 28	5.6	3	0 OF 2				
CALCIUM	3 OF 3	1968 83	1000 83	4340 00	708 50	4 OF 4	2766 25	2768 25	4090	1380	20F 2	3255.00	3255 00	4240.00	2270.00
CHROMIUM	3053	341 80	341 60	998 00	18 00	40F 4	155 50	155.50	208	129	1 OF 2	18.70	15.15	16.70	16.70
COBALT	20F 3	3 85	4 55	4 30	3 60	4 OF 4	114 89	114.69	235	65.5	2 OF 2	12.65	12.60	14.30	11.00
COPPER	3 OF 3	137 83	137 63	358 00	19 80	4 OF 4	217.75	217.75	306	142	2 OF 2	31.30	31.50	32 90	29.70
IRON	3 OF 3	11325 00	11325.00	20200 00	5725 00	40F 4	27375 00	27375 00	35500	13000	2 OF 2	13800.00	13800.00	15200.00	12000.00
LEAD	3 OF 3	 105 #5 	105 85	233 00	27 80	4 OF 4	435 75	435 75	758	267	2 OF 2	234.50	234 00	328.00	141.00
MAGNEBIUM	3053	1858 33	1856 33	3090 00	829.00	4 OF 4	2653 75	2653.75	3660	1810	2 OF 2	3000.00	3000.00	3330.00	2670.00
MANGANESE	30F 3	184 57	164 57	308 00	81.40	40F 4	458 25	458 25	864	110	2 OF 2	858.00	658 00	854 00	462.00
MERCURY	3 OF 3	35 33	35 33	99 2 0	1 80	4 OF 4	0 7 1	0 71	0.97	0.475	0 OF 2				
NICKEL	0 OF 3					40F4	80 89	60 69	186	45.7	0 OF 2				
POTASSIUM	3 OF 3	715 00	715 00	833 00	405 00	4 OF 4	1186.00	1166.00	1570	666	2 OF 2	1545.00	1545.00	2020.00	1070.00
BELENIUM	0 OF 3					0 OF 4					0 OF 2				
SILVER	0 OF 3					00F 4					0 OF 2				
SODIUM	20F 3	80.15	100 65	64 70	85 80	40F4	81375	813.75	1120	417	1 OF 2	426.00	244 50	428 00	428.00
VANADIUM	3 OF 3	17 28	17 20	38 10	6 25	4 OF 4	44 50	44 50	51.2	35.6	2 OF 2	31.10	30.90	41.40	20.60
ZINC	3 OF 3	100 30	100.30	380.00	63 80	4 OF 4	365 66	365 86	542	259.5	20F 2	148.00	148.00	166.00	130.00
THALLRIM	0 OF 3					0 OF 4					0 OF 2				

Frequency: Average Detected: Average Reported:

Number of detections in the number of samples velidated

cted: The arithmetic average of only detected concentrations

orted: The arithmetic average of the detected concentrations and one half the detection limit for undetected values reported

FINAL

TABLE 4-1 TABLE 4-1 SUMMARY OF SURFACE WATER AND SECIMENT ANALYTICAL RESULTS NYANZA III REMIEDIAL INVESTIGATION SUDBURY RIVER STUDY PAGE 6

	1			SEDMEN	IT - INORDA	NICO						·	······································	·····			
				U	atta mang												
INORGANICS			CORDERING W	ETLAND				HEARD POND			EASTERN WETLANDS DRILLING 0 TO 2 FOOT DEPTH INTERVAL						
	FREQUENCY	AVERAGE DETECTED	AVERAGE	MAXONALM		FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE DETECTEL	AVERAGE REPORTED	MAXIMUM	MINIMUM		
ALUMINUM	10 CF 10	12474 00	12474 000	19300 00	988 0 00	1 OF 1	10580 00	10580 00	10580.00	10580.00	23 OF 32	9356.26	8024 22	15400.00	4570.00		
ANTIMONY	0 OF 0					0 OF 1					0 OF 31						
ARSENIC	8 CF 10	5 42	0 220	11 80	3 90	1 OF 1	10 65	10 85	10.65	10.65	14 OF 32	2.63	1.71	7.00	0.91		
BARINM	10 OF 10	30.13	38 125	87 60	18 10	1 OF 1	89 60	89 80	89 80	89.60	32 OF 32	32.54	32.54	64.70	12.90		
BERYLLIUM	8 OF 8	0 01	0 816	1 80	0 20	0 OF 1				1	26 OF 32	2.10	1.80	8.20	1.20		
CADMIUM	0 0 F 8					0 OF 1					2 OF 31	2.60	0 65	3.40	1.60		
CALCIUM	10 OF 10	1363 00	1363 000	3000 00	326 00	1 OF 1	4200 00	4200.00	4200.00	4200 00	32 OF 32	2797.81	2797.81	13400.00	1290.00		
CHROMIUM	9 OF 10	25 02	24 010	101 00	11 60	1 OF 1	40 20	40 20	40 20	40.20	25 OF 32	51.71	41 82	424.00	4.80		
COBALT	0 CF 9	7 30	7 394	13.40	4 20	0 OF 1				1	17 OF 31	12.85	7.66	74.00	3.30		
COPPER	4 OF 10	34 30	18 013	83 40	14 20	1 OF 1	138.00	136 00	136 00	138.00	17 OF 32	57.10	31.60	319.00	8.10		
FION	10 OF 10	14261 30	14281 500	30700 00	8850 00	1 OF 1	16400 00	16400 00	16400.00	16400.00	25 OF 32	7641.60	6311.82	21900.00	3720.00		
LEAD	10 OF 10	71.18	71 180	754 00	0 40	1 OF 1	149 00	149.00	149 00	149.00	21 OF 32	302.42	201 90	5760.00	5.30		
MAGNEBIUM	10 OF 10	2082 00	\$082 000	3060 00	1400 00	1 OF 1	2370 00	2370 00	2370 00	2370.00	31 OF 32	1586.71	1554-16	2820.00	360.00		
MANGANESE	10 OF 10	297 92	207 015	1070 00	78 45	1 OF 1	335 50	335.50	335 50	335.50	30 OF 32	90 92	66.31	476.00	30.40		
MERCURY	5 OF 8	180	1 07 1	7 80	0 13	1 OF 1	3 50	3 50	3 50	3 50	26 OF 32	7.29	6.12	91.50	0.17		
NICKEL	2 OF 10	17 65	7 180	21 80	13 80	1 OF 1	11 30	11.30	11.30	11.30	14 OF 31	11.77	7.50	23.4	2.3		
POTASSIUM	10 OF 10	474 80	474 800	Ø23 00	205 00	1 OF 1	703 90	703 50	703 50	703.50	32 OF 32	512.27	512.27	934.00	150.00		
SELENIUM	1 OF 8	1 20	0 070	1 20	1 20	0 OF 1					0 OF 31						
BILVER	0 CF 0					0 OF 1					0 OF 24						
BODIUM	8 QF 10	151 10	113 400	555 00	77 80	1 OF 1	729 00	729 00	729 00	729 00	15 OF 32	867.73	369.11	3750.00	137.00		
VANADIUM	10 QF 10	29 07	20 005	87 90	17 70	1 OF 1	21 70	21.70	21.70	21.70	30 OF 31	13.92	13.61	22.60	6.10		
ZINC	4 CF 10	75 25	41 078	127 00	34 80	1 OF 1	302 00	302.00	302.00	302.00	23 OF 32	54.34	42 55	290.00	18.20		
THALLIUM	0 QF 8					0 OF 1					2 OF 31	1.30	0.48	1.40	1.20		

W92194F

ś

Number of detections in the number of samples validated Frequency: The arithmetic average of only detected concentrations

Average Detected: Average Reported:

The arithmetic everage of the detected concentrations and one half the detection limit for undetected values reported

FINAL

(

W92194F

4-8

TABLE 4~1

FABLE 3-1 SUMMARY OF SURFACE WATER AND SECIMENT ANALYTICAL RESULTS NYANZA IR REMIEDIAL INVESTIGATION SUDBURY RIVER STUDY PAGE 7

							SEDIMEN	T - ORGANIC	s				· ··· · ·		
ORGANIC	ne	BACI NCH 1, BUDBUR	GROUND / RESERVOR, I	RESERVOR 3			UNI	REACH 2					REACH 3		
	FREQUENCY	AVERACE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM	FREQUENC	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM
METHYLENE CHLORIDE	0 OF 4					2 OF 12	6 000	8 000	8 000	4.000	0.OF 8				
ACETONE	1 OF 4	to 00	255 25	10 0	10 0	8 OF 13	201 833	120 538	310 000	48 000	4 OF 8	182 50	131 21	310.0	37.0
1,2-DICHLOROETHENE (TOTAL)	0 OF 3					10F 0	31 000	8 722	31.000	31.000	0 OF 6				
METHYL ETHYL KETONE	0 OF 3					4 OF 8	47.000	27 781	80.000	26.000	30F 6	26 33	14 06	32.0	17.0
TRICHLOROETHENE	005 3					0.0F 9					0 OF 9				
BENZENE	0 05 3					0 OF 0					0 OF 9				
TETRACHLOROETHENE	0 OF 3					0 OF 9					0 OF 9				
TOLUENE	0 OF 3					0 OF 9					0 OF 9				
CHLOROBENZENE	0 OF 3					2 OF 1	23,500	9 444	27.000	20.000	0 OF D				
1,3-DICHLOROBENZENE	0 OF 3					0.0F 0					0 OF 9				
1.4-DICHLOROBENZENE	0 OF 3					2 OF 10	770.000	641.000	1100.000	440.000	10F 8	125 50	1101 72	125.5	125 5
1.2-DICHLOROBENZENE	007 3					3 OF 10	1460.000	878.000	1800 000	580.000	0.0F 9				
(3-AND/OR 4-)METHYLPHENOL	0 07 3					0 OF 9				• • • • • • • • • •	0 OF 9				
NITROBENZENE	0 OF 3					O OF 9					0 OF 9				
BENZOIC ACID	0 OF 3					0 OF 5					10F 9	340.00	5404 44	340.0	340.0
1.2.4 - TRICHLOROBENZENE	0 OF 3					2 OF 9	670 000	634 889	780.000	580.000	3 OF 9	125 33	1062.58	180.0	AR 0
NAPHTHALENE	0 OF 3					10F 9	1180.000	599 444	1180.000	1160 000	105.9	120.00	1111.39	120.0	120.0
	005.3					205 6	275 000	546.887	410.000	140 000		120 00	1111.00	120.0	120.0
ACENAPHTHMENE	1					100	202 750	404 111	500.000	93,000	205 0	540.00	A1A A1	880.0	120.0
ACENADITIENE						100	203 7 30	544 333	1235.000	83.000	305 0	100.00	1100 17	100.0	120 0
DIDENTOELIDAN						305 0	478 007	544 555	1233 000	450.000		100 00	1109.17	100 0	100.0
							005 000	111 000	1000 000	150.000	00000				
							444 000		1500.000	07.000				000.0	
							466 000	607.111	1500.000	87.000	204 9	180.00	1090 83	2000	120.0
	004 3										30 9	180.00	1019.17	240.0	140.0
PHENANIPHENE		150.00	3/8 / 5	160.0	140 0	1000 11	1775 500	1634 545	10850 0	480.000	901-9	1367 78	1367 78	7300 0	65 0
	000 3					0.05 10	755 750	753 850	2900 000	84.500	401-9	164 /5	1052 17	240 0	49 0
UI-N-BUITLPHIHALAIE	000 3					0.01-0					201-8	2855.00	1645 28	5200.0	110.0
FLUORANTHENE	3 OF 4	218 00	414 25	350 0	670	11 OF 11	2014 091	2014 091	9250.000	450.000	9 OF 9	2331 67	2331.67	12000 0	170.0
PYRENE	3 OF 4	180 67	385 50	270 0	52 0	10 OF 11	1819 000	1669 091	8900 000	510.000	9 OF 9	2461 11	2461 11	11000.0	130 0
BENZYL BUTYL PHTHALATE	0 OF 3					007 9					0 OF 9				
BENZO (A) ANTHRACENE	0 OF 3					00F 11	1081 111	1105 455	4400 000	420 000	7 OF 9	1195.00	1346 11	4500.0	55.0
CHRYSENE	2 OF 4	160-00	383 75	180 0	140 0	10 OF 11	966 500	1092 273	4200 000	210 000	8 OF 9	1741 88	1737 22	7700 0	100.0
BIS(2-ETHYLHEXYL) PHTHALATE	0 OF 3					OF 0	894 667	650 889	2000 000	138 000	5 OF 9	524 00	790 83	910 0	320.0
DI-N-OCTYLPHTHALATE	0 0 2 3					0 OF 0					0 OF 9				
BENZO (B) FLUORANTHENE	1 502.3	185.00	300 25	200 0	170 0	9 OF 11	973 889	1034 091	3550 000	350 000	80F 9	1470 00	1495 58	4400 0	110.0
BENZO (N) FLUORANTHENE	005 3					0 OF 10	921 667	969 500	3800 000	240 000	80F 9	1468 50	1494 22	3900 0	73.0
BENZO - A - PYRENE	105.3	e1 00	435 33	81 0	910	7 OF 10	1070 714	1027 500	3750 000	240 000	8 OF 9	1147.75	1209 11	4400 0	72 0
INDENO (1,2,3-CD) PYRENE	0 OF 3					2 OF 1	532 500	826 111	900 000	165 000	7 OF 9	474 36	785 61	1800 0	51.0
DIBENZO (A, H) ANTHRACENE	005 3					5 OF 0	88 500	514 222	93 000	60 000	2 OF 8	220 00	1109 67	310 0	130.0
BENZO (GHI) PERYLENE	0 OF 3					301.0	368 633	593 500	650 000	128 500	7 OF 9	457 78	772 72	1600 0	55.0
4.4 -DDE (P,P -DDE)	0.0F 3					10F 0	58 000	16 583	58 000	58 000	1 OF 9	60 00	32 08	60 0	60.0
4,4'-DDD (P,P -DDD)	0 OF 3					I OF 0	30 570	30 869	30 570	30 570	30F 9	169 33	84 78	370.0	53 0
4,4'-DDT (P,P'-DOT)	0 OF 3					4 OF 10	413 875	172 425	1400 000	71 000	0 OF 9				
GAMMA-CHLORDANE /2	005 3					0 OF 0					0 OF D				
PC8-124 (AROCLOR 1254)	0 07 3					0.0F 9					0 OF 9				
MONOMETHYLMERCURY DIMETHYLMERCURY	0(7 3					10F 3 00F 3	312	109 000	312	312	1 OF 18 1 OF 18	26 J 19 I	108 55 7 44	26 3 19 1	26 3 19.1

(

FINAL

٠

(

4-9

TABLE 4-1

SUMMARY OF SURFACE WATER AND BEDIMENT ANALYTICAL RESULTS NYANZA IN REMEDIAL INVESTIGATION SUDBURY RIVER \$1007 PAGE 5

(

DPANNC COMPORTING APRACH MARACH						-	•	SEDIMEN	IT - OFIGANICS	3							
DPDANC PECLEV PECLEV PECLEV PECLEV PECLEV DEPLEV DEPLEV<						··		!	UNITS ug/kg								
ARTINGE ARTINGE <t< th=""><th>ORGANIC</th><th></th><th>I</th><th>REACH 4</th><th></th><th></th><th>l</th><th>I</th><th>REACH 5</th><th></th><th></th><th colspan="6">REACH 6</th></t<>	ORGANIC		I	REACH 4			l	I	REACH 5			REACH 6					
METHYLE CLARINGE 0 07 9 0 07 1 0 07 2 6 45500 300 300 300 300 300 300 300 300 300 300 300 300 300 300 300 400 300 400 300 400 300 400 300 400 300 400 300 400 300 400 300 400 300 400 300 400 300 400 300 400 300 400 300 400 300 400 <		FREQUENCY	DETECTED	AVERAGE REPORTED	MAJOMUN	MINIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM	
ACTORE 107 9 9800 980 900 107 1 207 4 400 400 500 300 300 Lic DOCK MORENER (TOLK) 1007 1 1007 1007 1007 307 4 871 501 100 307 4 871 501 100 307 4 871 501 100 307 4 871 501 100 307 4 871 501 100 302 100 <td< td=""><td>METHYLENE CHLORIDE</td><td>0 OF 2</td><td></td><td></td><td></td><td></td><td>0 OF 1</td><td></td><td></td><td></td><td></td><td>0 OF 4</td><td></td><td></td><td></td><td></td></td<>	METHYLENE CHLORIDE	0 OF 2					0 OF 1					0 OF 4					
12DOKADOR INC. (100) 9 <td>ACETONE</td> <td>107 2</td> <td>2000 00</td> <td>1362.0</td> <td>2000 0</td> <td>2000 0</td> <td>0 OF 1</td> <td></td> <td></td> <td></td> <td></td> <td>2 OF 4</td> <td>465 00</td> <td>401 25</td> <td>570.0</td> <td>360.0</td>	ACETONE	107 2	2000 00	1362.0	2000 0	2000 0	0 OF 1					2 OF 4	465 00	401 25	570.0	360.0	
METHYLETYM ETYME PECK 100 9 9400 940	1,2-DICHLOROETHENE (TOTAL)	007 2					0 OF 1					0 OF 4					
TRICH CONCENTR-KE O/F	METHYLETHYL KETONE	107 2	\$4.00	59.5	54 0	54 0	0 OF 1					3 OF 4	82 17	59 13	110 0	30.5	
BENZEN 0.07 1 0.07 1 0.07 1 TOULEN 0.07 1 0.07	TRICHLOROETHENE	0 07 2					0 OF 1					0 OF 4					
DETABLE/LONDETIENE O OF I O OF <	BENZENE	0.07 2				1	0 OF 1				[0 OF 4					
COLURNE O OF I O OF I O OF I CALCHORGENZENE O OF I I O OF I	TETRACHLOROETHENE	0.07 2					0 OF 1				i	0 OF 4					
CHC/0706HX1RE 0 OF 1 0 OF 4 1.1 - DICH.0706HX1RE 0 OF 0 OF<	TOLUENE	0 OF 2					0 OF 1				1	0 OF 4					
1.3DICH CROPENTATE 0 OF 1 0 OF 0 OF 0 0 OF 0 0 OF 0	CHLOROBENZENE	0 07 2					0.0F 1					0 OF 4					
1.4DICHORORENT NE G-MANDROBENT NE G-MANDROBEN	1,3-DICHLOROBENZENE	0 07 2					0 OF 1					0 OF 4					
1,2-00:CLGROBENZENE 0 07 1 0 07 <td< td=""><td>1,4-DICHLOROBENZENE</td><td>0 07 2</td><td></td><td></td><td></td><td></td><td>0 OF 1</td><td></td><td></td><td></td><td></td><td>0 OF 4</td><td></td><td></td><td></td><td></td></td<>	1,4-DICHLOROBENZENE	0 07 2					0 OF 1					0 OF 4					
jachware	1.2-DICHLOROBENZENE	0.0# 2					D OF 1				1	0 OF 4					
NITFORENTE/E O OF I I O OF I I O OF I I I I I I I I I I I I I I I I I I	CI-AND/OR 4-IMETHYLPHENOL	0 (7 2					10F 1	130 00	130.0	130.0	130.0	0 OF 4					
DEPLOCID 207 2 1300 <th< td=""><td>NITROBENZENE</td><td>007 2</td><td></td><td></td><td></td><td></td><td>0 OF 1</td><td></td><td></td><td></td><td></td><td>0 OF 4</td><td></td><td></td><td></td><td></td></th<>	NITROBENZENE	007 2					0 OF 1					0 OF 4					
T24-TRCH (DAORE KENE) O GF O GF <tho <="" c<="" td=""><td>RENZOIC ACID</td><td>107 1</td><td>130 00</td><td>130.0</td><td>140.0</td><td>120.0</td><td>DOF 1</td><td></td><td></td><td></td><td></td><td>3 OF 4</td><td>395 00</td><td>6671.25</td><td>630.0</td><td>205.0</td></tho>	RENZOIC ACID	107 1	130 00	130.0	140.0	120.0	DOF 1					3 OF 4	395 00	6671.25	630.0	205.0	
NIPPERTMALENCE OCF I OCF I OCF I ACELMANTHRALENCE OCF I	1.2 4-TRICHLOROEENZENE	007 2					0 OF 1					0 OF 4					
2-NETHYLINEHTHYLENE ACEMANHTHYLENE ACEMANHTHYLENE ACEMANHTHYLENE ACEMANHTHYLENE OF ACEMANHTHYLENE OF ACEMANHTHYLENE OF DEFNOR PHTHALATE OF HUDGENA DEFNOR PHTHYLANE OF HUDGENA DEFNOR PHTHYLATE OF HUDGENA DEFNOR PHTHYLENE OF HUDGENA DEFNOR PHTHYLENE OF HUDGENA DEFNOR PHTHYLENE OF HUDGENA DEFNOR PHTHYLENE OF HUDGENA DEFNOR PHTHYLENE OF HUDGENA DEFNOR PHTHYLENE OF HUDGENA DEFNOR PHTHYLENE DEFNOR PHTHYLENE DEF	NAPHTHALENE	007.2				(DOF 1					0 OF 4					
ACEUNPHTMELNE OFF <	2_METHMINAPHTHALEME						0.04					0.06 4					
ACE NAPHITHENE OCF OCF <thocf< th=""> <</thocf<>	ACENAPHTHYLENE					j	105 1	180.00	180.0	160.0	160.0	0.04					
Digenzoprenan opf opf <thopf< th=""> <t< td=""><td>ACENAPHTHENE</td><td>0.07 2</td><td></td><td></td><td></td><td></td><td>0.015 1</td><td></td><td>100.0</td><td></td><td></td><td>105 4</td><td>74.00</td><td>2168.00</td><td>74.0</td><td>74.0</td></t<></thopf<>	ACENAPHTHENE	0.07 2					0.015 1		100.0			105 4	74.00	2168.00	74.0	74.0	
District Purpulation Corr Corr<							0.01					0.06 4		1.00.00			
Display Display <t< td=""><td>DETHYL PHTHALATE</td><td></td><td></td><td></td><td></td><td></td><td>0.000 1</td><td></td><td></td><td></td><td></td><td>0.06 4</td><td></td><td></td><td></td><td></td></t<>	DETHYL PHTHALATE						0.000 1					0.06 4					
Child Heiler Hild Yawme O CF I CF <thi cf<="" th=""> <thi cf<="" th=""> I CF <t< td=""><td>ELLODENE</td><td></td><td></td><td></td><td></td><td>1</td><td>0.05 1</td><td></td><td></td><td></td><td>[</td><td>105 4</td><td>180.00</td><td>2178 75</td><td>160.0</td><td>180.0</td></t<></thi></thi>	ELLODENE					1	0.05 1				[105 4	180.00	2178 75	160.0	180.0	
Name O C <tho c<="" th=""> <tho c<="" th=""></tho></tho>							0.05					0.06 4	100 00	21/0/5	100 0	100.0	
Pre-monumentance 100 1000	DUE MANTLEDE ME		134.04		170.0	170.0	105 1	540.00	540 O	540.0	540.0	405 4	1171 25	1171 25	1000.0		
Antimutative Ito if			1/0 00	1110	100	140.0	105 1	150.00	3400	150.0	150.0	205 4	205.00	2077.50	240.0	170.0	
Dirent Builter middle Cord				237.4			0.05	130.00	1300	1300	,300	0.05 4	200 00	2011 34	2400	170.0	
PLOCEMENTIFIENE 2 OF 2 OF 2 PR 00 2 PR 0 2 PR 0<						-			1100.0	1100.0		405 4	1075.00	1076.00	1000 0	1100.0	
Precedence Procession Procession <td>I CUMANTINE NE</td> <td></td> <td>240 00</td> <td>740.0</td> <td>2/00</td> <td>2000</td> <td>105 1</td> <td>1100 00</td> <td>1100 0</td> <td>1100 0</td> <td>1100 0</td> <td>405 4</td> <td>2075.00</td> <td>1873.00</td> <td>2000 0</td> <td>1600.0</td>	I CUMANTINE NE		240 00	740.0	2/00	2000	105 1	1100 00	1100 0	1100 0	1100 0	405 4	2075.00	1873.00	2000 0	1600.0	
BENZCI BUIL PHIMALATE OGF IOF IOF <thiof< th=""> IOF <thiof< th=""></thiof<></thiof<>			230 00	0 553	1000	820 0	107 1	1100 00	1100 0	620.0	630.0	205 4	2075.00	2073 00	2900 0	1300.0	
BENZO (QIANTIFICACE NE 1 CP 7 110 00 272 0 110 0 100 0 100 0 470	BENETL BUITL PHIMALATE						107 1	830.00	6300	430.0	0300	204 4	173 00	2002 30	2200	130.0	
CHATYSENE 2 05 7 205 8 205 8 205 8 205 8 205 8 205 8 205 8 205 0 100 0 100 0 100 0 100 0 100 0 306 4 1306 67 2130 00 1500 0 760 0 BIS/2-EIHYUHEXYLIPHITHALATE 6 0F 8 10F 4 210 0 2133 75 220 0 2133 75 220 0 2133 75 220 0 2133 75 220 0 2133 75 220 0 2133 75 220 0 2133 75 220 0 2133 75 220 0 2133 75 220 0 2133 75 220 0 2133 75 220 0 2133 75 220 0 2133 75 220 0 210 0 700 0 700 0 1400 0 1400 0 1400 0 1400 0 1400 0 1400 0 100 0 100 0 100 0 100 0 100 0 100 0 100 0 1400 0 1400 0 1400 0 100 0	HENZOWANTHHACE NE		110 00	777 0	1100	110.0	104 1	470.00	4/00	470.0	470 0	404 4	1017.30	1017.50	1500.0	510.0	
HS12(2-E1HYLHEXT(1)PHTHALATE 0 GF 2 1 GF 4 330 0 330 0 330 0 300 0 200 0	CHRYSENE	10 1	105 00	202.0	220 0	190 0	104 1	700 00	700.0	700 0	/00 0	304 4	1308.67	2730.00	1500.0	920.0	
DI-N-DC1YD/HTHALARE 0 OF 1 OF 4 20 00 213375 220 00 213375 220 00 213375 220 00 200 0	BIS 2-ETHTLEEXTLI PHIHALATE						107 1	330.00	330 0	J30 0	330.0	J UF 4	1410 07	2812 30	1900 0	780.0	
BENZO (R) FLUCHANTHENE Y GK Y 75 00 Y	DI-N-OCTYLPHTHALATE	000					0.04					10 4	220 00	213375	2200	220.0	
BER X20 (K) FLUCHANITENE TOP 2 TB0 00 TB0 0 TB0 00 TB0 0 TB0 0 <th< td=""><td>BENZO (B) FLUCHANTHENE</td><td>70 7</td><td>7/5 00</td><td>8/50</td><td>400.0</td><td>190 01</td><td></td><td>1400.00</td><td>1400 0</td><td>1400 0</td><td>1400.0</td><td>404 4</td><td>1617 30</td><td>1626 25</td><td>3400 0</td><td>700 0</td></th<>	BENZO (B) FLUCHANTHENE	70 7	7/5 00	8/50	400.0	190 01		1400.00	1400 0	1400 0	1400.0	404 4	1617 30	1626 25	3400 0	700 0	
BER X2 (- A - P/Tig ME 2 CF 2 M 0 00 140 00 150 0 160 0 20 0 320	BENZO (RI FLUCHANIHENE	107 2	00 001	242.0	190 0	190.0	0.04					304 4	863 33	2412 30	980 0	7700	
INDENO (1,2,3-CD) PYRENE I OF 2 IN 60 2376 II 0 II 0 <thii 0<="" th=""> II 0 II 0</thii>	BENZO-A-PYHENE	204 2	140.00	140 0	150.0	130.0	104 1	320.00	320.0	320 0	320 0	304 4	766 33	234123	1100 0	2100	
DIBENZO (A,H)ANTHRACENE 0 OF 0	INDENO (1,2,3-CD) PYRENE	102.5	BH 00	757 0	94 0	84 0	104 1	240 00	240 0	240 0	240 0	JOF 4	531.67	2148 /5	/40 0	200.0	
BENZO (3H)PERYLENE 1 0F 7 62 00 251 0 62 0 1 0F 1 270 00 270 0 270 0 3 0F 4 578 33 2183 75 820 0 240 0 4,4'-DDE (P,P'-DDE) 6 0F 7 0 0F 1 0 0F 1 0 0F 4 0 0F 4 0 0F 4 4,4'-DDT (P,P'-DDD) 0 0F 8 0 0F 1 0 0F 4 0 0F 4 0 0F 4 6,4'-DDT (P,P'-DDD) 0 0F 8 0 0F 1 0 0F 4 0 0F 4 6,4'-DDT (P,P'-DDD) 0 0F 8 0 0F 1 0 0F 4 0 0F 4 6,4'-DDT (P,P'-DDT) 0 0F 8 0 0F 1 0 0F 4 0 0F 4 6,4'-DDT (P,P'-DDT) 0 0F 8 0 0F 1 0 0F 4 0 0F 4 90,0F 12 0 0F 1 0 0F 4 0 0F 4 0 0F 4 0 0F 1 90,0F 12 0 0F 1 0 0F 1 0 0F 4 0 0F 1	DIBENZO (A, H) ANTHRACENE	0.02 5					0 OF 1					UOF 4					
4,4'-DDE (P,P'-DDE) 0 OF 1 0 OF 4 4,4'-DDE (P,P'-DDE) 0 OF 1 1 OF 4 4,4'-DDT (P,P'-DD1) 0 OF 2 0 OF 1 4,4'-DDT (P,P'-DD1) 0 OF 2 0 OF 1 0 OF 1 0 OF 4	BENZOIGHIPERYLENE	107.2	62 0 0	251.0	82.0	62 0	10F 1	270 00	270 0	270 0	270 0	3 OF 4	578 33	2183 75	820 0	240.0	
4,4'-DDD (P,P'-DDD) 0 CF 2 0 CF 1 1 CF 4 700 00 399 36 700 0 700 0 4,4'-DDT (P,P'-DDT) 0 CF 2 0 OF 1 0 OF 4 GAMMA-CHLORDANE /2 0 CF 2 0 OF 1 0 OF 4 PCB - 1254 (AROCLOR 1254) 0 CF 2 0 OF 1 0 OF 4 WONOWE HYLWE FICHAY 1 OF 7 76 3 76 3 76 3	4,4'-DDE (P.P'-DDE)	0 02 8					0 OF 1					0 OF 4					
4,4'-DDT (P,P'-DDT) 0 OF 0 OF 0 OF 0 OF 0 OF 4 GAMMA-CHLORDANE /2 0 OF 0 OF 0 OF 0 OF 0 OF 4 PCB-1254 (AROCLOR 1254) 0 OF 0 OF 0 OF 0 OF 0 OF 4 WONOME THY WERCURY 1 OF 1 OF 0 OF 0 OF 1 0 OF 1	(4,4°~DDD (₽.₽°~DDD)	0 04 8					0.07 1					10F 4	700 00	399-30	700 0	700 0	
GAMMA-CHLORDANE 72 0 OF 0	(4,4'-ODT (P.P'-DOT)	00.1				1	0 OF 1					UOF 4					
PCB - 1254 (AROCLOR 1254) 0 UF 2 WONOWE THYLINE RCLIRY 1 OF 7 76 30 76 3 76 3 0 0 F 1 0 UF 10 0 UF	GAMMA-CHLORDANE 12	0 04 2					0 OF 1					0 OF 4					
MONONE[HY1ME[HY1ME]] 1 17 7 7930 767 793 793 0 17 2 0 0 13	PCB-1254 (AROCLOR 1254)	0.05	84.1-	h -		.	0.00			-		0 <u>OF4</u>					
	INONOMETHYLMERCURY		78 30	26.7	78 3	78 3	0(7 2					0.04 13					

(

FINAL

.

1

010

TABLE 4-1 SUMMARY OF SURFACE WATER AND SECTMENT ANALYTICAL RESULTS NYANZA III REMIEDIAL INVESTIGATION SUDBURY RIVER STUDY PAGE 9

				•		SEDIMEN	- ORGANICS					•••	• • • • •	·
					-		UNITS ug/kg							
ORGANIC		í	PEACH 7				PEACHB					REACH 9		
	FREQUENCY	AVERAGE DE FECTED	AVERAGE	MAXIMUM	MINIMUM	FREQUENCY DETECTED	AVERAGE	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE	AVERIAGE REPORTED	MAXIMUM	MINIMUM
METHYLENE CHLORIDE	0 07 3					" NO SAMPLES COLLE	CTED IN THIS PE	ACH **		** NO S/	MPLESCOLLE	CTED IN THIS	REACH **	
ACETONE	1 207 3	1260 00	8710	1000 0	880.0									
1,2-DICHLOROETHENE (TOTAL)	007 3													
METHYLETHYLKETONE	205.3	177.00	1230	330.0	24 0				1					
TRICHLOROETHENE	007 3													
BENZENE	007 3								4					
TETRACHLOROETHENE	005 3					1			1					
TOLUENE	0.0F 1								1					
CHLOROBENZENE	0.0F 3													
1,3-DICHLOROBENZENE	005 3								1					
1,4-DICHLOROBENZENE	0.0F 3													
1,2-DICHLOROBENZENE	0.05 3													
(3-AND/OR 4-) METHYLPHENOL	005 3													
NITROBENZENE	005 3								1					
BENZOIC ACID	306.3	573 33	5730	1300.0	130.0				1					
1,2,4 - TRICHLOROBENZENE	007 3													
NAPHTHALENE	005 3													
2-METHYLNAPHTHALENE	107.3	78 00	4780	78 0	78 0									•
ACENAPHTHYLENE	007 3													
ACENAPHTHENE	107 3	110 00	490.0	1100	1100									
DIBENZOFURAN	0073													
DIETHYL PHTHALATE	007 3								1					
FLUORENE	105.3	140 00	488.0	140.0	140 0				1					
N-NITROSODIPHENLYAMINE	005 3								1					
PHENANTHRENE	305.3	503.33	5030	1200 0	220.0							•		
ANTHRACENE	107.3	210.00	521 0	2100	2100				1					
DI-N-BUTYLPHTHALATE	007 3								1					
FLUORANTHENE	307.3	#53 33	9530	1000 0	350.0	(í					
PYRENE	307 3	876.87	\$760	1700 0	350.0									
BENZYL BUTYL PHTHALATE	107.3	810.00	055.0	610.0	610.0				1					
BENZO(A)ANTHRACENE	3 07 3	430.00	430.0	000	180.0									
CHRYSENE	1 305 3	840 00	640.0	1200 0	280.0]					
BIS(2-ETHYLHEXYL) PHTHALATE	1 3 0F 3	720 00	646.0	1100 0	340.0									
DI-N-OCTYLPHTHALATE	005.3													
BENZO (B) FLUORANTHENE	305.3	518 67	5160	8100	230.0									
BENZO (K) FLUORANTHENE	1 305 3	446.67	448.0	850.0	190.0									
BENZO-A-PYRENE	1 105 1	530 00	720.0	0.000	180.0				1					
INDENO (1,2,3-CD) PYRENE	105.1	540.00	6310	540.0	540.0									
DIBENZO(A,H)ANTHRACENE	0.0F 3								1					
BENZO(GHI)PERYLENE	105 1	610 00	455.0	6100	610.0									
4.4 -DDE (P.PDDE)	0.05 3								l					
4,4'-DOD (P,P'-DDD)	005 1					1								
4,4'-DDT (P,P'-DDT)	005 3								[•		
GAMMA-CHLORDANE /2	1 105 3	18 00	1358-0	18 0	18 0									
PCB-1254 (AROCLOR 1254)	105 1	\$10.00	3338.0	5100	510.0				l	11				
MONOMETHYLMERCURY	0 CF 3					** NO SAMPLES COLLE	cted in this re	ACH **		** NO SAMPL	ES COLLECTE) IN THIS FIE AC		

۱

,

W921 ΰ

SUMMARY OF SURFACE WATER AND SETIMENT ANALYTICAL RESULTS

NYANZA IN REMIEDIAL INVESTIGATION SUDBURY RIVER STUDY

4 Ξı PAGE 10

TABLE 4-1

4.4'-000 P P -000)

4.4'-DOT (P.P.-DOT)

DIMETHYLMERCURY

GAMMA~CHLORDANE /2

MONOMETHYLMERCURY

PCB-1254 (AROCLOR 1254)

- - ----SEDIMENT - ORGANICS UNITS. ug/kg EASTERN WETLANDS - PHASE I CHEMICAL BHOOK CULVERT ORGANIC **REACH 10** COMPOUNDS **AVERAGE** AVERAGE AVERAGE AVERAGE AVERAGE AVERAGE FREQUENCY DETECTED REPORTED FREQUENCY DETECTED REPORTED MAXIMUM MINIMUM FREQUENCY DETECTED REPORTED MAXIMUM MINIMUM MAXIMUM MINIMUM ** NO BANFLES COLLECTED IN THIS FEACH ** 0 OF 2 0 OF 2 METHYLENE CHLORIDE 0 OF ACETONE 1 OF 2 1600.00 900 1600 0 1600 0 2 2 OF 2 72 25 72 25 130.0 14 5 2 OF 2 25.00 25 0 430 1,2-DICHLOROETHENE (TOTAL) 1 OF 450 00 237.75 450 0 450 0 0 OF 2 2 METHYL ETHYL KETONE 170 00 170 0 1 OF 72 00 37.7 TRICHLOROETHENE 1 OF 2 91 375 1700 2 72.0 0 OF 1 OF 2 00 20 BENZENE 2 2 27 0 OF 2 1 OF 2 2 00 27 2.0 TETRACHLOROE THENE 0 OF TOLUENE 0 OF 2 2 CHLOROBENZENE 1 OF 2 1600.00 808.375 1600 0 1600.0 2 OF 2 52 00 52 0 77 0 1,3-DICHLOROBENZENE 0 OF 2 1 OF 2 100 00 535 0 100 0 2 OF 465 00 1,4-DICHLOROBENZENE 1 OF 2 1800.00 12125 1600 0 1800.0 -2 465 0 590 0 2 OF 1650 00 1.05 7200.00 4012 5 7200 0 7200 0 1650 0 1900.0 1,2-DICHLOROBENZENE 2 2 1 OF 260 00 1827 5 260 0 200 0 0 OF (3-AND/OR 4-)METHYLPHENOL 2 2 1 OF 310.00 372 0 310.0 1 OF 2 650 00 737 5 650 0 650.0 2 NITHOBENZENE 0 OF 0 OF 2 2 BENZOIC ACID 1 OF 2 3100 00 1982 5 3100.0 3100 0 2 OF 2 835 00 835.0 670 0 1,2,4 - TRICHLOROBENZENE NAPHTHALENE 2 OF 1 OF 2 2300 00 1562 5 2300 0 2300.0 2 585 00 585 0 600 0 2-METHYLNAPHTHALENE 0 ()F 2 1 OF 2 160 00 307.0 180 0 1 OF 140 00 287 0 140 0 ACENAPHTHYLENE D OF 2 2 n OF 0 OF 2 2 ACENAPHTHENE 1 OF 92 00 92 0 0.05 2 2610 DIBENZOFURAN 2 0 OF A OF 2 DETHYL PHTHALATE 2 1 OF 0 OF 2 84.00 259 0 64 0 FLUORENE 2 2 OF 230 00 230 0 0 OF 2 2 250 0 N-NTROSODIPHENLYAMINE 2 OF 2 760 00 760 0 790 0 PHENANTHRENE 1 OF 2 620 00 722 5 620 0 620 0 ANTHRACENE 0 OF 2 2 OF 2 215 00 2150 250 0 0 OF 0 OF 2 DI-N-BUTYLPHTHALATE 2 2 OF 1300.00 1300.0 1400.0 1 OF 2 1800.00 12125 1600 0 1600.0 2 FLUORANTHENE 2 OF 2 1200 00 1200 0 1200.0 1700 00 1282 5 1700.0 1700 0 PYPENE 1 OF 2 0 OF 2 0 OF 2 BENZYL BUTYL PHTHALATE 2 OF 2 710 00 7100 8100 A OF 2 BENZO (A) ANTHRACENE 1 OF 1200 00 1012 5 1200 0 1200 0 2 OF 2 745 00 745 0 780 0 CHRYSENE 2 0 OF 2 OF 2 150.00 150.0 200.0 BIS(2-ETHYLHEXYL) PHTHALATE 2 0 OF 0 OF 2 2 DI-N-OCTYLPHTHALATE 2 OF 2 1475 00 1475 2500 0 200 0 2 OF 2 620 00 820 0 640 0 BENZO (B) FLUORANTHENE 2 OF 2 680 00 680 0 760 0 0 OF 2 BENZO (K) FLUORANTHENE 2 OF 7700 1 OF 860 00 842 5 860 0 660 0 2 720 00 720 0 2 BENZO-A-PYRENE 1 OF 330 0 0 OF 5 330.00 382 0 2 INDENO (1,2,3-CD) PYRENE 0 OF 0 OF 2 2 DIBENZO (A, H) ANTHRACE NE 1 OF 270.00 352 0 270 0 2 BENZO (GHI)PERYLENE 0 OF - 2 0 OF 2 4,4'-DDE (P,P'-DDE) 0.05 2

1 OF 2

0 OF 2

0 OF

0 OF

20

0 OF 11

** NO BANPLES COLLECTED IN THIS FEACH **

2

2

- 11

117 00

179 00

1170

229 Ö

75.5

67 59

.....

1170

129 0

0 OF 2

0 OF 2

0 OF 2

0 OF

2

** NO SAMPLES COLLECTED IN THIS PEACH **

ч INAL

7.0

72.0

2.0

2.0

27.0

100.0

340.0

1400.0

310.0

600.0

530.0

180.0

140.0

92.0

84.0

210.0

730 0

180.0

1200.0

1200.0

810.0

730.0

100.0

800.0

580.0

870.0

330.0

270.0

4-12

TABLE 4-1 SUMMARY OF SURFACE WATER AND SEDMENT ANALYTICAL RESULTS NYANZA III REMEDIAL INVESTIGATION SUDBURY RIVER STUDY PAGE 11

		-					SEDIMEN										
L								JNITS ug/kg									
ORGANIC			DUTFALL OPEF	ĸ	I		1	ACEWAY			COLD SPRING BROOK						
	FREQUENC	AVERAGE DETECTED	AVERAGE	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM	AVERAGE FREQUENCY DETECTED	AVERAGE REPORTED MAX	MUM MINH				
METHYLENE CHLORIDE	107 3	31 75	22 17	55 0	. 5	1 OF 4	62 00	542 38	62	62	" NO SAMPLES COLLE	CTED IN THIS PEACH	••				
ACETONE	107 1	34.00	10 47	34 0	34.0	2 OF 4	485 00	767.50	760	210							
1,2-DICHLOROETHENE (TOTAL)	1 0F 1	1 00	2 75	10	10	3 OF 4	738 87	566 88	1300	110							
METHYL ETHYL KETONE	0 07 2					1 OF 4	44 00	542.25	44	44							
TRICHLOROETHENE	3 0F 3	2 43	2 63	40	0.9	4 OF 4	17071 25	17071 25	44000	5							
BENZENE	007 3					0 OF 4											
TETRACHLOROETHENE	107 3	0 80	2 72		0.8	0 OF 4				}							
TOLUENE	1 1 07 1	0 40	2 85	04	04	0 OF 4				- 1							
CHLOROBENZENE	0 07 3					4 OF 4	5917 50	5917 50	19500	30							
1,3-DICHLOROBENZENE	007 3					0 OF 4											
1,4-DICHLOROBENZENE	0 07 3					0 OF 4				[
1.2-DICHLOROBENZENE	207.3	247 50	685 00	200.0	205.0	20F 4	5200 00	4300.00	7100	3300							
S-AND/OR 4-)METHYLPHENOL	0 0F 3					0 OF 4											
NITROBENZENE	2 07 3	350.00	65 (1 5)	500 0	200.0	0 OF 4											
BENZOIC ACID	0 07 3					OOF O											
1.2.4-TRICHLOROBENZENE	307 3	880 00	800 83	1300.0	aso 0	0 OF 4											
NAPHTHALENE	307.3	218.33	218.33	340.0	150 0	2 OF 4	5250.00	4081.25	6700	3600							
2-AFTHYI NAPHTHALENE	0.07.3	•••••				0.04	5100 00	400125	0,00	3000							
	1 10	282.50	201 11	110.0	275.0	0.0E 4											
ACENAPHTHENE	0.05	14.55		2.00		0.04											
						0.05 4				Ì							
CHETLAN DUTHALATE	in		714.17			0.05 4				1							
						205 4	2500.00	2004 26	2800	2200							
	1	300.00		200.0	220.0	0.05	200000	2800 23	2000	2200							
DAIEANANTHERE ARE	1 100	100 00	em m		2000	407 4	10110.00	10110.00	10000								
ANTUDACENE		130.00	660.67	105.0		105 4	2012 32	2024.06	2100	0,000							
			300 07		• • •	305 4	\$633.33	3231 23	3100	2/00							
									00000								
PLUCHANINENE	1 10 1		1300 07	1000 0	1000	405 4	14202 00	14202 50	20000	10050							
PTHENE	101	1800.00	1000 00	2500.0	1100.0	404 4	8400.00	9400 00	13000	6800							
BENZYL BUITL PHIHALATE		1303-000	1500	130 0	130.0	004 4											
BENZO (AJANTHHACENE	305 3		800 00	1050 8	540 0	305 4	6933 33	5712 50	11000	4300							
CHRYSENE	30 3	1070 07	10/0 0/	1400 0	0000	304 4	8350 00	5975 00	8600	3850							
BIS 2-ETHYLHEXYL) PHTHALATE	30 3	110 00		2000 0	320 0	104 4	2900 00	2793 75	2500	2500							
DI-N-OCTYUPHTHALATE	60 1					0.04				Í							
BENZO (B) FLUORANTHENE	305.3	ענגזי	1233 33	1750 0	e70 0	JOF 4	4066 87	4262 50	5200	2500							
BENZO (R) FLUORANTHENE	105 3	1000.00	473 33	1000 0	1000 0	3 OF 4	4118 67	4300 00	6300	2550							
BENZO-A-PYHENE	1 107 1	800 00	800 00	1200 0	500 0	JOF 4	3586 67	3887 50	4700	2600							
INDENO (1,2,3-CD) PYRENE	1 502 1	577 50	453 33	635.0	520.0	0 OF 4				Į							
DIBENZO (A, H) ANTHRACE NE	0.07 1					0 OF 4											
BENZO (GHI)PERYLENE	1 104 1	835 00	491 87	730 0	540 0	0 CTF 4											
4,4'-DDE (P.P'-DDE)	0 04 3	ł				2 OF 4	18 00	10 50	212	14 8							
4.4 -DDD (P.P -DDD)	0 0 0 1	ŀ			1	2 OF 4	83 20	32 04	108	58 4							
4.4'-DDT (P.P'-DDT)	0 04 3	l.				0 OF 4				ļ							
GAMMA-CHLORDANE 12	1 0.07 3	I				0 OF 4				1							
PCB-1254 (AROCLOR 1254)	0 07 3	l.				2 OF 4	350 49	199.08	589	11198							
							~ * * * * * * * *		TLUC DE ACLAN		11 NO CAMPLES COULE	OTTO IN THE OT LONG					

W92194F

TABLE 4–1 SUMMARY OF SUFFACE WATER AND BEDINENT ANALYTICAL RESULTS NYANZA III REMEDIAL INVESTIGATION SUDBURY RIVER STUDY PAGE 12

OROANIC			Units uging			· · ·	· · · ·	aslern Wellande Dr	lling		
COMPOUNDS								0 - 2 1001 - 88 48			
	FREQUENCY DETECTED REPORTED MAXIMUM		AVERAGE	AVERAGE REPORTED	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE	MAXIMUM	MINIM
NETHYLENE CHLORIDE	** NO SAMPLES COLLECTED IN THIS NEACH **		** NO SAMPLES COLLE	CTED IN THIS RE	ACH **		2 OF 32	625 50	139 52	1600.00	51
CETONE							6 OF 32	34 17	229 22	110.00	7
2-DICHLOROETHENE (TOTAL)	1						10 OF 32	572 03	26213	5500 00	, o
ETHYL ETHYL KETONE						[0 OF 32				-
RICHLOROETHENE							16 OF 32	313 38	216 36	3400.00	0
ENZENE		1				1	1 OF 32	4 00	110 86	4 00	4
ETRACHLOROETHENE	1					1	0 OF 32				
OLUENE						1	3 OF 32	B 33	110 81	16 00	2
HLOROBENZENE							15 OF 32	5553 73	2604 60	34000 00	3
3-DICHLOROBENZENE						1	10 OF 32	90.40	273 75	290 00	18 (
4-DICHLOROBENZENE						1	13 OF 32	906 69	579.81	3100 00	48 (
2-DICHLOROBENZENE	1						15 OF 32	3616 73	1677 69	13000 00	33
3~AND/OR 4-)METHYLPHENOL						1	1 OF 32	74 00	347 16	74 00	74 (
ITROBENZENE						ł	8 OF 32	101 25	1311 25	140.00	65
ENZOIC ACID	1						13 OF 32	292.27	324 83	1800-00	21
24-TRICHLOROBENZENE	J						9 OF 32	159 33	303 56	320 00	42 (
IAPHTHALE NE						1	10F 32	62 00	352 25	82 00	82 (
-METHYLNAPHTHALENE							1 OF 32	60.00	351 56	60 00	60 (
CENAPHTHYLENE						1	3 OF 32	31 33	329 66	45 00	21 0
CENAPHTHENE	1						1 OF 32	43 00	351 03	43 00	43 (
IBENZOF URAN							0 OF 32				
ETHYL PHTHALATE							1 OF 32	60 00	351 56	60.00	60 0
LUORENE							5 OF 32	53 60	308.38	75 00	8.0
I~NITROBODIPHENLYAMINE							13 OF 32	42.77	223 94	120.00	10.0
HENANTHRENE	1						7 OF 32	76 14	310 22	. /0 00	23 (
NTHRACENE		1					3 OF 32	45 00	329 53	91.00	21.0
I-N-BUTYPHTHALATE							14 OF 32	70 57	235 33	270 00	15 0
LUORANTHENE							19 OF 32	58 89	169 50	200.00	10 0
YRENE							1 OF 32	25.00	360-31	25.00	25 (
IENZYL BUTYL PHTHALATE	[- (2 OF 32	83 00	348 69	140 0 0	26 (
ENZO(A)ANTHRACENE	1					1	2 OF 32	125 50	348 63	190.00	61 (
HRYSENE						[3 OF 32	75 33	339 22	100.00	62.0
IS(2-ETHYLHEXYL)PHTHALATE							2 OF 32	22 50	337 97	36 00	9.0
N-N-OCTYUPHTHALATE							4 OF 32	80 50	320 84	160 00	41.0
ENZO (B) FLUORANTHENE							0 OF 32	0.00	0 00	0 00	00
IENZO (IQ FLUORANTHENE		- 1					10 OF 32	260 00	334 22	860 00	41 0
ENZO-A-PYRENE							1 OF 32	120 00	353 44	120 00	120 0
IDENO (1.2.3-CO) PYRENE	1						0 OF 32	0,00	0.00	0.00	00
HEENZOVA, HIANTHRACENE	{					1	1 OF 32	120 00	353 44	120 00	120 0
ENZO(GHI)PERYLENE							0 OF 32				
4 - DDE (P.P DDE)						1	0 OF 32				
4'-D00 (* .P'-D00)						1	0 OF 32				
4'-DDT (P.P'-DDT)							0 OF 32				
AMMA-CHLORDANE /2	1						0 OF 32				
CB - 1254 (AROCLOR 1254)	· · · · · · · · · · · · · · · · · · ·				15 11.12	ļ	0 OF 32				
	101		** NO BAMPLES COLLE	CIED IN THIS RE	ACH **		0 OF 0				•
TAVIDENTENE					·- · -	ł	0 +Du ini 19	ā īo	110.74	. 0.70	
VIENES (COTAL)							307 32	2 #7	110/8	5.00	0 /
HENOL						1	2 OF 32	62.00	340.75	86 00	54 0
- CHLOROPHE NOL	1						1 OF 32	20 00	345 18	20 00	20 0
ANY: CHLOBIDE											

i

FINAL

-13

1

W92194F

- TABLE 4-1 SUMMARY OF SUPFACE WATER AND SEDMENT ANALYTICAL RESULTS NYANZA III REMEDIAL INVESTIGATION SUDBURY RIVER STUDY PAGE 13

{

		-					SUAFACE WATE	A - TOTAL IN	ORGANICS				· · ·		
	t	- BACHD	NOUNO -			- 1		and age		1					
	REACH 1	BUDBURY RES	ERVOIR RESE	AVOIR NO 3				REACH 2			REACH 3				
COMPOUNDS	FREQUENCY	AVENAGE DE TECTED	AVERAOE REPORTED	MAXIMUM	MNNUM	FREQUENCY	AVE PAGE DE TECTED	AVE RAGE REPORTED	MAXIMUM	MNIMUM	FREQUENCY	AVE RAGE DE TECTED	AVERAGE	MAXIMUM	MNIMUN
ALUMNUM	1 2067	220 00	64 00	231 00	208 00	SOF 7	1178 50	525 30	3220 00	100 00	5 OF 5	128 70	126 70	168 00	92 50
ANTIMONY	0 OF 7					0 OF 7					0 OF 5				
ARGENIC	0.0F7					10F7	3 00	4 71	3 00	3 00	0 OF 5				
BARIUM	# OF 7	18 15	15.70	23 10	8 80	7047	33 81	33 81	132 00	12 60	2 OF 5	12 55	5 62	12 60	12 50
BERYLLIUM	0.047					1077	7 50	2 61	7 50	7 50	0 OF 5				
CADMIUM	0.047					1077	7 20	2 64	7 20	7 20	0 OF 5				
CALCIUM	1 10/1	10217 14	10217 14	12400 00	6200 00	7077	9317 86	9317 86	13000 00	6140 00	5 OF 5	7052 00	7052 00	7260 00	6790.00
CHROMIUM	0077					1 OF 7	7 00	5 31	7 00	7 00	0 OF 5				
COBALT	0 0 7					1 OF 7	27 70	25 39	27 70	27 70	0 OF 5				
COPPER	10/7	3 10	3 50	3 10	3 10	1077	27 70	13 52	27 70	27 70	0 OF 5				
IRON	50/7	867.29	785 50	1840.00	750 00	\$ OF 7	3943 80	2943 82	16100 00	458.00	5 OF 5	551 60	551 00	673 00	425 00
LEAD	10/7	21.10	3 90	21 10	21 10	2077	33 80	11.36	66 50	1 30	2 OF 5	1 00	1 00	1 00	1 00
MAGNEBIUM	1 1011	2450 08	2490 00	2900 00	1818 00	7 OF 7	2233 57	2233 57	3030 00	1570 00	5 OF 5	1730 00	1730 00	1780 00	1650 00
MANGANESE	0 OF 7	81 83	80.78	111 00	75 80	7047	1533 70	1533 70	9840.00	66 20	5 OF 5	63 38	63 38	83 60	54.60
MERCURY	0 OF 6					0 OF 7					0 OF 5				
NICKEL	1067	17.70	5 10	17 70	17 70	1077	77 40	22 01	77 40	77 40	0 OF 5				
POTABBIUM	10/1	2481 43	2481 43	3100.00	1610 00	7 OF 7	1971 43	1971 43	2560 00	1540 00	5 OF 5	1952 00	1952 00	2260 00	1730 00
SELENIUM	1 9067					00#7					1 OF 5	19300.00	3801 53	19300.00	19300.00
BILVER	1077	18 99	4 74	18 90	16 80	1077	68 80	9 58	68 90	68 90	3 OF 5	9 90	6 94	11 90	6.40
BODIUM	1011	20071-43	20071 43	38200 00	18000 00	7077	28757 14	28757 14	33800 00	19600 00	4 OF 5	19950 00	19820 00	20600 00	19700 00
THALLIUM	0.047					0 OF 7					0 OF 5				
VANADIUM	0017					2077	14 35	22 59	16 10	12 60	0 OF 5				
ZINC	1077	6 70	6 02	e 20	6 20	1077	125 00	24 81	125 00	125 00	1 OF 5	8 00	8.74	8 00	8 00

								A - FILTERED	INORGANICS						
		NEACH 1 BUDS		A			ı	REACH 2		REACH 3					
	FREQUENCY	AVE AADE DE TECTED	AVE ALLOE REPORTED	MAXIMUM	MNWVM	FREQUENCY	AVERAGE DETECTED	AVERAGE	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE DE TECTED	AVERAGE	MAXIMUM	MNIMUM
ALUMINUM	0 OF 6					2 OF 7	\$4.13	20 20	67 30	40 95	1 OF 1	60 60	60 60	60 60	80 80
ANTIMONY	0.04.8					0 OF 7					0 OF 1				
ARBENIC	0.00					1047	1 20	4 /3	1 20	1 20	0.01	12.00			13.00
BARIUM	1011	17 12	17.142	19 70	14 80	1017	et 17	21.17	33.00	12 20	1011	12 60	12 00	12 00	12 00
BEHTLLRJM	0.071					4047					0.041				
CADMIUM			11414 00	12400.00		1017	8081.43	0081.43	10700.00	8240.00	1051	7120.00	7120.00	7120.00	7120.00
CHROMINIA						6 OF 7	•••••	000.45	10,00 00	0140 00	O OF 1	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,			
COBALT	0.051				1	0.077					0 OF 1				
COPPER	0071				1	9 OF 7					O OF 1				1
IRON	4075	450 50	410 00	502 00	352 00	4 OF 7	508 38	387.71	623 00	281 00	1 OF 1	310 00	310 00	310 00	310 00
LEAD	0 07 5				-	0 OF 7					0 OF 1				
MAGNESIUM	S OF S	2770 00	2778 00	2000 00	2280 00	7 OF 7	2135 71	2135 71	2480 00	1530 00	1 OF 1	1700 00	1700 00	1700 00	1700 00
MANGANESE	4 0/ 1	56 95	15 00	87.00	24 80	6 OF 7	109 60	98 99	157 00	54 60	1011	48 60	48 60	48 60	48 60)
MERCURY	0 OF 5					0 OF 7					0 OF 1				
NICKEL	1 OF 8	7 00	3 90	7 00	7 00	0 OF 7					0 OF 1				
POTABSIUM	\$ OF \$	7646 00	2646.00	3010.00	2470.00	7 OF 7	1820 71	1920 71	2340 00	1530 00	1 OF 1	1720 00	1720 00	1720 00	1720 00
BELENIUM	0 OF 5					0 OF 7					0 0F 1				
BILVER	0 OF 5					0.047					0.0+1	10700 00		10700.00	10700.001
SODIUM	s of s	24148 08	34140.00	38000 00	20000.00	7077	54045.66	29092 66	34300.00	19100.00		19700-00	19700.00	19/00 00	19/00 00
THALLIUM	0 OF 7					0.047									
VANADIUM	0.07 6					0077						18 60	18 50	18.50	18 60
ZING	1 3041	18 43	18.11	\$1 eo	10.00	1 9097				-		10 50	- 10.30	10.50	10.201

ĺ

(

.
TABLE 4–1 SUMMARY OF SUFFACE WATER AND SEDMENT ANALYTICAL RESULTS NYANZA III REMEDIAL INVESTIGATION BUDBURY RIVER STUDY PAGE 14

·	t					· · · · · · ·		H - TOTAL ING JNITS ug/	DRGANICS		·				
INORGANIC COMPOUNDS		(REACH 4				I	NEACH 5				'	REACH 6		
	IREQUENCY	AVERAGE DE TECTED	AVERAGE REPORTED			FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MNIMUN
ALUMINUM	2072	714 88	714 00	1348.00		0 OF 2					3 OF 4	216 53	171 06	461 00	82 0
ANTIMONY	0 OF 2					0 OF 2					0 OF 4				
ARSENIC	0 OF 2					1 OF 2	1 10	1 05	1 10	1 10	0 OF 4				
BARIUM	1078	24.78	12.85	24 70	24 70	1 OF 2	12 75	16 30	12 75	12 75	1 OF 4	16 20	4 60	16 20	16 21
BERYLLIUM	0 0F 2					0 OF 2					0 OF 4				
CADMIUM	0 OF 2					0 OF 2					0 OF 4				
CALCIUM	5015	7300 00	7388 90	7868 00	8800 00	\$ OF \$	9722 50	Ø722 50	11600.00	7845 00	4 OF 4	9028 75	9028 75	9660 00	6785 00
CHROMIUM	1072		4 10	9 00	ေဆု	0 OF 2					0 OF 4				
COBALT	0.04 2					0 04 2				Í	0 OF 4				
COPPEN	0043					0012					0 OF 4				
	1011			2100 00	490.00	0.04.2					3014	469.33	463 75	507 00	446 00
				12 00							10F4	2 40	1 35	2 40	2 40
MALINEOTVAL	2012				1700 00	1011	2212 50	2212 30	2840.00	1/85 00	4 0F 4	2052 50	2052 50	2110 00	2000.00
MANUARCOL	1011	126 03	1/4 43	200.00	••• /•	1072	100.00	00 43	108.00	108 00	4074	98 35	96 35	167.00	70 10
NERCONT					1	0.041					0013				
						0.042		1810.00			4054				
ART CAN IN		1203 00	1205 00	1,00,00		2072	1870 00	1870 00	2100.00	100 001	4074	2156 25	2156 25	2425 00	1640.00
	0.053					A (14 2					0.064				
SODIUM	1 1011	10000.00	10000-00	21000.00	10000.00	2012	26275.00	28275.00	31100.00	21450 00	4 0 5 4	22500.00	22500.00	25 300 00	21000.00
THALLIUM						0.07.2	141.000		21.00 00		0 OF 4	22300 00	22,500,00	23300 00	21000 00
VANADIUM	0072					0 04 2					0 DE 4				
ZINC	0072					6 OF 2				1	O OF A				

		· ·	· 				WAFACE WATE	A - FILTERED INITS 491	INORGANICS						·
INORGANIC		•	NEACH 4				,	NEACH 5		1		,	EACH 0		ļ
	FREQUENCY	AVERAGE DE TECTED	AVERAGE REPORTED	MAJOMUM		FREQUENCY	AVERAGE DE TECTED	AVERAGE	MAXIMUM	MNIMUM	FREQUENCY	AVERAGE DE TECTED	AVE RAGE REPORTED	MAXIMUM	MNIMUM
ALUMINUM	1071	107 00	107 80	107 00	107 00	OF 1]	3 OF 3	110 57	110 57	174.00	78 30
ANTIMONY	1 00F1					0 OF 1					0 OF 3				
ARSENIC	10F1		88 80	80 80	96 90	0 OF 1				[O OF 3				}
BARIUM ·	0 0F 1				í	0 OF 1					0 OF 3				- 1
BERYLLIUM	0.0F1				1	0 OF 1					0 OF 3				
CADMIUM	0 0F1					8 OF 1					0 OF 3				
CALCIUM	1071	7700 00	7768-88	7700 00	7700 08	1 OF 1	11900 00	11800.00	11900 00	11900 00	3 OF 3	9003 33	9003 33	9380.00	8770 00
CHROMIUM	0 OF 1					0 OF 1				f	0 OF 3)
COBALT	0.0#1					0 OF 1					0 OF 3				1
COPPER	0.0#1					0 OF 1					0 OF 3				
IRON	10F1	254 85	366 80	394 00	396 00	8 OF 1					3 OF 3	429 00	429.00	525 00	330 00
LEAD	1011	1 10	1.50	1.80	1 80	OF 1					0 OF 3				1
MAGNEBIUM	1071	1740.06	1740 00	1740 00	1740 00	1 OF 1	2000 00	2660.00	2680 00	2680 00	3 OF 3	2056 67	2056 67	2120 00	2020 00
MANGANESE	10F1	41 30	41 30	41 30	41 30	0 OF 1					3 OF 3	68 17	68 17	75 80	58 60
MERCURY	0.071					0 OF 1				1	0 OF 3				1
NICKEL	0.0F1				1	0 OF 1					0 OF 3				
POTASSIUM	10F1	1318 88	1310 00	1310 00	1310 00	1 OF 1	2190 00	2190.00	2190.00	2190.00	3 OF 3	2313 33	2313 33	2550 00	2120 00
BELENIUM	0 OF 1					6 OF 1					0 OF 3				1
SILVER	8 OF 1					0 OF 1					0 OF 3				
SODIUM	10F1	20000 00	20808-80	20000 00	20000 00	1 OF 1	31300 00	31300.00	31300.00	31300.00	3 OF 3	15500 00	15500.00	23000 00	2100 00
THALLIUM	0.0F1					0 OF 1					0 OF 3				
VANADIUM	0.001				i i i	0 OF 1					0 OF 3				[
ZINC	10/1	22.00	27 80	22 80	22 80	0 OF 1				1	Ø OF 3				1

FINAL

· ·· .

- W92194F

TABLE 4-1 SUMMARY OF SUFFACE WATER AND BEDINE INT ANALYTICAL RESULTS NYANZA III REMEDIAL INVESTIGATION SUDBURY RIVER STUDY PAGE 15

1

INORGANIC	1	,	NEACH 7				1	REACH 8				,	R HCAE		
COMPOUNDS	1														
		AVERAGE	AVERAOL				AVERAGE	AVERAGE			f TOWENCH	AVERAGE	AVERAGE		
	THEOREM				هم شرمه	THEONERU	DETECTED	HEFOHIED	MACINICIA	MINIMUM	THEQUENCY	DETECTED	HEPOHIED	MAXIMUM	14 19 19
ALUMINUM	1074	101 00	37 16	103 00	103 00	** NO 5A	MPLES COLLEC	TED IN THIS PL	EACH **		** NO SAI	APLES COLLEC	TED IN THIS A	EACH **	
ANTIMONY	0.0F4														
RSENIC	0.0F4				1										
ARIUM	1074	21.30	7 30	21 30	21.30										
ERYLLIUM	0.0F4														
CADMIUM	0.074														
ALCIUM	4 07 4	11345.00	11365 00	13100 00	00 0008										
HROMIUM	0.074														
COBALT	0.074														
COPPER	0.074														
AON	3074	526 67	458 75	\$70.00	311 00										
LEAD	0 07 4														
MAGNEBIUM	4074	2540 00	2100 00	2910 00	2150 00										
ANGANESE	3074	#5 80	74 10	119 00	46 60										
MERCURY	8 07 4														
NCKEL	0 0# 4														
OTASSIUM	4 07 4	2167 50	2107 50	2380.00	1870 00										
ELENIUM	0074														
ILVER	1074	4 50	3 50	6 50	\$ 50										
IODIUM	4 0 7 4	24050 00	24850 00	27900 00	20800 00										
HALLIUM	0074														
ANADIUM	0.074)					
INC	0.044														

	· • -					ء ـ ـ ـ ـ ـ	URFACE WATE	NITS UDV			·				
		•	NE ACH 7					NEACH B				'	REACH D		
	FREQUENCY	AVERAGE DE TECTED	AVERADE	NAJONUN	MNIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MNIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MNIMU
ALUMINUM ANTIMONY ARBENIC	2 OF 4 0 OF 4 0 OF 4	131 10	72 34	199 00	106 00	** NO 8A	MPLES COLLEG	:1EO IN THIS P	REACH **		** NO 8A	MPLES COLLEC	TED IN THIS A	EACH	
BARIUM BERYLLIUM CADMIUM	1 OF 4 0 OF 4 0 OF 4	20 00	7 14	14 00	20 60										
CALCIUM CHROMIUM COBALT	4 OF 4 0 OF 4 0 OF 4	11412 98	1141 2 50	1 3400 00	861 0 00										
IRON LEAD	1 OF 4 0 OF 4	376 00	184 58	378 00	376 00										
MAGNESIUM MANGANESE MERCURY MICKEL	4 OF 4 3 OF 4 0 OF 4 0 OF 4	2917 90 07 40	2117 50 54 25	90 90 2000 00	2130 00 47 20										
POTASSIUM	4 OF 4 0 OF 4	1793 SO	1262 16	2540 00	1920 00										
SILVER SODIUM THALLIUM VANADIUM ZINC	1 OF 4 4 OF 4 0 OF 4 6 OF 4 0 OF 4	0 00 24625 00	9 00 24625 00	8 00 27800 00	00 206 05 00 206 05										

(

FINAL

Т

TABLE 4-1 SUMMARY OF SURFACE WATER AND SEDMENT ANALYTICAL RESULTS NYANZA III REMEDIAL INVESTIGATION SUDBURY RIVER STUDY PAGE 16

1

		-				·	SURFACE WATE	A - TOTAL INI INITS ug/	DROANICS						
	ļ		REACH 10					ASTERN WET	LANDS			CHEMICAL BRO	OK CULVERT		
	FREQUENCY	AVE MADE	AVERAGE REPORTED	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE	MAXIMUM	MNIMUM	FREQUENCY	AVERAGE DETECTED	AVE RAGE REPORTED	MAXIMUM	MINIMUM
ALUMNUM	** NO 84	WPLES COLLE	CTED IN THIS P	EACH **		10F2	758 00	421 20	756 00	756 00	• HO 5A	MPLES COLLEC	TED IN THIS R	EACH ··	
ANTIMONY						0 OF 2									
ARBENIC						0 OF 2									
BARIUM						\$ OF \$	8 5 3	8 50	B 4 0	7 85					
BERYLLIUM						0 OF 2				1					
CADMIUM						0 OF 2									1
CALCIUM						5 OF 5	14250 00	14250 00	14500 00	14000 00					
CHROMIUM						1042	/9 00	79.00	79.00	79 00					
COBALT						0072									
COPPER							14 60	1010 20	14 60	14 00				•	
IRON							1420 00	10/18/20	1420 00	1420 00					
LEAD					1		4 03	4 00	1245.00						
MALINEBIUM							100 73	100 70	1243 00	03.45					
MANGANESE						1012	2 06	2 10	3.00	0.37					
MERCONT						0.042			5.00	0.37					
ACTAGE INA	{					2062	974 50	974 50	1065.00	864 00					
SELENHIM						0 OF 2									
SH VER						0 OF 2									
ACOUNT						2072	11090.00	11090 00	18100 00	6080 00					
THALLIUM	1					0 OF 2									
VANADIUM	1					0.04.5				{					
ZINC	(0 OF 2									

		,	SUAPACE WATE	A - FILTERED INITS ug1	INORGANICS						
INORGANIC	REACH 10			ABTERN WET	ANDS		c c	HEMICAL BRO	OKCULVERT		
COMPOUNDS	AVERAGE AVERAGE MADMUM MINIMU	FREQUENCY	AVERAGE DE TECTED	AVERAGE	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MNIMUM
	** NO BAMPLES COLLECTED IN THIS REACH **	0 OF 2 0 OF 2 0 OF 2					** NO SA	MPLES COLLEC	TED IN THIS R	EACH **	
BARIUM BERYLLIVM		2 OF 2 0 OF 2 0 OF 2	6 60	6 60	6 10	5 50					
CALCIUM CHROMIUM COBALT		2 OF 2 8 OF 2 8 OF 2	12950 00	12950 00	14900 00	1 1 000 00					
COPPER IRON LEAD		0 OF 2 0 OF 2 1 OF 2	8 70	0.00	8 70	8 70					
MAGNEBIUM MANGANEBE MERCURY		1 OF 2 2 OF 2 9 OF 2	97 80 0 43	51 33 0 43	92 80 0 49	92 60 0 37					
POTASSIUM SELENIUM SILVER		2 OF 2 6 OF 2 0 OF 2	984 50	964 50	1140.00	829 00					
SODIUM THALLIUM VANADIUM ZINC		2 OF 2 0 OF 2 0 OF 2 0 OF 2	10290 00	10290.00	15900 00	4650 00			••• =		

FINAL

1

4-18

TABLE 4-1 SUMMARY OF SURFACE WATER AND SEDMENT ANALYTICAL REBULTS NYANZA III REMIEDIAL INVESTIGATION SUDBURY RIVER STUDY PAGE 17

[1	SUAFACE WATE	A - TOTAL ING UNITS UG/1	ORGANICS						
INORGANIC			DUTFALL CREE	ĸ				RACEWAY					COLD SPRING	BROOK	
	FREQUENCY	ÁVÉ MAQĚ DE FEC FED	AVERAGE	NAUMUM		FREQUENCY	AVERAGE DETECTED	AVERAGE	MAXIMUM	MNIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MNIMUM
ALUMNUM	1071	201.00	203 00	203 00	203 00	9 OF 1					1 OF 1	165.00	165 00	165 00	165 00
ANTIMONY	0.071					0 OF 1					0 OF 1				
ARBENIC	0.0/1					0 OF 1					0 OF 1				
BARIUM	10/1	10 50	10 90	10 50	10 50	1 OF 1	18.4	18.4	16.4	18.4	0 OF 1				
BERYLLIUM	0.0011					0 OF 1					0 OF 1				
CADMIUM	0 OF 1					0 OF 1				1	0 OF 1				
CALCIUM	1011	17700 00	17700 00	17700 00	17700 00	10F1	10025	10025	10025	10025	1 OF 1	9570 00	9570 00	9570 00	9570 00
CHROMIUM	10F1	5 60	5 60	5 60	5 60	8 OF 1					0 OF 1				
COBALT	10F1	2 30	2 30	2 30	2 30	0 OF 1					0 OF 1				
COPPER	6 OF 1				j	8 OF 1					0 OF 1				
IRON	10F1	1870.00	1879 00	1878 00	1879 00	1 OF 1	1070	1070	1070	1070	1 OF 1	668 00	688 00	668 00	688 00
LEAD	10/1	\$ 78	5.76	\$ 70	\$ 70	0 OF 1					0 OF 1				
MAGNEBIUM	1071	1830 00	1830 00	1830 00	1630 00	1071	2350	2350	2350	2350	1 OF 1	2490 00	2490.00	2490 00	2490 00
MANGANESE	1011	140.00	140.00	140.00	140 00	1011	263 5	263 5	283 5	283 5	1 OF 1	108 00	108 00	108.00	108 00
MERCURY	1001		0 40	0 48	0 48	0 OF 1					0 OF 1				
NICIDEL	6 OF 1					0 OF 1					0 OF 1				
POTABBIUM	1071	1790.00	1780.00	1790.00	1790 00	10F1	1875	1675	1675	1675	1 OF 1	2530 00	2530 00	2530 00	2530.00
BELENIUM	0 OF 1					0 OF 1					0 OF 1				
BILVER	0.071					00/1					1 OF 1	13 90	13 90	13 90	13 90
BODIUM	10F1	21400.00	21400 00	21400 00	21400 00	1 OF 1	33000	33000	33000	33000	1 OF 1	16500 00	16800 00	16800 00	16800 00
THALLIUM	80#1					0 OF 1					0 OF 1				
VANADIUM	0.0001					0 OF 1					0 OF 1				
ZINC	10/1	47 80	47 90	47 90	47 90	0 OF 1					O OF 1				

.

				- • • • • •	•		URFACE WATE	A - FILTERED	NORGANICS				·· ·		
INORGANIC			OUTFALL CREE		·	· · · · · · · · · · · · · · · · · · ·		ACEWAY					OLD SPRING	BROOK	
COMPOUNDS	FREQUENCY	AVERADE DE TECTED	AVERADE	MAXINUM	 MNIMUN	FREQUENCY	AVERAGE DETECTED	AVERAGE	MAXIMUM	MNIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM
ALUMNUM	1071	87 08	e7 co	67 00	e7 00	9 OF 1					** NO 8A	MPLES COLLEC	TED IN THIS R	EACH **	
ANTIMONY	0 0 F 1					9 OF 1									
ARBENIC	0 OF 1					0 OF 1									
BARIUM	10/1	8 40	9 40	8 40	8 40	1 OF 1	31.4	31.4	31.4	31.4					
JERYLLIUM	0.0F1					6 OF 1									
CADMIUM	0 OF 1					0 OF 1									
CALCIUM	10F1	17908 88	17400 00	17900 00	17900 00	1 OF 1	9580	9580	9580	9580					
CHROMIUM	0.0F1					9 OF 1									
COBALT	0.0/1					0 OF 1									
COPPER	0.041					0 OF 1									
IRON	10F1	666 66	808 00	00 00	990 00	1 OF 1	808	606	608	608					
LEAD	1 OF 1	2 40	2 40	2 40	2 40	0 OF 1									
MAGNEBIUM	1071	1200 00	1200.00	1280.00	1280.00	1 OF 1	2230	2230	2230	2230					
MANGANESE	10/1	173 00	123 00	123 00	123 00	1041	208 5	208 5	208 5	208 5					
MERCURY	10F1	0 42	• 42	0 42	0 42	0 OF 1									
NICKEL	0 OF 1					0 OF 1									
POTASSIUM	10/1	1780 80	1780 00	1788.00	1780.00	1 OF 1	1825	1825	1625	1825					
SELENIUM	0.0/1					0 OF 1									
BILVER	0 OF 1					0 OF 1									
BODIUM	1071	22000 00	\$3008 00	33000 00	\$\$000.00	1 OF 1	32500	32500	32500	32500					
THALLIUM	00/1					0 OF 1									
VANADIUM	0.0#1					0 OF 1									
ZINC	100	48.19	46.10	40 10	46 10	0 OF 1			_						



W92194F

4-19

TABLE 4-1 BUMMARTY OF BURFACE WATER AND BEDMENT ANALYTICAL REBULTB NYANZA HI REMEDIAL INVESTICIATION BUDBURY RIVER BITUDY PAGE 10

INORGANIC COMPOUNDS IVERACE AVERACE FREQUENCY DETECTED REPORTED MAXIMUM MINIMUM ALUMINUM ALUMINUM ALUMINUM ANTIMONY ARGENIC BARIUM BARIUM CADRUM COPPER IRON LEAD MAGNESEL PCTABSUM MAGNESEL PCTABSUM BLVERICM			WATER - YOTAL INORGANICS
AVERAGE AVERAG	INORGANIC	BOADEANO WETLANDS	HEARD POND
ALUMINUM ANTIMONY ANTIMONY ARBENC BANUM BERYLLIUM CADMIUM CADMIUM COBALT COPPER IRON LEAD IRON IRON IRON IRON IRON IRON IRON IRON		ÄVENÄDE ÄVERÄDE FREQUENCY DETECTED REPORTED MAJONUM MIM	AVE AAGE AVE AAGE AVE FREQUENCY DETECTED REPORTED MAXIMUM MINIMU
BODIUM VANADIUM VANADIUM	ALUMINUM ANTIMONY ARBENIC BARUM BERYLLIUM CADRIUM CALCIUM CHROMIUM COPPER IRON LEAD MAGNEBUM MAGNEBUM MAGNEBUM MAGNEBE POTASSIUM SILVER SCOULM THALLIUM YANADIUM	** NO SAMPLES COLLECTED IN THIS REACH **	** NO SAMPLES COLLECTED IN THIS REACH **

		ren - filtered inordanics Units upi
	BORDE RING WET LANDS	HEARD POND
	AVERAGE AVERAGE MATHAUM MINIMUM	ÁVERÁGE ÁVERÁGE FREGUENCY DETECTED REPORTED MAXIMUM MINIMU
ALUMINUM ANTIMONY ARBENIC BARIUM BERYLLIUM CADRIUM CADRIUM CORPER IRON LEAD MAGNEBUM MAGNEBUM MAGNEBUM MAGAESE MERCURY NICKEL POTABSIUM GELENIUM	** NO BAMPIFB COLIFCTED IN THIS REACH **	** NO BAMPLES COLLECTED IN THIS REACH **
SILVER BODIUM THALLIUM VANADIUM ZINC		

TABLE 4-1 SUMMARY OF SUFFACE WATER AND SEDIMENT ANALYTICAL RESULTS NYANZA HI FEMEDIAL INVESTIGATION SUDBURY RIVER STUDY PAGE 10

							SURFACE W	ATER - ORGAI	NICS						
		,	EACH 10				E	ASTERN WETL	ANDS			CHEMICAL BR	DOK CULVERT	-	
	FREQUENCY	AVE MADE DETECTED	AVENAGE REPORTED	MAJONKIM	MINIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE	MAXIMUM	MINIMUM
CHLOROMETHANE	** NO 8A	MPLES COLLE	CTED IN THIS P	EACH **		0 OF 2					** NO SA	MPLESCOLLE	CTED IN THIS	REACH **	
1,1-DICHLOROETHENE	}					0 OF 2									
1,2-DICHLOROETHENE (TOTAL)						1 OF 2	80	5 3	80	80					
METHYL ETHYL KETONE						0 OF 2									ł
TRICHLOROETHENE	ł					1 OF 2	85	4 5	65	6 5					
1,4-DICHLOROBENZENE						0 OF 2				1					1
1.2-DICHLOROBENZENE						1 OF 2	30	40	30	30					1
BIS(2-ETHYLHEXYL) PHTHALATE						0 OF 2				1					İ
GAMMA-BHC (UNDANE)	1.					0 OF 2									

l							SURFACE W	ATER - ORGA UNITS ug/1	NICS						
ORCIANIC				EK .				RACEWAY					COLD SPRING	BROOK	· · · · (
COMPOUNDS	FREQUENCY	AVENAGE DETECTED	AVERAGE REPORTED	MAXIMUM		FREQUENCY	AVE PAGE DE TECTED	AVERAGE	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE	AVE PAGE REPORTED	MAXIMUM	MINIMUM
CHLOROMETHANE	0 OF 1					8061					0 OF 1				
1.1-DICHLOROETHENE	0.071					0.0F1					0 OF 1				
1,2-DICHLOROETHENE (TOTAL)	10F1	12 0	120	120	120	10F1	2	2	2	2	0 OF 1				
METHYL ETHYL KETONE	0.04.0					0.0F1					0 OF 1				
TRICHLOROETHENE	1071	13.0	13.0	130	130	10F1	2	2	2	2	0 OF 1				
1.4 - DICHLOROBENZENE	1071	10	1.0	1.0	10	0 OF 1					0 OF 1				
1,2-DICHLOROBENZENE	1011	40	40	40	40	0.0F1					0 OF 1				
BIS(2-ETHYLHEXYL) PHTHALATE	1071	1.0	10	10	10	0.0F1					1 OF 1	58.0	58 0	58 0	58 0
GAMMA-BHC (UNDANE)	0.0#1					0.0F1					0 OF 1				

	UNITS ugt	
ORGANIC COMPOUNDS	BORDE RING WET LANDS	HEARD POND
	FREQUENCY DETECTED REPORTED MAXIMUM ANNIMUM	AVERAGE AVERAGE FREQUENCY DETECTED REPORTED MAXIMUM MINIMU
CHLOROMETHANE 1,1 - DICHLOROETHENE 1,2 - DICHLOROETHENE (TOTAL) METHYL ETHYL KETONE TRICHLOROETHENE 1,4 - DICHLOROBENZENE BIB(2 - ETHYLHEXYL) PHTHALATE DAMMA - BIC (LINDANE)	** NO BAMPLES COLLECTED IN THIS REACH **	** NO BAMPLES COLLECTED IN THIS REACH **

.

TABLE 4-1 BUMMARY OF BURFACE WATER AND BEDMENT ANALYTICAL REBULTS NYANZA III REMEDIAL INVESTIGATION BUDBURY RIVER STUDY PAGE 20

[–		BURFACE W	ATER - ORGA	NICS						
								UNITS up/	· ·						
ORGANIC COMPOUNDS		REACH 1 1	NDOURY RESI CIEDROUND -	ERVOR				REACH 2				r	REACH 3		
	FREQUENCY	DETECTED	REPORTED	MATINUM	MINIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MNIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE	MAXIMUM	MINIMUM
CHLOROMETHANE	1077	11.0		11.0	11 0	0 OF 7					0 OF 5				
1,1-DICHLOROETHENE	2017		24	30	30	0 OF 7					0 OF 5				
1,2-DICHLOROETHENE (TOTAL)	001					1047	10	23	10	10	0 OF 5				
METHYL ETHYL KETONE	0.041					0 OF 6					0 OF 3				
TRICHLOROETHENE	0077					0 04 7					0 OF 5				
1,4-DICHLOROBENZENE	001					0 OF 7				1	0 OF 5				
1.2 - DICHLOROBENZENE	0.01 1					0 OF 7					0 OF 5				
BIB (2-ETHYLHEXYL) PHTHALATE	0.011					6 OF 7					1 OF 5	10	4 2	10	10
GAMMA-BHC (LINDANE)	0.047					0 OF 7					0 OF 5				

L

4-21

W92194F

	1					1									
ORGANIC		•	NEACH 4					REACH 5				,	REACH 6		
COMPOUNDS	FRE CLUENCY	AVENADE DE TECHED	AVERAGE NE POR VED	MADMUM			AVE AAGE DE TE C TE D	AVE RAGE REPOR TED	MAXIMUM	MNIMUM	FREQUENCY	AVERAGE DE TEC TED	AVERAGE	MAXIMUM	MNIMUM
CHLOROMETHANE	.073					0 OF 2					0 OF 4				
1,1-DICHLOROETHENE	• OF 2					0 OF 2					0 OF 4				
1,2-DICHLOROETHENE (TOTAL)	0.04.5					0 OF 2					0 OF 4				
METHYL ETHYL KETONE	1078	10.0	7 8	18.8	18 0	0 OF 2					0 OF 4				
TRICHLORGE THE NE	1 In 1					0072					0 OF 4				
1,4-DICHLOROBENZENE	0 OF 2					0 07 2					0 OF 4				
1,2-DICHLOROBENZENE	0013					0.04.5					0 OF 4				
BIB(2-ETHYLHEXYL) PHTHALATE	072					0 OF 2					0 OF 4				
GAMMA-BHC (LINDANE)	072					0072					1 OF 4	0 0 1 5	0 023	0 015	0 0 1 5

· · · ·							BUAFACE W	ATER - ORGAN	NICS						
	T		- ·		• •	r				f					·· · - · -
ORGANIC	1		ACH 7					REACH 8				,	REACH 9		
COMPOUNDS	FREQUENCY	AVE ANDE DE TEC TED	AVERAGE REPORTED	MAJOWVM		FREQUENCY	AVE RADE DE TECTED	AVE RAGE REPORTED	MAXIMUM	MINIMUM	FREQUENCY	AVE RAGE DE TECTED	AVE RAGE REPORTED	MAXIMUM	MINIMUM
CHLOROMETHANE	• ~ •					** NO 84	WPLES COLE	CTED IN THIS	REACH **		** NO SA	MPLES COLLE	CTED IN THIS	REACH **	
1,1-DICHLOROETHENE	0.07.4														
1.2-DICHLOROETHENE (TOTAL)	0 OF 4														
METHYLETHYL KETONE	1 0 m 4														
TRICHLORGETHENE	004														
1.4 - DICHLOROBENZENE	8.0F.4														
1,2-DICHLOROBENZENE	8 11/4														
BIB(2-ETHYLHEXYL) PHTHALATE	0074														
GAMMA-BHC (LINDANE)	0.04.4					1									

FINAL



1 ŇN

TABLE 4-2 COMPARISON OF PARAMETERS IN SURFACE WATER TO STANDARDS NYANZA III REMEDIAL INVESTIGATION SUDBURY RIVER STUDY

		STAN U	DARDS G/L				DETECTED AN	ERAGE CO	NCENTRATIO	45			
PARAMETER					BÁCKGROUNE		WATERSHE	DAREA		TRIBUT	ARIES	OTHER AF	REAS
	MCL (1)	MCLO (1)	AWQC: ACUTE (2)	AWQC: CHRONIC (2)	REACH 1 SUDBURY RES RES. 3	EASTERN WETLANDS	CULVERT (4)	OUTFALL CREEK	RACEWAY	COLD SPRING BROOK	HEARD POND	BORDERING WETLANDS	EASTERN WETLANI BORINGS
INORGANICS:				_ \	••••••••••••••••••••••••••••••••••••••								
					220.00	756 00	NS	203.00	0.00	165.00	NS	NS	N
ANTIMONY	5	3	88.0	30.0	0.00	0 00	NS	0.00	0.00	0.00	NS	NS	N
ARSENIC (III)	50	•	360.0	190.0	0.00	0 00	NS	0.00	0.00	0.00	NS	NS	N
BARIUM	2000	2000			18.15	6 53	NS	10.50	16.40	0.00	NS	NS	N
BERYLLIUM	1	0	130 0	53	0 00	0 00	NS	0.00	0.00	0.00	NS	NS	N
CADMIUM	, s	5		11	0 00	0.00	NS	0.00	0.00	0.00	NS	NS	N
CALCIUM	, v	•		•	10217 14	14250.00	NS	17700.00	10025.00	9570.00	NS	NS	N
	100	100	1700.0	210.0	0.00	79.00	NS	5 60	0.00	0.00	NS	NS	N
COBALT					0.00	0.00	NS	2.30	0.00	0.00	NS	NS	N
COPPER	1300	1300	18.0	12.0	3 10	14.60	NS	0.00	0.00	0.00	NS	NS	N
IPON				1000.0	967.20	1420.00	NS	1870.00	1070.00	688.00	NS	NG	N
		•		10000	21.10	4.03	NG	5 70	0.00	0.00	NS		N
	5	v	030	J.2	2450.00	1187 50	NS	1830.00	2350.00	2400.00	NG		N N
MAURANERE					2450 00	107.30	113	1030.00	2330.00	2450.00	NO		N N
MANGANESE		•		0.013	0.00	2 00.73	NO	140.00	203.30	108.00	NG		N N
MERCONT				10012	17.70	209	NO	0.40	0.00	0.00	NO NO		N N
NUCKEL	100	100	1400 0	1000	2491 43	074 50	NO	1700.00	1676.00	2520.00	NG		n N
POTASSIUM			-		240143	0.00	NO	1790.00	1675.00	2030.00	NG		N N
SELENIUM		00	200	50	0.00	0.00		0.00	0.00	42.00			
SILVEN			• •	01	10 90	44000.00	NS	00.0	0.00	13.90	NG		ri N
SOLIOM					200/1.43	11090.00	NS	21400.00	33000.00	10000.00	NS	NS NS	N
THALLIUM	2	05	1400 0	40 0	000	0.00	NS	0.00	0.00	0.00	NS	NS NS	N
					0.00	0.00	NS	0.00	0.00	0.00	NS	NS NS	N
ZINC			120 0	1100	6.20	0.00	NS	47.90	0.00	0.00	NS	NS NS	N
ORGANIC COMPOUNDS										·			
CHLOROMETHANE					11 00	0.00	NS	0.00	0.00	0.00	NS	NS	N
1.1-DICHLOROETHENE	7	7	11600		3.00	0 00	NS	0.00	0.00	0.00	NS	NS	N
1.2-DICHLOROETHENE (CIS)	70	70	11600	1	0 00	8 00	NS	12.00	2.00	0.00	NS	NS	Ň
METHYL ETHYL KETONE	1				0.00	0.00	NS	0.00	0.00	0.00	NS	NS	N
TRICHLOBOETHENE	5	0	45000	21900	0.00	6 50	NSI	13.00	2.00	0.00	NS	NS	Ň
	. em	800	1120	763	0.00	0.00	NS	1.00	0.00	0.00	NS	NS	N
12-DICHLOROBENZENE	74	75	1120	763	0.00	3 00	NS	4 00	0.00	0.00	NS	NS NS	N
818/2-FTHYI HEXYI JPHTHALATE	''				0.00	0.00	NS	1 00	0.00	58.00	NS	NS	N
					0.00	0.00	Ng	0.00	0.00	0.00	NG	NG NG	N

(1) MCLS AND MCLOS FROM USEPA, APRIL, 1992 (2) AWGC FROM USEPA, MAY 1, 1991, TOXICITY TO AQUATIC ORGANISMS

(3) AWQC PRESENTED FOR CHROMIUM IS FOR THE TRIVALENT FORM

(4) NS - NO SAMPLES COLLECTED

(5) THE MCL PRESENTED FOR LEAD IS A PROPOSED VALUE (5/1/01) THE PRESENT DRINKING WATER ACTION LEVEL FOR LEAD IS 15 ug/l (4/92)

BOXED VALUES EXCEED MCL

TABLE 4-

W92194F

Ň

COMPARISON OF PARAMETERS IN SURFACE WATER TO STANDARDS NYANZA III REMEDIAL INVESTIGATION

PAGE 2

· · · · · · · · · · · · · · · ·			·	· · · · · · ·									
		STAN	DARIDS										
PARAMETER								F	REACHES				
	MCL (1)	MCLG (1)	AWQC: ACUTE	AWQC: CHRONIC (2)	2	3	4	5	6	7	8	9	10
INORGANICS:							····						
					1176.50	128.70	714.80	0.00	216.53	103.00	NS	NS	NS
	5	3	340.0	190.0	3.00	0.00	0.00	1 10	0.00	0.00	NS	NS	NG
	2000	2000	000.0		33.81	12.55	24 70	12.75	16.20	21.30	NS	NS	NS
BERMILIUM	1	0	130.0	53	7.50	0.00	0.00	0.00	0.00	0.00	NS	NS	NS
CADMIUM	5	5	3.9	1.1	7.20	0.00	0.00	0.00	0.00	0.00	NS	NS	NS
CALCIUM	-				9317.86	7052.00	7390.00	9722.50	9028.75	11365.00	NS	NS	NS
CHROMIUM (III)	100	100	1700.0	210.0	7.00	0.00	6.00	0.00	0.00	0.00	NS	NS	NS
COBALT					27.70	0.00	0.00	0.00	0.00	0.00	NS	NS	NS
COPPER	1300	1300	18.0) 12.0	27.70	0 00	0.00	0.00	0.00	0.00	NS	NS	NS
IRON				1000.0	3943.60	551.80	1515.00	0.00	469.33	528.67	NS	NS	NS
LEAD (5)	5	0	63 0	32	33.90	1.00[6.85	0.00	2.40	0.00	NS	NS	NS
MAGNESIUM				[2233 57	1730.00	1800.00	2212.50	2052.50	2560.00	NS	NS	NS
MANGANESE					1533.70	63,38	128.85	106.00	96.35	85.90	NS	NS	NS
MERCURY	2	2	24	0.012	0.00	0.00	0.00	0.00	0.00	0.00	NS	NS	NS
NICKEL	100	100	1400.0) 160.0	77.40	0 00	0.00	0.00	0.00	0.00	NS	NS	NS
POTASSIUM					1971.43	1952.00	1285.00	1870.00	2156.25	2167.50	NS	NS	NS
SELENIUM	50	50	20 0	5.0	0.00	19300.00	0.00	0.00	0.00	0,00	NS	NS	NS
SILVER			4.1	0.1	68.90	9.90	0.00	0.00	00.0	6.50	NS	NS	NS
SODIUM					28/5/.14	19950.00	19900.00	262/5.00	22500.00	24850.00	NS	NS	NS
THALLIUM	2	0.5	1400.0	40.0	0.00	0.00	0.00	0.00	0.00	0.00	NS	NS	NS
VANADIUM					14.35	0.00	0.00	0.00	0.00	0,00	NS	NS	NS
			120.0	110.0	125.00	8.00	0.00	0.00	0.00	0.00	NS	NS	NS
ORGANIC COMPOUNDS													<u> </u>
CHLOROMETHANE		_			0.00	0.00	0.00	0.00	0.00	0.00	NS	NS	NS
1,1-DICHLOROETHENE	7	7	11600		0.00	0.00	0.00	0.00	0.00	0.00	NS	NS	NS
1,2-DICHLORDETHENE (CIS)	70	70	11600		1.00	0.00	0.00	0.00	0.00	0.00	NS	NS	NS
METHYL ETHYL KETONE	-		AR.000	-	0.00	0.00	10.00	0.00	0.00	0.00	NS	NS	NS
		0	45000	21900	0.00	0.00	0.00	0.00	0.00	0.00	NS	NS NC	NS
	600	000	1120	/63	0.00	0.00	0.00	0.00	0.00	0.00	NS	NS NC	NS NO
	/5	/5	1120	/63	0.00	1.00	0.00	0.00	0.00	0.00	NO	ND	NS
				امم ا	0.00	1.00	0.00	0.00	0.00	0.00		NO	NO
GAMMA-BHU (LINDANE)	02	02	2	. 0.0	0.00	0.00	0.00	0.00	0.02	0.00	N)	CN3	115
· · · · · · · · · · · · · · · · · · ·													

(1) MCLS AND MCLGS FROM USEPA, AP RIL, 1992 (2) ANOC FROM USEPA, MAY 1, 1991, TOXICITY TO AQUATIC ORGANISMS

(3) AWOC PRESENTED FOR CHROMIUM IS FOR THE TRIVALENT FORM

(4) NS - NO SAMPLES COLLECTED (5) THE MCL PRESENTED FOR LEAD IS / PROPOSED VALUE (5/1/91). THE PRESENT DRINKING WATER ACTION LEVEL FOR LEAD IS 15 ug/1 (4/92)

BOXED VALUES EXCEED MCL

TABLE 4-3 SUMMARY OF FISH ANALYTICAL RESULTS: FILLET BY SPECIES NYANZA III REMEDIAL INVESTIGATION SUDBURY RIVER STUDY mg/kg WET WEIGHT

	T	Ĩ	SUDBURY RES	ERVOIR, LARG	EMOUTH			C	EDAR SWAM	PPOND, LARG	EMOUTHBASS	3
PARAMETER	FREQUE	INCY	DETECTED AVERAGE	REPORTED AVERAGE	MAXIMUM	MINIMUM	FREQUE	NCY	DETECTED AVERAGE	REPORTED AVERAGE	MAXIMUM	MINIMUM
ALUMINUM	3 OF	10	13.65	6.772	14.69	13.06	0 OF	7				
ANTIMONY	0 OF	10					1 OF	7	0.82	0.631	0.82	0.82
ARSENIC	1 OF	10	0 06	0.096	0.06	0.06	0 OF	7				
BARIUM	0 OF	10					0 OF	7				
BERYLLIUM	0 OF	10					0 OF	7				
CADMIUM	0 OF	10					0 OF	7				
CALCIUM	10 OF	10	400.95	400 950	513.22	303.53	7 OF	7	1402.71	1402.714	6460.00	179.00
CHROMIUM	5 OF	10	1.89	1.114	2.84	1.33	2 OF	7	0.43	0.194	0.64	0.22
COBALT	6 OF	10	0 05	0.229	0.08	0.03	0 OF	7				
COPPER	0 OF	10					0 OF	7				
IRON	2 OF	10	82.23	29.001	93.61	70.86	0 OF	7				
LEAD	0 OF	10					1 OF	7	1.17	0.239	1.17	1.17
MAGNESIUM	10 OF	10	379.13	379.128	519.28	264.71	7 OF	7	219.43	219.429	284.00	160.00
MANGANESE	7 OF	10	1.45	1.062	2.43	0.57	0 OF	7				
MERCURY	5 OF	10	0 89	0.489	1.18	0.55	4 OF	7	0.60	0.344	0.96	0.36
NICKEL	0 OF	10					1 OF	7	6.00	1.200	6.00	6.00
POTASSIUM	10 OF	10	5432.44	5432.440	6446.94	3897.49	7 OF	7	2720.00	2720 .000	3400.00	1940.00
SELENIUM	1 OF	10	1 80	0.460	1.80	1.80	6 OF	7	0.12	0.109	0.50	0.03
SILVER	0 OF	10					1 OF	7	2.20	0.397	2.20	2.20
SODIUM	10 OF	10	418.12	418,120	524.30	330.84	7 OF	7	529.71	529.714	984.00	376.00
THALLIUM	0 OF	10					0 OF	7				
VANADIUM	6 OF	10	0.49	0.325	0.76	0.31	0 OF	7				
ZINC	10 OF	10	32.36	32.359	61.62	9.72	7 OF	7	13.17	13.171	56.80	3.40

(

4-24

(

TABLE 4–3 SUMMARY OF FISH ANALYTICAL REDULTS FILLET BY SPECIES NYANZA III REMEDIAL INVESTIGATION SUDBURY RIVER STUDY mg/kg Wet weight PAGE 2

		(EDAR SWAM	P POND, YELLO	OW PERCH			Ş	OUTHVILLE P	OND, LARGEM	OUTH BASS	
PARAMETER	FREQUE	NCY	DETECTED AVERAGE	REPORTED AVERAGE	MAXIMUM	MINIMUM	FREQUE	INCY	DETECTED AVERAGE	REPORTED AVERAGE	MAXIMUM	MINIMUM
ALUMINUM	0 OF	10		<u></u>			0 OF	7		<u> </u>		
ANTIMONY	2 OF	10	0.22	0.524	0.25	0.19	0 OF	7				
ARSENIC	0 OF	10					0 OF	7				
BARIUM	0 OF	10					0 OF	7				(
BERYLLIUM	0 OF	10					0 O F	7				
CADMIUM	0 OF	10					0 OF	7				
CALCIUM	10 OF	10	2460 90	2460.900	8240.00	133.00	7 OF	7	5245.71	5245.714	12000.00	700.00
CHROMIUM	8 OF	10	0 25	0.222	0.56	0.16	5 OF	7	0.32	0.254	0.70	0.20
COBALT	0 OF	10					1 OF	7	0.48	0.497	0.48	0.48
COPPER	0 OF	10					2 OF	7	2.35	0.877	3.10	1.60
IRON	0 OF	10					1 OF	7	14.30	5.371	14.30	14.30
LEAD	0 OF	7					1 O F	7	0.77	0.239	0.77	0.77
MAGNESIUM	10 OF	10	247 40	247.400	298.00	214.00	7 OF	7	291.71	291 .714	376.00	228.00
MANGANESE	0 OF	10					1 OF	7	0.38	0.629	0.38	0.38
MERCURY	3 OF	10	3.33	1.004	9.60	0.20	4 OF	7	0.55	0.315	0.89	0.28
NICKEL	0 OF	10					0 O F	7				
POTASSIUM	10 OF	10	3196 00	3196 000	3380.00	2940.00	7 OF	7	3288.57	3288.571	3660.00	2980.00
SELENIUM	8 OF	10	0.56	0.454	4.00	0.03	7 OF	7	0.10	0.097	0.15	0.06
SILVER	0 OF	10					0 OF	7				
SODIUM	10 OF	10	456 60	456.600	534.00	336.00	7 OF	7	606.86	606.857	704.00	466.00
THALLIUM	1 OF	7	0.08	0.097	0.08	0.08	0 O F	0				
VANADIUM	0 OF	10					0 OF	7				1
ZINC	10 OF	10	7.72	7.720	12.40	<u>5.40</u>	6 OF	7	14.00	9.307	21,40	8.60

١

FINAL

.

.

i

		A	AILL POND, YE	LLOW PERCH	· _			ħ	AILL POND, LA	RGEMOUTHB	ASS	
PARAMETER	FREQUE	NCY	DETECTED AVERAGE	REPORTED AVERAGE	MAXIMUM	MINIMUM	FREQUE	NCY	DETECTED AVERAGE	REPORTED AVERAGE	MAXIMUM	MINIMUM
ALUMINUM	0 ÖF	10					0 OF	20		······································		
ANTIMONY	0 OF	10					0 OF	20				
ARSENIC	0 OF	10					0 OF	20				
BARIUM	0 OF	10					0 O F	20				
BERYLLIUM	0 OF	10					0 O F	20				
CADMIUM	0 OF	10					0 O F	20				
CALCIUM	10 OF	10	148 52	148.519	210.13	110.51	20 OF	20	135.82	135.817	391.44	87.16
CHROMIUM) 0 OF	10					0 O F	20				
COBALT	0 OF	10					0 OF	20				}
COPPER	2 OF	10	0.37	0 274	0.40	0.34	0 OF	20				
IRON	0 OF	10					0 O F	20				
LEAD	0 OF	10					0 O F	20				
MAGNESIUM	10 OF	10	240 49	240.489	257.67	223.87	20 OF	20	243.84	243.841	269.31	213.92
MANGANESE	10 OF	10	0.19	0.188	0.31	0.14	13 OF	20	0.13	0.093	0.20	0.08
MERCURY	2 OF	10	1.43	0.444	1.98	0.88	1 OF	20	1.12	0.250	1.12	1.12
NICKEL	0 OF	10					0 O F	20				i
POTASSIUM	10 OF	10	3546 22	3546.217	3861.39	3208.18	20 OF	20	4868.57	4868.566	33251.96	3070.58
SELENIUM	0 OF	10					0 O F	20				
SILVER	1 0F	10	0 02	0.092	0.02	0.02	0 O F	20				
SODIUM	10 OF	10	485 64	485.642	538.51	464.24	20 OF	20	551.66	551. 663	614.83	459.40
THALLIUM	0 OF	10					0 O F	20				
VANADIUM	1 OF	10	0.02	0.452	0.02	0.02	14 OF	20	0.04	0.175	0.05	0.02
ZINC	10 OF	10	3.55	3.550	3.91	3.27	20 OF	20	3.16	3.157	3,79	2.59

(

.

(

W92194F

FINAL

(

		Ē	RESERVOIR 2,	YELLOW PERC	ЭН		<u>+</u>	F	RESERVOIR 2,	LARGEMOUTH	BASS	
PARAMETER	FREQUEN	 1CY	DETECTED AVERAGE	REPORTED AVERAGE	MAXIMUM	MINIMUM	FREQUE	NCY	DETECTED AVERAGE	REPORTED AVERAGE	MAXIMUM	MINIMUM
ALUMINUM	1 OF	16	24.35	2.776	24.35	24.35	6 OF	34	13.88	3.274	16,56	12.42
ANTIMONY	0 OF	16					1 OF	31	15.50	0.962	15.50	15.50
ARSENIC	2 OF	16	0.06	0.080	0.07	0.05	3 OF	34	0.61	0.138	1.75	0.03
BARIUM	0 O F	16					0 OF	34				
BERYLLIUM	0 OF	16					0 OF	34				
CADMIUM	0 OF	16					1 OF	34	0.05	0.050	0.05	0:05
CALCIUM	16 OF	16	404.12	404.119	2393.96	92.30	28 OF	34	157.84	138.812	462.66	68.06
CHROMIUM	0 O F	16					8 OF	34	1.95	0.572	3.41	1.23
COBALT	2 OF	16	0.03	0.442	0.04	0.03	9 OF	33	0.06	0.380	0.13	0.04
COPPER	0 OF	16					4 OF	34	0.62	0.354	1.83	0.21
IRON	0 OF	16					6 OF	34	90.77	18.752	131.68	63.45
LEAD	0 OF	16					4 OF	34	0.41	0.081	0.53	0.29
MAGNESIUM	16 OF	16	252.09	252.087	308.26	209.43	28 OF	34	261.25	223.967	424.63	227.7 9
MANGANESE	14 OF	16	1.52	1.356	13.75	0.21	23 OF	34	0.55	0.382	1.99	0.07
MERCURY	12 OF	16	1.56	1.244	3.64	0.78	32 OF	34	2.87	2.719	7.60	0.80
NICKEL	0 OF	16					0 OF	34				
POTASSIUM	16 OF	16	3747.80	3747.796	5492.87	3068.97	28 OF	34	3694.25	2946.477	5263.24	3148.47
SELENIUM	0 OF	16					0 OF	34				0.00
SILVER	0 OF	16					1 OF	31	0.44	0.104	0.44	0.44
SODIUM	16 OF	16	437.83	437.828	561.54	249,91	28 OF	34	361.68	401.955	576.16	0.00
THALLIUM	0 OF	16					1 OF	31	1.35	0.135	1.35	1.35
VANADIUM	0 OF	16					13 OF	34	0.39	0.433	1.15	0.02
ZINC	15 OF	16	10,11	9.727	46.33	2.85	33 OF	34	14.44	14.144	85.36	0.31

W92194F

FINAL

-**W**-

TABLE 4–3 SUMMARY OF FIGH ANALYTICAL REBULTS: FRLLET BY BRECIES NYANZA III REMEDIAL INVESTIGATION SUDBURY RIVER STUDY mg/kg WET WEIGHT PAGE 5

	1	F	RESERVOIR 2,	BULLHEAD					RESERVOIR 1,	YELLOW PERC	ЭН	
PARAMETER	FREQUEN	NCY	DETECTED AVERAGE	REPORTED AVERAGE	MAXIMUM	MINIMUM	FREQUE	NCY	DETECTED AVERAGE	REPORTED AVERAGE	MAXIMUM	MINIMUM
ALUMINUM	6 OF	7	24.76	21.695	57.13	13.62	4 OF	18	22.67	5.239	28.02	12.64
ANTIMONY	0 OF	7					0 OF	14				
ARSENIC	0 OF	7					1 OF	20	0.45	0.114	0.45	0.45
BARIUM	0 OF	7					1 OF	21	1.0 6	0.167	1.06	1.06
BERYLLIUM	0 OF	7					0 OF	21				
CADMIUM	0 OF	7					0 OF	21				
CALCIUM	7 OF	7	370 69	442.115	578.34	81.17	21 OF	21	493.04	493.041	4833.78	120.80
CHROMIUM	5 OF	7	2.59	1.990	5.69	1.40	8 OF	21	2.18	1.004	3.28	1.33
COBALT	7 OF	7	0.09	0.085	0.24	0.03	10 OF	20	0.09	0.294	0.20	0.04
COPPER	3 OF	7	6.00	2.841	9 .72	4.03	0 OF	21				
IRON	5 OF	7	117.67	89.852	267.45	64.28	9 OF	21	96.85	43.504	140.25	74.48
LEAD	0 OF	7					0 OF	21				
MAGNESIUM	7 OF	7	282.08	282.079	392.74	186.33	21 OF	21	302.20	302.197	451.43	223.53
MANGANESE	7 OF	7	1 77	1.770	5.53	0.93	21 OF	21	2.73	2.725	33.81	0.16
MERCURY	6 OF	7	1 99	1.741	2.85	1.26	2 OF	21	0.76	0.219	0.76	0.76
NICKEL	0 OF	7					0 OF	21				
POTASSIUM	7 OF	7	5171.93	5171.931	7109.96	3456.60	21 OF	21	4279.21	4279.212	5963.71	2960.69
SELENIUM	0 OF	7					0 OF	21				•
SILVER	0 OF	7					0 O F	15				
SODIUM	7 OF	7	583.92	583.918	731.70	431.93	21 OF	21	464.32	464.320	637.93	261.49
THALLIUM	0 OF	7					0 O F	15				
VANADIUM	6 OF	7	0.59	0.523	1.16	0.31	9 OF	21	0.45	0.439	0.84	0.31
ZINC	7 OF	7	62.27	62.271	178.54	20.89	21 OF	21	21.32	21.160	68.03	2.95

(

FINAL

•

(

TABLE 4-3 SUMMARY OF FISH ANALYTICAL RESULTS FILLET BY SPECIES NYANZA III REMEDIAL INVESTIGATION SUDBURY RIVER STUDY mg/kg WET WEIGHT PAGE 6

.

		F	ESERVOIR 1,	LARGEMOUTH	BASS					RESERVOIR 1,	BULLHEAD	
PARAMETER	FREQUE	NCY		REPORTED AVERAGE	MAXIMUM	MINIMUM	FREQUE	NCY	DETECTED AVERAGE	REPORTED AVERAGE	MAXIMUM	MINIMUM
ALUMINUM	1 OF	21	22.25	1.822	22.25	22.25	• 7 OF	8	94.28	82.882	489.93	14.72
ANTIMONY	0 OF	30					0 OF	6				
ARSENIC	4 OF	30	0.04	0.081	0.04	0.03	0 OF	9			•	
BARIUM	0 OF	30					0 OF	9				
BERYLLIUM	0 OF	30					0 O F	9				
CADMIUM	0 OF	30					0 OF	9				
CALCIUM	30 OF	30	234.76	234.763	730.75	87.36	9 OF	9	754.49	754.494	4189.04	232.84
CHROMIUM	5 OF	30	3 31	0.989	8.32	1.54	7 OF	9	2.06	1.686	3.94	1.17
COBALT	10 OF	30	0.15	0.366	0.53	0.04	9 O F	9	0.07	0.071	0.13	0.03
COPPER	2 OF	30	5.16	0.792	6.31	4.01	1 OF	9	4.04	1.156	4.04	4.04
IRON	10 OF	30	168.67	56.742	302.61	61.09	6 OF	9	127.14	93.194	182.76	80.18
LEAD	0 OF	29					1 OF	9	2.16	0.273	2.16	2.16
MAGNESIUM	30 OF	30	300.56	300.559	592.38	224.30	9 OF	9	334.44	334.438	461.59	236.43
MANGANESE	25 OF	30	1.75	1.487	10.68	0.09	9 O F	9	3.78	3,776	9.09	1.05
MERCURY	24 OF	30	1.47	1.225	4.19	0.71	3 OF	9	0.92	0.381	1.20	0.70
NICKEL	0 OF	30					0 O F	9				
POTASSIUM	30 OF	30	3963.73	3 963 .733	6335.70	2580.56	9 OF	9	5201.66	5201.658	5992.34	3879.03
SELENIUM	0 OF	30					0 OF	9				
SILVER	0 OF	30					0 OF	6				
SODIUM	30 OF	30	475.67	489.004	625.67	33.48	9 O F	9	675.83	675.828	860.79	447.38
THALLIUM	0 OF	30					0 OF	6				
VANADIUM	9 OF	30	0.81	0.582	1.57	0.36	6 OF	9	0.47	0.347	0.82	0.35
ZINC	29 OF	30	31.72	31,368	262.46	2.86	8 OF	9	64.27	57.500	161.96	25.48

-

_

1

W92194F

FINAL

.

TABLE 4-3 SUMMARY OF FISH ANALYTICAL REBULTS: FILLET BY OPECIES NYANZA III REMEDIAL INVESTIGATION SUDBURY RIVER STUDY mg/kg WET WEIGHT PAGE 7

(

	Ī	5	AXONVILLE	POUNDMENT	LARGEMOUT	HBASS		5	SAXONVILLE II	MPOUNDMENT	, YELLOW PER	СН
PARAMETER	FREQUE	NCY	DETECTED AVERAGE	REPORTED AVERAGE	MAXIMUM	MINIMUM	FREQUE	NCY	DETECTED AVERAGE	REPORTED AVERAGE	MAXIMUM	MINIMUM
ALUMINUM	1 OF	23	14.01	3.930	14.01	14.01	4 OF	20	24.16	6.946	30.15	14.73
ANTIMONY	1 OF	23	0,14	0.580	0.14	0.14	0 O F	20				
ARSENIC	0 OF	23					1 OF	20	0.05	0.076	0.05	0.05
BARIUM	0 OF	23					0 OF	20				
BERYLLIUM	0 OF	23					0 OF	20				
CADMIUM	0 OF	23					0 OF	20				
CALCIUM	23 OF	23	2550.19	2550.194	19500.00	102.00	20 OF	20	2762 .61	2762.609	12200.00	340.02
CHROMIUM	15 OF	23	0.39	0.317	1.74	0.14	12 OF	20	1.35	0.929	4.28	0.14
COBALT	3 OF	23	0.13	0.452	0.32	0.03	10 OF	20	0.11	0.307	0.60	0.03
COPPER	0 OF	23					3 OF	20	3.96	0.856	6.10	1.67
IRON	1 OF	23	66.90	5.823	66.90	66.90	4 OF	20	123.12	30.449	155.91	69.21
LEAD	0 OF	23					1 OF	20	0.69	0.072	0.69	0.69
MAGNESIUM	23 OF	23	267.07	267.072	414.00	202.00	20 OF	20	333.84	3 33.83 5	499.59	189.00
MANGANESE	4 OF	23	2.04	0.543	5.94	0.41	15 OF	20	5.21	4.111	19.70	0.51
MERCURY	17 OF	23	0,94	0.700	1.80	0.20	15 OF	20	0.70	0.555	1.40	0.20
NICKEL	2 OF	23	1,17	0.470	1.24	1.10	0 OF	20				
POTASSIUM	23 OF	23	3439.44	3439.440	6636.87	2540.00	20 OF	20	4348.96	4348.957	6826.89	2360.00
SELENIUM	20 OF	23	0.19	0.195	0.62	0.03	10 OF	20	0.10	0.161	0.18	0.04
SILVER	1 OF	23	0.24	0.106	0.24	0.24	0 OF	20				
SODIUM	23 OF	23	644.78	644.780	1160.00	248.50	20 OF	20	502.54	502.543	848.00	195.64
THALLIUM	0 OF	23					1 OF	20	1.19	0.130	1.19	1.19
VANADIUM	2 OF	23	0.41	0.460	0.49	0.33	4 OF	20	0.95	0.474	1.22	0.64
ZINC	23 OF	23	8.84	8.844	42.73	1.69	20 OF	20	26.36	26.364	94.79	5.54

(

FINAL

(

•

ī

TABLE 4-3 SUMMARY OF FISH ANALYTICAL REBULTS FILLET BY BRECHS NYANZA HI REMEDIAL INVESTIGATION SUDBURY RIVER STUDY mg/kg WET WEIGHT PAGE 8

÷.

			SAXÖNVILLE II	MPOUNDMENT	BULLHEAD			F	AIRHAVEN BA	Y, LARGEMOU	ITH BASS	
PARAMETER	FREQUE	NCY	DETECTED AVERAGE	REPORTED AVERAGE	MAXIMUM	MINIMUM	FREQUE	NCY	DETECTED AVERAGE	REPORTED AVERAGE	MAXIMUM	MINIMUM
ALUMINUM	2 OF	ã-	15.85	9.945	19.00	12.70	0 OF	20				
ANTIMONY	0 OF	4					0 O F	20				
ARSENIC	1 OF	4	0 04	0.085	0.04	0.04	0 OF	20				
BARIUM	0 OF	4					0 O F	20				
BERYLLIUM	0 O F	4					0 O F	20				
CADMIUM	0 O F	4					0 OF	20				
CALCIUM	4 OF	4	405.31	405.311	513.72	310.32	20 OF	20	3283.80	3283.800	9900.00	116.00
CHROMIUM	1 OF	4	3 47	1.202	3.47	3.47	15 OF	20	0.55	0.439	3.40	0.14
COBALT	3 OF	4	0 04	0.155	0.06	0.02	2 OF	20	0.82	0.532	1.26	0.38
COPPER	0 OF	4					7 OF	20	2.12	0.954	3.40	1.20
IRON	1 OF	4	124 09	44.963	124.09	124.09	2 OF	20	19.15	4.733	23,40	14.90
LEAD	0 OF	4					2 OF	20	3.20	0.414	5.60	0.79
MAGNESIUM	4 OF	4	373 81	373.809	511.98	261.46	20 OF	20	241.45	241.450	314.00	186.00
MANGANESE	4 OF	4	1.11	1.108	2.27	0.64	1 OF	20	0.32	0.269	0.32	0.32
MERCURY	3 OF	4	0 74	0.601	0.88	0.60	19 OF	20	1.32	1.256	3.20	0.20
NICKEL	0 OF	4					1 OF	20	2.00	0.480	2.00	2.00
POTASSIUM	4 OF	4	6247.23	6247.230	7862.15	4479.02	20 OF	20	2945.00	2945.000	3520.00	2160.00
SELENIUM	0 OF	4					17 OF	20	0.08	0.073	0.19	0.03
SILVER	0 OF	4					0 OF	20				
SODIUM	4 OF	4	625 21	625 209	779.19	429.31	20 OF	20	658.80	658.800	1100.00	496.00
THALLIUM	0 OF	4					0 O F	17				
VANADIUM	0 OF	4					0 O F	20				
ZINC	4 OF	4	29.04	29.037	52.44	15.35	19 OF	20	8.85	8.409	12.80	5.30

FINAL

TABLE 4–3 SUMMARY OF FISH ANALYTICAL REBULTS: FILLET BY SPECIES NYANZA III REMEDIAL INVESTIGATION SUDBURY RIVER STUDY mg/kg WET WEIGHT PAGE 0

	F	AIRHAVEN BA	Y, YELLOW PE	RCH	· ··
PARAMETER	FREQUENCY	DETECTED AVERAGE	REPORTED AVERAGE	MAXIMUM	MINIMUM
ALUMINUM	0 OF 10				
ANTIMONY	0 OF 10				
ARSENIC	0 OF 10				
BARIUM	1 OF 10	1.10	1.910	1.10	1.10
BERYLLIUM	0 OF 10				
CADMIUM	0 OF 10				
CALCIUM	10 OF 10	9144.40	9144.400	28400.00	774.00
CHROMIUM	7 OF 10	0.64	0.480	1.40	0.20
COBALT	0 OF 10				
COPPER	3 OF 10	1.80	0.745	2.20	1.50
IRON	1 OF 10	23.80	5.195	23.80	23.80
LEAD	0 OF 10				Ì
MAGNESIUM	10 OF 10	285.40	285.400	474.00	202.00
MANGANESE	6 OF 10	13.00	8.040	33.20	5. 9 0
MERCURY	9 OF 10	0.72	0.650	1.80	0.33
NICKEL	0 OF 10				
POTASSIUM	10 OF 10	2887.00	2887.000	3580.00	1890.00
SELENIUM	10 OF 10	0.14	0.136	0.27	0.04
SILVER	0 OF 10				
SODIUM	10 OF 10	611.00	611.000	902.00	482.00
THALLIUM	0 OF 0				
VANADIUM	0 OF 10				
ZINC	10 OF 10	10.04	10.040	21.20	4.40

(

ł

٠

TABLE 4-3 SUMMARY OF FISH ANALYTICAL REBULTS FILLET BY SPECIES NYANZA NI REMEDIAL INVESTIGATION SUDBURY RIVER STUDY UQING WET WEIGHT PAGE 10

[1	1	UDBURY RESERVOI	R				CEDAR SWAMP PON	ID	
ł										
	i ni ni nav	. Aveniór l	ARGEMOUTH BASS			EDE OUEN	AVE DA OF	LARGEMOUTH BASS		
PARAMETER	FREQUENCY	AVEHAGE	AVEHAGE	MAX	MIN	FREQUEN	CY AVEHAGE	E AVEHAGE	MAX	MIN
	-	DETECTED	HEPORIED		• • • • •		DETECTE	D HEPORIED		
Methylens Chloride						0 of	2			
Acetone						0 of	2			
2 - Butanone						0 of	2			
Toluene						0 of	2			
Styrene						0 of	2			
Total Xylenes						0 of	2			
Phenol	1 OF	7 56 000	504 286	56	56	•				-
Bis(2 - Chloroethyl) ether	0 OF 1	1								
Benzyl Alcohol	0 OF 1	1				NOS	AMPLES COLLEC	STED IN THIS REACH		
2 - Methylphenol	OOF	7								
4 - Methylphenol	0 OF	1								
Nitrobenzene	0 OF 1	1				ļ				
Naphthalene	0 OF 1	1				1				
Diethylphthalate	0 OF 1	1								
Fluorene	1 OF 1	1 99 000	540 000	99	99					
Di-n-butylphthelete	1 OF 1	1 45 000	490 909	45	45	}				
Butylbenzylphthalate	0 OF 1	1								
Bis(2 - ethylhexyl)phthelete	0.05	•				1				
Di-n-octylphthelete	0 OF	•								
Benzo(b)fluoranthene	O OF 1	1				}				
Benzo(a)pyrene	0.01	1								
	ł		·· -		· · ·					
Heptechlor	6 OF	7				1				
Aldtin	OOF	•				[
Heptechlor Epoxide	0 OF]				
Endosultan I	0 OF					ļ				
Dieldrin	4.05	a 0.500	3 250	0.5	0.5					
A A'- DDE	B OF	8 19 563	19 563	41	35	í				
Enddo	LOF	a 0.500	5 313	0.5	0.5	Į –				
Endosullan il	OOF	•			•	1				
4 4'- DDD	7 OF	3 429	3 750	7	1	1				
Endosullan Sullate	2 OF	8 2 750	5 188	4 5	1					
4.4'-DDT	5 OF	0 900	2.813	2	0.5	1				
Methoxychior	OOF	•				1				
Endrin Ketone	OOF)				
alpha - Chlordene	1 OF	1 000	28 375	1	1					
gamma - Chlordana	0 OF	•				1				
Aroclor - 1248	0 OF					ļ				
Arociar - 1254	0 OF	3)				
Arocior - 1280	4 OF	8 87 500	63 750	95	20	l				
				·						
Methylmercury	3 OF	3 485	465	652	221	0 of	2			
ł	1					l				

1

.

I.

.

TABLE 4 - 3 SUMMARY OF FIBH ANALYTICAL REBULTS. FILLET BY SPECIES NYANZA HI REMEDIAL INVESITIGATION SUDBURY RIVER STUDY ug/kg WET WEIGHT PAGE 11

W92194F

			ILL POND	· · · · ·			N	ALL POND		
		۲	ELLOW PERCH				L	ARGEMOUTH BASS	5	
PARAMETER	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAX	MIN	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAX	MIN
Methylene Chloride										
Acetone										
2-Butanone										
Toluene						ł				
Styrene										
Total Xylenes										
		· ·-·· -				· · · · · · · · · · · · · · · · · · ·				
Phenol	0 OF 10	1				0 OF	1			
Bis(2 - Chloroethyl) ether	0 OF 10)				0 QF	2			
Benzyl Alcohol	0 OF 10)				0 OF	1			
2 - Methylphenol	0 OF 10)				0 O F	1			
4 - Methylphenol	0 OF 10)				0 OF	1			
Nitrobenzene	0 OF 10)				0 O F	2			
Naphthelene	0 OF 10					0 O F	2			
Diethylphthelate	0 OF 10					0 OF	2			
Fluorene	0 OF 10					0 OF	2			
Di-n-butylphthelate	0 OF 10					0 OF	2			
Butylbenzylphihaide	0 OF 10					0.04	2			
Bis(2-einymexyi)phinalais							2			
Di-n-octyphinalate							2			
Benzo(b)nuoraminene		,				0 OF	2			
		· · · · · · · · · · · · · · · · · · ·								
Heptachior	0 OF 10)				0 OF	2			
Aldrin	0 OF 10)				0 OF	2			
Heptechlor Epoxide	0 OF 10)				OOF	2			
Endosullan I	0 OF 10)				0 OF	2			
Dieldrin	0 OF 10)				0 O F	2			
4,4'-DDE	2 OF 10	27	13.05	30	24	0 OF	2			
Endrin	0 OF 10					0.01	2			
Endosullan II	0 OF 10					100	2			
4.4'-DDD	0 0F 10					100	2			
Endosullan Sullate						0.0F	2			
4,4°-DDI						0 OF	2			
Forder Kelope						0.05	2			
elaba - Chiardene							2			
amma - Chlordene		,				0.05	2			
Aroclor - 1248	1 05 10	, , 600	` #7	500	500		2			
Aroclor - 1254	0 OF 10)			500	0 OF	2			
Aroclor - 1260	0 OF 10)				0 OF	2			
						ļ				
Methylmercury	2 OF 10	550	150 6	730	370	2 OF	2 540	540	660	420

.....

. . .---

- ----

4-34

L____.

1

TA-E 4-3 SUMMARY OF FISH ANALYTICAL REBULTS FILLET BY SPECIES NYANZA HI REMEDIAL INVESTIGATION SUDBURY RIVER STUDY ug/kg WET WEIGHT PAGE 12

	1		RESERVOIR 2					R	ESERVOIR 2		
			YELLOW PERCH					B	ULLHEAD		
PARAMETER	FREQUENC	DETECT	GE AVERAGE IED REPORTED	MĂX	MIN	FREQUEN	CY	AVERAGE DETECTED	AVERAGE REPORTED	MAX	MIN
Methylene Chloride											
Acetone											
2 - Butanone											
Toluene						1					
Styrene	1										
Total Xylenes											
	· · · · · · · · · · · · · · · · · · ·			· · · · · · ·		· ·					
Phanol	0.05	17				0 OF	4				
Bis/2 - Chloroethyll ether	0.0F	17				0 OF	7				
Benzyl Alcohol	0 OF	17				0 O F	7				
2 - Methylphenol	0 OF	17				0 O F	4				
4 - Methylphenol	0 OF	17				0 ØF	4				
Nitrobenzene	0 O F	17				0 O F	7				
Naphthalene	0 OF	17				0 O F	7				
Diethylphthalate	0 OF	17				0 O F	7				
Fluorene	0 OF	17				0 O F	7				
Di-n-butylphthelate	0 OF	17				0 OF	7				
Butylbenzylphthelate	0 OF	17				0 O F	7				
Bis (2 - ethylhexyl) phthelete	0 O F	14				0 OF	0				
Di – n – octylphthelete	0 OF	14				0.01	0				
Benzo(b)fluoranthene	0 OF	17				0.0F	4				
Benzo(a)pyrene	0.01	17				0.01					
Heptachlor	0 OF	17				0 O F	7				
Aldrin	0 O F	17				2 OF	7	0.750	2.357	1	0.5
Heptachlor Epoxide	0 O F	17				0 O F	7				
Endosullan I	0 O F	17				0 OF	7				
Dieldrin	0 OF	17				2 O F	7	0.500	4.429	0.5	0.5
4,4'-DDE	13 OF	17 22	423 20.2	94 62	0.5	7 OF	7	4.571	4.571	7	1
Endrin	0 O F	17				0 O F	7				
Endosülfan II	0 OF	17				1 OF	7	1.000	5.286	1	1
4,4'~DDD	4 OF	17 5	125 97	06 7	2.5	4 OF	7	3.500	4.571	5.5	1.5
Endosullan Sullate	0 OF	17				0 OF	7	4 750	4 700		
4,4'-DDT	4 OF	17 1	013 87	38 15	0.05	201	4	1.750	4./80	2	1.5
Methoxychlor	0 OF	17				TOP	4	3.500	20.214	3,3 3.5	J.3
Endrin Ketone	0 OF	17				10F	',	2.300	5.300	2.J	۷.2
alpha - Chlordane	3 OF	17 6		71 1	0.5	0 0 0F	',	0 750	30.214	1	0.5
gamma - Chiordane	307	47	107 44.0	2 7 13	1		,	0.750	30.214	•	0.0
Arocior - 1248		17		51 7 20	10	3 05	Å	31 667	40 B33	35	30
Arocior - 1234		17 20	1220 1400 P78	50 750 61 62	10	3 OF	5	31 667	37 000	40	20
		.,									
Methyimercury	13 OF	13 15	26 92 1526 ·	92 3190	520	, ,	N	O SAMPLES C	OLLECTED IN THIS	REACH	

ł

1

1

1

FINAL

.

4-35

W92194F

TABLE 4-3 SUMMARY OF FISH ANALYTICAL REBULTS FILLET BY SPECIES NYANZA III REMEDIAL INVESTIGATION SUDBURY RIVER STUDY ugits WET WERGHT

PAGE 13

____ RESERVOIR 2 **RESERVOIR 1** LARGEMOUTH BASS YELLOW PERCH PARAMETER FREQUENCY AVERAGE AVERAGE MÁX MIN FREQUENCY AVERAGE AVERAGE MAX MIN DETECTED REPORTED DETECTED REPORTED Methylene Chloride Acetone 2 - Butanone Toluene Styrene **Total Xylenes** 0 OF 27 1 OF 20 2300.000 2300 Phenol 447.500 2300 0 O F 20 Bis(2-Chloroethyl) ether 0 OF 27 **Banzyl Alcohol** 0 OF 27 0 O F 20 2 - Methylphenol 27 0 OF 20 0 OF 4 - Methylphenol 27 0 O F 20 OOF Nitrobenzene 27 0 O F 20 0 OF Naphthalene 0 OF 27 0 O F 20 Diethylphthalate 0 OF 27 0 O F 20 Fluorene OOF 27 0 OF 20 0 OF 27 0 OF 20 Di-n-butylphthelate Butylbenzylphthelate 0 OF 27 0 OF 20 Bis(2-ethylhexyl)phthelete OOF 27 11 OF 17 1581.818 439.412 3300 600 27 0 OF 17 Di-n-octylphthalate 0 OF 27 0 OF 20 Banzo(b)fluoranthene 0 OF Benzo(a)pyrene 0 OF 27 0 OF 20 0 OF 27 0 OF 20 Heptachior 27 0 OF 20 Aldrin 0 OF Heptachlor Epoxide 27 0 OF 20 0 OF Endosullan 1 0 OF 27 0 OF 20 Dieldrin 0 OF 27 0 OF 19 4.4'-DDE 19 OF 27 25 763 20 500 68 5.5 9 OF 20 24.889 26.925 30 17 27 20 0 OF Endrin 0 OF 27 Endosullan II 0 OF 0 O F 20 4.4'-DDD 8 OF 27 3 625 6 963 7 0 O F 20 1 27 0 OF Endosullan Sullate 0 OF 20 4.4'-DDT 7 OF 27 1 143 6 389 2 0.5 0 OF 20 Methoxychlor 0 OF 27 0 OF 20 27 Endrin Ketone 0 OF 0 O F 20 alpha - Chlordane 3 OF 27 0 833 34,907 1 0.5 0 OF 20 gamma - Chiordane 4 OF 27 1 125 33 870 2 0.5 OOF 20 Arocior - 1248 0 OF 27 0 OF 20 Aroclor - 1254 27 83 519 350 20 6 OF 95 000 24 0 O F Aroclor - 1260 6 OF 27 60 500 76.593 93 16 0 OF 20 20 OF 21 1988 1898 57 4200 220 13 OF 13 685 685 1320 220 Methylmercury

FINAL

(

LE 4-3 SUMMARY OF FIGH ANALYTICAL REBULTS FILLET BY SPECIES NYANZA HI REMEDIAL INVESTIGATION SUDBURY RIVER STUDY ug/kg WET WEKGH? PAGE 14

		ħ	ESERVOIR 1					R	ESERVOIR 1		
		. 8	ULLHEAD					L.	ARGEMOUTH BASS		
PARAMETER	FREQUENCY	AVERAGE DETECTED	AVE RAGE REPORTED	MAX	MIN	FREQUEN	ICY	AVERAGE DETECTED	AVERAGE REPORTED	MAX	MIN
Methylene Chloride											
Acetone											
2 - Butanone											
Toluene											
Styrene											
iotal Xylenes											
	}		· ·· ····			↓					
Phenol	DOF 1	1				1 0F	30	2300	495	2300	230
Bis(2 - Chloroethyl) ether	0 OF 8					0 O F	30				
Benzyl Alcohol	0 OF 8					0 O F	30				
2-Methylphenol	0 OF 6	l i				0 O F	30				
4 - Methylphenol	0 OF E	1				0 O F	30				
Nitrobenzene	0 OF 8	1				0 OF	30				
Naphthalene	0.0F 8					0 O F	30				
Diethylphthalate	0 0 F 8)				0 O F	30				
Fluorene	0.01					0 O F	30				
Di-n-butylphthelete	0 OF 8					0 O F	30				
Butylbenzylphthalate	0 OF					DOF	30				
Bis (2 – ethylhexyl) phthelete	0 OF					11 OF	21	1581.818	1026.190	3300	60
Di-n-octylphthelate	0 OF					0 OF	20				
Benzo(b)fluoranthane	0.01						30				
Benzo(a)pyrene	0.01	•				0.05	30				
}	- ·-										
Heptachlor	0 OF 10					0 OF	30				
Aldrin	0 OF 10					0 OF	30				
Heptachlor Epoxide	0 OF 10)				0 OF	30				
Endosullan I	0 OF 10	2				0.0F	30				
Dieldrin	0.0F 10		48 300				30	04 800	13 067	20	
4,4'-DUE		14 3/3	13.300	32	3	906	30	24.009	13.007	30	14
Endrin						0.01	30				
Endosullan II			7.860	20	26		30				
		· · · · · · ·	1 0 3 0	20	3.3	0.0F	30				
	205 10	1000	4 450	•	1	0.0F	30				
Nethoryshine) /000	4.430	ŕ		0.0F	30				
	0.05 10	1				0 OF	30				
eloba - Chiotdana	8 OF 10) 1333	9 850	2.5	0.5	0 0F	30				
amma - Chlordene	5 OF 10) 1.000	15.600	1.5	0.5	0 OF	30				
Aroclor - 1248	0 OF 10)				0 OF	30				
Aroclor - 1254	3 OF 10	27 667	50.300	39	19	0 OF	30				
Aroclor - 1260	6 OF 10	35.500	40.800	69	15	0 OF	30				
Mathylmercury	1 OF	1 913	913	913	913	22 OF	23	1215	1166.96	3760	29

L.

4-37

FINAL

1

W92194F

TABLE 4–3 SUMMARY OF FISH ANALYTICAL REDULTS: FILLET BY SPECIES NYANZA III REMEDIAL INVESTIGATION SUDBURY RIVER STUDY Ug/kg WET WEIGHT PAGE 15

	7	BAXONVILLE IM	POUNDMENT			[SA	XONVILLE IMPOUN	DMENT	
	E DE OUTENOY	LARGEM	OUTH BASS		AAIN	ERECUEN	<u></u>	AVERAGE	YELLOW PERCI	H	
PARAMETER	THEODENCT	DETECTED	REPORTED		MIN	FREGUEN		DETECTED	REPORTED	MAA	MIN
Methviene Chloride	0 01	5				0 of	9				
Acetore	0 of	5				to 0	9				
2 - Butanone	0 01	5				0 of	9				
Toluene	0.01	5				0 of	9				
Styrene	0 of	5				0 of	9		•		
Total Xylenes	0 of	5				0 of	9				
Phenol	3 01	6 63 330	155.000	120	25	5 OF	17	33.600	315,471	48	2!
Bis(2 - Chloroethyl) ether	0 of	8				0 O F	18				
Benzyl Alcohol	0 01 1	8				0 O F	14				
2 - Methylphenol	0 of (6				0 O F	14				
4 – Methylphenol	0 of 4	8				0 O F	14				
Nitrobenzene	0 of	8				0 O F	18				
Naphth alene	1 01	8 1900 000	484 688	1900	1900	0 OF	18				
Diethylphthelate	0 01	6				0 OF	18				
Fluorene	0 01	8				0 OF	18				
Di-n-butyiphtheiste	1 01	8 62 000	237 125	62	62	1 OF	18	28.000	324.889	28	2
Butylbenzylphthalate) 0 of (6				1 0F	18	120.000	330.000	120	12
Bis(2-ethythexyl)phthelate	5 of	5 251 800	252 000	680	28	1 OF	8	58.000	116.625	58	5
Di-n-octylphthalate	l o of	5				0 O F	8				
Benzo(b)fluoranthene	l lo 0	8				0 OF	18				
Benzo(a)pyrene	0 of	•				0 O F	18				
Heptachlor	1 of	8 8 000	2 250	6	6	2 OF	20	3.750	1.800	6	1.
Aldrin	l 0 of	8				0 OF	20				
Heptachior Epoxide	0 of 1	1				1 OF	20	4.000	1.925	4	4
Endosulfan I	0 of					1 OF	20	1.500	1.800	1.5	1.
Dieldrin	l lo of l	•				4 OF	20	0.375	2.650	0.5	
4,4'-DDE	8 of	8 9 3 5 9	9 31 1	13	3.5	16 OF	20	6.684	5.723	13.6	:
Endrin	1 01	8 2 500	2.750	2.5	2.5	2 O F	20	4.000	3.550	7.5	0.5
Endosullan II	loof	8				0 O F	20				
4,4'-000	8 of	8 4 386	4 161	8 74	2	16 OF	20	5.173	4.539	14	
Endosullan Sullate	0.0	8				1 O F	20	1.000	3.500	1	
4,4'-DDT	2 0	■ 1 250	2.000	1.5	1	3 OF	20	1.167	3.025	2	0.5
Methoxychlor	l lo 0	8				0 OF	20				
Endrin Ketone	0 01	8				0 O F	20				
alpha Chlordane	0 of	8				1 0F	20	2.000	17.350	2	:
gamma – Chlordane	0 01	8				3 O F	20	0.670	14.350	1	0.9
Aroclor - 1248	0 01	8		_		0 O F	20				
Arociot - 1254	3 01	20 000	18 875	20	20	6 O F	17	25.000	21.176	40	1
Aroclor - 1260	5 of	8 90 400	83.775	110	77.9	3 O F	15 	49.700	33.940	65.1	
Methylmercury	7 of	6 520 43	645 36	1370	340	8 of	13	356.13	253.77	542	148

.....

- - -- -- -- --

4-38

FINAL

24-3 SUMMARY OF FISH ANALYTICAL REBULTS FILLET BY SPECIES NYANZA NI REMEDIAL INVESTIGATION SUDBURY RIVES STUDY ug/kg WET WEIGHT PAGE 10

	T.	5	AXONVILLE IMPOU	NDMENT				F	AIRHAVEN BAY		
			BULLHEAD					V	RGEMOUTH BASS		,
PARAMETER	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAX	MIN	FREQUEN	ICY	AVERAGE DETECTED	AVERAGE REPORTED	MAX	MIN
		LES COLLECTER	DIN THIS REACH			1					
Methylene Chloride						2 OF	19	7175.000	766.447	14000	350
Acetone						5 OF	19	1128.000	466.053	1600	940
2-Butanone						0 O F	19				
Toluene						1 O F	19	5.000	12.105	5	
Styrene						1 O F	19	2.000	11.947	2	:
Total Xylenes						1 OF	19	2.000	11.947	2	:
Phanol	0 OF 4					19 OF	19	1393.368	1393.368	8200	64
Bis(2-Chloroethyl) ether	0 OF 4					1 O F	19	40.000	117.368	40	40
Benzyl Alcohol	0 OF 4					1 OF	19	31.000	116.105	31	3
2-Methylphenol	0 OF 4					1 OF	19	33.000	116.211	33	33
4 - Methylphenol	0 OF 4					13 OF	19	202.846	176.947	1900	3;
Nitrobenzene	0 OF 4					1 OF	19	24.000	116.526	24	24
Naphthalane	0 OF 4					0 O F	19				
Diethylphthalate	0 OF 4					0 O F	19				
Fluorene	0 OF 4					0 O F	19				
Di-n-butylphthalate	0 OF 4					9 O F	19	41.111	84.211	110	2
Butylbenzylphthalate	0 OF 4					2 OF	19	202.000	129.158	370	34
Bis (2 - ethylhexyl) phthalate	0 OF 0					2 OF	19	32.500	111.316	37	28
Di-n-octylphthalate	0 OF 0					2 OF	19	45.500	112.684	46	45
Benzo(b)fluoranthene	0 OF 4					1 OF	19	87.000	118.789	87	87
Benzo(a)pyrene	0 OF 4	-				1 OF	19	26.000	115.579	26	26
Heptachlor	0 OF 4					0 OF	17				
Aldrin	0 OF 4					0 OF	17				
Heptachlor Epoxide	0 OF 4					0 O F	18				
Endosullan I	0 OF 4					0 O F	18				
Dieldrin	2 OF 4	0 000	3 250	0	0	0 O F	15				
4,4'-DDE	4 OF 4	12 000	12 000	20	8	19 OF	19	19.109	19.227	36	
Endrin	1 OF 4	7 000	6.375	7	7	0 O F	15				
Endosullan II	1 OF 4	0 000	4.625	0	0	0 OF	18				
4,4'-DDD	4 OF 4	8 250	6.500	8	5	0 O F	18		•		
Endosultan Sulfate	0 OF 4					0 OF	18				
4,4'-DDT	1 OF 4	2 000	5.125	2	2	0 O F	14				
Methoxychlor	1 OF 4	1 000	22.750	1	1	0 O F	18				
Endrin Ketone	0 OF 4					0 OF	18				
alpha – Chlordane	0 OF 4					0 O F	18				
gamma – Chiordane	3 OF 4	1 000	6.250	1	1	0 O F	18				
Aroclor - 1248	0 OF 4					0 O F	18				
Aroclor - 1254	3 OF 3	23 333	23.333	25	20	19 OF	19	223.421	218.205	460	90
Aroclor ~ 1260	4 OF 4	40 000	40 000	70	20	1 OF	18	30.000	15.833	30	30
Methylmercury	NO	SAMPLES COLLE	CTED IN THIS REAC	CH		19 OF	19	686.32	686.32	1200	430

W92194F

FINAL

1

TABLE 4-3 SUMMARY OF FISH ANALYTICAL REPULTS: FILLET BY SPECIES NYANZA HI REMEDIAL INVESTIGATION SUDBURY RIVER STUDY Ug/kg WET WEIGHT PAGE 17

	[FA	IRHAVEN BAY			
			YE	LLOW PERCH			
PARAMETER	FREQUEN	Ċ¥ 	AVERAGE DETECTED	AVERAGE REPORTED	MAX	MIN	
					20		
Methylene Chloride	100	10	26 000	15 200	20	20	
Acetone	100	10	276 000	209 500	200	260	
2 ~ Butanone Takana	200	10	275 000	110.130	300	230	
	100	10	20 000	13.030	20	20	
Styrene Total Xylenes	0 OF	10					
Phenol	10 OF	10	481.900	481.900		59	
Biar2 - Chloroethyll ether	0.0F	10					
Benzyl Alcohol	4 OF	10	136.250	128.500	420	26	
2 - Methylphenol	OOF	10					
4 - Methylphenol	● OF	10	61.000	66.900	100	26	
Nitrobenzene	0 OF	10					
Naphthalene	0 OF	10					
Diethylphthelete	1 OF	10	33 000	113 800	33	33	
Fluorene	0 OF	10					
Di-n-butylphthalate	5 OF	10	120 000	135 000	280	25	
Butylbenzylphthelate	5 OF	10	989 600	629 800	3000	12	
Bis(2 – ethylhexyl)phthalate	2 OF	10	32 500	104 500	37	28	
Di-n-octylphthelete	2 O F	10	36 000	105.700	43	29	
Benzo(b)fluoranthene	0 O F	10					
Benzo(a)pyrene	0 OF	10					
<u> </u>	ł						
Heptachlor	0 OF	10					
Aldrin	0 OF	10					
Heptachlor Epoxide	0 OF	10					
Endosullan I	0 OF	10					
Dieldrin	0 OF	10					
4,4'-DDE	10 OF	10	15 436	15.436	35	6	
Endrin	0 OF	10					
Endosullan II	0 O F	10					
4,4'-DDD	0 O F	10					
Endosulfan Sulfate	0 O F	10					
4,4'-BDT	0 O F	10					
Methoxychlor	0 O F	10					
Endrin Kelone	0 O F	10					
alpha Chlordane	0 O F	10					
gamma – Chlordane	0 O F	10					
Aroclor - 1248	0 OF	10					
Aroclor - 1254	● OF	10	174 256	158.250	330	72	
Aroclor - 1260	0 OF	10					
Methylmercury	7 OF	10	420	328 5	480	330	

FINAL

.

٠

W92194F

.

NYANZA III REMEDIAL INVESTIGATION SUDBURY RIVER STUDY

PAGE 18

	•				F	RESULT				
	F	ESERVOIR	1		RESERVOIR	2	9	SAXONVILLE MPOUNDMI	ENT	SUDBURY RESERVOIR
	BASS	PERCH	BULLH	BASS	PERCH	BULLH	BASS	PERCH	BULLH	BASS
ORGANIC COMPOUNDS. (ug/kg)										
ALPHA-BHC	1.50					1.00			0.50	
HEPTACHLOR						0.50		(5. 50	1
ALDRIN	0.50		1	1.50	1.00	0.50	•]	3.50	
HEPTACHLOR EPOXIDE		1		3.50	1.50	6.50		4.00	14.00	1
DIELDRIN	1 50	1 00		0.50	2.00	5.00	0.50		7.00	
4,4'-DDE (P,P'-DDE)	280.00	55.00	29.00	170.00	80 _. 00	64.00	35.00	24.00	240.00	74.00
ENDRIN		1		5.00						
ENDOSULFAN II (BETA)	18.00		1.50				0.50	10.00	2.50	10.00
4,4'-DDD (P,P'-DDD)	57 00	20.00	4.50	44.00	39.00	28.00	6.50	16.00	100.00	12.00
ENDOSULFAN SULFATE	[[0.50		1.50	4.50	{	2.50	2 00	15.00	0.50
		0.50	[1.50	7.00	2.00	2.50	3.00	15.00	0.50
		1			1.00	2.00	3.00	0.50		
	7.60	ľ		1.00	1.00	7.50	0.50	1.50	10.00	
GAMMA-CHLUHUANE /2	0.00	1	1	1.00	1.00	7.50	0.50	1.50	10.00	
ALPHA-OHLUNDANE /2	9.00			1.50	140.00	110.00			260.00	
PCB - 1254 (AROCLOH 1254)	760.00	120.00	00.03	700.00	150.00	210.00	90.00	50.00	630.00	110.00
- 1200 (ANOCLON 1200)	/00.00	120.00	00.00	100.00	100.00	210.00	00.00	00.00	000.00	110.00
MONOMETHYLMERCURY	652	399	102	2140	551	NS	598	58.3	NS	244
INORGANIC PARAMETERS (mg/kg)		1								
ARSENIC	0 0349								0.0318	
BARIUM	[]	2.48	1.15		1.68		0.5598			}
CALCIUM	27875.6	18454.9	7732.51	2258.044	17950.03	624.09	10494.51	3033.39	1242.82	909.35
COBALT	0 0644	0.0644	0.07		0.0753	0.031	0.0577	0.0345	0.0244	
IRON	160.21	163.84	230.15		106.33		126.72			
LEAD	} j)	0.397		0.31					j
MAGNESIUM	649.88	487.95	285.85	262.51	557.5948	200.85	482.88	353.95	160.17	276.89
MANGANESE	7.24	59.13	31.66	0.5554	11.65	0.5923	2.972	6.26	1.35	
MERCURY	3 809			3.4	1.9565	1.5238	1		0.9	
POTASSIUM	2103.94	2484 3	2179.633	3399.22	2948.73	3189.56	3650.63	2994.83	2273.83	3120.72
SODIUM	2042.78	1799 96	1589.62	1173 38	1704.72	1478.93	1451.82	1356.55	907.84	1210.41
ZINC	31.95	28 71	26.99	12	25.6	13.87	26.12	45.7	15.0423	12.21

W92194F

FINAL

Results of the monthly water quality sampling exhibited expected seasonal variation in the concentrations of various water quality components. Results are further discussed and summarized in Appendix L.

4.1.2 Summary Tables

Data are reported from the laboratory and presented in the summary tables in micrograms/kilogram (ug/kg), milligrams/kilograms (mg/kg) and dry weight for sediment samples. These units are closely equivalent to parts per billion (ppb) and parts per million (ppm), respectively. Water samples are reported in micrograms/liter (ug/L) and milligrams/liter (mg/l). Again, the units are nearly equivalent to ppb and ppm, respectively. Data will be discussed and presented on Study Area figures in units of ppb and ppm.

Summary Tables 4-1, 4-2, and 4-3 are presented to accompany the discussion presented in this section. Table 4-1 presents a statistical summary of the results from analyses of surface water and sediment samples. Table 4-2, presents a comparison of Water Quality Criteria with contaminant levels in the Study Area. Tables 4-3 and 4-4 present summaries of contaminant concentrations in fish tissue. Table 4-7 presents a summary of the Eastern Wetlands soil and sediment analyses.

Table 4-1 presents range, frequency, and averages of contaminants detected. This table is organized by Reach, showing the statistics for each section of the Study Area as described in Section 2.0.

Two average values are presented in Table 4-1 for each contaminant detected. Average reported values reflect an average of all the laboratory data presented in Appendices A through D. A value reported by data validation process followed by a "u" is the detection limit for the analysis. The actual concentration of the contaminant is expected to be less than the detection limit and also below lab quality control limits. According to EPA Region I policy, one half of the u-qualified value is calculated into the reported average concentration for use in human health risk assessment calculations.

When this process is performed, the calculated average concentrations for some parameters are higher than the maximum concentrations reported by the laboratory. These occurrences indicate a high detection limit in some samples, and a trace of the compound positively detected in others. As a result, the average value is skewed by the high detection limits attained by the laboratory.

To compensate for this, averages of samples with positive (above detection limits) detected contaminant concentrations are also

W92194F

FINAL

TABLE 4–4 SUMMARY OF FISH ANALYTICAL RESULTS: FILLET BY REACH NYANZA III REMEDIAL INVESTIGATION SUDBURY RIVER STUDY mg/kg WET WEIGHT

			·· — ··		SUDBURY RES	ERVOIR			CEDAR SWAM	P POND		
PARAMETER	FREQUE	NCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM	FREQUEN	ICY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM
ALUMINUM	3 OF	10 [·]	13.65	6.772	14.69	13.06	0 OF	17				
ANTIMONY	OOF	10					3 OF	17	0.420	0.568	0.820	0.190
ARSENIC	1 OF	10	0.06	0.096	0.06	0.06	0 OF	17				
BARIUM	0 OF 1	10					0 OF	17				
BERYLLIUM	OOF	10					0 OF	17				
CADMIUM	0 OF	10					0 OF	17				1
CALCIUM	10 OF	10	400.95	400.950	513.22	303.53	17 OF	17	2025.176	2025.176	8240.000	133.000
CHROMIUM	5 OF	10	1.89	1.114	2.84	1.33	10 OF	17	0.288	0.211	0.640	0.160
COBALT	6 OF	10	0.05	0.229	0.08	0.03	0 OF	17				
COPPER	OOF	10					0 OF	17				
IRON	2 OF	10	82.23	29.001	93.61	70.86	0 OF	17				ł
LEAD	0 OF	10					1 OF	14	1.170	0.120	1.170	1.170
MAGNESIUM	10 OF	10	379.13	379.128	519.28	264.71	17 OF	17	235.882	235.882	298.000	160.000
MANGANESE] 7 OF	10	1.45	1.062	2.43	0.57	0 OF	17				
MERCURY	5 OF	10	0.89	0.489	1.18	0.55	7 OF	17	1.770	0.732	9.600	0.200
NICKEL	0 OF	10					1 OF	17	6.000	0.494	6.000	6.000
POTASSIUM	10 OF	10	5432.44	5432.440	6446.94	3897.49	17 OF	17	3000.000	3000.000	3400.000	1940.000
SELENIUM	1 OF	10	1.80	0.460	1.80	1.80	14 OF	17	0.368	0.312	4.000	0.030
SILVER	0 OF	10					1 OF	17	2.200	0.164	2.200	2.200
SODIUM	10 OF	10	418.12	418.120	524.30	330.84	17 OF	17	486.706	486.706	984.000	336.000
THALLIUM	0 OF	10					1 OF	14	0.080	0.049	0.080	0.080
VANADIUM	6 OF	10	0 49	0.325	0.76	0.31	0 OF	17				
ZINC	10 OF	10	32 36	32.359	61.62	9.72	17.OF	17	9.965	9.965	56.800	3.400

W92194F

(

FINAL

TABLE 4-4 SUMMARY OF FISH ANALYTICAL REBULTS: FILLET BY REACH NYANZA III REMEDIAL INVEBTIGATION SUDBURY RIVER STUDY mg/kg WET WEIGHT PAGE 2

	······································	_	SOUTHMILLE P	OND			MILL POND					
PARAMETER	FREQUEN	CY	DETECTED AVERAGE	REPORTED AVERAGE	MAXIMUM	MINIMUM	FREQUEN	ICY	AVERAGE DETECTED	Average [•] Reported	MAXIMUM	MINIMUM
ALUMINUM		7'	·····				0 OF	- 30		L		
ANTIMONY	0 OF	7					0 OF	30				
ARSENIC	0 OF	7					0 OF	30				
BARIUM	0 OF	7					0 OF	30				
BERYLLIUM	0 OF	7					0 O F	30				
CADMIUM	0 OF	7					0 OF	30				1
CALCIUM	7 OF	7	5245.714	5245.714	12000.000	700.000	30 OF	30	140.051	140.051	391.445	87.162
CHROMIUM	5 OF	7	0.316	0.254	0.700	0.200	0 OF	30				
COBALT	1 OF	7	0.480	0 497	0.480	0.480	0 OF	30				
COPPER	2 OF	7	2.350	0.877	3.100	1.600	2 OF	30	0.370	0.091	0.404	0.340
IRON	1 OF	7	14.300	5.371	14.300	14.300	0 OF	30				
LEAD	1 OF	7	0.770	0.239	0.770	0.770	0 OF	30				
MAGNESIUM	7 OF	7	291.714	291.714	376.000	228.000	30 OF	30	242.723	242.723	269.305	213.920
MANGANESE	1 OF	7	0 380	0.629	0.380	0.380	23 OF	30	0.152	0.125	0.313	0.078
MERCURY	4 OF	7	0.548	0.315	0.890	0.280	3 OF	30	1.328	0.315	1.981	0.881
NICKEL	0 OF	7					0 OF	30				
POTASSIUM	7 OF	7	3288.571	3288.571	3660.000	2980.000	30 OF	30	4427.783	4427.783	33251.955	3070.579
SELENIUM	7 OF	7	0.097	0.097	0.150	0.060	0 OF	30				
SILVER	0 OF	7					1 OF	30	0.025	0.031	0.025	0.025
SODIUM	7 OF	7	606.857	606.857	704.000	466.000	30 OF	30	5 29.656	529.656	614.829	459,403
THALLIUM	0 OF	0					0 OF	30				
VANADIUM	0 OF	7					15 OF	30	0.035	0.267	0.046	0.021
ZINC	6 OF	7	14.000	9.307	21.400	8.600	30 OF	30	3.288	3.288	3.912	2.592

(

.

(

•

TABLE 4-4 SUMMARY OF FISH ANALYTICAL REBULTS FILLET BY REACH NYANZA III REMEDIAL INVESTIGATION SUDBURY RIVER STUDY mg/kg WET WEIGHT PAGE 3

[RES	ERVOI	12		RESERVOIR 1									
PARAMETER	FREQUEN	ICY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM	FREQUE	NCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM		
ALUMINUM	13 OF	57 [⊥]	19.708	5.396	57.127	12.420	12 OF	47	64.407	16.928	489.928	12.643		
ANTIMONY	1 OF	54	15.499	0.552	15.499	0.030	0 OF	50						
ARSENIC	5 OF	57	0 390	0 105	1.754	0.000	5 OF	59	0.119	0.080	0.114	0.030		
BARIUM	0 OF	57				1	1 OF	60	1.061	0.058	0.167	0.167		
BERYLLIUM	0 OF	57					0 OF	60						
CADMIUM	1 0F	57	0 052	0 030	0.052	0.052	0 OF	60						
CALCIUM	51 OF	57	264 320	250.532	2393.962	68.057	60 OF	60	403,120	403.120	4189.044	87.358		
CHROMIUM	13 OF	57	2 199	0 586	5.689	1.230	20 OF	60	2.420	1.099	8.318	1.174		
COBALT	18 OF	56	0 067	0 361	0.240	0.027	29 OF	59	0.103	0.297	0.532	0.031		
COPPER	7 OF	57	2 929	0 560	9.723	0.210	3 OF	60	4.787	0.570	6.313	0.040		
IRON	11 OF	57	102 996	22 220	267.448	63.450	25 OF	60	132.847	57.576	302.612	61.086		
LEAD	4 OF	57	0 406	0 048	0.529	0.290	1 OF	59	2.157	0.042	2.157	2.157		
MAGNESIUM	51 OF	57	261 232	238 997	424.628	186.331	60 OF	60	306.214	306.214	592.383	223.530		
MANGANESE	44 OF	57	1.053	0 826	13.752	0.075	55 OF	60	2.453	2.264	10.682	0.095		
MERCURY	50 OF	57	2 451	2 185	7.600	0.784	29 OF	60	1.367	0.746	4.185	0.696		
NICKEL	0 OF	57					0 OF	60						
POTASSIUM	51 OF	57	3913 867	3444 710	7109.957	3068.974	60 OF	60	4259. 839	4259.839	6335.704	2580.562		
SELENIUM	0 OF	57				ļ	0 OF	60						
SILVER	1 OF	54	0 444	0 060	0 444	0.444	0 OF	51						
SODIUM	51 OF	57	416 072	434.371	731.696	249.910	60 OF	60	501.721	501.721	860.792	33.482		
THALLIUM	1 0F	54	1.346	0.078	1.346	1.346	0.OF	51						
VANADIUM	19 OF	57	0 450	0.322	1.161	0.020	24 OF	60	0.591	0.497	1.572	0.310		
ZINC	55 OF	57	19 348	18.814	178.539	0.305	58 OF	60	32.442	31 .715	262.464	2.861		

1

FINAL

1

TABLE 4-4 SUMMARY OF FISH ANALYTICAL REBULTS: FILLET BY REACH NYANZA III REMEDIAL INVESTIGATION SUDBURY RIVER STUDY mg/kg WET WEIGHT PAGE 4

[SAXONMILLE I	MPOUNDMEN	† · · · · · · · · · · · · · · · · · · ·		FAIF	HAVEN	BAY		·····	
PARAMETER	FREQUEN	CY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM	FREQUEN	NCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM
ALUMINUM	7 OF	47	20.337	5.726	30.153	12.698	0 OF	30		·-·· I	I	
ANTIMONY	1 OF	47	0.140	0.284	0.140	0.040	0 OF	30				
ARSENIC	2 OF	47	0.044	0.040	0.046	0.000	0 OF	30				l
BARIUM	0 OF	47					1 OF	30	1.100	0.637	1.100	1.100
BERYLLIUM	0 OF	47					0 OF	30				1
	0 OF	47					0 OF	30				
CALCIUM	47 OF	47	2458.040	2458.040	19500.000	102.000	30 OF	30	5237.333	5237.333	28400.000	116.000
CHROMIUM	28 OF	47	0.916	0.653	4.278	0.140	22 OF	30	0.581	0.453	3.400	0.140
COBALT	16 OF	47	0.103	0.365	0.600	0.025	2 OF	30	0.820	0.355	1.260	0.000
COPPER	3 OF	47	3.960	0.364	6.100	1.670	10 OF	30	2.026	0.884	3,400	1.200
IRON	6 OF	47	113.913	19.633	155.911	66.900	3 OF	30	20.700	4.887	23.800	14.900
LEAD	1 OF	47	0.690	0.031	0.690	0.690	2 OF	30	3.195	0.276	5.600	0.790
MAGNESIUM	47 OF	47	304.566	304.566	511.977	189.000	30 OF	30	256.100	256.100	474.000	186.000
MANGANESE	23 OF	47	3.947	2.109	19.700	0.405	7 OF	30	11.189	2.859	33.200	0.320
MERCURY	35 OF	47	0.823	0.630	1.800	0.200	28 OF	30	1.129	1.054	3.200	0.200
NICKEL	2 OF	47	1.170	0.230	1.240	1,100	1 OF	30	2.000	0.320	2.000	2.000
POTASSIUM	47 OF	47	4065.429	4065.429	7862.154	2360.000	30 OF	30	2925.667	2925.667	3580.000	1890.000
SELENIUM	30 OF	47	0.159	0.164	0.620	0.030	27 OF	30	0.099	0.094	0.270	0.030
SILVER	1 OF	47	0.240	0.052	0.240	0.240	0 OF	30				i
SODIUM	47 OF	47	582.588	582.588	1160.000	195.639	30 OF	30	642.867	642.867	1100.000	482.000
THALLIUM	1 OF	47	1.190	0.055	1.190	1.190	0 OF	17				
VANADIUM	6 OF	47	0.768	0.427	1.218	0.330	0 OF	30				
ZINC	47 OF	47	18.018	18.018	94.793	1.690	29 OF	30	9.261	8.953	21.200	4.400

(

(

•

.

TÁ 4-4 IA 4-4 SUMMARY OF FISH ANALYTICAL RESULTS. FILLET BY REACH NYANZA III REMEDIAL INVESTIGATION SUDBURY RIVER STUDY US/KG WET WEIGHT PAGE 5

W92194F

			BUDBURY RESER	VOIR	CEDAR SWAMP POND					
PARAMETER	FREQUENCY	AVERAGE	AVERAGE REPORTED	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM
Methylene Chloride						0 of 2	·			
Acetone						0 of 2	!			
2 - Butanone						0 of 2				
Toluene						0 of 2	1			
Styrene						0 of 2	1			
Total Xylenes						0 of 2				
Phenol	1 OF 7	7 56 000	504 286	56 000	56 000					
Bis(2 - Chloroethvil ether	0 OF 11	1								
Benzyl Alcohol	0 OF 11					NO SAMP	IES COLLECT	ED IN THIS REACH	4	
2-Methylphenol	O OF 7	,				1				
4 - Methylphenol	0 OF 7	,								
Nitrobenzene	0 OF 11	1								
Naphthalene	0 OF 11	1								
Diethylphthalate	0 0 F 11	1								
Fluorene	1 0F 11	99 000	540 000	99.000	99.000	1				
Di-n-butylphthelete	1 OF 11	45 000	490.909	45 000	45.000					
Butylbanzylphthalate	0 OF 11	ł								
Bis(2 - ethylhexyl)phthalete	0 OF 6)				1				
Di-n-octylphthalate	0 OF 6	3								
Benzo(b)fluoranthene	0 OF 11	I				1				
Benzo(a)pyrene	0 OF 11	1								
Hentechlor	8 OF 7	,		······		1				
Aldrin	0.05									
Hentechlor Enovide	0.05					1				
Endosullan I	0.05					1				
Dietdrin	4.05	0 500	3 250	0 500	0.500					
4.4'~DDE	I OF	19 563	19 563	41 000	3 500	1				
Endrin	I OF	0 500	5.313	0.500	0.500	1				
Endosulfan II	O OF	1			•••••					
4.4'-DDD	7 OF	3 4 2 9	3.750	7.000	1.000					
Endosullan Sullate	2 OF 8	2.750	5 188	4 500	1.000	1				
4.4'-DDT	SOF 6	0.900	2.813	2 000	0.500					
Methoxychlor	0 OF	1	_							
Endrin Ketone	OOF	1				1				
alpha - Chlordane	1 OF 8	1 000	26.375	1 000	1.000	1				
gamma - Chlordane	0 OF	J				J				
Aroclor - 1248	0 OF	L				[
Aroclor - 1254	O OF 3	l i				1				
Aroclor - 1260	4 OF 8	87 500	83 750	95 000	20.000	ł				
	ļ									
Methylmercury	3 OF 3	485	485	652	221	0 OF 2				
L	1					l				

i.

. ____

FINAL

TABLE 4-4 SUMMARY OF FIGH ANALYTICAL RESULTS FILLET BY REACH NYANZA IN REMEDIAL INVESTIGATION SUDBLIRY RIVER STUDY UD/Kg WET WERDHT

PAGE

gamma - Chlordane

Araclor - 1248

Arocior - 1254

Aroclor - 1260

Methylmercury

1

0 OF

1 OF

0 OF

0 OF

4 OF

.....

12

12

12

12

12

500

646

72.5

215 60

500

730

لتر			MILL POND										
	PARAMETER	FREQUENCY	AVERAGE	AVERAGE REPORTED	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE DETECTED					
	Methylene Chloride												
	Acetone												
	2 ~ Butanone												
	Toluene												
	Styrene												
	Total Xylenes												
ł	0+												
	Phenol Bio (2 - Chloroothuft other						0.05 51	•					
	Bis(2 - Chloroethyl) emer	0.05 12					0.05 51						
	2 - Mathyiphanai	0.05 11					0.05 48	L					
	4 - Methylphenol	0 OF 11					0 OF 48						
	Nitrobenzene	0 OF 12					0 OF 51						
	Naphthelene	0 OF 12				1	0 OF 51						
4	Diethylphthalate	0 OF 12		I.			0 OF 51						
Å	Fluorene	0 OF 12	È				0 OF 51						
ö	Di-n-butylphthelate	0 OF 12	2				0 OF 51						
	Butylbenzylphthelate	0 OF 12	?				0 OF 51						
	Bis(2-ethylhexyl)phthelete	0 OF 12	1				0 OF 41						
	Di-n-octylphthelate	0 OF 12	2				0 OF 41						
	Banzo(b)fluoranthene	0 OF 12	2				0 OF 51						
	Benzo(a)pyrene	0 OF 11	2				0 OF 51						
	Hentechior	0.0F 12					0.OF 51						
	Aldrin	0.05 12					2 OF 51	0 750					
	Heptachior Epoxide	0 OF 12					0 OF 51						
	Endosullan I	0 OF 12					0 OF 51						
	Dieldrin	0 OF 12	2				2 OF 51	0.500					
	4,4'-DDE	2 OF 12	244.75	10.875	i 30	24	39 OF 51	20.846					
	Endrin	0 OF 12	2				0 OF 51						
	Endosulfan II	0 OF 12	2				1 OF 51	1.000					
	4,4'-DDD	0 OF 11	?				18 OF 51	J.969					
	Endosulfan Sulfate	0 OF 12	2				0 OF 51						
	4,4'-DDT	0 OF 12					13 OF 51	1.196					
	Methoxychior	0 OF 12					1 OF 51	3.500					
	Endrin Kelone	0 OF 12					101 51	2.500					
	alpha – Chiordane	0 OF 12	-				ι 60F 51	0.633					

FINAL

RESERVOIR NO. 2

REPORTED

0.324

0.608

18.245

0.725

7.549

6.952

3 598

0.755

33.137

36.755

91.770

76.459

1756.47

9 OF

0 OF

16 OF

14 OF

33 OF

500

370

51

51

50

49

34

1.056

129.875

42.143

1806.36

1

0.5

68

1

7

2

3.5

2.5

1

2

730

4200

93

0.5

0.25

0.5

1

1

0.05

3.5

2.5

0.5

0.5

10

10

220

MAXIMUM

MINIMUM

W92194F

E 4-4 SUMMARY OF FIBH ANALYTICAL REBULTD. FILLET BY REACH NYANZA IN REMEDIAL INVEBTIGATION SUDBURY RIVER STUDY UD/NG WET WEIGHT PAGE 7

~

			R	ESERVOIR NO. 1		SAXONVILLE IMPOUNDMENT						
PARAMETER	FREQUENC	Y AVE	RAGE	AVERAGE REPORTED	MAXIMUM	MINIMUM	FREQUE	NCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM
Methviene Chloride							0 OF	14				
Acetone							0 OF	14				
2 - Butanone							0 OF	14				
Toluene	1						0 OF	14				
Styrene	1						0 O F	14				
Total Xylanes							0 O F	14				
Phenol	205	58 91	MAD 000	A10 345	2100	2300	R ÓF	27	44 749	233.074	120	
Ris/2 - Chloroethyll ether	0.05	50 23		410.045	2000	2000	0.0F	30		200.0/4	120	21
Benzyl Alcohol	0.05	5.8					0.0F	26				
2 - Mathyloheool	0.01	50					A OF	24				
A - Methylphenol	0.05	50					0.0F	24				
Nitrobenzene	0.05	44					0.OF	30				
Nanhthaiana	0.05	58					1 OF	30	1900 000	129 250	1900	1000
Diethybhthelate	0.0F	58					0 OF	30		120.200	1500	1300
Elugrane	I D OF	58					0 OF	30				
Di-n-butvlobthelate	0.05	58					2 OF	30	45 000	258 167	62	28
Butylbenzyiphthalate	0 OF	58					1 0F	30	120 000	198.000	120	120
Bis (2 - ethylhesyll phthelete	22 OF	44 15	581.818	659 545	3300	600	6 O F	13	219 500	168 692	680	28
Di-n-octviphthelate	0.0F	43			••••		0 OF	13				
Banzo(b)fluoranthene	0 OF	58					DOF	30				
Benzo(a)ovrene	8 OF	58					0 OF	30				
							0 O F	0				
Heptachlor	0 OF	60					0 OF 3 OF	0 32	4,500	1 688	6	15
Aldrin	0 OF	60					0 OF	32			-	1.0
Heptachlor Epoxide	0.0F	60					1 OF	32	4.000	1 203	4	4
Endosulfan I	0 OF	60					1 0 F	32	1.500	1,125	1.5	15
Dieldrin	0 OF	59					4 OF	32	0.250	2.063	0.5	0.05
4.4'-DDE	26 OF	60	21 854	17.725	32	17	28 OF	32	8 208	7,404	20	3
Endrin	0 OF	60					4 OF	32	4.375	3,703	7.5	0.5
Endosullan It	OOF	60					0 O F	32			_	
4,4'-DDD	4 OF	60	11.375	1.308	20	3.5	28 OF	32	5.102	4.689	14	1
Endosullan Sullate	0 O F	60					1 OF	32	1.000	2.188	1	1
4,4'-DDT	2 OF	60	1.000	0.742	1	1	6 OF	32	1.334	3.031	2	0.5
Methoxychlor	0 OF	60					1 OF	32	1.000	2.844	1	1
Endrin Ketone	0 O F	60					0 O F	32				
alpha – Chiordane	● OF	60	1.333	1.642	2.5	0.5	1 OF	32	2.000	10.844	2	2
gamma – Chlordane	5 OF	60.	1.000	2.600	1.5	0.5	6 OF	32	0.835	10.000	1	0.5
Aroclar-1246	0 OF	60					0 O F	32				
Aroclor - 1254	3 OF	60	27 887	8.383	39	19	12 OF	28	23.333	20.179	40	15
Arocior - 1260	6 OF	60	35.500	6.800	69	15	12 OF	25	63.425	46.870	110	20
Methylmercury	36 OF	37 1	1015.22	99 0.75	3760	220	15 OF	12	526.13	402.95	1370	148

× m.-

W92194F

FINAL

TABLE 4 - 4 BUMMARY OF FIBH ANALYTICAL REBULTS FILLET BY REACH NYANZA IN REMEDIAL INVESTIGATION BUDBURY RIVER STUDY UD/Kg WET WEIGHT PAGE 0

		_	F	AIRHAVEN BAY			
PARAMETER	FREQUENCY		AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM	
Mathviene Chioride	3 OF	29	4792 887	507 397	14000	28	
Acetone	8 OF	29	1075 000	398 276	1600	810	
2 - Butanone	2 OF	29	275 000	40.741	300	250	
Toluene	2 OF	29	15 500	12.707	26	5	
Styrene	1 OF	29	2 000	7 828	2	2	
Total Xylenes	1 OF	29	2 000	7.628	2	2	
Phenol	29 OF	29	1079 069	1079 089	8200	59	
Bis(2-Chloroethyl) ether	1 OF	29	40 000	76 897	40	40	
Benzyl Alcohol	5 OF	29	115 200	120 379	420	26	
2 - Methylphenol	1 OF	29	33 000	76 136	33	33	
4 – Methylphenol	22 OF	29	144 818	139.000	1900	26	
Nitrobenzene	1 OF	29	24 000	76.345	24	24	
Naphthalene	0 OF	29					
Diethylphthalate	1 OF	29	33.000	39.241	33	33	
Fluorene	0 OF	29					
Di-n-butylphthalete	14 OF	29	69 286	101.724	280	21	
Butylbenzylphthelate	7 OF	29	784 571	301.793	3000	12	
Bis(2 - ethylhexyl)phinalate	4 0 F	29	32 500	106.966	37	28	
Di-n-octylphthelate	4 OF	29	40 750	110 276	46	29	
Benzo(b)fluoranthene	100	29	87 000	17.626	67	87	
Benzo (a) pyrene	TOP	29	28 000	/3/24	20		
Heptachlor	0 OF	27					
Aldrin	0 OF	27					
Heptachlor Epoxide	0 OF	28					
Endosullan I	0 OF	20					
Dieldrin	0 OF	25				_	
4,4'-DDE	29 OF	29	17 842	17.920	36	5	
Endrin	0.05	25					
Endosullan II	OOF	28					
4,4'-000	DOF	28					
	0.01	28					
4,4 - UUT Methawahire	0.01	24					
MutroxyChior Endris Keisse	0.05	20 28					
einhe - Chiordene	0.05	28					
	0.05	28					
	0.0F	28					
Arochr = 1254	28 OF	29	207 618	197 531	460	72	
Aroclor - 1280	1 OF	28	30 000	10.179	30	30	
Methylmercury	28 OF	29	814.82	562.93	1200	330	

(

FINAL

.

عه

4-50
presented in Table 4-1. These values are an average of those values listed in Appendices A through D which were either unqualified or qualified with a "j". Values qualified with a "j" indicate a positive detection, although the value is approximate.

This value is used throughout this Section to compare Reachspecific data to background concentrations. For brevity, this Average Detected Concentration is abbreviated to ADC.

4.2 <u>Potential Sources of Contamination</u>

Section 4.2.1 presents the Nyanza Site-related contaminants identified by EPA as being discharged to the environment at the Nyanza Site. Many potential point and non-point sources which could contribute a variety of contaminants to the river system also exist within the Study Area. These "off-site" sources are discussed in Section 4.2.2 and are grouped into landfills, potential oil/hazardous material release sites, wastewater discharges, and non-point sources.

4.2.1 Nyanza Site-Related Contaminants

The US EPA has compiled a list of contaminants which are known to exist in the soils or groundwater on the Nyanza Site and have been linked to on-site discharges. The list includes:

> trichloroethene 1,2-dichloroethene chlorobenzene nitrobenzene dichlorobenzenes 1,2,4-trichlorobenzene aniline naphthalene phenols benzidine antimony cadmium chromium arsenic lead mercury

These compounds are referred to in this RI report as "Site-related contaminants" which should not be confused with "Site Specific Contaminants", a group of aniline compounds specific to the dye manufacturing process. Usage of these aniline compounds has been documented at the Site, however, were not detected in Operable Unit III investigations.

_____ W92194F

4.2.2 Contamination Documented at the Nyanza Site

The following section summarizes reported contamination in media on the Nyarta Site (Operable Unit I) and in groundwater migrating from the site (Operable Unit II). This section summarizes background sections of the Draft Final RI Report, Nyanza II (Ebasco, 1991).

Contaminants which have been identified on the Nyanza Site include volatile organic compounds (VOCs), semi-volatile organic compounds (SVOCs), and metals in all media (groundwater, soils, surface water, and sediments).

Soil sampling results indicated that trichloroethene (TCE) and chlorobenzene, the primary VOC contaminants, are present in the 10^2 and 10^3 ppb range respectively. SVOC contamination was found to be limited in areal extent. Nitrobenzene, dichlorobenzenes (DCBs), 1,2,4-trichlorobenzene, aniline, and naphthalene were detected in the 10^2 to 10^4 ppb range. Chromium and lead were detected at levels up to 10^2 mg/kg and 10^3 ppm respectively.

Principal VOCs detected in onsite groundwater include TCE, 1,2dichloroethene (1,2-DCE), and chlorobenzene. Primary SVOCs detected in groundwater include nitrobenzene, aniline and DCBs. A wide variety of additional SVOCs including 1,2,4-trichlorobenzene, phenol, naphthalene and benzidine were also detected. Cadmium, chromium, arsenic, mercury and lead were also repeatedly detected in groundwater at the Site.

A plume(s) of organic compound contamination has migrated toward the north and east approximately one-half mile downgradient of the Site, to the vicinity of the Mill Pond/Raceway area of the River. Analine, DCBs, TCE and nitrobenzene have been detected in the plume(s) which occurs both in unconsolidated overburden and bedrock. No volatiles were detected in the surface water samples in Mill Pond. Volatiles in sediments included acetone, methylene chloride, and 2-butanone.

Mercury was detected in the seep water on the eastern flank of Megunko Hill, facing the Eastern Wetlands. Other heavy metals are not commonly detected in groundwater or surface water immediately off-site.

4.2.3 Off-Site Contaminant Sources

Many potential sources of contamination to the Sudbury River exist within the Nyanza III Study Area. This section provides an overview of potential sources of contamination to the Sudbury River other than the Nyanza site. This discussion is based on information collected from DEP and EPA files, and conversations

with municipal offices within the Sudbury River drainage basin. For purposes of this discussion, potential sources are grouped into landfills, potential oil/hazardous material release sites and wastewater discharges, and non-point sources. Research was not conducted to determine the historical contributors of contaminants to the River.

4.2.3.1 Landfills

A minimum of 12 active or inactive landfills are located within the Sudbury River drainage basin (Figure 4-1). Five waste disposal areas are located upstream from the Nyanza Site. Baystate Abrasives in Westborough operated seven dump sites which contain cyanide wastes and phenolic resins. These sites are currently undergoing environmental assessment by the Massachusetts DEP. Baystate Abrasives also operates a landfill for used abrasive materials west of Cedar Swamp. Two unlined, privately owned landfills have been operated by E.L. Harvey & Sons since the 1960s in the headwaters area of the Sudbury River. The eastern landfill is capped with silt, whereas the western landfill is uncapped, but a groundwater monitoring well network has been established. The former Ashland Landfill is also located upstream from the Nyanza site, adjacent to the Hopkinton Reservoir.

Seven landfills are located within the Sudbury River drainage basin downstream from the Nyanza Site. The Natick Landfill, located south of Lake Cochichuate, accepts residential and commercial solid The facility is unlined and has operated since 1946. wastes. Groundwater at the landfill has been monitored for the past eight The former Framingham Landfill was closed in 1985. years. The facility was capped and has a leachate collection and treatment system in place. An abandoned ash landfill is located west of Farm Pond in Framingham, within 300 yards of the Sudbury River. This landfill is unlined and is not monitored. Solid waste from Framingham, Southborough, and Ashland is currently hauled for incineration or landfilling elsewhere.

The Sudbury and Wayland landfills and the Joint Wayland-Sudbury septage treatment plant are located in wetlands west of the Sudbury River. The Wayland Landfill includes an unlined portion which was capped in 1988. A new, contiguous lined section of the landfill has a leachate collection system. A groundwater monitoring network has been established at the Wayland Landfill. The Sudbury Landfill is located directly south of the Wayland Landfill, and is monitored quarterly. Results of monitoring have indicated the presence of various organic and inorganic contaminants.

The Town of Concord operates a landfill located on the Sudbury River drainage divide near the confluence of the Sudbury River with the Concord River. The Concord Landfill is lined and has a



leachate treatment system. This facility is not believed to impact the Sudbury River.

4.2.3.2 <u>Potential Oil/Hazardous Material Release Sites and</u> <u>Wastewater Discharges</u>

Numerous potential oil/hazardous material release sites and wastewater discharges are located within the Sudbury River drainage basin (Figure 4-2). The release and disposal site locations were obtained from the DEQE lists of oil/hazardous material release sites and confirmed disposal sites. The wastewater discharge locations were obtained from the DEQE Division of Water Pollution Control reports (DEQE, 1987).

The release and disposal sites represent a wide range of potential sources, including service stations; dry cleaning establishments; electronics firms; manufacturers; electroplating shops; research and laboratory facilities; and spill sites. Potential contaminant sources are located throughout the drainage basin. The majority of potential release or disposal sites are located in the more densely populated and the commercial areas of Ashland and Framingham.

Two major wastewater dischargers are located downstream from the Nyanza site. The Marlborough East Wastewater Treatment Plant is located about six miles west of the Sudbury River on Hop Brook in Marlborough. Treated wastewater is discharged to Hop Brook, which flows into Wash Brook, and then to the Sudbury River at Route 20 in Wayland. The treatment plant has a design flow of 5.5 million gallons per day. A wastewater treatment plant operated by Raytheon discharges wastewater near Route 20 in Wayland. The wastewater consists of a combination of electroplating process wastewater (design flow of 25,000 gallons per day), sanitary wastewater (design flow of 65,000 gallons per day), and cooling tower and boiler blowdown water.

4.2.3.3 Non-Point Sources

Non-point source discharges represent a third category of potential contamination sources of the Sudbury River. Non-point source discharges are generally collected in developed areas by stormdrainage systems prior to discharge to the environment. These discharges represent overland flow from commercial, industrial, and residential areas. Stormwater from communities along the Sudbury River is discharged directly to the River or to natural water bodies that ultimately discharge to the River. Stormwater from urban areas may pick up wastes ranging from fly ash and dust from industrial processes, to organic debris and litter.

The highway network throughout the study area serves as a continuing source of additional vehicular and roadwash related



4~56

contaminants. Vehicular traffic contributes significant levels of lead, chromium, asbestos, slowly biodegradable petroleum products and other contaminants generated from processes such as tire wear which enter the environment as roadwash (Midwest Research Institute, 1975). Deposition rates of roadway surface contaminants have been shown to be highly correlated with total traffic (Biospherics, Inc., 1974), and would therefore be expected to be highest along the major roadways. Use of highway de-icing salts provides an additional potential source of sodium and chloride contamination to the Sudbury River.

Legally permitted direct wastewater discharges could also be significant source of contaminants in the River. In addition, it is likely that non-point runoff contributes heavy metals that are also Nyanza Site-related contaminants including lead, chromium and selenium, as well as polynuclear aromatic hydrocarbons and other organic compounds. In differentiating between Nyanza Site-related contamination and contaminants from other sources in the Study Area, the following should be noted:

- Mercury has been directly linked with the Nyanza Site; in addition, no other potential sources of this metal have ·been identified in the Study Area.
- o The highest concentrations and frequency of mercury contaminated samples have been noted in waterbodies nearest the Site (those in Reach 2). Concentrations and detection frequencies generally decrease through Reach 10.
- Nyanza Site-related contaminants may also be discharged to the Sudbury River by numerous potential point and nonpoint sources throughout the Study Area.

Lead, for example, may enter the Sudbury River from Nyanza Site runoff, a direct industrial wastewater discharge, stormwater runoff from roadways or in a landfill leachate. An extensive list of nonsite related contaminants were also detected in Sudbury River sediments throughout the Study Area. These contaminants include PAHs, DDT with associated degradation products, PCBs, volatile organic compounds and heavy metals.

4.2.3.4 Summary of Contaminants found in the Study Area

Not all chemical contaminants present in the river system are a result of discharges from the Nyanza Site. However, several conclusions can be drawn:

o Mercury, a major contaminant in fish, sediment, and surface water has been directly linked with the Nyanza

W92194F

4-57

Site; and no other source of this metal has been identified in the Study Area. Mercury will serve as the indicator of Nyanża-related contamination.

- o Methylmercury has been detected in sediments downstream of the site. While mercury is defined as a indicator, this compound can be similarly used.
- Chromium may be used as an indicator compound because it is a Nyanza Site-related contaminant which is not generally associated with most other point and non-point source discharges in the vicinity.
- Lead can be attributable to Nyanza activities, however, lead does not serve as an indicator compound because of the potential availability to the river system from other, non-point sources which cannot be accurately quantified.
- Polynuclear Aromatic Hydrocarbons cannot serve as indicator compounds because they were not found on Site and they are readily available to the river system from non-point sources.
- Organic compounds unique to dye manufacturing that have been identified at the Nyanza site cannot serve as "tracer" compounds for site-related contamination, since those compounds have not been detected in the Sudbury River.
- Volatile organic compounds are common contaminants associated with discharges and spills from fuel transfer stations, asphalt runoff, and various manufacturing and other light industry sources. These compounds degrade in the environment in a relatively short period of time, and therefore are not good indicator compounds for historic discharges.
- o Pesticides are not Site-related contaminants but are present in several sediment samples in apparently unrelated occurrences. Pesticides are significant in the ecological risk assessment due to the persistence and bioaccumulation potential of these compounds.
- PCBs are not Site-related discharge compounds and are present in only one River sediment sample and two raceway sediment samples. However, PCBs are significant in the ecological risk assessment for the same reason as pesticides.

W92194F

4-58

4.3 Results From Analysis of Data: Surface Water/Sediment

This Section discusses the distribution of contaminants defined by RI data collection activities. This section describes the contaminants found in sediment, surface water, and biota tissue, organized by Reach.

4.3.1 Background Sample Locations

Contaminant background values were calculated from analysis of samples collected from Reach 1, the Sudbury Reservoir and Reservoir 3. Water and sediment samples were collected in these areas. The entire length of Reach 1 is hydraulically upgradient of the Nyanza Site. Sample stations within the Reach were selected, to the extent possible, to be removed from cultural influence. The Sudbury Reservoir and Reservoir 3 were was also chosen as background locations. The Reservoirs are within the Sudbury River drainage basin, yet are removed from the influence of the Nyanza Site, as shown on Figure 2-A.

The background sediment database is comprised of nine sample locations, which include two vertical profile samples. The total number of samples, therefore, is 11, all of which were analyzed for inorganics and four for organics. As shown on Table 4-1, chromium was detected in nine of the 11 samples and the average detected concentration (ADC) of chromium sediments was 24.26 ppm. The highest detected concentration was 55.20 ppm. Mercury was detected in two of nine samples at an ADC of 1.05 ppm. The maximum detected concentration of mercury in background samples was 1.59 ppm. Low concentrations of several PAH compounds and acetone, a volatile organic compound (VOC), were also detected in the background sediment samples.

Sample locations and associated mercury and chromium results can be located on the Figure 2 series of oversized maps (Volume IV).

The background surface water database is comprised of seven samples, all of which were analyzed for inorganics and organics. Water data is presented in Table 4-1 and sample locations are shown on oversize Figure 3-A in Volume IV. Concentrations of both mercury and chromium were below detection limits for all samples. Lead concentrations in one surface water sample exceeds the MCL. Table 4-2 compares MCL and AWQC standards to detected averages of contaminants detected in surface water samples. The only organic contaminants detected were chloromethane (11 ppb) and 1,1 dichloroethene (3 ppb in two samples). Neither compound is pervasive in the Study Area surface waters.

4.3.2 Site Drainage System and Tributaries

This section provides a summary of data collected in the Eastern Wetlands, which is the direct receptor of Site surface runoff. This section will demonstrate that the Wetlands is a source area for mercury found in the Sudbury River and is a likely source area for some organic contaminants which were detected in sediments immediately down gradient of the wetlands.

4.3.2.1 Eastern Wetland

Ten surficial sediment samples were collected (Figure 2-D) for inorganic analysis and two sediment samples were collected for organic analysis. At these two locations, water samples were collected for both organic and inorganic analyses.

Sediment samples indicated a high frequency of detectable concentrations of mercury and chromium. The average detected concentration of mercury was 44.84 ppm and 136.79 ppm for chromium. Maximum concentrations were 152.0 ppm and 462.0 ppm for the is Aluminum also above background respective metals. Many organic compounds including 4,4' DDD (not concentrations. Site-related compound) are present in area sediments in significant concentrations. Site-related compounds include 1,2-DCE (ADC of 72.25 ppb), TCE (one detect-170 ppb), chlorobenzene (one detect -1,600 ppb), and 1,2-dichlorobenzene (1,2 DCB) (one detect - 7,200 ppb).

was detected in surface waters above background Mercury concentrations and above MCL and AWQCs. Mercury was detected in two samples at a maximum of 3.8 ppb and chromium was detected in one sample at 79.0 ppb which is above background concentrations. Other inorganics were present comparable to background concentrations. Site-related compounds, Trichloroethene, 1,2-DCE and 1,2 DCB were detected in water at concentrations less than 10 ppb.

4.3.2.2 <u>Chemical Brook Culvert</u>

Two sediment samples were collected from Chemical Brook culvert. The Culvert carries outfall from the Eastern Wetlands and Chemical Brook to the Outfall Creek, which is located north of Front street and 0.2 miles east of the downtown Front St./Main St. crossings on the north side of Ashland Center. No water samples were collected from the Culvert. See Figures 2-C and 2-D for sediment sample locations and analytical results.

A broad range of organics were detected, including many compounds seen in the Eastern Wetlands. Site-related compounds include 1,2-DCE (ADC of 25.0 ppb), TCE (one detect - 72 ppb), chlorobenzene

(ADC of 52.0 ppb) and 1,2-DCB (ADC of 1650). Nitrobenzene, a Siterelated compound, was detected at a concentration of 310 ppb. The only detected occurrences of Nitrobenzene in the Study Area was in the Culvert and the immediately downgradient Outfall Creek. Also of note is an occurrence of dibenzofuran (non Site-related compound) at a concentration of 92 ppb (SD3-201). Numerous PAHs were detected at concentrations over 1,000 ppb. These compounds are expected in this area, as this culvert serves as a stormwater drain for a large commercial/industrial parking area.

Inorganic contaminant concentrations in the Culvert are generally lower than those detected in the Eastern Wetlands. However, mercury and chromium were detected in both sediment samples at maximum concentrations of 7.1 ppm and 135 ppm, respectively.

4.3.2.3 Outfall Creek

Three sediment samples were collected. All were analyzed for inorganics and two for organics. One water sample was collected.

Elevated concentrations of mercury and chromium were noted in all sediment samples collected from this Creek. The ADC for mercury was 35.33 ppm with a maximum concentration of 99.20 ppm. The ADC for chromium was 341.8 ppm with a maximum concentration of 988.0 Aluminum iron present above ppm. and were background Volatile compounds detected in the Eastern concentrations. Wetlands were detected again in the Outfall Creek at concentrations less than 10 ppb. Site-related benzenes are present including the furthest downgradient occurrence of nitrobenzene (2 detects maximum of 500 ppb); 1,2-dichlorobenzene (2 detects - maximum of 290.0 ppb); and 1,2,4 trichlorobenzene (3 detects - maximum 1300 ppb). An extensive list of PAH compounds were detected in the 10' ppb range.

The water sample analysis detected lead at 5.7 ppb, which was above the MCL and the chronic AWQC level. Barium, cobalt and zinc were detected above MCL and AWQC levels and background concentrations. Site-related organic compounds including 1,2-DCE, TCE, and DCBs were present at concentrations less than 15 ppb. These organic compounds have been detected in upgradient areas tracing back to the Eastern Wetlands.

4.3.2.4 Raceway

Two sediment and one surface water samples were collected from the Raceway as a part of follow-up Phase II sampling. Samples were collected upgradient of the confluence of the Outfall Creek and the Raceway (Figure 2-C). At the time of sample collection, a bluecolored light sheen was noted on the surface of the water flowing

W92194F

out of the culverted section of the Raceway. No source for this sheen could be identified.

Two-depth sediment samples were collected in two locations in the Raceway. Mercury and chromium were present in elevated concentrations in all samples. The ADC for mercury was .71 ppm and the maximum concentration was .97 ppm? the ADC for chromium was 155.5 ppm and the maximum concentration was 208 ppm. Other metals including arsenic, beryllium, cadmium, cobalt, lead, nickel, and vanadium were also detected at concentrations above background levels.

Several Site-related organic contaminants were present in these samples: 1,2-dichloroethene was detected at an ADC of 736.67 ppb, trichloroethene had an ADC of 17,071.25 ppb, chlorobenzene had an ADC of 5,917.5 ppb and 1,2-dichlorobenzene had an ADC of 5,200 ppb. Non-Site-related volatile organic contaminants MEK, acetone and methylene chloride were also detected at lower concentrations. An extensive list of PAH compounds were detected in the 10³ to 10⁴ ppb range.

Pesticides 4,4'-DDD and 4,4'-DDE were detected in both sample locations at an ADC of 83.2 ppb and 18 ppb, respectively. A potential localized source of these compounds is located in Mill Pond, which is upgradient of these sample locations. Similar occurrences of pesticides were noted in the main branch of the River (Reach 2) and are discussed further in Section 4.3.3.12.

One PCB compound, Aroclor 1254, was detected in both upper and lower sample intervals of one sample. A positive detection of PCBs was found at one other sample location (near the Saxonville Impoundment) in the Study Area. The source of these contaminants is not known.

Site-related contaminants including 1,2-DCE, chlorobenzene, TCE, and mercury, generally were present in higher concentrations in the deeper interval samples. This suggests that contaminants in the Raceway could be the result of discharge of contaminated groundwater in this area.

Site-related inorganic contaminants were not detected in the surface water sample. Only barium was detected above background concentrations.

Two Site-related organic contaminants, 1,2-DCE and TCE, were detected in the Raceway. Both occurred at concentrations of 2 ppb.

4.3.2.5 <u>Cold Spring Brook</u>

Cold Spring Brook is a tributary of the Sudbury River. It flows into the River upgradient of Reservoir 2. Two sediment samples were collected from this area and analyzed for inorganics. One water sample was collected and analyzed for inorganics and organics.

Mercury and methylmercury were not detected in the sediments of Cold Spring Brook. Chromium was detected below background concentrations. Lead was detected in sediments at a maximum of 328 ppm.

The water sample indicated the presence of one phthalate compound bis(2-ethylhexyl)phthalate, at a concentration of 56 ppb. This compound is a common contaminant associated with plastics and its presence could be an artifact from sample collection procedures.

Chromium, lead, and mercury were not detected in the surface water sample collected from this location. Other inorganics were detected at concentrations comparable to background levels. Silver exceeded AWQC levels. MCL and AWQCs have not been established for other inorganics detected in the surface water sample.

4.3.2.6 <u>Heard Pond</u>

Heard Pond, located west of the Sudbury River in Reach 7, is a small sized eutrophic pond surrounded in most part by Cattail Marsh and Red Maple Swamp. This pond was considered a separate reach, since it is reported to be occasionally flooded by the Sudbury River.

One sediment sample was collected from Heard Pond and analyzed for inorganics. Mercury was detected at 3.5 ppm and chromium at 40.20 ppm, both above background concentrations. Other metals were detected at concentrations comparable to background levels.

4.3.3 Sudbury River

4.3.3.1 <u>Reach 1 (Includes Cedar Swamp Pond)</u>

Seven sediment samples were collected for inorganic analysis and three for organic analysis. Six surface water samples were collected for both inorganic and organic analyses. Reach 1 is a background Reach, and includes all of the background samples collected within the River. Since Reach 1 samples, in part, established background levels of contaminants, only positive detections of Site-related contaminants will be discussed.

Two sediment samples contained concentrations of mercury above

detectable limits, SD3-103 (1.6 ppm) and 104 (.5 ppm). Chromium was detected in concentrations of 6.6 ppm (SD3-209) to 55.2 ppm (SD3-208). These samples are less than 300 feet apart, indicating a wide variability in concentrations as a function of the sample point. Acetone was detected in one sample (SD3-152 at 10 ppb) and several PAH compounds were present. These compounds may have been introduced to the River through road run off.

Volatile compounds were detected in three water samples at concentrations below 11 ppb. None of the compounds persist into Reach 2.

No methylmercury was detected in background sediment samples.

4.3.3.2 <u>Reach 2 (Includes Mill Pond)</u>

Reach 2 is the potential receptor of the organic groundwater plume migrating from the Nyanza Site (Ebasco, 1991) and is the receptor, through the Outfall Creek and Raceway, of Site run-off from the Eastern Wetlands. Two distinct groupings of samples are present in Reach 2 with regard to Site-related contaminants; those above the confluence of the Raceway and the River are more characteristic of background samples; while those below this confluence generally contain Site-related contaminants above background levels.

Twelve sediment samples were collected for inorganic and thirteen for organic analyses. Seven water samples were collected for both inorganic and organic analyses.

Mercury and chromium concentrations in sediments increased sharply The mercury ADC was 6.81 ppm as compared to in this Reach. background levels of 1.04 ppm and the ADC for chromium was 36.32 as compared to the background concentration of 24.26 ppm. More significantly, maximum detected concentrations for mercury and chromium were 30.60 ppm and 216.0 (both in SD3-115), respectively, as compared to background maximums of 1.59 ppm and 55.22 ppm. Manganese was elevated above background concentrations, but other inorganics were comparable to background levels. Site-related organic contaminants which were detected in Reach 2 sediment include 1,4- and 1,2 - dichlorobenzene. Concentrations were between 440 and 1800 ppb.

A wide range of PAH compounds were detected in Reach 2, reflecting the commercial/industrial nature of the area. Concentrations were in the 10³ to 10⁴ ppb range, an order of magnitude greater than those found in the Outfall Creek, suggesting an input from an off-Site source. Dibenzofuran (2 detects) was found at a maximum concentration of 1060 ppb (SD3-204). This compound was also noted in the Chemical Brook Culvert.

W92194F

4-64

4,4'-DDD and 4,4'-DDE were detected in Mill Pond sediments through to the confluence of the River and the Raceway. The highest concentration of 1400 ppb occurred in sample SD3-156, in Mill Pond. Decreasing concentrations were noted in successive samples, suggesting a source in Mill Pond and down-river transport and subsequent dispersion.

Mercury was not detected in Reach 2 water samples, however chromium was detected in one sample at 7 ppb (SW3-130). Other inorganics were sharply elevated compared to background samples. The only organic contaminant detected was 1,2 DCE (one detect) at 1.0 ppb. Mercury and selenium were not detected in surface water samples, other inorganics with established MCLs and AWQCs were present in concentrations which exceeded these levels. These organics include Beryllium, Cadmium and Lead.

Three sediment samples were analyzed for methylmercury. A concentration of 312 ppb was detected in one sample; others were below detection limits.

4.3.3.3 <u>Reach 3 - Reservoir 2</u>

A total of 23 sediment samples were collected for inorganics and nine for organic analyses. Five water samples were analyzed for inorganics and organics.

The highest concentrations of mercury and chromium in the River sediments were detected in this Reach (Reservoir 2). The average detected concentration for mercury is 17.43 ppm and for chromium, 305.62 ppm. Maximum detected concentrations are 49.10 ppm (SD3-122) and 2620.00 ppm (SD3-117) respectively. Also, the frequency of detection was above 90 percent for both metals. Two occurrences of Site-related organic contaminants were noted in sediments; 1,4dichlorobenzene and 1,2,4-trichlorobenzene. The maximum concentration of both compounds was 180 ppb. Numerous PAH compounds occurred in the scattered sediment samples in the 10^3 to 10^4 ppb range.

DDD and DDE were detected in four scattered samples in concentrations up to 370 ppb.

The nature of the contamination in Reach 3 is similar to Reach 2, with the noted elevation of mercury and chromium concentrations.

Mercury and chromium were not detected in water samples collected from Reach 3. Other inorganics were detected at concentrations comparable to background with the exception of selenium, which was detected in one of five samples at a concentration of 19,300 ppb. This concentration exceeds MCL and AWQCs. There is no explanation for this occurrence of selenium. Barium, lead and zinc exceed

established AWQCs. One phthalate compound was noted at trace concentrations in one water sample.

Eighteen sediment samples were analyzed for methylmercury including nine shallow sediment samples and nine deep samples. Monomethylmercury was detected in one sample, at 26.3 ppb in the SD3-130 shallow interval sediment.

4.3.3.4 Reach 4 - Reservoir 1

Thirteen sediment samples were collected and analyzed for inorganics and two for organics. Two water samples were analyzed for both inorganic and organics.

Mercury and chromium concentrations in sediments decreased from Reach 3. The ADC for mercury was 3.69 ppm and for chromium 76.26 ppm. Maximum concentrations of both metals dropped significantly, but the frequency of detection (indicating a widespread distribution) remained high. Aluminum occurred in concentrations above background levels, but other inorganics were detected at concentrations comparable to those found in background. Several common organic compounds are present in several sediment samples. Acetone, methyl ethyl keytone (MEK) and benzoic acid were present No Site-related organic in relatively low concentrations. contaminants were detected in Reach 4 sediments. PAHs were detected in the range of 90 to 400 ppb, significantly less than concentrations found in Reaches 2 and 3. This Reach is characterized by a more residential/rural setting, with less contribution from road run-off and industrial activity.

Chromium was detected in one water sample at a concentration of 6 ppb (SW3-113). Mercury was not detected. Other inorganics concentrations were comparable to background levels. Lead exceeded MCL and AWQC levels. The only organic detected in one surface water sample was MEK, at a concentration of 10 ppb.

Seven sediment samples were analyzed for methylmercury which was detected in one deep interval sample at 79.3 ppb (SD3-132).

4.3.3.5 <u>Reach 5</u>

Six sediment samples were collected and analyzed for inorganics and one for organic compounds. Two water samples were analyzed for both inorganic and organic compounds.

Mercury and chromium concentrations dropped from those detected in upstream Reaches, however remained above background. The ADCs for the metals are 1.17 ppm and 30.62 ppm, respectively. Frequency of detects remained high. Concentrations of lead and silver are above background levels. Other inorganics were comparable to those found

in background. Scattered concentrations of PAHs and other semivolatile compounds were detected up to 1400 ppb.

Mercury and chromium were not detected in surface water and were not detected in downgradient Reaches 6 or 7, which was the downgradient extent of water sampling. Barium and arsenic concentrations exceeded established MCLs. No organics were detected in surface water samples.

Methylmercury was not detected in sediments collected in this Reach.

4.3.3.6 <u>Reach 6 - Saxonville Impoundment</u>

Twenty one sediment samples were collected for inorganics and four for organics analysis. Four water samples were collected for both inorganic and organic analyses.

Mercury and chromium concentrations in sediments increased within the Saxonville impoundment. ADCs were 5.00 ppm and 69.84 ppm for the respective metals. A trend emerged; these contaminants are concentrated in the low flow velocity reservoirs and impoundments. This trend will be discussed later in this Section and in Section 5.0.

Cobalt, cadmium (frequency 4 of 4) and magnesium are present in concentrations above background levels while other inorganics concentrations were comparable to background levels. Organic contamination was present at higher concentrations than upgradient Reaches, in this more industrialized area. Acetone, MEK, benzoic acid were present in concentrations up to 630 ppb. PAH compounds were widespread, at concentrations up to 3400 ppb.

DDD was present in one sample (SD3-138) at a concentration of 700 ppb.

Inorganics concentrations in surface water were comparable to background levels. Barium and lead exceeded MCL and AWQC levels. Only one organic compound, a pesticide, Lindane, was present at a concentration of 0.015 ppb in one sample.

Methylmercury was not detected in sediment samples.

4.3.3.7 <u>Reach 7</u>

Thirteen sediment samples were collected for inorganic and three for organic analyses. Four water samples were collected for both inorganic and organic analyses. Mercury and chromium concentrations in sediments dropped to ADCs of 2.08 ppm and 61.59 ppm, respectively. Frequency of detection was 62 percent for mercury and 77 percent for chromium, indicating a general decrease in the pervasive nature of contamination. Cadmium was present, again at levels above background concentrations. Acetone, MEK and benzoic acid were again detected in sediments up to 1600 ppb. The pervasive PAH compounds were present in concentrations similar to those found in the previous Reach. Α pesticide, gamma chlordane 2, and a PCB, Aroclor 1254, were present at low concentrations; each was detected in one sample. Other inorganics are comparable to background. Arsenic was present in relatively low concentrations, however a maximum of 40.60 ppm was detected in one sample.

Inorganic constituents in surface water were comparable to background levels. Barium and silver were detected in concentrations above MCL and AWQCs respectively. No organic compounds were detected in surface water.

Methylmercury was not detected in sediment samples.

4.3.3.8 Reach 8 - Great Meadows Natural Wildlife Refuge

Six sediment samples were analyzed for inorganic compounds. Sediment organic analyses were not conducted in Reaches 8 through 10, nor were surface waters collected in these Reaches.

Mercury and chromium concentrations again dropped. The ADC for mercury was 1.62 ppm; chromium fell below background concentrations. Cadmium was above background levels and arsenic was detected in one sample at a concentration of 30 ppm. Other inorganics were comparable to background levels.

4.3.3.9 <u>Reach 9 - Fairhaven Bay</u>

This Reach includes Fairhaven Bay. Although there is no impoundment on the Bay, this part of the River exhibits characteristics of an impoundment with the associated increase in mercury and chromium concentrations.

Three sediment samples were analyzed for inorganics. The ADC of mercury and chromium were 3.15 ppm and 50.27 ppm, respectively levels twice those found in the upgradient Reach 8. Frequency of detection also rose, indicative of the relatively more widespread nature of contamination in the Bay compared to adjacent River runs. Cadmium was again present in concentrations above background levels. Arsenic was also elevated above background due to a maximum concentration of 64.60 ppm in one sample of three. Other inorganics were found at levels comparable to background.

W92194F

4-68

4.3.3.10 <u>Reach 10</u>

This was the last Reach sampled. The downgradient end of this Reach is delineated by the confluence of the Sudbury and the Assabet Rivers.

Five sediment samples were collected and analyzed for inorganics. Mercury was detected in two of five samples with an ADC of 0.37 ppm and a maximum detection of 0.53 ppm. The background ADC for mercury was 1.05 ppm with a maximum detected concentration of 1.59 ppm. Concentrations of mercury in Reach 10 sediments were thus comparable to background levels. Chromium remained below background concentrations.

4.3.3.11 Bordering Wetlands

Ten samples were collected from six locations in the bordering wetlands within Reaches 2, 3, and 4. Four of these sample locations were in Reach 2 and one each in Reaches 3 and 4. At four of these sample locations (two in Reach 2, and Reaches 3 and 4), samples were collected from two depth intervals. All samples were analyzed for inorganic compounds, and two were analyzed for methylmercury.

The bordering wetlands samples were located above the normal River level, but within the potential seasonal flood zone. Analysis of these samples provides information regarding the presence of contaminants transported from the River bed sediments to the wetlands. This information is used in the risk assessment.

The following discussion presents a comparison of results from bordering wetlands samples and those from the open water areas of the associated reach.

Reach 2: Three wetland areas were sampled in this Reach: one at the confluence of the Outfall Creek and the Raceway (SD-113, 114); a second at Mill Pond (sd-115), and a third downgradient of the confluence (SD-159).

Mercury and chromium were present in sediments at concentrations comparable to background concentrations and below the averages for Reach 2. Other inorganics were also comparable to background concentrations. Organic analyses were not conducted on these samples.

The wetland at the Raceway confluence has been impacted by Site-related inorganic contaminants. Mercury and chromium concentrations in sample SD3-113 were 7.6 and 101 ppm respectively, significantly above detected background concentrations and above average detected concentrations for Reach 2. Mercury and chromium were below detectable limits in sample SD3-114. Concentrations of other inorganics were comparable to Reach 2.

Methylmercury was not detected in the bordering wetlands, although it was detected in one sample of three collected from Reach 2 (River) sediments.

Reach 3: Some heavy metals including copper (15.4 mg/kg), chromium (11.8 mg/kg), and cobalt (4.8 mg/kg) in the bordering wetlands samples (SD3-161) were at least one order of magnitude less than those found in Reservoir 2 sediments. Mercury, in particular, was detected at concentrations three orders of magnitude less than the average detected concentration in the reservoir.

Reach 4: Mercury was not detected in this bordering wetland, while Reservoir 1 samples showed an ADC of 3.69 ppm. Chromium was detected in the wetland at 15 ppm, compared to the reservoir ADC of 66.8 ppm. Nickel and zinc were both detected at one order of magnitude less than the average detected concentration for reservoir samples.

Four locations were sampled in discrete vertical intervals. Two upper interval mercury results did not pass data validation. Considering the remaining two samples, mercury concentrations decreased significantly with depth. Chromium showed no trend. Other heavy metal concentrations also decreased with depth.

4.3.3.12 <u>Pesticide Analyses</u>

Scattered occurrences of various pesticides, in particular 4,4'-DDT (DDT) and degradation products 4,4-DDD' (DDD) and 4,4-DDE' (DDE) were detected in the Study Area.

The pesticide DDT was detected in 13 sediment samples located in Reaches 2 through 6 and in the Eastern Wetlands as summarized below:

<u>Sample</u>	<u>Reach</u>	DDT	DDD	DDE	<u>Other</u>
SD3-156	2	1400 ug/kg			
SD3-157	2	115 ug/kg	33 ug/kg		
SD3-202	2	71 ug/kg		58 ug/kg	
SD3-204	2	105 ug/kg			
SD3-115	2		117 ug/kg		
SD3-106	EW		180 ug/kg		
SD3-118	3		53 ug/kg		
SD3-122	3		60 ug/kg		,
SD3-125	3		370 ug/kg		
SD3-129	3		130 ug/kg		
SD3-138	5		700 ug/kg		
SD3-148	7				18 ug/kg*
SD3-244	Race	way	104 ug/kg	14.8 ug/kg	15.5 ug/kg**
SD3-245	Race	way		21.2 ug/kg	

Gamma ChlordaneHeptachlor Epoxide

The highest concentration of DDT occurred near the outlet of the Mill Pond (SD3-156), at a concentration of 1400 ppb. Consecutive down River samples SD3-157, 202 and 204 contained decreasing concentrations of DDT and DDD, suggesting a release into Mill Pond and subsequent down River transport and dilution of contaminated sediment. Pesticides were also detected in the Raceway, which is also downgradient from Mill Pond.

Other downstream occurrences of pesticides were primarily DDD, with the highest concentration of 700 ppb occurring in sample SD3-138, located in the Oxbow Lake of Reach 5. One sample, SD3-148, contained 18 ppb of Gamma Chlordane. This sample was located downstream of the Saxonville Dam in Reach 7. These detections appear to occur randomly throughout the Study Area.

One occurrence of a pesticide in a surface water sample was located in Reach 7, in the vicinity of Heard Pond. The sample, SW3-117 contained Gamma-BHC, or Lindane, at a concentration of 0.015 ppb. Lindane was not detected in sediment samples.

Pesticides are discussed in more detail in Section 5.0 and in the risk assessments, where this group of chemicals is of greater significance.

4.3.3.13 Polychlorinated Biphenyls (PCBs) Analyses

One sediment sample, SD3-150, collected in Reach 7, contained 510 ppb of PCBs. There is no obvious source of this compound in the vicinity of this sample location. PCBs were also found in two sediment samples collected in the Raceway at 589 ppb and 111 ppb.

PCBs were not detected in any surface water samples. PCBs are discussed further in the risk assessments.

4.3.3.14 Inorganic Leachability Analysis

Two extraction procedure analyses were performed, Toxicity Characteristic Leaching Procedure (TCLP) and ACOE elutriate analysis. TCLP analysis is intended to identify those wastes which pose a hazard because of their potential to leach significant concentrations of specific toxic constituents. The procedure is used to evaluate leaching potential of hazardous materials in a worst case scenario, such as disposal in a landfill.

Two separate extraction procedures are used for TCLP. In this procedure, two extracts are made, one is analyzed for volatile organic compounds, while the second is analyzed for metals and other organics. The extraction procedure takes into account sample-specific parameters such as grain size, pH and percent solids to determine the proper strength of the extraction fluid.

Specific to this study, the method is a measure of the stability of inorganics adsorbed to sediment particles. This is determined by leaching the sediment sample with an aqueous solution with a pH of 5, and quantifying the concentration of inorganics which are leached.

River water with a pH of 5 would be characteristic of a stressed river system. Measured pH values in surface water were consistently in the six to seven range throughout the monthly water quality sampling events (see Appendix L). A pH of 5 was measured once in November 1989 at sample location SD3-110 during the Phase I sampling event. This measurement is considered an outlier because other measurements collected in the vicinity, SW3-109 and 111, indicated a neutral reading of pH 7.

Mercury was detected in the leachate at a frequency of 2 of 24 TCLP samples. Concentrations of total mercury to TCLP extract mercury are presented below:

<u>Sample f</u>	Total Mercury	TCLP
SD3-112	99.2 ppm	.29 ppm
SD3-115	30.6 ppm	.23 000

Relatively small amounts of mercury were leached in less than 10 percent of the samples analyzed. This analysis also detected levels of leachable lead on the average of 91.35 ppm (all reaches tested), and chromium at 35.39 ppm (Reaches 2, 3, 4). Other metals were also detected by this procedure, including arsenic (average of 8.77 ppm), cadmium (average of 6.62 ppm), copper (average of 21.5

W92194F

ppm) and silver (average of 4.8 ppm).

The results indicate that mercury and chromium are relatively stable within the River sediments and are not readily transferred to the aqueous environment under normal conditions. These results also indicate that dredged sediments would likely leach some metals in a landfill situation. This data will be further evaluated in the feasibility study process.

Mercury and chromium were also detected by ACOE elutriate analysis, which was performed on 27 shallow sediment samples during Phase I. Elutriate analysis of sediments is performed to determine the effect of turbulent river water on the associated sediments and the possibility of contaminant transfer from sediment to surface water. River water and sediment is agitated to simulate a dredging scenario. The concentrations of the analytes are measured in the water after agitation.

These results indicate that a disturbance of the sediments could cause relatively small amounts of mercury and chromium to transfer from the sediment to the surface water. These analyses resulted in maximum concentrations of 1.1 ppb of mercury at location SD3-110 and 504 ppb chromium at location SD3-150. These values exceed chronic ambient water quality criteria (2.0 ug/l, chromium; 0.012 ug/l mercury). These results will be further evaluated in the Feasibility Study.

4.3.3.15 Total and Dissolved Inorganics in Water

Several water samples were analyzed for total and dissolved metals to determine the speciation of contaminants between those being adsorbed onto particulates and those in the dissolved phase. This information will be utilized in determining contaminant transport mechanisms and is a consideration for the bioavailability of contaminants to biota.

The results indicate the presence of mercury and chromium in several water samples and the presence of these and other metals at low concentrations in the dissolved phase. Mercury was detected in three unfiltered sample locations between the Eastern Wetlands and the inlet of Reservoir 2. The highest concentration was 3.85 ppb (SW3-104). The only apparent contaminant distribution pattern is the occurrence of all samples in proximity to the Nyanza Site. Results are listed below:

Unfiltered Mercury	Filtered Mercury		
3.80 ppb	0.49 ppb		
0.37 ppb	0.37 ppb		
0.48 ppb	0.42 ppb		
	<u>Unfiltered Mercury</u> 3.80 ppb 0.37 ppb 0.48 ppb		

Chromium was detected in four surface water samples. Samples were located in the Eastern Wetlands (SW3-104), in the Outfall Creek (SW3-106), the inlet to Reservoir 2 (SW3-113), and in Reach 2, downstream from Mill Pond (SW3-130). Sample results are summarized below:

<u>Sample No.</u>	<u>Unfiltered</u> Chromium	Filtered Chromium
SW3-104	79.0 ppb	non-detect
SW3-106	5.6 ppb	non-detect
SW3-113	6.0 ppb	no analysis
SW3-130	7.0 ppb	non-detect

The sample results are widely scattered with a random distribution. The absence of chromium in the filtered samples indicates the chromium is probably adsorbed to suspended particulates. This will be discussed further in Section 5.0.

Metals often present in unfiltered samples but noted to be absent in their dissolved fraction include barium, beryllium, cadmium, copper, cobalt, iron and vanadium.

4.3.3.16 Analysis of Vertical Distribution of Contaminants

Thirty-seven sediment samples were collected from two depth intervals, zero to six inches and six to 12 inches at selected sample stations. Sample stations were chosen to obtain a representative indication of vertical distribution of contaminants in sediments throughout the Study Area.

Mercury and chromium concentrations increased with depth in several samples collected from Mill Pond. This is shown by SD3-154, 155, and 156 in Figure 2-C. The upstream Raceway sample (SD3-244) also exhibited this pattern. In samples collected downstream and in the reservoirs, this pattern was not observed.

Samples from the deep intervals were also analyzed for organic compounds. No organics were detected in many of the samples. One sample, SD3-156 (Mill Pond), showed concentrations of 1,4dichlorobenzene (1100 ppb) and 1,2-dichlorobenzene (580 ppb), in deep intervals as well as PAH compounds which were reported in corresponding shallow intervals. The dichlorobenzene concentrations cannot be compared to the data for the shallow interval, as the data for dichlorobenzene in the shallow interval was rejected during data validation processes.

In the Eastern Wetlands, mercury and chromium are often present in higher concentrations in the deeper interval such as samples SD3-107, 108 and 109. This is discussed further in Section 4.7.

W92194F

4-74

FINAL Four vertical profile samples were collected from Mill Pond in the

area where the Site related contaminated groundwater plume is projected to intersect Reach 2. One of the purposes of these samples was to determine if Site specific organics (aniline etc.) were present. These compounds were not detected in either sample.

4.4 <u>Results from Analysis of Data: Fish Tissue</u>

4.4.1 Summary Tables

Table 4-3 and 4-4 present frequency, average and maximum concentrations of contaminants in fish offal and fillet tissue detected during the Operable Unit III sampling work. These tables presents statistics for the Study Area as described in Section 2.0, and also present the data collected in the background area as a subset of the first group. The data are presented in wet weight.

Two average values are presented for each parameter in this table. Detected averages are calculated by averaging only the values positively detected in the samples. Reported averages were calculated by using half detection limits for results reported below detection limits.

4.4.2 Inorganic Contaminants

Most average concentrations of metals in fish tissue collected throughout the Study Area exceed by a factor of 2 the average concentrations calculated for background locations. Magnesium, copper, potassium, nickel, selenium, and sodium were exceptions, indicating that the concentrations of these metals in fish tissues downstream of the Nyanza Site are not elevated above background concentrations.

Mercury was detected in fish tissue samples from 0.1 ppm to 9.6 ppm. The detected average throughout the Study Area was 2.02 ppm. Since mercury is bioaccumulated, it tends to be found in higher concentrations in older predatory fish. The data generally support this rule, with some aberrations.

The maximum concentration of mercury (9.6 ppm) was detected in a two year old yellow perch collected from Cedar Swamp Pond (a Background location within Reach 1). This occurrence may be the result of a transcription or laboratory error, however this cannot be confirmed. Although this result seems unlikely to represent reliable data, no reason could be identified to reject this data.

This occurrence represents the difficulty posed by discussing individual data points; a discussion of data groups better represents distribution of contaminants.

Fish tissue samples contained levels of mercury in the 1 to 2 ppm range in areas downstream of the Raceway confluence. Upstream of this confluence, the detections were between 0.1 ppm and 1.0 ppm although the yellow perch discussed in the preceding paragraph raised the detected average to 1.19 ppm in Background areas. Levels of mercury in bass tended to be one order of magnitude higher than those in perch and bullhead. This may be a factor of age or feeding habits. As bass are a larger, more aggressive fish, they may feed on a higher trophic level than do perch, and actually will feed on perch. This exposes the bass to higher concentrations of bioaccumulated and bioconcentrated contaminants.

No obvious trends were exhibited by the other inorganic data in fish tissues. Other heavy metals were detected in fillet and offal samples as shown in Tables 4-3 and 4-4, including arsenic (maximum of 0.114 ppm); Colbalt (maximum of 1.26 ppm); lead (maximum of 5.6 ppm); Selenium (maximum of 4.0 ppm) and vanadium (maximum of 1.57 ppm). Antimony, cadmium, nickel, silver and thallium were detected in five or less of the 258 samples collected.

4.4.3 Organic Contaminants

As shown in Table 4-4, monomethylmercury was detected in 77 of 82 samples of fish tissue analyzed for this contaminant in ranges between 221 and 4200 ppb. Tables 4-3 and 4-4 show that ADCs for methlmercury in fish fillet are generally 50 percent of the ADCs for total mercury. Exceptions to this observation do exist, however; ADCs for these compounds were much closer in all species of fish collected from Reservoir 1, and the ADC for methylmercury actually exceeded the ADC for total mercury in yellow perch collected from Reservoir 2. This exceedence is possible because different individuals yellow perch were analyzed for each of the two parameters. Offal data shows a similar trend, but limited data points make any trend difficult to confirm.

A number of pesticides and PCB compounds were detected in fish tissues. Most significant was a repeated occurrence of PCBs Aroclor 1254 and 1260, in concentrations as high as 760 ppb. Also significant was a repeated occurrence of 4,4' DDE, 4,4' DDD and 4,4' DDT. DDE was detected in over 75 percent of fish tissue samples analyzed, and concentrations ranged from detection limits to 250 ppb. Other pesticides were detected less frequently, including chlordanes, dieldrin, endrins, and endosulfan compounds.

These compounds were detected in less than 50 percent of the samples analyzed, and were generally detected at concentrations less than 10 ppb.

Very few organic compounds were detected in background samples. Compounds that were detected were pesticides, a PCB compound, a

W92194F

phenol and a PAH compound. Of the pesticides, DDE, DDD, DDT and dieldrin were as frequently detected as were PCBs. These compounds were most frequently detected throughout the Study Area. Average concentrations of pesticides in the Study Area were less than twice those calculated for background.

Some of these pesticide compounds are bioaccumulated in the fatty tissues of fish, rather than the muscle. Viscera (offal) was analyzed separately from fillet meat in the Phase I collection activities, and as expected, some organics, notably DDE and PCBs (1260), were found at higher levels than in the viscera samples. Bioaccumulation is discussed at length in Section 7.0.

Other organic compounds detected in fish tissue include several volatiles, several phthalate compounds, phenols, and several PAH compounds. Most notable was a frequent detection of phenol at an average concentration of 825 ppb. Most other contaminants were detected in less than 5% of the total number of samples, and these compounds were generally not detected in background samples.

All data values specific to reach and species are presented in Tables 4-3 and 4-4.

4.5 <u>Benthic Survey Results</u>

Results of the sorting and preliminary identification of macroinvertebrates are presented in Table 4-5. Table 4-5 indicates the total number of specimens per group and the percent occurrence of the specimens sorted at each station. Only eight samples from the first 28 sampling stations (SD3-101 to 130/Reaches 1 to 3) contained at least 100 macroinvertebrates; four of these stations were background locations or reference stations. Overall, samples from stations SD3-107 to SD3-130 (Reaches 2 and 3) contained fewer organisms (especially insect larvae), which reflect greater environmental stresses, than did samples from stations SD3-131 to 150 (Reaches 4 to 7). Table 4-6 contains grain size, TOC and dissolved oxygen data collected at river locations for evaluation purposes.

Reference station SD3 101 was dominated by Chironomidae (midges). Their presence was probably indicative of oxygen deprivation in the deep water sediment of this location behind the Sudbury Reservoir dam. Reference stations SD3-103 and 104 (Reach 1) were upstream of the Site and reflected better water quality than the most severely impacted stations of SD3-111 to 116, which contained the fewest macroinvertebrates. Two of these stations (SD3-110 and 111) contained no macroinvertebrates. Station SD3-110 was located at the confluence of Chemical and Trolley Brooks. Sample SD3-106 was collected at a surface water location in the eastern portion of the Eastern Wetlands. Data from stations SD3-106 and 110 should not be

W92194F

4-77

BENTHIC SAMPLES SUMMARY DATA - PHASE I NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

FINAL

	·										_			
LOCATION	INSECTS	er.1	MIDGES	8 7	OTHERS	9(1		S •()			SEDIMEN		INTRATE	
	1100. /	74	(140. /	70 J	(110. 7	70,	(i to . /	7 4 j	SPECIMENT	00	M	5	G	W
SD3- 101	0/	a	104 /	100	0/	0	0/	٥	104	100 %	- «	_ •	~	~
SD3- 101 -DU	i o/	ō	100 /	95	0/	ō	5/	5	105	-%	- %	100 %	- 76	- 76
SD3- 102	3/	11	18/	67	4/	15	2/	7	27	10 %	- %	- %	90 %	- 70
SD3- 103	9/	26	9/	26	17 /	49	0/	0	35	75 %	- %	25 %	- 4	
SD3- 103-DU	8/	8	7/	7	83 /	83	2/	2	100	15 %	5%	80 %	- %	- %
SD3- 104	24 /	- 36	39/	59	3/	5	0/	0	66	60 %	- %	- %	40 %	- %
SD3- 105	4/	- 4	27 /	25	72 /	67	41	4	107	70 %	30 %	- %	- %	- %
SD3- 106	0/	0	7/	21	4/	12	221	67	33	20 %	80 %	- %	- %	- %
SD3- 106 - DU	0/	0	0/	0	0/	0	25 /	100	25	35 %	65 %	~ %	- %	- %
SD3- 110	0/	0	0/	0	0/		0/	0	0	-%	- %	50 %	50 %	- %
SU3- 111	0/	0	0/	0	0/		10/	~~~~	0	30 %	- %	- %	70 %	- %
SD3- 112 SD3- 115	21	20	0/	0	5/	71	0/	32	13	100 %	- 76	-%	- %	40 %
SD3- 115	21	17	0/	ŏ	4/	22	61	50	12	100 %	- %	- %	- %	. – %
SD3- 117	21/	20	39/	37	40/	38	5/	5	105	40 %	- 70	/3 76 60 M	25 %	- %
SD3- 118	6/	10	27 /	43	26/	41	4/	6	63	100 %	- 76	00 %	- %	- %
SD3- 119	0/	0	27/	25	71/	67	8/	8	106	100 %		- 70	~ %	- %
SD3- 120	6/	6	72/	72	13/	13	9/	9	100	100 %		- *	- 76	- 70
SD3- 121	0/	ō	78/	72	9/	8	21 /	19	106	100 %		- *	- 70	- 76
SD3- 122	2/	2	47 /	55	20 /	24	16 /	19	85	100 %	- %	- %	- %	
SD3- 123	0/	0	16 /	64	8/	32	1/	- 4	25	100 %	- %	- %	- %	- %
SD3- 124	2/	-14	6/	43	4/	29	2/	14	14	40 %	- %	60 %	- %	- %
SD3- 125	0/	0	25/	74	5/	15	4/	12	· 34	100 %	- %	- %	- %	- %
SD3- 126	1/	4	8/	35	0/	0	14 /	61	23	10 %	- %	- %	90 %	- %
SD3- 127	2/	2	67 /	83	0/	0	12 /	15	81	100 %	- %	- %	- %	- %
SD3- 128	0/	0	12/	63	5/	26	2/	11	19	90 %	10 %	- %	- %	- %
SD3- 129	21	4	2//	33	0/		2/	0	29	35 %	- %	70 %	- %	- %
SU3- 130	5 D/		01/	80	3/	•	3/	-	83	50 %	- %	50 %	- %	- %
503-131	15/	15	30/	27	50/	50		-	103	10 %	- %	10 %	- %	- %
SD3- 132	14/	14	58/	56	23/	20	£/ 8/		100	100 %	70 %	15 %	- %	- %
SD3- 134	13/	12	63/	59	31 /	291	0/	ŏ	107	90 %	10 %	- 76	- *	- %
SD3- 135	15/	15	59/	58	27/	27	0/	ō	101	60 %	40 %	- 78	- 7	
SD3- 136	10 /	10	3/	3	76 /	75	13/	13	102	100 %	- %	- %	- 76	
SD3- 137	6/	4	421	30	92/	66	0/	0	140	100 %	- 5	- %	- *	
SD3- 138	41 /	40	21 /	21	40 /	39	0/	0	102	50 %	50 %	- %	- ~	
SD3- 139	7/	7 !	3/	3	94 /	89	2/	2	1 1 06 1	65 %	35 %	- %	- %	- 5
SD3- 139-DU	: 6/	5	1 5/	4	99/	88	2/	2	112	60 %	40 %	- %	- %	- 5
SD3- 140	24 /	23	17/	16	60 /	58	3/	3	104 i	100 %	- %	- %	- %	- %
SD3- 141	5/	5	12/	12	78 /	76	7/	7	102	90 %	10 %	- %	- %	- %
SD3- 142	4/	4	20/	19	80 /	77	0/	0	104	40 %	60 %	- %	- %	- %
SD3- 143	14 /	13	: 5/	5	85 /	82	0/	0	l 104 [.]	70 %	30 %	- %	- %	- %
SD3- 144	. 11/	10	62/	59	28 /	27	4/	4	1 05 :	25 %	75 %	- %	- %	- %
SD3- 144 - DU	1 7/	6	78/	72	20/	18	4/	4	109	30 %	70 %	- %	- %	- % i
SD3- 145	5/	5	14/	13	64/	π	6/	6	109	25 %	75 %	- %	- %	- %
503-146	8/	7		61	32/	27	5/ 20/	5	118	100 %	- %	- %	- %	- %
SU3- 14/	5/	10		46	i 13/ . 34/	30	/ قنه بعر	40	i 50 -	100 %	- %	- %	- %	- %
5UJ- 146	1//	17	40/	40		47 60	10/	13	101	100 %	- %	- %	- %	- %
	10/	17 24	12/	10	. /J/ , 24/	- 09) - 25 i	3/ 12/	12	100 67	40 40	35 %	- %	- %	- %
303- 130	ا دری	541		50	64/		161	12		100 %	- 76	- 76	- %	- % (

OTHERS = AMPHIPODA (SCUDS - SIDE SWIMMERS), GASTROPODA (SNAILS), PELECYPODA (CLAMS)

OD = ORGANIC DETRITUS

M = AQUATIC MACROPHYTES

S ≃ SAND

G = GRAVEL

W = WOOD FRAGMENTS/STICKS

TAB	L E	4-6
-----	-----	-----

GRAIN SIZE, TOC, DISSOLVED OXYGEN IN SEDIMENT NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

SAMPLE IDENTIFICATION TRAFFIC REPORT NUMBER OF DESIGNATION	503- 102-0101 5041A-114	503+106+0101 5041A+91	SD3-106-0101 5041A-91 LAB_DUPL/CATE	SD 3 - 106 - 0101 - DU 5041A - 92 DUPLICATE	SD3-110-0101 5041A-97	SD3-112-0101 5041A-39	SD3-115-0101 5041A-42
DATE SAMPLED	11/30/89	11/27/89	11/27/89	11/27/89	11/27/89	11/09/89	11/09/89
X FINER THAN							
1 1/2"		••	NA	100	••		NA
1	••	••		98	100	100	(ORGANIC)
3/4"	••	••		98	99	91	
1/2"	••	100		98	97	55	
3/8"	100	99		98	95	40	
#4	97	99		97	88	17	
#8	95	98		96	81	4	
#10	95	97		96	79	2	
#16	93	96		96	73	1.6	
#30	89	86		94	59	1.1	
#40	87	85		92	50	0.9	
#50	85	81		90	42	0.7	
#80	83	76		84	28	0.5	
#100	62	75		83	25	0.4	
#200	74.8	73.0		75.2	15.8	0.3	
0.037 mm					4.0	0.1	
0.036 mm							
0.035 mm							
0.034 mm	32.6	37.8					
0.035 mm				44.8			
0.031 mm							
0.023 mm	11.4				4.0	0.1	
0.022 mm		29 6		33.3		•••	
0.021 mm							
0.014 mm						0.1	
		21 6		74 4	4.0	0.1	
	V.U			20.4	4.0	0.1	
		20 4		22.0	4.0	0.1	
	0.U K 7	15 1		16 1	3.2	0.0	
	3.7			10.1	J.E	0.0	
				11 C	1 2	0.0	
	2.7	Y.2		(),) E T	3.2	0.0	
	1.7	2.1		2.1	3.6	0.0	
TOC (mg/kg)	HA	9500		10500	5700	27400	12100
Dissolved Oxygen (ms/L)	13	NR		MR	9	10	10

NA - Not Analyzed NR - Not Recorded

FINAL

1

W92194F

.

(

W92194F

TABLE 4-6 GRAIN SIZE, TOC, DISSOLVED OXYGEN IN SEDIMENT NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

PAGE 2

SAMPLE IDENTIFICATION TRAFFIC REPORT NUMBER OC DESIGNATION	\$03-117-0101 5041A-22	SD3-118-0101 5041a-12	SD3+119-0101 5041A+14	SD3-120-0101 S041A-10	SD3-120-0101 5041A-10	SD3-122-0101 5041A-23	SD3 - 124 - 0101 504 1A - 09
DATE SAMPLED	11/06/90	11/02/89	11/03/89	11/02/89	11/02/89	11/06/90	11/01/89
X FINER THAN							
1 1/2"	100	••	••	••	••	••	
1"	89	••	• •	••	••		••
3/4"	89	• •	••	••	••	• •	••
1/2"	79	••		••	••		100
3/8"	76	••	••	••	••		98
#4	68	••	••	•-	••	••	84
#8	59	• -	••	•-			70
#10	57	100	100	••	••	100	67
#16	52	9 9	99	100	••	99	56
#30	47	96	97	99	100	99	44
#40	43	92	96	98	99	98	38
#50	41	87	94	97	98	98	33
#80	37	72	87	92	94	98	23
#100	35	68	83	88	90	97	20
#200	25,2	53.3	60.3	65.4	68.1	92.9	9.2
0.037 mm							3.4
0.036 mm			20.1				
0.035 mm	12.0	25.6		25.7			
0.034 mm					31.3		
0.033 mm							
0.031 mm						60.5	
0.023 mm	10.3		15.9				3.4
0.022 mm		21.5		21.6	25.3		
0.021 mm						48.2	
0.014 mm							
0.013 mm	6.9	16.4	12.7	18.5	17.2	32.8	3.4
0.010 mm							3.4
0.009 mm	5.2	15.4	8.5	14.4	15.2	23.6	
0.007 mm	4.0	12.3	6.3	12.3	7.1	17.4	2.7
0.006 mm							
0.003 mm	2.3	7.2	4.2	8.2	4.0	8.2	2.0
0.001 mm	1.7	4.1	3.2	5.1	0.0	4.1	1.3
TOC (mg/kg)	14400	2700	27000	28600	28600	15900	28500
UISSULVED UXYDEN (MD/L)	C I	7	R.C.	10	10	NK	У

(

NA - Not Analyzed NR - Not Recorded

(

FINAL

ĺ

.

SAMPLE IDENTIFICATION TRAFFIC REPORT NUMBER OC DESIGNATION	503-124-0101-00 5041A-09 LAB_DUPLICATE	SD 3- 125-0101 5041A-25	SD3+129-0101 5041A-33	SD3 - 129 - 0101 - DU 504 1a - 34	SD3-130-0101 5041A-35	SD3-132-0101 5041A-109	SD3-135-0101 5041A×79
DATE SAMPLED	11/01/89	11/07/89	11/08/89	11/08/89	11/08/89	11/29/89	11/16/89
% FINER THAN							
1 1/2"		••	••	••	••	••	
1"	· •	••	· ·	• •	• •	• •	••
3/4"	100	• •	••	••	100	••	••
1/2"	96	• •	••	••	97		••
3/8"	94	100	••		95		••
#4	83	99	••		87	100	••
#8	71	98		••	78	99	
#10	68	97	100	100	77	99	100
#16	55	97	95	96	72	98	98
#30	33	93	88	88	62	92	96
#40	31	92	85	86	55	85	96
#50	22	01	84	85	46	76	94
#80	16	87	81	82	20	55	01
#100	13	86	80	81	26	49	RQ
#200	9.4	72 2	71.0	74.6	17 7	31 4	78 3
0 037 mm	2.0				7 7	21.4	10.5
0.036 mm	2.0				1.1	10.0	
0.035 mm						10.9	24.5
0.034 mm		30 1					24.5
0.033 mm		30.1	44.7	/ A 7			
				40.7			
0.021 mm	3.0						14 7
	č. U	36 0			4.0	0.9	10.3
		23.4	17.0				
			37.9	42.3			
0.014 mm	2.0		•• •				
0.015 mm		20.8	31.5	51.1	4.6	6.9	10.2
0.010 mm	2.0				4.6		
0.009 mm	•	16.6	27.3	26.9		6.9	10.2
0.007 mm	2.0	13.5			2.3	5.0	6.1
0.006 mm	•		20.0	20.7			
0.003 mm	2.0	8.3	11.6	12.4	1.5	4.0	6.1
0.001 mm	2.0	4.2	13.7 R	4.1	0.0	3.0	1.0
TOC (mg/kg)	28500	47600	17000	15200	13400	4700	620 0 0
Dissolved Oxygen (mg/L)	Ŷ	9.5	9	9	9.4	12	NR

NA - Not Analyzed NR - Not Recorded

4-81

FINAL

Ĺ

TABLE 4.6 TABLE 4-0 GRAIN SIZE, TOC, DISSOLVED DAVGEN IN SEDIMENT NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACH/SETTS

W92194F

4-82

PAGE 4

SAMPLE IDENTIFICATION TRAFFIC REPORT NUMBER	SD3 - 136 - 0101 504 1A - 104	SD 3 - 138 - 0101 504 1A - 106	SD3-139-0101 5041A-65	503-139-0101-DU 5041A-66	SD3-140-0101 5041A-69	SD3-146-0101 5041A-53	SD3-148-0101 5041a-50
DATE SAMPLED	11/28/89	11/28/89	11/15/89	11/15/89	11/15/89	11/13/89	11/13/89
% FINER THAN							
1 1/2"	••	• •	NA	NA	••		. .
1"	••	••	(ORGANIE)	(ORGANIC)			
3/4"	· •	100				••	• -
1/2"		99			100		
3/8"	••	98			99	••	••
N 4	100	98			98		
#8	99	96			98		100
#10	98	96			98		99
#16	98	96			96	100	98
#30	96	94			93	99	96
#40	93	92			90	97	95
#50	90	88			87	96	92
#8 0	74	82			81	93	77
#100	67	79			77	93	69
#200	45.3	68.6			57.6	83.2	34.9
0.037 mm							
0.036 mm	19.4				19.8		13.9
0.035 mm							
0.034 mm							
0.033 mm		47.3					
0.031 mm						61.3	
0.023 mm	17.3				15.8		7.9
0.022 mm							
0.021 mm		40.7				44.4	
0.014 mm							
0.013 mm	15.3	29.7			9.9	33.8	79
0.010 mm							6.0
0.009 mm	15.3	26.4			9.9	29.6	
0.007 mm	12.2	19.8			4.9	2710	4.0
0.006 mm						25. 3	
0.00 3 mm	6.1	9.9			4.0	16.9	3.0
0.001 mm	5.1	6.6			1.0	7.4	0.0
TOC (mg/kg)	5600	16892	13600	8100	8200	9650	24500
Dissolved Oxygen (mg/L)	14	11	10	10	10	10	12

(

NA - Not Analyzed NR - Not Recorded

(

(

.

(

TABLE 4-6 GRAIN SIZE, TOC, DISSOLVED OXYGEN IN SEDIMENT NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS PAGE 5

W92194F

4-83

SAMPLE IDENTIFICATION TRAFFIC REPORT MUMBER	503-149-0101 5041A-57	SD3 - 150 - 0101 504 1A - 60
QC DESIGNATION		
DATE SAMPLED	11/14/89	11/14/89
X FINER THAN		
1 1/2"	NA	••
1	(ORGANIC)	100
3/4"		Y 5
1/2"		92
3/8"		89
#4 #0		64
#0		80
# {U 414		au 74
#10 #10		70
#30		70
#40		00 50
#80) y
#100		11
#200		10 5
0.037		17.J & 4
0.036 mm		9.4
0.035 m		
0.034 mm		
0.033 mm		
0.031 mm		
0.023 mm		A 4
0.022 mm		4.4
0.021 mm		
0.014 mm		
0.013 mm		4.8
0.010 mm		4.8
0.009 mm		
0.007 mm		3.2
0.006 mm		
0.003 mm		3.2
0.001 mm		0.0
TOC (mg/kg)	20000	16700
Dissolved Oxygen (mg/kg)	7	10

1

NA - Not Analyzed NR - Not Recorded

1

compared with data collected from River stations (these stations are not in a tributary of the Sudbury River) and should be used for reference only.

Station SD3-117, which was strongly influenced by the relative high water quality discharge of Cold Spring Brook, contained a relatively balanced macroinvertebrate community and was not as environmentally stressed as the surrounding stations.

At least 100 macroinvertebrates were found in each sample from stations SD3-131 to 150 (Reaches 4 to 7) with the exception of station SD3-147, where only 50 macroinvertebrates were present in the sample. Despite the lower number of organisms, the percent occurrence of insects at SD3-147 was similar to that of adjacent stations. All stations from stations SD3-131 to 150 contained populations of aquatic insects, Chironomidae (Diptera-midge larvae), and "others" which were composed primarily of clams (Pelecypoda), snails (Gastropoda), and scuds (Amphipoda). Aquatic worms (Oligochaeta) were dominant organisms at seven stations and absent from 11 stations.

Station SD3-138, which is located in the Oxbow Lake (Reach 5), contained the highest percentage (40 percent) of aquatic insects observed in samples from the survey. Composition of aquatic insects at stations SD3-131 to 150 ranged from 4 percent to 24 percent. This indicates the presence of stressed benthic habitats; however, the conditions were improved compared to those found at stations SD3-106 to 130 (Reach 2 and 3).

The stations in Reservoir No. 2 and areas upstream (Reaches 2 and 3), especially stations SD3-106 to 116, showed signs of the highest degrees of environmental stress. The areas with lower degrees of environmental stress (as evidenced by the overall higher occurrences of aquatic insect larvae) were at Reservoir No. 1 (stations SD3-132 to 135/Reach 4) and stations SD3-147 to 150 (Reach 7).

4.5.1 Grain Size/TOC

The results of the benthic sample analysis were compared with grain size and TOC. Grain size of the sediment affects the diversity of the organisms present in the sample zone due to the habitat available. However, the benthic samples were analyzed in a semiquantitative manner only. Since the number of organisms were not counted beyond 100, an accurate comparison of the number of organisms present to any data cannot be precisely performed.

The data does show a general trend of few organisms present in sandy and rock areas. However, these areas tend to be those which were found to be most heavily contaminated (i.e. the Outfall Creek

W92194F

and the Raceway) which further complicates the issue.

TOC present in the environment is important as both a food source and habitat to the benthic organisms present.

No obvious relationship was noted between the TOC data and results of the benthic samples. This is most likely due to contaminant stress. Data for these parameters are presented in Tables 4-5 and 4-6.

4.6 <u>Invertebrate Tissue Analysis</u>

Biologists from the EPA Environmental Services Division (ESD) collected and analyzed larvae of the caddis fly (family Hydropsychidae) to determine the concentrations of contaminant metals in the tissues of animals in this trophic level. The collection procedures and locations used for this study are summarized in Section 2.0 of this report. These animals were selected because their feeding habits are expected to provide them with a relatively high exposure dose from contaminants in the suspended sediment.

Due to the limits of sample quantity, the organism tissue was analyzed only for three metals: mercury, chromium, and lead. Results of these analyses are included in Appendix F.

The results discussed in Appendix F show that the concentrations of mercury in the tissue of caddis fly larvae collected from the upstream Reach (ECF-01, ECF-02) may not be significantly lower than that of any other sample. The highest concentration of mercury (wet weight) was detected in ECF-03 (Reach 2) and the highest concentration of mercury (dry weight) was reported to be the same in two samples ECF-02 (background) and ECF-03 (Reach 2). However, data quality limitations make these assumptions questionable.

Chromium was reported above detection limits in one sample, ECF-04, a duplicate sample in Reach 4. Chromium was reported in the other duplicate at detection limits. Chromium was reported below detection limits at all other locations, including background.

The highest lead concentration was in the first background sample (ECF-01) by dry and wet weight analysis. Other high levels of lead were found in samples collected from Reaches 2 and 7.

Evaluation of this data indicates that the data represents a qualitative reference point. Metals were detected in the tissues of these organisms, but the concentrations of mercury and other heavy metals reported are approximated and lack significant differentiation from background values to quantify risks to these or higher organisms in the food chain. This data does, however,

indicate that these organisms may act as a transport pathway for mercury to enter the food chain (Nolan, 1920).

4.7 <u>Eastern Wetlands Soil Boring Program</u>

4.7.1 Background

Phase I sampling located organic and inorganic contaminants in the Wetlands (see Section 4.3.2.1) at elevated concentrations. The Eastern Wetlands boring program was conducted as part of the Phase II Study Area investigation, to define the horizontal and vertical extent of contamination in the Wetlands.

Analytical data is presented in Appendix C and summarized in Table 4-7. This data has been summarized separately from the other sediment sampling data because samples were collected in a different manner and constitute a different information base.

The Wetlands are dominated by two small ponds averaging two to three feet in depth and separated by a low dike as shown on Figures 1-D and 1-E. The ponds are surrounded by a wetland inundated with up to one foot of water. A small feeder stream which drains the eastern portion of the Nyanza Site discharges to the Wetland to the southwest of the southern pond. A second feeder stream which has been redirected reportedly also drained the Nyanza Site and fed into the west side of the northern pond.

Tricil Environmental Response Company, as part of the Operable Unit I remediation effort, excavated contaminated soils and sediment from the Trolley Brook Wetlands and feeder streams, within the eastern border of the Site to the south of Megunko Road. Soils were excavated to a depth of four feet. The southern pond feeder stream was excavated during remediation to a depth of one to two feet to the extent of the Site boundary.

4.7.2 Samples

Thirty boreholes were advanced in the Wetlands using mechanized and hand-operated augers. Samples were collected to a depth of six feet in the ponds. Four sample intervals were collected: sediment surface to a depth of two feet, from four to five feet, and from five to six feet. Borings were advanced to a depth of two feet in surrounding wetland areas and samples collected from two discrete one-foot intervals. The upper sample interval often was a mixture of recent alluvial sediments and underlying soil. Any vegetation was preferentially removed prior to collecting the sample. A surface sediment sample was collected in the southern pond feeder stream and one borehole was advanced in Trolley Brook, which is the only surface water outlet for the Wetlands.
Caution should be exercised in comparing borehole data to River sediment data. Various physical and chemical processes have both accumulated and dispersed contaminants in River sediments. Contaminants are readily adsorbed onto sediments making this type of deposit a contaminant "sink". The soils underlying the Eastern Wetlands, on the other hand, are generally coarser grained. Contaminants do not accumulate in coarse grained soil to the degree as in finer grained sediments. For this reason a "background" sediment concentration cannot be compared to contaminant concentrations in coarser soils.

4.7.3 Surficial Geology

Minimal recent alluvial sediment accumulation was encountered in the ponds. This was expected considering the wetlands are less than 60 years old, formed through filling of low lying land for real estate development (EPA, 1982). The history of the Wetlands is described in Section 2.0.

A thin vegetative cover was found to overlay a peat deposit which was locally up to two feet thick (Figure 3-3). Regional glaciofluvial soils consisting of silty fine sands were encountered beneath the peat deposits. A coarser, silty gravel underlies the northern pond. This unit was not encountered under the southern pond. The gravel may be eroded or pinched out north of the southern pond. Monitoring instruments frequently detected volatile organic vapors in the gravel unit, indicating that VOCs may be migrating preferentially through this deeper unconsolidated unit.

4.7.4 Analytical Results

Many organic and inorganic contaminants have been detected in samples throughout the Wetlands. This discussion will focus on the Site-related contaminants. The most notable contaminant occurrence is the presence of volatile organic compounds (in concentrations in the range of 10' ppb) in the upper sediments of the northern pond, and the pervasive high concentrations of mercury in the upper sample intervals and at lower concentrations to depths of four to six feet. Sample results are illustrated on Figure 2-D.

4.7.4.1 Organic Compounds

A number of Site-related contaminants were detected in significant concentrations in the Wetlands. Chlorobenzene was detected in the central portion of the northern pond in concentrations up to 34,000 ppb at boring SD3-235. The contamination extended into the central portion of the pond where concentrations at depth range from 23 (SD3-226) to 360 ppb (SD3-235). TCE was detected in concentrations up to 1300 ppb (SD3-233) in the general area of the chlorobenzene contamination. The highest concentrations of TCE was generally in

- W92194F

the upper sample interval and decreased with depth. Occasionally, TCE and other volatile organic compounds were detected at slightly higher concentrations in the bottom gravel unit such as TCE in borings SD3-229, 231, 232, and 236. Chlorobenzene exhibited a similar pattern in borings SD3-229, 231, 233, and 238. A third VOC, 1,2-Dichloroethene (1,2-DCE), was also detected in the same area in concentrations up to 5500 ppb (upper interval of boring SD3-233). Contaminant levels generally decreased with depth but increased slightly in the bottom gravel unit. The vertical distribution of contaminants is illustrated in Table 4-7.

Traces of organic contaminants were also detected in the southern pond but are apparently randomly distributed.

Several semi-volatile compounds were detected in the same area as the volatile compounds, most notably 1,2-DCB. The highest concentration (13,000 ppb) was detected in the top interval of boring SD3-234. Concentrations of 1,2-DCB ranging from 1,200 to 8,300 ppb were detected in several other borings in the vicinity. Concentrations vary greatly with depth; some concentrations decreased sharply (SD3-234) while others were persistent at depth such as in borings SD3-233 and 236.

Other semi-volatile compounds detected include several PAH compounds and phthalates. As with the volatiles, the semi-volatile compounds were detected at significantly lower concentrations in the southern pond.

4.7.4.2 <u>Inorganic Contaminants</u>

The primary inorganic contaminants in the Eastern Wetlands were mercury and chromium. Both were widespread throughout the Wetland system and vertical distributions were similar.

The highest concentrations of both mercury and chromium found in the northern pond during this boring program were detected in the surface interval of SD3-235. At this location, mercury was present at 13.8 ppm and chromium at 101.0 ppm. Concentrations of both metals were elevated between SD3-235 and SD3-237, located to the north at the discharge Culvert. Otherwise, both metals were distributed throughout the pond in a somewhat random pattern.

The highest concentrations of mercury and chromium detected during this boring program were found in samples collected from the stream feeding the south pond (SD3-218). This sample was collected from the surface to six inches, while other top interval samples collected during the boring program were from the surface to 12 or 24 inches. The metals were likely to be present at significantly higher concentrations on the ground surface and decrease with depth. This sample may represent a high biased sample.

W92194F

(

7

SUMMARY OF ANALYTICAL RESULTS EASTERN WETLANDS BORING PROGRAM NYANZA III - SUDBURY RIVER STUDY

	-					EAST	EPH WETLAND	8 BEDIMENT	- TOTAL INOR	GANICS					
· · · · · · · · · · · · · · · · · · ·							<u>l</u>	INITS: mg/kg							
INORGANIC	1		DEFTH = 0-2F	T			C	EPTH = 2-41	FT.		DEPTH = 4-5FT.				
COMPOUNDS	FREQUENCY	DETECTED	AVERAGE REPORTED	MAXMUM	MINIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXIMUM	MINIMUM	FREQUENCY	AVERAGE DETECTED	AVERAGE REPORTED	MAXMUM	MINIMUM
ALUMINUM	23 OF 32	8358 28	8024 22	15400.00	4370 00	23 OF 31	8831 98	7734 24	13900 00	5945 00	17 OF 21	6764 71	6065 48	11300.00	3520.00
ARBENIC	14 OF 32	2 63	1 32	7 00	0.01	17 OF 31	1 78	1 08	3 60	0.71	18 OF 21	2 04	1.70	6 10	0.64
BARUM	32 OF 32	N M	32 54	64 70	12 90	31 OF 31	29 58	29.58	53.10	12.00	21 OF 21	28 95	26 95	47 80	11.40
BERYLLIUM	20 OF 32	2 10	1 80	6 20	1 20	21 OF 31	1 27	1.01	1.70	1.15	14 OF 21	1.21	0 97	1.30	1.10
CADMIUM	207 32	2 80	0 63	3 40	1 90	2 OF 31	1 50	0 57	1.50	1.50	2 OF 21	1.30	0 59	1.30	1.30
CALCIUM	32 OF 32	2797 81	2797 81	13400 00	1280 00	31 OF 31	2138 45	2138.45	3250.00	1330.00	21 OF 21	2318 57	2318 57	3800.00	1340.00
CHROMIUM	25 OF 32	\$1.71	41 82	424 00	4 80	29 OF 31	20.06	19 13	51.85	4.90	19 OF 21	21 27	19 63	93.90	7.80
COBALT	17 OF 32	12 85	7 44	74 00	3 30	25 OF 31	5 05	4 31	7 40	3 40	17 OF 21	6 25	5 31	14.10	3.70
COPPER	17 OF 32	57 10	31 80	318 00	6 10	14 OF 31	14 33	6 38	54.50	6 50	13 OF 21	16 12	11 24	45 60	8.00
IFION	25 OF 32	7641 60	6311 82	21900 00	3720 00	30 OF 31	77 19 87	7598 61	11900.00	3100 00	21 OF 21	8478 57	8478 57	14100.00	4890.00
LEAD	21 OF 32	308 42	201 80	6760 00	6 30	18 OF 31	150 88	77.33	1740.00	3 20	14 OF 21	175 81	118 72	2360.00	3.20
MAGNESIUM	31 OF 32	1588 71	1554.10	2620 00	300 00	31 OF 31	2372 58	2372 58	3390 00	985 00	21 OF 21	2664 29	2664 29	4670 00	1320.00
MANGANESE	30 OF 32	60 62	00 31	478 00	30 40	31 OF 31	95 27	85 27	141.50	52 90	21 OF 21	115 32	115 32	218.00	61.00
MERCURY	26 OF 32	7 20	5 63	01 10	0 17	22 OF 31	1 45	1 03	12.00	0.17	10 OF 21	2 50	1.23	10.30	0.17
NICKEL	16 OF 32	1274	7 50	23.4	7.3	24 OF 31	10 32	8.75	15 20	8 80	11 OF 21	12 84	8 24	21 30	8.60
POTASSIUM	32 OF 32	512 27	612 27	834 00	190 00	30 OF 31	840 90	911.01	2060 00	265 00	21 OF 21	1529 05	1529 05	3710.00	607.00
BODIUM	15 OF 32	° 887 73	300 11	3750 00	137 00	11 OF 31	304 95	180 66	594 00	134.50	9 OF 21	284 00	168 63	571.00	125.00
THALLIUM	1 OF 32	1 30	0 40	1 40	1 20	2 OF 31	1 35	0 48	1 60	1.10	0 OF 21		0 00		
VANADIUM	30 OF 32	13 82	13.10	22 80	8 10	31 OF 31	16 35	18 35	26 70	5 90	21 OF 21	18 69	18.69	30 40	7.50
ZING	23 07 32	<u>M M</u>	42 65	290 00	18 20	27 OF 31	37.72	36 59	75.70	22.30	19 OF 21	47.72	45 22	81 90	20.00

1

, '

TABLE 4-7

W92194F

SUMMARY OF ANALYTICAL RESULTS EASTERN WETLANDS BORING PROGRAM NYANZA III – SUDBURY RIVER STUDY PAGE TWO

			E	ASTERN WETL	ANDS SEDIM	ENT - TOTAL IN	IORGANICS			
					UNITS #	g/kg				
INORGANIC	DEPTH = 5-OFT					(DEPTH = 6-7 FT.			
COMPOUNDS	FREQUENCY	AVERAGE DETECTED	AVERAGE	MAXMUM	MINIMUM	FREQUNCY	AVERAGE	AVERAGE REPORTED	MAXIMUM	MINIMUM
ALUMINUM	17 OF 18	7848 47	7408 58	17200 00	3400 00	1 OF 1	4840 00	4840 00	4840.00	4840.00
ARBENIC	15 OF 19	1 26	1 08	2 70	0 83	0 OF 1				
BARIUM	18 OF 18	30 05	30 05	53 20	11 80	1 OF 1	18 40	18 40	18 40	18 40
BERYLLIUM	14 OF 18	1 23	1 04	1.40	1 10	0 OF 1				
CADMIUM	1 OF 10	1 30	0 54	1 30	1 30	0 OF 1				
CALCIUM	18 OF 18	2717 37	2717 37	4960 00	1310 00	1 OF 1	1660 00	1680 00	1660 00	1660.00
CHROMIUM	17 OF 18	21.08	10 27	63 60	7 20	0 OF 1				
COBALT	15 OF 10	6.86	5 71	10 80	3 30	1 OF 1	4 00	4 00	4 00	4 00
COPPER	13 OF 10	11 🗰	0.50	20 60	5 90	1 OF 1	8 70	6 70	8 70	6.70
IRON	18 OF 18	10381 05	10381 05	30000 00	5380 00	10F 1	8480 00	6460 00	6480 00	6480.00
LEAD	13 OF 18	156 55	110 👀	1940 00	2 80	1 OF 1	3 00	3 00	3 00	3 00
MAGNEBIUM	19 OF 18	3527 37	3527 37	8630 00	1280 00	1 OF 1	1750 00	1750.00	1750.00	1750.00
MANGANESE	18 OF 18	148 80	148 80	277 00	63 50	1 OF 1	67.00	87.00	67 00	87 00
MERCURY	13 OF 18	1 50	1 03	10 80	0 19	1 OF 1	0 33	0.33	0.33	0.33
NICKEL	11 OF 10	12 34	8 57	16 60	8 60	0 OF 1				
POTASSIUM	18 OF 18	1802 74	1802 74	3570 00	864 00	1 OF 1	1340 00	1340 00	1340 00	1340 00
SODIUM	7 OF 19	215 43	129 58	271 00	163.00	0 OF 1				
THALLIUM	0 OF 19		0 00			0 OF 1				
VANADIUM	18 OF 18	24 87	24 87	88 50	0 40	1 OF 1	12 00	12 00	12.00	12.00
ŻINC	17 OF 18	40 85	48.81	78 20	25 60	1 OF 1	42 70	42.70	42.70	42 70

.

(

SUMMARY OF ANALYTICAL RESULTS EASTERN WETLANDS BORING PROGRAM NYANZA III — SUDBURY RIVER STUDY PAGE THREE

W92194F

							EASTERNWET	LANDS SEDIM	ENT - ORGAN	1C9					
	•·			• · · •				NT9: ug/kg	<u>-</u> .						
ORGANIC			EPTH = 0-2	<u>n</u>				DEPTH = 2-41	ন) ••••••••••••••••••••••••••••••••••••	DEPTH = 4-51	FT	
COMPOUNDS	FREQUENCY	DETECTED	REPORTED	MAXIM	Manma Ar	FREQUENCY	DETECTED	REPORTED	MAXIMUM	MINIMUM	FREQUENCY	DETECTED	REPORTED	MAXIMUM	MNIMUM
VINYL CHLORIDE	10732	46 00	222 03	130.00	6 00	2 OF 31	25 25	35.39	39.00	11 50	2 OF 21	13.00	42.17	21.00	5.00
METHYLENE CHLORIDE	101.25	625 50	138 52	1800.00	51 00	1 OF 31	17 00	19 15	17 00	17.00	0 OF 21				
ACETONE	0 CF 32	34 17	224 22	110.00	7 00	E OF 31	34 63	44 10	96.00	5.00	7 OF 21	53 00	56 28	100.00	8.00
1,2-DICHLOROETHENE (TOTAL)	10 OF 32	\$72.03	342 13	5500 00	0 30	11 OF 31	71 55	40 48	300.00	1.00	9 OF 21	41.78	36 62	150.00	4.00
1.2-DICHLOROETHANE	007 22					1 OF 31	1 00	18 84	1 00	1.00	0 OF 21				
TRICHLOROETHENE	10 CF 32	312.30	216.30	3400 00	0 30	14 OF 31	94.39	57.15	970.00	2 00	13 OF 21	37 65	41.83	200.00	0.50
BENZENE	107 22	4 00	118.88	4 00	4 00	1 OF 31	3 00	18 87	3 00	3 00	1 OF 21	3 00	20.60	3 00	3.00
TOLUENE	3 07 38	8 33	11881	16 00	2 00	2 OF 31	1 50	16 62	2 00	1.00	1 OF 21	0 60	20 48	- 0 60	0.60
CHLOROBENZENE	19 07 22	8953 73	2004 00	34000 00	3 00	14 OF 31	414 25	168.19	2000.00	5.00	12 OF 21	304 50	174.90	2100.00	4.00
ETHYLBENZENE	10732	0.70	118.78	ê 70	0 70	0 OF 31					0 OF 21				
TOTAL XYLENES	3 07 32	2 67	110.27	8 00	1 00	4 OF 31	1 20	16 77	2 00	0.80	1 OF 21	2 00	20 55	2 00	2.00
PHENOL	2 OF 32	et 60	340.75	88 00	56 00	0 OF 31					1 OF 21	35.00	228.76	35.00	35.00
2-CHLOROPHENOL	107.32	30 00	345.16	20 02	20 00	0 OF 31					0 OF 21				
1,3-DICHLOROBENZENE	10 CF 32	90-40	87376	200.00	16 00	4 OF 31	34 38	228.94	58 00	17.00	3 OF 21	57 67	213.00	76.00	33.00
1.4-DICHLOROBENZENE	13 07 38	808 89	878.01	3100.00	48 00	11 OF 31	227 62	248.02	620.00	27.00	7 OF 21	262 29	255.05	780,00	29.00
1.2-DICHLOROBENZENE	18 07 38	3616 73	1877 60	13000 00	33 00	12 OF 31	1215 82	630.11	4900.00	71.00	11 OF 21	968.45	633.71	5100 00	30.00
4-METHYLPHENOL	105 22	74 00	347 16	74 00	74 00	1 OF 31	14 00	250.71	14 00	14.00	0 OF 21				
BENZOIC ACID	6 CF 38	101 25	1311 25	140.00	e5 co	# OF 31	52 63	1025 90	110.00	15 00	2 OF 21	171 50	1058.61	310,00	33.00
1.2.4-TRICHLOROBENZENE	130732	202.27	324 63	1600 00	21 00	11 OF 31	126.95	212.23	290.00	30 00	5 OF 21	280.80	243.95	610.00	45.00
NAPHTHALENE	807 22	198.33	301 90	320.00	42 00	3 OF 31	36 63	234 65	64 00	19 00	3 OF 21	31.00	209.19	48 00	23.00
2-METHYLNAPHTHALENE	107.32	82 00	362 28	62 00	62 00	1 OF 31	16 50	244.10	18 50	18 50	1 OF 21	27 00	229.38	27.00	27.00
ACENAPHITHYLENE	107.12	60 00	351 50	60 00	60 00	1 OF 31	63 50	250.68	63 50	63 50	1 OF 21	150.00	235 24	150.00	150.00
ACENAPHTHENE	3 04 39	31 30	376 68	45 00	21 00	1 OF 31	18 00	248.10	16 00	18 00	1 OF 21	29 00	229.48	29 00	29.00
DIRENZOFLIBAN	107.39	43.00	361 03	43 00	43 00	1 OF 31	23 50	248.42	23 50	23 50	1 OF 21	32 00	229.62	32 00	32.00
FILIDRENE	10739		351 56	60 00	60 00	1 OF 31	78 00	252.73	79.00	79 00	1 OF 21	53 00	230.62	53 00	53.00
						8 OF 31					0 OF 21	•••			
	4 CE 10		308.38	75.00	75 m	9 CE 31	62 (0)	34247	es m	34.00	1 06 21	19.00	229.00	19.00	18.00
	4 (75.34					10531		1344 88	en m		0.06.21	10.00	120.00		10.00
		48 77		(30.00		105.11	110.00	267.88	31000	310 00	20521	187.50	212 14	350.00	25.00
	1.0.4		14.00	120.00		1 CF 31		211.00	47.50	4.00	105 21	110.00	233.33	110.00	110.00
		4.5	576.63			105.31	17.00	240.21	67.30	17 00	206.21	20.60	210.33	21.00	30.00
UI-N-BUITUPHINALAIE	101		444 11	100	1.00	1 OF 31	17 00	248.21	17.00	17.00	20521	20 50	210.33	21.00 000.00	20.00
PLOUPANITERE						2 CF 35	977 24	204.00	640.00	11 00	20521	404.50	20110	990.00	30.00
			100.11			105.11	11.00	25378	11.00	14 30	20721	25.00	231.30	25 00	25.00
BUTTLEENZTUPHTPALATE			544.00	23 00		10531	270.00	231 03	370.00	270.00	10521	400.00	251.43	490.00	480.00
BENZOWAN INVICENCE	10 1			140.00		10531	270.00	236.37	2/000	27000	10521	=20.00	25145	420.00	#200.000
				190,00		1 (JF 3)	354.00	206.90	33400	2000	10521	620.00	206 71	A1 00	41.00
	10.2						- 20	22300			100 21	12 (27	228.67	12.00	19.00
			33/ 87		000	10731	9 50	247.87	0.00	270.00	10541	12.00	220.07	400 00	16.00
ULNLOID)FLUCHUNTHENE		14 H)	950 94	140.00	4100	1 UP 31	\$/4.00	206.3/	2/000	«/U00	10721	990.00	201.40 000 00	750.00	700.00
Genzo (k) Ruonanthene	007.22		••• <i>•</i> =			00-31					10-21	/30.00	202.00	/30.00	/30,00
BENZONIPYRENE	10 CF 32	200.00	334 22	00 00	41 00	7 UF 31	65 00	220.18	290.00	18 00	20-21	31800	246.48	5/U.QO	
INDENO(1,2,3-cd)PYRENE	10732	130.00	353 44	120.00	120.00	10431	200.00	254.11	200.00	200.00	10121	270.00	240.45	2/1100	2/0.00
DIBENZ (A, H) AN THEACENE	0 07 32					104-31	26.00	252.03	20 00	20 00	10121	84.00	232.57	#4 00	94.00
BENZO(gh)PERYLENE	10734	120.00	353 44	120.00	120.00	1 0F 31	210.00	254 84	210.00	210.00	1 07 21	260.00	240.48	200.00	200.00

.

1

4-91

.

1

TABLE 4-7

SUMMARY OF ANALYTICAL REBULTS EASTERN WETLANDS BORING PROGRAM NYANZA III - SUDBURY RIVER STUDY PAGE FOUR

W92194F

	EASTERN WETLANDS SEDIMENT - ORGANICS											
	UNITS ug/ug											
ORGANIC	l		2EPTH = \$-01	FT				DEPTH = 6-7	<u>. </u>			
COMPOUNDS		AVENUCE	AVENADE				AVERAGE	AVERAGE				
	FREQUENCY	DETECTED	REPORTED			PREGUENCY	DETECTED	HEPOHIED	MAXIMUM	MINIMUR		
VINYL CHLORIDE	1 OF 18	10 00	6.79	10 00	10 00	0 OF 1						
METHYLENE CHLORIDE	0 OF 19					B OF 1						
ACETONE	# OF 18	37 00	10.34	100.00	7 00	1 OF 1	170	170	170	17		
1,2-DICHLOROETHENE(TOTAL)	0 CF 10	37.39	18.71	95 00	2 00	1 OF 1	17	17	17	1		
1,2-DICHLOROETHANE	0 OF 19					0 OF 1						
TRICHLOROETHENE	12 OF 10	29 40	19 01	110.00	0 80	1 OF 1	79	79	79	7		
BENZENE	1 OF 10	4 00	3 29	4 00	4 00	0 OF 1						
TOLLIENE	2 OF 19	3 00	3 37	4 00	2 00	0 OF 1						
CHLOROBENZENE	10 OF 19	140.00	74 🗗	360.00	11.00	1 OF 1	250	250	250	25		
ETHYLBENZENE	0 OF 10					0 OF 1						
TOTAL XYLENES	1 OF 10	0 80	323	0 80	0 #0	0 OF 1						
PHENOL	2 OF 10	63 50	216.42	120.00	47.00	0 OF 1						
2-CHLOROPHENOL	1 OF 19	45 00	221 64	45 00	45.00	0 OF 1						
1,3-DICHLOROBENZENE	0 OF 10					0 OF 1						
1.4-DICHLOROBENZENE	2 OF 10	48 50	21195	54 00	43 00	1 OF 1	40	40	40	4		
1,2-DICHLOROBENZENE	8 OF 18	182 17	207 79	500.00	48 00	1 OF 1	240	240	240	24		
4-METHYLPHENOL	0 OF 10					0 OF 1						
BENZOIC ACID	2 OF 10	14 50	1012 05	17 00	12 00	0 OF 1						
1.2A-TRICHLOROBENZENE	1 OF 18	66 00	223 11	60 00	ee co	0 OF 1						
NAPITINALENE	0 CF 10					0 OF 1						
2-METHYLNAPHTHALENE	0.07 10					0 OF 1						
ACENAPHTHYLENE	0 OF 18					0 OF 1						
ACENAPHTHENE	3 OF 19	22.33	198.64	23 00	21.00	0 OF 1						
DIBENZOFURAN	0 CF 10					0 OF 1						
FLUORENE	0 CF 10					0 OF 1						
4 & DINTRO-2-METHY PHENOL	0 CF 10					0 OF 1						
N-NITROROUGHEN YAMNE	0.05.10					0 OF 1						
PENTACH OBOPHENOL	0.07.10					6 OF 1						
PHENANTHEENE	0 CF 10					6 OF 1						
	8 OF 18					0.0001						
	100.00	11.00	918 78	31 00		0 CE 1						
EL INDANTHENE	0.05 10		2.0.0	•••		0 CE 1						
PVDENE	3.05.10			31.00		0.051						
	1.05.16	27.00		74.00	24 00	8 OF 1						
		/4 00		74.00	~~~~	0.051						
					Į	0.051						
						0.071						
					1	0001						

4
9
N

.

(

From the point where the feeder stream enters the south pond, both metals displayed high correlation within samples, following what is probably an ancestral stream bed. The series of samples included, from south to north, SD3-224, 240, 221 and 223. The last sample in the series was located at the channelway through the dike. A sample collected at the channel discharge in the north pond, SD3-213 also exhibited elevated mercury and chromium concentrations. The ancestral stream bed was not positively identified in the northern pond, but based on elevated mercury and chromium concentrations can be speculated to extend from SD3-213 to 228, 233, 235, 236, 226, and 237.

4.7.4.3 <u>Eastern Wetlands Discussion</u>

The horizontal distribution of contaminants suggests that two different waste streams were introduced to the Wetlands from two distinct locations. The first an obvious source, is the south pond feeder stream, where elevated levels of Site-related metal contaminants occur. Site-related organic contaminants are not present in this area. A second feeder stream transporting a mixed metals - organic compound wastes, may have discharged into the north pond in the vicinity of SD3-235. This theory is supported by the detection of the highest Site-related organic contamination in the Wetlands coincident with the occurrence of metals.

Concentrations of representative contaminants noted on Figure 2-D can be followed from the Wetlands to SD3-200 and 201 in Chemical Brook Culvert. The more persistent contaminants can be traced further downstream through the Outfall Creek (Figure 2-C) to the Sudbury River. Further discussion of contaminant distribution is found in Sections 4.8 and 5.0.

4.8 <u>Summary and Discussion of Distribution of Contaminants in</u> the Study Area

Both organic and inorganic wastewater and sludge discharges have been documented at the Nyanza Site and various organic and inorganic Contaminant constituents are well documented as occurring in River sediments throughout the Study Area.

This section summarizes the nature and extent of contaminants in the Study Area, from the Site source area through to Reach 10 of the Sudbury River. The discussion will focus on mercury in sediments, since this contaminant is most clearly linked to the Nyanza Site. Chromium in sediments will also be discussed. Like mercury, it is linked to the Site, is persistent in the environment, and displays the same trends as mercury.

4.8.1 Inorganic Contaminants in the Study Area

4.8.1.1 <u>Sediments</u>

Figure 4-3 illustrates the average detected concentrations of mercury and chromium from Reach 1, which represents background, through the Study Area to Reach 10.

Results of the Eastern Wetlands drilling program are not considered in the graphic analysis since they constitute a separate base of information. Phase I samples collected from this area are comparable, and are included in Figure 4-3 (see the discussion in Section 4.7.2).

Mercury contamination in sediments displays a distinctive pattern. High Average Detected Concentrations (ADCs) are associated with slow water flow and depositional areas such as within an impoundment. River runs with relatively high water flow velocities exhibit low ADCs. This is well illustrated in Figure 4-3.

Maximum concentrations follow a similar general trend. However, these results cannot exhibit a reliable pattern because they represent single sample points.

The highest concentrations of mercury in sediments occur in the Eastern Wetlands area, which drains the eastern flank of the Nyanza Site (Figure 2-A and 2-C). The ADC is 44.84 ppm (maximum of 152 ppm), compared to a background ADC of 1.05 ppm (maximum of 1.59 ppm).

As the sediment bed load is transported downstream through Chemical Brook Culvert to the Outfall Creek, and thus to the raceway and river sediment from other sources begins to mix with contaminated sediments from the Eastern Wetlands and formerly, the Site. This mixing results in a decrease in mercury concentration as the Siterelated sediment enters the Raceway and subsequently, the Sudbury River. However, there is a dramatic rise in mercury concentrations in River sediment at the Raceway confluence. Concentrations of mercury in samples upgradient of this confluence are at or below background levels, while the samples below the confluence exhibit average detected concentrations above background.

Samples collected from the Raceway-also show low concentrations of mercury in the sediment due to the lack of depositional areas resulting from high water flow velocities.--

Mercury concentrations in Reach 3 rise sharply. Reach 3 consists of a short River run and Reservoir 2. Within the Sudbury River, Reservoir 2 contains the highest concentrations of inorganic contaminants. The ADC and maximum concentration drop in Reach 4

W92194F





FIGURE 4-3

R = Reach #, EW = Eastern Wetlands, OC = Outfall Creek

but remain at relatively high levels. The average and maximum drop considerably in Reach 5, a swift flowing River run.

The ADC and maximum concentration of mercury rises in Reach 6, the impoundment above the Saxonville Dam. These values drop again in Reaches 7 and 8, which are slow, meandering River runs. Mercury concentration rises again in Reach 9, Fairhaven Bay, before dropping below background concentrations in Reach 10, a River run.

Chromium, another Site-related contaminant, was also detected in Eastern Wetland sediments, however, was detected in the Outfall Creek and Reach 3 (Reservoir 2) sediments in higher concentrations than in the Eastern Wetlands. Otherwise, chromium follows a similar pattern as mercury; lower concentrations in River runs and higher in impoundments. Chromium was detected below background levels in Reach 10.

Lead is another site-related inorganic contaminant and is found at elevated levels throughout the Study Area. However, lead may be attributed to the existence of point and non-point sources described in Section 4.2.3.

The pattern of lead distribution in the study area does not follow that of mercury and chromium. ADCs and maximum concentrations in the Eastern Wetlands are lower than those in background area, and are highest in the Raceway. These levels drop to background concentrations in Reach 2 (ADC=58, Max 295) and remain in that range in Reaches 3 and 4. Concentrations in Reaches 5 and 6 are significantly higher (ADCs=237, 285 ppm; and maximum=809, 876 ppm, respectively), and then drop to background levels in Reach 8.

4.8.1.2 Surface Water

Mercury and chromium occurred at low concentrations in several surface water samples. These samples were located from the Eastern Wetlands to the inlet of Reservoir 2. The only apparent distribution pattern is that all samples are in proximity of the Site.

Lead was also detected in surface water samples throughout Reaches 1 through 5, the Eastern Wetlands, and the associated discharge culvert. No distribution pattern was apparent.

4.8.2 Organic Contaminants in the Study Area

Several organic compounds identified as Site-related contaminants were found in several Study Area media.

Site-related organic contaminants were detected in Eastern Wetlands in the range of 10^1 to 10^4 ppb. sediments include These dichlorobenzene, chlorobenzene, trichloroethene, and Concentrations decreased within a short distance dichloroethene. downstream of the Wetlands and these organics were not detected in significant concentrations in the River Reaches. Minimal concentrations (less than 13 ppb maximum concentration) of volatile organic compounds were detected in River water samples.

Eight organic compounds were identified in the Final Work Plan as being used by the dye industry during the history of production at Nyanza. These compounds were referred to as "Site-specific organics" and include 1-napthalamine, 2-napthalamine, aniline, pnitrotoluene, o-nitrochlorobenzene, benzidine, o-tolidine and odianisidine. This designation should not be confused with "Siterelated contaminants" as defined by EPA (Section 4.2.1).

Sediment and surface water samples were collected and analyzed for these compounds during both Phase I and II. Detailed shallow and deep interval sediment sampling was conducted in an area in Reach 2, extending from Mill Pond to the confluence of the Raceway and elevated the Outfall Creek. Α groundwater plume with concentrations of aniline mapped by Ebasco is projected to intersect the Sudbury River at this point. None of the Sitespecific organic compounds were detected in the river.

The groundwater - surface water relationship was not investigated as part of this study, however, several possibilities exist to explain the absence of the Site-specific organics in the River:

- o these organic compounds are unstable in the environment and readily breakdown in the alluvial environment
- the groundwater plume discharges to the River, but not at a sufficient rate to contribute measurable quantities of the contaminants to the River
- o the groundwater plume does not discharge to the River
- o contaminants are discharged to the River and are rapidly diluted

Polynuclear Aromatic Hydrocarbon compounds were also detected in sediments in the Study Area. PAHs are primarily formed during incomplete combustion or pyrolysis of organic materials. As such, the compounds are typically found in areas impacted by roadways, railways and other urban runoff. This trend is evidenced by their presence in Reaches located near these features, including the Culvert and Outfall Creek, and Reaches 2, 3 and the upper portions of Reach 7. These contaminants were detected less frequently and at lower concentrations in the Eastern Wetlands. None of these compounds were detected in the surface waters.

Mercury can be transformed by various bacterial processes to methylmercury (mono- and di-methylations), but this organo-metallic compound is very unstable in the environment. Samples collected from the Study Area indicate very low levels of methylmercury in the sediment; none was detected in the water. Methylmercury in these media is generally dependent on the pH of the waters. Waters with a low pH (less than 5) are more likely to allow methylation of mercury (Bloom, 1989). The Study Area waters generally maintain a pH greater than 6.0 and methylmercury will not be a stable compound in sediment or surface water. Methylmercury tends to be more prevalent in fish tissue than is mercury in its elemental form.

Scattered occurrences of pesticides and PCBs are noted in sediments throughout the Study Area. One pesticide, Lindane, was detected in one water sample at a concentration of 0.015 ppb. Pesticides and PCBs readily bioaccumulate in biota, and sampling analyses indicate that these compounds have accumulated in Study Area fish and benthic biota. This will be discussed further in Section 7.0.

The final group of organic compounds detected in the Study Area are the phthalate compounds. These chemicals are generally considered to be derived from manufacturing and breakdown of plastics. These compounds were principally detected in Reaches 3, 6, and in the Outfall Creek. Only one phthalate compound, bis(2-ethylhexyl) phthalate, was detected in three surface water samples at a maximum concentration of 56 ppb.

W92194F

4-98

FINAL

5.0 CONTAMINANT FATE AND TRANSPORT

The occurrence of Site-related contaminants in the River in concentrations above background in several media (including water, and biota) indicates that these contaminants are sediment, migrating and accumulating in the environment. After a chemical is released to the environment, it can be accumulated, transported, or Accumulation and transport of contaminants have transformed. occurred in the River. Mercury, in particular, has accumulated in sediments and has been bioaccumulated in biota and upper trophiclevel fish. This section focuses on the primary contaminant fate and transport processes occurring in the River, including the bulk transport of contaminated sediments by the River and fate processes that play a role in contaminant uptake by biota. This section also includes process and contaminant-specific discussions.

5.1 <u>Sediment Transport in Aquatic Systems</u>

Many contaminants are readily adsorbed onto and transported with sediments. The most hazardous Study Area contaminants, such as mercury and pesticides, are among those most likely to be found in high concentrations in and transported with sediments because of the high partitioning coefficients associated with such contaminants. For this reason, sediment transport is considered the primary process for movement of contaminants in the Study Area.

Sediments originate from River inflows, shore erosion, and, to a Once introduced into water, lesser extent, waste disposal. sediments are dispersed primarily by currents and are either deposited to the bottom, where they can become a permanent part of bed sediments, or resuspended and further transported. Deposition to bottom sediments depends on particle size, settling rates, and flocculation. Particle diameters can range over two orders of magnitude, and settling rates can range over four orders of Thorough knowledge of particle sizes, magnitude (Lick, 1984). settling rates, and flocculation effects is important for determining settling and deposition rates and contaminant release An in-depth analysis of from sediments to overlying waters. requires extensive sediment transport processes sediment characterization that is beyond the scope of this study; however, this section presents some generalizations on sediment transport.

The net flux of sediment (NF) at the sediment/water interface can be described as the difference between the entrainment rate (E) and the deposition rate (D).

NF = E - D

D and E are considered to be independent processes, and NF = E when there is no suspended sediment and, therefore, no deposition is

W92194F

present (Lick, 1984). Conversely, NF = D in the absence of entrainment. Entrainment depends on: turbulent stress at the sediment/water interface, water content of the deposited sediments, grain size and specific gravity of the sediments, density of compacted sediment, and the activity of benthic biota and bacteria (Lick, 1984).

Currents and, to a lesser degree, wind- or boat-generated wave action at reservoir shorelines are the most important sediment transport processes because they are directly responsible for the transport of sediments, cause turbulence resulting in dispersion, and cause shear stress which results in sediment transport at the sediment/water interface (Lick, 1984).

Based on partitioning coefficients, many of the chemicals of concern (COCs) in this study including mercury, PCBs, and DDT are likely to be concentrated in River sediments. This assumption is supported by the relatively low surface water concentrations compared to relatively high sediment concentrations. Under calm conditions, surface water concentrations are likely to remain relatively low; however, release of some contaminated sediments to surface waters is likely during periods of turbulence at the sediment/water interface. Currents, bioturbation, and wave action likely to result in the resuspension and transport of are contaminated sediments. Except for those COCs expected to biomagnify in food chains, such as mercury, high concentrations of COCs adsorbed to sediment particles are not likely to be the primary hazard to site biota because of decreased bioavailability. It is more critical that sediments containing high concentrations of COCs may continue to be important sources of contamination to surface waters whenever the sediment/water interface is disturbed, such as during high water flows resulting from spring runoff and rain events.

An important element of sediment transport is the speciation or forms of contaminants being transported both into and out of aquatic systems. Forms of chromium and mercury are highly toxic and different forms of some organic chemicals, such as DDT, DDD, and DDE, are associated with different degrees of toxicity. For example, inorganic mercury introduced into aquatic systems can be deposited to bed sediments without change. In the bed sediments, bacterial methylation of inorganic mercury may occur, potentially increasing the hazards associated with those sediments. The relative concentrations of methylmercury and inorganic mercury are usually considered to be critical for evaluating sediment toxicity. However, concentrations of chemicals in sediments might not be in equilibrium, and inorganic mercury/ methylmercury concentrations that contribute to total sediment mercury are probably rarely Because of the dynamics associated with sediment mercury static. contamination, all mercury in sediments is often assessed as

W92194F

methylated or potentially methylated.

Speciation of chromium was investigated through laboratory analysis. Six sediment samples were analyzed for hexavalent chromium. Concentrations of total chromium ranged from nondetectable to 2,620 ppm; hexavalent chromium was not detected in any of these samples.

5.2 <u>Persistence</u> (T_{1/2})

Ultimate chemical fate is commonly expressed as overall environmental half-life $(T_{1/2})$. Overall aqueous half-lives are based on the maximum degradation rates associated with the most important degradation processes in an aqueous medium (Howard et al., 1991). These processes include photolysis, hydrolysis, aerobic biodegradation, and anaerobic biodegradation. Table 7-5 lists overall half-lives for the final Ecological Risk Assessment COCs for surface waters, sediments, and biota.

Overall half-lives are media-dependent; therefore, soil and sediment half-lives are likely to differ from surface water halflives. Biodegradation is the dominant degradation process in soil, except for those chemicals that undergo rapid hydrolysis. Biodegradation is the dominant degradation process in surface water (Howard et al., 1991).

5.3 <u>Partitioning Coefficients</u>

Interstitial water ingestion by organisms is a primary entry point of contaminants into the food chain. Partitioning coefficients are useful for estimating interstitial water concentrations of organic chemicals from measured sediment concentrations. The coefficients used in the ecological risk assessment (Section 7.0) include Kd, Koc, and Kow. These coefficients are defined as follows (EPA, 1989a):

where

Kd = Koc + foc

W1161 (

Kd = <u>conc. of chemical in soil or sediment (ug/kg)</u> conc. of chemical in water (ug/L)

and

Koc = partitioning coefficient for chemical/ organic carbon

foc = fraction of organic carbon in sediment/soil

Koc values are available in the accepted literature for most organic chemicals. Kd values are often estimated from general Koc values and from site-specific foc values. The resulting Kd values

W92194F

can be used to estimate water concentrations from measured sediment concentrations, or vice versa. If Koc values are unavailable for a chemical of concern, they can be estimated from the octanol/water partition coefficient, Kow. Octanol is considered to be a surrogate for lipid.

Kow <u>concentration of chemical in octanol</u> concentration of chemical in water

Regression equations relate Kow to Koc, and are chemical classspecific. The following equation is appropriate for Koc estimations for pesticides and all other chemicals for which no specific equation exists (EPA, 1988b):

Log Koc = 0.544 log Kow + 1.377

For polynuclear aromatic hydrocarbons (PAHs), the following equation is appropriate (EPA, 1988b):

Log Koc = 0.937 log Kow - 0.006

5.4 <u>Bioconcentration Potential (BCF)</u>

Chemicals can be transported from sediments, interstitial waters, or surface waters to biota. The chemical uptake rate can exceed the depuration rate, resulting in bioconcentration in biota exposed to those chemicals. The site of bioconcentration is chemical- and organism-specific, and many organic chemicals have an affinity for lipid; therefore, these chemicals are deposited in fatty tissue. Chemicals that are partitioned to lipid might not be available to cause adverse effects to host organisms until these lipids are Metals can be partitioned to nearly any tissue metabolized. depending on the metal and the exposed organism. The fact that metals can be partitioned to muscle tissue is of special concern contaminants stored in a particularly toxic form for (methylmercury) in edible fish tissue.

The bioconcentration factor is used to relate chemical uptake to chemical depuration; it is defined as the net accumulation of a chemical via aqueous uptake in excess of depuration, or

BCF <u>chemical concentration in biota</u> chemical concentration in water

BCFs are most commonly based on whole-body values (wet weight); however, BCFs are often determined for edible portions of fish and shellfish for which human consumption is a consideration. BCFs are based on uptake through water only and do not include uptake through dietary means. If both dietary and water uptake routes are

W92194F

considered, the term is defined as bioaccumulation factor (BAF). Generally, BAFs exceed BCFs for the same species and the same Therefore, BAFs are preferred over BCFs for assessing chemical. overall uptake and food chain effects. In general, however, BAF data are lacking for most chemicals and most species. Therefore, bioaccumulation potential is more commonly expressed as BCF (EPA, For this assessment, BCF values were used to assess 1985h). bioaccumulation potential because BAF data were lacking for many species and chemicals of concern. The food chain model used to estimate bioaccumulation in the aquatic food chain predicts BAFs; predicted BAF values were used to estimate tissue these concentrations in aquatic biota.

Reported BCFs used in this study, which were obtained primarily from EPA Ambient Water Quality Criteria Documents (1980-88), are based on lipid-normalized BCFs. This adjustment is intended to make all measured BCFs for a material comparable, regardless of species or tissue with which the BCF was measured (EPA, 1985h).

There is general agreement on the significance of BCF values. Most commonly, BCF values less than 1,000 are not considered to be significant. Recently, EPA (1991) recommended that chemicals with log Kow values of 3.5 or less be considered to have low potential to bioconcentrate. Log Kow is directly related to log BCF, and the EPA cutoff value of 3.5 (log Kow) reflects unadjusted BCFs of approximately 3,000. Therefore, COCs associated with BCFs greater than 3,000 were considered candidates for final media-specific COCs.

A few highly lipophilic (attracted to lipid) chemicals, such as methylmercury, DDT, and PCBs, tend to biomagnify in food chains or food webs. Biomagnification is defined as increasing biotic chemical concentration in successively higher trophic levels. For chemicals with high biomagnification potential, top-level predators commonly present the highest tissue concentrations of chemicals in food chains. Biomagnification also appears to be correlated to Long-lived species, and especially older exposure duration. individuals of long-lived species, tend to have the highest tissue concentrations of such biomagnified chemicals as mercury or PCBs. The ingestion of contaminated prey is considered the primary route of exposure for chemicals with high biomagnification potential. Other exposure routes, such as water intake, are considered to be likely but of little concern in comparison to the ingestion of food contaminated with highly lipophilic chemicals.

The preferred method for determining biomagnification of chemicals of concern is by measuring contaminant concentrations in tissue samples of several representative food chain organisms encompassing the primary trophic levels of the food chain. However, analytical considerations (methodology limitations, expense, and time constraints) often preclude complex food chain sampling. An appropriate alternative method for evaluating biomagnification is through the combined use of selective tissue sampling and food chain modeling. Computer-based food chain models, such as those of Thomann (1989), are useful for predicting organic chemical distribution in some aquatic food chains.

5.5 Fate and Transport Modeling

Sampling results specific to the Study Area include extensive laboratory-measured sediment and surface water COC concentrations and measured tissue concentrations of the primary COCs. Because data are extensive and the chemicals of concern are so well characterized, complex computer modeling of chemical fate and transport is considered unnecessary for this study. The decision not to use computer modeling fate and transport is based on a primary assumption that extensive measured data in sediments, surface water, and biota, are associated with lower levels of uncertainty than predicted values based on generalized computer models.

The prediction of future exposure concentrations in the various media clearly dictates the use of computer-based fate and transport models. However, future exposure scenarios are not the focus of this study, and the simple fate and transport models described above (Kd, $T_{1/2}$, BCF), in addition to the extensive sample data available, should adequately describe chemical fate and transport for this study.

5.6 <u>Bediment Transport</u>

5.6.1 Introduction

This section discusses sediment transport based on stream flow velocity data obtained during monthly sampling events from November 1989 through December 1990. Flow velocities were not measured at all sampling locations during each month. All flow velocities were obtained using the "6/10" method of measurement; this method obtains flow velocities from a point above the stream bed that is approximately 6/10 of the total stream depth. The average depth at the points of stream flow measurement in Reach 1 through Reach 7 was two to three feet; therefore, flow velocities referenced in this section were obtained at a depth of 0.8 to 1.2 feet beneath the surface. Particle size data for each respective Reach were obtained during Phase I and Phase II sampling events; this work occurred during a time-frame that was suitable for making comparisons.

Contaminants are most likely to be adsorbed onto silts and clays;

however, these particles generally require higher flow velocities than sands to become entrained in the stream. The most mobile particles are fine sands that do not possess the high absorption tendencies of silts and clays. The following sections discuss contaminant transport entrainment, the resuspension of bed-load sediment, and traction and transport of bed-load along the River The maximum recorded flow velocities and the highest bottom. recorded percentages of fine sand, silt, and clay were used to determine a "worst-case" scenario of the viability of contaminant transport by sediment entrainment and bed-load traction. Table 5-1 summarizes the data discussed in the following sections. Α description of each Reach is presented in Section 1.2.1 and illustrated in Figures 1-A and 1-B.

5.6.2 Reach 1

This section of the Sudbury River is approximately 6.5 miles long and drops 100 feet in elevation along its course. Stream flow velocities were calculated from Southville Pond to the Nyanza Site. Flow velocity in the Southville Pond was below measurable limits; immediately downstream of the Cordaville Dam, which however, impounds the pond, maximum velocities of 15.24 cm/sec were measured. Flow velocities of 39.62 cm/sec were recorded 500 feet downstream from Cordaville Dam. Stream flow decreased to 3.05 cm/sec approximately 2,000 feet upstream from the Nyanza Site; however, approximately 1,400 feet upstream of the Nyanza Site, flow Stream flow velocities reached a maximum of 91.44 cm/sec. velocities reached entrainment levels (over 30 cm/sec) during April, June, August, October, November, and December. (Velocities above 30 cm/sec will begin the entrainment of fine sand and approach the point at which silt and clay will become fully mobile in the stream) (Friedman and Sanders, 1978).

Sediment size data for Reach 1 were obtained from only the downstream half of the Reach beginning at Southville Pond. The sediments in the pond and immediately downstream of the pond's spillway consist of predominantly coarse to medium sand with minor accumulations of small pebbles. Approximately 1,500 feet upstream from the Nyanza Site, the Sudbury River bed consists of fine sand grading to coarse silt.

Velocities greater than 90 cm/sec will entrain coarse pebbles and are more than sufficient to entrain clay. Maximum flow velocities exceeding 90 cm/sec were measured. Flow velocities of 30 cm/sec or greater were measured during several months in 1990. Contaminant transport by sediment entrainment and bed-load traction will occur under these conditions.

FINAL

٠.

Table 5-1. Summary of Hydraulic Data for the Sudbury Riv	ver
--	-----

•

Stream	Length in	Elevation	Flow Veloci	ty (cm/sec)	Sediment Size			
Reach	Miles	Drop	Min	Max.	Upstream	Downstream		
1	6.5	100	3.05	91.44	. NA	medium to coarse sand		
2	1.16	minimat	1.00	9.14	medium to very fine	fine sand, coarse silt		
3	1.83	NA	<1.00	30.48	pebbles to fine sand coarse to medium silts	very fine sand to coarse silt		
4	0.73	minimal	<1.00	NA	very fine sands, slits and clays	NA		
5	1.59	locally steep	NA	9.14	various sizes of sand	NA		
6	1.23	minimal	<1.00 or below measurable limits	3.05	coarse sand to coarse silt	coarse silt, very fine sand		
7	3.37	minimal	<1.00 or below measurable limits	NA	coarse to fine sand	very fine sands, coarse to medium silts		

(

ſ

ł

η

5.6.3 Reach 2

This section of the River, stretching from Mill Pond to the confluence of Cold Spring Brook and the River, exhibits a very low gradient, much less than that of Reach 1. The wetlands bordering this reach suggest a stream flow velocity low enough to allow the accumulation of vegetative growth. Measured flow velocities in Mill Pond were less than 1.00 cm/sec. Velocities near the end of this reach measured between 3.05 and 9.14 cm/sec.

Bottom sediments in Mill Pond ranged from medium to very fine sand (upstream) to fine sand/coarse silt (downstream). The rest of the reach was characterized by a stream floor consisting of pebbles and coarse sand, with less than ten percent of the sediments analyzed smaller than medium sand.

The maximum flow velocity of 9.14 cm/sec was observed twice during 1990. Such a velocity is insufficient to entrain fine sand, the most mobile sediment size. Contaminant transport by traction of fine sand along the stream floor is an uncommon event. Traction within Reach 2 would require an unusually smooth stream bed and consistent bottom currents greater than those measured. The occurrence of this phenomenon is highly unlikely. A wide range of measured flow velocities indicates that portions of this Reach, particularly Mill Pond, are depositional.

The channelized portions of the Reach have measured flow velocities sufficient to maintain sediment in suspension.

5.6.4 Reach 3

This section of the Sudbury River consists of Reservoir 2 and a short length of River channel before Reservoir 2. Reservoir 2 consists of two separate surface water bodies connected by a channel-way. Velocities of water flowing into the southern section of Reservoir 2 reached a maximum of 30.48 cm/sec, while flow at the downstream end was measured at 3.05 cm/sec. At the farthest downstream sampling location (upstream of the Reservoir 2 spillway), flow velocity was less than 1.00 cm/sec. Maximum flow velocities in this Reach were observed during April 1990.

Just before the River enters the upstream section of Reservoir 2, the bed-load consists of a wide range of sediment sizes, from pebbles to fine sand (SD3-117 and 158). Throughout the upstream section of Reservoir 2, bottom sediments are dominated by coarse and medium silts (SD3-118 and 120). The downstream section of Reservoir 2 contains a large percentage of coarse to very fine sand. Minor coarse silt also occurs.

Before entering Reservoir 2, water velocities reached a maximum of

W92194F

30.48 cm/sec, which would indicate the beginning stages of fine sand entrainment. However, both sections of Reservoir 2 consistently exhibited maximum flow velocities of 3.05 cm/sec, which will not support any entrainment or bed-load traction. Therefore, contaminant transport by entrainment and traction is viable only in the short length of stream channel upstream of Reservoir 2.

The River portion of this Reach will maintain sediment in suspension. The low-flow conditions in Reservoir 2 indicate a depositional environment.

5.6.5 Reach 4

This section of the Sudbury River consists of Reservoir 1 and several associated wetland areas. No data on flow velocities of water in this reach were available; however, it can be assumed that water velocities in this reservoir would be similar to those observed at other surface water impoundments along the Sudbury River (less than 1.00 cm/sec).

Sediments in Reach 4 were silts and clays with some fine sand size particles. Transport by sediment entrainment or traction is not possible in Reach 4 under normal flow conditions. This Reach is also likely to be depositional in nature.

5.6.6 Reach 5

This Reach consists of slightly meandering open waterway. Immediately downstream of Reach 4, the River gradient increases, resulting in greater flow velocity. The only stream flow velocity measurement available was obtained immediately downstream of the dam that impounds Reservoir 1. The flow velocity at this sampling location was 9.14 cm/sec.

Pebbles are the dominant sediment of the River floor approximately 2,000 to 4,000 feet downstream of the dam. Over the entire length of the Reach, sands make up the largest percentage of River sediment deposits.

A flow of 9.14 cm/sec is probably not representative of the entire Reach. However, the extent of Reach 5 is narrow and well-defined, suggesting that average stream flow velocity is sufficiently high that host lithologies are resistant enough to decrease the meandering of the stream. Because sand of various sizes is the predominant sediment, and since the maximum flow velocities reported are too low to entrain fine sand, contaminant migration by sediment entrainment is not possible under normal flow conditions. However, as stated in earlier sections, 9.14 cm/sec can support the traction of finer sands if physical conditions of the stream floor

are conducive to the traction process. Therefore, traction should not be eliminated as an occasional method of transporting contaminant-laden sediment downstream.

Velocities in the range of 9 cm/sec are sufficient to keep fine sand and smaller size fractions in suspension. Any sediment in suspension, therefore, is likely to be transported through Reach 5. This is supported by the low percentage of finer grained size fractions in sediment samples along this reach, including SD3-136, 174, 175, and 176.

Sediments in the Oxbow Lake in Reach 5, including samples SD3-138, 177, and 178, are classified as medium to coarse sand and sandy silts. The Oxbow Lake is a cut-off meander of the River. The finer grain sizes found here are probably a result of an infilling process, including bank erosion and slumping that has occurred since the oxbow was cut off from the River.

5.6.7 Reach 6

Reach 6 consists of two impoundments contained by the Fenwick Street and Saxonville Dams. This Reach is choked with submerged and floating leafy vegetation.

Flow data collected at the Fenwick Street Dam impoundment indicate a maximum flow of 3.05 cm/sec. Flow velocity measured in the Saxonville Dam impoundment consistently was below measurable limits.

No sediment grain size data were available for the surface water impoundment behind the Fenwick Street Dam due to the high organic content of the sample. Bottom sediments within the Saxonville Dam impoundment ranged from coarse sand to coarse silt, with the coarse silt and very fine sand predominant in the downstream sections.

Low-flow velocities and fine sediments collected in this Reach indicate that water flowing through these impoundments cannot maintain sediment in suspension. This Reach is a depositional area.

5.6.8 Reach 7

Reach 7 of the Sudbury River represents a large floodplain. In characteristic nature, this section consists of broad meanders created as the gradient flattens considerably and approaches horizontal. Stream flow velocities were recorded as below measurable limits or 0 cm/sec. Again, the River is covered by lush growths of floating vegetation suggestive of extremely slow-moving water.

W92194F

Coarse to fine sand dominated the bottom sediments in the upstream third of this reach. The presence of small pebbles was reported in one sample location, SD3-183. Further downstream of this section, coarse and medium sands were replaced by very fine sands and coarse to medium silts.

It is likely that both sediment entrainment and traction of bedload are not possible under normal stream flow conditions in this Reach, and that large portions are depositional areas, with water flow velocities too slow to maintain sediments in suspension.

No grain-size samples were collected from reaches 8,9, or 10.

5.6.9 Eastern Wetlands

The Eastern Wetlands received runoff and lagoon discharges from the Nyanza Site for many years. This wetland area might be a continuing source of contamination to the Sudbury River.

The Eastern Wetlands are impounded by an old trolley bed to the west, an exposed bedrock knob to the east, and Megunko Road to the north (Figure 2-D). The Wetlands consist of several acres of ponded water that discharges to a culvert passing under Megunko Road to Trolley Brook, which subsequently enters Chemical Brook Culvert.

Water flow in the Wetlands is not measurable. Flow velocity and volume at the discharge point ranges from negligible during dry periods to substantial during rain events and spring runoff. Most contaminant (and sediment) transport will occur during these events.

Recharge of the Wetlands by groundwater occurs seasonally, however, the interaction of the surface water and groundwater was not the focus of this investigation. Groundwater recharge as a potential contaminant pathway is addressed by Ebasco (1991) in the Operable Unit II groundwater study.

5.6.10 Chemical Brook Culvert/Outfall Creek

The transport of contaminants through Chemical Brook Culvert was investigated through a visual and remote video inspection. Appendix H contains details of the Culvert video inspection.

An attempt was made to quantify water flow using a current meter; however, the data may not accurately reflect flow velocities due to turbulent water flow at most Culvert access points. Water flow velocities exiting sections of Culverts into manholes appeared to be sufficient to keep fine grained sediments in suspension and to transport them through the Culvert. However, minor accumulations

W92194F

of fine sediments were noted in manholes at the upstream end of the Culvert.

The Culvert is a constructed series of pipes and connected catch basins. Because the catch basins are deeper and larger in diameter than the Culvert pipe, water entering them will flow turbulently. Sediment will be deposited in the backwater eddy of the turbulent flow. Several inches of silt-size sediment were noted in two manholes in the upstream end of the Culvert. Sediment in other manholes consisted of sandy gravel with cultural debris. No other significant accumulations of sediment were noted during the video inspection.

Much of the downstream portion of the Culvert has an open, stream bed-type bottom. The Culvert discharges to the Outfall Creek before it empties into the Raceway. The Outfall Creek is a swiftmoving channelized stream. No accumulations of sediments were noted in the Creek.

The surface water/groundwater relationship was not the focus of this study, so information is not available to determine if the Culvert and Outfall Creek are receiving or recharging groundwater.

5.6.11 Raceway

The Raceway is a channelized, fast-moving waterway built to drive power-generating equipment and to supply water to historic mill sites (Figure 2C).

Coarse sand and gravel characterize the Raceway bottom. Some wellestablished deposits of fine-grained sediments are present in limited areas throughout the length of the Raceway.

No flow velocity data were collected here. Visual estimates indicate that the water velocity is sufficient to transport fine sand and smaller sediments through the length of the Raceway under normal flow conditions. Seasonal flooding conditions can cause the influx of sediments and redistribute fine-grained sediment deposits through the length of the Raceway; however, this feature is not considered a depositional area under normal flow conditions.

5.7 <u>Chemical-Specific Fate and Transport</u>

Results of sampling indicate that several chemicals or classes of chemicals are expected to dominate both the human health and ecological risk assessments. These chemicals include mercury, chromium, lead, polynuclear aromatic hydrocarbons, chlorinated benzenes, pesticides (DDT, DDE, chlordane), and possibly other organic chemicals such as acetone and trichloroethene. In general, fate and transport of a chemical in a given chemical class

(polynuclear aromatic hydrocarbons, metals, pesticides) are likely to follow similar processes as other chemicals within that class. The following sections contain a generalized review of fate and transport processes for chemicals or classes of chemicals, of potential concern.

5.7.1 Chromium

Chromium is an essential element for mammals and possibly other organisms, and in nearly all environmentally important compounds it exists as either hexavalent (Cr^{+6}) or trivalent (Cr^{+3}) chromium (FWS, 1986a). Hexavalent chromium in the environment is usually the result of domestic and industrial discharges, while trivalent chromium is the form generally found in biological tissues. Under certain conditions, hexavalent chromium is reduced to trivalent chromium, thereby reducing potential toxicity. Under these conditions, trivalent chromium exists as an essentially nontoxic precipitate.

Hexavalent chromium is many times more soluble (and hence more bioavailable) than trivalent chromium; therefore, high surface water concentrations of hexavalent chromium pose a significant hazard to aquatic life. In contrast, trivalent chromium has a low solubility, and thus significant surface water concentrations are extremely unlikely.

Hexavalent chromium analyses were performed on six sediment samples collected in the Study Area. The results of these analyses indicate hexavalent chromium concentrations are below detection limits (10 ug/L) for all samples.

Because sediments contaminated with chromium are likely to contain primarily trivalent chromium, chromium toxicity is unlikely under naturally occurring conditions unless surface water chromium concentrations approach levels considered toxic.

Aqueous transport processes of chromium include surface water transport (primarily dissolved toxic hexavalent chromium) and sediment/suspended solids transport (primarily less toxic trivalent chromium). Chromium speciation and behavior in soils is not well known, but chromium adsorbed to soil particles is readily transported during surface runoff to aqueous systems. Chromium adsorbed to soil and sediment should be considered persistent, but the dynamics of chromium oxidation/reduction under natural conditions are not well studied; the potential toxicity of contaminated systems can change over time or under certain conditions. Whether oxidation of trivalent chromium (to Cr⁺⁶) or reduction of hexavalent chromium (to Cr⁺³) is the dominant process appears to depend on aeration and pH. Environmental fate processes

for most metals, including chromium, are limited in number, and neither hexavalent nor trivalent chromium significantly bioconcentrates in fish or invertebrates (EPA, 1985c).

Four of 33 unfiltered water samples (SW3-103, 104, 106 and 113) contained chromium above detectable limits. The highest concentration of chromium in these samples was 79 ppm (SW3-104, Eastern Wetlands). Three of the unfiltered water samples that contained chromium above detectable levels were filtered and then analyzed (see Section 4.3.3.15.). Chromium was not detected in the filtered samples, indicating it is not readily desorbing from sediments and entering the aquatic environment.

5.7.2 Lead

Lead is similar to most metals except mercury in fate and transport properties, including expected sorption to sediments and soils, low bioconcentration potential and environmental persistence (EPA, 1985f). Although lead appears to have an affinity for soils and sediments, surface water concentrations of lead which are known to result in adverse chronic effects to aquatic life, are relatively common. The most environmentally important form of lead is Pb⁺², but several other forms are both soluble and toxic.

Lead is most soluble and bioavailable under conditions of low pH, low organic content, low concentration of suspended solids and low concentrations of calcium, iron, manganese, zinc, and cadmium (FWS, 1988). The water quality criteria measured in the Study Area were above these thresholds of concern. If aqueous lead and lead compounds are present, they tend to concentrate in the water surface microlayer. However, most lead entering natural waters is precipitated to bed sediments as hydroxides or carbonates (FWS, 1988). Desorption from bed sediments is slow. Water flow rate affects both lead speciation and migration. At low stream flow, lead is likely to be removed rapidly from the water column by adsorption onto sediment particles and subsequent sedimentation (FWS, 1988). As with many metals, lead is mobilized and released from sediments when pH decreases rapidly.

Lead adsorbed to sediment particles is likely to be transported to and in aquatic systems by processes similar to those described for chromium. Possible fate processes for lead in aquatic systems include: long-term sorption to bed sediments; desorption and release from sediments under low pH conditions; uptake by aquatic organisms; and possibly methylation by fish and other aquatic life. The mechanism and potential biological effects of lead methylation are unclear (FWS, 1988). Freshwater bioconcentration factors for lead suggest a relatively low potential for significant bioconcentration. Lead does not biomagnify in food chains.

W92194F

5.7.3 Mercury

Mercury occurs in the natural environment primarily as inorganic mercury (Hg^{+2}) . Its sources can include the direct industrial discharge of inorganic mercury or the oxidization of elemental mercury to inorganic mercury. Inorganic mercury, in turn, can be methylated by both aerobic and anaerobic bacteria that are present in freshwater sediments and in the slime coat, intestines, and liver of fish (FWS, 1987a). Methylmercury is more toxic to biota than inorganic mercury; therefore, the conversion of inorganic mercury to methylmercury is a critical process. Analyses performed on 22 surface water samples collected in the Study Area revealed methylmercury concentrations below detection limits in all the samples. Similar analyses were performed on 64 sediment samples; less than eight percent of the sediment samples exhibited detectable concentrations of monomethylmercury. Dimethylmercury was detected in only one sediment sample.

Primary fate processes of mercury include adsorption to sediments and suspended particulates; relatively rapid biotic uptake (especially methylmercury) through ingestion of contaminated sediments, prey, and water; methylation of inorganic mercury; slow depuration (elimination and metabolism); bioconcentration in tissue; and biomagnification in the food chain. Freshwater bioconcentration factors for inorganic mercury range from 1,800 to nearly 5,000, while similar factors for methylmercury range from 10,000 to 85,700 (EPA, 1984g).

Mercury is transported to and within aquatic systems by processes generally similar to those described for the other metals of concern. Results of TCLP tests performed on sediments collected in the Study Area suggest that sediment-bound mercury is likely to remain bound to sediment particles. Even at pH 5.0, which is less than pH values expected in the Study Area, aqueous mercury was below detection limits using TCLP methodology. The relatively low concentrations of mercury in surface water samples measured from compared to the relatively high sediment the Study Area, concentrations detected, support this assumption. Because of the potential for methylation, rapid uptake, and slow depuration, mercury is transported in significant concentrations through food The high degree of both bioaccumulation (uptake through chains. food and water) and toxicity, results in the classification of mercury as one of the most toxic chemicals in the environment.

5.7.4 Polynuclear Aromatic Hydrocarbons

Polynuclear or polycyclic aromatic hydrocarbons (PAHs) are a diverse class of compounds consisting of substituted and unsubstituted polycyclic and heterocyclic aromatic rings. PAHs are

naturally occurring, but most PAHs found in the environment today are the result of the incomplete combustion of organic compounds. PAHs are widely distributed in the environment, and have been

PAHs generally sorb strongly to soil and, in aquatic systems, adsorption to sediments, suspended particulates, and biota is likely. Photolysis is likely to be the primary degradation process in surface waters; however, photolysis rates of specific compounds vary widely within the class of PAHs. In addition, PAHs in surface waters can evaporate, disperse into the water column, or undergo chemical oxidation. Experimental evidence indicates that the ultimate fate of PAHs adsorbed to sediments is biodegradation and biotransformation by benthic organisms (EPA, 1980h). Complete degradation times are reported to range from less than one day to six weeks for particulate-bound PAHs (FWS, 1987b). Biotic uptake of PAHs from water and food is generally fairly rapid, yet bioconcentration in most freshwater organisms, especially fish, is low because of rapid metabolism. Mammals, fish, and possibly other organisms are efficient metabolizers of PAHs, yet mammalian and recent fish studies suggest that intermediate metabolites might be highly toxic, mutagenic, or carcinogenic to the host. Therefore, even though bioconcentration potential is low, considered unacceptable carcinogenic and noncarcinogenic risk levels might be associated with elevated environmental concentrations of PAHs.

detected in air, groundwater, surface water, soil, sediments, and

plant and animal tissue (EPA, 1980h).

Run-off from roadways is a significant contributor to surface water and sediment PAH contamination. PAHs that are transported to surface waters from run-off are likely to adsorb to sediment particles and be transported downstream. Thus, PAHs are transported by sediment transport processes. Resuspension of sediments release PAH-contaminated sediments to the water column, where photolytic processes are likely to result in degradation of these compounds. In addition, releases of sediment-adsorbed PAHs to surface waters can increase PAH bioavailability and potential for adverse effects to aquatic life. While surface water concentrations of PAHs remained low in samples collected in the Study Area, sediment samples revealed significant contamination with a variety of PAH compounds.

5.7.5 Chlorinated Bensenes

Chlorinated benzenes include 1,2-, 1,3-, 1,4-dichlorobenzene, and nine other chlorinated benzenes identified by the degree and location of chlorination on the benzene rings (EPA, 1980c, 1980e). Environmental persistence varies with the type of compound. Compounds that are more halogenated tend to be more resistant to biodegradation and, therefore, are more persistent (EPA, 1980c). Chlorinated benzenes are not likely to biomagnify within food chains. However, a 1972 report of hexachlorobenzene in the eggs of common terns that had apparently eaten contaminated fish, suggests that hexachlorobenzene can biomagnify (EPA, 1980c). The implications associated with this finding are unclear and further study is warranted.

Dichlorobenzenes are readily soluble in lipids and are relatively volatile (EPA, 1980e). The log Kow for 1,4-dichlorobenzene is 3.37 (EPA, 1980e), suggesting that this compound is not likely to bioconcentrate significantly in biota. Reported freshwater BCFs are less than 100 (EPA, 1980e). However, mammalian studies suggest that excretion of dichlorobenzenes is quite slow, indicating a cause for concern under conditions of long exposure durations.

Transport of chlorinated benzenes is not well characterized. Based on partitioning coefficients, aqueous transport is likely under natural conditions. Dichlorobenzenes are relatively immobile compounds in the environment and tend to be adsorbed and transported with sediments. Chlorinated benzenes in sediments and soils can be relatively persistent if these compounds are highly chlorinated; therefore, sediment-bound compounds may or may not significantly to surface water contamination. contribute Chlorinated benzenes were measured above detection limits in three surface water samples and 24 sediment samples collected in the The contaminated sediments in the Study Area are Study Area. likely to serve as potential sources of aqueous contamination.

5.7.6 Pesticides

Pesticides of potential concern include DDT and its metabolites (primarily DDD and DDE) and chlordane. DDT and its metabolites are highly persistent in soil and water; they are also widely dispersed by erosion, runoff, and volatilization, and have low water solubility and high lipophilicity, resulting in bioaccumulation in wildlife and humans (EPA, 1980d). The log Kow for DDT and its metabolites ranges from approximately 4.0 to over 6.1, which explains the relatively high lipophilicity of these compounds. Measured freshwater BCFs range from approximately 2,000 (<u>Procambarus alleni</u>, crayfish) to greater than 4 million (kiyi, a freshwater fish) (EPA, 1980d).

The maximum estimated biodegradation half-life of DDT in soil and under aerobic conditions in surface water is 15.6 years (Howard et al., 1991). The use of DDT has been banned in the United States for nearly 20 years, yet it is still observed in significant concentrations in both soils and waters from a variety of sites. This might indicate that the estimated half-life is too conservative. Thirteen sediment samples from the Study Area contained DDT or its metabolites at concentrations above detection limits. Because of the properties stated above, DDT and its

W92194F

metabolites are likely to be transported with sediments and within biota. Aqueous transport is probably unlikely under most conditions, and surface water samples collected in the Study Area did not contain detectable concentrations of DDT, DDD, or DDE.

Chlordane is a broad-spectrum pesticide that was produced in a variety of formulations from 1947 until 1983, when its use in the United States was restricted to termite control. Although the aqueous half-life of some chlordane isomers is short (<18 hours), soil half-lives can exceed 14 years (FWS, 1990). In general, biomagnification of chlordane in freshwater environments is unexpected. Chlordane is absorbed readily by mammals and birds through skin, diet, and inhalation. The potential for chlordane absorption by other groups of animals is unknown. Data are also unavailable on the degradation of chlordane by photolysis, photooxidation, or reduction. A hydrolysis half-life of nearly 200,000 years has been estimated (Howard et al., 1991), indicating that hydrolysis is not a primary degradative process.

Chlordane transport is likely to be similar to that of DDT and its metabolites; therefore, the transport of suspended sediments contaminated with chlordane might be the primary route of surface water transport. Concentrations of chlordane in surface waters with low suspended solids are likely to remain low. In support of this argument, chlordane was not detected in any surface water samples collected within the Study Area. Sediment half-lives for chlordane are likely to be similar to those of soils; therefore, chlordane is likely to be relatively persistent in aerobic sediments. Only one sediment sample collected in the Study Area

5.7.7 Polychlorinated Biphenyls

PCBs consist of a mixture of chlorinated biphenyls that contain a varying number of substituted chlorine atoms on the aromatic rings (EPA, 1980g). PCBs are classified according to the degree of chlorination, and are designated by the manufacturer as Aroclors. Less chlorinated Aroclors include 1016, 1221, 1232, 1242, and 1248. Increasing chlorination is designated by higher Aroclor numbers (1254, 1260, 1262, 1268, and 1270). In addition to being toxic and environmentally persistent, PCB mixtures can contain small quantities of highly toxic contaminants, such as polychlorinated dibenzofurans (PCDFs). PCDFs may be responsible for certain toxic effects in humans and animals that are associated with PCBs (EPA, 1980g).

Fate and transport of PCBs depends on specific properties associated with degree of chlorination, which can differ significantly (the water solubility ranges from less than 3 ug/L to 200 ug/L;) (EPA, 1980g). Long-range atmospheric transport of PCBs

W92194F

by wind, rain, and snow is well documented (FWS, 1986b). Volatilization is apparently rapid in aqueous systems, but PCB sediments limits the rate of volatilization. sorption to Adsorption to bed sediments is likely under most naturally occurring freshwater aquatic conditions, and PCBs can remain in sediments for at least eight to 15 years (FWS, 1986b). Reported Kows for PCBs range from 10,000 to 20,000 (EPA, 1980g), suggesting a moderate to high bioconcentration potential and a relatively high degree of lipophilicity. Reported freshwater BCFs range from 2,700 (midge) to 274,000 (fathead minnow) (EPA, 1980g). Most BCF data is based on moderately chlorinated Aroclors such as 1242, 1248, and PCBs are included with the few chemicals (DDT 1254. and methylmercury) that are known to biomagnify in food chains; therefore, low aqueous concentrations of PCBs in surface waters can pose serious hazards to aquatic life and organisms that consume contaminated aquatic species.

In general, PCB transport processes are likely to be similar to those of DDT (aqueous transport is most likely limited to transport of contaminated suspended sediment and uptake and transport through food chains). The primary fate processes of PCBs include deposition to bed sediments and biotic uptake and deposition to fatty tissue. PCBs were measured above detection limits in three sediment samples, and were not detected in any surface water samples collected in the Study Area. However, concentrations of PCBs were detected in fish tissue; this subject is discussed in Section 7.0.

5.7.8 Chemical Fate and Transport Summary

The primary transport processes for Study Area contaminants include seepage from soils, overland transport in surface runoff, and in-Contaminants associated with stream transport. suspended particulate matter probably will be transported in flowing waters, settle out in quiet waters, and ultimately be deposited in bed Because of this cycle, impoundment sediments in the sediments. Study Area are generally more contaminated than River sediments. Sediments in Reservoirs Nos. 1 and 2 are more contaminated than those in downstream impoundments. Contaminant release or mobilization from bed sediments are likely only under certain conditions, such as low pH or significant disturbance of upper layer sediments.

All Site contaminants are expected to have an affinity for sediments and soil and, in general, concentrations of contaminants in these media greatly exceed those of surface water. Contaminants that occur at relatively elevated concentrations in surface water should be considered the most hazardous because exposure duration is constant for aquatic life. Evaluating the bioavailability of contaminants is critical for an accurate assessment of the

W92194F

potential for adverse effects to biota, and contaminants dissolved in surface water are more likely to be bioavailable than those sorbed to sediments or soil.

Contaminated sediments can be a long-term source of surface water contamination for persistent chemicals such as metals, pesticides, and, under anaerobic conditions, some PAHs. However, chromium, mercury, and lead are not desorbed readily from sediments, as indicated by the frequency of nondetectable concentrations in TCLP and filtered sample analyses. Desorption is unexpected unless conditions of low pH (lower than has been measured in the River) or significant sediment perturbation exist. Ingestion of contaminated sediments or soils is generally considered a minor exposure pathway for most aquatic or terrestrial biota in comparison to aqueous exposure, but it can be a potentially significant exposure route for humans, especially children. Exposure sources, pathways, and routes are discussed in detail in the appropriate sections for the human health and ecological risk assessments.

The distribution of contaminants, as discussed in Section 4.0, supports the conclusions of this Fate and Transport discussion. Contaminants that have an affinity for sediment particles are found in much higher concentrations in the reservoirs than in the River In addition, sediments in Reservoir 2 generally exhibit runs. higher concentrations of contaminants than Reservoir 1. Contaminant concentrations in reservoir sediments were consistently lower down-River; contaminant concentrations in River segments where relatively high water flow velocities were measured were lower than those encountered in slower moving reaches of the River and considerably lower than those in reservoirs.

-- - -_____ ____ ۶ 1 è 6

· . . . ·

FINAL

6.0 BASELINE PUBLIC HEALTH RISK ASSESSMENT

6.1 <u>Objectives</u>

Section 6.0 presents the methodology for and the results of a baseline public health risk assessment conducted for the Sudbury River Study Area. The objective of the risk assessment is to predict potential current and future risks to the public from the organic and inorganic chemicals detected in surface waters, sediments, and fish tissues. Site-related and study area contaminants were considered separately in the risk assessment to evaluate the contribution of the risk posed by Site-related risks to the overall study area risks. The baseline public health risk assessment for the Study Area was conducted according to guidelines presented in the following references:

- Supplemental Risk Assessment Guidance for the Superfund Program, EPA Region I, June 1989 (EPA 901/5-89-001).
- Risk Assessment Guidance for Superfund, Volume I, Human Health Evaluation Manual (Part A), EPA, December 1989 (EPA 540/1-89-002).

The baseline risk assessment is organized according to the following outline:

- <u>Hazard Identification/Selection of Chemicals of Potential</u> <u>Concern</u>. A subset of all chemicals detected in environmental samples collected from the Study Area is selected to represent Study Area contaminants in the risk assessment. These chemicals are frequently referred to as "indicator chemicals" or "chemicals of concern." Chemicals of concern are selected based on contaminant occurrence and distribution, persistence, toxicity, and mobility. Information on the known or suspected adverse noncarcinogenic or carcinogenic health effects of each indicator compound is presented.
- o <u>Dose-Response Assessment</u>. Available health-based standards and criteria and dose-response parameters (toxicity criteria) are summarized for each chemical of concern.
- o <u>Exposure Assessment</u>. The Study Area is described with respect to the receptors who are potentially exposed to the contaminated environmental media. The risk assessment identifies exposure mechanisms by which human

W92194F
receptors potentially contact the chemicals of concern. Exposure scenarios are developed which allow a quantitative estimation of the "dose" to which an individual may theoretically be exposed as a result of contact with Study Area chemicals of concern.

o <u>Risk Characterization</u>. Toxicity criteria, such as Carcinogenic Slope Factors and Reference Doses are used to estimate the potential for adverse health effects that may occur as a result of human exposure to Study Area contaminants. Separate risk calculations were performed for site-related and study area contaminants. These calculations were performed to provide an indication of whether risks posed by site related contaminants were a major contribution to overall study area risks.

6.2 <u>Hazard Identification</u>

This section provides the rationale for the selection of contaminants of concern (COC). The COCs will be used to characterize the potential for noncarcinogenic effects and the carcinogenic risks associated with the exposure to surface water and sediments and the consumption of fish taken from the Study Area. The chemical occurrence and distribution tables presented in Section 4.0 of the RI provide a basis for selection of contaminants of concern.

6.2.1 Selection of Study Area Chemicals of Concern

The following factors were considered in selecting the COCs for the Study Area:

- o Occurrence and distribution of the chemicals
- o Environmental fate and mobility of the chemicals
- o Chemical toxicity
- o Chemical persistence

The concentration-toxicity screening procedure presented in the Risk Assessment Guidance for Superfund, Volume I, Human Health Evaluation Manual (Part A) (EPA, December 1989) was used to identify those chemicals that are most likely to contribute significantly to risks associated with human exposure to contaminated environmental media. Each chemical detected was assigned a risk factor (or chemical score), based on the maximum concentration of the chemical in the environmental medium and the toxicity criteria for the chemical. The toxicity criteria used to conduct the concentration-toxicity screening are presented on Table 6-1. The primary reference for the toxicity criteria is the

W92194F

FINAL

TABLE 6-1 TOXICITY CRITERIA FOR CHEMICALS OF CONCERN SELECTION NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

	Reference	Cancer Slope Factor					
	Dose Oral Poute	Factor Oral Poute of					
Chemical	of Exposure	Exposure					
	(mg/kg/day)	(mg/kg/day) ⁻¹					
Acetone	1.0E-01 ⁽¹⁾						
Benzene		2.9E-02 A ⁽¹⁾					
2-Butanone	5.0E-02 ⁽²⁾						
Chlorobenzene	2.0E-02 ⁽¹⁾						
Chloromethane		1.3E-02 B2 ⁽²⁾					
1,1-Dichloroethene	9.0E-03 ⁽¹⁾	6.0E-01 C ⁽¹⁾					
1,2-Dichloroethene	1.0E-02 ⁽²⁾						
Ethylbenzene	1.0E-01 ⁽¹⁾						
Methylene chloride	6.0E-02 ⁽¹⁾	7.5E-03 B2 ⁽¹⁾					
Styrene	2.0E-01 ⁽¹⁾	3.0E-02 B2 ⁽¹⁾					
Tetrachloroethene	1.0E-02 ⁽²⁾	5.1E-02 B2 ⁽²⁾					
Toluene	2.0E-01 ⁽¹⁾						
Total Xylenes	2.0 ⁽¹⁾						
Trichloroethene		1.1E-02 B2 ⁽²⁾					
Vinyl chloride		1.9 A ⁽¹⁾					
Acenaphthene	6.0E-02 ⁽¹⁾						
Acenaphthylene**	4.0E-03 ⁽²⁾						
Anthracene	3.0E-01 ⁽¹⁾						
Benzo(a)anthracene*		5.8 B2 ⁽⁴⁾					
Benzo(a)pyrene		5.8 B2 ⁽⁴⁾					
Benzo(b)fluoranthene*		5.8 B2 ⁴⁰					
Benzo(g,h,i)perylene	4.0E-03 ⁽²⁾						
Benzo(k)fluoranthene*		5.8 B2 ⁽⁴⁾					
Benzoic Acid	4.0(1)						
Benzyl alcohol	3.0E-01 ⁽²⁾						
Bis(2-chloroethyl)ether		1.1 B2 ⁽²⁾					
Bis(2-ethylhexyl)phthalate	2.0E-02 ⁽²⁾	1.4E-02 B2 ⁽¹⁾					
Butybenzylphthalate	2.0E-01 ⁽¹⁾						

TABLE 6-1 TOXICITY CRITERIA FOR CHEMICALS OF CONCERN SELECTION NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS PAGE TWO

	Reference	Cancer Slope
	Dose	Factor Oral Poute of
Chemical	of Exposure	Exposure
	(mg/kg/day)	(mg/kg/day) [.]
2-Chlorophenol	5.0E-03 ⁽¹⁾	
Chrysene*	•	5.8 B2 ⁽⁴⁾
Dibenz(a,h)anthracene*		5.8 B2 ⁽⁴⁾
Dibenzofuran	4.0E-03 ⁽³⁾	
1,3-Dichlorobenzene	3.0E-2	
1,4-Dichlorobenzene		2.4E-02 B2 ⁽²⁾
1,2-Dichlorobenzene	9.0E-02 ⁽¹⁾	
Diethylphthalate	8.0E-01 ⁽¹⁾	
Di-N-Butylphthalate	1.0E-01 ⁽¹⁾	
Di-n-octylphthalate	2.0E-02 ⁽²⁾	
Fluoranthene	4.0E-02 ⁽¹⁾	
Fluorene	4.0E-02 ⁽¹⁾	
Indeno(1,2,3-cd)pyrene*		5.8 B2 ⁽⁴⁾
2-Methylnaphthalene	NA	NA
2-Methylphenol	5.0E-02 ⁽¹⁾	
3-/4-Methylphenol	5.0E-02 ⁽¹⁾	
Naphthalene	4.0E-03 ⁽²⁾	
Nitrobenzene	5.0E-04 ⁽¹⁾	
N-Nitrosodiphenylamine		4.9E-03 B2 ⁽¹⁾
Phenanthrene**	4.0E-03 ⁽²⁾	
Phenol	6.0E-01 ⁽¹⁾	
Pyrene	3.0E-02 ⁽¹⁾	
1,2,4-Trichlorobenzene	1.31E-03 ^{cb}	
Aldmin		1 75+1 820
Aldrin	3.0E-05**	1.7E+1 B2
Chlordane	6.0E-05 ⁽¹⁾	1.3 B2 ⁽¹⁾
Chlordane 4,4'-DDD	6.0E-05 ⁽¹⁾	$1.3 B2^{(1)}$ 2.4E-01 B2 ⁽²⁾

TABLE 6-1 TOXICITY CRITERIA FOR CHEMICALS OF CONCERN SELECTION NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS PAGE THREE

......

	Reference	Cancer Slope
	Dose Oral Route	Factor Oral Route of
Chemical	of Exposure	Exposure
· ·	(mg/kg/day)	(mg/kg/day) ^{.1}
4,4'-DDT	5.0E-04 ⁽¹⁾	3.4E-01 B2 ⁽¹⁾
Dieldrin	5.0E-05 ⁽¹⁾	
Endosulfan	5.0E-05 ⁽¹⁾	
Endrin	3.0E-04 ⁽¹⁾	
Endrin Ketone	NA	NA
Heptachlor	5.0E-04 ⁽²⁾	4.5 B2 ⁽²⁾
Heptachlor epoxide	1.3E-05 ⁽¹⁾	9.1 B2 ⁽¹⁾
Lindane	3.0E-04 ⁽¹⁾	1.3 B2 ⁽²⁾
Methyoxychlor	5.0E-03 ⁽¹⁾	
Polychlorinated biphenyls		7.7 B2 ⁽¹⁾
Aluminum	NA	NA
Antimony	4.0E-04 ⁽¹⁾	
Antimony Arsenic	4.0E-04 ⁽¹⁾ 3.0E-04 ⁽¹⁾	1.8A ⁽¹⁾⁽⁵⁾
Antimony Arsenic Barium	4.0E-04 ⁽¹⁾ 3.0E-04 ⁽¹⁾ 5.0E-02 ⁽²⁾	1.8A ⁽¹⁾⁽⁵⁾
Antimony Arsenic Barium Beryllium	4.0E-04 ⁽¹⁾ 3.0E-04 ⁽¹⁾ 5.0E-02 ⁽²⁾ 5.0E-03 ⁽¹⁾	1.8A ⁽¹⁾⁽⁵⁾ 4.3A ⁽¹⁾
Antimony Arsenic Barium Beryllium Cadmium	4.0E-04 ⁽¹⁾ 3.0E-04 ⁽¹⁾ 5.0E-02 ⁽²⁾ 5.0E-03 ⁽¹⁾ 5.0E-04 ⁽¹⁾	1.8A ⁽¹⁾⁽⁵⁾ 4.3A ⁽¹⁾
Antimony Arsenic Barium Beryllium Cadmium Calcium	$4.0E-04^{(1)}$ $3.0E-04^{(1)}$ $5.0E-02^{(2)}$ $5.0E-03^{(1)}$ $5.0E-04^{(1)}$ NA	1.8A ⁽¹⁾⁽⁵⁾ 4.3A ⁽¹⁾ NA
Antimony Arsenic Barium Beryllium Cadmium Calcium Chromium (+3)(6)	$4.0E-04^{(1)}$ $3.0E-04^{(1)}$ $5.0E-02^{(2)}$ $5.0E-03^{(1)}$ $5.0E-04^{(1)}$ NA $1.0^{(2)}$	1.8A ⁽¹⁾⁽⁵⁾ 4.3A ⁽¹⁾ NA
Antimony Arsenic Barium Beryllium Cadmium Calcium Chromium (+3)(6) Cobalt	$4.0E-04^{(1)}$ $3.0E-04^{(1)}$ $5.0E-02^{(2)}$ $5.0E-03^{(1)}$ $5.0E-04^{(1)}$ NA $1.0^{(2)}$ NA	1.8A ⁽¹⁾⁽⁵⁾ 4.3A ⁽¹⁾ NA
Antimony Arsenic Barium Beryllium Cadmium Calcium Chromium (+3)(6) Cobalt Copper	$4.0E-04^{(1)}$ $3.0E-04^{(1)}$ $5.0E-02^{(2)}$ $5.0E-03^{(1)}$ $5.0E-04^{(1)}$ NA $1.0^{(2)}$ NA NA	1.8A ⁽¹⁾⁽⁵⁾ 4.3A ⁽¹⁾ NA NA NA
Antimony Arsenic Barium Beryllium Cadmium Calcium Chromium (+3)(6) Cobalt Copper Iron	4.0E-04 ⁽¹⁾ 3.0E-04 ⁽¹⁾ 5.0E-02 ⁽²⁾ 5.0E-03 ⁽¹⁾ 5.0E-04 ⁽¹⁾ NA 1.0 ⁽²⁾ NA NA NA	1.8A ⁽¹⁾⁽⁵⁾ 4.3A ⁽¹⁾ NA NA NA NA
Antimony Arsenic Barium Beryllium Cadmium Calcium Chromium (+3)(6) Cobalt Copper Iron Lead	4.0E-04 ⁽¹⁾ 3.0E-04 ⁽¹⁾ 5.0E-02 ⁽²⁾ 5.0E-03 ⁽¹⁾ 5.0E-04 ⁽¹⁾ NA 1.0 ⁽²⁾ NA NA NA NA	1.8A ⁽¹⁾⁽⁵⁾ 4.3A ⁽¹⁾ NA NA NA NA NA
Antimony Arsenic Barium Beryllium Cadmium Calcium Chromium (+3)(6) Cobalt Copper Iron Lead Magnesium	4.0E-04 ⁽¹⁾ 3.0E-04 ⁽¹⁾ 5.0E-02 ⁽²⁾ 5.0E-03 ⁽¹⁾ 5.0E-04 ⁽¹⁾ NA 1.0 ⁽²⁾ NA NA NA NA NA NA	1.8A ⁽¹⁾⁽⁵⁾ 4.3A ⁽¹⁾ NA NA NA NA NA NA
Antimony Arsenic Barium Beryllium Cadmium Calcium Chromium (+3)(6) Cobalt Copper Iron Lead Magnesium Manganese	4.0E-04 ⁽¹⁾ 3.0E-04 ⁽¹⁾ 5.0E-02 ⁽²⁾ 5.0E-03 ⁽¹⁾ 5.0E-04 ⁽¹⁾ NA 1.0 ⁽²⁾ NA NA NA NA NA NA NA NA NA NA	1.8A ⁽¹⁾⁽⁵⁾ 4.3A ⁽¹⁾ NA NA NA NA NA NA

• •

TABLE 6-1 TOXICITY CRITERIA FOR CHEMICALS OF CONCERN SELECTION NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS PAGE FOUR

Chemical	Reference Dose Oral Route of Exposure (mg/kg/day)	Cancer Slope Factor Oral Route of Exposure (mg/kg/day) ⁻¹
Nickel	2.0E-02 ⁽¹⁾	
Potassium	NA	NA
Selenium	5.0E-03 ⁽¹⁾	
Silver	3.0E-03 ⁽²⁾	
Sodium	NA	NA
Vanadium	7.0E-03 ⁽²⁾	
Zinc	2.0E-01 ⁽²⁾	
Thallium	7.0E-05 ⁽²⁾	

- NA Neither RfDs or CSFs are available for these chemicals.
- (1) U.S. EPA Integrated Risk Information System (12-3-91).
- ⁽²⁾ U.S. EPA Health Effects Assessment Summary Tables, Annual FY-1991 (EPA, January 1991).
- ⁽³⁾ U.S. EPA, ECAO provisional RfD for Nyanza.
- ⁽⁴⁾ U.S. EPA Drinking Water Health Advisory Memorandum (EPA, April 1991).
- ⁽⁵⁾ Based on a unit risk of 5 x $10^{-5}/\mu g/L$.
- ⁽⁶⁾ Sediment samples collected at the Nyanza Site were analyzed for Total Chromium and hexavalent chromium. Hexavalent chromium was <u>not</u> detected in the sediments; thus, the RfD for trivalent chromium is used in this risk assessment.
- According to EPA Region I guidance, the reference dose for naphthalene is used as a surrogate for the noncarcinogenic PAH compounds.
- ** According to EPA Region I guidance, a 5.8 (mg/kg/day)⁻¹ cancer slope factor is used to evaluate carcinogenic PAHs.

A blank space indicates that either a CSF or RfD is not available for this compound.

tion System (IRIS) (EPA

FINAL

EPA's Integrated Risk Information System (IRIS) (EPA, December 1991). Toxicity criteria available from EPA Region I, the EPA Health Effects Assessment Summary Tables (HEAST) FY 1991, and the EPA Drinking Water Health Advisory Memorandum (EPA, April 1991) were used when toxicity criteria were not available on IRIS.

The concentration-toxicity screen uses the following expression to develop a risk factor for each chemical:

 $Ri = Ci \times Ti$

Where:

- Ri = Risk factor for chemical i in the environmental medium
- Ci = Maximum concentration of chemical i in the environmental medium
- Ti = Toxicity value for chemical i in the environmental medium (i.e., either the Cancer Slope Factor (CSF) or 1/[Reference Dose (RfD)])

The ratio of the risk factor for each chemical to the total risk factor (ERi) for all chemicals is designated the relative risk for that chemical.

Tables 6-2, 6-3, and 6-4 present the results of the toxicity screen and the contaminant occurrence and distribution statistics (range of positive detections, number of positive detections/number of samples) that support the selection of COCs listed on Table 6-1. As a general guidance, Study Area chemicals meeting the following criteria were selected as chemicals of concern:

- Chemical demonstrated a relative risk factor greater than
 0.001 and was detected more than once in an environmental medium.
- Chemical demonstrated a relative risk factor greater than
 0.0001 and was detected in at least five percent of the environmental samples.

Additionally, any chemical listed as a Class A (Human Carcinogen) or Class B-1 (Probable Human Carcinogen: Limited evidence of carcinogenicity in human from epidemiologic studies) carcinogen in IRIS was selected as a COC. The EPA has identified the following chemicals as Nyanza sitespecific contaminants:

ο	Trichloroethene	0	Phenols
0	1,2-Dichloroethene	ο	Benzidene
ο	Chlorobenzene	0	Antimony
0	Nitrobenzene	ο	Cadmium
ο	The Dichlorobenzenes	ο	Chromium
ο	1,2,4-Trichlorobenzene	ο	Arsenic
ο	Aniline	0	Lead
ο	Naphthalene	ο	Mercury

These contaminants are considered to be site related. They were detected at the Nyanza Site and/or in groundwater underlying or downgradient of the site. All Nyanza site-specific contaminants were selected as COCs.

The following subsections provide a detailed discussion of the COC selection rationale.

6.2.1.1 Volatile Organic Chemicals of Concern

According to CLP analytical results, the following Class A carcinogens were detected in environmental media sampled in the Study Area:

Chemical	Carcinogen Class	Surface Water Statistics	Sediment Statistics ⁽¹⁾	Fish Statistics
Vinyl chloride	A	ND	2.7%	ND
Benzene	A	ND	2.7%	ND

ND: Not detected.

⁴⁴ Number of positive detections/number of samples collected, expressed as a percentage.

Although these chemicals were detected infrequently, benzene and vinyl chloride are selected as COCs since they are both known human (Class A) carcinogens.

The following volatile organics were selected as COCs because they were detected in at least five percent of the samples analyzed and demonstrated a relative risk value greater then 0.0001.

Chemical	Surface Water Statistics ⁽¹⁾	ERi	Sediment Statistics ⁽¹⁾	ERi	Fish Statistics ⁽¹⁾	ER i
1,2-Dichloroethene	11%	3x104(N)	26%	5.8x10*(N)	HØ	•
Trichloroethene	8.3%	3.52x10 ³ (C)	34%	1.3x104(C)	ND	-
Chiorobenzene	ND	-	32%	1.81x10 ³ (N)	ND ND	-
Hethylene chloride	ND	-	9%	3x10*(C)	6.5%	5.21x10 ³ (C)
Acetone	ND	•	342	3x10 ⁵ (N)	13%	1.4x10 ⁻⁴ (N)

N = Noncarcinogenic relative risk value.

C = Carcinogenic relative risk value.

ERi = Relative Risk

⁽¹⁾ Number of positive detections/number of samples collected, expressed as a percentage.

It should also be noted that 1,2-dichloroethene, trichloroethene, and chlorobenzene have been identified by the EPA as Nyanza site-specific contaminants.

1,1-Dichloroethene is selected as a COC because the relative risk calculated for this chemical (ERi = 4.43×10^{-2} , surface water) exceeds 1×10^{-3} and it was detected in surface water samples twice. However, it should be noted that the maximum concentration of 1,1-dichloroethene in surface waters was detected at a background sample location.

Based on the selection criteria previously discussed, the following volatile organic chemicals were not selected as COCs:

- o Chloromethane
- o 2-Butanone (methyl ethyl ketone)
- o Toluene
- o Styrene
- o Total xylenes

The occurrence and distribution statistics and toxicity screen results that support the decision not to select these chemicals as COCs are presented in Tables 6-2, 6-3, and 6-4.

6.2.1.2 <u>Semivolatile Organic Chemicals of Concern</u>

Semivolatile organic chemicals (semi-VOCs) detected in environmental media samples collected in the Study Area included several polyaromatic hydrocarbons (PAHs), chlorinated benzene compounds, nitrobenzene, phenolics, and phthalates. As detailed in Section 4, these compounds were particularly prevalent in sediments.

None of the semi-VOCs detected in the Study Area are classified as Class A or B-1 carcinogens. The carcinogenic PAHs are classified as B-2 carcinogens (probable human carcinogen: sufficient evidence of carcinogenicity in animals; inadequate evidence of carcinogenicity in humans).

The following semi-VOCs were selected as COCs because they were detected in at least five percent of the samples analyzed and demonstrated a relative risk value greater than 0.0001 (Note: Some chemicals have both cancer and non-cancer toxicity values):

TABLE 5-2 SELECTION OF CHEMICALS OF CONCERN FOR SEDIMENTS NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

	No of	Range of Positive Detects [Avg.]		Background		Τοχιεί	ly Screen	Ĩ	Location of		Comments	
Parameter	Detections No of			Concentration Ra [Avg.]	ange	Noncard	inogens	Carcinogens/WOE		Maximum Positive		Selected as a COC?
	Samples Collected	(mg/kg	5)(1)	(mg/kg)		Ri	ERi	Ri	ERi	Detection		
VOLATILE ORGANIC CHEMICALS												
Methylene chloride	רוור	0 004 - 1 6	(0 133)	ND		2.67x10-1	3×10-5	1 2x10-2	3x10-5,82	Eastern Wet.	No**	Selected for fish.
Vinyl chloride	2/74	0 006 - 0 13	[0 068]°	ND				2.47x10-1	6 4x10-4,A	Eastern Wet.	Yes	Class A carc
Acetone	26/77	0 010 - 2 6	[0 304]	0 01 (0	0 0 1]*	2 6x10 - 1	3x10-5			Reach 4	No**	Selected for fish.
1,2 Dichloroethene	19/73	0 0003 - 5 5	[0 21]	ND		5 5x10+2	5.8x10.4			Eastern Wet.	Yes	Nyanza***
2-Butanone (methyl ethyl ketone)	15/69	0 017 - 0 45	[0 127]	ND		9	1x10 ⁻⁵			Eastern Wetlands	No	Low toxicity score
Trichloroethene	25/74	0 0003 - 44	(1 755)	ND				4.84x10-1	1.3x10 ⁻³ ,82	Raceway	Yes	Nyanza***
Benzene	2/74	0 002 - 0 004	[0 003]*	ND				1 16x10-4	<1x10-5,A	Eastern Wet	Yes	Class A carc.
Tetrachloroethene	2/74	0 0008 - 0 002	[0 0014]*	ND		2x101	<1x105	1.02x10-4	<1x10 ⁻⁵ ,82	Culvert	No	Low concentration
Toluene	4/74	0 0004 - 0 016	{0 0064}*	ND		8×10-2	< 1x 10.5			Eastern Wet	No	Low toxicity score
Chlorobenzene	24/74	0 003 - 34	[2 22]	ND		1 7x10+3	1 81x10-3			Eastern Wet	Yes	Nyanza***
Ethybenzene	1/74	0 0007	10 0007]*	ND		7x103	< 1x 10 5			Eastern Wet.	No	Low toxicity score
Total xylenes	3/74	0 001 - 0 005	[0 0027]*	ND		2.5x10-3	<1x10.5			Eastern Wet.	No	Low toxicity score

Phenol	2/74	0 056 - 0 068	[0 062] .	ND	1 13x10-1	< 1x10 5			Eastern Wet.	No**	Selected for fish.
1,3 Dichloro benzene	11/74	0 018 - 0 29	(0 289)	ND	9 67	<1x105			Eastern Wet.	Yes	Nyanza***
1,4 Dichlorobenzene	19/74	0.048 - 3.1	[0 695]	ND			7 44x10 2	1 9×10 4,82	Eastern Wet.	Yes	Nyanza***
1,2 Dichlorøbenzene	25/74	0 033 - 13	[1 875]	ND	1 44x10-2	1 5x10 4			Eastern Wet.	Yes	Nyanza***
3/4-Methyl phenol	3/74	0 074 - 0 26	[0 154]*	ND	52	1x10-5			Eastern Wet.	No**	Selected for fish.
2 chlorophenol	1/74	0 0 2 0	10 0201-	ND	4	< 1x10 5			Eastern Wet.	No	Low det freq
Nitro benzene	12/74	0 065 - 0 65	10 2051*	ND	1 3x10-3	1 38x10 3			Eastern Wet.	Yes	Nyanza***

(

6-10

•

(

,

TABLE 6-2 SELECTION OF CHEMICALS OF CONCERN FOR SEDIMENTS NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS PAGE TWO

FAGE ING												
	No of Positive	Range	ot	Background			Τοχιει	ty Screen		Location of		
Parameter	Detections/ No of	Positive D JAvg	Positive Detects [Avg.]		n Range }	Noncard	linogens	Carcino	Carcinogens/WOE		Selected as a COC?	Comments
	Samples Collected	{mg/1g		(mg/k)	g) 	Ri	ERI	Ri	ER	Detection		
SEMIVOLATILE ORGANIC CHEMICALS (CONTINUED)												
Benzoic acid	22/74	0 021 - 1 6	[0 332].	ND		4 0x10-1	< 1x10 ⁻⁵			Eastern Wet.	No	Low toxicity score
1,2,4 Trichlorobenzene	20/74	0 042 3 1	[0 579]	ND		2 37=10+3	2 52×10-3			Eastern Wet.	Yes	Nyanza***
Napthalene	11/74	0 082 - 6 7	{0 785}	ND		1 68×10+3	1 76×10-3			Raceway	Yes	Nyanza***
2-Methyl napthalene	5/74	0 06 - 0 41	{0 396}	ND						Reach 2	Yes	No toxicity criteria
Acenapthylene	14/74	0 021 - 0 88	[0 425]	ND		2.2 x 10+2	2.3x10-4			Reach 3	Yes	(2)
Acenapthene	7/74	0 043 1 235	[0 6 36]	ND		2 06×10 + 1	2x105			Reach 2	No	Low toxicity score
Dibenzofuran	3/74	0 15 1 06	(0 503)	ND		2 65×10+2	2 8x10 4			Reach 2	No	Low det freq
Diethyl phthalate	2/74	0.06 - 0.66	{0 382}	ND		8 25x10-1	< 1x10 5			Outfall Creek	No	Low toxicity score
Fluorene	16/74	0.025 - 2.8	[0 754]	ND		7x10+1	7x10-5			Raceway	No	Low toxicity score
N-nitrosodiphenylamine	20/74	0 01 0 29	10 1041-	ND				1.42x10-3	< 1 x 10·5, B2	Outfall Creek	No	Low toxicity score
Phenanthrene	47/77	0 023 - 16	[1 353]	0 14 - 0 16	[0.15]*	4x10+3	4 5x10-3			Raceway	Yes	(2)
Anthracene	25/75	0 021 - 3 1	(0 76)	ND		1 03x10+1	1x10-5			Raceway	No	Low toxicity score
Di-n-butyl phthalate	16/74	0 015 - 5 2	[0 545]	ND		5 2x10+1	6x10-5			Reach 3	No	Low toxicity score
Fluoranthene	62/77	0.01 - 20	[1 647]	0 067 - 0 32	[0 219]*	5x10+2	5 3x10-4			Raceway	Yes	(2)
Pyrene	43/77	0 025 - 13	[1 481]	0 052 - 0 27	[0 181]*	4 33x10+2	4 6x10-4			Raceway	Yes	(2)
Butyl benzyl phthalate	7/74	0 0 26 0 63	[0 582]	ND		3 15	< 1x10 5			Reach 5	No	Low toxicity score
Benzo (a) anthracene	35/76	0 055 11	(0 963)	ND				6 38x10+1	0.17,B2	Raceway	Yes	(2)
Chrysene	41/77	0 062 - 8 6	[1 099]	0 14 - 0 18	[0 16]*			4 99x10+1	0.13,82	Raceway	Yes	(2)
Bis(2-ethylhexyl) phthalate	25/74	0 009 2 5	[0 78]	ND		1 25x10+2	1 3x10-4	3 5×10-2	9x10 ⁻⁵ ,82	Raceway	Yes	(2)

.

ł

TABLE 6-2 W92194F

SELECTION OF CHEMICALS OF CONCERN FOR SEDIMENTS

NYANZA OPERABLE UNIT 3

MIDDLESEX COUNTY, MASSACHUSETTS

PAGE THREE

Parameter	No of Positive Range of Detections/ Positive Detects No of [Avg]	Background		Τοχιείτ	y Screen		Location of		
		Positive Detects [Avg]	Concentration Range [Avg] (mg/kg)	Noncarcinogens Carcinogens/WOE		Maximum Positive	Selected as a COC?	Comments	
	Samples Collected	(mg/kg)(*)		Ri	ERi	Ri	ERi	Detection	

SEMIVOLATILE ORGANIC CHEMICALS (CONTINUED)

Di-n-octyl phthalate	5/74	0 041 - 0 22	[0 108]*	ND		1.1x10+1	1x10-5			Reach 6	No	Low toxicity score
Benzo(b)fluoranthene	39/77	011-52	[1 421]	0 17 - 0 20	[0 185]*			3 02x10+1	0.08, 82	Raceway	Yes	(2)
Benzo(k)fluoranthene	40/75	0 041 - 6 3	[0 949]	ND				3 65x10+1	0.09, 82	Raceway	Yes	(2)
Benzo(a)pyrene	34/75	0 072 - 4 7	[0 897]	0 091	[0 091]*			2 73x10+1	0 07, 82	Raceway	Yes	(2)
Indeno(1,2,3-cd)pyrene	18/74	0 051 - 1 8	[0 401]	ND				1 04x10+1	0.027, B2	Reach 3	Yes	(2)
Dibenz(a,h)anthracene	5/74	0 093 - 0 31	[0 147]*	ND				1.8	4 67x10 ³ , B2	Reach 3	Yes	(2)
Benzo(g,h,i)perylene	19/74	0 055 - 1 6	[0 786]	ND		4x10+2	4 2x10-4			Reach 3	Yes	(2)

PESTICIDES AND POLYCHLORINATED BIPHENYL COMPOUNDS

DDE	4/74	0.06	[0 022]	ND			2.04x10-2	5x10-5,82	Reach 3	No	Selected for fish.
DDD	8/74	0 053 - 0 7	[0 104]	ND			1.68x10-1	4 4x10 ⁻⁴ , B2	Reach 6	Yes	(2)
DDT	4/75	0 071 - 1 4	[0 172]	ND	2.8x10+3	2.98x10-3	4.76x10-1	1.24x10 ⁻³ , B2	Reach 2	Yes	(2)
Gamma chlordan e	1/74	0.018	[0 018]*	ND	3.0x10+2	3.2x10.4	2.34x10-2	6x10 ⁻⁵ , 82	Reach 7	No	Low det. freq.
Arochlor 1254	3/74	0 112 - 0 589	[0 404]*	ND			4.54	1.18x10 ⁻² , B2	Raceway	Yes	(2)

METHYLATED MERCURY COMPOUNDS

Monomethyl mercury	5/46	0 026 - 0 312	(0 082)	ND	1 04x10+3	1 11x10-3		Reach 2	Yes	Nyanza***
Dimethyl mercury	1 /46	0 019	[0 007]	ND	6 33×10+1	7x10-5		Reach 3	Yes	Nyanza***

TABLE 6-2 SELECTION OF CHEMICALS OF CONCERN FOR SEDIMENTS NYANZA OPERABLE UNIT 3

MIDDLESEX COUNTY, MASSACHUSETTS

W92194F PAGE FOUR

	No of	Range.					Τοχιεί	ly Screen		Location of		
Parameter	Detections/ No. of	Positive D Avg	etects 1	Range [Avg		Noncard	tinogens	Carcinog	gens/WOE	Maximum Positive	Selected as a COC?	Comments
	Samples Collected	(mg/i-g		(mg/k)	(mg/kg)		ERi	Ri	ERi	Detection		
METALS												
Aluminum	168/177	1.840 - 22,900	[10,339 1]	2,870 - 16,600	[9200 9]					Reach 4	No	(3)
Antimony	3/154	67-18	[4 05]	69	[6 1]	4 5x10-4	0 047			Reach 3	Yes	(2)
Arsenic	134/177	0 75 - 64 6	(9 5)	25-211	[8 7]	2 13x10+5	0.22	1 15x10+2	03, A	Reach 9	Yes	Nyanza***
Barium	175/177	11.4 - 272	[67 1]	13 3 - 178	[66 7]	5 44×10+3	5.79x10-3			Reach 7	Yes	(2)
Beryllium	98/157	01-1015	[1 5]	0 37 - 1 8	[0 85]	2 03x10+3	2.16x103	4 36x10+1	0 11, A	Raceway	Yes	(2)
Cadmium	27/160	13-199	[4 3]	ND		3 98×10+4	0 042			Reach 3	Yes	Nyanza***
Calcium	176/177	326 18 400	[31367]	588 - 4,020	[2161 9]					Eastern Wet.	No	(3)
Chromium (III)	153/177	4 8 - 2.620	(98 8)	66.552	[22 4]	2 62x10+3	2 8×10-3			Reach 3	Yes	Nyanza***
Cobalt	101/167	2 6 - 235	[13 6]	37-188	[10 7]					Raceway	No	(3)
Copper	131/177	8 1 - 454	[99]	89-3404	[73 0]					Reach 3	Yes	No toxicity criteria
Iron	170/77	1.980 110.000	16,739.7)	4,110 - 110,000	(22,124-1)					Background	No	(3)
Lead	166/177	2 4 - 5,760	[159 3]	5 - 248	[85 1]					Eastern Wet.	Yes	Nyanza***
Magnesium	176/177	195 - 5,030	[2,267 8]	762 0 - 3,280	[1877 8]		•			Reach 3	No	(3)
Manganese	174/177	28 4 5,860	[475 3]	65 8 · 1,640	[434 5]	5 86x10+4	0 062			Reach 8	Yes	(2)
Mercury	124/164	0 13 - 152	[7 8]	0 5 - 1 59	[0 27]	5 07x10+5	0 539		·	Eastern Wet	Yes	Nyanza***
Nickel	85/174	2 3 - 186	[19.2]	3 2 - 51	[11 4]	9 3x10 · 3	0.01			Raceway	Yes	(2)
Potassium	147/177	92.2 2.260	[696 7]	224 - 1,355	[672 6]					Reach 3	No	(3)
Selenium	43/156	03-72	[1 33]	29-31	[1 0]	1 44x10+3	1 5x10 3		·	Reach 7	Yes	(2)
Silver	1/125	64	[1 9]	ND		2 13x10+3	2.27x103			Reach S	No**	Selected for Surface Water

TABLE 6-2

SELECTION OF CHEMICALS OF CONCERN FOR SEDIMENTS

NYANZA OPERABLE UNIT 3

MIDDLESEX COUNTY, MASSACHUSETTS

PAGE FIVE

6**M**

Ñ

4F

	No of Positive	Range of	Background		Toxicity	y Screen		Location of		
Parameter	Detections/ No. of	Positive Detects [Avg.]	Concentration Range [Avg.]	Noncarcinogens Carcinogens		Carcinogens/WOE		Maximum Positive	Selected as a COC?	Comments
	Samples Collected	(mØ/rØ)(+)	(mg/kg)	Ri	ERi	Ri	ERi	Detection		

METALS (CONTINUED)

Sodium	97/177	20 2 - 4,630	[384 6]	389 · 72	[186.2]				Reach 3	No	(3)
Vanadium	166/176	46-83	[28 7]	48 45 - 10 2	[20 2]	1 19x10+4	0.012		Reach 6	Yes	(2)
Zinc	152/177	10 5 - 765	[157 3]	629 - 13 5	[133 5]	3 83x10+3	4 07x10-1		Reach 5	Yes	(2)
Thallium	2/138	12-14	[0 46]	ND		2x10+4	0 02 1		Eastern Wet	Yes	(2)

ND = Not detected

Avg z Average

Ri = Risk factor

ERi # Ratio of risk factor for chemical to total of risk factors for all chemicals

WOE = Carcinogenic weight of evidence

COC = Chemical of concern

6-14

(1) Range of positive detections and average concentration for sediment samples collected during the RI (background samples as well as samples downstream of the Nyanza Site). The average concentration presented is calculated using one half the sample quantitation limit (or one half the contract required detection limit) for nondetect values. Data for all reaches with at least one positive detection were used to calculate the average. If the average is greater than the maximum detected concentration, the average of positive detections(*), only, is presented.

(2) Chemical meets COC selection criteria

(3) No toxicity criteria is available for this chemical. With the exception of aluminum and cobalt, these chemicals are macronutrients.

** Although this chemical does not meet the selection criteria for a COC in sediments, it is listed as a COC for surface waters and/or fish. Conservatively, the chemical will be evaluated in the risk assessment.

*** This chemical has been identified by the EPA as a Nyanza site specific contaminant

				SELECTION OF MIE	CHEMICALS NYANZA DDLESEX CO	TABLE 6-3 OF CONCERN OPERABLE UI UNTY, MASSA	FOR SURFA	CE WATER				
	No of Positive	Rar	nae of	Backor	ound	T	Τοχια	ity Screen		Location of		
Parameter		Positiv	e Detects	Concentrat	ion Range	Noncar	Noncarcinogens Carcino		gens/WOE	Maximum Positive	Selected	Comments
	Collected	lu	g/L} ⁽¹⁾	lug	/1)	Ri	ERi	Ri	ERi	Detection		
VOLATILE ORGANIC CHEM	HCALS											· · · · · · · · · · · · · · · · · · ·
Chloromethane	1/36	11	[5 9]	11	[5.9]			1.43x10-1	3 52×10 B2	Background	No	(2) Low det. freq.
1,1-Dichloroethene	2/36	3	[2 6]	3	[2 6]	3.33x10+2	8x105	1.8	4.43x10-2 C	Background	Yes	(2)(3)
1,2-Dichloroethene	4/36	1 12	(3.7)	ND		1 2x10+3	3x10-4			Outfall Creek	Yes	Nyanza***
2-Butanone (methyl ethyl ketone)	1/32	10	(7 5)	ND		2x10+2	5x10-5			Reach 4	No	Low det. freq. Low toxicity score
Trichloroethene	3/36	2 - 13	(6)	ND				1 43x10-1	3 52x10-3 82	Outfall Creek	Yes	Nyanza***
SEMIVOLATILE ORGANIC C	HEMICALS											······································
1,4 Dichlorobenzene	1/36	1	[1]	ND		T		2.4x10-2	5.9x10-4C2	Outfall Creek	Yes	Nyanza***
1,2 Dichlorobenzene	2/36	3 4	(3 5)*	ND		4 44x10+1	1x10·5			Outfall Creek	Yes	Nyanza***
Bis(2-ethyl, hexyl) phthalate	3/36	1 - 58	[11-4]	ND		2 9x10+3	7 2x10-4	8 12x10-1	2x10-282	Cold Spring Brook	Yes	(3)
PESTICIDES												
Gamma BHC (Lindane)	1/36	0 02	[0 02]	ND		6.67x10+1	2x10-5	2.6x10-2	6.4x10-4B2	Reach 6	No	Low det freq.
METALS												
Aluminum (1) (F)	19/36 10/26	82 - 3,220 41 - 174	[254] [63 6]	209 - 231 ND	[84]					Reach 2	No	(4)
Antimony (T) (F)	0/36			ND						ND	No**	Selected for sediments
Arsenic (T) (F)	2/36 2/26	1 1 - 3 1 2 - 50 9	[2 05]* [10 5]	ND ND		1x10+4	2 48x10-3	5.4	0.133 A	Reach 2	Yes	Nyanza***
Barium (T) (F)	23/36 18/26	7 6 - 132 5 5 - 33 6	(15 02) [15 0]	98-23.1 148-197	[15.7] [17 5]	2.64x10+3	6 5x10 ^{.4}			. Reach 2	Yes	(3)
Jeryllium (T) (F)	1/36 0/26	75	[2 61	ND ND		1 5x10+3	3 7x10-4	3 23x10+1	0.794 A	Reach 2	Yes	(3)

1

W92194F

6-15

(

FINAL

7

.

TABLE 6-3

SELECTION OF CHEMICALS OF CONCERN FOR SURFACE WATER NYANZA OPERABLE UNIT 3

(

MIDDLESEX COUNTY, MASSACHUSETTS PAGE TWO

W92194F

6-16

		No of Positive	Range	01	Backgrou	nd		Toxicity	Screen		Location of		
Parameter	r	Detections/ No. of Samples	Positive D [Avg	etects }	Concentration [Avg.]	Range	Noncard	inogens	Carcino	gens/WOE	Maximum Positive	Selected as a COC?	Comments
		Collected	(µg/L)	(U)	(µg/L)	(ug/L)		ERi	Ri	ERi	Detection		
METALS (CONTINU	IED)			_									
Cadmium	(T) (F)	1/36 0/26	72	[2 8]	ND ND		1 44x10+4	3 57x10-3			Reach 2	Yes	Nyanza***
Calcium	(T) (F)	36/36 26/26	6.140 - 17,700 6.240 - 17,900	[9822 2] [10,546 5]	6,200 - 12,400 9,680 - 12,400	[10,217] [11,676]					Outfall Creek	No	(4)
Chromium	(T) (F)	4/36 0/26	56-79	{17 5}	ND ND		7 9x10+1	2x 10 ⁻⁵			Eastern Wet.	Yes	Nyanza***
Cobalt	(T) (F)	2/36 0/26	23-277	[22 5]	ND ND						Reach 2	No	(4)
Copper	(T) (F)	3/36 0/26	3 1 - 27 7	(9 3)	3 1 ND	[3.1]*					Reach 2	Yes -	No toxicity criteria
Iron	(T) (F)	27/36 16/26	311 - 16,100 281 - 860	{1206 73} [384 5]	750 - 1,640 352 - 502	[765 5] [410]					Reach 2	No	(4)
Lead	(T) (F)	11/36 3/26	10-665 15-87	[5 2] [4 4]	21 1 ND	[4]					Reach 2	Yes	Nyanza***
Magnesium	(T) (F)	36/36 26/26	1,130 - 3,030 1,290 - 2,960	[21376] [21927]	1,610 - 2,900 2,290 - 2,960	[2,450] [2,776]					Reach 2	No	(4)
Manganese	(T) (F)	33/36 21/26	48 6 - 9,840 24 6 - 208 5	(372 5) [76 7]	75 8 - 111 24 6 - 87	[80 8] [55 7]	9 84x10+4	2 44x10-2			Reach 2	Yes	(3)
Mercury	(T) (F)	3/32 3/26	0 37 - 3 8 0 37 - 0 49	{1 6} [0 43]	ND ND		1 27×10+4	3.14x10-3			Eastern Wet.	Yes	Nyanza***
Nickel	(T) (f)	2/36 1/26	177-774 70	[13 6] [3 8]	177	(5 1) (3 8)	3 87×10+3	9.6x10 ⁻⁴			Reach 2	Yes	(2)(3)

(

.

.

TABLE 6-3 SELECTION OF CHEMICALS OF CONCERN FOR SURFACE WATER NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS PAGE THREE

W92194F

	No of Positive	Range of	Background		Toxicity	y Screen		Location of		
Parameter	Detections/ No of Samples	Positive Detects [Avg]	Concentration Range [Avg.]	Noncarc	inogens	Carcino	gens/WOE	Maximum Positive	Selected as a COC?	Comments
	Collected)	(µg/L)(1)	(ug/L)	Ri	ERi	Ri	ERi	Detection		

METALS (CONTINUED)

Potassium	(1) (F)	36/36 26/26	864 - 3,100 829 - 3,010	[2013 3] [2053]	1,810 - 3,100 2,470 - 3,010	[2481 4] [2,646]				Background	No	(4)
Selenium	(T) (F)	1/36 0/26	19,300	[3,861 5]	ND ND		3 86x10+6	9.57x10 ⁻¹		Reach 3	No**	Selected for Sediments.
Silver	(T) (F)	7/36 1/26	65-689 8	[6 8] (3 9]	16 9 ND	[4 74]	2.3x10+4	5.69x10-3		Reach 2	Yes	(3)
Sodium	(T) (f)	35/36 26/26	6,080 - 38,200 2,100 - 39,000	{24,378 6} {25,593 5}	16,000 - 38,200 26,600 - 39,000	[28,887] [34,140]				Background	No	(4)
Thallium	(T) (F)	0/36			ND					ND	No**	Selected for Sediments.
Vanadium	(T) (F)	2/36 0/26	12 6 - 16 1	[14-4]	ND ND		2.3x10+3	5.7x10-4		Reach 2	Yes	(3)
Zinc	(T) (F)	4/36 6/26	6 2 - 125 10 - 46 1	[16 1] [18 5]	62	{6.2]* {12.1}	6 25x10+2	1.5x10 ^{.4}		Reach 2	Yes	(3)

Ŧ = Total (unfiltered) metals results

= Filtered metals results

ND = Not detected

= Average Avo

= Risk factor Ri

ERi = Ratio of risk factor for chemical to total of risk factors for all chemicals

WOE = Carcinogenic weight of evidence

COC = Chemical of concern

(3) Chemical meets COC selection criteria.

(4) No toxicity criteria is available for this chemical. With the exception of cobalt and aluminum, these chemicals are macronutrients.

.. Although this chemical does not meet the selection criteria for a COC in surface waters, it was selected as a COCs for sediments or fish. Conservatively, the chemical will be evaluated in the risk assessment.

*** This chemical has been identified by EPA as a Nyanza site-specific contaminant.

(1) Range of positive detections and average concentration for surface water samples collected during the RI (i.e., background samples as well as samples downstream of the Nyanza Site). The average concentration presented is the average calculated using one half the sample quantitation limit (or one half the contract required detection limit) for nondetect values. Data for all reaches with at least one positive detection were used to calculate the average. If the average is greater than the maximum detected concentration, the average of positive detections(*), only, is presented.

(2) Chloromethene, 1, 1-Dichloroethene, and nickel (filtered) were detected at background sample locations, only

TABLE 6-4
SELECTION OF CHEMICALS OF CONCERN FOR FISH (FILLETS)
NYANZA OPERABLE UNIT 3
MIDDLESEX COUNTY, MASSACHUSETTS

Ξ
Q
N
Ч
9
4
ㅋ

No. of Positive Detections/	Range of Positive Detects		Sudbury Reservoir		Τοχιζι	ty Screen		Location of			
Parameter	Detections/ No of Samples	Poulive D. [Avg	riecis J	Concentration Range	Noncard	inogens	Carcino	igens/WOE	Maximum Positive	Selected as a COC?	Comments
ļ	Collected	(mg/) (1)	91	(mg/kg)	Ri	ERi	Ri	ERi	Detection		
VOLATILE ORGANIC CHEN	1ICALS										••••••••••••••••••••••••••••••••••••••
Methylene chloride	3 of 45	0.028 14	[0 507]		2 33x10+2	2.02×10-3	1.05×10-1	7.27x10-3 82	Fairhaven Bay	Yes	(2)
Acetone	6 of 45	0 810 - 1 6	[0 398]		1 6x 10 + 1	1 4x10 ^{.4}			Fairhaven Bay	Yes	(2)
2-Butanone	2 of 45	0 250 - 0 3	[0 041]		6	5x10-5			Fairhaven Bay	No	Low toxicity score
Toluene	2 of 45	0 005 - 0 026	{0 013}		1 3x10 1	< 1x10.5			Fairhaven Bay	No	Low toxicity score
Styrene	1 of 45	0 002	{0 003}*		1x10-2	< 1 x 10 5	6x10-5	< 1x10.5 82	Fairhøven Bay	No	Low toxicity score
Total Xylenes	1 of 45	0 002	[0 005].		1x10-3	< 1x10.5			Fairhaven Bay	No	Low toxicity score
SEMIVOLATILE ORGANIC	CHEMICALS									-	
Phenol	40 of 180	0 025 - 8 2	[0 536]	0 056 [0.056]*	1 37x10+1	1 2x10 ⁻⁴			Fairhaven Bay	Yes	(2)
Bis(2 chloroethyl)ether	1 of 191	0.04	 0 04}*				4.4x10-2	3.05x10 382	Fairhaven Bay	No	Low det. frequency
Benzyl alcohol	5 of 186	0 026 - 0 42	[0 120]		14	1x10-5			Fairhaven Bay	No	Low toxicity score
2-Methyl phenol	1 of 177	0 033	[0 033]-		6 6x10 1	1x10 ⁻⁵			Fairhaven Bay	No	Low toxicity score
4-Methyl phenol	22 of 177	0 026 - 1 9	[0 139]		3.8x10+1	3 3x10-4			Fairhaven Bay	Yes	(2)
Nitro benzene	1 01 191	0 024	[0 024]		4 8x10+1	4 2x10-4			Fairhaven Bay	Yes	Nyanza***
Napthalene	1 of 191	19	[0 129]		4 75x10+2	4 12x10-3			Saxon. Res	No**	Selected for sediments
Diethylphthalate	1 of 191	0 033	(0 033)*		4 13x10-2	<1x10.5			Fairhaven Bay	No	Low toxicity score
fluorene	1 of 191	0 099	(0 099]*	0 099 [0.099]*	2.48	2x10.5			Sudbury Res.	No	Low toxicity score
Di-n-butyl phthalate	17 of 191	0 021 - 0 280	{0 23}	0 045 [0 045]*	28	2x10 5			Fairhaven Bay	No	Low toxicity score
Butyl benzyl phthalate	8 of 191	0012-30	[0 249]		1 5x10+1	1 3×10 4			Fairhaven Bay	No	Low det. frequency
Bis(2·ethyl hexyl) phthalate	32 of 145	0 028 - 3 3	10 41		1 65x 10 + 2	1 43x10 3	4 62x10-2	3 2x10 382	Reservoir 1	Yes	(2)

(

FINAL

•

•	•	E	c

TABLE 6-4 SELECTION OF CHEMICALS OF CONCERN FOR FISH (FILLETS) NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS PAGE TWO

W92194F

PAGE INVO												
	No of Positive	Range	of	Sudbury R	eservoir		Toxici	ty Screen		Location of		
Parameter	Detections/ No of Samples	Positive De [Avg]	precis 	Concentratio [Avg	Concentration Range [Avg.]		cinogens	Carcino	gens/WOE	Maximum Positive	Selected as a COC?	Comments
	Collected	(1)	,, 	(mg/k	g)	Ri	ERi	Ri	ERi	Detection		
SEMIVOLATILE ORGANIC	CHEMICALS (CON											
Di(n-octyl)phthalate	4 of 144	0 029 - 0 046	[0 040]*			23	2x10-5			Fairhaven Bay	No	Low toxicity score
Benzo(b)fluoranthene	1 01 191	0 087	[0 078]					5 05x10-1	0.035 B2	Fairhaven Bay	No**	Selected for sediments
Benzo(a)pyrene	1 of 191	0 026	[0 036].					1.51x10-1	1.04x10-2 #2	Fairhaven Bay	No**	Selected for sediments
PESTICIDES/PC8s												
Heptachlor epoxide	1 of 191	0 004	[0 001]			3 08x10+2	2 67x10-3	3.64x10-2	2.52x10-3 82	Saxon Res.	No	Low det. frequency
Endosulfan I	1 of 191	0 0015	[0 001]							Saxon Res.	-	See total Endosulfan
Dieldrin	10 of 187	0 00005 - 0 0005	[0 0004]*	0 0005	[0.0005]*	1x10+1	9 0x 10-5	8x10-3	5 5x10-482	Saxon Res.	Yes	(2)
4,4 DDE	132 01 192	0 0005 - 0 068	(0 016)	0 0035 - 0 041	[0.019]			2.31x10-2	1 6x 10-3 #2	Reservoir 2	Yes	(2)
Endrin	5 of 188	0 0005 - 0 0075	[0 004]	0 0005	[0 0005]*	2 5x10+1	2 2x10-4			Saxon Res.	No	Low toxicity score
Endosulfan II	1 of 191	0 00 1	[0 001]							Reservoir 2	· ·	See total Endosulfan
4,4 DDD	55 of 191	0 001 - 0 02	[0 004]	0 001 - 0.007	[0.004]			4 8x10-3	3 3x10-482	Reservoir 1	Yes	(2)
Endosulfan sulfate	3 of 191	0 001 - 0 0045	10 0031	0 001 - 0 0045	[0.0027]*					Sudbury Res	-	See total Endosulfan
4,4 DDT	26 of 187	0 00005 - 0 002	[0 0012]*	0 0005 - 0 002	[0 0009]*	4	3x10-5	6.8x10-4	5x10-582	Sudbury Res.	No**	Selected for sediments
Methoxychlor	2 of 191	0 001 - 0 0035	[0 0033]			7x10-1	1x10 \$			Reservoir 2	No	Low toxicity score
Endrin ketone	1 of 191	0 0025	[0 001]							Reservoir 2	No	Low det. frequency
Alpha chlordane	14 of 191	0 00050 0025	0 001}*	0 001	[0.0001]*					Reservoir 1	-	See total chlordane
Gamma chlordane	20 of 191	0 0005 - 0 002	[0 001]*							Reservoir 2	•	See total chlordane
Aroclor 1248	1 of 191	05	[0 073]							Mill Pond	-	See total PCBs
Aroclor 1254	59 of 182	0 01 - 0 73	[0 068]				_			Reservoir 2	•	See total PCBs
Aroclor 1260	37 of 182	0 01 - 0 11	(0 036)	0 020 - 0 095	[0 064]					Saxon Res.	· ·	See total PCBs
Heptachlor	3 of 189	0 0015 - 0 006	[0 002]			1 2x10+1	1 0x10 4	2.7×10-2	1 87x10-3 82	Saxon Res.	Yes	(2)
Aldrin	2 of 190	0 0005 - 0 001	[0 0003]			3 33x10+"	2 9x10 4	1 7x10-2	1 18×10-3 82	Reservoir 2	Yes	(2)

.

1

. /

TABLE 6-4

SELECTION OF CHEMICALS OF CONCERN FOR FISH (FILLETS)

.

NYANZA OPERABLE UNIT 3

MIDDLESEX COUNTY, MASSACHUSETTS PAGE THREE

W92194F

	No of Positive Range of	Sudbury Reservoir		Toxici	ty Screen		Location of Maximum Positive		Comments	
Parameter	Detections/ No. of Samples	(Avg)	Concentration Range		inogens	Carcino		gens/WOE		Selected as a COC?
	Collected	(1)	(mg/kg)	Ni	ERi	Ri	ERÌ	Detection		

PESTICIDES/PCBs (CONTINUED)

Total PCBs	•	•	•			1.03x10+1	7.1x10-182	•	Yes	(2)
Total Endosulfan		•	•	1.6x10+2	1 4x10-3				Yes	(2)
Total Chlordane		•	•	8 33x10+1	7 2x10-3	6.5x10-3	4.5x10-482		Yes	(2)

METALS/METHYL MERCURY

Aluminum	35 of 245	12 4 - 489 93	[8 944]	13 06 - 14 69	{6 77]				T	Reservoir 1	No	(3)
Antimony	5 of 245	0 03 - 15 5	[0 448]			3 87x10+4	3 35x10 1			Reservoir 2	Yes	(2)
Arsenic	13 of 257	0 03 - 1 75	(0 078)	0.06	{0 06}*	5 83x10+3	5 06x10-2	3.16	2.19x10-1A	Reservoir 2	Yes	Nyanza***
Barium	2 of 258	0 17 - 1 1	[0 251]			2.2x10+1	1 9x10-4			Fairhaven Bay	No**	Selected for sediments
Cadmium	1 of 258	0.05	[0 030]			1x10+2	9x10-4		1	Reservoir 2	Yes	Nyanza***
Calcium	252 of 258	68 06 - 28,400	(1,513 47)	303 5 - 513.2	{400.1}					Fairhaven Bay	No	())
Chromium	103 of 258	0 14 - 8 32	[0 702]	13.28	[1.14]	8.32	7x10.5			Reservair 1	Yes	Nyanza***
Cobalt	72 of 256	0 02 - 1 26	{0 341}	0 03 - 0 08	[0 05]*		[Fairhaven Bay	No	(3)
Copper	27 of 258	0 04 - 9 72	[0 514]							Reservoir 2	Yes	No toxicity criteria
Iron	48 of 258	14 3 - 302 61	[28 996]	70 9 - 93 6	[29 0]					Reservoir 1	No	(3)
Lead	10 of 254	0 29 - 5 60	[0 085]							Fairhaven Bay	Yes	Nyanza***
Magnesium	252 of 258	160 - 592 38	{275 65}	264 7 - 519 3	[379.1]					Reservoir 1	No	(3)
Manganese	160 of 258	0 075 - 33 81	[1 604]	0 57 - 2 4	[1.06]	3 32x10+2	2 88x 10-3			Fairhaven Bay	Yes	(2)
Mercury	161 of 258	02-96	{6 B21]	0 55 - 1 20	[0 89]*	3.2x10+4	2 77x10-1			Cedar Pond	Yes	Nyanza***
Nickel	4 01 258	11-60	[0 285]			3=10+2	2 60x 10-3			Cedar Pond	Yes	(2)

TABLE 6-4

SELECTION OF CHEMICALS OF CONCERN FOR FISH (FILLETS) NYANZA OPERABLE UNIT 3

MIDDLESEX COUNTY, MASSACHUSETTS

W92194F

PAGE FOUR										
Parameter	No of Positive Detections/ No of Samples	Range of Positive Detects	Sudbury Reservoir Concentration Range [Avg.]		Toxicit	y Screen		Location of		
				Noncarcinogens Carcinogens/WOE		Maximum Positive	Selected as a COC?	Comments		
]	Collected	(1)	(mg/kg)	Ri	ERi	Ri	ERi	Detection		

METALS (CONTINUED)

				*****						Y
Potassium	252 of 258	1.890 - 33,251	{3,844.81}	3,097 5 - 6,446 9	[5,432 4]			Mill Pond	No	(3)
Selenium	79 of 258	003-40	[0 190]	18	[0 46]	8x10+2	6.93x10-3	Ced.Sw.Pond	Yes	(2)
Silver	4 of 246	0 03 - 2 2	[0 063]			7 33×10+2	6.36x10-3	Ced.Sw.Pond	Yes	(2)
Sodium	252 of 258	33 5 - 1,160	[519 86]	330 8 - 524 3	[418-1]			Saxon Res.	No	(3)
Thallium	3 of 223	0 08 - 1 35	(0 065)			1.93x10+4	1.67x10-1	Reservoir 2	Yes	(2)
Vanadium	70 of 258	0 02 - 1 57	[0 390]	0 31 - 0 76	{0 325}	2.25×10+2	1.95x10-3	Reservoir 1	Yes	(2)
Zinc	252 of 258	0 31 - 262 46	[18 401]	97-616	(32 35)	1 31x10+3	1.14x10-2	Reservoir 1	Yes	(2)
Methyl mercury	117/129	0 370 - 4 2	[0 957)	0 221 - 0 652	[0 485]	14x10+4	1.21x10-1	Reservoir 2	Yes	Nyanza***

6-21

ND = Not detected

Avg = Average

RI = Risk factor

ERi = Ratio of risk factor for chemical to total of risk factors for all chemicals

WOE = Carcinogenic weight of evidence

COC = Chemical of concern

(2) Chemical meets COC selection criteria

** Although this chemical does not meet the selection criteria for a COC in fish, it was selected as a COC for surface waters and/or sediments.

*** This chemical has been identified by EPA as a Nyanza site specific contaminant

⁽¹⁾ Range of positive detections and average concentration for fish samples collected during the RI (i.e., background samples as well as samples downstream of the Nyanza Site). The average concentration presented is calculated using one half the sample quantitation limit (or one half the contract required detection limit) for nondetect values. Data for all reaches with at least one positive detection were used to calculate the average. If the average is greater than the maximum detected concentration, the average of positive detections(*), only, is presented.

⁽³⁾ No toxicity criteria is available for this chemical. With the exception of aluminum and cobalt, these chemicals are macronutrients. Conservatively, the chemical will be evaluated in the risk assessment.

FINAL

Chemical	Surface Water Statistics ⁽¹⁾	ERi	Sediment Statistics ⁽¹⁾	ERi	Fish Statistics ⁽¹⁾	ERI
1,4-Dichlorobenzene	3%	5.9x10 ^{-41C1}	26%	1.9x10*(C)	ND	-
1,2-Dichlorobenzene	5.5%	1x10 ^{-5(N)}	33%	1.5x10 ⁻⁴ (N)	ND	-
1,2,4-Trichlorobenzene	ND	-	27%	2.52x10 ³ (N)	ND	-
Nitrobenzene	ND	-	16%	1.38x10 ⁻³ (N)	22%	4.1x10 ⁴ (N)
Phenol	ND	-	2.7%	1x10 ⁻⁵ (N)	12.4%	1.2x10*(N)
4-Methylphenol	ND	-	4%	1x10 ⁵ (N)	22.1%	3.3x10 ⁴ (N)
Bis(2-ethylhexyl) phthalate	8.3%	2x10 ⁻² (C)	34%	1.3x10 ⁴ (N)	11.2%	2.3x10 ⁻³ (C)
Carcinogenic PAHs	ND	-	(2)	(2)	(3)	(3)
Noncarcinogenic PAHs	ND	•	(2)	(2)	(3)	(3)

N = Noncarcinogenic relative risk value.

C = Carcinogenic relative risk value.

ERi = Relative Risk Value.

⁽¹⁾ Number of positive detections/number of samples collected, expressed as a percent.

⁽²⁾ See Table 6-3, for relative risk values.

See Table 6-5 for relative risk values.

As detailed on Tables 6-2 and 6-4, the following PAHs were selected as COCs:

0	Naphthalene	0	Chrysene
0	Acenaphthylene	ο	Benzo(b)fluoranthene
ο	Phenanthrene	ο	Benzo(k)fluoranthene
ο	Fluoranthene	ο	Benzo(a)pyrene
ο	Pyrene	0	Indeno(1,2,3-cd)pyrene
0	Benzo(a) anthracene	ο	Dibenz(a,h)anthracene
		0	Benzo(g,h,i)perylene

PAHs were prevalent only in sediments. Six were detected in background sediment samples at maximum concentrations 5- to 10-fold less than those detected in Study Area samples. As discussed in Section 4.0, PAHs may have entered the Sudbury River system from numerous contaminant sources. PAHs are produced by the combustion process and, consequently, are frequently found along roadways traversed by motor vehicles. Unlike phenol, nitrobenzene, and the chlorinated benzene compounds listed in the preceding table, most PAHs have not been identified by EPA as site-specific contaminants for the Nyanza Site. Naphthalene is the only PAH designated as a Nyanza site-specific contaminant. Toxicity criteria do not exist currently for 2-methyl naphthalene. This chemical will be discussed qualitatively in the risk assessment.

1,3-Dichlorobenzene is selected as a COC because it was designated a site-specific compound for Nyanza.

W92194F

6-22

Based on the selection criteria presented in the introduction to this section, the following semi-volatile organic compounds were not selected as COCs:

- o 2-Chlorophenol
- o Benzoic acid
- o Dibenzofuran
- o Diethylphthalate
- o Fluorene
- o Anthracene
- o Di-n-butyl phthalate
- o Butyl benzyl phthalate
- o Di-n-octyl phthalate
- o Bis(2-chloroethyl)ether
- o Benzyl alcohol
- o 2-Methylphenol
- o Diethyl phthalate
- o Acenaphthene
- o N-Nitrosodiphenylamine

Tables 6-2, 6-3, and 6-4 present the occurrence and distribution statistics and toxicity screen results that support the decision not to select these chemicals as COCs. None of these compounds are Nyanza site-specific contaminants.

6.2.1.3 <u>Pesticide/PCB Chemicals of Concern</u>

Several pesticides/polychlorinated biphenyl compounds (PCBs) were detected in sediments and fish tissue samples collected from the Study Area. The following pesticides/PCBs were selected as COCs because they were detected in at least five percent of environmental samples and demonstrated a relative risk factor greater than 0.0001:

Chemical	Surface Water Statistics ⁽¹⁾	ERi	Sediment Statistics ⁽¹⁾	ERi	Fish Statistics ⁽¹⁾	ERi
4,4-000	MD	•	112	4.4x10 ⁴ (C)	33x	3.3x104(C)
4,4-DDE	ND	•	5.41	5x10*(C)	71.31	1.6x10 ⁻³ (C)
4,4-DDT	MD	•	5.3%	2.98x10 ⁻³ (N)	15.4%	5x10*(C)
PCBs	ND	•	4.12	1.18x10 ² (C)	33.3X ^m	7.1x10 '(C)®
Dieldrin	ND	•	NED	-	6.6%	5.5x104(C)
Chiordane	NED	-	1.43	3.2x10 ⁻⁴	12.8%	7.2x10 ³ (N)

FINAL

- N = Noncarcinogenic-relative risk value
- C = Carcinogenic-relative risk value
- ERi = Relative Risk.
- ⁽¹⁾ Number of positive detections/number of samples collected, expressed as a percent.
 - See Table 6-5, relative risk factor presented for total PCBs.

Although heptaclor, aldrin, and endosulfan (and its metabolite, endosulfan sulfate) were detected in less than five percent of the fish tissue samples analyzed, they were included as COCs because the relative risk factors for these contaminants are greater than 0.001 (see Table 6-4).

The following pesticides did not meet the frequency of detection criteria and/or the toxicity screen criteria and were not selected as COCs:

- o Heptachlor epoxide
- o Endrin
- o Endrin ketone
- o Gamma BHC (Lindane)

Dieldrin, 4,4-DDE, endrin, 4,4-DDD, endosulfan sulfate, 4,4-DDT, and chlordane and Aroclor-1260 were all detected in background samples as well as Study Area samples. The maximum concentration detected in Study Area samples was no greater than 10-times the maximum background sample concentrations. As discussed in Section 4, pesticides/PCBs may have entered the Study Area waterways from various contaminant sources. These chemicals have not been designated by EPA as site-specific contaminants for the Nyanza Site.

6.2.1.4 Inorganic Chemicals of Concern

Mercury, chromium, lead, cadmium, antimony, and arsenic were identified in Section 4 as predominant Nyanza site contaminants. As shown on Tables 6-2, 6-3, and 6-4, these five metals meet the established COC selection criteria. Additionally, the maximum and/or average concentration of these metals in Study Area sediment and surface water samples are greater than two times the concentrations detected at background sample locations. All are selected as COCs for the Study Area risk assessment.

Barium, beryllium, manganese, nickel, selenium, silver, thallium, vanadium, and zinc meet one or more COC selection criteria for surface waters, sediments, or fish and are included as COCs. However, for several of these metals, particularly barium, copper, vanadium, and zinc, the concentrations detected in Study Area samples are similar to concentrations detected in background (less

W92194F

6-24

than two times background) sediment, surface water, and fish tissue samples. Others (e.g. thallium) were detected very infrequently. Toxicity criteria do not exist currently for copper. This metal will be discussed quantitatively in the risk assessment.

The following inorganic chemicals were not selected as COCs for the Study Area risk assessment:

- o Aluminum
- o Calcium
- o Cobalt
- o Iron
- o Magnesium
- o Potassium
- o Sodium

Several are macronutrients (potassium) which are necessary for health maintenance and all are known to be relatively nontoxic when compared to the heavy metals previously selected as COCs. Aluminum and cobalt do not fall into this category. Toxicity criteria are not currently available for these metals. Neither is a Nyanza site-specific chemical. Because of the large number of COCs already selected for the Study Area risk assessment, it is unlikely that the exclusion of these metals as COCs will significantly affect the results of the risk assessment. However, both metals will be discussed in the uncertainty analyses presented in Section 6.6.

The COCs selected for the Sudbury River Study Area are summarized in Table 6-5.

6.2.2 Toxicity Profiles

The purpose of this section is to identify the potential health hazards associated with exposure to the chemicals of concern identified in Section 6.2.1. A toxicological evaluation of each indicator chemical was conducted to characterize its inherent toxicity. The evaluation consisted of the review of scientific data to determine the nature and extent of the human health hazards associated with exposure to the various chemicals. Based on the scientific data review, a toxicity profile for each indicator chemical was developed.

Toxic effects considered in these profiles include noncarcinogenic (toxic) and carcinogenic health effects. Noncarcinogenic health effects are generally assumed to occur only at doses exceeding a certain "threshold dose." Toxicological endpoints, routes of

W92194F

FINAL

TABLE 6-5 CHEMICALS OF CONCERN NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

Volatile Organia Ch	
Vinyl chloride	Chlorobenzene
1,2-Dichloroethene	Methylene chloride
Trichloroethene	Acetone
Benzene	1,1-Dichloroethene
Semi-Volatile Organic	Chemicals
1,3-Dichlorobenzene	Acenapthylene
1,4-Dichlorobenzene	Phenanthrene
1,2-Dichlorobenzene	Fluoranthene
Nitrobenzene	Pyrene
1,2,4-Trichlorobenzene	Benzo(a)anthracene
Bis(2-ethylhexyl)phthalate	Chrysene
Phenol	Benzo(b)fluoranthene
3-Methylphenol	Benzo(k)fluoranthene
4-Methylphenol	Benzo (a) pyrene
Napthalene	Dibens (2, 2, 3-ca) pyrene
2-metnyl naptnalene	Dibenz (a, n) anthracene
	Benzo(g, n, 1) perylene
Pesticides/PC	B s
4,4-DDD, DDE, and DDT	Dieldrin
Polychlorinated biphenyl compounds	Aldrin
Chlordane	Heptachlor
Endosulfan I, II	
(and endosulfan sulfate)	
Metals/Alkylated	Metals
Monomethyl mercury	Copper
Dimethyl mercury	Lead
Mercury	Manganese
Arsenic	Nickel
Antimony	Silver
Barium	Selenium
Beryllium	Thallium
Cadmium	Vanadium
Chromium	Zinc

exposure, and doses in human and/or animal studies are discussed in the profiles.

Carcinogenic health effects are associated with exposure to a chemical capable of promoting, initiating, or causing a malignant neoplasm. Routes of exposure and doses in human and/or animal studies are discussed. Also presented is the EPA's weight-of-evidence for a compound's carcinogenicity (Group A, known human carcinogens; Group B, probable carcinogens, Group C, possible carcinogens, Group D, not classifiable as to its carcinogenicity).

The available toxicological information indicates that many of the indicator chemicals have both noncarcinogenic and carcinogenic health effects in humans and/or in experimental animals. Although the indicator chemicals may cause adverse health and environmental impacts, dose-response relationships and the potential for exposure must be evaluated before the risks to receptors can be determined. Dose-response relationships correlate the magnitude of the dose with the probability of toxic effects, as discussed in the following section.

All toxicity profiles are presented in Appendix K. A brief summary of the toxicological properties of each chemical of concern is also provided in the following paragraphs. The toxicity profiles provide the qualitative weight-of-evidence that Study Area contaminants pose actual or potential hazards to human health and the environment.

6.2.2.1 <u>Acetone</u>

Adverse noncarcinogenic health effects (human receptors) reported for acetone exposure through inhalation include: nausea, vomiting, muscle weakness, and difficulty breathing. At high concentrations of exposure through inhalation, the compound can have a narcotic effect. Prolonged repeated exposure by inhalation will produce headaches (human receptors). Acetone is a kidney toxin (test animals (rats) and humans (oral route of exposure)). Dermal exposure to high concentrations of acetone or pure product may cause excessive drying of the skin (human receptors).

6.2.2.2 <u>Aldrin</u>

Aldrin, one of the cyclodiene insecticides, is a neurotoxicant and has been classified as a B-2 carcinogen (test animal=mice; oral route of exposure). Studies have demonstrated that it produces liver tumors in test animals (mice). The effect of concern listed in the HEAST FY 1991 tables for the RfD is liver lesions (oral route of exposure test animal=rat).

W92194F

6.2.2.3 <u>Antimony</u>

Long-term repeated exposure to antimony, by the ingestion route of exposure, can reduce life span, alter blood makeup, and cause heart damage (test animals (rat)). Exposure to antimony by inhalation may cause difficulty in breathing (human receptors). Skin rashes are often produced as a result of direct dermal contact with the chemical (human receptor).

6.2.2.4 <u>Arsenic</u>

Arsenic has been implicated in the production of skin cancer in humans. There is also extensive evidence that inhalation of arsenic compounds causes lung cancer in workers; elevated incidence of chromosome aberrations have been reported in humans exposed to arsenic compounds. Arsenic compounds have been reported to be teratogenic, fetotoxic, and embryotoxic in several animal species, and an increased incidence of multiple malformations among children born to women occupationally exposed to arsenic has been reported. Arsenic compounds also cause noncancerous, possibly precancerous, skin changes in exposed individuals.

The current chronic Reference Dose (RfD) for Arsenic is 3×10^4 mg/kg/day, based on the results of an epidemiological study conducted in Taiwan (Tseng, 1977). The noncarcinogenic end point of concern was keratosis (formation of horny growths on the skin) and hyperpigmentation (human receptor; oral route of exposure). The RfD is currently under review by the EPA RfD group.

Inorganic arsenic is classified as a Class A human carcinogen (a known human carcinogen). The cancer slope factor (CSF) for the inhalation route of exposure is 50 $(mg/kg/day)^{-1}$ and is based on epidemiological studies of smelter workers. A unit risk of 5 x 10⁻⁵ $(ug/L)^{-1}$ (oral route of exposure) proposed by the Risk Assessment Forum, is currently under review.

6.2.2.5 <u>Barium</u>

Increased blood pressure has been observed in experimental animals (rats) routinely exposed to barium in drinking water. Barium is also toxic to the nervous system, the muscular system, and gastrointestinal system when ingested at high concentrations. The soluble barium salts are more toxic than the insoluble barium salts (Clements, 1985). This is probably due to the fact the soluble barium salts are more likely to be absorbed than insoluble barium salts.

W92194F

6-28

FINAL

6.2.2.6 <u>Benzene</u>

Benzene is classified as a known human carcinogen (Class A), based on the incidence of leukemia in individuals exposed in the work place. Numerous epidemiologic and case studies have reported an increased incidence or casual relationship between leukemia and benzene exposure. An increased incidence of neoplasia has also been reported in rats and mice exposed by inhalation and gavage. The current Health Advisory available for benzene is based on animal studies which demonstrated noncarcinogenic liver effects in animals exposed through inhalation. Both oral and inhalation Reference Doses are under review by the U.S. EPA. The carcinogenic slope factor (oral) is 2.9×10^{-2} (U.S. EPA, IRIS, December 1991).

6.2.2.7 <u>Beryllium</u>

Short-term exposure to beryllium by inhalation can cause formation of scar tissue in the lungs, breathing difficulty, and weight loss (human receptors). Skin exposure to beryllium can cause a skin rash at the point of contact. Beryllium is currently classified as a probable human carcinogen based on the results of occupational studies and animal studies (rats) demonstrating a possible relationship between beryllium exposure and lung cancer. Other cancers were also noted.

6.2.2.8 <u>Bis(2-ethylhexyl)phthalate</u>

The RfD and CSF available for bis(2-ethylhexyl)phthalate are based on animal studies detecting adverse noncarcinogenic liver effects and liver tumors in test animals (guinea pigs) exposed through the oral route of exposure.

6.2.2.9 <u>Cadmium</u>

There is suggestive evidence linking cadmium with cancer of the prostate in humans (U.S. EPA, 1980). An increased incidence of tumors has not been seen in animals exposed to cadmium orally, but four of the five available studies were inadequate by current standards (Clement, 1983).

The evidence from a large number of studies on the mutagenicity of cadmium is equivocal; it has been hypothesized that cadmium is not directly mutagenic but impedes repair (Clement, 1983). Cadmium is a known animal teratogen and reproductive toxin. It has been shown to cause renal dysfunction in both humans and animals. Other toxic effects attributed to cadmium include immunosuppression (in animals), anemia (in humans), pulmonary disease (in humans), a

possible effect on the endocrine system, defects in sensory function, and bone damage.

The current oral RfD (chronic) for cadmium is 5×10^4 . A toxicokinetic model was used to establish this value. The noncarcinogenic end point of concern is renal damage (proteinuria). An uncertainty factor of ten is associated with this RfD. Cadmium is classified as a probable human carcinogen (B-1) for the inhalation route of exposure.

6.2.2.10 <u>Chlordane</u>

Chlordane, a cyclodiene insecticide, is a pesticide which degrades very slowly in the environment. Clinical effects observed as the result of human exposure include muscle tremors, anemia, and leukemia. Reproductive effects have been reported in animal studies (Lappenbusch, 1988). The RfD and CSF available for chlordane are based on animal studies demonstrating liver necrosis and liver tumors in test animals (rats, oral route of exposure). Chlordane has been designated a Class B-2 carcinogen by the EPA. As noted for many other pesticides/PCBs, chlordane bioconcentrates significantly in fish tissue.

6.2.2.11 <u>Chlorobenzene</u>

A carcinogenicity study showed that chlorobenzene caused neoplastic nodules in the liver of male rats but was not carcinogenic in female rats or in mice. Occupational studies suggest that chronic exposure to monochlorobenzene vapor may cause blood dyscrasia, hyperlipidemia, and cardiac dysfunction in humans. Like many organic solvents, monochlorobenzene is a central nervous system depressant in overexposed humans, but no chronic neurotoxic effects have been reported.

The oral Reference Dose for chlorobenzene is $2 \times 10^{-2} \text{ mg/kg/day}$. The uncertainty assigned this Reference Dose is 1,000. There is a medium level of confidence for this Reference Dose because of a low assessment of toxicity possible in the study. Carcinogenicity of chlorobenzene is under review by a U.S. EPA work group (U.S. EPA, IRIS, September 1990).

6.2.2.12 <u>Chromium</u>

Skin rashes (human receptor) may occur as a result of dermal contact with certain chromium compounds (chromium III). Exposure by inhalation of chromium (VI) can cause irritation, inflammation, ulceration, and perforations of the nasal cavity. Exposure to

W92194F

chromium (VI) compounds can cause kidney damage. Occupational health studies demonstrate a cause-effect relationship between chromium exposure in workers and lung cancer.

6.2.2.13 <u>Copper</u>

A deficiency of copper, an essential element, may result in anemia, loss of pigment, reduced growth, and loss of arterial elasticity. However, persons who are overexposed may exhibit Wilson's Disease (disorder of Cu metabolism) or liver cirrhosis (Lappenbusch, 1988).

6.2.2.14 <u>4,4-DDT; 4,4-DDE; and 4,4-DDD</u>

Dichlorodiphenyl trichloroethane was a widely used insecticide. The central nervous system and liver are primary target organs for this compound and its degradation products. Tremors have been observed in the chronically exposed (Lappenbusch, 1988). The RfD and CSF available for DDT and its degradation byproducts are based on liver effects and liver tumors observed in test animals (oral route of exposure rats, mice, hamsters). 4,4-DDT; 4,4-DDE; and 4,4-DDD have been designated as Class B-2 carcinogens by the EPA.

6.2.2.15 <u>1,2-Dichlorobenzene; 1,3-Dichlorobenzene; and</u> <u>1,4-Dichlorobenzene</u>

The liver is a target organ for all three dichlorobenzenes. "Liver effects" and/or "liver and kidney effects" are listed as the "effect of concern" for the available RfDs (inhalation and oral routes of exposure; test animal=rats). Additional bioeffects may include anemia and skin lesions (Lappenbusch, 1988). The CSF available for 1,4-dichlorobenzene is based on animal studies that detected liver tumors in test animals (mice) exposed through the oral route of exposure.

6.2.2.16 <u>1,1-Dichloroethene (1,1-DCE)</u>

The liver and kidney are primary target organs for 1,1-DCE. The compound is currently classified as a possible human carcinogen (Class C). Inhalation exposure has an anesthetic effect.

6.2.2.17 <u>1.2-Dichloroethene (1.2-DCE)</u>

1,2-DCE is a central nervous system and liver toxin. The current Reference Dose for the oral route of exposure is based on animal studies (test animal=rats) reporting an adverse effect on the blood system. Inhalation exposure to elevated concentrations of 1,2-DCE can cause nausea, vomiting, muscle weakness, tremors, and cramps. Repeated exposure by this route can effect proper functioning of

W92194F

FINAL

the liver. Short-term, high-concentration exposures can effect the central nervous system.

6.2.2.18 <u>Dieldrin</u>

Dieldrin, one of the cyclodiene insecticides, is a neurotoxicant and has been classified a B-2 carcinogen. CSFs are available for the inhalation and oral routes of exposure. Studies have demonstrated that it produces liver tumors in test animals (mice). The effect of concern listed in the HEAST FY 1991 tables for the RfD, is liver lesions (oral route of exposure, test animal=rats). The BCF for dieldrin exceeds 4,500; the chemical bioconcentrates significantly in fish tissues (Lappenbusch, 1988). The use of this chemical has been discontinued in the U.S. because of its carcinogenicity (Hodgson et al., 1988).

6.2.2.19 <u>Endosulfan</u>

Mild kidney lesions have been observed in rats exposed to endosulfan in their diet. The uncertainty factor assigned to the RfD currently available for this compound is 3000. Endosulfan is described in Sax as a highly toxic organochlorine pesticide which does not accumulate significantly in human tissue. It is a central nervous stimulant producing convulsions (receptor not specified -Sax, 1991).

6.2.2.20 <u>Heptachlor/Heptachlor Epoxide</u>

Heptachlor, a cyclodiene insecticide, is a moderately persistent insecticide. In the body, heptachlor is metabolized to heptachlor epoxide. Adverse bioeffects resulting from exposure to heptachlor/heptachlor epoxide include neurological effects like irritability as well as blood disorders (anemia, leukemia) (Lappenbusch, 1988). The RfD for heptachlor was derived based on studies demonstrating increased liver weight in test animals exposed through the oral route of exposure. Heptachlor has also produced liver tumors in mice.

6.2.2.21 <u>Lead</u>

There is evidence that several lead salts are carcinogenic in mice or rats, causing tumors of the kidneys after either oral or dermal administration. Data concerning the carcinogenicity of lead in humans are inconclusive. There is equivocal evidence that exposure to lead causes genotoxicity in humans and animals. The available evidence indicates that lead presents a hazard to reproduction and exerts a toxic effect on conception, pregnancy, and the fetus in humans and experimental animals.

W92194F

Many lead compounds are sufficiently soluble in body fluids to be toxic. Lead poisoning may cause peripheral neuropathy in adults and children; permanent learning disabilities that are clinically undetectable in children may be caused by exposure to relatively low levels. Short-term exposure to lead can cause reversible kidney damage, but prolonged exposure at high concentrations may result in progressive kidney damage and possibly kidney failure. Anemia is an early manifestation of lead poisoning. Lead is classified as a probable human carcinogen. A CSF has not been published to date by the EPA. (U.S. EPA, IRIS, September 1990).

A RfD for lead is not currently available for either the oral or inhalation route of exposure. However, lead is considered a highly toxic metal. Recently, the EPA presented a strategy to address health and environmental problems caused/suspected to occur as a result of public exposure to this multi-media contaminant (EPA, February 1991). An Action Level of 15 ug/L for lead in drinking water supplies was established in June 1991.

6.2.2.22 <u>Manganese</u>

ł

The central nervous system is a primary target organ for manganese. Inhalation of manganese dusts may result in lung disease. Long-term manganese inhalation exposure can affect the central nervous system and the liver. An iron-deficient diet may magnify these effects.

6.2.2.23 Mercury and Mercury Compounds

Elemental mercury is not highly toxic as an acute poison. Soluble mercuric salts are highly poisonous on ingestion, with oral LD_{50} values of 20 to 60 mg/kg reported. Mercurous compounds are less toxic when administered orally. Acute exposure to mercury compounds at high concentrations causes a variety of gastrointestinal symptoms and severe anuria with uremia. Signs and symptoms associated with chronic exposure involve the central nervous system and include behavioral and neurological disturbances.

In general, organic mercury compounds (methyl mercury) are more toxic than inorganic compounds. Neurotoxicity is a toxic endpoint for methylmercury compounds. Although brain damage due to prenatal exposure to methyl mercury has occurred in human populations, no conclusive evidence is available to suggest that mercury causes anatomical defects in humans.

1

The current oral RfD (chronic) for inorganic mercury and methyl mercury is 3×10^4 mg/kg/day. The end points of concern are neurotoxicity and kidney effects. The RfD is based on epidemiological studies and rat studies which evaluated the ingestion and parenteral routes of exposure. An uncertainty factor of 1,000 and 10 have been assigned to the RfDs for inorganic mercury and methyl mercury, respectively.

6.2.2.24 Methyl Napthalene

2-Methyl Napthalene is described as moderately toxic via the ingestion and intraperitoneal routes of exposure (receptor not specified). If the compound is heated, acrid smoke and irritating fumes are emitted. The LD-50 for rats is 1630 mg/kg (oral route of exposure) (Sax, 1991).

6.2.2.25 Methylene Chloride

Methylene chloride is currently categorized as a probable human carcinogen through the ingestion and inhalation routes of exposure. The results of animal studies (rats-oral route of exposure) indicate that it is a liver toxin. At high concentrations, the compound is a severe skin and eye irritant (human receptor).

6.2.2.26 <u>3-Methylphenol/4-Methylphenol</u>

The current RfD for 3-/4-methylphenol was derived based on the results of animal studies demonstrating reduced body weight gain and neurotoxicity in chronically exposed (oral route of exposure) test animals (rats).

6.2.2.27 <u>Nickel</u>

Short-term inhalation exposure to nickel can cause pneumonia-like symptoms. Long-term inhalation exposures can produce nasal injury, cancer of the nasal cavity, and lung cancer (human receptor). Skin contact may produce a skin rash at the point of contact (human receptor). The current Reference Dose for nickel is based on animal studies (rat) demonstrating reduced body weight as a result of nickel exposures (oral route of exposure).

6.2.2.28 <u>Nitrobenzene</u>

Headaches, vertigo, and methemoglobinemia have been observed in workers exposed to nitrobenzene. The current oral RfD for nitrobenzene is 5×10^4 mg/kg/day. The end points of concern are hematological, adrenal, renal, and hepatic lesions observed in test

W92194F

6-34

animals (mice) exposed through the oral route of exposure. The oral RfD for nitrobenzene may change in the near future pending the outcome of a review currently underway by the Oral RfD Work Group.

The current Ambient Water Quality Criteria (AWQC) for the protection of human health is 30 ug/L (assuming water and fish consumption) and is based on organoleptic end points. The AWQC lowest effects concentration (LEC) for the protection of freshwater aquatic life is 27,000 ug/L (acute limit).

6.2.2.29 Phenol

The current RfD for phenol is based on the chemical's adverse health effects on the fetus. Other target organs for this compound are the liver, kidney, lung, and heart (Lappenbusch, 1988). Phenol has been shown to be a promoter of carcinogenesis in experimental animals (Hodgson et al., 1988).

6.2.2.30 Polychlorinated Biphenyl Compounds (PCBs)

Acneform eruptions, abnormal liver function, increased abortions and premature deliveries, and immuno-suppression have been observed in humans exposed to PCBs through the oral route of exposure. PCBs have been designated Class B-2 carcinogens by the EPA. Liver tumors have been detected in mice exposed through the oral route of exposure (Lappenbusch, 1988). PCBs are known to bioconcentrate significantly in fish tissue.

6.2.2.31 Polynuclear Aromatic Hydrocarbons (PAHs)

Polynuclear aromatic hydrocarbons are a large, diverse class of chemicals found in the environment as complex mixtures. Proliferating tissues (the intestinal epithelium) are particularly susceptible target organs for the PAHs (Lappenbusch, 1988). The RfDs which are available for naphthalene, acenaphthylene, phenanthrene, fluoranthene, and pyrene specify the liver, kidney, skin, and/or decreased body weight gain as target organs or effects of concern (test animal=rats). Several PAHs have been designated as Class B-2 carcinogens and it is the carcinogenic potential of these chemicals that is usually of prime concern in public health risk assessment (several test species-oral and inhalation route of exposure). The CSFs available for benzo(a)pyrene are derived based on animal studies demonstrating tumors of the respiratory tract and stomach in test animals exposed through the inhalation and oral routes of exposure, respectively. Many PAHs cause tumors in the skin and epithelial tissues of test animals. An excess mortality from lung cancer has been observed among workers exposed to large

amounts of PAH-containing materials such as coal gas, tars, and coke oven emissions (Lappenbusch, 1988).

6.2.2.32 <u>Selenium</u>

Acute (high-dose) exposure to selenium can adversely affect the liver, kidney, myocardia, and nervous system. Chronic selenium toxicity is manifest by depression, nervousness, dermatitis, and gastrointestinal disturbances. Selenium is an essential element in animals and probably in humans (Sax, 1991).

6.2.2.33 <u>Silver</u>

Argyria (a bluish discoloration) has been observed in patients exposed therapeutically to silver. Acute (high-dose) exposures (oral route) can result in lesions of the liver, kidney, bone marrow, and lung as well as severe gastrointestinal distress (convulsions and vomiting). Chronic exposure may adversely affect numerous organ systems and impair immunological resistance (Sax, 1991).

6.2.2.34 <u>Thallium</u>

As noted in Table 6-1, thallium (RfD = 7×10^{-5} ; oral route of exposure, chronic) has one of the lowest RfDs available for metals currently listed in IRIS or in the HEAST tables. The effect of concern specified is increased SGOT (serum glutamic oxalacetic transaminase) and serum LDH (lactic dehydrogenase) levels and alopecia (absence or loss of hair) in test animals (rats) exposed through the oral route of exposure. Chronic toxicity includes reddening of the skin, polyneuritis, alopecia, and cataracts. Hepatic and renal damage has been observed after chronic exposure (Hodgson et al., 1988).

6.2.2.35 <u>Zinc</u>

Anemia has been observed in patients exposed therapeutically to zinc. However, zinc is an essential trace element. Excess amounts may adversely affect the gastrointestinal system (i.e. cause vomiting) and alter the distribution of other trace elements in the human body (Sax, 1991).

6.2.2.36 <u>1.2.4-Trichlorobenzene</u>

The current RfD for trichlorobenzene is derived from the results of animal studies demonstrating porphyria in test animals (rats, oral route of exposure). Porphyria is a disorder in which there is increased formation and excretion of porphyrins or their precursors. Porphyrins are a group of pigments found in living cells that act as the prosthetic group for respiratory pigments.

6.2.2.37 <u>Trichloroethene (TCE)</u>

TCE is a skin and eye irritant. Inhalation of high concentrations can produce a numbing effect and heart failure. Repeated exposure can cause headaches, drowsiness, and damage internal organs. TCE is a liver toxin and is currently classified as a probable human carcinogen.

6.2.2.38 <u>Vanadium</u>

The primary health effect from airborne exposure to vanadium is irritation of the skin, eyes, and respiratory tract. Prolonged inhalation exposure can cause coughing, bronchitis, chest spasms, and chest pain.

6.2.2.39 Vinyl Chloride

Vinyl chloride has been designated a Class A (known human carcinogen) by the EPA. Studies have demonstrated that it produces liver tumors in animals exposed through various routes of exposure. The principal toxicological concern is that chronic poisoning with vinyl chloride leads to angiosarcoma of the liver in humans. Vinyl chloride is a mutagen and it adversely affects reproduction in male and female test animals (Hodgson et al., 1988).

6.3 <u>Dose-Response Assessment</u>

An important component of the risk assessment process is the relationship between the dose of a compound (amount to which an individual or population is exposed) and the potential for adverse health effects resulting from exposure to that dose. Dose-response relationships provide a means by which potential public health impacts may be evaluated. This section discusses the toxicity criteria that will be used to characterize the public health risk associated with exposure to Study Area COCs and identifies standards/criteria available for Study Area COCs.

Tables 6-6 presents regulatory standards or guidelines available for the chemicals of concern selected in Section 6.2. Target organs are listed in Table 6-7. These values and the toxicity criteria (Reference Doses and Carcinogenic Slope Factors) presented in Table 6-1 are used in the risk characterization process which will be described in Section 6.5.
As discussed in succeeding sections, the risk assessment for the Study Area will be conducted inpart, by a comparison of contaminant levels to existing standards and criteria. Additionally, anticipated human exposure dosages resulting from potential human contact with contaminated environmental media will be evaluated using the Reference Doses and Carcinogenic Slope Factors in Table 6-1. The methodology for estimating exposure dosages is presented in Section 6.4.

6.3.1 Toxicity Criteria (Dose-Response Parameters)

6.3.1.1 <u>Reference Dose (RfD)</u>

As defined in IRIS (the EPA Integrated Risk Information System), an RfD is an estimate (with uncertainty spanning perhaps an order of magnitude) of a daily exposure to the human population (including sensitive subgroups) that is likely to be without an appreciable risk of deleterious effects during a lifetime. RfDs are developed for chronic and/or subchronic human exposure to hazardous chemicals and are based on the assumption that thresholds exist for certain toxic effects. The RfD is usually expressed as an acceptable dose (mg) per unit body weight (kg) per unit time (day). The RfD is derived by dividing the no-observed-adverse-effect-level (NOAEL) or the lowest-observed-adverse-effect-level (LOAEL) by an uncertainty factor (UF) times a modifying factor. The use of uncertainty factors and modifying factors is discussed in the EPA, Office of Research and Development (ORD) Health Effects Assessment Summary Tables, FY 1991.

The uncertainty factor used in calculating the RfD reflects scientific judgement regarding the various types of data used to estimated RfD values. An uncertainty factor of 10 is usually used to account for variation in human sensitivity when extrapolating from valid human studies involving subchronic (for subchronic RfDs) or long-term (for chronic RfD) exposure of average, healthy subjects. An additional 10-fold factor is usually used for each of the following extrapolations: from long-term animal studies to the case of humans, from a LOAEL to a NOAEL, and from subchronic studies to a chronic RfD. An additional uncertainty or modifying factor, ranging from >0 to 10, is applied to reflect professional assessment of the uncertainties of the study and database not explicitly addressed by the above uncertainty factors (such as completeness of the overall data base). The default value for this modifying factor is 1.

W92194F

TABLE 6-6 REGULATORY REQUIREMENTS FOR CHEMICALS OF CONCERN NYANZA OPERABLE UNIT 3 MIDDLEBEX COUNTY, MASSACHUSETTS

				Federal Ambient W	ater Quality	Commonwealth of	FDA
	EPA SOWA Ste	ndards (ug/L)	EPA Drinking Water	Criteria for Huma	n Health	Massachusetts	Action
Chemical Of Concern				Ingestion of Water	Ingestion of		Leveis
	MCL (Statue)	MCLG (Status)	(ug/L)	Plus Consumption	Aquatic Life	Drinking Water	Consumption
	(ug/L)	(ug/L)	(1)	of Aquatic Life	Only	Standards/Guidelines	of Fish
	(1)	(1)		(ug/L)	(ug/L)	(4)	(ppm)
				(2)	(2)		(5)
Volatile Organic		<u></u>					
Vinyl Chloride	2 (F)	0 (F)	10 (L.T child)	2 (3)	525 (3)	2	
Methylene Chloride	5 (P)	0 (P)	500 (L.T child)	0.19	15.7	5	
Acetone						700	
1,1-Dichloroethene	7 (F)	7 (F)	7 (Lifetime)	0.033	1.85	7	1
1,2-Dichloroethene(cis)	70 (F)	70 (F)	70 (Lifetime)	700(3)	140,000(3)	70	1
Trichloroethene	5 (F)	0 (F)		2.7	80.7	5	
Benzeñe	5 (F)	0 (F)	235 (10 day-child)	0.66	40	5	1 1
Chlorobenzene	100 (F)	100 (F)	100 (Lifetime)	660(3)	21000(3)	100	
Semivolatile Organic							
1,4-Dichlorobenzene	75 (F)	75 (F)	75 (Lifetime)	400(3)	2600(3)	5	{
1,2-Dichlorobenzene	600 (F)	600 (F)	600 (Lifetime)	2700(3)	17000(3)	600	{ }
Nitrobenzene				30			
1,2,4-Trichlorobenzene	9(P)	9 (P)	9(Lifetime)	1		1	
Bis(2-ethylhexyl)	4 (P)		9(Lifetime)	15000 (1.8(3))	50000 (5.9 (3))	10	
phthalate							
Phenol			4000 (Lifetime)	21000(3)	4600000(3)	ĺ	{ }
4-Methylphenol							} }
		Í					

HOLLOR

TABLE 6-6

REGULATORY REQUIREMENTS FOR CHEMICALS OF CONCERN

NYANZA OPERABLE UNIT 3

1

MIDDLESEX COUNTY, MASSACHUSETTS

PAGE 2

W92194F

		- 4- 4 6 - 9		Federal Ambient W	ater Quality	Commonwealth of	FDA
	EPA SUWA Sta	ndards (ug/L)	EPA Drinking water	United a Water	Ingestion of	Massachusens	Action
Chemical Of Concern	MCL (Bratus)	MCL G. (Status)	(upt)	Blue Consumption	Aquetic Life	Drinking Weter	Consumptio
	(und)	(uo/l)	(0g) L) (1)	of Aquatic Life	Aqualic Lib Only	Standarde/Guidelines	of Fieb
		(ug/L)	(')	(ug/l)	(ug/l)		
		(')		(3972)	(39/12/		(5)
Polyaromatic Hydrocarbons						· · · · · · · · · · · · · · · · · · ·	
Naphthalene			20 (Lifetime)				1
2-Methylnaphthalene							
Acenaphthylene				0.0028(3)	0.031 (3)		ł
Phenanthrene				0.0028(3)	0.031 (3)		
Fluoranthene				3000(3)	370(3)		1
Pyrene				960(3)	11000(3)		
Benzo(a)anthracene	0.1 (P)	0 (P)		0.0028(3)	0.031 (3)		
Chrysene	02 (P)	0 (P)		0.0028(3)	0.031 (3)		1
Benzo(b)fluoranthene	02(P)	0 (P)		0.0028(3)	0.031 (3)		
Benzo(k)fluoranthene	02(P)	0 (P)		0.0028(3)	0.031 (3)		
Benzo(a)pyrene	02(P)	0 (P)		0.0028(3)	0.031 (3)		
Indeno(1,2,3-cd)pyrene	02(P)	0 (P)		0.0028(3)	0.031 (3)		
Dibenz(a,h)anthracene	0.3 (P)	0 (P)		0.0028(3)	0.031 (3)		
Benzo(g,h,i) perylene	02 (P)	0 (P)					
Pesticidea							
Aldrin			0.3 (L.T. – child)	0.00013(3)	0.00014(3)		0.3
4,4'-DDD				0.00083(3)	0.00084(3)		ł
4,4'-DDE				0.00059(3)	0.00059(3)		5
4,4,'-DDT				0.000024(3)	0.000024		5
Dieldrin			0.5 (L.T child)	0.033			
Heptachlor	04(F)	0 (F)	5 (L.T. – child)	0.00021 (3)	0.00021 (3)	0.2	0.3
Chlordane (Total)	2 (F)	0 (F)	0.5 ug/L (L.T child)	0.00057(3)	0.00059(3)	0.5	0.3
Endosulfan I							
Endosulfan II							
Endosulfan Sulfate							

TABLE 8-8

REGULATORY REQUIREMENTS FOR CHEMICALS OF CONCERN

NYANZA OPERABLE UNIT 3

MIDDLESEX COUNTY, MASSACHUSETTS

V REGULA V NYANZA V MIDDLES V PAGE 3 4

	EPA BOWA Sta	nderde (ugA.)	EPA Drinking Water	Federal Ambient W Criteria for Humar	ater Quality 1 Health	Commonwealth of Massachusetts	FDA Action
Chemical Of Concern	MCL (Bistus)	MCLQ (Status)	(vg/L)	Ingestion of Water Plus Consumption	In gestion of Aquatic Life	Drinking Water	Levels Consumption
			(17		() my		(nem)
		(1)		(091.)	(09)()	(*)	(ppm) (5)
		L		L			······································
PCB							
Aroclor 1248	05(F)	0 (6)		0 000044(3)	0.000045(3)	05	
Aroclor 1254	05(F)	0(#)		0 000044(3)	0 000045(3)	0.5	
Aroclor 1260	05(F)	0 (f)		0 000044(3)	0 000045(3)	0.5	1
Metals/Alkylated Metals							
Monomethyl Mercury				{			
Dimethyl Mercury		1					
Mercury	267	2(7)	2 (Lilatima)	0 14(3)	0.15(3)	2	1
Antimony	319	3 🔊	3 (Lifetime)	14(3)	4300(3)		1
Arsenic	0.05 mgl. (1)			0 0022	0.0175	50	
Berlum	5000 (P)	2000 (P)	5000 (Lifetime)			1000	
Beryllium	10	0 (*)	4000 (L.T ch#d)	0 0077 (3)	0.13(3)		
Cadmium	5 (#)	\$ (F)	5 (Lifetime)	10	170(3)	10	
Chromium (Total)	100 (F)	100 (F)	100 (Lifetime)	50	3400(3)	50	
Copper	1300 (**)	1300 (**)		1300 (3)		1300	{
Lead	15 (Action Lavel	•		50 (3)		50	
Manganese	50 (F) Seconder	v]					
Nickel	100 (P)	100 🝙	100 (Liletime)	610(3)	4600(3)		
Selenium	50 (F)	50 (F)		100(3)	6800(3)	10	1
Silver	100 (F)		100 (Lifetime)	105(3)	65000(3)	50	
Theilum	119	05(7)	0 4 (Lifetime)	1 7 (3)	6 3(3)		ł
Vanadium			20 (Liletime)				
Zinc	5000 (F) Second	1 vrat				1	

* - Unless otherwise noted

KEY: I - Interim - Undereview

- P Proposed
- F Final
- LT Long Term

FOOTNOTES:

(1) EPA Drinking Water Regulations and Health Advisory Memorandum ~ Office of Drinking Water - April 1991.

(2) AWQC values presented in IRIS (12-3-81)

(3) AWQC values presented in EPA Headquaters Office of Water Bummary Tables

(see Appendix K) are presented in those cases when values were not available on IRIS (The 10~6 risk value is presented for carchogens)

(4) Commonwealth of Massachusetta Drinking Water and Standards and Ouldelines Memorandum (see Appendit K)

(5) ASTDR - Health Assessment Guidence Manual (ASTDR - July, 1990) (see Appendix IQ)

TABLE 6-7 CHEMICALS OF CONCERN AND TARGET ORGANS NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

Chemical	Target Organ:		Chemical	Target Organ:
	Oral / Inhalation			Oral / Inhalation
		-		
Volatile Organic			Pesticides	
Vinyl Chloride			4,4,'-DDD	
Methylene Chloride	Liver/Liver		4,4,'-DDE	
Acetone	Liver, Kidney/		4,4,'DDT	Liver/
1,1-Dichloroethene	Liv,Kid/Liv,Kid		Dieldrin	Liver/
1,2-Dichloroethene			Chlordane (Total)	Liver (Females)/
Trichloroethene			Endosulfan I	Kidney/
Benzene			Endosulfan II	Kidney/
Chlorobenzene	Liver, Kidney/		Endosulfan Sulfate	
			Aldrin	Liver/
Semivolatile Organic			Heptachlor	Liver/
1,4-Dichlorobenzene				
1,2-Dichlorobenzene	Liver/		PCBs	
Nitrobenzene	Liver, Kidney/		Arocior 1248	
1,2,4-Trichlorobenzene	Epidemis/		Aroclor 1254	
Bis(2-ethylhexyl)	Liver/	i.	Arocior 1260	
phthalate		ł		
Phenol	Fetus/	i	Metals/Alkylated Metals	:
3/4-Methylphenol	None/		Monomethyl Mercury	CNS/
			Dimethyl Mercury	CNS/
(Polvaromatic Hydrocarbons)		;	Mercury	Brain, Kidney/
Nachthaiene	Body Weight/		Antimony	Longevity, Blood/
2-Methvinaphthaiene	Body Weight/		Arsenic	Skin/
Acenaphthylene	Epidemis/		Banum	Blood/
Phenanthrene	Body Weight/		Bervilium	
Fluoranthene	Lver Kidnev/		Cadmium	Kidney/
Pyrana	Kidnev/		Chromum (VI)/III	Liver/
Benzo(a)anthracene			Copper	
Chrysene			Lead	
Benzo (b) fil ioranthene			Manganese	CNS/CNS
Benzo (k) fluoranthene			Nickel	Body Weight/Lung
Benzo(a)nurene	-		Selenium	Selenosis/
Indena (1 2 3-cd) ovrepe	· · · · · ·		Silver	Arovria/
	·		Thallium	
			Vanadium	
Benzol(g.n.)perviene			Zioo	Anomial
				Anemia

FINAL

_ __

6.3.1.2 <u>Carcinogenic Slope Factor (CSF)</u>

CSFs are applicable for estimating the lifetime probability (assumed 70-year lifespan) of human receptors contracting cancer as a result of exposure to known or suspected carcinogens. This factor is generally reported in units of (mg/kg/day)⁻¹ and is derived through assumed low-dosage responses determined from human or animal studies. Cancer risk and CSFs are most commonly estimated through the use of a linearized, multistage, mathematical extrapolation model applied to animal bioassay results. The value used in reporting the slope factor is the upper 95 percent confidence limit.

The available toxicity criteria presented in Table 6-1 are summarized from IRIS (EPA, December 1991). Toxicity criteria available from EPA Region I, the EPA Health Effects Assessment Summary Tables (HEAST) - FY 1991, and the EPA Drinking Water Health Advisory Memorandum (EPA, April 1991), are presented when toxicity criteria are not available from IRIS.

6.3.2 Criteria/Standards

Table 6-6 presents available regulatory standards or guidelines for the Study Area chemicals of concern. These values will be used in the risk characterization presented in Section 6.5.

6.3.2.1 <u>Maximum Contaminant Levels (MCLs)</u>

MCLs are enforceable standards promulgated under the Safe Drinking Water Act (SDWA) and are designed for the protection of human health. MCLs are based on laboratory or epidemiological studies and apply to all public water systems. A public water system is defined as a system which provides water to the public for human consumption and which has at least 15 service connections or regularly serves an average of at least 25 individuals daily for a minimum of 60 days per year. MCLs are designed for prevention of human health effects associated with lifetime exposure (70-year lifetime) of an average adult (70 kg) consuming two liters of water per day, but also reflect the technical feasibility of removing the contaminant. These enforceable standards also reflect the fraction of the toxicant expected to be absorbed by the gastrointestinal tract.

The federal SDWA standards undergo periodic review by the Environmental Protection Agency and are subject to change as new toxicological information becomes available. Recent proposed standards and criteria are included in Table 6-6 for the chemicals of concern.

Surface waters within the Study Area are listed by the State of Massachusetts as Class B waters, suitable for recreational use and propagation of wildlife. The use of Study Area surface waters as a drinking water supply source is not considered within this assessment. MCLs have been presented on Table 6-6 for reference purposes only.

6.3.2.2 <u>Maximum Contaminant Level Goals (MCLGs)</u>

MCLGs are specified as zero for carcinogenic substances, based on the assumption of nonthreshold toxicity, and do not consider the technical or economic feasibility of achieving these goals. MCLGs are non-enforceable guidelines based entirely on health effects. The MCLs have been set as close to the MCLGs as is considered technically and economically feasible.

6.3.2.3 <u>Ambient Water Quality Criteria (AWQC)</u>

AWQC are not enforceable regulatory guidelines. They may be used for identifying human health risks and acute and chronic toxic effects in aquatic organisms. AWQCs consider acute and chronic effects in both freshwater and saltwater aquatic life, and adverse carcinogenic and noncarcinogenic health effects in humans from ingestion of both water (2 liters/day) and aquatic organisms (6.5 grams/day), from ingestion of water alone (2 liters/day) and from ingestion of aquatic organisms alone (6.5 grams/day). The AWQCs for protection of human health for carcinogenic substances are based on the EPA's specified incremental cancer risk range of one additional case of cancer in an exposed population of 10,000,000 to 100,000 persons (the 10^{-7} to 10^{-5} range).

6.3.2.4 <u>Health Advisories (HAs)</u>

HAs are guidelines developed by the EPA Office of Drinking Water for non-regulated contaminants in drinking water. These guidelines are designed to consider both acute and chronic toxic effects in children (assumed body weight of 10 kg) who consume 1 liter of water per day or for adults (assumed body weight of 70 kg) who consume 2 liters of water per day. Health Advisories are generally available for acute (1-day), subchronic (10-day), and chronic (long-term) exposure scenarios. Health advisory guidelines are designed to consider only threshold effects and, as such, are not used to set acceptable levels of known or probable human carcinogens.

W92194F

6-44

Table 6-6 presents available U.S. Food and Drug Administration (FDA) Action Levels for chemicals in fish tissue and Commonwealth of Massachusetts Drinking Water standards and guidelines, when available.

6.4 Exposure Assessment

The purpose of this section is to evaluate the potential for human exposure to the chemicals of concern identified in the surface water, sediment, and fish tissue samples collected in the Sudbury River Study Area. This section identifies actual or potential routes of exposure, characterizes the exposed populations, and presents the methodology used to estimate the degree or magnitude of exposure.

To determine whether there is an actual exposure or a potential for exposure in the future, the most likely pathways of chemical release and transport as well as the human and environmental activity patterns within the Study Area must be considered. Α complete exposure pathway has three components: (1) a source of chemicals that can be released to the environment; (2) a route of contaminant transport through an environmental medium; and (3) an exposure or contact point for a human receptor. These components are addressed in the following subsections. In the final subsection, exposure scenarios will be developed based on indicator chemical concentrations, chemical fate and mobility, and relevant demographic information. The exposure scenarios will allow quantitative estimation of human intakes as a result of contact with Study Area contaminants.

6.4.1 Sources of Contamination

As detailed in Section 1, several contaminant sources exist within the Sudbury River Study Area:

- o The Nyanza Superfund Site
- o Twelve active or inactive landfill sites)
- o Potential oil/hazardous materials release sites
- o Wastewater discharge sites

Additionally, non-point sources such as road runoff are suspected to have contributed to contaminant concentrations detected in the surface water, sediment, and fish tissue samples taken from the Study Area.

6.4.1.1 The Nyanza Site Contaminant Source Area

The Nyanza Superfund Site is a 35-acre site located on Megunko Road in Ashland, Massachusetts, approximately 22 miles west of Boston. The site was occupied from 1917 through 1978 by several companies involved in manufacturing textile dyes and dye intermediates. During that period, the following wastes were disposed into vaults, tanks, lagoons, and surface drainage systems at various onsite locations:

- o Partially treated process wastewater
- o Chemical sludge from the wastewater treatment process
- o Solid process wastes (chemical precipitate and filtercake
- o Solvent recovery distillation residue
- o Off-specification products

Based on historical monitoring, the EPA has identified the following metals as Nyanza site-specific contaminants:

- o Antimony
- o Arsenic
- o Cadmium
- o Chromium
- o Lead
- o Mercury

Previous Nyanza site investigations indicate that onsite soils, sludges, and sediments were contaminated with heavy metals such as mercury, chromium, lead, and cadmium. Cadmium, chromium, arsenic, mercury, and lead were also repeatedly detected in groundwater at the Site. As discussed in Section 4 and summarized in Tables 6-2, 6-3 and 6-4, all of these metals were detected in surface waters, sediments, and fish tissues samples collected in the Study Area.

The EPA has identified the following organics as Nyanza sitespecific contaminants:

- o Trichloroethene o Chlorobenzene
- o Dichlorobenzenes
- o Aniline

- o 1,2-Dichloroethene
- o Nitrobenzene
- o 1,2,4-Trichlorobenzene
- o Naphthalene
- o Phenolic compounds
- o Benzidine

Based on past Nyanza site investigations, trichloroethene and chlorobenzene were detected in onsite soils and sediments at concentrations exceeding 100 μ g/kg and 1,000 μ g/kg, respectively. Nitrobenzene; the dichlorobenzenes; 1,2,4-trichlorobenzene; aniline; and naphthalene were detected in the 100 to 1,000 μ g/kg

W92194F

concentration range. As reported in this RI report, chlorobenzene; the dichlorobenzenes; trichloroethene; and 1,2-dichloroethene were detected in the sediments/soils of the Eastern Wetlands which border the Nyanza Site. Aniline, the dichlorobenzenes, trichloroethene, and nitrobenzene have been detected in the unconsolidated overburden and bedrock aquifers underlying and downgradient of the Nyanza site (Ebasco, 1991). With the exception of aniline and benzidine, all of these organic contaminants were detected in surface water, sediments, and/or fish tissue samples collected during the Sudbury River study.

6.4.1.2 Offsite Contaminant Sources

Section 4.2.2 described the landfills, oil/hazardous materials release sites, wastewater discharges, and nonpoint sources which contribute to contaminant concentrations detected mav in environmental samples collected from the Sudbury River. Several of the landfills are currently undergoing environmental assessment by the Massachusetts Department of Environmental Protection. These offsite sources may act as sources of the nonsite-specific contaminants detected in the River. For example, the landfills may contribute to the phthalate, PAH, and pesticide/PCB contamination detected in environmental samples. Nonpoint contaminant sources such surface drainage and stormwater runoff from light industrial areas, highways, and railways may contribute to the petroleum related contaminants (benzene and PAHs) and pesticide/PCB contamination detected in the Study Area.

6.4.2 Contaminant Transport and Migration

The fate and migration of Study Area contaminants was discussed in Section 5.0. Summarizing from Section 5.0, the principal mechanisms by which contaminants appear to be migrating from source areas (and potentially resulting in exposure to human and ecological receptors) are as follows:

- o Surface drainage/stormwater runoff from contaminated sites/soils areas (the Nyanza Site) may transport contaminants to surface waters/sediments of the Sudbury River.
- o Chemicals introduced to the Sudbury River may be transported through the surface water in the dissolved phase or adsorbed to entrained sediments.

- o Chemicals in surface waters/sediments may be exchanged between these media through adsorption/desorption processes.
- o Chemical transport to and from surface waters/sediments and the groundwater may occur as the groundwater (possibly) discharges to and is recharged by surface waters.

As described in Sections 4.0 and 5.0 of this report, surface water drainage from the Nyanza site is considered the primary contaminant flow path to the river. Site surficial runoff drains to the Eastern Wetlands to the east of the site and to Chemical Brook to the north. Surface waters from the Eastern Wetlands drain into Trolley Brook which converges with Chemical Brook to form a confluence at the upstream end of the Chemical Brook Culvert. The culvert carries surface waters through Ashland. Outfall Creek is formed by the discharge of surface waters from the culvert. Outfall Creek flows north for a short distance and then discharges to the lower end of the Raceway which in turn discharges to the Sudbury River.

6.4.3 Receptor Identification and Exposure Routes

This section identifies the human receptors potentially exposed to contaminated environmental media and provides the methodology used to estimate the degree or magnitude of exposure.

6.4.3.1 <u>Site Characterization and Receptor Identification Site</u> <u>Characterization</u>

For this report, the Sudbury River Study Area is defined as 33 miles of the Sudbury River drainage basin from the head waters at Cedar Pond in Westborough to the confluence of the Sudbury and Assabet Rivers which form the Concord River in Concord, Massachusetts. As described in Section 3.0, the Study Area is composed of various types of open water bodies (the River, tributaries to the river, several reservoirs and ponds) and associated wetlands which are generally accessible to the public.

Land use along the Sudbury River Study Area varies, as described in Section 3.2.2. Although the land in the vicinity of Cedar Swamp is largely undeveloped, residential, commercial, and industrial properties as well as wetlands and meadows abut the river. The Sudbury River and its tributaries in the Study Area flow through size from 4,000 to several communities that range in The area is classified as rural to urban 65,000 residents. From the Framingham/Wayland town line northward suburban.

W92194F

approximately 9 miles to Route 117, the River is bordered by the Great Meadows National Refuge.

The Sudbury River in Ashland is classified as a Class B surface water, which is protected for the propagation of fish, other aquatic life, and wildlife, and for primary and secondary contact recreation. As described in Section 3.2.1, water use in certain areas of the Study Area is restricted. However, much of the river (and the river front) is used for recreation (boating, hiking, fishing).

While it is understood that some surface water bodies within the Study Area are used more extensively for recreational purposes than others, for the purposes of this risk assessment, it is assumed that all areas of open water are similar, in that persons are likely to swim, fish, or wade in any area. Although the reservoirs are somewhat restricted areas, a certain amount of trespass is expected. Therefore, legal and physical restrictions of the reservoirs are not considered. It is assumed that residents who own property abutting the reservoir areas will use them as recreational resources.

The following tributaries and wetlands are included in the Study Area:

- o The Raceway
- o Trolley Brook
- o The Eastern Wetlands
- o Cold Spring Brook
- o Chemical Brook
- o Chemical Brook Culvert
- o Outfall Creek

The following paragraphs briefly describe these surface water bodies/wetlands and characterize each in terms of the potential for human exposure to surface water/sediments/fish.

Eastern Wetlands

The Eastern Wetlands includes the two small ponds and their outlet to the north, as described in Section 3. This is an area which has unrestricted access and can be used by the public without restriction. The area is wooded, with substantial marsh areas on the eastern sides of the ponds. This characteristic may dissuade public use of the area. However, the outlet to the wetlands, Trolley Brook, passes under a public roadway, flows to Chemical Brook and then into a large culvert to the north of the site.

W92194F

Public exposure and contact with the surface waters and sediments is more likely in these areas.

Chemical Brook Culvert

The Chemical Brook Culvert, referred to as the "culvert", is an underground pipeline constructed of corrugated pipe, concrete pipe, and stone blocks. This culvert passes through several concrete catch basins and receives water from various sources as it passes in a northeasterly direction under the town center of Ashland. The culvert is completely isolated from the general public because of its construction and location. Therefore, the public is unlikely to be exposed to the waters and sediments in the Chemical Brook Culvert.

Outfall Creek

The outfall for the Chemical Brook Culvert is an open creek which runs northeast to the Raceway upstream from its confluence with the Sudbury River. This creek is a moderately flowing stream, varying in depth from 6 to 18 inches. Access to the outfall creek is unrestricted and adjacent woodlands abut several residential properties.

Raceway

The Raceway flows out of Mill Pond separate from the Sudbury River and flows under a large factory building before becoming an open watercourse. The Raceway is a fast moving waterway with a gravel and sand bottom. The Raceway parallels the River, receiving the Chemical Brook outfall creek waters and then rejoining the Sudbury River. Access to the Raceway is also unrestricted and adjacent woodlands abut public and private properties.

Sudbury River

The reaches of the Sudbury River, described in Section 1.2.1, are considered alike in that each reach is subject to unrestricted access by the residential population in the adjoining towns. In the case of some reaches, the potential for interaction increases, especially in the areas of the Wildlife Refuge and areas of the River which are more heavily fished. The characteristics of the Sudbury River range from slow moving and meandering river conditions (including shallow lakes and ponds) to quickly moving riffle/run areas of limited extent. The River is expected to be used for various recreational purposes including fishing, boating, swimming and wading, as well as numerous play activities by children.

W92194F

6-50

Cold Spring Brook

Cold Spring Brook is a tributary to the Sudbury River flowing from the south and connecting to the River at the junction of Reaches 2 and 3. Since this is an upstream location not influenced by the Sudbury River, this stream is considered a separate reach.

Heard Pond

Heard Pond is a moderate-sized eutrophic pond located to the west of the Sudbury River at Reach 7. This pond is nearly surrounded by wetlands. There is no information confirming or disproving the possibility of a hydraulic connection between the River and Heard Pond although a source reported that the pond is affected by the River during flood conditions. Heard Pond is therefore considered a separate reach from the River but not a background location.

Bordering Wetlands

Bordering wetlands within the Study Area are grouped as a separate section. These wetlands include the woodlands and floodplains adjacent to the river and its tributaries which are only inundated during flood-stage river conditions. The bordering wetlands are generally unrestricted from public access and include some private properties which abut the river and reservoirs. The activities which are expected to occur in these areas include hiking, bird watching, and numerous play activities by children, the same as those expected to occur in yards of private homes.

Recreational Facilities

Much of the study area is utilized for recreational purpose although there are no formally regulated recreational zones such as municipal playgrounds or swimming beaches. The only formal recreational area is the Great Meadows National Wildlife Refuge, located at reaches 7 and 8. This area is frequented by numerous visitors, who are restricted to non-intrusive activities such as hiking and bird watching.

There are small boat launching areas in many reaches, including Mill Pond (Reach 2), Heard Pond (Reach 7) and both upstream and downstream of Fairhaven Bay (Reach 9) these areas in particular could be used for swimming or other recreational activities, but it is important to note that except for the MDC Reservoirs No. 1 and 2, the entire study area is available for public access. Children will access water bodies wherever it is convenient, and light boat access can be obtained at nearly any street crossing.

W92194F

6.4.3.2 <u>Receptors of Concern</u>

Based on the contaminant occurrence and distribution, contaminant fate and transport, and water/land use information presented in previous sections of this report, there are two contaminated environmental media which human receptors are likely to contact in the Study Area: surface waters and sediments. Human receptors that could be potentially exposed to Study Area contaminants include:

- o Individuals using the surface water bodies and river front of the Study Area for recreational purposes. These receptors may contact contaminated surface waters/ sediments while wading, boating, hiking, or swimming. Recreational facilities do exist within the Study Area. Additionally, although the use of some of the Study Area surface bodies is restricted, trespass is a possibility.
- o Sports or subsistence fishermen who fish the open surface water bodies of the Study Area (Discussions with local town people indicate that local ethnic groups heavily fish certain areas of the Sudbury River Study Area).
- o Individuals residing along the river front or any of the wetlands investigated during the RI.

6.4.3.3 <u>Exposure Routes</u>

This subsection presents the methodology that will be used to estimate potential exposure doses incurred by a human receptor who may contact contaminated surface waters/sediments in the Study Area or who may consume fish taken from the Study Area.

The exposure scenarios developed in this section are based on estimated parameters such as exposure rates, durations, and frequency. Exposure doses are calculated using mathematical models which account for these estimated parameters as well as for the contaminant concentrations detected. While most of the exposure scenarios presented consider the recreational use of surface water bodies within the Sudbury River Study Area, the sediment exposure scenarios also consider exposures incurred by individuals <u>residing</u> close to river front areas or wetlands occasionally inundated with surface waters. The fish ingestion scenarios consider both subsistence and sports fishermen as receptors of concern. Table 6-8 summarizes the input parameters for the exposure scenarios. Exposure dose calculations and results are presented in Appendix K.

W92194F

TABLE 6~8(1)

.....

1

SUMMARY OF IMPUT PARAMETERS FOR EXPOSURE SCENARIOS MYANIA OPERABLE UNIT 3 NIDDLESEX COUNTY, NASSACHUSETTS

Environmental Medi a	Land/Surface Water Use Scenario	Route of Exposure	Receptor of Concern	Contact Rate	Skin Surface Area (cm ²)	Exposure Time (hrs/day)	Exposure Frequency (days/yr)	Exposure Duration (years)	Exhibit Number
Surface Water	Recreational	Accidental	Child				50	6	
		Ingestion	Teen	0.05 L/hr	NA	2.0 hrs/day	150	12	6-1
	1	}	Adult	1		ļ.	50	12	
1	{	Dermal	Child		3,490		50	6	
	1	Absorption	Teen	NA	6,207	2.0 hrs/day	150	12	6-2
		1	Adult		19,400	1	50	12	
Sediments	Recreational	Accidental	Child	200 mg/day			50	6	
	ł	Ingestion	Teen	100 mg/day	NA NA	NA	150	12	6-1
			Adult	50 mg/day		ļ	50	12	
		Dermal	Child				50	6	
	ļ	Absorption	Teen	500 mg/day	$2,000 \ \mathrm{cm}^2$	NA	150	12	6-3
		[Adult	1			50	12	
]	Residential	Accidental	Child	200 mg/day				6	
1		Ingestion	Teen	100 mg/day	NA	NA	270	12	6-1
			Adult	50 mg/day		1		12	
1		Dermal	Child					6	
	{	Absorption	Teen	500 mg/day	$2,000 \text{ cm}^2$	NA	270	12	6-3
	1		Adult		-			12	
Fish	Subsistence Fishing	Ingestion Adult 0		0.132 kg/day	NA	NA	350	30 6-4	
9	Sport Fishing	Ingestion	Adult	0.054 kg/day	day NA	NA	350	30	- 0~4

NA Not applicable (1) The reader is r

(1) The reader is referred to Appendix K for a detailed explanation and listing of references supporting the use of the input parameters presented.

.

Several types of open water bodies exist within the Study Area: stream flow zones of the Sudbury River; tributaries, such as the Raceway, to the River; and the Eastern Wetlands. Assuming the recreational use of these surface water bodies, the following human exposure scenarios are possible:

- o Accidental ingestion/dermal absorption of contaminants during swimming or wading in surface waters of the Study Area
- o Accidental ingestional/dermal absorption of contaminants in sediments
- o Ingestion of finfish taken from surface water bodies within the Study Area

Recreational Exposure Scenarios: Surface Waters

The exposure dose equation presented in Exhibit 6-1 will be used to estimate an exposure dose resulting from the accidental ingestion of chemicals in surface water during swimming. The receptors of concern for the exposure scenario are a 70-kg adult, a 37-kg adolescent (teenager), and a 15-kg child. The exposure rate, exposure frequency, and exposure duration estimates are NUS estimates based primarily on climatic conditions in the Study Area and the assumption that the adolescent is the receptor most likely to be using the Study Area surface water bodies for recreational purposes. The water ingestion rate (50 mL/hour) is an EPA estimate (EPA, December 1989).

The exposure dose equation for dermal contact with surface water is presented in Exhibit 6-2. The receptor body weight, exposure rate, exposure frequency, and exposure duration assumptions are the same as those assumed for the accidental-ingestion-of-surface-water exposure scenario. It is assumed that the total skin surface area of the adult receptor $(19,400 \text{ cm}^2)$ is exposed to surface water contaminants while swimming. The skin surface area for the adolescent receptor is calculated assuming that the total body surface area $(13,300 \text{ cm}^2)$ is exposed while swimming and the feet and lower legs (20 percent body surface area = 2,660 cm^2) are exposed while wading. The skin surface area for the adolescent receptors presented in Table 6-8 is a time-weighted value assuming that the adolescent receptor swims 50 days/year and wades 100 days/year, respectively. It is assumed that 50 percent of the child's total body surface area $(6,980 \text{ cm}^2)$ is exposed to surface water contaminants while swimming/wading. Contaminant-specific

W92194F

EXHIBIT 6-1

INCIDENTAL INGESTION OF CHEMICALS IN SURFACE WATER AND SEDIMENTS NYANZA OPERABLE UNIT 3

Exposure Dose (mg/kg-day) = <u>C x CR x ER x EF x ED x ABS x 10-6 kg/mg*</u> BW X AVG x 365 days/year

- C = Measured/Estimated Media Concentration (mg/L, water; mg/kg, sediment)
- CR = Contact Rate or Ingestion Rate (IR) (L/hr (0.05L/hr)
 for surface waters; mg/day for sediments (e.g. 200
 mg/day-child). See Table 6-8.
- ER = Exposure Rate or Time (hrs/day) (applicable to surface water only) (2 hrs/day)
- EF = Exposure Frequency (days per year) (Adults 50 days/yr; Teens - 150 days/yr; Child - 50 days/yr)
- ED = Exposure Duration (years/lifetime) (Adults + 12 yrs; Teen -12 yrs; Child - 6 yrs)
- BW = Receptor Body Weight (Kg) (Adult 70kg; Teen 37.2 kg; Child 14.5 kg)
- AVG = Number of years over which the exposure is averaged (70 years for carcinogenic effects, ED for noncarcinogenic effects)
- ABS = Absorption through the gut (unitless)

Volatile organic compounds - 100% Semivolatile organic compounds - 100% (Sediments - PCBs - 30%) Pesticides - 100% (Sediments - high absorption to soils - 30%) Inorganics - 100% (Sediments - Lead - Adults 30%, Children 50%)

Notes:

- * The conversion factor of 10⁴ kg/mg is used for the ingestion of sediments only. Ingestion of water scenario requires no such conversion factor.
- o Example calculations are presented in Appendix K.
- Input parameters for the exposure scenarios are presented in Table 6-8. See Appendix K for contaminant-specific absorption factors.

References:

EPA, March 1991 EPA, June 1989 EPA, December 1989 EPA, May 1989

EXHIBIT 6-2 DERMAL CONTACT WITH CHEMICALS IN SURFACE WATER NYANZA OPERABLE UNIT 3

Absorbed Dose $(mg/kg/day) = C \times PC \times SSA \times ER \times EF \times ED$ BW x AVG x 365 days/year x CF

- C = Measured/Estimated Surface Water Concentration (mg/L)
- PC = Permeability Constant (cm/hour). See Appendix K for contaminant specific PC values. The default value is 8 x 10^4 cm/hr. (EPA, December 1989)
- SSA = Contact Rate (Skin Surface Area Available for Contact cm²). Adults 19,400 cm²; Teens 6,207 cm²; Children 3490 cm²)

ER = Exposure Rate (hours per day) (2 hrs/day)

- EF = Exposure Frequency (days/year) (Adults 50 days/yr; Teen - 150 days/yr; Children - 50 days/yr)
- ED = Exposure Duration (years/lifetime) (Adults 12 yrs; Teens - 12 yrs; Child - 6 yrs)
- BW = Receptor Body Weight: Kg (Adult 70 kg; Teen 13.71 g; Child - 14.51 g)
- AVG = Number of years over which the exposure is averaged (70 years for carcinogenic effects; ED for noncarcinogenic effects)
- CF = Volumetric Conversion Factor for Water (1,000 cm³/L)

Notes:

- o In the absence of contaminant-specific permeability constants, the permeability constant for water (0.0008 cm/hr) may be used. (See Appendix K for contaminant-specific permeability constants used in risk assessment.)
- o Example calculations are presented in Appendix K.
- Input parameters for the exposure scenario are presented in Table 6-8.

References:

EPA, March 1991 EPA, December 1989 EPA, May 1989 permeability constants are used if available. (In accord with EPA Region I guidance, permeability constants, available in the recently published <u>Interim Guidance for Dermal Exposure Assessment</u> ((EPA, March 1991) were used whenever possible.)) In the absence of contaminant-specific permeability constants, the permeability constant for water (0.0008 cm/hr) will be used for organic contaminants. The reader is referred to Exhibit 6-2 and Appendix K for the references supporting the exposure parameter assumptions and permeability constants used to evaluate the dermal-contactwith-surface-water exposure scenario.

Recreational Exposure Scenarios: Sediments

Exhibit 6-1 presents the exposure dose equation for the accidentalingestion-of-sediments exposure scenario. The receptor body weight, exposure frequency, and exposure duration assumptions are the same as those assumed for the surface water exposure scenarios. The sediment ingestion rates are those presented in the EPA Risk Assessment Guidance for Superfund, Volume I (EPA, December 1989) and available supplemental guidance (EPA, March 1991). The contaminant absorption rates are those presented in the Region I Supplemental Risk Assessment Guidance of the Superfund Program (EPA, June 1989).

exposure dose equation for dermal-contact-with-sediment The exposure scenario is presented in Exhibit 6-3. The receptor body weight, exposure frequency, and exposure duration assumptions are the same as those stated for the surface water exposure scenarios. The soil contact rate and contaminant absorption rates are those presented in the Region I risk assessment guidance document. The dermal absorption rate assumed for semi volatile organics other than PAHs and PCBs (10%) is more conservative than the absorption rate assumed for PAHs and PCBs (5%). If the true absorption rate for all semi volatiles is 5%, the absorbed dose for semi volatiles other than PAHs and PCBs may be overestimated. Given the fact that the risks associated with the accidental-ingestion-of-sediment exposure scenario predominate, the use of a 5% or 10% absorption rate for semivolatiles will not significantly alter results of the The dermal absorption factors assumed for risk assessment). mercury (0.01) and methyl mercury (0.77) were provided by the Massachusetts Department of Environmental Protection. The current EPA Region I guidance for the dermal adsorption factor for mercury The assumed rate of 0.01 therefore is more conservative is 0.05. than current EPA Region I guidelines and the risks associated with sediment contact scenarios may be over estimated, but will not significantly alter results of the risk assessment.

W92194F

EXHIBIT 6-3 DERMAL CONTACT WITH CHEMICALS IN SEDIMENTS NYANZA OPERABLE UNIT 3

Absorbed Dose $(mg/kg/day) = C \times SCR \times RAF \times EF \times ED \times 10-6 kg/mg$ BW x AVG x 365 days/year

- C = Measured/Estimated Contaminant Concentration in Sediment
 (mg/kg)
- SCR = Soil/Sediment Contact Rate (mg soil/day of exposure)
 (500 mg/day)
- RAF = Absorption Factor for Soils/Sediments (unitless)
 Volatile Organic compounds 50%
 Semivolile organic compounds 10%
 Polyaromatic hydrocarbons 5%
 Polychlorinated biphenyl compounds 5%
 Pesticides high sorption to soils 5%
 low sorption to soils 50%
 Inorganics (except mercury) negligible
 Mercury 1%
 Methylmercury 77%
- EF = Exposure Frequency (days/year) (Adult 50 days/yr; Teen -150 days/yr; Child - 50 days/yr)
- ED = Exposure Duration (years/lifetime) (Adult 12 yrs; Teen - 12 yrs.; Child - 6 yrs)
- BW = Receptor Body Weight: Kg (Adult 70 kg; Teen 37 kg; Child 14.5 kg)
- AVG = Number of years over which the exposure is averaged (70 years for carcinogenic effects; ED for noncarcinogenic effects).

Notes:

- o Example calculations are presented in Appendix K.
- Input parameters for the exposure scenarios are presented in Table 6-8.
- o See Appendix K for contaminant-specific absorption factors.

References:

EPA, March 1991 EPA, December 1989 EPA, June 1989

Fish Ingestion Exposure Scenario

The ingestion-of-fish exposure scenario (Exhibit 6-4) considers consumption of fish taken from the Study Area by sports and subsistence fishermen. The receptor of concern is an adult receptor (70 kg) who consumes the fish 350 days/year over a 30-year lifetime. Based on EPA guidance, the sports fisherman and subsistence fisherman consume 0.054 kg/day and 0.132 kg/day, respectively. (EPA, March 1991) the fraction of fish taken from the Sudbury River is assumed to be 0.25 and 0.75 for sports and subsistence fishermen, respectively.

Residential Exposure Scenario: Sediments

Bordering wetlands within the Study Area include vegetated wetlands, swamps, and marshes adjacent to the open water area. Unlike the Eastern Wetlands, most of these wetlands are contain no surface water during dry periods.

Because residential areas adjoin or are located in the vicinity of the bordering wetlands, residents of these areas may be in frequent contact with contaminated soils/sediments. Consequently, Table 6-8 includes exposure assumptions for <u>residents</u> exposed to the soil/sediment through the accidental-ingestion and dermal-contact exposure scenarios. All exposure assumptions for the residential exposure scenario are the same as those presented for the recreational exposure scenarios except that the exposure frequency (days/year exposed) increases to 270 days/year.

6.5 <u>Risk Characterization</u>

Risk characterization evaluates the potential for adverse health effects from site media by integrating information developed during the exposure and toxicity assessments. EPA guidelines (EPA, September 1986) for the use of dose-additive models were used to combine the risks for individual chemicals to estimate the cumulative risks for mixtures found in the Study Area, assuming that the toxicological endpoints are the same. An average and reasonable maximum-case scenario are presented based on the evaluation of maximum and average contaminant concentrations.

6.5.1 Methodology for Estimation of Carcinogenic Risks

Carcinogenic risks can be estimated by combining information on the strength or potency of a known or suspected carcinogen (Carcinogenic Slope Factor, Table 6-6) with an estimate of the individual exposure doses of a chemical.

W92194F

EXHIBIT 6-4 INGESTION OF FINFISH NYANZA OPERABLE UNIT 3

Exposure Dose $(mg/kg/day) = C \times IR \times FI \times EF \times ED$ BW x AVG x 365 days/year

- C = Measured/Estimated Contaminant Concentration in Fish
 Flesh (mg/kg).
- IR = Ingestion Rate
 0.054 kg/day for sport fishing
 0.132 kg/day for subsistence fishing
- FI = Fraction of consumed fish taken from Sudbury River
 Sportsmen 0.25
 Subsistence fishermen 0.75
- EF = Exposure Frequency (days/year) (350 days/year)
- ED = Exposure Duration (years/lifetime) (30 years)
- BW = Receptor Body Weight (70 kg)
- AVG = Number of years over which the exposure is averaged
 - o 70 years for carcinogenic effects
 - o ED for noncarcinogenic effects

Notes:

- o Example calculations are presented in Appendix K.
- Input parameters for exposure scenarios are presented in Table 6-8.

References:

EPA, December 1989 EPA, 1986 EPA, March, 1991

W92194F

Carcinogenic risk may be estimated as follows:

 $Risk^{(1)} = (CSF) (Dose)$

Where:

- CSF = Carcinogenic Slope Factor (slope of the doseresponse curve in (mg/kg-day)⁻¹ from Table 6-2).
- Dose = Amount of a contaminant absorbed by a receptor in mg/kg-day.
- ⁽¹⁾ The carcinogenic risk formula presented is applicable to situations where the carcinogenic risk is not expected to exceed 10^{-2} .

The resultant risk value $(1 \times 10^{-6} \text{ or a } 1-\text{in-1},000,000 \text{ chance})$ can also be applied to a given population to determine the number of excess cases of cancer that could be expected to result from exposure $(1 \times 10^{-6} \text{ is one additional case of cancer in } 1,000,000 \text{ exposed persons}).$

The total risk for exposure to multiple compounds is presented as the summation of the risk for the individual contaminants. Risks can be calculated in this manner under the following assumptions:

- o There are no antagonist/synergistic effects between chemicals.
- o All chemicals produce the same result (cancer).
- o Cancer risks from various exposure routes are additive, if the exposed populations are the same (EPA, 1986).

6.5.2 Methodology for Estimation of Noncarcinogenic Risks

Potential health risks resulting from exposure to noncarcinogenic compounds are estimated by comparing the maximum daily dose calculated for an exposure to an acceptable intake dose, such as a chronic or subchronic reference dose. If the ratio between an exposure dose and the RfD exceeds unity, there is a potential health risk associated with exposure to that chemical (EPA, September 1986). The Dose/RfD ratio is not a mathematical prediction of the severity or probability of toxic effects; it is simply a numerical indicator of the potential for adverse effects. The ratio of the exposure dose to the Reference Dose is sometimes

referred to as the Hazard Quotient (HQ). The summation of HQs for several compounds is frequently referred to as the Hazard Index (HI).

Conservatively, a total HI for any exposure route is calculated by summing the Dose/RfD rations (HQs) for the individual chemicals of concern (EPA, September 1986). To provide a better indication of risks, Dose/RfD ratios should be summed according to the target organ affected, for example, the Dose/RfD ratios for those chemicals affecting the liver should be summed separately from those chemicals affecting the nervous system.

The target organ or toxicity endpoint of concern identified on the tables in this section are those that were specified on IRIS or the Health Effects Assessment Summary Tables (EPA, January 1991) as the endpoint of concern for the Reference Dose. Appendix K summarizes the toxicological effects and target organs reported in the literature for the Study Area indicator compounds. The Hazard Index and RfDs are subject to the uncertainties described in this section.

6.5.3 Risk Assessment Results

This section presents the results of the risk assessment conducted for the Sudbury River Study Area. As discussed previously, mercury and methyl mercury are the principal contaminants of concern. Mercury and methyl mercury have been detected above background concentrations in sediments and fish of the Sudbury River and other surface water bodies downstream of the Nyanza Site. Mercury has also been detected in fish tissue samples above FDA Action Levels.

The following subsections present the carcinogenic and noncarcinogenic risk assessment results, assuming the recreational and residential exposure scenarios presented in Section 6.4. Each Sudbury River reach and several surface water bodies are discussed individually.

The reader should note that the hazard quotients for the organic mercury compounds were not considered (i.e. added) during the calculation of hazard indicies presented in the risk results tables. Hazard quotients developed for total mercury were considered in the development of hazard indices.

6.5.3.1 <u>Risk Assessment Results for Background Surface Water and</u> <u>Sediment Exposure Scenarios - River Reach 1 and the</u> <u>Sudbury Reservoir - Recreational</u>

Reach 1 and the Sudbury Reservoir are upstream/upgradient of the Nyanza Site, its surficial drainage, and groundwater contaminant plume. Analyses of samples collected from Reach 1 and the Sudbury Reservoir provide background COC concentrations for the Study Area.

Tables 6-9 and 6-10 present risk assessment results for COC concentrations detected in background sediments and surface waters, respectively. The accidental-ingestion and dermal contact routs of exposure (surface water and sediments) were evaluated assuming recreational land/water use scenarios. Hazard indices calculated for the accidental-ingestion exposure route exceed those calculated for the dermal-contact exposure route. Noncarcinogenic risks estimated for the sediment exposure scenarios exceed those estimated for the surface-water exposure scenarios. Only the hazard quotient for arsenic exceeds 0.1 (accidental ingestion of sediments by a child receptor, reasonable maximum case). The hazard quotients and hazard indices calculated for surface water and sediment exposure scenarios do not exceed unity when maximum or average COC concentrations are evaluated. If hazard indices are summed for the accidental-ingestion and dermal-contact exposure routes for each media and then combined for surface waters and sediment exposures, the total hazard index does not exceed unity. These results indicate that adverse noncarcinogenic health effects would not be anticipated under the conditions of the recreational exposure scenarios defined in Section 6.4.

The cancer risks (CR) estimated for the sediment exposures (combined accidental-ingestion and dermal-contact exposure routes) are 1.8 x 10⁻⁵ and 8.2 x 10⁻⁶ when the reasonable maximum and average-case scenarios are evaluated for all COCs detected. Arsenic, beryllium, and the carcinogenic PAHs are the COCs contributing to the estimated excess lifetime cancer risk. Cancer risks estimated for the surface water exposure scenarios (combined accidental-ingestion and dermal-contact exposure routes) are 1.7 x 10⁴ and 1.5 x 10⁴ when the reasonable maximum- and averagecase scenarios are evaluated. 1,1-Dichloroethene is the only COC contributing to the risk. As a point of reference, the 1 \times 10⁴ to 1 x 10⁴ cancer risk range is often evaluated in the development of health-based standards/criteria and in the determination of cleanup goals at hazardous waste sites.

6-63

6-64

TABLE 6 - 9A RIBK ASBESSMENT RESULTS FOR SEDIMENT EXPOSURE SCENARIOS BACKGROUND - REACH 1 AND SUDBURY RESERVOR NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

							NONCARO	NOGENC	RISK ANA	LYSIS RESL	JLTS		r					······
CONTAMINANTS of CONCERN		TRATION KC	EXPOSURE RECEPTO	FACTOR A = TEEN	P I€D	HAZ	ARD QUO	TIENTS CI	ID	HAZ	ARD QUO	TIENTS; TE	EN	HAZARD QUOTIENTS: ADULT				TOXIC END- POINT
	MAX	AVG	INGESTION	DERMAL CONTACT	(MG - KG -			DE	RMAL NTACT			DERM	IAL ACT		STION	DERI		
				L	DAY	MAX	AVG	MÁX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	1
ABSENIC*	21.1	. 74	111F-08	0 00E + 00	3 00E - 04	1 3E - 01	5 3E - 02	0 0E + 00	0 0E + 00	7 BE - 02	3 2E - 02	0.0E+00	0.0E+00	8.9E - 03	2.9E-03	0.0E + 00	0 0E + 00	SKIN
ANTIMONY		6 05	1 11E-08	0 00E + 00	4 00E - 04	3 2E - 02	2 8E - 02	0 0E + 00	0 0E + 00	19E-02	1 7E-02	0.0E + 00	0.0E+00	1.7E-03	1.5E-03	0 0E + 00	0.0E + 00	BLOOD
CHROMIUM*	55 2	22.4	1 11E - 08	0 00E + 00	1.00E+00	1 OE - 04	4 1E - 05	0 0E + 00	0 0E + 00	6 1E-05	2.5E - 05	0.0E + 00	0.0E+00	5.4E-08	2.2E-08	0.0E + 00	0.0E+00	UVER
LEAD"	248	85 09	5 55E - 07	0 00E + 00		0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	0.0E + 00	0.0E + 00	CNS			
MERCURY*	1 50	0 27	1 11E - 08	5 55E - 08	3 00E - 04	9 7E - 03	1 6E - 03	2 4E - 04	4 1E -05	5 9E - 03	1 OE - 03	2.9E - 04	5.0E ~ 05	5.2E-04	8.8E-05	5.2E - 05	8.6E-06	CNS
ACETONE	0 0 1	0 01	1 11E-08	2 78E - 08	1 00E - 01	1 8E - 07	1 8E - 07	2 3E - 07	2 3E - 07	1 1E-07	1.1E-07	2 8E - 07	2.8E-07	9.8E - 09	9 8E 09	4.9E - 08	4.9E-08	UK
PHENANTHRENE	0 18	0 15	1 11E-08	2 78E - 07	4 00E - 03	7 3E - 05	0 0E - 05	9 1E - 08	8 6E - 06	4 4E - 05	4 2E - 05	1.1E-05	1.0E-05	3.9E - 08	3.7E-06	2.0E-06	1.8E-06	N9
FLUORANTHENE	0 32	0 210	1 11E-08	2 78E - 07	4 00E - 02	1 5E - 05	1 OE - 05	1 6E - 06	1 3E ~ 06	89E-08	6 1E-06	2.2E-06	1.5E-08	7.8E-07	5.4E-07	3.9E - 07	2.7E - 07	L/K
PYRENE	0 27	0 1007	1 11E-08	2 78E -07	3 00E - 02	1 8E - 05	1 1E-05	2 1E - 08	1.4E - 06	1 0E - 05	6.7E-06	2.5E-08	1.7E-08	6.6E - 07	5.9E - 07	4.4E 07	2.9E-07	KIDNEY
CHRYSENE	0 18	0 18	1 11E-08	2 78E - 07		0 DE + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E + 00	0.0E + 00	0 OE + 00	0.0E + 00	0.0E + 00	0.0E + 00	NS
BENZO(B)FLUORANTHEN	02	0 185	1 1 1E-06	2 70E - 07		0 0E + 00	0.0E + 00	0.0E+00	0.0E + 00	0.0E + 00	0.0E + 00	0.0E + 00	0 0E + 00	0.0E + 00	NS			
BENZO(A)PYRENE	0.001	0.001	111E-00	2 70E - 07		0 OE + 00	0 0E + 00	0 0E + 00	0 OE + 00	0.0E + 00	0.0E+00	0.0E + 00	0.0E+00	0.0E + 00	0.0E + 00	0.0E + 00	0.0E + 00	NS
BARIUM	170	80 60	111E-00	0 00E + 00	5 00E - 02	6 5E - 03	2 4E - 03	0 0E + 00	0 0E + 00	4.0E - 03	1.5E - 03	0.0E+00	0.0E+00	3.5E-04	1.3E-04	0 0E + 00	0.0E + 00	BLOOD
BERYLUUM	18	0 85	1 11E - 08	0 00E + 00	5 00E - 03	6 6E - 04	3 1E - 04	0 0E + 00	0 0E + 00	4.0E-04	1.9E-04	0.0E+00	0.0E+00	3.5E – 05	1.7E-05	0 0E + 00	0 0E + 00	NS
COPPER	340 4	73 02	1 1 1 E - DE	0 00E + 00		0 0E + 00	0.0E + 00	0 0E + 00	0.0E + 00	0.0E+00	0.0E + 00	0.0E + 00	0 0E + 00	0.0E + 00	NS			
MANGANESE	1840	434 51	1 1 1 E - OB	0 00E + 00	1 00E - 01	3 OE - 02	7 9E - 03	0 0E + 00	0 0E + 00	1.8E - 02	4.8E-03	0.0E + 00	0.0E+00	1.6E – 03	4.3E-04	0 0E + 00	0.0E + 00	CNS
NICKEL	51	11 35	1116-00	0 00E + 00	\$ 00E - 02	4 7E - 03	1 OE - 03	0 0E + 00	0 0E + 00	2.0E-03	6 3E - 04	0.0E+00	0.0E+00	2.5E-04	5.6E - 05	0.0E + 00	0.0E + 00	le w
SELENUM	31	1 04	1 1 1E-08	0 00E + 00	6 00E - 03	1 1E - 03	3 8E - 04	0 0E + 00	0 0E + 00	8 9E - 04	2 3E - 04	0.0E + 00	0.0E+00	0.1E-05	2.0E – 05	0.0E + 00	0.0E + 00	SELENOS
VANADIUM	48 45	20.18	1 11E-08	0 00E + 00	7 OOE - 03	1 3E - 02	5 3E - 03	0 0E + 00	0 0E + 00	7 7E - 03	3 2E – 03	0.0E + 00	0.0E+00	6.6E 04	2.8E-04	0.0E + 00	0.0E+00	NS
ZINC	829	133 5	1 1 1 E - 08	0 00E + 00	2 00E - 01	5 7E - 03	1 2E - 03	0 0E + 00	0 0E + 00	3.5E - 03	7.4E-04	0.0E + 00	0.0E+00	3.1E-04	6.5E-05	0 0E + 00	0.0E + 00	BLOOD
NYANZA SITE CONTAMINA	NTS			HAZAPOIN	жx — —	1 7E -01	6 3E - 02	24E-04	4 1E-05	1.0E-01	5 OE - 02	2.9E-04	5.0E-05	9.1E-03	4.4E-03	5.2E - 05	8.8E-06	7
OTHER SUDBURY RIVER C	ONTAMIN	ANTS		HAZARDING	XE X	0 1E - 02	1 9E - 02	1 3E - 05	1 1E - 05	3 7E - 02	1 1E-02	1.6E - 05	1.4E-05	3.3E - 03	1.0E - 03	2.8E 06	2.4E-08	
	20 1				¥ X	2 3E - 01	1.05-01	2 6F - 04	5 3E - 05	1 AE - 01	a 26_02	3 1E - 04	8 4E - 05	1 2F - 02	54E-03	5 5E05	1 1E - 05	

TOXICITY ENDPOINTS ABBREVIATIONS NS NOT SPECIFIED L/X: UVER AND KIDNEY BW: BODY WEIGHT CNS: CENTRAL NERVOUS SYSTEM

TABLE 6-98 RISK RESULTS FOR SEDIMENT EXPOSURE BACKGROUND NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

!

			·	CARCINOGE		NALYSIS I	RESULTS		
CONTAMINANTS of		TRATION	EXPOSURE	FACTOR		CANCER F	RISKS		CANCER SLOPE FACTOR
CONCERN	MAX	AVG	INGESTION	DERMAL CONTACT	ACCIDENTAL INGESTION		DERM CONT	AL	(MG/KG/D) 1 / WEIGHT
					MAX	AVG	MAX	AVG	EVIDENCE
ARSENIC*	21 1	8 74	3 64E-07	0.00E+00	1.4E-05	5.7E-06	0.0E+00	0.0E+00	1.80E+00 A
ANTIMONY*	69	8 05	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
CHROMIUM*	55 2	22 4	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
LEAD*	248	85 09	1.82E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MERCURY*	1.59	0 27	3.64E-07	1.51E-08	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ACETONE	0.01	0.01	3.64E-07	7.56E-07	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
PHENANTHRENE	0.16	0.15	3.64E-07	7.56E-08	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
FLUORANTHENE	0.32	0 219	3 64E-07	7.56E-08	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
PYRENE	0.27	0.1807	3 64E-07	7.56E-08	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
CHRYSENE	0.18	0.18	3.64E-07	7.56E-08	3.8E-07	3.4E-07	7.9E-08	7.0E-08	5.80E+00 B2
BENZO(B)FLUORANTHEN	0.2	0.185	3.84E-07	7.56E-08	4.2E-07	3.9E-07	8.8E-08	8.1E-08	5.80E+00 B2
BENZO(A)PYRENE	0.091	0 091	3.64E-07	7.56E-08	1.9E-07	1.9E-07	4.0E-08	4.0E-08	5.80E+00 B2
BARIUM	178	66.68	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
BERYLLIUM	1.8	0 85	3 64E-07	0.00E+00	2.8E-06	1.3E-06	0.0E+00	0.0E+00	4.30E+00 A
COPPER	340.4	73.02	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MANGANESE	1640	434 51	3 64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
NICKEL	51	11.35	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
SELENIUM	3.1	1.04	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
VANADIUM	48.45	20.19	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ZINC	629	133.5	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0F+00	

NYANZA SITE CONTAMINANTS	CANCER RISK	1.4E-05	5.7E-06	0.0E+00	0.0E+00
OTHER SUDBURY RIVER CONTAMINANTS	CANCER RISK	3.8E-06	2.2E-08	2.1E-07	1.9E-07
ALL CHEMICALS OF CONCERN	CANCER RISK	1.8E-05	8.0E-06	2.1E-07	1.9E-07

FINAL

W92194F

TABLE 0 - 10A RISK ASSESSMENT RESULTS FOR SURFACE WATER EXPOSURE SCENARIOS BACKGROUND - REACH 1 AND SUDBURY RESERVOIR NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

	T						NONCAR	NOGENC	RISK ANAL	YSIS RESUL	.15							
CONTAMINANTS of CONCERN	CONCENTR MG/	NG NG	EPPOSUAE RECEPTOR	- TEEN	PFD (MG - KG -	HAZA		ENTS: CHI	ம	HAZAF	ID QUOTIEN	ITS: TEEN		HAZARD		is: Adult		TOXICITY END- POINT
	MAX	AVG	INGESTION	DEFINAL CONTACT	DAY)	ACCIDEN	TAL	DERM	AL ICT	ACCIDEN		DERIMA	с. .т	ACCIDENTA	AL	DERMA	L ST	
						Max	DVA	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	
LEAD.	0 0211	0 004	1 11E-03	1 38E - 04		0 0E+00	0 0E+00	0 0E+00	0 0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	CNS
1,1-DICHLOROETHENE	0 003	0 0020	1 118-03	2 19E-03	9 00E - 03	3 0E - 04	26E-04	34E-04	29E-04	3.7E-04	3.2E - 04	7.3E-04	6.3E-04	6.5E-05	5.7E-05	4.0E-04	3.5E-04	L/K
BARIUM	0 0231	0 0157	1 11E-03	1 30E - 04 1 30E - 04	5 00E - 02	4 2E - 04 0 0E + 00	29E-04 00E+00	29E-05 00E+00	20E-05 00E+00	5.1E-04 0.0E+00	3.5E-04 0.0E+00	6.4E-05 0.0E+00	4.3E-05 0.0E+00	9.0E-05 0.0E+00	6.1E-05 0.0E+00	3.5E-05 0.0F+00	2.4E-05 0.0E+00	BLOOD NS
MANGANESE	0 111	0 0808	1 11E-03	1 38E-04	1 00E-01	1 0E-03	74E-04	7 1E-05	5.1E-05	1.2E-03	9.0E-04	1.5E-04	1.1E-04	2.2E-04	1.8E04	8.4E-05	6.1E-05	CNS
SILVER	0 0177	0 0051	1 11E-03	1 38E - 04 1 38E - 04	2 00E - 02 3 00E - 03	6 1E-04	2 3E - 04 1 4E - 03	36E-05	10E-05	9.8E-04 6.3E-03	2.6E - 04 1.8E - 03	1.2E-04 7.6E-04	3.5E-05 2.2E-04	1.7E-04 1.1E-03	5.0E-05 3.1E-04	6.7E-05 4.3E-04	1.9E-05 1.2E-04	ARGYRIA
ZINC	0 0082	0 0062	1 11E-03	1 38E-04	2 00E-01	2 8E - 05	2 6E - 05	2 0E - 08	2 0E - 06	3.4E-05	3.4E-05	4.3E-08	4.3E-08	6.1E08	0.1E-08	2.4E-08	2.4E-06	BLOOD
NYANZA SITE CONTAMINA	ANTS.			HAZARO INI	DEX	0 0E+00	0 0E + 00	0 0E + 00	00E+00	00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0 0E+00	0 OE + 00	
OTHER SUDBURY RIVER	CONTAMINA	175		HAZAFID INI	DEX	7 7E -03	3 0E - 03	8 8E - 04	4 8E - 04	9 4E - 03	3.6E - 03	1.9E-03	1.0E-03	1.7E-03	6.4E-04	1.0E-03	5.8E-04	
ALL CHEMICALS OF CON	LL CHEMICALS OF CONCERN HAZAFD INDE			DEX	7.7E-03	3.0E-03	6.6E-04	4.8E-04	9.4E-03	3.8E 03	1.9E-03	1.0E-03	1.7E-03	8.4E-04	1.0E-03	5. <u>8E</u> -04]	
TOXOCITY ENDPOINT					NOPOINTS A	BREVIATIO	: en c	NS: NOT S CNS: CEN	PECIFIED	L/K: LIV DUS SYSTE	er and Kid M	NEY						

Ń

6-66

.

(

T/ _E 6-10B RISK RESULTS FOR SURFACE WATER BACKGROUND NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

					CARCINO	GENIC RIS	KRESULTS	3	
CONTAMINANTS of CONCERN	CONCENTR/ MG/	ATION KG	EXPOSURE	FACTOR		CANCEP	RISKS		CANCER SLOPE FACTOR
	MAX	AVG	INGESTION	DERMAL CONTACT	ACCIDEN	TAL DN	DERM CONTA	AL	(MG/KG/D)-1 /
					MAX	AVG	MAX	AVG	WEIGHT OF EVIDENCE
LEAD*	0.0211	0.004	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
1,1-DICHLOROETHENE	0.003	0.0026	3.02E-04	6.70E-04	5.4E-07	4.7E-07	1.2E-06	1.0E-06	6.00E-01 C
COPPER	0.0231	0.0157	3.02E-04 3.02E-04	4.21E-05 4.21E-05	0.0E+00	0.0E+00 0.0E+00	0.0E+00	0.0E+00	
MANGANESE NICKEL	0.111	0.0808 0.0051	3.02E-04 3.02E-04	4.21E-05 4.21E-05	0.0E+00 0.0E+00	0.0E+00 0.0E+00	0.0E+00 0.0E+00	0.0E+00 0.0E+00	
SILVER ZINC	0.0169 0.0062	0.0047 0.0062	3.02E-04 3.02E-04	4.21E-05 4.21E-05	0.0E+00 0.0E+00	0.0E+00 0.0E+00	0.0E+00 0.0E+00	0.0E+00 0.0E+00	
NYANZA SITE CONTAMIN	ANTS*		CANCER RI	SK	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
OTHER SUDBURY RIVER CONTAMINANTS				SK	5.4E-07	4.7E-07	1.2E-06	1.0E-06	
ALL CHEMICALS OF CON	CERN			SK	5.4E-07	4.7E-07	1.2E-06	1.0E-06	

BEPH: BIS(2ETHYL HEXYL)PHTHALATE

FINAL

1

W92194F

6.5.3.2 <u>Risk Assessment Results for Background Fish Ingestion</u> <u>Scenarios - Sudbury Reservoir, Southville Pond, and Cedar</u> <u>Swamp Pond</u>

The Sudbury Reservoir, Southville Pond, and Cedar Swamp Pond are located upstream of the Nyanza Site. Tables 6-11, 6-12, and 6-13 present risk assessment results assuming that sports fishermen and subsistence fishermen consume fish taken from these surface water bodies.

The Sudbury Reservoir

Hazard indices calculated for the subsistence fishermen consuming fish taken from the Sudbury Reservoir (Table 6-11) exceed unity the reasonable maximum case is evaluated. when (Mercury (Max#HQ=5.3) and methyl mercury (Max#HQ=2.9) are the principal COCs contributing to the risk. The hazard index calculated for subsistence fishermen also exceeds unity when average contaminant concentrations are evaluated. The hazard index calculated for COCs affecting the central nervous system and the kidney (mercury, methyl mercury, manganese) exceeds unity when the subsistence fisherman is the receptor of concern. Hazard indices calculated for other toxic endpoints (or target organs) do not exceed unity. These results indicate that adverse health effects are possible for subsistence fishermen (reasonable maximum- and average-case) routinely ingesting fish taken from the Sudbury Reservoir. Hazard indices calculated assuming that a sports fisherman is the receptor of concern do not exceed unity. However, the maximum and average mercury concentrations detected in fish tissue samples exceed the current FDA Action Level for mercury (1 mg/kg) by less than a factor of 2. The maximum concentration of methyl mercury (C_____= 0.652 mg/kg) is approximately one-half the Action Level.

The excess lifetime cancer risks estimated for sports fishermen and subsistence fishermen exceed 1×10^{-5} and 1×10^{-4} , respectively, when average or maximum COC concentrations are evaluated. The principal contaminants contributing to the risk are the pesticides and PCBs which have not been identified as predominant Nyanza Site contaminants. One Nyanza Site contaminant, arsenic, does contribute to the estimated cancer risk.

The Southville Pond

Hazard indices calculated for COC concentrations in the Southville Pond fish tissue samples (Table 6-12) exceed unity only when the subsistence fisherman is evaluated as the receptor of concern.

W92194F

TABLE (RISK ASSESSMENT RESULTS FOR FISH INCL. STION EXPOSURE SCENARIOS SUDBURY RESERVOR NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

						NONCARCIN	OGENIC RISK A	NALYSIS RESULT	19				CARCINOG	ENIC RISK	ANALYSIS F	ESULTS	
CONTAMINANTS OF CONCERN	CONCEN MQ/	KO	EXPOSURE	FACTOR	RFD (MG KG		HAZARD	QUOTIENTS		TOXICITY END- POINT	EXPOSURE	FACTOR	CA	NCER FISK	5		CANCER SLOPE FACTOR
	MAX	AVG	SPORTS	BUBBB- -TENCE	DAY)	SPORTSF	SHERMEN	SUBSIS	STENCE		SPORTS FISHER-	SUBSIS- TENCE	SPORT F	SHERMEN	SUBSIST FISHEF	TENCE	(MG/KG/D)~1
ļ			MEN	MEN		MAXIMUM	AVERAGE	MAXIMUM	AVERAGE	1	MEN	MEN	MAXIMUM	AVERAGE	MAXIMUN	AVERAGE	EVIDENCE
PHENOL *	0.000	0.064	1 875 - 04	1 30F - 03	6 00F-01	17F-05	175-06	1.3 - 04	13E-04	FETUS	7.93E-05	581E-04	0.0E+00	0.0E+00	0.0F+00	0.0E+00	
ARSENIC	0.00	0.04	105-04	1 3F -03	100F-04	375-02	37E-02	2万-01	275-01	SKIN	7.93E-05	5.81E-04	8.65-08	8.6E-08	6 3E 05	6.3E-05	1.80E+00.4
CHROMIUM*	2 64	1 1 14	185-04	1 30E - 03	1 00E+00	5 3E-04	2 IE-04	39E-03	1.5E03	LIVER	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MERCURY	1 10	0.00	1 8E-04	1 36E - 03	3 00E - 04	7 3E 01	5 5E - 01	53E+00	4.0E+00	CNS	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
METHYL MERCURY*	0 652	0 485	1 85E - 04	1 30E-03	3 00E ~ 04	4 Œ01	3 0E - 01	2 9E + 00	2 Æ+00	CNS	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ENDOSULFAN SULFATE	0 0045	0 00275	1855-04	1 36E - 03	5 00E ~ 05	1 TE-02	10E-02	12E-01	7 5E02	NS	7.93E-05	5 81E-04	0.0E+00	0.0E+00	0 0E + 00	0.0E+00	
DIELDRIN	0 0005	0 0005	1 85E - 04	1 36E - 03	5 00E ~ 05	16E-03	16E-03	14E-02	14E-02	LIVER	7.93E-05	5.81E-04	6.3E-07	6.3E-07	4.6E-08	4.6E-08	1.60E+01 E
4,4DOD	0 007	0 00375	1 85E-04	1 36E - 03		000€+00	0 0E + 00	0 0E + 00	0 0E + 00	LIVER	7 93E-05	5.81E-04	1.3E-07	7.1E08	9.6E-07	5.2E ~07	2.40E01 B
4,4-DDE	0.041	0.01956	1 8E-04	1 36E - 03		0 0E+00	0 0E + 00	0 0E + 00	0 0E + 00	LIVER	7.93E-05	5.81E-04	1.1E-08	5.3E-07	8.1E-08	3.9E-08	3.40E-01 E
4,4-DOT	0 002	0 0000	1 80E - 04	1 36E - 03		0 0€+00	0 0E + 00	0 0E + 00	0 0E+00	LIVER	7.93E-05	5.81E-04	5.4E-08	2.Æ-08	4.0E-07	1.8E-07	3.40E-01 E
ALPHA-CHLORDANE	0 001	0 001	1 80E-04	1 36E - 03	0 OCE ~05	3.1€-03	3 1E-03	2 3E - 02	2.3E-02	LIVER	7.93E-05	5.81E-04	1.0E-07	1.0E 07	7.6E~07	7.6E~07	1.30E+00 E
AROCLOR-1280	0 095	0 06375	1 8EE-04	1 36E - 03		0 0E+00	0 0E + 00	0 0E + 00	0.0E+00	NS	7.93E-05	5.61E-04	5.6E - 05	3.9E05	4.3E-04	2.9E~04	7.70E+00 E
MANGANESE	2 4 3	1 062	185E-04	1 36E - 03	1 00E ~ 01	45€-03	2.0E-03	3 3E - 02	1.4E-02	CNS	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
BELENIUM	1.	0 46	1 80E - 04	1 3E - 03	6 00E 03	87E-02	17E-02	4 9E - 01	1.Æ-01	SELENOSIS	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
VANADIUM	0 76	0 325	1 85E - 04	1 3E - 03	7 00E ~ 03	2 0E - 02	8 6E - 03	15E-01	8 3E-02	NS	7.93E-05	5.61E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ZINC	61 62	32 399	1 80E - 04	1 36E - 03	2 00E ~ 01	57E-02	3.0E - 02	4 2E - 01	2 Æ - 01	BLOOD	7.93E-05	5.61E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	L
NYANZA SITE CONTAMIN	ANTS*		HAZARD IN	DEX		7 6 E - 01	5 9E -01	5 6€ + 00	4.3E+00		CANCER RI	SKS:	8.6E-06	6.6E-06	6.3E - 05	6.3E 05]
OTHER SUDBURY RIVER	CONTAMIN	NTS	HAZARD IN	DEX		1.7E-01	7.3E-02	1.2E+00	5.3E-01		CANCER RI	SKS:	6.0E-05	4.0E-05	4.4E-04	3.0E - 04	
ALL CHEMICALS OF CON	CERN		HAZARDIN	DEX		● 3E -01	6 6E - 01	8 9E + 00	4.6E+00		CANCER RI	9K8:	8.9E - 05	4.9E-05	5.0E - 04	3.6E-04	

BEPH:BIS(2-ETHYL HEXYL)PHTHALATE

TORCITY ENDPOINT ABBREVIATIONS.

LAK LIVER AND KIDNEY

NS NOT SPECIFIED BW. BODY WEIGHT CNS: CENTRAL NERVOUS SYSTEM

FINAL

.

TABLE 6-12 RISK ASSESSMENT RESULTS FOR FISH INGESTION EXPOSURE SCENARIOS SOUTHVILLE POND NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

	1					NONCARCIN	DCIENIC RISK ÅN	IALYSIS RESULT	8				CARCINOG	ENIC RISK A	NALYSIS RI	ESULTS	
CONTAMINANTS OF CONCERN		RATION KO	EXPOSURE	FACTOR	PFD (MG - KG -		HAZARD	QUOTIENTS		TOXICITY END- POINT	EXPOSURE	FACTOR	са	NCER RISKS	3		CANCER SLOPE EACTOR
	MAX	AVG	SPORTS FIGHER-	9U8819- - TENCE FISHER-	DAY)	SPORTSFI	SHERMEN	SUBSISTENCE FISHERMEN			SPORTS SUBSIS- FISHER- TENCE MEN FISHER-		SPORT FISHERMEN		N SUBSISTENCE FISHERMEN		(MG/KG/D)-1 / WEIGHT OF
			MEN	MEN		MAXIMUM	AVERAGE	MAXIMUM	AVERAGE	-		MEN	MAXIMUM	AVERAGE	MAXIMUM	AVERAGE	EVIDENCE
CHROMIUM*	07	0 254	1 85E - 04	1 36E - 03	1 00E+00	1 3E - 04	4 7E - 05	9 5E -04	34E-04	LIVER	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
LEAD*	077	0 230	1 80E - 04	1 3E -03		0 0E+00	0 0E+00	0 OE + 00	0 0E + 00	CNS	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	1
MERCURY*	0 89	0 315	1 89E - 04	1 36E - 03	3 OCE -04	8 5E -01	1 9E - 01	4 0E + 00	1.4E+00	CNS	7.93E-05	5.81E-04	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	
COPPER	31	0 887	1 85E - 04	1 36E - 03		0 0E+00	0 0E + 00	0 0E+00	00E+00	NS	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MANGANESE	0.36	0 30	1 85E-04	1 36E-03	1.00E-01	7.0E-04	7.0E-04	52E-03	52E-03	CNS	7.93E-05	5.61E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
SELENIUM	0 15	0 087	1 85E-04	1 36E -03	6 00E - 03	5 5E - 03	3 E - 03	4 1E - 02	2 6E - 02	SELENOSIS	7.93E-05	5.81E-04	0 0E+00	0.0E+00	0.0E+00	0.0E+00	
ZINC	214	9 307	1 85E - 04	1 38E - 03	2 00E -01	2 Œ - 02	8.6E - 03	1.5E -01	8.3E-02	BLOOD	7.93E - 05	5.81E-04	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	
NYANZA SITE CONTAMIN	INTS.		HAZARD	DEX		5.5E-01	1.9E-01	4.0E+00	1.∉E+00		CANCERR	18K9:	0.0E+00	0.0E+00	0.0E+00	0.0E+00	7
OTHER SUDBURY RIVER	CONTAMIN	etw	HAZARD IN	DEX		2 GE - 02	1 3E - 02	19E-01	9.5E-02		CANCERR	ISKS:	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ALL CHEMICALS OF CON	ICEFIN		HAZARO I	DEX		87E-01	2 1E-01	4 2E+00	1.5E+00		CANCER R	ISKS:	0.0E+00	0.0E+00	0.0E+00	0.0E+00	

W92194F

BEPH:BIS(2-ETHYL HEXYL)PHTHALATE

TORCITY ENDPOINT ABBREVIATIONS:

NS: NOT SPECIFIED BW: BODY WEIGHT CNS: CENTRAL NERVOUS SYSTEM

L/K. LIVER AND KIDNEY

1

.

(TABLE V									1		
í						PISK ASSES	SMENT RESULTS	FORFISH	JTION EXPOSU	E SCENARIO	3								
								CEDAR SWAN	P POND										
							NY	ANZA OPERAB	LE UNIT 3										
							MDDLESE	K COUNTY, MAS	SBACHUSETTS										
	Ţ	NONCARCINDRENIC RISK ANALYSIS RESULTS											CARCINOGENIC RISK ANALYSIS REGULTS						
CONTAMINANTS OF	CONCEN	TRATION	EXPOSURE FACTOR		NFD (MQ-	NONCARCINOGENIC RISK CHARACTERIZATION		NONCAPICINIDIENIC RISK CHARACTERIZATION		TOXICITY END	EXPOSURE	FACTOR-					CANCER		
CONCERN											CANCER RISKS BLOPE								
			sponts Filten-	 - TENCE FIDHER-	KQ - DAY)	HAZAPO OLIOTENTS RECEPTOR = SPORTSFISHERMEN		HAZARD QUOTENTS RECEPTOR - SUBSIGITENCE FIBHERMEN		POINT									
	MAX	AVO									Sports Fisher- Men	BUBSES- TENCE FISHER-	SPORT FISHERMEN		SUBSISTENCE FISHERIMEN		(MG/KG/D)-1 /		
	MG/MG	MG/NG												i		· · · · · · · · · · · · · · · · · · ·	WEIGHT OF		
		l	MEN	MEN		MAIGMUM	AVERAGE	IMAXIMUM	AVEPLACE	·		MEN	MAXMUM	AVERAGE	MAXIMUM	AVERAGE	EVIDENCE		
ANTIMONY*	0.00	0 586	1 85E - 04	1 34E - 03	4 00E - 04	366-01	2 6 - 01	2 EE + 00	1.9E+00	BLOOD	7.83E-05	5.81E04	0.0E+00	0.0E+00	0.0E+00	0.0E+00			
CHROMIUM	0.0	0211	1 80E -04	1 26-03	1 00E+00	12-04	3 E - 05	● Æ-04	2 9E-04	LIVER	7.93E-05	5.61E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	1		
LEAD*	1 17	0 12	1 85E-04	1 3E-03		0 0E+00	0 Œ + 00	0 0E + 00	0.0E+00	CNS	7.93E-05	5.01E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00			
MERCURY	•	0732	1 85E - 04	1 34E - 03	3 OCE -04	5 9E + 00	4 年-01	4 E +01	3.3E+00	CNS	7.93E - 05	5.61E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	,		
NICKEL		0 494	1 858 - 04	1 JE - 03	2 00E - 02	5 5E - 02	4 EE - 03	4 1E-01	3 Œ-02	BW	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00			
SELENIUM		0312	18E-04	1 3E -03	6 00E - 03	1 5 -01	1 2E - 02	1 1E+00	6 5E ~ 02	SELENOSIS	7.83E-05	5.61E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00			
SILVER	22	0 184	1 8E-04	1 JE - 03	3 00E - 03	1 -01	1 OE - 02	9 E - 01	74E-02	ARGYRIA	7 93E - 05	5.81E - 04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	,		
THALLIUM	0.00	0 048	1 ME - 04	1 3E-03	7 00E - 05	2 1E -01	t 3E −01	15E+00	9 5E - 01	NS	7.83E-05	581E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	, i		
ZINC	50 (1 1 105	1 0E-04	1 3E - 03	2 00E -01	8 5 - 02	9 Z - 03	3 🕮 - 01	6.6E-02	BLOOD	7.93E-05	5.61E-04	0.0E+00	0.0E+00	0 0E + 00	0.0E+00	1		
C																	יבי בי		
NYANZA SITE CONTAMINANTS" HAZARD INDEX				U 3E + 00	7 1E-01	4 CE + 01	5.2E+00			rsks:	0.0E+00	0.0E+00	0.0E+00	0.0E+00					
OTHER SUDBURY RIVER CONTAMINANTS HAZARD INDEX				8 CE - 01	1 6E - 01	4 ≪ +00	1.2E+00		CANCER RISKS: 0.0E+00 0.0E+00 0.0E+00 0.			0.0E + 00	,						
ALL CHEMICALS OF CONCERN HAZAPO INDEX				0 0E + 00	8 6 € -01	5 1E+01	6 € + 00		CANCERR	ISKS:	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00)				

، W92194F

BEPH:BIS(2-ETHYL HEXYL)PHTHALATE

TORCITY ENDPOINT ABBREVIATIONS

NS: NOT SPECIFIED BW: BODY WEIGHT CNS: CENTRAL NERVOUS SYSTEM

•

LAK: LIVER AND KIDNEY

Mercury is the principal contaminant contributing to the risk; the hazard quotient for this contaminant exceeds unity when maximum (HQ=4) or average (HQ=1.4) concentrations are evaluated. These results suggest that adverse noncarcinogenic health effects are possible for the subsistence fisherman routinely consuming fish taken from the Southville Pond. However, the hazard index calculated for the average-case scenario only slightly exceeds unity. Additionally, the maximum concentration of mercury detected in the fish tissue samples (0.89 mg/kg) approaches but does not exceed the FDA Action Level (1 mg/kg) for mercury in fish. Methyl mercury was not detected in the fish tissue samples.

Cancer risk estimates are not presented for the Southville Pond because Cancer Slope Factors (CSFs) are not available for any of the COCs detected. Of the chemicals listed on Table 6-12, only lead has been designated a carcinogen (Class B-2) by the EPA.

Cedar Swamp Pond

Hazard indices calculated for sports fishermen and subsistence fishermen consuming fish taken from Cedar Swamp Pond (Table 6-13) exceed unity when the reasonable maximum case is evaluated. The hazard index calculated for subsistence fishermen also exceeds unity when average contaminant concentrations are evaluated. Mercury is the principal contaminant contributing to the risk; the hazard quotient calculated for mercury exceeds unity in the aforementioned cases (HQs = 43, 3.3, 5.9). The hazard quotient presented for antimony also exceeds unity when the subsistence fisherman is considered the receptor of concern. However, it should be noted that antimony was detected very infrequently in the fish tissue samples collected from the Sudbury River Study Area. These results suggest that adverse noncarcinogenic health effects are possible for the subsistence fisherman routinely consuming fish taken from the Cedar Swamp Pond. The maximum concentration of mercury detected in the Cedar Swamp fish tissue samples (C_____9.6 mg/kg) exceeds the FDA Action Level for mercury in fish. However, the average mercury concentration $(C_{reg}=0.732 \text{ mg/kg})$ is below the FDA Action Level. Methyl mercury was not detected in fish tissue samples.

Cancer risk estimates are not presented for the Cedar Swamp Pond because CSFs are not available for any of the COCs detected. Of the COCs listed on Table 6-13, only lead has been designated a carcinogen (Class B-2) by the EPA.

W92194F

6-72

6.5.3.3 Risk Assessment Results for River Reach 2 and Mill Pond

As described in Section 1.0, Reach 2 extends from the Pleasant Street impoundment to the Union Street (Route 135) bridge in Ashland. The Reach is directly impacted by Nyanza Site discharges downstream of Mill Pond and is geographically the most proximate of the Reaches to the Nyanza Site. In comparison to background, several Nyanza Site contaminants were detected in Reach 2 surface waters and sediments:

- o 1,2-Dichloroethene
- o 1,2-Dichlorobenzene
- o 1,2,4-Trichlorobenzene
- o Arsenic
- o Lead
- o Chlorobenzene
- o 1,4-Dichlorobenzene
- o Naphthalene
- o Chromium
- o Mercury
- o Monomethyl mercury

The average concentration of mercury in Reach 2 sediments $(C_{evg}=3.81 \text{ mg/kg})$ is 14 times the average concentration of mercury in background sediment samples $(C_{evg}=0.27 \text{ mg/kg})$. The maximum concentration of monomethyl mercury for Study Area sediments was detected in a sample collected from Reach 2. The maximum concentration of cadmium $(C_{max}=7.2 \text{ ug/L})$, lead $(C_{max}=66.5 \text{ ug/L})$, and beryllium $(C_{max}=7.5 \text{ ug/L})$ in surface waters exceeds proposed Federal SDWA standards or Action Levels. (Federal SDWA standards are mentioned as a point of reference only; surface waters of the Study Area are not used as a domestic water supply source.)

Tables 6-14 and 6-15 present risk assessment results for COC concentrations detected in sediments and surface waters of Reach 2. The accidental-ingestion and dermal-contact routes of exposure (surface waters and sediment) were evaluated assuming recreational land/water use scenarios. The hazard quotients and hazard indices calculated for surface-water and sediment exposure scenarios approach 0.5 but do not exceed unity in any of the cases presented. Only the hazard quotient for mercury and manganese exceed 0.1 (accidental ingestion of sediments, surface waters, child receptor/teen receptor, reasonable maximum case). If hazard indices are summed for the accidental-ingestion and dermal-contact exposure routes for each medium and then combined for surface water

W92194F
TABLE 8-14A RISK ASSESSMENT RESULTS FOR SEDIMENT EXPOSURE SCENARIOS REACH NO 2 NYANZA OPERABLE UNIT 3 MODOLESEX COUNTY, MASBACHUSETTS

							NONCARC	NOGENC	RISKANA	LYSIS RESU	113							
CONTAMINANTS of CONCERN	CONCEN	KG	EXPOSURE RECEPTOR	FACTOR R = TEEN	₩FD	HAZ		NENTS CI	#LD	HAZ		NENTS: TE	EN	HAZAR	DQUOTIENT	S: ADULT		TOXIC END- POINT
	MAX	DVA	INGENTION		(MG -	ACCH	DENTAL	DE		ACCIDENT		DEAM	AL.	ACCID	ENTAL	DERI	AL	
		[CONTROL	DAY	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	1 1
		I												-++.		_		
1,2-DICHLOROETHENE*	0 031	0 0087	1 11E-08	278E-08	1 00E - 02	57E-06	1 6E - 08	7 1E-08	2 0E - 08	3.4E-08	9.7E-07	8.6E-06	2.4E-08	3.0E - 07	6.5E~08	1.5E-08	4.3E~07	BLOOD
CHLOROBENZENE*	0 0 27	0 0064	1 11E~06	278E-08	2 ODE - 02	2 6E - 06	8 6E - 07	3 1E - 08	1.1E-00	1.5E-08	5.2E-07	3.7E-08	1.3E-08	1.3E-07	4.6E - 08	6.6E-07	2.3E 07	LIVER
1,2-DICHLOHOBENZENE		0 670	111E-06	8 98E - 07	9 00E - 02	376-05	1 BE-05	91E-08	4 52 -00	2.26-05	1.1E-05	1.1E-05	5.4E-08	2.0E-05	9.6E -07	2.0E - 08	9.8E-07	UVER
1,4-DICHLOROBENZENE	I .''	0 841	111E-00	5 55E - 07		0 0E + 00	0.0E + 00	0 0E + 00	0.0E+00	0 0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E + 00	0.0E+00	UVER
124TRICHLOHOBENZENE	0 78	0 8369	1116-00	0 DDE -07	1 306 - 03	116-03	9.0E-04	2.7E-04	2 2E -04	6.5E-04	5.5E-04	3.2E-04	2.7E-04	5.7E-05	4.8E05	5.7E-05	4.6E-05	SKIN
NAPHTHALENE		0 5994	111E-00	2 /DE -07	4 00E -03	54E-04	276-04	0.7E-05	34E-05	3.3E-04	1./E-04	0.2E~05	4.2E-05	2.9E~05	1.5E-05	1.4E-05	7.3E-08	BW
AHSENIC"			111E-00	0.000 + 00	3 000 -04	# 1E -02	4 26 -02	0.02+00	0 00 + 00	5 5E-02	2.65-02	0.00 + 00	0.02+00	4.9E-03	2.3E-03	0.0E+00	0.0E+00	SKIN
	218	54 07	4445.07	0.000 + 00	1000 +00	392-04	0.2E-00	0.02+00	0.00 + 00	2.42-04	3.0E - 03	0.00+00	0.0E+00	2.16-03	3.3E-08	0.02+00	0.05+00	CNR
	10.0	3.41	1115-00	6 85C - 00	3 00E - 04	105-01	2 35 - 02	476-03	5 8E - 04	115-01	1 45 - 02	5.7E - 02	7.15-04	1.05-02	1.25-02	105 03	1.25-04	CNS
	0,00	0.100	1115-00	4 995 - 08	3 000 - 04	105-01	2 3E -02	376-03	135-01	125-01	4.05-04	J./E-03	1.0E-02	1.05-02	3.45 05	7.00-03	275.04	CNS
MONOMETHICMERCONT	00.1	0.100		-206-00	9.00E - 04	1 00 - 00	002-04	3.72-00	1.02 - 00	1.22-05	4.00 - 04	4.42 - 03	1.02-03	1.02 -04	3.02 -03	1.82-04	2.72-04	CINS
DICHLOROMETHANE	0 000	0 008	1 11E-06	270E-08	0 00E - 02	2 4E - 07	1 6E - 07	3 0E - 07	2 3E - 07	1.5E-07	1.1E-07	3.7E-07	2.8E-07	1.3E-08	8.6E - 09	6.5E-08	4.9E-06	UVER
ACETONE	0 31	0 1205	1 11E-08	2 78E-06	1 00E - 01	57E-08	2 2E - 08	7.1E-08	2.8E-08	3.4E-08	1.3E-08	8.8E - 08	3.3E-08	3.0E~07	1.2E-07	1.5E-08	5.9E-07	L/K
BEHP	2	0 8508	1.11E-08	5 55E - 07	2 00E - 02	1 8E-04	5.9E-05	4.6E-05	1.5E-05	1.1E-04	3.6E-05	5.6E-05	1.6E-05	9.8E-08	3.2E-08	9.8E-08	3.2E-08	UVER
2-METHYLNAPHTHALEN	0 41	0 275	1 11E-08	2 78E-07		0 0E + 00	0.0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	BW
ACENAPTHYLENE	0 5	0 4861	111E-08	2 78E -07	4 00E - 03	2 3E - 04	2 2E-04	2.9E-05	2.6E-05	1.4E-04	1.3E-04	3 5E - 05	3.4E-05	1.2E-05	1.2E-05	6.1E-06	5.9E-06	NS
PHENANTHRENE	10 65	1 8345	1 11E-08	2 78E - 07	4 00E - 03	4 9E - 03	7.5E-04	6.1E-04	9.3E-05	3.0E-03	4.5E-04	7.4E-04	1.1E-04	2.6E-04	4.0E-05	1.3E-04	2.0E-05	BW
FWORANTHENE	9 25	2 0 1 4	1 11E-08	2 78E - 07	4 00E - 02	4 2E - 04	9 2E - 05	5.3E-05	1 1E-05	2.6E-04	5.6E-05	6.4E-05	1.4E-05	2.3E-05	4.9E - 06	1.1E-05	2.5E-08	L/K
PYRENE	1	1 869 1	1 11E-08	2 78E - 07	3 00E - 02	54E-04	1.0E-04	6 6E-05	1 3E - 05	3.3E-04	8.2E-05	8.2E-05	1.5E-05	2.9E-05	5.4E-06	1.5E~05	2.7E-08	KIDNEY
BENZO(A)ANTHRACENE	44	1 1055	111E~08	278E-07		0 0E + 00	0 0E + 00	0 0E+00	0 0E + 00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	NS
CHRYSENE	42	1 0823	111E-08	2 78E - 07		0 0E + 00	0.0E+00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E + 00	0.0E+00	NS
BENZO(B)FLUORANTHEN	🖡 3 55	1 0341	111E-06	2.78E-07		0 0E + 00	0 0E + 00	0 0E+00	0 0E + 00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	NB
BENZO (K) FLUORANTHEN	€ 38	0 8695	1 11E-08	2 78E-07		0 0E + 00	0 0E + 00	0 0E + 00	0 0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	0.0E + 00	NS
BENZO(A)PYRENE	3 75	1 0275	1 11E-08	278E-07		0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	NS
INDENO(123-CD)PYRENE	00	0 6261	1 11E-08	2 76E - 07		0 0E+00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E + 00	0.0E+00	0.0E + 00	0.0E + 00	NS
DIBENZ(AH)ANTHRACENE	1 0 0 8 3	0 0005	1 11E-06	2 78E-07		0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	NS			
BENZO(GHI)PERYLENE	0 05	0 5035	1116-06	2 78E-07	4 00E - 03	3 0E - 04	2 7E - 04	3 7E - 05	3.4E ~05	1.8E-04	1.8E-04	4.5E-05	4.1E-05	1.6E ~ 05	1.5E-05	8.0E - 08	7.3E – 06	NS
BARIUM	206	71.48	1 11E-08	0 00E + 00	5 00E - 02	7 8E - 03	2 8E - 03	0 0E + 00	0.0E+00	4.8E-03	1.8E-03	0.0E + 00	0.0E+00	4.1E-04	1.4E04	0.0E+00	0.0E+00	BLOOD
BERYLUUM	1.	0 87	1 11E-08	0 00E + 00	5 00E - 03	8 9E - 04	3 2E - 04	0 0E + 00	0.0E+00	4.2E-04	1.9E-04	0.0E+00	0.0E+00	3.7E05	1.7E-05	0.0E + 00	0.0E + 00	NS :
COPPER	184	40.18	1 11E-08	0 00E + 00		0 0E+00	0 0E + 00	0.0E+00	0 0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	NS
MANGANEBE	4100	667 87	1 11E-06	0 00E + 00	1 00E - 01	7 5E - 02	1 8E - 02	0 0E + 00	0 0E + 00	4.6E - 02	9.8E-03	0.0E+00	0.0E+00	4 0E 03	6.5E-04	0.0E+00	0.0E+00	CNS
NICKEL	1 10 1	7 47	1 11E-06	0 00E + 00	2 00E - 02	1 7E - 03	€ 6E - 04	0 0E + 00	0 0E + 00	1.1E-03	4.1E-04	0.0E+00	0.0E + 00	8.3E-05	3.7E - 05	0.0E + 00	0.0E+00	BW
VANADIUM	45 8	18 8	1 11E-00	0 00E + 00	7 00E - 03	1 2E - 02	5 2E - 03	0 0E + 00	0 0E + 00	7.3E-03	3.1E-03	0.0E+00	0.0E+00	6 4E - 04	2 8E - 04	0.0E + 00	0.0E + 00	NS
ZINC	330	128 44	1 11E-00	0 00E + 00	2 00E -01	3 DE ~03	1 2E-03	0 0E + 00	0.0E+00	1.8E-03	7.0E-04	0.0E+00	0.0E+00	1.8E-04	8 2E - 05	0.0E + 00	0.0E + 00	BLOOD
4.4-DDE	0 058	0.0100	3 33E -07	2 78E-07		0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0 0E + 00	0.0E+00	0.0E + 00	0.0E + 00	NB			
4,4DDD	0 0306	0 0306	3 33E - 07	2 78E-07		0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E + 00	0.0E + 00	NS			
4,4-DDT	L14	0 1724	3 33E - 07	2 78E-07	l	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	NS			

NYANZA SITE CONTAMINANTS	HAZARD INDEX	2 01 6 7E - 02	2 67E-03 2.1E-03 *	1.7E-01 4.1E-02 1.1E-0	2 2.0E-03 *	1.5E-02	3.8E -03	1.0E - 03	4.6E-04 *
OTHER SUDBURY RIVER CONTAMINANTS	HAZARD INDEX	1 1E-01 27E-0	2 85E-04 20E-04	6.5E-02 1.7E-02 1.0E-0	3 2.4E-04	5.7E-03	1.5E - 03	1.6E - 04	4.2E - 05
ALL CHEMICALS OF CONCERN		3	9.5E-03 2.3E-03	2.4E-01 5.6E-02 1.2E-0	2 2.8E-03	2.1E-02	5.1E-03	2 0E - 03	5.0E-04
		BREVIATIONS	NS: NOT SPECIFIED	L/K: UVER AND KIDNEY	BW: BODY WE	івнт с	NS: CENTRAL	NERVOUS S	YSTEM

(

FINAL

.

(

BEHP:BIS(2-ETHYL HEXYL)PHTHALATE

W92194F

TABLE 8-14B

PISK REBULTS FOR BEDIMENT EXPOSURE

REACH NO 2 NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

· · · · · · · · · · · · · · · · · · ·	· · ·	·			<u> </u>				
				CARONOG	ENIC RISK	NALYSIS	RESULTS		
									CANCER
CONTAMINANTS	CONCEN	TRATION	EXPOSURE	FACTOR		CANCER F	NSK8		SLOPE
of	MO/	Ka							FACTOR
CONCERN									(MG/KG/D)-1
			INGE \$ I ION	CONTACT	ACCIDEN		UEHM		
	MAA	~~		CONTACT	INGEBIN	, m	CONT		WEIGHT
					MAX	AVG	MAX	AVG	ENDENCE
		·· · ····		•	· · · · · · · · · · · · · · · · · · ·			L	
1.2-DICHLOROETHENE*	0 031	0 0067	3 84E - 07	7 58E - 07	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
CHLOROBENZENE*	0 027	0 0094	3 84E - 07	7 50E -07	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
1,2-DICHLOROBENZENE	1.8	0 878	3 84E - 07	1 51E -07	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
1,4-DICHLOROBENZENE	11	0 841	3 84E - 07	1 51E - 07	0 0E - 00	5 8E - 09	4 OE ~09	2 3E - 09	2.4E - 02 B2
124TRICHLOROBENZENE	0 70	0 0300	3 84E - 07	1 51E - 07	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
NAPHTHALENE*	1 10	0 5994	3 84E - 07	7 50E - 08	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
ARSENIC*	14 0	6 85	3 84E - 07	0 00E + 00	9 8E - 09	4 8E -08	0 0E + 00	0 0E + 00	1 8E+00 A
CHROMIUM*	210	34	3 84E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
LEAD"	205	56 07	1 82E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
MERCURY*	30 0	3 81	3 84E - 07	1 51E - 08	0 0€ + 00	0 0E + 00	0 0E + 00	0 0E + 00	
MONOMETHYLMERCURY	0 312	0 100	3 64E - 07	1 16E - 08	0 OE + 00	0 OE + 00	0 0E + 00	0 0E + 00	
DICHLOROMETHANE	0 006	0 000	3 84E - 07	7 56E - 07	2 2E - 11	1 OE - 11	4 5E - 11	3 4E - 11	7 5E - 03 B2
ACETONE	0 31	0 1205	3 84E - 07	7 50E - 07	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
BEHP	2	0 6506	3 84E - 07	1 51E - 07	1 OE - 08	3 3E - 09	4 2E -08	14E-09	1 4E - 02 BZ
2-METHYLNAPHTHALEN	0 41	0 275	3 44E - 07	7 50E - 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
ACENAPTHYLENE	0.5	0 4861	3 84E - 07	7 56E - 08	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
PHENANTHRENE	10 85	1 6345	3 84E - 07	7 56E - 08	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
FLUORANTHENE	0 25	2 014	3 84E - 07	7 50E - DB	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
PYRENE		1 0091	3 84E - 07	7 56E - 08	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
BENZO(A)ANTHRACENE	44	1 1055	3 84E - 07	7 50E - 08	8 3E - 08	2 3E - 08	1 9E - 08	4 8E ~07	5 8E + 00 82
CHAYSENE	4 2	1 0923	3 84E - 07	7 58E - 08	8 9E - 08	2 3E - 08	1 8E - 08	4 8E -07	5 8E + 00 B2
BENZO(B)FLUORANTHEN	3 55	1 0341	3 84E - 07	7 58E - 08	7 5E - 08	2 2E - 08	1 8E - 06	4 5E -07	5 8E + 00 B2
BENZO(K)FLUORANTHEN	5 3 8	0 9995	3 84E - 07	7 50E - 08	8 OE - 08	21E-08	1 7E - 06	4 3E ~07	5 6E + 00 B2
BENZO(A)PYRENE	3 75	1 0275	3 84E - 07	7 50E - 08	7 BE - 08	2 2E - 08	1 8E - 08	4.5E - 07	5 8E + 00 B2
INDENO(123-CD)PYRENE	09	0 8281	3 84E -07	7 50E - 08	1 BE - 06	1 JE - 00	3 9E - 07	2.7E ~ 07	5.8E+00 B2
DIBENZ (AH) ANTHRACENE	0 063	0 0005	3 84E - 07	7 50E - 08	2 OE - 07	1 BE - 07	4 1E-08	3 8E ~ 08	5 8E + 00 B2
BENZO(GHI)PERYLENE	0 65	0 5035	3 84E - 07	7 50E - 08	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
BARIUM	208	71.48	3 84E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
BERYLUUM	10	0 87	3 84E - 07	● 00€ + 00	3 OE - 08	1 4E - 06	0 0E + 00	0 0E + 00	4 3E +00 A
COPPER	184	40.18	3 84E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
MANGANESE	4100	867 87	3 64E - 07	00E+00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
NICKEL	10 1	7 47	3 84E - 07	0 00€ + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
VANADIUM	45 6	18.8	3 84E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
ZINC	330	120 44	3 84E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
4,4-DDE	0 056	0 0100	1 QOE -07	7 50E - 06	2 2E -09	2 1E - 00	1 5E - 08	4 3E - 10	3 4E - 01 B2
4,4-000	0 0300	0 0300	1 QOE - 07	7 50E - 00	8 OE - 10	8 OE - 10	5 5E - 10	5 5E ~ 10	2 4E - 01 82
4,4-DDT	14	0 1724	1 00E -07	7 59E - 08	5 2E - 00	8 4E - 08	3 8E - 06	4 4E09	3.4E-01 B2

NYANZA SITE CONTAMINANTS	CANCER RISK	9 8E - 00	4 0E -08	4 DE -00	2 JE - 09
OTHER SUDBURY RIVER CONTAMINANTS	CANCER RISK	4 7E -06	1 4E - 05	9 1E-08	2 6E - 06
ALL CHEMICALS OF CONCERN	CANCER RISK	5 OE - 05	1 9E - 05	8 1E-00	2 8E - 08

BEHP:BIS(2-ETHYL HEXYL)PHTHALATE

TABLE 6 - 15A RISK ASSESSMENT RESULTS FOR SURFACE WATER EXPOSURE SCENARIOS REACH NO 2 NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

	[NONCAR	NOGENIC	PISK ANAL	YSIS RESUL	.19				•••••			
CONTAMINANTS of CONCERN	CONCEN MQ/	NO NO	EXPOSURE RECEPTOR	FACTOR	NFD (MG- KG-	HAZAI		ENTS: CHIL	.D	HAZAF	10 QUOTIEN	ITS: TEEN		HAZARC	QUOTIENT	's: adult		TOXICITY END- POINT
	MAX	AVG	NGESTICN		DAY)	ACODEN	TAL DN	DEPM/	NL CT	ACCIDENT	TAL N	DERMA	L	ACCIDENTA	L	DERMAL		
						MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	
1,2-DICHLOROETHENE*	0001	1	1 11E-03	2 06E - 03	1 00E-02	9 1E-05	9 1E-05	9 5E - 05	9 5E - 05	1.1E-04	1.1E-04	2.1E-04	2.1E04	2.0E-05	2.0E-05	1.1E-04	1.1E-04	BLOOD
ARSENIC*	0 0003	0 003	1 11E-03	1 30E - 04	3 00E - 04	9 1E - 03	9 1E-03	64E-04	64E-04	1.1E-02	1.1E-02	1.4E-03	1.4E-03	2.0E-03	2.0E-03	7.6E-04	7.6E-04	SKIN
CADMUM*	0 007	0 0028	1 118-03	1 38E - 04	6 00E - 04	1 3E - 02	5 2E - 03	9 2E - 04	3 6E - 04	1.8E-02	6.3E-03	2.0E-03	7.8E-04	2.6E-03	1.1E~03	1.1E-03	4.3E-04	KIDNEY
CHROMIUM*	0 007	0 0063	1 11E-03	1 36E - 04	1 00E+00	8 4E - 08	4 8E 06	4 5E - 07	34E-07	7.8E-08	5.9E - 06	9.7E-07	7.3E-07	1.4E-06	1.0E-06	5.3E-07	4.0E-07	LIVER
LEAD*	0 087	0 0114	1 11E-03	1 38E-04		0 0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	CNS
BARIUM	0 132	0 0336	1 11E-03	1 38E-04	5 00E - 02	2 4E - 03	6 2E - 04	1.7E-04	4 3E - 05	2.9E-03	7.5E-04	3.6E-04	9.3E-05	5.2E-04	1.3E-04	2.0E04	5.1E-05	BLOOD
BERYLUUM	0 008	0 0028	1 11E-03	1 38E-04	5 00E - 03	1 4E - 03	4.8E-04	9 6E - 06	3 3E - 05	1.7E-03	5.8E-04	2.1E-04	7.2E-05	2.9E-04	1.0E-04	1.1E-04	4.0E-05	NS
COPPER	0 028	0 0135	1 11E-03	1 36E-04		0 0E+00	0 0E+00	0 0E+00	0 0E+00	0 0E+00	0 0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	-10E+00	0.0E+00	NS
MANGANESE	9 84	1 6337	1 11E-03	1 38E-04	1 00E - 01	9 0E - 02	14E-02	6 3E - 03	9 8E - 04	1.1E-01	1.7E-02	1.4E-02	2.1E-03	1.9E-02	3.0E-03	/ 5E-03	1.2E-03	CNS
NICKEL	0 077	0 022	1 11E-03	1 36E-04	2 00E - 02	3 6E - 03	1 0E - 03	2 5E - 04	7 0E - 05	4 3E-03	1.2E-03	5.3E-04	1.5E-04	7.6E-04	2.2E-04	2.9E-04	8.4E05	BODY WT
SILVER	0 069	0 0090	1 11E-03	1 30E - 04	3 00E - 03	2 1E - 02	2 9E - 03	1 5E - 03	20E-04	2 6E - 02	3 5E 03	3.2E - 03	4.4E-04	4.5E-03	6.2E-04	1.7E-03	2.4E-04	ARGYRIA
VANADIUM	0 0 16	0 0144	1 11E-03	1 36E - 04	7 00E - 03	2 1E - 03	1 9E - 03	1.5E-04	1.3E-04	2.6E-03	2.3E-03	3.2E-04	2.8E-04	4.5E04	4.0E-04	1.7E-04	1.6E-04	NS
ZINC	0 125	0 0246	1 11E-03	1 36E - 04	2 00E -01	87E-04	1.1E-04	4.0E-05	7.9E-08	6.9E ~ 04	1.4E-04	8.6E-05	1.7E-05	1.2E-04	2.4E-05	4.7E-05	9.4E-08	BLOOD
NYANZA SITE CONTAMN	ANTS"			HAZARDIN	SEX	2 2E - 02	1 4E - 02	1.7E-03	1 1E-03	2.7E-02	1.8E-02	3.6E - 03	2.4E-03	4.8E-03	3.1E-03	2.0E-03	1.3E-03	
OTHER SUDBURY RIVER	CONTAM	NANTS		HAZARD IN	JEX	1 2E -01	2 1E-02	6 4E -03	1.5E-03	1.5E-01	2.6E-02	1.8E-02	3.2E-03	2.6E - 02	4.5E-03	1.0E-02	1.7E-03	
ALL CHEMICALS OF CON	CERN			HAZARD IN	X X	1 4E-01	3.5E - 02	1 0E - 02	2.6E-03	1.7E-01	4.3E-02	2.2E-02	5.5E-03	3.1E-02	7.6E-03	1.2E-02	3.1E-03	J
						BBREVIATIC	NS:	NS NOT S	PEOFIED	L/K: UV		NEY						

BEPH: BIS(2ETHYL HEXYL)PHTHALATE

NS NOT SPECIFIED L/K: UVER CNS. CENTRAL NERVOUS SYSTEM L/K: UVER AND KIDNEY

BLE 6--15B

RISK ASSESSMENT RESULTS FOR SURFACE WATER EXPOSURE SCENARIOS REACH NO. 2 NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

					CARCINO	GENIC RIS	K RESULT	S	
CONTAMINANTS of CONCERN	CONCEN MG/	ITRATION KG	EXPOSURE	FACTOR		CANCEF	RISKS		CANCER SLOPE FACTOR
	MAX	AVG	INGESTION	DERMAL	ACCIDEN	TAL	DERM	AL	(MG/KG/D)-
				CONTACT	INGESTIC	N	CONT	ACT	
					MAX	AVG	MAX	AVG	WEIGHT OF
	{								EVIDENCE
1,2-DICHLOROETHENE*	0.001	0.001	3.02E-04	6.28E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ARSENIC*	0.003	0.003	3.02E-04	4.21E-05	1.6E-06	1.6E-06	2.3E-07	2.3E-07	1.80E+00
CADMIUM*	0.007	0.0028	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
CHROMIUM*	0.007	0.0053	3.02E-04	4.21E05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	1
LEAD*	0.067	0.0114	3.02E-04	4.21E05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
BARIUM	0.132	0.0338	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
BERYLLIUM	0.008	0.0026	3.02E-04	4.21E-05	9.7E-06	3.4E-06	1.4E-06	4.7E-07	4.30E+00
COPPER	0.028	0.0135	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MANGANESE	9.84	1.5337	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
NICKEL	0.077	0.022	3.02E-04	4.21E~05	0.0E+00	0.0E+00	0. 0E +00	0.0E+00	
SILVER	0.069	0.0096	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
VANADIUM	0.016	0.0144	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
	0.125	0.0248	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
NYANZA SITE CONTAMINA	ANTS*		CANCER R	SK	1.6E-06	1.6E-06	2.3E-07	2.3E-07]
OTHER SUDBURY RIVER (ONTAMI	NANTS		ISK	9.7E-06	3.4E-06	1.4E-06	4.7E-07	
ALL CHEMICALS OF CON	CERN		CANCER R	ISK	1.1E-05	5.0E-06	1.6E-06	7.0E~07	

BEPH: BIS(2ETHYL HEXYL)PHTHALATE

6-77

FINAL

W92194F

::

and sediment exposures, the total hazard index does not exceed unity. These results indicate that adverse noncarcinogenic health effects would not be anticipated for a receptor contacting Reach 2 surface waters and sediments under the conditions of the recreational exposure scenarios defined in Section 6.4.

The cancer risks estimated for sediment exposure scenarios exceed 6×10^{-5} and 2×10^{-5} (risks summed for the accidental-ingestion and dermal-contact exposure routes) when the reasonable maximum and average cases are evaluated, respectively. Cancer risks estimated for the accidental-ingestion exposure route exceed those estimated for the dermal exposure route. The principal COCs contributing to the estimated cancer risk are the carcinogenic PAHs. Two Nyanza site-specific contaminants, arsenic and 1,4-dichlorobenzene, contribute to the excess lifetime cancer risk. However, the risk estimated for Nyanza Site contaminants does not exceed 1 x 10^{-5} even when maximum contaminant concentrations are evaluated. As a point of reference, excess lifetime cancer risks estimated for COC concentrations in background sediment samples were 1.8 x 105 and 8.2 x 10^{-6} when the reasonable maximum- and average-case scenarios are evaluated, respectively.

The cancer risks estimated for the surface water exposure scenarios slightly exceed 1 x 10^{-5} and 5 x 10^{-6} (risks summed for accidentalingestion and dermal-contact exposure routes) when the reasonable maximum- and average-cases are evaluated, respectively. Risks estimated for the accidental-ingestion exposure route predominate. Beryllium and arsenic, a Nyanza Site contaminant, are the only COCs contributing to the estimated risk. Both metals have been designated by the EPA as Class A human carcinogens. The risk levels estimated for the background surface waters were 1.7 x 10^{-6} and 1.5 x 10^{-6} when the reasonable maximum and average cases were evaluated, respectively.

Hazard indices calculated for COC concentrations detected in Mill Pond fish tissue samples (Table 6-16) exceed unity when the reasonable maximum case is evaluated. The hazard index also exceeds unity when average contaminant concentrations are evaluated and the subsistence fisherman is considered the receptor of concern. Mercury and methyl mercury are the principal COCs contributing to the risk; the hazard quotient calculated for mercury exceeds unity in the aforementioned cases. The maximum concentration of mercury detected in Mill Pond fish tissue samples exceeds the FDA Action Level for mercury in fish:

 C_{max} (mercury) = 2 mg/kg C_{max} (methyl mercury) = 0.7 mg/kg

W92194F

TABLE (* FRISK ASRESSMENT RESULTS FOR FISH JITION EXPOSURE SCENARIOS MILL POND NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

	T					NONCARCIN	OCIENIC RISK AN	ALYSIS RESUL	19				CARCINOG	ENIC RISK A	NALYSIS RI	ESULTS	
CONTAMINANTS OF CONCERN	CONCEN MG/M3	TRATION	EXPOSURE	FACTOR	PFD (MG- KG-		HAZARD QU	OTIENTS		TOXICITY END- POINT	EXPOSURE	FACTOR	СА	NCER RISKS	3		CANCER SLOPE FACTOR
	MAX	AVG	SPORTS	SUBBIS- -TENCE	DAY)	SPORTSF	BHERMEN	9U888	STENCE		SPORTS FISHER-	SUBSIG- TENCE	SPORT F	SHERMEN	SUB9IST FISHERI	ENCE MEN	(MG/KG/D) - 1 /
		1	MEN	FILMER-		MAXIMUM	AVERAGE	MAXIMUM	AVERAGE	-	MEN	HSHEH-	MAXIMUM	AVERAGE	MAXIMUM	AVERAGE	EVIDENCE
MERCURY*	1 981	0 315	1 85E - 04 1 85E - 04	1 36E - 03 1 36E - 03	3 00E - 04 3 00E - 04	12E+00 45E-01	1 0E - 01 1 3E - 01	9 0E + 00 3 3E + 00	14E+00 97E-01	CNS CNS	7.93E-05 7.93E-05	5.81E-04 5.81E-04	0.0E+00 0.0E+00	0.0E+00 0.0E+00	0.0E+00 0.0E+00	0.0E+00 0.0E+00	
4,4-DDE	0 03	0 01088	1 85E - 04	1 38E -03		0 0E+00	0 0E + 00	0 0E + 00	0 0E + 00	LIVER	7 93E - 05	5.81E-04	8.1E-07	2.6E-07	5.9E-06	2.1E-06	3.40E-01 8
AROCLOR - 1248 COPPER	0 404	0 0725	1 85E - 04	1 36E - 03	1 00E_01	0 0E+00 0 0E+00	0 0E + 00 0 0E + 00 2 1E - 04	0 0E +00 0 0E +00	0 UE + 00 0.0E + 00 1.7E = 03	NS NS CNR	7.93E-05 7.93E-05 7.91E-05	5.81E-04 5.81E-04	3.1⊵-04 0.0E+00	4.4E-05 0.0E+00	2.2E-03 0.0E+00	3.2E -04 0.0E +00	7.70E+00 Ba
SILVER	0 025	0 025	1 85E-04	1 36E - 03	3 00E -03 7 00E -03	15E-03	1 5E - 03 9 2E - 04	1.1E-02 8.9E-03	1.1E-02 6.6E-03	ARGYRIA	7.93E-05	5.81E-04	0.0E+00 0.0E+00	0.0E+00 0.0E+00	0.0E+00 0.0E+00	0.0E+00 0.0E+00	
	3 012	3 228	1 85 -04	1 36E - 03	2 00E 01	3 E-03	3 0E - 03	2 /E 02	2.2E-02	BLOOD	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
NYANZA BITE CONTAMIN	IANTS*	· ·	HAZAPO IN	DEX		12 + 00	1 9E -01	9 0E + 00	1 Æ+00		CANCER R	SKS:	0.0E+00	0 0E + 00	0.0E+00	0.0E+00	ן
OTHER SUDBURY RIVER	CONTAMIN	etru	HAZARD IN	DEX		7 Œ - 03	67E-03	5 1E - 02	4 2E - 02		CANCERRI	SK8:	3.1E-04	4.5E-05	2.2E-03	3.3E-04	
ALL CHEMICALS OF CON	CERN		HAZARD IN	DEX		122+00	2 0E -01	9 0E + 00	1.5E+00	- <u>-</u>	CANCER RI	SK9:	3.1E-04	4.5E-05	2.2E-03	3.3E -04	

BEPH: BIS(2-ETHYL HEXYL)PHTHALATE

TOROTTY ENDPOINT ABBREVIATIONS

NS NOT SPECIFIED BW BODY WEIGHT CNS: CENTRAL NERVOUS SYSTEM

L/K LIVER AND KIDNEY

.

FINAL

These results indicate that adverse noncarcinogenic health effects are anticipated for the sports fisherman (reasonable maximum-case) and subsistence fishermen routinely consuming fish taken from Mill Pond.

Cancer risks estimated for Mill Pond fish tissue samples exceed 1×10^4 when the reasonable worst-case is evaluated. Estimated excess lifetime cancer risks also exceed 1×10^4 when the average case is evaluated and the subsistence fisherman is the receptor of concern. Aroclor-1248 and 4,4-DDE are the only COCs contributing to the estimated cancer risk. These risks are approximately 5 times those estimated for the Sudbury Reservoir fish tissue samples.

6.5.3.4 Risk Assessment Results for the Eastern Wetlands

The Eastern Wetlands receive surface water runoff directly from the Nyanza Site and constitute a headwaters area of a small tributary (Trolley Brook) of the Sudbury River. In contrast to most of the River Reaches, several organic and inorganic Nyanza Site contaminants were detected in the both sediment and surface water samples collected from the area:

- o Trichloroethene
- o Chlorobenzene
- o 1,2-Dichlorobenzene
- o 1,2,4-Trichlorobenzene
- o Arsenic
- o Lead
- o 1,2-Dichloroethene
- o Nitrobenzene
- o 1,4-Dichlorobenzene
- o Naphthalene
- o Chromium
- o Mercury
- o Monomethyl mercury

Trichloroethene and mercury were detected at least once in surface water samples at concentrations in excess of Federal SDWA MCLs. The average mercury concentration detected in the Eastern Wetland sediments (C_{ex} =35.9 mg/kg) is more than 100 times the concentration detected in the background sediment samples (C_{ex} =0.27 mg/kg)

Risk assessment results for the Eastern Wetlands are presented on Tables 6-17, 6-18, and 6-19. Risk assessment results for the surficial samples (Table 6-17) are presented separately from those

W92194F

TABLE 0-17A RISK ASSESSMENT RESULTS FOR SEDIMENT EXPOSURE SCENARIOS EASTERN WETLANDS NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

							NONCAR		RISK ANA	LYSIS RESI	JLTS							
CONTAMINANTS of CONCERN	CONCEN	(TRATIO) KG	EXPOSURE RECEPTOR	FACTOR	A ₩FD	HAZ	ARD QUO	TIENTS C	HILD	HAZ	ARD QUOT	IENTS: TE	EN	HAZAR	ID QUOTIENT:	S: ADULT		TOXIC END- POINT
	MAX	AVG	INGESTION	DERMAL	(MG -	ACCI	DENTAL	DE	AMAL	ACCIDEN	TAL	DERM	AL	ACCIE	DENTAL	DEA	WAL	
				CONTACT	KG	INGE	TION	co	NTACT	INGESTIC	N	CONT	ACT	INGE	STION	CON	ACT	-
······································	ļ	1.	- 1	I	DAY	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	+
TRICHLOROETHENE	0 17	0 0914	1 11E-00	2 78E - 08		0 0E + 00	0 0E + 00	0 0E + 00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	NS			
1.2 - DICHLOROETHENE*	013	0 0723	1 11E-08	2 78E -08	1 00E - 02	2 4E - 05	1 3E - 05	3 OE - 05	1 6E - 05	1 4E - 05	8 OE - 08	3.6E - 05	2 0E - 05	1.3E-06	7.1E-07	6.4E - 06	3.5E-06	BLOOD
CHLOROBENZENE*	1.6	0 8084	1 11E-00	2 78E - 09	2 00E - 02	1 5E - 04	7 4E - 05	1 8E - 04	9 2E - 05	8 8E - 05	4 5E - 05	2 2E - 04	1.1E-04	7.8E-06	3.9E - 06	3.9E - 05	2.0E - 05	UVER
NITROBENZENE*	0 65	0 65	1 11E-00	5 55E - 07	5 00E - 04	2 4E - 03	2 4E - 03	5 9E - 04	5 9E ~ 04	1 4E - 03	1.4E-03	7 2E - 04	7.2E-04	1.3E-04	1.3E-04	1.3E-04	1.3E – 04	L/K
1,2-DICHLOROBENZENE	77	4 013	1 1 1 E - OB	5 55E - 07	8 00E - 02	1 5E - 04	8 1E-05	3 7E - 05	2 OE - 05	8 9E - 05	5 0E - 05	4 4E - 05	2.5E~05	7.8E-08	4.4E 06	7.8E - 06	4.4E-08	UVER
1.4 - DICHLOROBENZENE	1 18	1 2125	1 11E-00	5 55E - 07		0 0E + 00	0 0E + 00	0 0E + 00	0.0E + 00	0.0E+00	0.0E + 00	0.0E + 00	0.0E + 00	0.0E + 00	LIVER			
124TRICHLOROBENZENE	31	1 8625	1 11E-08	5 55E - 07	1 30E - 03	4 4E - 03	2 8E - 03	1 1E - 03	8 9E - 04	2 6E - 03	1.7E - 03	1.3E – 03	8.4E-04	2.3E-04	1.5E-04	2.3E - 04	1.5E-04	SKIN
NAPHTHALENE*	23	1 5825	1 11E-00	2 78E - 07	4 00E - 03	1 1E-03	7 1E-04	1 3E - 04	8 9E - 05	8 4E 04	4 3E - 04	1.6E - 04	1.1E-04	5.6E - 05	3.8E - 05	2.8E-05	1.9E - 05	BW
ARSENIC*	127	0 59	1 11E-08	0 00E + 00	3 00E - 04	7 7E - 02	4 OE - 02	0 0E + 00	0 0E + 00	4 7E - 02	2.4E - 02	0.0E + 00	0.0E + 00	4.1E-03	2.1E-03	0.0E + 00	0 0E + 00	SKIN
CHROMIUM*	462	123 88	1 11E-08	0 00E + 00	1 00E + 00	● 4E - 04	2 3E - 04	0 0E + 00	0 0E + 00	51E~04	1.4E-04	0.0E + 00	0.0E+00	4.5E - 05	1.2E-05	0.0E + 00	0 0E + 00	UVER
LEAD	142	72 58	5 55E - 07	0 00E + 00		0 DE + 00	0 0E + 00	0.0E + 00	0.0E + 00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	0.0E + 00	CNS			
MERCURY*	152	35 878	1 11E - 00	5 55E - 00	3 00E - 04	● 3E - 01	2 2E -01	2 3E - 02	5 5E - 03	5 6E 01	1.3E-01	2.8E-02	6.6E-03	5.0E-02	1.2E-02	5.0E – 03	1.2E-03	CNS
MONOMETHYLMERCURY	0 220	0 0676	1 11E-00	4 20E -00	3 00E - 04	1 4E -03	4 1E-04	2 7E - 03	7 9E-04	8.5E-04	2.5E~04	3.3E – 03	9.6E-04	7.5E - 05	2.2E – 05	5.6E~04	1 7E 04	CNS
ACETONE		0.	111E-08	2 78E - 08	1 00E - 01	2 0E - 05	1 8E - 05	3 7E - 05	2 1E-05	1 8E - 05	1.0E-05	4.4E-05	2.5E-05	1.6E-08	8 8E 07	7 8E ~ 06	4.4E - 06	L/K
3/4METHYLPHENOL	0 28	0 26	1118-00	5 55E - 07	5 00E - 02	9 5E - 08	9 5E - 00	2 4E - 08	2 4E - 08	5 BE ~ 08	5.6E - 08	2.8E-08	2.9E-08	5.1E-07	5.1E-07	5.1E-07	5.1E-07	BW,CNS
PHENANTHRENE	0 62	0 62	1 11E-00	2 78E - 07	4 00E - 03	2 BE - 04	2 8E - 04	3 5E - 05	3 5E - 05	17E-04	1.7E-04	4.3E - 05	4.3E ~ 05	1.5E - 05	1.5E-05	7.8E - 08	7.8E - 08	BW
FLUORANTHENE	10	1 2125	1 11E-00	2 78E - 07	4 00E - 02	7 3E - 05	6 5E - 05	● 1E-06	0 9E 00	4.4E-05	3.4E-05	1.1E – 05	8.4E-08	3.9E - 08	3.0E - 06	2.0E - 06	1.5E-06	L/K
PYRENE	17	1 2025	1116-00	2 78E - 07	3 00E - 02	1 0E - 04	7 7E-05	1 3E 05	9 8E - 08	6 3E - 05	4.7E-05	1.6E~05	1.2E-05	5.5E-08	4.1E-06	2 8E - 06	2.1E-08	KIDNEY
CHRYSENE	12	1 0125	111E-00	2 78E -07		0 0E + 00	0 0E + 00	0.0E+00	0.0E + 00	0.0E+00	0.0E + 00	0.0E+00	0.0E + 00	0.0E + 00	NS			
BENZO(B)FLUORANTHEN	F 25	2 77	1 11E-08	2 78E - 07		0 0E + 00	0 0E + 00	0 0E+00	0 0E + 00	0.0E+00	0.0E + 00	0.0E+00	0.0E + 00	0.0E + 00	NS			
BENZO(A)PYRENE	0.80	0 8425	1 11E-00	2 70E -07		0 0E + 00	0 0E + 00	0 0E+00	0.0E+00	0 0E+00	0.0E+00	0 0E + 00	0.0E + 00	0.0E + 00	NS			
BARIUM	577	33 615	1 11E - 00	0 00E + 00	5 00E - 02	2 IE - 03	1 2E - 03	0 0E + 00	0 0E + 00	1.3E 03	7.5E-04	0.0E+00	0.0E+00	1.1E-04	0.6E-05	0.0E + 00	0.0E+00	BLOOD
BERYLUUM	4	2 21	1 11E-00	0 00E + 00	5 00E - 03	1 5E - 03	0 1E - 04	0 0E + 00	0 0E + 00	8.9E-04	4.9E-04	0.0E + 00	0.0E+00	7.8E-05	4.3E-05	0.0E+00	0.0E + 00	NS
СОРРЕЯ	120	63 73	1 11E - 00	0 00E + 00		0 0E + 00	0.05+00	0.05+00	0.05+00	00E+00	0.06+00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	NS
MANGANESE	1320	200 11	1 11E - 00	0 00E + 00	1 00E - 01	242-02	D.3E-U3	0.0E+00	0.05+00	1.5E-02	3.2E~03	0.00+00	0.0E+00	1.36-03	2.8E-04	0.02+00	0.0E+00	CNS
NICKEL	40 4	10 143	1116-00	0 00E + 00	2 00E - 02	372-03	W 3E -04	0.05+00	0.000	2.2E-03	3.8E~U4	0.00 +00	0.0E+00	202~04	506-05	0.00 + 00	0.05+00	BW SELENCO
SELENIUM		143	1116-00	0 00E + 00	3006-03	242-03	326-04	0.0E+00	0.000 + 00	1.46-03	3.26-04	0.00 + 00	0.00 + 00	1.3E-04	2.86 -03	0.00 + 00	0.02 +00	SELENUA
VANADIUM	3/4	1/ 3/	1116-00	0.000 + 00	7002-03	# 8E ~ 03	402-03	0.02+00	0.05+00	5 82 -03	200-03	0.00 + 00	0.02+00	3.2E - 04	2.56~04	0.02 + 00	0.00 + 00	INS INS
			1118-00	0.005+00	2006-01	1 52 -03	0.05.00	0.000 + 00	0.00000	0 05 400	3.0E - 04	0.02+00	0.02+00	0.0E + 00	3.2E-03	0.02+00	0.02+00	Ng
NYANZA SITE CONTAMINA OTHER SUDBURY RIVER (ANT8 CONTAMIN			HAZARD IN	DEX	1 0E + 00 4 5E - 02	2 0E - 01 1 4E - 02	2 5E - 02 9 6E - 05	7 OE - 03 7.5E - 05	• • 28E-01	1.6E - 01 6.7E - 03	3.1E-02 1.2E-04	8.5E-03 * 9.1E-05 *	5.4E-02 2.4E-03	1.4E - 02 7.7E - 04	5 4E 03 2.1E 05	1.5E - 03 1.6E - 05	· .
ALL CHEMICALS OF CON	DERN			HAZARD IN	DEX	1.1E+00	2 8E - 01	2 5E - 02	7 0E - 03	84E-01	1.7E-01	3.1E-02	6.8E - 03	5.7E-02	1.5E-02	5.4E-03	1.5E-03	
······				TOXCITY F		BREVIATIO	DNS:	NS NOT	PECIFIED			NEY	BW: BODY V	VEIGHT (CNS: CENTRA	L NERVOUS	SYSTEM	97

BEPH BIS 2 - ETHYL HEXYLIPHTHALATE

.

6-81

TABLE 6 - 17B RISK RESULTS FOR SEDIMENT EXPOSURE EASTE RN WETLANDS NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

L L				CARCINOGE	ENIC RISK A	NALYSIS F	RESULTS		
CONTAMINANTS of	CONCEN	TRATION	EXPOSURE	FACTOR		CANCERF	ISKS		CANCER SLOPE FACTOR
CONCERN	MAX	AVG	INGESTION	DEFINAL CONTACT	ACCIDEN	TAL DN	DERM CONT/	AL VCT	(MG/KG/D) 1 / WEIGHT
					MAX	AVG	MAX	AVG	
	0.17	0.0914	3 MAE-07	7 54E-07	6.8E-10	37E-10	145-09	7.6E-10	1 10F-02 B2
12-DICHLOBOETHENE*	0.13	0 0723	3 64E-07	7 56E-07	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
CHI OBOBENZENE*	1.6	0 8064	3 64E-07	7 56E-07	0.0E+00	0.0E+00	0 0E+00	0.0E+00	
NITROBENZENE*	0 65	0 65	3 64E - 07	1 51E-07	0.0F+00	0 0F + 00	0 0F+00	0.0E+00	
12-DICHLOBOBENZENE	72	4 013	3 64E-07	1 51E-07	0 0F + 00	0.0E+00	0.0F+00	0.0E+00	
1.4-DICHLOROBENZENE	16	1 2125	3 64E - 07	1.51E-07	1.4E-06	1.1E-08	5.8E-09	4.4E-09	2.40E-02 B2
124TRICHLOBOBENZENE	31	1 9625	3 64E-07	1 51E-07	0 0F + 00	0 0F+00	0 0F+00	0.0F+00	
NAPHTHALENE*	23	1 5625	3 64E-07	7 56E-08	0 0E + 00	0 0E + 00	0 0E+00	0 0E+00	
ARSENIC*	127	6.59	3 64E - 07	0.00F+00	8 3E-06	4 3E-06	0 0F + 00	0.0E+00	1 80F + 00 A
CHROMIUM .	462	123 68	3 64E - 07	0 00E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
LEAD*	142	72 58	1 82E - 07	0 00E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	
MERCURY*	152	35 878	3 64E-07	1.51E-08	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	
MONOMETHYLMERCURY	0 229	0 0676	3 64E - 07	1.16E - 06	0 0E + 00	0 0E + 00	0 0E+00	0.0E+00	
ACETONE	16	0 9	3 64E-07	7.56E-07	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
3/4METHYLPHENOL	0 26	0 26	3 64E-07	1 51E-07	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
PHENANTHRENE	0 62	0 62	3 64E-07	7 56E-08	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
FLUORANTHENE	18	1 2125	3 64E - 07	7 56E-08	0 0E + 00	0 0E+00	0 0E+00	0.0E+00	
PYRENE	17	1 2625	3 64E-07	7 56E-08	0 0E + 00	0.0E+00	0 0E+00	0.0E+00	
CHRYSENE	12	1 0125	3 64E-07	7.56E-08	2.5E-06	2.1E-06	5.3E-07	4.4E-07	5.80E+00 B2
BENZO(B)FLUORANTHENE	2 5	2 77	3 64E - 07	7.56E-08	5 3E-06	5 6E-06	1 1E-08	1.2E-06	5.80E+00 B2
BENZO(A)PYRENE	0 86	0 8425	3 64E-07	7 56E-06	1 8E-08	1.8E-06	3 8E-07	37E~07	5.80E+00 B2
BARIUM	57 7	33 615	3 64E-07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E+00	
BERYLUUM	4	2 2 1	3 64E-07	0 00E + 00	6 3E-06	3 5E-06	0 0E+00	0.0E+00	4.30E+00 A
COPPER	120	63 73	3 64E-07	0 00E + 00	0 0E + 00	0.0E+00	0 0E + 00	0 0E+00	
MANGANESE	1320	200 11	3 64E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E+00	0.0E+00	
NICKEL	40.4	10 143	3 64E-07	0 00E + 00	0 0E + 00	0 0E+00	0 0E + 00	0.0E+00	
SELENIUM	6 5	1 43	3 64E-07	0 00E+00	0 0E + 00	0.0E+00	0 0E + 00	0.0E+00	
VANADIUM	37.4	17 57	3 64E-07	0 00E+00	0 0E + 00	0 0E+00	0 0E+00	0.0E+00	
ZINC	164	65 383	3 64E-07	0 00E + 00	0 0E + 00	0 0E+00	0.0E+00	0.0E+00	
44-DDD	0 177	0 0755	1 09E-07	7.56E-06	4.6E-09	2.0E-09	3.2E-09	1.4E-09	2.40E-01 B2

NYANZA SITE CONTAMINANTS	CANCER RISK	6 3E-06	4.3E-06	7.2E-09	5 2E-09 *
OTHER SUDBURY RIVER CONTAMINANTS	CANCER RISK	* 1.6E-05	1.3E-05	2.0E-08	2.0E-06 ·
ALL CHEMICALS OF CONCERN	CANCER RISK	2 4E-05	1.8E-05	2.0E-06	2.0E-06

BEHP:BIS(2-ETHYL HEXYL)PHTHALATE

6-82

FINAL

(

TABLE 6 - 16A FISK ASSESSMENT FESULTS FOR BEDIMENT EXPOSURE SCENARIOS EASTERN WETLANDS DRILLING NYANZA OPERABLE UNIT 3 MIDDLEGEX COUNTY, MASSACHUSETTS

							NONCAR	NOGENIC	RISK ANA	LYSIS RESU	LTSFOR	ADULT, TE	EN, AND CH	ILD RECEPTO	RS			<u></u>
CONTAMINANTS of CONCERN	CONCEN	TRATION KO	EXPOSURE RECEPTO	FACTOR R = TEEN	A ≢D	HA2 Pier	CEPTOR -	TIENTS CHILD		HAZ	ARD QUOT EPTOR =	IENTS		HAZARI	DQUOTIENT	8 T		TOXIC END- POINT
	MAX	DYA	NGESTION	DEPMAL CONTACT	(MQ - KQ -	ACCI INGE	DENTAL		RMAL NTACT				AL NCT		ENTAL STION		ALL ACT	
		L		J		-	ATG	MAX	AVG	MAA	AVG	MAA	AVU	MAA	AVG	MAA	AVG	<u> </u>
IRICHLOROETHENE*	34	0 2104	1 11E-08	2 78E -08		0 0E + 00	0.0E+00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0 0E + 00	0.0E + 00	0.0E+00	0.0E+00	NS
1,2-DICHLOROETHENE*	55	0 2821	111E-08	2 78E-08	1 00E - 02	1 0E - 03	4 BE - 05	13E-03	6 OE - 05	6 1E-04	2.9E - 05	1.5E-03	7.3E-05	54E-05	2.6E-06	2.7E-04	1.3E - 05	BLOOD
CHLOROBENZENE*	34	2 8048	1 11E-08	278E-08	2 00E - 02	3 1E - 03	2.4E-04	3 9E - 03	3 OE - 04	19E~03	14E-04	4.7E-03	3.6E-04	1.7E-04	1.3E-05	8.3E-04	6.4E-05	LIVER
NITROBENZENE*	0 14	0 1013	1 11E-08	6 66E - 07	5 00E - 04	61E-04	3.7E-04	1 3E-04	8 3E - 05	3 1E-04	2.3E - 04	1.6E-04	1.1E-04	2.7E-05	2.0E - 05	2.7E – 05	2.0E-05	L/K
1,2-DICHLOROBENZENE	13	1 8777	111E-08	6 66E - 07	9 00E - 02	2 6E - 04	3 6E - 05	8 6E - 05	9 5E - 08	1.6E~04	2 3E - 05	8.0E-05	1.2E05	1.4E-05	2.0E-06	1.4E-05	2.0E-06	LIVER
1,3-DICHLOROBENZENE	0.29	0 2730	1 11E-08	5 55E - 07	3 00E - 02	1 0E - 05	1.7E-05	4 4E-06	4 2E - 08	1 1E-05	1.0E-05	5.4E-08	5.1E-08	9 5E - 07	8 8E-07	9.5E-07	8.9E - 07	NS
1,4-DICHLOROBENZENE	31	0 5798	1 11E-08	6 66E - 07		0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	LIVER
124TRICHLOROBENZENE	0.32	0 3030	1 11E-08	5 50E - 07	1 30E - 03	4 5E - 04	4 3E - 04	1.1E-04	1.1E-04	2.7E-04	26E-04	1.4E-04	1.3E-04	2.4E05	2.3E-05	2.4E-05	2.3E-05	SKIN
NAPHTHALENE*	0.085	0.002	1 11E-08	270E-07	4 00E - 03	376-05	3.7E-05	47E-06	47E-08	2 3E - 05	2.3E-03	5.7E-06	5.7E-08	2.0E-08	2.0E-06	1.0E-06	1.0E-06	BW
ARBENIC		1 /1		0 000 + 00	3 OUE - 04	4 32 - 02	105-02	0.05+00	0.02+00	2.00-02	0.3E-03	0.05+00	0.00 +00	2.3E-03	3.8E - 04	0.02+00	0.0E+00	KIDNEY
		41.02	1112-00	0.000 + 00	1005-04	775-04	240-03	0.000	0.05 +00	475-04	4.85-05	0.02+00	0.02+00	415-05	4 15-08	0.00 + 00	0.02+00	
	6260	201.0	5 56F - 07	0.000 + 00	1002 100	0.00 + 00	n 0E+00	0.000 + 00	0.00 + 00	0.0E+00	0.0E+00	0.0E+00	0.02 +00	0.0E+00	0.0E+00	0.0E+00	0.00 + 00	CNS
MERCURY*	018	8 12	1 11E-00	5 55E - 08	3 00E - 04	5 6E - 01	3 7E - 02	1.4E-02	8 3E - 04	3 4E-01	2.3E-02	1.7E-02	1.1E-03	3.0E - 02	2.0E-03	3.0E - 03	2.0E-04	CNS
BENZENE	0 004	0 004	1 11E-00	278E-08		0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E + 00	NS
DICHLOROMETHANE	10	0 1385	111E-08	2 78E - 08	0 00E - 02	4 8E - 05	4 2E-08	0.1E-05	5 3E - 08	3.0E - 05	2.6E-08	7.4E-05	8.5E-08	2.6E-06	2.3E - 07	1.3E-05	1.1E-06	LIVER
ACETONE	0 11	0 0342	1 11E-08	2 76E - 08	1 00E-01	2 0E - 08	8 2E - 07	2 5E - 08	7 8E - 07	1.2E-08	3 6E - 07	3.1E-08	9.5E-07	1.1E-07	3.3E-08	5.4E-07	1.7E-07	L/K
ВЕНР	0 030	0 0225	1 11E-08	5 55E - 07	2 00E - 02	3 3E - 08	2.1E-08	82E-07	5.1E-07	2.0E-08	1.2E-08	1.0E-08	8.2E-07	1.6E - 07	1.1E-07	1.8E - 07	1.1E-07	LIVER
3/4METHYLPHENOL	0 074	0 074	1 11E-08	8 55E - 07	5 00E - 02	27E-08	2.7E-08	8 8E-07	8 8E ~ 07	1.8E-08	1.6E - 08	8.2E-07	6.2E-07	1.4E-07	1.4E-07	1.4E-07	1.4E07	BW,CNS
2-METHYLNAPHTHALEN	0.08	0.08	1 11E-00	2 78E - 07		0 0E+00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0 0E + 00	0.0E+00	BW
ACENAPTHYLENE	0 045	0 0313	1 11E-08	2 70E-07	4 00E - 03	21E-05	1.4E-06	2 6E - 08	1 BE - 08	1.2E-05	8.7E-08	3.1E-08	2.2E-08	1.1E-08	7.7E - 07	5.5E-07	3.8E - 07	NS
PHENANTHRENE	0 27	0 0781	1 11E-08	2 70E - 07	4 00E - 03	1 2E - 04	3 6E - 05	15E-05	4 5E - 08	7.5E-05	2.2E - 05	1.9E - 05	5.4E-08	6.6E - 06	1.9E-08	3.3E - 06	9.6E - 07	BW
FLUORANTHENE	0 26	0 1685	1 11E-00	2 70E - 07	4 00E - 02	1 2E ~ 05	7 7E - 08	1.5E-08	9.7E-07	7 2E - 06	4.7E-08	18E~06	1.2E-08	8.4E-07	4.1E-07	3.2E - 07	2.1E-07	L/K
PYRENE	0 025	0 025	1 11E-08	2 78E-07	3 00E - 02	1 6E - 08	1.6E - 08	1.9E-07	1.9E-07	9 3E - 07	9 3E - 07	2.3E-07	2.3E-07	8 2E - 08	8 2E - 08	4.1E-08	4.1E-08	KIDNEY
BENZO(A)ANTHRACENE	0 19	0.120	1 11E-08	2 78E - 07		0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	NS
CHRYSENE	01	0 0783	1 11E-08	2 78E - 07		0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0 0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	NS
BENZO(K)FLUORANTHEN	E 0.000	0 20	1 11E-08	2 78E - 07		0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	NS
BENZO(A)PYRENE	510	0 12	1 11E-08	2 70E - 07		0 0E + 00	0 0E + 00	0.0E+00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0 0E + 00	0 0E + 00	0.0E+00	0.0E + 00	NS
DIBENZ(AH)ANTHRACENE	0 12	0 12	1 11E-08	2 78E - 07		0 0E + 00	0 0E + 00	0 OE + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E +00	0.0E + 00	NS
BARIUM	64.7	32 64	1 11E-08	0 00E + 00	5 00E - 02	24E-03	12E-00	0 0E + 00	0 0E + 00	1.4E-03	72E-04	0 0E+00	0.0E+00	1.3E-04	6.4E-05	0.0E+00	0.0E+00	BLOOD
BERYLLIUM		10	1 11E-00	0.005+00	6 00E - 03	30E-03	6 GE - 04	0 0E+00	0.0E+00	162-03	405-04	0.02+00	0.0E+00	1.0E-04	3.56-05	0.05+00	0.0E+00	NS
COPPER	316	31.0	1 11E-00	0 00E + 00		0.000 + 00	0 0E + 00	0.0E+00	0 0E +00	0.02+00	0.05+00	0.0E+00	0 0E+00	0.0E+00	0.45.05	0.0E+00	0.0E+00	NS CNO
MANGANESE	470	66.31	1116-00	0 00E+00	1 00E - 01	#/E-03	186-00	0.0E+00	0.0E+00	0 3E - 03	9 8E~04	0.05+00	0.0E+00	4.7E-04	8.46-03	0.00 + 00	0.0E+00	DW
NICKEL	234	18	111E-04	0.000 + 00	2006-02	216-03	0 82 -04	0.0E+00	0.000 + 00	1 36 - 03	420-04	0.0E+00	0.0E+00	1.16-04	J.7E-03	0.0E+00	0.02+00	NO
THALLIUM			1116-00	0 000 + 00	7006-08	372-02	126-02	0.000	0.0E+00	2 20 - 02	7.JE~W	0.02+00	0.02+00	2.02-03	0.4E-04	0.02+00	0.02+00	No
ZINC	22 0	42 55	1 11E-00	0 00E+00	2 00E-01	20E-03	3 0E - 03	0 0E+00	0.0E+00	1.6E - 03	2.2E-03 2.4E-04	0.0E+00	0.0E+00	3.2E-04 1.4E-04	2.1E-05	0.0E+00	0.0E+00	BLOOD
	•		-	-												-		
NYANZA SITE CONTAMINI	INTS "			HAZARD IN	XX	ê 2€ ~ 0i	5 1E - 02	1 8E - 02	1 5E - 03	3.8E - 01	3 18-02	2.4E-02	1.8E-03	3.3E - 02	2.7E-03	4.2E - 03	3.2E-04]
OTHER SUDBURY RIVER C		ANTS		HAZARO INI	DEX.	8 1E-02	2 OE - 02	8 5E - 05	1 5E - 05	37E-02	1.2E-02	1 0E-04	1.8E-05	3.3E - 03	1.1E-03	1 8E - 05	3.1E-06	h 1

 OTHER SUDBURY RIVER CONTAMINANTS
 HAZARD INDEX
 0 1E - 02 2 0E - 02 0 5E - 05 1 5E - 05 3 7E - 02 1.2E - 02 1 0E - 04 1.8E - 05 3.3E - 03 1.1E - 03 1 8E - 05 3.1E - 06

 ALL CHEMICALS OF CONCERN
 HAZARD INDEX
 0 8E - 01 7 1E - 02 1 9E - 02 1 5E - 03 4 1E - 01 4 3E - 02 2.4E - 02 1.8E - 03 3.8E - 03 4 2E - 03 3.3E - 04

 TOXICITY ENDPOINTS ABBREVIATIONS:
 NS NOT SPECIFIED
 L/K: LIVER AND KIDNEY
 BW: BODY WEIGHT
 CNS: CENTRAL NERVOUS SYSTEM

BEHP:BIS(2-ETHYL HEXYL)PHTHALATE

FINAL

- /

TABLE 0 - 108 RISK RESULTS FOR SEDIMENT EXPOSURE WETLANDS DRILLING NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

_ _ _ _ _ _ _

			l	CAPCINOG	ENIC FISK A	NALYBIS	ESULTS		CANCER
CONTAMINANTS of	CONCEN MG/	TRATION	EXPOSUR	FACTOR		CANCERF	VSKS		SLOPE FACTOR
CONCERN	MAX	AVG	NGESTIO	DERMAL CONTACT	ACCIDEN		DERM CONT/	AL VCT	(MG/KG/D) - 1 / WEIGHT
					MAX	AVG	MAX	AVG	EVIDENCE
	34	0 2164	3 84E - 07	7 58E - 07	14E-08	8 7E - 10	2 6E - 06	1 6E - 09	1 10E - 02 B
2-DICHLOROETHENE*	55	0 2821	3 84F - 07	7 50E - 07	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
CHLORO BENZENE*	34	2 6048	3 84E - 07	7 50E - 07	0 0E + 00	0 0E + 00	0 0E+00	0 0E + 00	
NITROBENZENE*	0 14	0 1013	3 84E - 07	1 51E - 07	0 0E + 00	0 0E + 00	0.00 + 0.0	0 0E + 00	
12-DICHLOBOBENZENE	13	1 8777	3 BAE - 07	1 515-07	0.0E+00	0 0E + 00	0.0E+00	0.0E+00	
13-DICHLOROBENZENE	0 29	0 2736	3 84F - 07	1 51E - 07	0.0E+00	0 0F + 00	0.0E+00	0.0E+00	
4-DICHLOBOBENZENE	31	0.5796	1 64F - 07	1 BIE - 07	27E-08	5 1F-09	1 1E - 08	2 1E - 08	2 40F - 02 B
124TRICHLOBOBENZENE	0.32	0 3038	3 64F - 07	1 515-07	0 0F+00	0 0F + 00	0.0E+00	0.0E+00	
NAPHTHALENE*	0.082	0.082	3 84F - 07	7 56E - 08	0.0F+00	0 0E+00	0.0E+00	0.0E+00	
ABBENIC*		1 71	3 ALE - 07	0.000 + 00	4 6E - 08	1 1E-08	0.05+00	0.05 + 00	1 80E + 00 A
CADMILIM*	14	0.05	1 84E 07	0.005 +00	0.05.400	0.05 +00	0.05 + 00	0.05 + 00	10021001
	424	41 82	1 84E - 07		0.05 +00	0.05 +00	0.05 + 00	0.05 +00	
EAD	6760	201.0	1 426 - 07	0.005.00	0.05+00	0.05 + 00	0.05 + 00	0.00 + 00	
AERCURVA			3 846 07		0.02 +00	0.02 0.00	0.02 + 00		
MENCONT"	913	• 12	3 842 - 0/	1012-08	0.05+00	0.05+00	0.05+00	0.05+00	
BENZENE	0 004	0 004	3 84E - 07	7 56E - 07	42E-11	4 2E - 11	8 8E - 11	8 8E - 11	2 80E - 02 A
DICHLOROMETHANE	1.0	0 1395	3 84E - 07	7 58E - 07	4 4E - 09	3 8E - 10	9 1E-09	7 9E - 10	7 50E - 03 B
ACETONE	011	0 0342	3 84E ~ 07	7 50E - 07	0 0E+00	0 0E + 00	0 0E + 00	0 0E + 00	
BEHP	0 0.36	0 0225	3 84E - 07	1 51E - 07	1 BE - 10	1 1E - 10	7 8E - 11	4 6E - 11	1 40E - 02 B
3/4METHYLPHENOL	0 074	0 074	3 84E - 07	1 51E - 07	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
2-METHYLNAPHTHALEN	0.06	0.06	3 84E - 07	7 56E - 08	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
ACENAPTHYLENE	0 045	0 0313	3 84E - 07	7 56E - 08	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
PHENANTHRENE	0 27	0 0781	3 84E - 07	7 58E - 08	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
FLUOBANTHENE	0 26	0 1685	3 84F - 07	7 58F - 06	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
PYRENE	0.025	0.025	3 846 - 07	7 50E - 08	0.0E+00	0 0E + 00	0.0F+00	0.0E+00	
ENZO/AIANTHBACENE	0.10	0 126	3 846 - 07	7 54E - 08	4 OF - 07	2 7E - 07	8.3E-08	5 5E - 08	5 60E + 00 B
CHRYSENE	0.1	0.0781	1845-07	7 645 - 08	2 16 - 07	175-07	445-00	1 4E - 00	5 60E + 00 B
BENZONOEL UOBANTHEN	: 0 MA	0.94	3 845 - 07	7 546 - 08	1.05-08	5.95 - 07	1.05-07	12E-07	5 60E +00 D
		0.19	3 845 - 07	7 500 - 00	9 46 - 07	9 55 - 07	6 35 - 08	A 35 - 08	
	0.12		3 4 4 4 4	1 300 - 00	9 6E 07	2 50 - 07	836 00	5 30 - 00	5 000 0000
	44.7	1012	3002-07	7 DEC - UE	2 30 - 07	2000-07	9 JE - 08	5 36 - 08	3000 + 00 0
		32 34		0.002.00	0.02000		0.02+00	0.05+00	
			3 842 - 07	0002+00	1.32 - 08		0.02+00	0.02+00	4 JUE +00 A
JITEN .	310	31 0	3 842 - 07	0002+00	0.05+00	002+00	V 02 +00	000000	
WANGANESE	478	88 31	3 84E ~ 07	0 00E + 00	0 0E + 00	0 0E+00	U DE +00	0 0E+00	
NICKEL	734	75	3 842 - 07	0 00E +00	0 0E + 00	0 0E + 00	0 0E +00	0 0E +00	
HALLUM	14	0 40	3 84E - 07	0 00E + 00	0 0E + 00	0 0E +00	0 0E + 00	0 0E + 00	
/ANADIUM	72 0	13 61	3 84E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0 DE +00	0 0E + 00	
ZINC	290	42 55	3 84E - 07	0 00E + 00	0 0E + 00	0 0E+00	0 0E + 00	0 0E + 00	

NYANZA SITE CONTAMINANTS	CANCER RISK	4 9E - 08	1 1E-08	4 OE - 08	3 9E - 08
OTHER SUDBURY RIVER CONTAMINANTS	CANCER PISK	1 ØE - 05	4 3E - 08	● 2E - 07	3 2E - 07
ALL CHEMICALS OF CONCERN	CANCER FISK	2 0E - 05	5 5E - 06	€ 8E - 07	3 2E - 07

BEHP:BIS(2-ETHYL HEXYL)PHTHALATE

FINAL

6-84

4

TALLÉ 0-19A RIBK ASSESSMENT RESULTS FOR SURFACE WATER EXPOSURE SCENARIOS EASTERN WEITLANDS NYANZA OPETABLE UNIT 3 MODLESEX COUNTY, MASSACHUSETTS

	T						NONCAR	INCOENC	AISK ANAL	YSIS RESUL	LTS							
CONTAMINANTB of CONCEFIN	CONCENTR MG	ATION I	PECEP TOP	e factor	RFD (MG - KG -	HAZA		D QUOTIENTS: CHILD			ad Quotier	NTS: TEEN		HAZAR		TS: ADULT		TOXICITY END- POINT
	WAX	AVG	INCESTIC	DEFINAL	DAY)	ACCIDEN	TÁL DN	DEAM	AL NCT	ACCIDEN	TAL	DERMA	L	ACCIDENT	AL N	DERMA	L	
						MAX	AVO	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	1
TRICHLOROE THENE* 1,2-DICHLOROE THENE* 1,2-DICHLOROBENZENE CHROMIUM* LEAD* MERCURY* BARIUM COPPER	0 0006 0 003 0 003 0 0053 0 0053 0 0054 0 0094	0 0045 0 0253 0 025 0 079 0 004 0 0021 0 0085 0 0146	1 11E-00 1 11E-00 1 11E-00 1 11E-00 1 11E-00 1 11E-00 1 11E-00 1 11E-00	2 16E - 03 2 06E - 03 1 12E - 02 1 30E - 04 1 30E - 04 1 30E - 04 1 30E - 04 1 30E - 04	1 00E - 02 9 00E - 02 1 00E + 00 3 00E - 04 5 00E - 02	0 0E +00 7 3E -04 3 0E -08 7 2E -05 7 2E -05 1 2E -02 1 2E -02 1 7E -04 0 0E +00	0 0E + 00 4 8E - 04 3 0E - 05 7 2E - 05 0 0E + 00 6 4E - 03 1 8E - 04 0 0E + 00	0 0E+00 7 6E-04 1 7E-04 5 0E-08 0 0E+00 6 1E-04 1 2E-05 0 0E+00	0 0E+00 5 0E-04 1 7E-04 5 0E-08 0 0E+00 4 5E-04 1.1E-05 0 0E+00	0.0E+00 69E-04 3.7E-05 6.8E-05 00E+00 1.4E-02 2.1E-04 00E+00	0.0E+00 5.9E-04 3.7E-05 6.0E-05 0.0E+00 7.8E-03 1.9E-04 0.0E+00	0.0E+00 1.6E-03 3.7E-04 1.1E-05 0.0E+00 1.7E-03 2.6E-05 0.0E+00	0.0E+00 1.1E-03 3.7E-04 1.1E-05 0.0E+00 9.7E-04 2.3E-05 0.0E+00	0.0E+00 1.6E-04 6.5E-08 1.5E-05 0.0E+00 2.5E-03 3.7E-05 0.0E+00	0.0E+00 1.0E-04 6.5E-06 1.5E-05 0.0E+00 1.4E-03 3.3E-05 0.0E+00	0.0E+00 9.1E-04 2.1E-04 6.0E-08 0.0E+00 9.6E-04 1.4E-05 0.0E+00	0.0E+00 6.0E-04 2.1E-04 6.0E-08 0.0E+00 5.3E-04 1.3E-05 0.0E+00	NS BLOOD UVER UVER CNS CNS BLOOD NS
MANGANESE	0 108	0 1007	1 11E-03	1 38E-04	1 00E -01	9 9E - 04	02E-04	0 9E - 05	64E-05	1.2E-03	1.1E-03	1.5E-04	1.4E-04	2.1E-04	2.0E-04	8.2E-05	7.6E-05	CNS
NYANZA SITE CONTAMIN	ANTS"			HAZARD IN	DEX	1 2E - 02	7 OE - 03	17E-03	1 1E-03	1 5E-02	8 5E-03	3.8E-03	2.4E-03	2.7E-03	1.5E-03	2.1E-03	1.3E-03]
OTHER SUDBURY RIVER	CONTAMINAL	179		HAZARD IN	DEX	1 2E - 03	1 1E-03	8 1E~05	7 5E-05	1.4E-03	1.3E-03	1.7E-04	1.6E-04	2.5E-04	2.3E~04	96E-05	8.9E-05	
ALL CHEMICALS OF CON	CEPN			HAZARD IN	DEX	1 4E - 02	8 1E - 03	1 8E-03	1.2E-03	1.6E-02	9.8E-03	4.0E-03	2.6E-03	2.0E-03	1.7E-03	2.2E-03	1.4E-03]
BEPH: BIS(2ETHYL HEXYL)PHTHALATE				NOPOINTS A	BREVIATIO	2NS	NS NOT S	PECIFIED	L/K: LIV DUS SYSTE	efiand Kid M	NEY						

W92194F

1

.

1

TABLE 6-19B RISK RESULTS FOR SURFACE WATER EASTERN WETLANDS NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

					CARCINO	GENIC RIS	K RESULTS	5	
CONTAMINANTS of CONCERN	CONCENTRA MG/	TION KG	EXPOSURE	FACTOR		CANCER	RISKS		CANCER SLOPE FACTOR
	MAX	AVG	INGESTION	DERMAL	ACCIDEN	TAL	DERM	AL	(MG/KG/D)-1
	i			CONTACT	INGESTIC	DN	CONTA	NCT	1
					MAX	AVG	MAX	AVG	WEIGHT
	1								OF
									EVIDENCE
TRICHLOROETHENE*	0.0065	0.0045	3.02E-04	6.61E-04	2.2E-08	1.5E-08	4.7E-08	3.3E-08	1.10E-02 B2
1,2-DICHLOROETHENE*	0.008	0.0053	3.02E-04	6.28E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
1,2-DICHLOROBENZENE	0.003	0.003	3.02E-04	3.43E-03	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
CHROMIUM*	0.079	0.07 9	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
LEAD*	0.0053	0.004	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MERCURY*	0.0038	0.0021	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
BARIUM	0.0094	0.0085	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
COPPER	0.0146	0.0146	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MANGANESE	0.108	0.1007	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	l
NYANZA SITE CONTAMINA	ANTS*		CANCER RI	SK	2.2E-08	1.5E-08	4.7E-08	3.3E-08	
OTHER SUDBURY RIVER	CONTAMINAN	TS		SK	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ALL CHEMICALS OF CON	CERN		CANCER RI	SK	2.2E-08	1.5E-08	4.7E-08	3.3E-08	

BEPH: BIS(2ETHYL HEXYL)PHTHALATE

::

6-86

presented for the drilling samples (Table 6-18) because the average concentrations are significantly different. The accidentalingestion and dermal-contact routes of exposure (surface waters and sediments) were evaluated assuming recreational land/water use scenarios.

FINAL

Table 6-17 presents risk assessment results for sediment exposure Noncarcinogenic risks estimated for the accidentalscenarios. ingestion exposure route predominate. Mercury is the principal contaminant of concern; the hazard quotient presented for mercury approaches unity (>0.9) when the maximum contaminant concentrations are evaluated and a small child is considered the receptor of concern for the accidental-ingestion exposure route. The hazard index calculation for chemicals targeting the kidney equals unity (= 0.95) when maximum contaminant concentrations are evaluated and a small child is the receptor of concern (risks summed for the accidental ingestion and dermal contact route). The hazard index equals unity and slightly exceeds unity (HI=1.1) when the reasonable maximum-case scenario is evaluated for Nyanza site contaminants and for all COCs respectively (accidental exposure route; a small child is the receptor of concern). As discussed previously, adverse noncarcinogenic health effects are possible when the HQ or HI exceeds unity.

In contrast to the risk assessment results presented for many of the Sudbury River reaches, the hazard index calculated for the Nyanza Site contaminants detected in the Eastern Wetlands clearly exceeds that calculated for other Sudbury River contaminants.

Hazard quotients and hazard indices presented for the surface water exposure scenarios (Table 6-19) do not exceed 0.1 even when the noncarcinogenic risks are summed for the accidental-ingestion and dermal-contact exposure routes.

The excess lifetime cancer risks estimated for the sediment and surface water exposure scenarios are 2.6×10^{-5} and 2.0×10^{-5} (risks summed for the accidental-ingestion and dermal-contact exposure routes) when the reasonable maximum- and average-case scenarios are evaluated, respectively. Risks estimated for the sediment exposure scenarios exceed those estimated for the surface water exposure scenarios. The principal contaminants contributing to the risk are the carcinogenic PAHs, beryllium and arsenic, a Nyanza Site contaminant. As a point of reference, the 1E-04 to 1E-06 risk range is often evaluated in the development of health-based standards/criteria and in the determination of cleanup goals at hazardous waste sites. Cancer risks estimated for background COC concentrations in surface waters and sediments were 2E-05 and

W92194F

9.7E-06 (risks summed for the accidental-ingestion and dermalcontact exposure route) for the reasonable maximum and average case, respectively.

Risk Assessment Results for Eastern Wetlands Drilling Samples

Several organic and inorganic Nyanza Site contaminants were detected in samples (0- to 2-foot interval samples) collected during the drilling program conducted in the Eastern Wetlands:

Trichloroethene	ο	1,2-Dichloroethene
Chlorobenzene	0	Nitrobenzene
1,2-Dichlorobenzene	ο	1,3-Dichlorobenzene
1,4-Dichlorobenzene	0	1,2,4-Trichlorobenzene
Naphthalene	ο	Arsenic
Chromium	0	Lead
Mercury		
	Trichloroethene Chlorobenzene 1,2-Dichlorobenzene 1,4-Dichlorobenzene Naphthalene Chromium Mercury	TrichloroetheneoChlorobenzeneo1,2-Dichlorobenzeneo1,4-DichlorobenzeneoNaphthaleneoChromiumoMercuryo

The average mercury concentration detected in the sediment samples $(C_{xyg}=6.12 \text{ mg/kg})$ was greater than 20 times the average concentration detected in background sediment samples $(C_{xyg}=0.27 \text{ mg/kg})$.

Table 6-18 presents the risk assessment results for COC concentrations detected in the 0- to 2-foot sediment samples collected during the Eastern Wetlands drilling program. Mercury is the principal contaminant contributing to the noncarcinogenic risk. The chemical-specific hazard quotient for mercury is greater than 0.5 when a small child is evaluated as the receptor of concern. However, chemical-specific hazard quotients and hazard indices calculated for each exposure route do not exceed unity even when the reasonable maximum case is evaluated. Also, if hazard indices are summed for the accidental-ingestion and dermal-contact exposure routes, the total hazard index does not exceed unity. These results indicate that adverse noncarcinogenic health effects would not be anticipated under the conditions of the recreational exposure scenarios defined in Section 6.4.

Excess lifetime cancer risks estimated for maximum and average COC concentrations in the 0- to 2-foot samples slightly exceed 2E-05 and 5.5E-06 (risks summed for accidental-ingestion and dermal-contact exposure routes), respectively. As a point of reference, cancer risks estimated for background sediment concentrations were 1.8E-05 and 8.2E-06 for the reasonable maximum and average case scenarios, respectively. The carcinogenic PAHs, beryllium, and arsenic, a Nyanza Site contaminant, are the principal contaminants contributing to the estimated risk. It should be noted that cancer

FINAL

risks estimated for Nyanza Site contaminants are less than those estimated for "other Sudbury River contaminants."

6.5.3.5 Risk Assessment Results for Chemical Brook Culvert

Chemical Brook Culvert is fed by Chemical Brook and Trolley Brook. Water from the culvert discharges to Outfall Creek. Chemical and Trolley Brooks were primary surficial drainage routes from the Nyanza Site and the Eastern Wetlands. Several organic and inorganic Nyanza Site contaminants were detected in sediment samples collected from the culvert:

0	Trichloroethene	0	1,2-Dichloroethene
0	Chlorobenzene	0	Nitrobenzene
0	1,2-Dichlorobenzene	0	1,3-Dichlorobenzene
0	1,4-Dichlorobenzene	0	1,2,4-Trichlorobenzene
0	Naphthalene	0	Arsenic
0	Chromium	0	Lead
0	Mercury		
	—		

The average mercury concentrations detected in the Brook Culvert sediments (C_{avg} =6.8) were greater than 25 times the average concentration detected in background sediments (C_{avg} =0.27 mg/kg).

Table 6-20 presents risk assessment results for COCs detected in sediment samples collected from the Culvert. The accidentalingestion and dermal-contact routes of exposure (surface waters and sediments) were evaluated assuming recreational land/water use scenarios. Although mercury was detected in excess of background concentrations, the chemical-specific hazard quotients and hazard indices presented for each exposure route do not exceed unity. If hazard indices are summed for the accidental-ingestion and dermalcontact exposure routes, the total hazard index does not exceed unity. These results indicate that adverse noncarcinogenic health effects are not anticipated for a receptor contacting the sediments under the conditions of the recreational exposure scenarios defined in Section 6.4.

Arsenic, the carcinogenic PAHs, and beryllium are the principal contaminants contributing to the estimated excess lifetime cancer risks presented on Table 6-20. Cancer risks estimated for several carcinogenic PAHs exceed 1E-06. Arsenic is the only Nyanza Site contaminant demonstrating a cancer risk exceeding 1E-06. The cancer risks estimated for the sediment exposures are 1.6E-05 and 1.4E-05 (risk summed for the accidental-ingestion and dermalcontact routes) when the reasonable maximum- and average-case

W92194F

TABLE 8-20A RISK ASSEBSMENT RESULTS FOR SEDIMENT EXPOSURE SCENARIOS CHEMICAL BROOK CULVERT NYANZA OPERABLE UNIT 3 MODILESEX COUNTY, MASSACHUSETTS

							NONCARC	NOGEN	RISK ANA	LYSIS RESI	ILTS							
CONTAMINANTS of CONCERN	CONCEI			FACTOR R = TEEN	₽FD	н	AZARD QU CHIL	OTIENT D		HA	teen	TIENT		ŀ	1AZARD QUC ADUL	DTIENT T		TOXIC END POINT
	MAX MG/KQ	AVG MG/KG	INGESTION	-DERMAL CONTACT	(MG - KG -			DERN CONT.						ACCIDENTA	L	DERMAL CONTACT	AVG	-
·····	<u></u>	I						A	AVU	MAA			AVG		AVG		AVG	+
TRICHLOROETHENE	0 072	0 0377	1 11E-00	2 78E - 00		0 0E + 00	0.0E+00	0.0E+00	0 0E + 00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	NS
1,2-DICHLOROETHENE*	0 043	0 025	1 11E-08	2 70E - 00	1 DOE 02	7 9E 06	4 6E - 06	9 8E - 06	57E-06	4.8E-08	2.8E-06	1.2E-05	6.9E~06	4.2E - 07	2.4E-07	2.1E-06	1.2E – 08	BLOOD
CHLOROBENZENE	0 077	0 0 5 2	1 11E - 00	2 78E - 08	2 00E - 02	7 OE - 06	47E-06	6 6E - 08	5 9E - 08	4.3E-08	2.9E-08	1.1E-05	7.2E~0 0	3.8E-07	2.5E - 07	1.9E-08	1.3E-08	UVER
NTROBENZENE*	0 31	0 31	1 11E-00	5 55E - 07	6 00E - 04	1 1E-03	1 1E-03	2 6E - 04	2 8E - 04	6.9E-04	0.9E-04	3.4E-04	3.4E04	0.1E-05	8.1E-05	6.1E-05	6.1E-05	L/K
1,2-DICHLOROBENZENE		1 05	111E-00	5 55E - 07	00E-02	3 9E - 05	3 3E - 05	9.6E-06	0.4E-06	2.3E-05	2.0E-05	1.2E-05	1.0E~05	2.1E-08	1.8E-06	2.1E-08	1.8E-08	UVER
1,3-DICHLOROBENZENE	1 01	01	111E-00	5 55E - 07	3 00E - 02	01E-06	0.1E-00	1.5E-00	1.5E-08	3.7E-06	3.7E-08	1.9E-06	1.9E~08	3 3E - 07	3.3E-07	3.3E-07	3.3E-07	NS
1,4-DICHLOHOBENZENE	0.50	0 485	1112-00	0 55E - 07	1 905 09	0.0E+00	0.0E+00	0.05+00	0.05+00	0.0E+00	U.UE + 00	0.02+00	0.0E+00	0.0E+00	0.0E+00	0.02+00	0.0E+00	
124 THICHLOHOBENZENE		0 0 0 35	111E-00	9 765 07	1 JOE - 03	12E-03	12E-03	3.12-04	2 0E -04	1.42-04	7.1E-04	3.7E-04	3.0E~04	0.5E-05	8.3E-05	7 36 00	8.3E-05	
ABGENIC		41	1116-00	0.005 + 00	1 00E - 04	4 0F 02	2 65 -04	0 0F + 00	0 0F + 00	2 4E -02	1 AF _ 02	4.2E-03	0.0E+00	2 2E = 03	1.4E-03	0.05+00	0.00+00	SKIN
CHROMUM*	1.15	82.3	1116-08	0.000 + 00	1 00E + 00	2 5E - 04	1 5E - 04	0 0F + 00	0.0E+00	1.5E-04	0 1F-05	0.0E+00	0.0E+00	1.3E-05	8 1E-08	0.0E+00	0.0E+00	UVER
LEAD"	57 5	38 25	5 55E - 07	0 00E + 00		0 0E + 00	0 0E+00	0 0E + 00	0 0E + 00	0 0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	CNS
MERCURY	71		1 11E - 08	8 55E - 08	3 00E - 04	4 3E - 02	4 1E - 02	1 1E-03	1 OE - 03	2.8E-02	2.5E-02	1.3E-03	1.3E~03	2.3E - 03	2.2E-03	2.3E-04	2.2E-04	CNS
BENZENE	0 002	0 002	1 11E-00	2 78E - 08	1	0 0E+00	0 0E + 00	0 0E + 00	0.0E+00	0 0E + 00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	NS
BEHP	0 2	0 15	1 11E - 08	5 65E - 07	\$ 00E - 02	1 8E - 05	1 4E - 05	4 6E - 08	3.4E - 06	1.1E-05	8.3E-06	5.8E-08	4.2E~08	9.8E-07	7.3E - 07	9.6E – 07	7.3E-07	UVER
2-METHYLNAPHTHALEN	014	0 18	1118-00	2 78E - 07	_	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E + 00	0.0E+00	0.0E + 00	8W
ACENAPTHYLENE	014	0 14	1 11E-00	2 78E - 07	4 00E - 03	€ 4E - 05	6 4E - 05	6 0E - 08	8 OE - 08	3 9E - 05	3.9E-05	9.7E-08	9.7E~08	3.4E-08	3.4E-08	1.7E-06	1.7E-06	NS
PHENANTHRENE	0.76	0 70	1 11E -08	2 70E - 07	4 00E -03	3 8E -04	3 5E 04	4 5E - 05	4 3E - 05	2.2E-04	2.1E-04	5.5E-05	5.3E-05	1.9E-05	1.9E-05	9.7E-06	9.3E-06	BW
FWORANTHENE		13	1 11E-00	278E-07	4 00E -02	842-03	59E-03	8 05 -08	7.4E-06	3.0E-05	3.8E-05	9.7E-08	B.UE~08	3.4E-06	3.22-08	1./12-08	1.6 - 08	KIDNEY
PTHENE BENTOWN			1112-00	2702-07	3 005 -04	7 3E - 03	7 32 - 03	0.05+00	0.05+00	440-00	9.40-03	0.05+00	0.05 +00	0.05+00	3.0E - 00	2.02-08	2.02-00	NONET
CHOVEENE	0.74	0745	1116-00	2785-07		0.05+00	0.0E+00	0.0E+00	0.00 + 00	0.0E+00	0.02+00	0.0E+00	0.0E+00	0.000	0.0E+00	0.02+00	0.0E+00	NS
IRENZO/BIELUORANTHEN		0 82	1116-00	2 78E -07		0.0E+00	0.0E+00	0.0E+00	0 0F+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0 0F + 00	NS
BENZOINFLUORANTHEN	E 070	0.00	1 11E-00	2 78E - 07		0 0E+00	0 0E + 00	0 0E+00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	NS
BENZOLAPYRENE	0 77	0 72	1 11E-00	2 78E -07		0 0E+00	0 0E+00	0 0E+00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E + 00	NS
INDENO(123-CO)PYRENE	0 31	0 33	1 11E-08	2 78E - 07		0 0E+00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0 0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	NS
BENZO(GHI)PERYLENE	0 21	0 27	1 11E-00	2 70E - 07	4 00E - 03	1 2E - 04	1 2E 04	1 5E - 05	1.5E-05	7.6E~05	7.5E-05	1.9E05	1.9E-05	6.6E~06	6.6E - 06	3.3E - 08	3.3E - 06	
BARIUM	25 0	18.05	1 11E-00	0 00E + 00	8 00E - 02	● 5E - 04	8 8E - 04	0 0E + 00	0 0E + 00	5 6E - 04	4.1E-04	0.0E+00	0.0E + 00	5.1E-05	3.8E - 05	0.0E + 00	0.0E + 00	BLOOD
BERYLUUM	0.54	0 45	1 11E-00	0 00E + 00	5 00E - 03	2 0E - 04	1 OE - 04	0 0E + 00	0 0E + 00	1 2E-04	1.0E-04	0.0E + 00	0.0E+00	1.1E~05	6.8E - 06	0.0E+00	0.0E + 00	NS
COPPER	02 5	63 35	1 11E-00	0 00E + 00	_	0 0E+00	0 0E + 00	0 0E+00	0 0E + 00	0 0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	0.0E + 00	NS
MANGANESE		90 8	1 11E-00	0 00E + 00	1 00E - 01	2 2E - 03	1 7E - 03	0 0E + 00	0 0E + 00	1.3E-03	1.0E-03	0.0E+00	0.0E+00	1.2E04	8.9E-05	0.0E+00	0.0E + 00	CNS
VANADIUM	1	12	1 11E-00	0 DOE + 00	7 00E -03	47E-03	J.1E-03	0 0E + 00	0 0E + 00	2.9E-03	1.9E-03	0.0E+00	0.0E + 00	2.5E-04	1.7E-04	U.UE + 00	0.0E+00	NS
ZINC	T 103	97.05	1118-06	U DOE + 00	2 00E-01	0 4E - 04	8 5€ -04	U 0E + 00	U 0E +00	6.7E-04	5.2E-04	U.UE + 00	U.0E+00	1 5.0E-05	4.52-05	0.0E+00	0.0E + 00	IBLOOD

NYANZA SITE CONTAMINANTS	HAZARD INDEX	8 6E-02 7 0E-0	2 17E-03 1.7E-03	5 3E-02 4.3E-02 2.1E-0	3 2.0E-03 4.6E-0	3 3.8E - 03	3.7E-04 3.6E-04
OTHER SUDBURY RIVER CONTAMINANTS	HAZARD INDEX	07E-03 72E-0	3 90E-05 87E-05	59E-03 4.4E-03 1.1E-0	4 1.1E-04 5 2E-04	4 3.8E-04	1.9E-05 1.9E-05
ALL CHEMICALS OF CONCERN	HAZARD INDEX	9 6E - 02 7.8E - 0	2 18E-03 18E-03	5 8E-02 4.7E-02 2.2E-0	3 2.1E-03 5.1E-0	3 4.2E - 03	3.8E-04 3.8E-04
	TOXICITY ENDPOINTS	BBREVIATIONS:	NS NOT SPECIFIED	L/K: UVER AND KIDNEY	BW: BODY WEIGHT	CNS: CENTRA	L NERVOUS SYSTEM

BEHP BIS(2 ~ ETHYL HEXYL)PHTHALATE

FINAL

ł

TABLE 6-20B RISK RESULTS FOR SEDIMENT EXPOSURE CHEMICAL BROOK CULVERT NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

					CARCINO	SENIC RIS	K ANALYS	S	
CONTAMINANTS of	CONCEN	ITRATION	EXPOSURE	FACTOR	.	CANCER	AISKS		SLOPE FACTOR
CONCERN	MAX	AVG	INGESTION	DERMAL CONTACT	ACCIDEN	TAL DN	DERM CONT/	AL ACT	(MG/KG/D)-1 / WEIGHT
	WU/KU	MU/AU			MAX	AVG	MAX	AVG	EVIDENCE
TRICHLOROETHENE*	0 072	0 0377	3 64E 07	7 56E - 07	2.9E - 10	1.5E-10	6.0E - 10	J.1E-10	1.10E-02 B
1.2-DICHLOROFTHENE*	0 043	0 025	3 64E - 07	7 56E - 07	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
CHLOROBENZENE*	0 077	0 052	3 64E - 07	7 56E-07	0.0E+00	0.0E+00	0.0E+00	0 0E + 00	
NITROBENZENE*	0 31	0 31	3 64E - 07	1 51E - 07	0 0E + 00	0.0E+00	0.0E+00	0 0E + 00	
1,2-DICHLOROBENZENE	19	1 65	3 64E - 07	1 51E - 07	0 0E + 00	0.0E+00	0 0E + 00	0 0E + 00	
1.3-DICHLOROBENZENE	01	01	3 64E - 07	1 51E-07	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
1,4-DICHLOROBENZENE	0 59	0 465	3 64E - 07	1 51E-07	5.2E-09	4.1E~09	2.1E-09	17E-09	2.40E-02 B
124TRICHLOROBENZENE	0 87	0 835	3 64E - 07	1 51E-07	0 0E + 00	0 0E+00	0 0E + 00	0 0E + 00	
NAPHTHALENE*	08	0 565	3 64E - 07	7 56E - 06	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
ARSENIC*	66	4 3	3 64E - 07	0 00E + 00	4 3E - 08	2 6E - 06	0.0E+00	0 0E + 00	1.80E+00 A
CHROMIUM*	135	82 3	3 64E - 07	0 00E + 00	0.0E+00	0.0E+00	0.0E+00	0 0E + 00	
LEAD*	57 5	38 25	1 82E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0.0E + 00	0 0E + 00	
MERCURY*	71	68	3 64E - 07	1 51E-08	0 0E + 00	0 0E + 00	0.0E+00	0 0E + 00	
BENZENE	0 002	0 007	3 64E - 07	7 56E - 07	2 1E-11	2 1E-11	4 4E-11	4 4E-11	2 90E~02 A
ВЕНР	02	0 15	3 64E - 07	1 51E-07	1 0E - 09	7.6E - 10	4 2E - 10	3 2E ~ 10	1.40E-02 B
2-METHYLNAPHTHALENE	0 18	0 18	3 64E ~ 07	7 56E-08	0 0E + 00	0 0E + 00	0.0E+00	0 0E + 00	
ACENAPTHYLENE	0 14	0 14	3 64E - 07	7 56E - 08	0 0E + 00	0 0E + 00	0.0E+00	0 0E + 00	
PHENANTHRENE	0 79	0 76	3 64E - 07	7 56E - 08	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
FLUORANTHENE	14	13	3 64E - 07	7 56E - 08	0 0E + 00	0 0E + 00	0 0E+00	0 0E+00	
PYRENE	12	1 2	3 64E - 07	7 56E-08	0 0E + 00	0 0E + 00	0.0E+00	0 0E+00	
BENZO(A)ANTHRACENE	0.81	0 7 1	3 64E - 07	7 56E - 08	17E-08	1 5E - 08	3 5E - 07	3 1E - 07	5 80E + 00 B
CHRYSENE	0 76	0 745	3 64E - 07	7 56E - 08	16E-08	1 6E - 06	3 3E - 07	3 3E ~ 07	5.80E+00 B
BENZO(B)FLUORANTHEN	0 64	0 62	3 64E - 07	7 56E - 08	18E-08	1 7E - 08	37E-07	36E-07	5.80E+00 B
BENZO(K)FLUORANTHEN	0.78	0 64	3 64E - 07	7 56E - 06	1 6E - 08	14E-08	3 4E - 07	3 OE ~ 07	5 80E+00 B
BENZO(A)PYRENE	0 77	0 72	3 64E - 07	7 56E - 08	16E-08	1 5E - 06	3.4E-07	32E~07	5.80E+00 B
INDENO(123-CD)PYRENE	0 33	0 33	3 64E - 07	7 56E - 08	7.0E-07	7.0E - 07	1.4E-07	1 4E ~ 07	5.80E+00 B
BENZO(GHI)PERYLENE	0 27	0 27	3 64E - 07	7 56E - 08	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
BARIUM *	25 9	18 65	3 64E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E+00	
BERYLLIUM	0 56	0 45	3 64E - 07	0 00E + 00	88E-07	7.0E - 07	0.0E+00	0 0E + 00	4.30E+00 A
COPPER	92 5	63 35	3 64E - 07	0 00E + 00	0 0E + 00	0.0E+00	0 0E+00	0 0E + 00	
MANGANESE	119	90.6	3 64E - 07	0 00E + 00	0 0E+00	0 0E + 00	0.0E+00	0.0E+00	
VANADIUM	16	12	3 64E - 07	0 00E+00	0.0E+00	0.0E+00	0.0E+00	0 0E+00	
ZINC	103	82 95	3 64E - 07	0 00E+00	0 0E + 00	0.0E+00	0 0E + 00	0 0E + 00	

NYANZA SITE CONTAMINANTS	CANCER RISK	4 3E - 06	28E-08	2 7E - 09	2 0E - 09
OTHER SUDBURY RIVER CONTAMINANTS	CANCER RISK	9 9E - 08	9.2E-06	1.9E-08	1 6E 06
ALL CHEMICALS OF CONCERN	CANCER RISK	1 4E - 05	1 2E - 05	1.9E-08	1 BE - 06

BEHP:BIS(2-ETHYL HEXYL)PHTHALATE

scenarios are evaluated, respectively. As a point of reference, cancer risks estimated for background sediment concentrations were 1.8E-05 and 8.2E-06 for the reasonable maximum and average case, respectively. Thus, the cancer risks estimated for Chemical Brook Culvert sediments are similar to background levels.

6.5.3.6 <u>Risk Assessment Results for Outfall Creek</u>

Chemical Brook Culvert discharges to Outfall Creek. The following Nyanza Site contaminants were detected in surface waters and sediments of Outfall Creek:

- o Trichloroethene
- o Nitrobenzene
- o 1,2,4-Trichlorobenzene
- o 1,4-Dichlorobenzene
- o Chromium
- o Mercury
- o 1,2-Dichloroethene
- o 1,2-Dichlorobenzene
- o Naphthalene
- o Arsenic
- o Lead

The average concentration of mercury in the creek sediments is greater than 100 times the concentration detected in background sediment samples. The maximum concentration of trichloroethene detected in creek surface waters (13 μ g/L) exceeds the current Federal SDWA MCL (5 μ g/L).

Risk assessment results for Outfall Creek are presented on Tables 6-21 and 6-22. The accidental-ingestion and dermal-contact routes of exposure (surface waters and sediments) were evaluated assuming recreational land/water use scenarios. Table 6-21 summarizes risk assessment results for sediment exposure scenarios. Although mercury was detected at concentrations 100 times background, the chemical-specific hazard quotients and hazard indices presented on Table 6-21 do not exceed unity. The HQ calculated for mercury does not exceed 0.6 in many of the cases presented. If hazard indices are summed for the accidental-ingestion and dermal-contact exposure routes, the total hazard index does not exceed unity. If the noncarcinogenic risks presented for the surface water exposure scenarios (Table 6-22) are also considered (the risks associated with the sediment-exposure scenarios are added to those associated with the surface water exposure scenarios), the total hazard index does not exceed unity. These results suggest that adverse noncarcinogenic health effects are not anticipated for a receptor

W92194F

TABLE 0-21A RISK ASSESSMENT REBULTS FOR BEDIMENT EXPOSURE SCENARIOS OUTFALL CREEK NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

							NONCAR		RISK ANA	LYSIS RESI	ULTS							
CONTAMINANTB of CONCERN		ITRATION	EXPOSURE RECEPTO	FACTOR	H FD	HAZA		ENTS: CHI	10	HAZ		NENTS: T	EEN	HAZ	ARD QUOTIE	NTS ADULT		TOXIC END- POINT
	MAX	AVO	INGEBTION	DERMAL	(MQ - KQ -	ACO	DENTAL	DE CO		ACCIDEN		DERM	AL ACT	ACCID	STION	DERN	AL	
			1		DAY	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG]
			I								.							1
TRICHLOROETHENE	0 004	2 43	1 11E-00	2 70E - 00		0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.02+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	INS
1,2-DICHLOROETHENE*	0 001	1	1 11E-00	2 / 8E - 08	1006-02	100 - 07	185-04	2 36-07	2.3E-04	1.1E-07	1.1E-04	2.00-0/	2.85-04	A 95 - 08	9.85 ~06	4.9E - 06	4.8E-03	BLOOD
NTROBENZENE*	03	035	1112-00	5 55E - D/	5 DOE - D4	1 BE-03	1.3E-UJ	402-04	J 2E -04	1.1E-03	7.86~04	1.00 -04	3.90-04	9.8E - UJ	0.0E-03	3 35 07	0.8E-03	
1.2 - DICHLOROBENZENE	020	0 24/3	1116-00	B 55E - 07	1 306 - 02	1 85 - 01	116-00	486-04	2 8E ~ 04	11E-01	5.1C~00	5 86 -04	3 4E -04	9.85-05	2.7E-07	9 AE 05	2.72-07 8.0E-05	ISKIN
NADUTHALENET	0.14	0 2183	1116-00	2 78E - 07	4 005 - 03	1 BE - 04	1 0F - 04	18F-05	1 2E - 05	R 4E - 05	8 1E - 05	2 4E - 05	15E-05	6 3E - 06	5 3E -06	4 2E - 08	27E-08	BW
ABBENIC*	41	7 61	1 116-06	0.006 + 00	3 00F - 04	2 SE - 02	1 8E - 02	00F+00	0 0E+00	15E-02	1 1E-02	0.0E+00	0.0E+00	1 3E - 03	9.6E-04	0.0E + 00	0 0E+00	SKIN
CHROMIIM		341.8	1115-06	0 00E + 00	1006+00	1 8F - 03	B 2E - 04	0 0F+00	0 0E + 00	1.1E-03	3 6E - 04	0 0E + 00	0.0E+00	9.7E-05	3.3E-05	0.0E+00	0.0E+00	UVER
LEAD-	233	105 85	3 35E - 07	0 00E + 00		0 0E + 00	0 0E + 00	0 0E+00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	0.0E+00	0.0E+00	CNS
MERCURY*	99.2	35 33	1 11E-00	5 55E - 00	3 00E - 04	0 0E - 01	2 2E - 01	1 5E - 02	54E-03	37E-01	1.3E-01	1.8E-02	6 5E - 03	3 2E - 02	1.2E-02	3.2E-03	1.2E-03	CNS
			1															
DICHLOROMETHANE	0 055	8 0222	111E-00	278E-00	0 00E - 02	17E-06	6 7E - 07	2 1E-00	84E-07	1.0E~06	4.1E-07	2.5E - 08	1.0E-06	9.0E-08	3 8E ~ 08	4 5E - 07) 8E-07	UVER
ACETONE	0 034	0 0 1 57	1 11E-00	2 70E - 08	1 00E - 01	8 2E - 07	2 9E - 07	7 8E - 07	3 8E - 07	3.8E ~ 07	1.7E - 07	9.4E-07	4.4E-07	3 3E - 08	1.5E-08	1.7E – 07	7.7E-08	U/K
BEHP	2	0 86	1 11E-00	5 55E - 07	2 00E - 02	1 8E - 04	8 8E - 05	4 0E - 05	2 2E - 05	1 1E-04	5.3E - 05	5.8E - 05	2.7E-05	9.6E - 06	4.7E-08	9.6E - 06	4.7E-06	UVER
ACENAPTHYLENE	0 31	0 2033	1 11E-00	2 70E - 07	4 00E - 03	1 4E - D4	1.2E-04	1 BE - 05	1 5E - 05	9 8E - 05	7.3E – 05	2.2E-05	1.8E-05	7.8E-08	6.4E-08	3.8E - 06	3.2E - 08	NS
PHENANTHRENE	0 03	0 03	1 11E-00	2 70E - 07	4 00E - 03	4 2E - 04	2 9E - 04	5 3E - 05	3 6E - 05	2 6E ~ 04	1 7E - 04	8.5E-05	4.4E05	2.3E - 05	1.5E - 05	1.1E-05	7.7E - 08	BW
FWORANTHENE	10	1 3087	1 11E - 08	2 70E - 07	4 00E - 02	7 3E - 05	6 OE - 05	1E-00	7 5E - 08	4 4E ~ 05	3 6E - 05	1.1E-05	0.1E-00	3.9E - 08	3.2E - 08	2.0E - 06	1.6E-08	L/K
PYRENE	25	1.	1 11E - 06	2 78E - 07	3 00E - 02	1 SE - 04	1 18-04	1 9E - 05	1 4E-05	9 3E ~ 05	0.7E~05	2.3E-05	1.7E-05	6 2E → 06	5.9E-06	4.1E-06	2.9E-06	KIDNEY
BENZO(A) ANTHRACENE	1 05	08	1 11E-00	2 78E - 07		0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	NS
CHRYSENE	14	1 0767	1 11E-00	270E-07		0 OE + 00	0.000000	0.000+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	002+00	0.0E + 00	0.0E+00	NS
BENZO(B)FWORANTHEN	178	1 2333	1 11E-00	278E-07		0.000 + 000	0000000	0000000	0 0E + 00	0.0E+00	0.02+00	0.05+00	0.02+00	0.0E+00	0.02+00	0.05+00	0.0E+00	INS .
BENZO(K)FLUORANTHEN		0 4733	111E-00	2 / UE - 0/		0.05+00		0000+00		0.0E+00	0.000	0.00 + 00	0.000	0.000	0.00+00	0.00 + 00	0.0E+00	INS NO
BENZO(A)PTHENE	12	0.4533	1116-00	2702-07					0.05 + 00	0.05+00	0.00 + 00	0.02+00	0.05+00	0.02+00	0.02+00	0.00 + 00	0.02+00	NO
INDENU(123-CD)PTHENE	0 0 3 7	0 4533	1116-00	2786-07	4 005 - 01	1 16 - 04	2 2 E - 04	425-05	2 AF - 05	205-04	1 4F - 04	5 1E - 05	34E-05	1.8E - 05	1.2E-05	896-06	6 0E - 06	NR
BADUIAA	58.5	31.84		0.005.400	B 006 - 07	916-01	12E-01		0.0E+00	1.3E~03	7 OF -04	0.0E+00	0.0E+00	1 1E-04	8 2E - 05	0.0E+00	0.0E+00	BLOOD
BERVELIUM	29	1 07	1115-06	0 00F + 00	5 00E - 03	0 OF - 04	3 9E - 04	0 0E+00	0 0E + 00	4.0E-04	2.4E-04	0.0E+00	0.0E+00	4.3E-05	2.1E-05	0.0E + 00	0.0E+00	NB
COPPER	354	137 83	1116-08	0 00E+00		0 0E+00	0 0E + 00	0 0E+00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E + 00	0.0E+00	NS
MANGÁNESE	308	184 57	1 11F-06	0.000 + 00	1 00E - 01	5.6E-01	3 0E - 03	0 0E+00	0 0E + 00	3.4E-03	1.8E-03	0.0E+00	0.0E+00	3.0E-04	1.8E-04	0.0E + 00	0.0E+00	CNS
VANADILIM	30.1	17.24	1 11E-08	0 00E+00	7 00E - 03	# 4E - 03	4 5E - 03	0 0E+00	0 0E+00	5.7E-03	2.7E-03	0.0E+00	0.0E+00	5.0E-04	2.4E-04	0.0E+00	0.0E+00	NS
ZINC	300	108.3	1 11E-08	0 00E+00	2 00E - 01	3 6E - 03	1.5E-03	0.0E+00	0 0E + 00	2.2E - 03	0.2E-04	0.0E+00	0.0E+00	1.9E-04	8.1E-05	0.0E+00	0.0E + 00	BLOOD
	han		•															
NYANZA RITE CONTAMINA	NTR			HAZARD IN	DEX	8 3E - 01	2.4E-01	1 6E - 02	6 2E - 03	3.9E-01	1.4E-01	1.9E-02	7.8E -03	3.4E-02	1.3E-02	3.4E - 03	1.3E-03	۰ ۲
OTHER SUDBURY RIVER C	ONTAMIN	ANTE		HAZARDIN	DEX	2 3E - 02	1 1E-02	1 9E - 04	1 2E - 04	1.4E - 02	7.0E-03	2.3E-04	1.5E – 04	1.2E - 03	6.1E-04	4.1E-05	2.8E - 05	

6 6E -01 2 5E -01 1 6E -02 0 3E -03 4.0E -01 1 5E -01 2 0E -02 7.7E -03 3.5E -02 1.3E -02 3.5E -03 1.4E -03

BW: BODY WEIGHT

CNS: CENTRAL NERVOUS SYSTEM

NS: NOT SPECIFIED L/K: LIVER AND KIDNEY

BEHP:BIS(2-ETHYL HEXYL)PHTHALATE

HAZARD INDEX

TOXICITY ENDPOINTS ABBREVIATIONS:

ALL CHEMICALS OF CONCERN

FINAL

TABLE 0-21B RISK RESULTS FOR SEDMENT EXPOSURE OUTFALL CREEK NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

				CARCINOGE	ENIC RISK A	NALYSIS I	RESULTS		
CONTAMINANTS of	CONCEN	TRATION	EXPOSURE	FACTOR		CANCER F	NSKS		CANCER SLOPE FACTOR
CONCERN	MAX	AVG	INGESTION	DEFINAL CONTACT	ACCIDEN	TAL XN	DERM	AL ACT	(MG/KG/D) 1 / WEIGHT
					MAX	AVG	MAX	AVG	OF EVIDENCE
TRICHLOBOETHENE	0 004	2 63	3 64E-07	7 56E-07	1 8E-11	1.1E-08	3.3E-11	2 2E-08	1 10E-02 82
1.2-DICHLOROETHENE*	0 001	1	3 64E-07	7.58E-07	0 0E+00	0.0E+00	0 0E + 00	0.0E+00	
NITROBENZENE*	05	0 35	3 64 E - 07	1 51E-07	0.0F+00	0.0E+00	0.0E+00	0.0E+00	
12-DICHLOBOBENZENE	0 29	0 2475	3 64F-07	1 51E-07	0.0E+00	0.0F+00	0 0F+00	0 0F + 00	
124TRICHLOROBENZENE	13	0 8008	3 64E-07	1 51E-07	0.0E+00	0.0E+00	0 0E + 00	0 0E + 00	
NAPHTHALENE*	0.34	0 2183	3 64F - 07	7 56E-08	0.0E+00	0.0E+00	0 0F + 00	0 0F + 00	
ARSENIC*	41	2 93	3 64E-07	0 00E + 00	27E-06	19E-06	0 0E + 00	0 0E+00	1.80E+00 A
CHROMIUM*	968	341.8	3 64E - 07	0 00E + 00	0.0F+00	0 0E + 00	0 0E + 00	0 0E + 00	
LEAD*	233	105 95	1 82F - 07	0 00E+00	0.0E+00	0 0E+00	0.0F+00	0 0E + 00	
MERCURY*	99 2	35 33	3 64E-07	1.51E-08	0.0E+00	0 0E + 00	0.0E+00	0 0E + 00	
DICHLOROMETHANE	0 055	0 0222	3 64E-07	7 56E-07	1.5E- 10	8 0E-11	3 1E- 10	1 3E- 10	7.50E-03 82
ACETONE	0 034	0 0157	3 64E - 07	7 56E-07	0 0E + 00	0 0E+00	0.0E+00	0 0E+00	
BEHP	2	0 96	3.64E-07	1.51E-07	1.0E-08	4.9E-09	4.2E-09	2.0E-09	1.40E-02 B2
ACENAPTHYLENE	0 31	0 2633	3 64E-07	7.56E-08	0.0E+00	0.0E+00	0.0E+00	0 0E + 00	
PHENANTHRENE	0 93	0 63	3 64E-07	7 58E-08	0.0E+00	0.0E+00	0.0E+00	0 0E+00	
FLUORANTHENE	16	1 3067	3 64E-07	7.56E-08	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	
PYRENE	25	1.0	3 64E-07	7 56E-08	0.0E+00	0.0E+00	0.0E+00	0 0E + 00	
BENZO(A)ANTHRACENE	1 05	0.8	3 64E-07	7 58E-08	2.2E-08	1.7E-08	4 8E-07	3 5E - 07	5.80E+00 B2
CHRYSENE	14	1 0787	3 64E-07	7.58E-08	3 0E-06	2.3E-06	8.1E~07	4.7E-07	5.80E+00 B2
BENZO(B)FLUORANTHEN	1 75	1 2333	3 64E-07	7 56E-06	37E-06	2.6E-08	7.7E-07	54E-07	5.60E+00 B2
BENZO(K)FLUORANTHEN	1	0 4733	3 64E~07	7 56E~08	2.1E-06	10E-08	4.4E-07	2 1E-07	5.80E+00 82
BENZO	12	0.0	3 64E~07	7.56E-06	2.5E-08	1.9E-08	5 3E-07	3 9E-07	5.60E+00 B2
INDENO(123-CD)PYRENE	0 635	0 4533	3 64E-07	7 56E-08	1.3E-06	● 6E07	2.6E-07	2.0E-07	5.80E+00 82
BENZO(GHI)PERYLENE	0 73	0 4917	3 64E-07	7 56E-08	0 0E+00	0 0E + 00	0.0E+00	0.0E+00	
BARIUM	58 5	31 58	3 64E-07	0 00E+00	0.0E+00	0.0E+00	0.0E+00	0 0E+00	
BERYLUUM	22	1 07	3 64E-07	0 00E+00	34E-06	17E-08	0.0E+00	0 0E+00	4.30E+00 A
COPPER	359	137 83	3 64E-07	0 00E + 00	0 0E+00	0.0E+00	0.0E+00	0.0E+00	
MANGANESE	309	164 57	3 64E-07	0 00E+00	0 0E+00	0.0E+00	0.0E+00	0.0E+00	
VANADIUM	36 1	17 28	3 64E-07	0 00E+00	0 0E+00	0 0E+00	0 0E+00	0 0E + 00	
ZINC	390	106.3	3 64E-07	0 00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	

NYANZA SITE CONTAMINANTS	CANCER RISK	2.7E-08	1.9E-06	3.3E-11	2 2E-08
OTHER SUDBURY RIVER CONTAMINANTS	CANCER RISK	1.8E-05	1.2E-05	3.1E-06	2.2E-06
ALL CHEMICALS OF CONCERN	CANCER RISK	2.1E-05	1.4E-05	3.1E-08	2.2E-08

BEHP:BIS(2-ETHYL HEXYL)PHTHALATE

6-94

(

TA 6-22A RISK ASSESSMENT RESULTS FOR SURFACE WATER EXPOSURE SCENARIOS OUTFALL OREEK NYANZ A OPERABLE UNIT 3 MODLESEX COUNTY, MASSACHUSETTS

CONTAMINANTS of CONCERN	CONCENTRY MGV	kg Kg	EDEPOSURE RECEPTOR	FACTOR	RFD (MG - KG -	HAZAF		ENTS: CHI	ம	HAZAR	id quotien	ITS: TEEN		HAZARC		is: Adult		TOXICI END- POIN
	MAX	AVG	INGESTION	DERMAL CONTACT	DAY)	ACCIDEN	TÁL DN	DERM	AL VCT	ACCIDENT	AL N	DEFIMA	L ST	ACCIDENT	NL	DERMA	ī	
	(MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	
TRICHLOROE THENE*	0013	0013	1 11E-03	2 10E - 03		0 0E+00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	NS
1.2-DICHLOROETHENE*	0 0 12	0 012	1 11E-03	2 05E - 03	1 00E - 02	1 1E-03	1 1E-03	1.1E-03	1 1E-03	13E-03	1.3E-03	2.5E-03	2.5E-03	2.3E-04	2.3E-04	1.4E-03	1.4E-03	BLOCO
1.4-DICHLOROBENZENE	0 001	0 001	1 11E-03	1 12E - 02	1	0 0E+00	0 0E+00	0 0E+00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0 0E + 00	0.0E+00	0.0E+00	LIVER
1.2-DICHLOROBENZENE	0 004	0 004	1 11E-03	1 12E - 02	9 00E - 02	4 1E-05	4 1E-05	2 3E - 04	2 3E - 04	49E-05	4.9E~05	5.0E-04	5.0E-04	8.7E-06	6.7E-08	2.8E-04	2.8E-04	UVER
CHROMIUM*	0 0056	0 0056	1 11E-03	1 38E - 04	1 00E+00	51E-08	51E-08	3 6E - 07	36E-07	8.2E-08	82E-08	7.7E-07	7.7E-07	1.1E-08	1.1E-08	4.3E - 07	4.3E-07	UVER
LEAD*	0 0067	0 0067	1 11E-03	1 38E - 04	¦	00E+00	0 0E + 00	0 0E + 00	0 0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	CNS
MERCURY	0 00048	0 0005	1 11E-03	1 38E - 04	3 00E - 04	15E-03	1 5E - 03	10E+04	1.0E-04	1.8E-03	1.8E-03	2.2E-04	2.2E-04	3.1E-04	3.1E-04	1.2E+04	1.2E-04	CNS
BEHP	0 001	0 00 1	1 11E-03	1 10E-04	2 00E - 02	4 6E-05	4 6E - 05	2 5E - 08	2 5E - 06	5.6E-05	5 6E - 05	5.5E-08	5.5E-08	9.8E - 08	9.8E-06	3.0E-08	3.0E-06	UVER
BARIUM	0 0105	0 0 105	1 11E-03	1 38E-04	5 00E - 02	19E-04	19E-04	1 3E - 05	1.3E-05	2 3E-04	2.3E-04	2.9E-05	2.9E-05	4.1E-05	4.1E-05	1.6E+05	1.6E-05	BLOOD
MANGANESE	0 14	0 14	1 11E-03	1 385 - 04	1 00E-01	13E-03	1 3E - 03	89E-05	8.9E-05	1.6E-03	1.6E-03	1.9E-04	1.9E-04	2.7E-04	2.7E-04	1.1E-04	1.1E-04	CNS
ZINC	0.0479	0 0479	1 11E-03	1 38E-04	2 00E -01	2.2E-04	2 2E-04	1.5E-05	1.5E-05	2.7E-04	2.7E-04	3.3E-06	3.3E-05	4.7E-05	4.7E-05	1.8E-05	1.8E-05	BLOOD
NYANZA SITE CONTAMIN	ANTS	-		HAZANO IN))))	2 0E - 03	2 6E - 03	15E-03	15E-03	3.2E-03	3.2E-03	3.2E-03	3.2E-03	5.6E-04	5.6E-04	1.6E-03	1.6E-03	1
OTHER SUDBURY RIVER	CONTAMINA	179		HAZARD IN	DEX	1 7E - 03	1 7E - 03	1 2E-04	1.2E-04	2.1E-03	2.1E-03	2.6E-04	2.6E-04	3.7E-04	3.7E-04	1.4E-04	1.4E-04	
ALL CHEMICALS OF CON	CERN			HAZARD IN	DEX	4.3E-03	4 3E-03	1.0E-03	1.0E-03	5.3E - 03	5.3E-03	3.4E-03	3.4E-03	9.3E-04	9.3E-04	1.96-03	1.9E-03	

BEPH: BIS(2ETHYL HEXYL)PHTHALATE

TABLE 6-22B RISK RESULTS FOR SURFACE WATER OUTFALL CREEK NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

					CARCINO	SENIC RISI	KRESULTS	3	
CONTAMINANTS of CONCERN	CONCENTRA MG/	ATION KG	EXPOSURE	FACTOR		CANCER	RISKS		CANCER SLOPE FACTOR
	MAX	AVG	INGESTION	DERMAL	ACCIDEN				(MG/KG/D) 1
	l				MAX	AVG	MAX	AVG	/ WEIGHT
						110	IVICUA	710	OF
									EVIDENCE
TRICHLOROETHENE*	0.013	0.013	3.02E-04	6.61E-04	4.3E-08	4.3E-08	9.5E-08	9.5E-08	1.10E-02 B2
1,2-DICHLOROETHENE*	0.012	0.012	3.02E-04	6.28E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
1,4-DICHLOROBENZENE	0.001	0.001	3.02E-04	3.43E-03	7.3E-09	7.3E-09	8.2E-08	8.2E-08	2.40E-02 B2
1,2-DICHLOROBENZENE	0.004	0.004	3.02E04	3.43E-03	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
CHROMIUM*	0.0056	0.0056	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
LEAD*	0.0057	0.0057	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MERCURY*	0.00048	0.0005	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
BEHP	0.001	0.001	3.02E-04	3.37E-05	4.2E-09	4.2E-09	4.7E-10	4.7E-10	1.40E-02 B2
BARIUM	0.0105	0.0105	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MANGANESE	0.14	0.14	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+.00	0.0E+00	
ZINC	0.0479	0.0479	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	

NYANZA SITE CONTAMINANTS*	CANCER RISK	5.0E-08	5.0E-08	1.8E-07	1.8E-07
OTHER SUDBURY RIVER CONTAMINANTS	CANCER RISK	4.2E-09	4.2E-09	4.7E-10	4.7E-10
ALL CHEMICALS OF CONCERN	CANCER RISK	5.5E-08	5.5E-08	1.8E-07	1.8E-07

BEPH: BIS(2ETHYL HEXYL)PHTHALATE

96-96

FINAL

Arsenic, beryllium, and the carcinogenic PAHs are the principal contaminants contributing to the excess lifetime cancer risks estimated for the sediment exposure scenarios. Cancer risks estimated for "other Sudbury River contaminants" not related to Site discharges exceed those estimated for the Nyanza Site contaminants. Estimated cancer risks range from 2.1×10^{-5} (accidental-ingestion, reasonable worst-case scenario) to 2.2 x 10^{6} (dermal-contact, average-case scenario). As a point of reference, cancer risks estimated for background sediment concentrations were 1.8 x 10⁻⁵ and 8.2 x 10⁻⁶ for the reasonable maximum and average case, respectively. Cancer risks estimated for the surface water exposure scenarios do not exceed 1×10^{-6} in any case presented on Table 6-22. Cancer risks estimated for COC concentrations in Outfall Creek would not exceed 1 x 10^4 even if the risks associated

6.5.3.7 Risk Assessment Results for the Raceway

associated with the sediment exposure scenarios.

exposure scenarios defined in Section 6.4.

The Raceway is fed by Outfall Creek and, in turn, discharges to the Sudbury River. The following organic and inorganic Nyanza Site contaminants were detected in sediment and/or surface water samples collected from this tributary to the Sudbury River:

with the surface water exposure scenarios were summed with those

0	Trichloroethene	0	1,2-Dichloroethene
0	Chlorobenzene	0	1,2-Dichlorobenzene
0	Naphthalene	0	Arsenic
0	Cadmium	0	Chromium
0	Lead	0	Mercury

No COCs were detected in surface water samples at concentrations in excess of Federal SDWA Primary MCLs. The average mercury concentration detected in the sediments ($C_{reg}=0.71 \text{ mg/kg}$) is approximately three times the concentration detected in background samples ($C_{reg}=0.27 \text{ mg/kg}$).

Tables 6-23 and 6-24 present risk assessment results for COC concentrations detected in sediments and surface waters, respectively. The hazard quotients and hazard indices calculated do not exceed unity in any of the cases presented. Mercury is not the predominant contaminant contributing to the estimated noncarcinogenic risk. The hazard quotient calculated for one compound only, arsenic, exceeds 0.1 (accidental-ingestion scenario, child and teen receptors). If hazard indices are summed for the

W92194F

TABLE 6~23A RISK ASSESSMENT RESULTS FOR SEDIMENT EXPOSURE SCENARIOS RACEWAY NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

							NONCARC	NOGENC	RIBK ANA	LYDIS RESI	JLTS							
CONTAMINANT8 of CONCERN	CONCEN	TRATION KG	EXPOSURE RECEPTOR	FACTOR	PIFD	HAZ		TENTS CH	4LD	HAZ	ARD QUOT	NENTS: TE	EN	HAZARI	QUOTIENT	S: ADULT		TOXIC END - POINT
	MAX	AVG	INGESTION		(MG - KG -	ACCIE		DE		ACCIDEN		DERM		ACCID	ENTAL	DERN		
					DAY	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	1
TRICHLOROETHENE*	44	17 071	111E-08	2 78E - 08		0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E + 00	0.0E + 00	0.0E + 00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	0.0E+00	NS
1,2-DICHLOROETHENE*	13	0 567	1 11E-08	2 78E - 08	1 00E - 02	2 4E - 04	1 OE - 04	3 0E - 04	1 3E - 04	1.4E-04	6.3E~05	3.6E - 04	1.8E-04	1.3E05	5.5E-06	6.4E-05	2.8E-05	BLOOD
CHLOROBENZENE*	19.5	5 8175	1 11E-00	278E-08	2 00E - 02	1 8E - 03	5 4E ~ 04	2.2E - 03	8 8E - 04	1 1E 03	3.3E04	2.7E - 03	8.2E-04	9.5E - 05	2.9E - 05	4.8E-04	1.4E-04	UVER
1,2 - DICHLOROBENZENE	71	4 3	111E-06	8 55E ~ 07	00E-02	1 4E - 04	8 7E - 05	3 8E - 05	2 2E - 05	8 8E - 05	5.3E-05	4.4E - 05	2.7E+05	7.7E-08	4.7E-08	7.7E-06	4.7E-08	UVER
NAPHTHALENE*	67	4 0813	1 11E-08	2 78E - 07	4 00E - 03	3 1E - 03	1 9E - 03	3 6E - 04	2 3E-04	1 9E - 03	1.1E-03	4.7E-04	2.8E-04	1.6E-04	1.0E-04	8 2E ~05	5.0E-05	BW
ARSENIC*	37 2	24 96	1 11E-08	0 00E + 00	3 00E - 04	2 3E -01	1.5E - 01	0 0E + 00	0 0E + 00	1.4E-01	9.2E - 02	0 0E + 00	0.0E+00	1.2E-02	6.1E-03	0.0E + 00	0.0E + 00	SKIN
CADMIUM*	58	2 28	1 11E-08	0 00E + 00	5 00E 04	2 OE - 02	8.3E-03	0 0E + 00	0 0E + 00	1 2E 02	5.1E-03	0.0E + 00	0.0E + 00	1.1E-03	4.5E-04	0.0E+00	0.0E + 00	KIDNEY
CHROMIUM"	208	155 5	1 11E-08	0 00E + 00	1 00E + 00	3 8E - 04	2 8E ~ 04	0 0E + 00	0 0E + 00	2.3E-04	1.7E-04	0.0E + 00	0.0E+00	2.0E-05	1.5E-05	0.0E + 00	0.0E + 00	LIVER
LEAD*	750	435 75	5 55E - 07	0 00E + 00		0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E + 00	0.0E + 00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E + 00	0.0E + 00	CNS
MERCURY*	0 97	0 7 1	1 11E-08	5 55E - 00	3 00E - 04	5 9E 03	4.3E 03	1.5E-04	1 1E-04	3 6E 03	2.8 E - 03	1.8E ~04	1.3E-04	3.2E - 04	2.3E-04	3.2E - 05	2.3E - 05	CNS
DICHLOROMETHANE	0 062	0 082	1 11E-08	2 78E - 08	00E - 02	1 9E - 06	1 9E - 08	2.4E - 06	2 4E - 08	1 1E-06	1 1E-08	2.9E - 06	2.9E~06	1.0E-07	1.0E-07	5.1E-07	5.1E-07	UVER
ACETONE	076	0 485	1 11E-08	2 78E - 08	1 00E - 01	1 4E - 05	6 9E - 06	1.7E - 05	1.1E-05	84E-08	5.4E - 08	2.1E - 05	1.3E-05	7.4E-07	4.7E-07	3.7E - 06	2.4E-06	L/K
BEHP	25	2 5	111E-08	5 55E - 07	\$ 00E - 02	2 3E 04	2 3E - 04	5.7E – 05	5.7E-05	1.4E-04	1.4E+04	6.9E - 05	6.9E~05	1.2E-05	1.2E - 05	1.2E – 05	1.2E – 05	LIVER
PHENANTHRENE	10	12 113	1 11E-08	2 70E -07	4 00E - 03	7 3E - 03	5.5E - 03	9 1E - 04	6 9E - 04	4.4E-03	3.4E – 03	1.1E-03	8.4E-04	3.9E - 04	3.0E – 04	2.0E - 04	1.5E-04	BW
FWORANTHENE	20	14 203	1 11E-00	2 78E -07	4 00E - 02	• 1E-04	6 5E - 04	1 1E-04	8 1E-05	5.6E-04	4.0E-04	1.4E-04	9.9E-05	4.9E - 05	3.5E-05	2.4E - 05	1.7E-05	L/K
PYRENE	13	• 4	1 11E-08	2 78E - 07	3 00E - 02	7 9E - 04	57E-04	9 9E - 05	7 2E - 05	4 BE 04	3.5E-04	1.2E-04	8.7E-05	4.2E-05	3.1E-05	2.1E-05	1.5E-05	KIDNEY
BENZO(A)ANTHRACENE	11	8713	1 11E-08	278E-07		0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E + 00	0.0E + 00	0.0E+00	0.0E + 00	0.0E + 00	0.0E+00	0.0E+00	NS
CHRYSENE		5 975	1 11E - 08	2 70E - 07		0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E + 00	0.0E+00	0 0E + 00	0.0E + 00	0.0E + 00	0.0E+00	0.0E + 00	0.0E + 00	NS
BENZO(B)FLUORANTHEN	E 52	4 203	1 11E-00	2 70E -07		0 0E + 00	0.0E + 00	0.0E + 00	0.0E + 00	0.0E + 00	0.0E + 00	0.0E + 00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	NS
BENZO(K)FLUORANTHEN	E 03	4 3	1 11E-08	2 78E -07		0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	NS
BENZO(A)PYRENE	47	3 686	1 11E - 08	278E-07		0.0E+00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E +00	0.0E + 00	NS
BARIUM	721	44 1	1 11E-00	0 000 + 00	9 00E -02	2 66 -03	165-03	0.0E+00	0.0E+00	1.0E-03	9.0E-04	0.0E+00	0.0E+00	1.4E-04	8.6E-05	0.0E+00	0.0E+00	BLOOD
BERYLUUM	10 15	7 08	111E-00	0.000 + 00	5 00E -03	37E-03	2 65 -03	0 0E + 00	0 0E + 00	2 3E - 03	10E-03	0.0E+00	0.0E+00	2.0E-04	1.4E-04	0.0E+00	0.0E+00	NS
COPPER	306	217 75	11112-00	0 000 + 00		0 0E + 00	0.05+00	0.000 + 00	0 0E + 00	0.0E+00	0.06+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	NS
MANGANEBE		436 23	1112-00	0 000 + 00	1000 -01	1 86 - 02	8.3E -03	0.05+00	0.05+00	9 GE U3	5.1E-03	0.0E+00	0.0E+00	B.3E-04	4.5E - 04	0.0E+00	0.0E+00	CNS
NUCKEL			1112-00	0 002 + 00	2 000 - 02	176-02	126-03	0.000	0.05+00	1.02 -02	0.0E~03	0.02+00	0.00 + 00	9.1E-04	4.46-04	0.00 + 00	0.000	8.
TINC		-	1116-00	0.002 + 00	1 00E -01	1 3E -02	3 35-03	0.05 + 00	0.05.00	B.1E-03	205 03	0.02 +00	0.00 +00	7.20-04	1.00-04	0.00+00	0.05+00	NS NO
	0 0 0 1 2	0.0106	3 335 -07	9 785 - 07	2002-01	0.05.00	0.0E + 00	0.05+00	0.02+00	3.0E-03	2.00-03	0.02+00	0.00+00	2.72-04	0.00 + 00	0.02 +00	0.00+00	Ne
4,4-000	0.0212	0.000	3 3 3 5 - 07	2 700 -07		0.05+00	0.00 + 00	0.02 +00	0.05+00	0.00 + 00	0.00 + 00	0.02+00	0.00+00	0.02 +00	0.02+00	0.05+00	0.02+00	NO
AROCLOR 1254	0.000	0 0 32	335-07	A 00E + 00		0.05+00	0.00 +00	0.05+00	0.05+00	0.02+00	0.00 + 00	0.02+00	0.02+00	0.02+00	0.00 + 00	0.02 +00	0.02+00	NB
INNULUN 1234			3 352 -07					5 0E 7 00	000 700	0.05 7 00	0.02+00	5.0E + 00	U.UE + UU	0.02700	J.02 + 00	0.02 700	0.02 + 00	113
NYANZA SITE CONTAMINA	NTS-			HAZARD IN	DEX	2 0E -01	1.7E-01	3.1E-03	1 2E - 03	1 8E - 01	1.0E-01	3.8E-03	1.4E-03	1.4E-02	9.0E-03	8.6E-04	2 5E - 04	
OTHER SUDBURY RIVER C	ONTAMIN	ANTS			DEX	e 7E - 02	4.3E-02	1.2E-03	0.1E-04	4.1E-02	2.6E-02	1.5E - 03	1.1E-03	3.6E - 03	2.3E - 03	2.8E~04	2.0E-04	
ALL CHEMICALS OF CONC	ZEAN			HAZARO IN	DEX	3.3E-01	2.1E-01	4 3E - 03	2.1E-03	2.0E - 01	1.3E-01	5.2E 03	2.5E-03	1.7E-02	1.1E-02	9.2E <u>~04</u>	4.5E - 04	

NS: NOT SPECIFIED L/K: UVER AND KIDNEY

TOXICITY ENDPOINTS ASSREVIATIONS

BEHP:BIS(2-ETHYL HEXYL)PHTHALATE

W92194F

86-9

FINAL

BW: BODY WEIGHT CNS: CENTRAL NERVOUS SYSTEM

TABLE 6~238 RISK RESULTS FOR SEDIMENT EXPOSURE RACEWAY NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

	-	-		CARCINOG	ENIC RISK A	NALYSIS	RESULTS		
CONTAMINANTS of	CONCEN	TRATION	EXPOSURE	FACTOR		CANCER	AISKS		CANCER SLOPE FACTOR
CONCERN	мах	AVG	INGESTION	DERMAL CONTACT	ACCIDEN	TAL DN	DERM CONTA	AL	(MG/KG/D)-1 / WEIGHT
					MAX	AVG	MAX	AVG	OF EVIDENCE
TRICHLOROETHENE*	44	17 071	3.84E-07	7.56E-07	1.8E-07	6.8E-08	37E-07	1.4E-07	1.10E-02 B2
1.2-DICHLOROETHENE*	13	0 567	3 64E - 07	7.56E-07	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	
CHLOROBENZENE*	19 5	5 9175	3 64E - 07	7.56E-07	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	
1,2~DICHLOROBENZENE	71	4 3	3 64E - 07	1.51E-07	0.0E + 00	0.0E+00	0.0E + 00	0.0E + 00	
NAPHTHALENE*	67	4 0813	364E-07	7.56E-08	0.0E+00	0 0E + 00	0 0E + 00	0.0E+00	
ARSENIC	37 2	24 96	3 64E - 07	0 00E + 00	2 4E - 05	1.6E - 05	0.0E+00	0.0E+00	1.80E+00 A
CADMIUM*	56	2 28	3 64E - 07	0.00E+00	0 0E + 00	0.0E+00	0.0E + 00	0.0E + 00	
CHROMIUM*	208	155 5	3 64E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0.0E + 00	0.0E + 00	
LEAD*	758	435 75	1 62E - 07	0 00E + 00	0 0E + 00	0.0E+00	0.0E + 00	0.0E+00	
MERCURY	0 97	0 71	3 64E - 07	1.51E-08	0 0E + 00	0.0E + 00	0 0E + 00	0 0E + 00	
DICHLOROMETHANE	0 062	0 062	3 64E-07	7 56E-07	1 7E - 10	1.7E-10	3 5E - 10	3.5E - 10	7.50E-03 82
ACETONE	0 76	0 485	3 64E - 07	7 58E-07	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	
BEHP	25	25	3 64E - 07	1.51E-07	1.3E - 08	1.3E-08	5.3E - 09	5.3E-09	1.40E-02 B2
PHENANTHRENE	18	12 113	3 64E - 07	7.56E - 08	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	
FLUORANTHENE	20	14 263	3 64E - 07	7 58E - 08	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	
PYRENE	13	94	3 64E - 07	7 56E - 08	0 0E + 00	0.0E + 00	0 0E + 00	0.0E+00	
BENZO(A)ANTHRACENE	11	5 7 13	3 64E - 07	7 56E-08	2 3E - 05	1 2E ~ 05	4.8E - 06	2.5E-06	5.80E+00 B2
CHRYSENE	86	5 975	3 64E - 07	7.56E-08	1.8E - 05	1.3E ~ 05	3 8E - 06	2.6E~06	5.80E+00 B2
BENZO(B)FLUORANTHEN	52	4 263	3 64E - 07	7 56E-08	1 1E-05	9 OE ~ 08	2.3E - 06	1.9E-06	5.60E+00 B2
BENZO(K)FLUORANTHEN	63	4 3	3 64E - 07	7 56E - 08	1 3E - 05	9.1E-06	2 6E - 08	1.9E-06	5.80E + 00 B2
BENZO(A) PYRENE	47	3 688	3 64E - 07	7 56E - 08	9 9E - 08	8 2E - 08	2.1E-08	1.7E-08	5.80E + 00 B2
BARIUM	72 1	44 1	3 84E - 07	0 00E + 00	0 0E + 00	0.0E+00	0 0E + 00	0.0E+00	
BERYLLIUM	10 15	7 06 (3 64E - 07	0 00E + 00	1 6E-05	1.1E~05	0.0E+00	0 0E + 00	4.30E+00 A
COPPER	306	217 75	3 64E - 07	0 00E + 00	0 0E + 00	0.0E+00	0.0E + 00	0.0E+00	
MANGANESE	864	456 25	3 64E - 07	0.00E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	
NICKEL	186	89 89	3 64E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E + 00	
VANADIUM	51 2	44 5	3 64E - 07	0 00E + 00	0 0E + 00	0.0E+00	0 0E + 00	0.0E + 00	
ZINC	542	365 88	3 84E - 07	0 00E+00	0 0E + 00	0.0E+00	0 0E + 00	0.0E+00	
4,4~DDE	0 0212	0 0 105	1 09E - 07	7.56E-08	7 9E - 10	1.3E~09	5.4E-10	2.7E-10	3.40E-01 B2
4,4-DDD	0 106	0 035	1 09E - 07	7 56E - 08	2 8E-09	6.4E~10	2.0E - 09	5.8E-10	2.40E-01 B2
AROCLOR 1254	0 589	0,199	1.09E-07	0.00E+00	4.9E-07	1.7E-07	0.0E+00	1.2E-07	7.70E+00 B2

NYANZA SITE CONTAMINANTS*		2.5E-05	1.6E-05	37E-07	1.4E-07
OTHER SUDBURY RIVER CONTAMINANTS	CANCER RISK	9 2E - 05	6 2E - 05	1 6E - 05	1.1E-05
ALL CHEMICALS OF CONCERN	CANCER RISK	1.2E-04	7.9E-05	1.6E - 05	1. <u>1E-05</u>

BEHP:BIS(2-ETHYL HEXYL)PHTHALATE

TABLE 0-24A RISK ASSESSMENT RESULTS FOR SURFACE WATER EXPOSURE SCENARIOS RACEWAY NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

CONTAMINANTS of	CONCENTR	ATION KB		FACTOR	₩0- ₩0-	HATA			RISK ANAL	YSIS RESUL		IS TEEN		HA7AST				TOXICITY END POINT
	WAX	AVG	INGESTION		DAY	ACCIDEN		DERM		ACCIDENT	TAL N	DERMA	L भ	ACCIDENT	AL N		T	
						MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	}
TRICHLOROE THENE • 1,2 - DICHLOROE THENE •	0 002 0 002	0 005	1 11E-03 1 11E-03	2 10E - 03 2 05E - 03	1 00E - 02	0 0E+00 1 8E-04	0 0E+00 1 6E-04	0 0E+00 1 9E-04	0 0E+00 1.9E-04	0.0E+00 2.2E-04	0.0E+00 2.2E-04	0.0E+00 4.1E-04	0.0E+00 4.1E-04	0.0E+00 3.9E-05	0.0E+00 3 9E-05	0.0E+00 2 3E-04	0.0E+00 2.3E-04	NS BLOOD
BARIUM MANGANESE	0 0 164	0 0 164	1 11E-03 1 11E-03	1 38E - 04 1 38E - 04	5 00E - 02 1 00E - 01	3 0E - 04 2 4E - 03	3 0E - 04 2 4E - 03	2 1E-05 1 7E-04	2 1E-05 1 7E-04	3 6E - 04 2.9E - 03	3 6E - 04 2.9E - 03	4.5E-05 3.6E-04	4.5E-05 3.6E-04	6.4E05 5.2E04	6.4E - 05 5.2E - 04	2.5E-05 2.0E-04	2.5E-05 2.0E-04	BLOOD CNS
NYANZA SITE CONTAMIN	ÂN 19			HAZARD IN	bēji	1 BE-04	1 BE - 04	19E-04	19E-04	2.2E-04	2.2E-04	4.1E-04	4.1E-04	3.9E-05	3 9E - 05	2.3E-04	2.3E-04	1
OTHER SUDBURY RIVER		N TS			DEX	27E-03	2 7E - 03	1 9E-04	1 9E-04	3 3E - 03	3 3E - 03	4 1E-04	4.1E-04	5 8E - 04	5.8E - 04	2 2E - 04	2 2E - 04	
ALL CHEMICALS OF CON	CERN	- –		HAZARD IN	DEX	2 9E - 03	2 9E - 03	3 6E - 04	3 8E - 04	3.5E-03	3.5E-03	8.2E-04	8.2E-04	6.2E-04	6.2E-04	4.5E-04	4.5E-04]
BEPH: BIS(2ETHYL HEXAL	PHTHALATE				NOPOINTS A	BOREVIATIO	NS [.]	NS NOT S	PECIFIED	L/K: LIV OUS SYSTE	er and Kid M	NEY						

t

W92194F

FINAL

.

TABLE 6-24B RISK RESULTS FOR SURFACE WATER RACEWAY NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

					CARCINO	GENIC RIS	K RESULTS	5	
CONTAMINANTS of CONCERN	CONCENTRA MG/	ATION KG	EXPOSURE	FACTOR		CANCEF	RISKS		CANCER SLOPE FACTOR
	MAX	AVG	INGESTION	DERMAL CONTACT	ACCIDEN	TAL DN			(MG/KG/D)-1
					MAX	AVG	MAX	AVG	WEIGHT OF
									EVIDENCE
TRICHLOROETHENE*	0.002	0.002	3.02E-04	6.61E-04	6.6E-09	6.6E-09	1.5E-08	1.5E-08	1.10E-02 B2
1,2-DICHLOROETHENE*	0.002	0.002	3.02E-04	6.28E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
BARIUM	0.0164	0.0164	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MANGANESE	0.2635	0.2635	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
NYANZA SITE CONTAMIN	ANTS*		CANCER RI	SK	6.6E-09	6.6E-09	1.5E-08	1.5E-08	
OTHER SUDBURY RIVER	CONTAMINAN	ITS		SK	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ALL CHEMICALS OF CON	CERN		CANCER RI	SK	6.6E-09	6.6E-09	1.5E-08	1.5E-08	

BEPH: BIS(2ETHYL HEXYL)PHTHALATE

W92194F

accidental-ingestion and dermal-contact exposure routes for each medium and then combined for surface water and sediment exposures, the total hazard index does not exceed unity. Based on these results, adverse noncarcinogenic health effects are not anticipated under the conditions of the recreational exposure scenarios defined in Section 6.4.

The excess lifetime cancer risk estimated for the surface water and sediment exposures slightly exceed the 1E-04 and 9E-05 cancer risk levels (risks summed for the accidental-ingestion and dermal-contact exposure routes) when maximum and average contaminant concentrations are evaluated, respectively. Cancer risk levels associated with surface water exposures do not exceed 1 x 10°. As a point of reference, cancer risks estimated for COC concentrations detected in background surface waters and sediments were 2E-05 and 9.7E-06 for the reasonable maximum and average case, respectively. Thus, risks summed for Raceway surface waters and sediments exposures are an order of magnitude higher than background risks. The principal COCs contributing to estimated risk are the carcinogenic PAHs, the PCB Aroclor-1254, arsenic, beryllium and trichloroethene. Although arsenic and trichloroethene are Nyanza Site contaminants, the total cancer risk estimated for "other Sudbury River COCs" exceeds that estimated for all Nyanza Site The total cancer risk estimated for Nyanza Site contaminants. contaminants alone was approximately 2.5E-05 and 1.6E-07 when maximum and average contaminant concentrations were evaluated, respectively. Cancer risks estimated for the surface water exposure scenarios are minimal when compared to those estimated for sediments.

6.5.3.8 <u>Risk Assessment Results for Cold Spring Brook</u>

Cold Spring Brook is a tributary of the Sudbury River which is fed by discharges from the Cold Spring Culvert. Neither surface water body receives surface runoff from the Nyanza Site. Contaminant concentrations in surface water and sediment samples collected from the Brook are minimal compared to concentrations detected in areas that are obviously impacted by Nyanza Site contamination such as the Eastern Wetlands. No Nyanza Site contaminants were detected in surface water samples collected from the Brook. Arsenic, chromium, and lead were the only Nyanza Site contaminants detected in sediment samples. Mercury was not detected in environmental samples collected from the Brook.

Tables 6-25 and 6-26 present risk assessment results for COC concentrations detected in sediments and surface waters, respectively. The accidental-ingestion and dermal-contact routes

W92194F

6-102

TABLE 8-25A RISK ASSESSMENT RESULTS FOR SEDIMENT EXPOSURE SCENARIOS COLD SPRING BROOK NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

7

							NONCARC	NOGENC	RISK ANA	LYSIS RESL								
CONTAMINANTS of CONCERN		TRATION KG	EXPOSURE RECEPTOR	FACTOR	PIF D	HAZ	ARD QUO	DENTS C	HILD	HAZ		NENTS: TE	EN	HAZAR		S: ADULT		-TOXIC END- POINT
	MAX	AVG	INGESTION	DERMAL CONTACT	(MG - KG -		DENTAL	DE CO	RMAL NTACT	ACCIDENT		DERM	AL ACT		DENTAL	DERI		
	L.			I	DAY	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	
ARSENIC	43	35	1 11E-08	0 00E + 00	3 00E - 04	2 BE - 02	2 1E -02	0 0E + 00	0 0E + 00	1 6E - 02	1.3E-02	0.0E + 00	0.0E+00	1.4E-03	1.1E-03	0.0E + 00	0.0E+00	SKIN
CHROMIUM*	187	15 15	1 11E-00	0 00E + 00	1 00E + 00	3 1E - 05	2 BE - 05	0 0E + 00	0.0E + 00	1.9E - 05	1.7E-05	0.0E + 00	0.0E+00	1.6E-08	1.5E-06	0.0E + 00	0.0E + 00	UVER
LEAD*	358	234	8 55E - 07	0 00E + 00		0 0E + 00	0 0E + 00	0 0E + 00	0.0E + 00	0.0E + 00	0.0E + 00	0.0E + 00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E + 00	CNS
BARIUM	e5 5	50	1 11E-00	0 00E + 00	5 00E - 02	24E-03	2 2E - 03	0 0E + 00	0 0E + 00	1.5E-03	1 3E - 03	0.0E + 00	0.0E+00	1.3E-04	1.2E~04	0.0E+00	0.0E+00	BLOOD
BERYLUUM	10	1	1 11E-00	0 00E + 00	5 00E - 03	€ 9 E 04	3.7E - 04	0 0E + 00	0 0E + 00	4 2E - 04	2.2E-04	0.0E + 00	0.0E+00	3.7E-05	2.0E - 05	0.0E+00	0.0E+00	NS
COPPER	32 🖲	31 5	1 11E-00	0 00E + 00		0 0E + 00	0.06+00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	NS
MANGANESE	854	656	111E-0	0.00E+00	1 00E - 01	106-02	126-02	0.02+00	0.0E+00	9.5E-03	7.3E-03	0.0E+00	0.0E+00	8.4E-U4	8.4E~U4	0.0E+00	0.0E+00	CNS
VANADIUM	414	30 0	1 TTE - 00	0.000 + 00	7 00E - 03	1 TE - 02	8.1E-UJ	0.05+00	0.000	0 00 -03	4.92-03	0.00 + 00	0.000	0.8E-04	4.35~04	0.05+00	0.00 + 00	NS NOOD
<u>ZINC</u>	! •		['''E-00		1006-01	1 56 - 03	1.46 - 03	002400	0027001		0.20-04	0.02 + 00	U.DE Y OU	8.12-03	, <u>22</u> - 03	0.02 + 00	0.02 + 00	82000
NYANZA SITE CONTAMIN	NTB			HAZARO IN	XEX .	2 8E - 02	2 1E - 02	0 0E + 00	0 0E + 00	1.8E-02	1.3E-02	0.0E + 00	0.0E+00 *	1.4E-03	1.1E-03	0 0E+00	0.0E+00	
OTHER SUDBURY RIVER	ONTAMIN	ANTS		HAZARD IN	X X	3 1E - 02	2 4E - 02	0.0E+00	0 0E + 00 '	1.9E-02	1.5E - 02	0.0E+00	0.0E+00 *	1.7E-03	1.3E~03	0.0E + 00	0.0E + 00	
ALL CHEMICALS OF CON	ZERN			HAZARD IN	ж×	57E-02	4 5E - 02	0.0E + 00	0 0E + 00	3.5E - 02	2.8E-02	0.0E + 00	0.0E+00	3.1E-03	2.4E-03	0.0E+00	0.0E+00	

TOXICITY ENDPOINTS ABBREVIATIONS NS NOT SPECIFIED L/K: LIVER AND KIDNEY BW: BODY WEIGHT CNS: CENTRAL NERVOUS SYSTEM

6-103

TABLE 6–25B RISK RESULTS FOR SEDIMENT EXPOSURE COLD SPRING BROOK NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

CONTAMINANTS of	CONCEN MG/I	TRATION	EXPOSURE	CARCINOGE FACTOR	ENIC RISK A	NALYSIS F	RESULTS		CANCER SLOPE FACTOR
CONCERN	мах	AVG	INGESTION	DERMAL CONTACT	ACCIDEN INGESTIC	TAL DN	DERM CONTA	AL ACT	(MG/KG/D)-1 / WEIGHT
					MAX	AVG	MAX	AVG	EVIDENCE
ARSENIC* CHROMIUM* LEAD*	4.3 16.7 328	3.5 15.15 234	3.64E-07 3.64E-07 1.82E-07	0.00E+00 0.00E+00 0.00E+00	2.8E-06 0.0E+00 0.0E+00	2.3E-06 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00	1.80E+00 A
BARIUM BERYLLIUM COPPER MANGANESE VANADIUM	65.5 1.9 32.9 854 41.4	59 1 31.5 658 30.9	3.64E-07 3.64E-07 3.64E-07 3.64E-07 3.64E-07 3.64E-07	0.00E+00 0.00E+00 0.00E+00 0.00E+00 0.00E+00	0.0E+00 3.0E-06 0.0E+00 0.0E+00 0.0E+00	0.0E+00 1.6E-06 0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	4.30E+00 A

NYANZA SITE CONTAMINANTS	CANCER RISK	2.8E-06	2.3E-06	0.0E+00	0.0E+00 *
OTHER SUDBURY RIVER CONTAMINANTS	CANCER RISK	3.0E-06	1.6E-06	0.0E+00	0.0E+00 *
ALL CHEMICALS OF CONCERN	CANCER RISK	5.8E-06	3.9E-06	0.0E+00	0.0E+00

FINAL

6-104

W92194F

TAULE 0-26A RISK ASSESSMENT RESULTS FOR SURFACE WATER EXPOSURE SCENARIOS COLD SPRING BROOK NYANZ A OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

CNS CENTRAL NERVOUS SYSTEM

	1		-				NONCAR	NOGENC	RISK ANAL	YSIS RESUL	TS							
CONTAMINANTB of CONCERN	DONCENTRATION EXPOSU MOV KG RECEPT		EXPOSURE FACTOR		AFD (MG - KG - HAZARD QUOTIENTS: CHILD			ю	HAZARD QUQTIENTS: TEEN				HAZARD QUOTIENTS: ADULT				TOXICITY END- POINT	
	THAN T	AVG	INCESTICK	DEPMAL CONTACT	DAY)	ACCIDEN	TÁL DN	DERM	AL KCT	ACCIDEN	AL N	DERMA	L CT	ACCIDENT/	NL N	DERMA	L 2T	÷
						MAX	DVA	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	
BEHP MANGANESE SILVER	0 058 0 108 0 0139	0 058 0 108 0 0139	1 11E - 03 1 11E - 03 1 11E - 03	1 10E - 04 1 38E - 04 1 38E - 04	2 00E - 02 1 00E - 01 3 00E - 03	2 0E - 03 9 9E - 04 4 2E - 03	2 6E - 03 9 9E - 04 4 2E - 03	1 5E - 04 8 9E - 05 3 0E - 04	1 5E - 04 6 9E - 05 3 0E - 04	3.2E-03 1.2E-03 5.1E-03	3.2E - 03 1.2E - 03 5.1E - 03	3.2E-04 1.5E-04 6.4E-04	3.2E-04 1.5E-04 6.4E-04	5.7E-04 2.1E-04 9.1E-04	5.7E-04 2.1E-04 9.1E-04	1.8E 04 8 2E 05 3.5E 04	1.8E-04 8.2E-05 3.5E-04	UVER CNS ARGYRIA
NYANZA SITE CONTAMIN	ANTS			HAZAPO IN	DEX	0 0E+00	0 0E+00	0 0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
OTHER SUDBURY RIVER		178		HAZARD IN	DEX	7 9E - 03	7.9E-03	5 1E-04	5.1E-04	9.6E-03	9.8E-03	1.1E-03	1.1E-03	1.7E-03	1.7E-03	6.1E-04	6.1E-04	
ALL CHEMICALS OF CON	CERN			HAZARD IN	DEX	7.0E-03	7 9E-00	5 1E-04	5.1E-04	9.0E-03	9.8E-03	1.1E-03	1.1E-03	<u>1.7E - 03</u>	1.7E - 03	8.1E-04	8.1E-04	}
					NDPOINTS A	BEREVIATIO	ONS [.]	NS. NOT S	PECIFIED	L/K: LIV	er and Kid	NEY						

BEPH: BIS (2ETHYL HEXYL)PHTHALATE

W92194F

1

TABLE 6–26B RISK RESULTS FOR SURFACE WATER COLD SPRING BROOK NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

			CARCINOGENIC RISK RESULTS									
CONTAMINANTS of CONCERN	CONCENTRA MG/	TION KG	EXPOSURE	FACTOR		CANCER SLOPE FACTOR						
	MAX	AVG	INGESTION	DERMAL CONTACT	ACCIDEN INGESTIC	TAL DN	DERM CONTA	AL \CT	(MG/KG/D)-1 /			
					MAX	AVG	MAX	AVG	WEIGHT OF EVIDENCE			
BEHP MANGANESE SILVER	0.058 0.108 0.0139	0.058 0.108 0.0139	3.02E-04 3.02E-04 3.02E-04	3.37E-05 4.21E-05 4.21E-05	2.5E-07 0.0E+00 0.0E+00	2.5E-07 0.0E+00 0.0E+00	2.7E-08 0.0E+00 0.0E+00	2.7E-08 0.0E+00 0.0E+00	1.40E–02 B2			
NYANZA SITE CONTAMIN	ANTS*		CANCER R	ISK	0.0E+00	0.0E+00	0.0E+00	0.0E+00]			
OTHER SUDBURY RIVER CONTAMINANTS			CANCER R	ISK	2.5E-07	2.5E-07	2.7E-08	2.7E-08				
ALL CHEMICALS OF CON	CERN		CANCER R	ISK	2.5E-07	2.5E-07	2.7E-08	2.7E-08				

BEPH: BIS(2ETHYL HEXYL)PHTHALATE

FINAL

.

of exposure (surface waters and sediments) were evaluated assuming recreational land/water use scenarios. The hazard quotients and hazard indices calculated for COC concentrations in surface waters and sediments do not exceed unity in any of the cases presented. None of the hazard quotients presented exceed 0.1. Also, if hazard indices are summed for the accidental-ingestion and dermal-contact exposure routes for each medium and then combined for surface water and sediment exposures, the total hazard index does not exceed unity. These results indicate that adverse noncarcinogenic health effects would not be anticipated for a receptor contacting surface waters and sediments under the conditions of recreational exposure scenarios defined in Section 6.4.

The excess lifetime cancer risks estimated for the surface water and sediment exposures are approximately 6E-06 and 4E-06 (risks summed for the accidental-ingestion and dermal-contact exposure routes) when the reasonable maximum-case and average-case scenarios are evaluated, respectively. These cancer risks are less than those estimated for background sample concentrations. Arsenic, beryllium, and bis(2-ethylhexyl)phthalate are the COCs contributing to the estimated excess lifetime cancer risk.

6.5.3.9 Risk Assessment Results for Reach 3

Sudbury River Reach 3 is a primary sedimentation area directly downstream of Reach 2. The Reach ends at Dam No. 2 which divides Reservoirs No. 2 and No. 1. Several Nyanza Site contaminants were detected in Reach 3 surface waters and sediments:

0	Mercury	0	Chromium
0	Methyl mercury	0	Lead
0	Arsenic	0	1,4-Dichlorobenzene
0	Antimony	0	1,2,4-Trichlorobenzene
0	Cadmium	0	Naphthalene

The average concentration of mercury (C_{reg} =15.98 mg/kg) in Reach 3 sediment samples was 60 times the average mercury concentration in background sediment samples. Monomethyl mercury and dimethyl mercury were detected in Reach 3 sediments.

Table 6-27 and 6-28 present risk assessment results for COC concentrations detected in sediments and surface water of Reach 3. The accidental-ingestion and dermal-contact routes of exposure (surface waters and sediments) were evaluated assuming recreational land/water use scenarios. The majority of the noncarcinogenic risk is attributable to Nyanza site-specific COCs. The hazard quotients presented for arsenic and mercury exceed 0.1 in one or more cases

W92194F

TABLE 0-27A RIBK ABBEBSMENT RESULTS FOR SEDIMENT EXPOSURE SCENARIOS REACH NO 3 NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

.....

. .

			<u>, </u>				NONCAR	CINOGENIC	RISK ANA	LYSIS RESL	JLTS			·· ······			······	<u>.</u>
CONTAMINANTS of CONCERN	CONCENTRATION MG/KG MAX AVG		ON EXPOSURE FACTOR		FACTOR RFD		HAZARD QUOTIENTS: CHILD HAZARD QUOTIENTS: TEEN HAZARD QUOTIENTS: ADULT								TOXIC END- POINT			
			INDERTION	DESTICH DEPMAL CONTACT		ACCIDENTAL NGESTION		DEFMAL CONTACT		ACCIDENTAL		DERMAL CONTACT		ACCIDENTAL INGESTION				1.
					DAY	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	1
													0.05.00	0 OF 1 00	0.05.00	0.05.00		
1,4-DICHLOHOBENZENE	01230	01250	1 11E-00	5 356-07		0.000	0 0E+00	0.0E+00	0.0E+00	0.0E+00	1 15 04	775 05	0.0E+00	1.4E 06	0.02+00	1.4E 06	0.0E+00	CIVEN
NADUTHALENE		0 1200	1115-00	9 785 -07	4 005 - 03	8.65-04	8.60-04	0 36-03	# #E-03	3.15-04	1.10-04	A 35-08	8.45-08	2.95-04	205-00	1.55-04	9.4E-00	SKIN
ADDENIC+	21.6	7 41	1116-04	2/00-07	1006-04	135-03	3 30-03		0.05+00	3.3E-03	3.35-03	0.0E+00	0.05+00	2.02-08	2.85-00	0.05+00	1.50-00	OW NI
ANTINONY			1116-00	0.005 .00	4 006 - 04	A #E-07	105-02	0.05+00	0.05+00	505-02	1 16-02	0.00 +00	0.05+00	A 4E-03	105-03	0.05+00	0.00 + 00	
CADMILINA			1116-00		4 000 - 04	7 16 01	1.00 02	0.02+00	0.05+00	A 46 02	1 16-02	0.02+00	0.05+00	3.95-03	0.75-04	0.02 +00	0.02+00	KINEY
CHROMIUM	200	201.02	1115-04	0.000 .00	1005-04	7 JE-02	B 3C 04	0.05+00	0.00 + 00	3.42-02	1.16-02	0.02 +00	0.00 + 00	382-03	205.05	0.02+00	0.02+00	
LEAD4	2020		A 856-07	0.000 + 00	1002700	UJ	0.05-04	0.05+00	0.05+00		0.00-04	0.00 +00			5.0C-03	0.0E+00	0.00 + 00	CNS
MERCURY			1115-00	4 84E_AA	3.006-04	116-01	876_03	■ 1E_01	24F_01	205-01	5 0C + 00	1.05-02	3.0E-03	1.8E_02	5.02 +00	1 AF_03	5.0C+00	CNS
MONOMETHY	0.0200	0.0263	1115-00	4 285 - 08	3006-04	1.65-04	185-04	115-04	2 4E=03	A 7E_05	0.75-05	375-04	3.7E-04	8.8E~08	J.2E-03	8.8E+05	3.2E~04	CNG
DIMETHYLMERCURY*	0 0191	0 0074	1 11E-00	4 20E - 00	3 00E-04	1 2E-04	4 5E-05	2 2E-04	8 7E-05	7.1E-05	2.8E-05	2.7E-04	1.1E-04	6.2E-06	2.4E~00	4.8E-05	1.9E-05	CNS
ACETONE	0 31	0 1312	1 11E-08	2 78E-08	1 00E-01	57E-08	2 4E-08	7.1E-00	3 0E-06	3 4E-08	1.5E-08	8.6E-06	3.6E-06	3.0E-07	1.3E-07	1.5E-06	6.4E-07	LIK
BEHP	ا م	0 7905	1 11E-08	5 65E-07	2 00E - 02	8 3E - 05	7 2E-05	2 1E-05	1 8E-05	5.1E~05	4.4E~05	2.5E-05	2.2E-05	4.5E-08	3.9E-06	4.5E-06	3.9E-08	LIVER
ACENAPTHYLENE	0 80	0 8188	1116-00	2 78E-07	4 00E-03	4 0E - 04	37E-04	5 0E-05	47E-05	2.4E-04	2.3E-04	0.1E05	5.7E-05	2.2E-05	2.0E-05	1.1E-05	1.0E-05	NS
PHENANTHRENE	7 3	1 3878	1 11E-08	2 78E-07	4 00E-03	3 3E-03	6 2E-04	4 2E-04	7.8E-05	2.0E-03	3.8E-04	5.1E-04	9.5E-05	1.8E-04	3.3E~05	8.9E-05	1.7E-05	BW
FL UOBANTHENE	12	2 3317	1 11E-06	2 78E-07	4 00E-02	5 5E-04	1.1E-04	8 8E-05	1.3E-05	3.3E-04	0.5E-05	8.3E-05	1.6E05	2.9E05	5.7E-08	1.5E-05	2.9E-06	lux
PYRENE	1 11	2 4811	1118-06	2 70E-07	3 00E-02	07E-04	1.5E-04	0.4E-05	1.0E-05	4.1E-04	9.1E-05	1.0E-04	2.3E-05	3.0E-05	8.0E-08	1.8E-05	4.0E-08	KIDNEY
BENZO(A)ANTHRACENE	4.5	1 3461	1 11E-00	2 76E-07		0 0E + 00	0 0E+00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	NS
CHRYSENE	77	1 7372	1 11E-00	2 78E-07		0 0E + 00	0 0E + 00	0 0E + 00	0 0E +00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	NS
BENZO(B)FLUORANTHEN	È 44	1 4050	1 11E-00	2 70E-07		0 0E + 00	0.0E+00	0 0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	NS
BENZOKAFLUORANTHEN	Ė 30	1 4942	1 11E-00	2 78E-07		0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	0.0E+00	NS
BENZO(A)PYRENE	44	1 2001	1 11E-00	2 78E-07		0 0E + 00	0 0E + 00	0 0E+00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	INS
INDENO(123-CD)PYRENE	1	0 7858	1 11E-00	2 70E-07		0 0E + 00	0 0E+00	0 0E+00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0 OE + 00	0.0E+00	NS
DIBENZ(AH)ANTHRACENE	0 3 1	0 22	1 11E-00	2 78E-07		0 0E + 00	0 0E + 00	0 0E+00	0 0E + 00	0.0E+00	0 0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	NS
BENZO(GHI)PERYLENE	1.	0 7727	1 11E-06	2 78E-07	4 00E-03	7 3E-04	3.5E-04	9 1E-05	4 4E-05	4.4E-04	2.1E-04	1.1E-04	5.4E-05	3.9E-05	1.9E-05	2.0E-05	9.5E~08	NS
BARIUM	110		1 11E-06	0 00E+00	5 00E-02	4 2E-03	2 0E-03	0 0E+00	0.0E+00	2.6E-03	1.6E-03	0.0E+00	0.0E+00	2.3E04	1.4E-04	0.0E+00	0.0E+00	BLOOD
BERYLLIUM	43	1 62	1 11E-06	0 00E+00	6 00E-03	1 8E - 03	5 6E-04	0 0E+00	0 0E + 00	9.6E-04	3.4E-04	0.0E+00	0.0E+00	8.4E-05	3.0E-05	0.0E + 00	0.0E +00	NS
COPPER	454	184.48	1 11E-00	0 00E + 00		0 0E + 00	0 0E+00	0 0E+00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	N9
MANGANESE	847	301 42	1 11E-00	0 00E + 00	1 00E-01	1.8E-02	7.1E-03	0 0E + 00	0 0E + 00	9.4E-03	4.3E-03	0.0E+00	0.0E+00	8.3E-04	3.8E-04	0.0E+00	0.0E+00	CNS
NICKEL	84 8	20 43	1 11E-06	0 00E+00	2 00E-02	8.1E-03	2.8E-03	0.0E+00	0 0E + 00	4.8E-03	1.8E-03	0.0E+00	0.0E+00	4.3E~04	1.4E-04	0.0E+00	0.0E+00	BW
SELENIUM	•	081	1 11E~08	0 00E + 00	5 00E-03	1 8E-03	3 OE - 04	0 0E + 00	0 0E + 00	8.9E-04	1.6E-04	0.0E+00	0.0E+00	7.8E-05	1.8E~05	0.0E+00	0.0E+00	SELENOSIS
VANADIUM	67 6	35 02	1 11E-06	0 00E + 00	7 00E-03	18E-02	■ 1E-03	0 0E + 00	0 0E + 00	1.1E-02	5.6E-03	0.0E+00	0.0E+00	9.5E-04	4.9E-04	0.0E + 00	0.0E +00	NS
ZINC	435	177.06	1 1 1E-06	0 00E + 00	2 00E-01	4 0E - 03	1 8E - 03	0 0E + 00	0 0E + 00	2.4E-03	9.6E-04	0.0E+00	0.0E+00	2.1E-04	6.7E~05	0.0E+00	0.0E+00	BLOOD
4,4-DDE	0.06	0 0 321	3 33E-07	2 78E-07		0 0E + 00	0 0E + 00	0 0E+00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	0.0E+00	NS
4,4-DDD	0 37	0.0646	3 336-07	2 70E-07		0 0E+00	0.0E+00	0.0E+00	0 0E + 00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	NS
			.			L					118 4-			3.35 65	0.75			<u> </u>
INYANZA BITE CONTAMINA	W18			NAZARD IN		4 2€~01	1 86-01	4 4 E−03	\$ PE-03	3.8E-01	1.1E-01	1.0E-02	3.0E-03	3 3E-02	0.7E-03	1.8E-03	5.3E-04	
OTHER SUDBURY RIVER	CONTAMIN	IANTS		HAZARD IN	DEX	5 8E - 02	2 8E - 02	7 4E-04	2 2E-04	3.5E-02	1.8E-02	0.0E-04	2.7E-04	3.1E-03	1.4E~03	1.6E-04	4 8E-05	ļ
ALL CHEMICALS OF CON	CERN			HAZARD IN	DEX	6 8E-01	2.1E-01	0 1E-03	2 7E-03	4.1E-01	1.3E-01	1.1E-02	3.3E-03	3.7E-02	1.1E-02	2.0E-03	5.8E-04	_
							DNS.	NS NOT	PECIFIED			NEY	BW: BODY V	NEIGHT C		L NERVOUS	SYSTEM	

6-108

FINAL

(

BEHP:BIS(2-ETHYL HEXYL)PHTHALATE

W92194F

TABLE 8-278 RISK REBULTS FOR SEDMENT EXPOSURE REACH NO 3 NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

1

				CARCINOGI			REBULTS														
CONT AMINANTS	CONCENTRATION MG/KG		CONCENTRATION		CONCENTRATION		CONCENTRATION		CONCENTRATION		CONCENTRATIO		CONCENTRATION		EXPOBURE	EXPOSURE FACTOR		CANCERF	VGKS		CANCER SLOPE FACTOR
CONCERN	MAX	AVO	INGESTION	DERMAL CONTACT	ACCIDENTAL		DERM CONT	AL ACT	(MG/KG/D) - 1 / WEIGHT												
					MAX	AVG	MAX	AVG	OF EVIDENCE												
1,4-DICHLOROBENZENE	0 1255	0 1255	3 84E-07	1 51E-07	1 1E-08	1 1E-00	4 8E-10	4 8E-10	2 40E-02 B2												
124TRICHLOROBENZENE	0 16	0 1253	3 84E-07	1 51E-07	0 0E+00	0 0E + 00	0 0E + 00	0 0E + 00													
NAPHTHALENE*	0 12	0 12	3 84E - 07	7 50E-08	0 0E+00	0 0E + 00	0 0E + 00	0 OE + 00													
ARISENIC*	21 5	7 @1	3 84E - 07	0 00€ + 00	1 4E-05	5 OE-08	0 0E + 00	0 0E + 00	1 80E + 00 A												
ANTIMONY*	18	4 11	3 64E-07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00													
CADMIUM*	18.8	4 88	3 84E-07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00													
CHROMIUM*	2620	202 02	3 84E-07	0 00E + 00	0 0E+00	0 0E + 00	0 0E + 00	0 OE + 00													
LEAD*	285	137 87	1 82E-07	0 00E + 00	0 0E+00	0 0E + 00	0 0E + 00	0 0E + 00													
MERCURY*	54.0	15 98	3 84E - 07	1 51E-00	0 0E+00	0 0E + 00	0 0E + 00	0 0E + 00													
MONOMETHYI MERCURY	0.0263	0 0263	3 84E-07	1 10E-08	0.0F+00	0.05+00	0 0E + 00	0 0F + 00													
DIMETHYLMERCURY*	0 0 1 0 1	0 0074	3 84E - 07	1 18E-08	0 0E +00	0 0E + 00	0 0E + 00	0 0E + 00													
ACETONE	0 31	0 1312	3 84E - 07	7 58E - 07	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00													
8EHP	0 0 1	0 7909	3 84E-07	1 51E-07	4 8E-08	4 0E - 09	1 9E-09	176-09	1 40E-02 82												
ACENAPTHYLENE	0 88	0.8100	3 84E-07	7 50E-00	0 0E+00	0 0E + 00	0 0E + 00	0 0E + 00													
PHENANTHRENE	73	1 3676	3 84E-07	7 58E-08	0 0E+00	0 0E + 00	0 0E + 00	0 0E + 00													
FLUORANTHENE	12	2 3317	3 84E - 07	7 58E-00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00													
PYRENE	11	2 4011	3 84E-07	7 58E - 08	0 0E+00	0 0E + 00	0 0E + 00	0 0E + 00													
BENZOLAIANTHRACENE	4 5	1 3461	3 84E - 07	7 58E - 08	8 5E-08	2 8E-08	2 OE - 00	5 0E-07	5 80E + 00 82												
CHRYSENE	77	1 7372	3 84E - 07	7 506 - 00	1 8E-05	37E-08	34E-08	7 8E - 07	5 80E + 00 B2												
BENZO(B)FLUORANTHEN	4 4 4	1 4950	3 84E-07	7 50E-00	# 3E-0#	3 2E - 08	1 9E-08	8 8E-07	5 80E + 00 B2												
BENZOKIFLUORANTHEN	30	1 4942	3 84E-07	7 58E-00	8 2E-08	3 2E-08	17E-08	6 SE-07	5 80E + 00 82												
BENZO(A)PYRENE	44	1 2001	3 84E-07	7 56E - 00	0 3E-00	2 8E - 08	1 9E-08	5 3E-07	5 80E + 00 B2												
INDENO(123-CD)PYRENE	1.8	0 7858	3 84E - 07	7 58E - 08	3 8E-08	17E-08	7 9E-07	3 4E-07	5 80E +00 B2												
DIBENZ(AH)ANTHRACENE	110	0 22	3 84E-07	7 56E - 06	6 5E-07	4 8E-07	14E-07	9 6E-08	5 80E + 00 B2												
BENZO(GHI)PERYLENE	1.6	0 77 77	3 84E-07	7 56E-08	0 0E+00	0 OE + 00	0 0E + 00	0 OE + 00													
BARILM	110	88.85	3 84E-07	0 00€ + 00	0 0E+00	0 0E+00	0 0E + 00	0 0E + 00													
REBALIUM	43	1.52	3 84E-07	0 00F + 00	87E-08	24F-08	0 DE+00	0 0F + 00	4 30E + 00 A												
COPPER	454	184 48	3 84F-07	0.00F + 00	0.0E+00	0.05+00	0 0E + 00	0.05 +00													
MANGANESE	847	301 42	3 845-07	0 00F + 00	0.0E+00	8 OF + 00	0 0F+00	0.05 + 00													
NICKEI		20 41	3 845-07	0.00E + 00	0.05+00	0.0E +00	0.0E + 00	0.05 400													
RELENILIM			1845-07	0.005.00	0.05.00	0.06.400	0.05+00	0 00 400	ļ												
		200	3 846 - 47	0.005.000	0.05.00	0 0E + 00	0.05.400	0.05.00													
		199 66	3 445 . 07	000E 000	0.05.00		0.05.00														
		0.0354	J 042-0/	7 845 44	0 0E+00	406-04		A 2E - 10	3 405 - 01 04												
	0.00	0.0321	1000-07	7 500 - 00	8 76 - 00		A 75-04	1 80 00	3405-0184												
4,4-000	0.37	0.0040	1002-07	7 592-09	#7E~0 0	1 26-00	0/C-UU	106-08	2402-0182												
NYANZA BITE CONTAMINA	CANCER NIBI	₹l	148-05	5 0E-08	4 6E-10	4 0E-10	l														
OTHER SUDBURY RIVER C	CANCER RIS	C C	8 4E~05	2 0E - 05	1 2E-05	3 8E-00															
	COM			r	7.85.04		1 26-04	3.45-04	1												
ALL UTENIUALS OF CONC			CHARGE IN MIGH		/ 92~00	4 DE-UD	1.40-03	9.00-00	1												

BEHP:BIS(2-ETHYL HEXYL)PHTHALATE

FINAL

١
TABLE 6-28A RISK ASSESSMENT RESULTS FOR SURFACE WATER EXPOSURE SCENARIOS REACH NO. 3 NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

	· [NONCARC	NOGENC	RISK ANAL	YSIS RESUL	TS							
CONTAMINANTB of CONCERN	CONCENTR/ MG/	KG	EPPOSURE RECEPTOR	FACTOR	AFD (MG- 103-	HAZAI		ENTS: CHI	D	HAZAR		NT9: TEEN		HAZAR		S: ADULT		TOXICITY END- POINT
CONCENT	NAX .	AVG	INGESTION	DERMAL	DAY)	ACCIDEN	TAL DN	DERM	AL LCT	ACCIDENT		DEAMA	L ST	ACCIDENT/	AL N	DERMA	L	
		l	ļ			MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	
LEAD*	0 001	0 001	1 11E-03	1 38E-04		0 0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	CNS
BEHP	0 001	0 001	1.11E-03	1.10E-04	2.00E-02	4.8E-05	4.8E-05	2.5E-08	2.5E-08	5.8E-05	5.6E - 05	5.5E-08	5.5E-08	9.6E-06	9.8E-08	3.0E-06	3.0E-08	UVER
BARIUM	0.0126	0 0006	1 11E-03	138E-04	500E-02 100E-01	2 3E-04 7 6E-04	1.0E-05 5.8E-04	1.6E-05 5.3E-05	7.2E07 4.0E05	2.8E-04 9.3E-04	1.2E-05 7.0E-04	3.5E-05 1.2E-04	1.5E-06 6.7E-05	4.9E-05 1.6E-04	2.2E-08 1.2E-04	1.9E~05 6.3E-05	8.5E-07 4.8E-05	BLOOD CNS
SELENIUM	19 3	3 86 15	1 11E-03	1 38E-04	5 00E-03	3 5E+00	7.1E-01	25E-01	4 9E-02	4.3E+00	6.6E-01	5.3E-01	1.1E-01	7.6E-01	1.5E-01	2.9E-01	5.9E-02	SELENOS
ZINC	0 008	0,008	111E-03	1.38E-04	2.00E-01	3.7E-06	3.7E-05	2.58-04	2.5E-06	4.4E-05	4.4E-05	5.5E-04	5.5E-08	7.8E-08	7.8E-08	3.0E-04	3.0E-06	BLOOD
NYANZA SITE CONTAMIN	NANTS'			HAZARD IN		0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
OTHER SUDBURY RIVER		179			DEX	3 5E+00	7.1E-01	2.5E-01	4.9E-02	4.3E+00	8.6E-01	5.3E-01	1.1E-01	7.8E-01	1.5E-01	2.9E-01	5.9E-02	
ALL CHEMICALS OF COM	NCEFIN			HAZARD IN	DEX	3.5E+00	7.1E-01	2.5E-01	4.9E-02	4.3E+00	8.6E-01	5.3E-01	1.1E-01	7.6E-01	1.5E-01	2.9E-01	5.9E-02	
BEPH: BIS/2ETHYL HEXY	LIPHTHALATE				NDPOINTS A	BBREVIATIC	2NS :	NS: NOT S	PECIFIED	L/K: LIV DUS SYSTE	er and Kic M	NEY						

BEPH: BIS(2ETHYL HEXYL)PHTHALATE

TABLE 6-28B RISK RESULTS FOR SURFACE WATER REACH NO 3 NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

			CARCINOGENIC RISK RESULTS										
CONTAMINANTS of CONCERN	CONCENTRA MG/	ATION KG	EXPOSURE	FACTOR		CANCER	RISKS		CANCER SLOPE FACTOR				
	MAX	AVG	INGESTION	DERMAL	ACCIDEN	TAL	DERM	AL	(MG/KG/D)-				
				CONTACT	INGESTIC	DN	CONT	ACT	1				
					MAX	AVG	MAX	AVG	WEIGHT				
						{	1	:	OF				
	1								EVIDENCE				
LEAD*	0.001	0.001	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00					
BEHP	0.001	0.001	3.02E-04	3.37E05	4.2E-09	4.2E-09	4.7E-10	4.7E-10	1.40E-02				
BARIUM	0.0126	0.0006	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00					
MANGANESE	0.0836	0.0634	3.02E-04	4.21E~05	0.0E+00	0.0E+00	0.0E+00	0.0E+00					
SELENIUM	19.3	3.8615	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00					
SILVER	0.0119	0.0069	3.02E-04	4.21E05	0.0E+00	0.0E+00	0.0E+00	0.0E+00					
ZINC	0.008	0.008	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	· · · · · · · · · · · · · · · · · · ·				
NYANZA SITE CONTAMI			CANCER R	SK	0.0E+00	0.0E+00	0.0E+00	0.0E+00					
		ITS		ISK	4.2E-09	4.2E-09	4.7E-10	4.7E-10					
ALL CHEMICALS OF COI	NCERN		CANCER R	ISK	4.2E-09	4.2E-09	4.7E-10	4.7E-10					

BEPH: BIS(2ETHYL HEXYL)PHTHALATE

::

FINAL

.

presented. Noncarcinogenic risk estimated for the accidentalingestion-of-sediment exposure route exceeds those presented for the dermal-contact exposure route. If hazard indices are summed for the accidental-ingestion and dermal-contact routes of exposure, the hazard index for the sediment exposure scenarios does not exceed unity. These results suggest that adverse noncarcinogenic health effects would not be anticipated under the conditions of the recreational exposure scenarios for sediments defined in Section 6.4.

If selenium (maximum surface water concentration = 19,300 μ g/L) is excluded from the assessment, the hazard index calculated for the surface water exposures does not exceed unity (Table 6-28). Assuming the accidental ingestion of surface waters while swimming or wading, the hazard quotient for selenium is greater than unity when maximum contaminant concentrations are evaluated. However, it should be noted the selenium was detected only once in Sudbury River surface waters. Selenium was not detected in the sediment samples collected from Reach 3. The positive detection of 19,300 μ g/L selenium in one surface water sample may not be indicative of overall water quality in Reach 3. Lead (maximum concentration ($C_{max}=1$ μ g/L)) was the only Nyanza Site contaminant detected in Reach 3 surface water samples. The concentration detected is lower than the current EPA Action Level for Lead (15 μ g/L).

The excess lifetime cancer risks estimated for the surface water and sediment exposure scenarios slightly exceed 8 x 10^{-5} and 2.5 x 10^{-5} (risks summed for the accidental-ingestion and dermalcontact exposure routes) when maximum and average COC concentrations are evaluated, respectively. As a point of reference, cancer risks estimated for COC concentrations detected in background surface waters and sediments were 2 x 10^{-5} and 9.7 x 10° , respectively. The principal COCs contributing to the risk are the carcinogenic PAHs, arsenic, and beryllium. Cancer risks estimated for the "other Sudbury River contaminants" exceed those estimated for the Nyanza Site contaminants. Cancer risks associated with the surface water exposure scenarios are minimal when compared to those associated with the sediment exposure scenarios.

Hazard indices calculated for all COCs detected in Reservoir No. 2 fish tissue samples exceed unity for all cases presented in Table 6-29. Hazard quotients presented for antimony, arsenic, mercury, methyl mercury, thallium, and zinc exceed unity in one or more cases presented. (It should be noted that anitomy arsenic, and thallium were detected very infrequently in fish tissue samples

W92194F

6-112

TABLE & -29 RMSK ASSESSMENT RESULTS FOR FISH INGESTION EXPOSURE SCENARIOS RESERVOIR NO 2 NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

						NONCARCIN		CARCINOGENIC RISK ANALYSIS RESULTS									
CONTAMINANTS OF CONCERN	CONCENT MQ/	RATION KG	EXPOBURE	FACTOR	- AFD (MG- KG-		HAZARD	QUOTIENTS		TOXICITY END- POINT	EXPOSURE	FACTOR	CAI	NCER RISKS	3		CANCER SLOPE FACTOR
	MAX	AV3		BUBBIB- - TENCE	DAY)	SPORTSF	ISHERMEN	SUBS	BIENCE		BPORTS FISHER-	SUBSIS-	SPORT FK	SHERMEN	BUBSISTI FISHERI	ENCE MEN	(MG/KG/D)1 /
	ļ		FINHERI~	FICHER-		MAXIMUM	AVERAGE	MAXIMUM	AVERAGE	-	MEN	FISHER-	MAXIMUM	AVERAGE	MAXIMUM	AVERAGE	EVIDENCE
								5 (F - 0)		81.000	7.075.06		0.0E · 00	0.05.00	0.05.00	0.0E · 00	
ANTIMONY	15 449	0 302		1382-00	4002-04	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	202-01	7.05.00	1 100 + 00		7.032-05	5010-04	0.000+00	0.02+00	1.000+00	0.02+00	1.005.00.4
AHSENK"	1/54	0.00	1002-04	1302-00	3002-04	1 10 00	0.02-02	1.65.001		SKIN KEY	7.030-00	5.61E-04	2.32 - 04	1.3E - 00	1.85 -03		1.80E+00 A
CADMIUM-	0.002	0.03	1000 -04	1 302 -03	1005-00	1.5-01	1 16-02	7 25 - 01	7.05-04	INCH	7.62 - 05	5.01C-04	0.000+00	0.02+00	0.00 + 00	0.02+00	
	0.030	0.048	1 855 - 04	1362-00	7002700	0.05.00	0.05+00	0.05+00	0.05+00	CNR	7.01 -05	5.010-04	0.000+00	0.02+00	0.00.+00	0.000+00	
	0.540	9.046	105-04	1 300 - 00	1 ME-04	45.00	1 25 + 00	0 0£ + 00	0.02+00	CNR	7.000-00	5.01C-04	0.000	0.000 + 000	0.00.+00	0.02+00	
METHYL MERCURY*	42	1 75847	1852-04	1 362 -03	3 00E -04	2 €+00	1.1E+00	1 9E+01	7.9E+00	CNS	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ENDOSULFAN II	0 001	0 00073	1 85E - 04	1 34E -03	5 00E - 06	37E-03	2 TE - 03	27E-02	2 0E - 02	KIDNEY	7.93E-05	5.81E04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
DELORIN	0 0005	0 0006	1 85E - 04	1 30E - 03	8 00E - 08	1 EE - 03	1 6E - 03	1 Æ-02	1.4E02	LIVER	7.93E-05	5.81E-04	6 3E −07	6.3E-07	4.6E-08	4.6E-06	1.80E+01 B2
4,4-000	0 007	0 00307	1 8E-04	1 3E - 03		00000	0 0E+00	0 0E+00	0.0E+00	LIVER	7.93E-05	5.81E-04	1.3E-07	7.5E-08	9.8E-07	5.5E-07	2.40E-01 82
4,4-DOE	0 008	0.01825	1 85E - 04	1 34E - 03		0 0€+00	0 0E + 00	0 0E+00	0 0E + 00	LIVER	7.93E-05	5.81E-04	1.6E-06	4.9E-07	1.3E05	3.6E 08	3.40E-01 82
4,4-DUT	0 002	0 0012	1 85E - 04	1 3EE - 03		0 0E+00	0 0E + 00	0 0E + 00	0 0E + 00	LIVER	7.93E-05	5.81E-04	5.4E-08	3.2E-06	4.0E-07	2.4E-07	3.40E-01 B2
ALPHA-CHLORDANE	0 001	0 00063	1 85E - 04	1 3E - 03	6 OOE - 05	3 ₩Ε - 03	5 8E - 03	2 3E - 02	19E-02	LIVER	7.93E-05	5.81E-04	1.0E-07	8.6E-06	7.8E-07	6.3E-07	1.30E+00 82
GAMMA-CHLORDANE	0 002	0 00 108	1 8E - 04	1 3E - 03	8 OOE - 05	0 2E - 03	3 3E - 03	4 5E - 02	24E-02	LIVER	7.93E-05	5.81E-04	2.1E-07	1.1E-07	1.5E-08	8.0E07	1.30E+00 B2
ALDRIN	0 001	0 00032	1 85E-04	1 30E - 03	3 00E 06	82E-03	\$ 0E - 03	4 5E - 02	1.5E-02	LIVER	7.93E-05	5.81E-04	1.3E-08	4. € −07	9 9E~08	3.2E-08	1.70E+01 62
AROCLOR-1254	0 73	0 08177	1 85E-04	1 36E - 03		00E+00	0 0E + 00	0 0E + 00	0 0E + 00	NS	7.93E-05	5.81E-04	4.5E-04	5.6E-05	3.3E - 03	4.1E-04	7.70E+00 B2
AROCLOR-1280	0 093	0 07546	1 82 - 04	1 36E - 03		0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	NS	7.93E-05	5.61E-04	5.7E-05	4.7E-05	4.至-04	3.4E-04	7.70E+00 82
COPPER	0 723	0 56	1 85E-04	1 SE - 03		0 0E+00	0 0E + 00	0 0E + 00	0 0E + 00	NS	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MANGANESE	13 752	0 829	1 8E-04	1 3E - 03	1 DOE -01	2 0E - 02	1.5E-03	19E-01	1.1E-02	CNS	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
SILVER	0 444	0.00	1 85E-04	1 3E - 03	3 00E - 03	276-02	3 7E - 03	2 0E-01	2.7E-02	ARGYRIA	7.93E05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	1
THALLIUM	1 340	0 078	1 85E - 04	1 36 - 00	7 OOE - 08	36E+00	2 1E - 01	2 E +01	1.5E+00	NS	7.93E05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
VANADIUM	1 181	0 322	1 8E-04	1 36E - 03	7 00E - 03	3 € - 02	8 5E - 03	2 2E - 01	62E-02	NS	7.93E-05	5.81E-04	0.0E+00	0 0E + 00	0.0E+00	0.0E+00	
ZINC	178 530	18 814	1 85E - 04	1 3E -03	2 00E-01	1 7E - 01	1 7E-02	12 + 00	1.3E-01	BLOOD	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	L
NYANZA BITE CONTAMIN	ANTS		HAZAPID IN	DEX		1 5 + 01	176+00	9 E + 01	2.0E+01		CANCER RI	SKS:	2.5E-04	1. 5E - 05	1.8E-03	1.1E-04]
OTHER SUDBURY RIVER	CONTAMINA	ena	HAZARDIN	DEX		3 € + 00	2 CE -01	2 Œ +01	16E+00		CANCER RI	5KS:	5.1E-04	1.0E04	3 Æ-03	7.7E−04	
ALL CHEMICALS OF CON	CERN			DEX		1 元 +01	1.EE+00	1 2 + 02	2 2E + 01		CANCER RI	9K9:	7.6E-04	1.2E-04	5.5E~03	8.8E-04	

BEPH:BIS(2-ETHYL HEXYL)PHTHALATE

TOROTTY ENDPOINT ABBREVIATIONS

NO NOT SPECIFIED BW. BODY WEIGHT CNS: CENTRAL NERVOUS SYSTEM LK: LVER AND KIDNEY

collected during the Sudbury River RI). The hazard indices presented for the Nyanza Site contaminants exceed those estimated for the "other Sudbury River contaminants." All of these results suggest that adverse noncarcinogenic health effects may be anticipated for the sports fishermen or subsistence fishermen consuming fish taken from Reservoir No. 2 under the conditions of the exposure scenarios defined in Section 6.4. The maximum and average mercury and methyl mercury concentrations detected in the fish tissue samples exceed the current FDA Action Level.

Cancer risks estimated for the sports fishermen and subsistence fishermen exceed 1×10^4 and 1×10^3 for the reasonable maximumand average-case scenarios, respectively. Several pesticides/PCBs contribute to the estimated risk. However, arsenic is the only Nyanza Site contaminant contributing to the estimated excess lifetime cancer risk. As a point of reference, risks estimated for Sudbury Reservoir fish tissue samples do not exceed 5×10^4 even when the subsistence fisherman is the receptor of concern and maximum contaminant concentrations are evaluated.

6.5.3.10 Risk Assessment Results for Reach 4

Sudbury River Reach 4 is also a sedimentation area. The Reach includes Reservoir No. 1 and ends at Dam No. 1 at the Winter Street crossing in Framingham. The following Nyanza Site contaminants were detected in Reach 4 surface waters and sediments:

0	Arsenic	0	Lead
0	Cadmium	0	Mercury
0	Chromium	0	Methyl mercury

The average mercury concentration detected in Reach No. 4 sediments $(C_{reg} = 3.38 \text{ mg/kg})$ is 12 times concentrations detected in background sediment samples $(C_{reg} = 0.27 \text{ mg/kg})$.

Tables 6-30 and 6-31 present risk assessment results for COC concentrations detected in sediments and surface waters of Reach 4. The accidental-ingestion and dermal-contact routes of exposure (surface waters and sediments) were evaluated assuming recreational land/water use scenarios. The hazard quotients and hazard indices presented do not exceed unity for any of the cases presented. Only the hazard quotient for arsenic exceeds 0.1 in one case presented. If hazard indices are summed for the accidental-ingestion and dermal-contact exposure routes and then combined for surface water and sediment exposures, the total hazard index does not exceed unity. These results indicate that adverse noncarcinogenic health effects would not be anticipated for a receptor contacting Reach 4

W92194F

6-114

Г	

TABLE 0 - 30A FISK ASSESSMENT RESULTS FOR SEDIMENT EXPOSURE SCENARIOS REACH NO 4 NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

- ----

	L						NONCARC	NOGENC	RISK ANA	LYDIS RESU	LTS							
CONTAMINANT8 of CONCERN		TRATION	EXPOSURE RECEPTO	FACTOR R = TEEN	RFD	HAZ	ARD QUOT	NENTS CH	LD	HAZ		TENTS: TE	EN	HAZAR				TOXIC END- POINT
	MAX	AVG	INGESTION	DERMAL CONTACT	(MG - KG -	ACCIL		DEI	AMAL MTACT	ACCIDENT	AL N	DERM CONTA	AL ACT	ACCID	ENTAL	DERN	AAL ACT	
					DAY	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	L
	32.3	10 68	1 11E-00	0 00E + 00	3 00E - 04	2 OE - 01	6 5E - 02	0 0E + 00	0 0E + 00	1 2E - 01	4 0E - 02	0.0E + 00	0.0E + 00	1.1E-02	3.5E-03	0.0E + 00	0.0E+00	SKIN
			1116-00	0.002+00	5 00E - 04	346-02	1 95 -04		0.000	355-02	1.1E-02	0.02+00	0.000	2.95-05	7.05-04	0.000+00	0.00 + 00	LIVER
CHHOMIUM"	224		8 885 - 02	0.000 + 000	1002+00		0.05+00		0.05+00	2 JE - 04	0.05+00	0.02 +00	0.00 + 00	2.22~00	0.05+00	0.02+00	0.00+00	CNR
		1 14	1116-04	5 55E - 08	1005-04	4 4F - 02	2 16-02	1 1E-03	5 1E -04	27E-02	1 3E - 02	1 4F -03	8 3E-04	2 4 E - 03	1 1E-03	245-04	1 1E - 04	CNS
	0.0203	0.0287	1116-00	4 28F - 06	3 00E - 04	4 8F - 04	1 AE - 04	9 3F - 04	3 1E-04	2 8E 04	8 8F -05	1.1E-03	3.8E-04	2 6E - 05	67E-06	2.0E-04	8 7E - 05	CNS
	00.00														0.72 00		0.72 00	
ACETONE	2.6	1 382	111E-08	2 78E-08	1 00E - 01	4 7E - 05	2 5E - 05	5 9E - 05	3 1E - 05	2 9E - 05	1 5E~05	7.2E-05	3.8E-05	2.5E-06	1.3E-08	1.3E-05	6.7E-06	L/K
PHENANTHRENE	017	0 17	1 11E-00	2 70E-07	4 00E - 03	7 0E -05	7 8E - 05	97E-06	97E-06	4 7E - 05	47E-05	. 1.2E-05	1.2E-05	4 2E - 06	4 2E - 08	2 1E-08	2.1E-08	BW
FLUORANTHENE	0 27	0 26	1 11E-08	2 78E -07	4 00E - 02	1 2E -05	126-05	1 5E - 06	1 5E - 06	7 5E - 06	7.2E~06	1.9E-08	1.6E-06	6.6E - 07	6.4E-07	3.3E - 07	3.2E - 07	L/K
PYRENE	0 20	0 255	1 11E-06	2 78E - 07	3 00E - 02	1 8E - 05	1 6E - 05	2 0E - 08	19E-06	9 6E - 06	9.4E - 08	2.4E-08	2.4E-08	8.5E - 07	8.3E - 07	4 2E ~ 07	4.2E 07	KIDNEY
BENZO(A) ANTHRACENE	011	0 1 1	1 11E-08	2 78E -07		0 0E + 00	0 0E + 00	0.0E + 00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	0.0E + 00	0.0E + 00	NS			
CHRYSENE	0 22	0 205	1 11E-00	2 78E-07		0 0E + 00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E + 00	0.0E+00	NS			
BENZO(B)FLUORANTHEN	04	0 275	1 11E-00	2 70E -07	1	0 0E + 00	0 0E + 00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	0.0E + 00	NS			
BENZOKAFLUORANTHEN	E 015	0 15	1116-06	2 70E-07		0 0E + 00	0.0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	0.0E + 00	0.0E + 00	NS
BENZO(A)PYRENE	0 15	0 14	1 11E - 08	2 70E - 07		0 0E + 00	0 0E + 00	0.0E + 00	0.0E + 00	0 0E + 00	0.0E + 00	0.0E+00	0.0E + 00	0.0E+00	NS			
NDENO(123 - CD)PYRENE	0 094	0 084	1 11E-00	2 78E - 07		0 0E + 00	0 0E + 00	0.0E + 00	0 0E + 00	0.0E+00	0.0E + 00	0.0E + 00	0.0E + 00	0.0E + 00	NS			
BENZO(GHI)PERYLENE	0 002	0 002	1 1 1E-00	2 78E - 07	4 00E ~ 03	4 2E - 05	4 2E ~ 05	5 3E - 08	5 3E - 06	2 6E - 05	2 8E - 05	8.4E-06	6.4E-08	2.3E-08	2.3E - 08	1.1E~06	1.1E-06	NS
BARIUM	135	ee e 1	1 11E-00	0 00E + 00	6 00E - 02	4 OE - 03	2 6E - 03	0 0E + 00	0 0E + 00	3 OE - 03	1.8E - 03	0.0E + 00	0.0E+00	2.8E-04	1.4E-04	0.0E + 00	0.0E + 00	BLOOD
BERYLUUM	30	0.8	1 11E-00	0 00E+00	5 00E - 03	1 3E - 03	2 9E - 04	0 0E + 00	0 0E + 00	8 OE - 04	1.8E-04	0 0E + 00	0.0E+00	7.0E – 05	1.8E-05	0.0E + 00	0.0E + 00	NS
COPPER	332	115 85	1 11E-00	0 00E + 00	1	0 0E + 00	0 0E + 00	0 0E + 00	0.0E + 00	0.0E+00	0 0E + 00	0 0E + 00	0.0E + 00	0.0E + 00	NS			
NICKEL	63	28.08	1 11E-00	0 00E + 00	2 00E - 02	5 8E - 03	2 6E - 03	0 0E + 00	0 0E + 00	3 5E - 03	1.8E-03	0.0E+00	0.0E+00	3.1E-04	1.4E-04	0.0E + 00	0.0E+00	BW
SELENIUM	4	174	1 1 1 E - 00	0 00€ + 00	5 00E - 03	1 5E -03	84E-04	0 0E + 00	0 0E + 00	8.9E-04	3.9E-04	0.0E + 00	0.0E+00	7.8E-05	3.4E-05	0 0E + 00	0.0E + 00	SELENOSIS
VANADIUM	071	34 63	1118-00	0 00E + 00	7 00E - 03	1 BE - 02	0 OE - 03	0.0E+00	0 0E + 00	1.1E-02	5.5E-03	0.0E+00	0.0E+00	9.4E-04	4 BE - 04	0.0E+00	0.0E+00	INS
NYANZA SITE CONTAMINA	NTS			HAZARD INC	λŧ.	3 0E - 01	1.0E-01	1.1E-03	5 1E-04	1 6E - 01	6 3E - 02	1.4E-03	6.3E-04	1.8E-02	5.5E-03	2.4E-04	1.1E-04	1

	TOXICITY ENDPOINTS	BREVIATIONS	NB NOT SPECIFIED	L/K: UVER AND KIDNEY	BW: BODY WEIGHT	CNS: CENTRA	NERVOUS SY	STEM
ALL CHEMICALS OF CONCERN	 HAZARDINDEX	3 3E-01 1 2E-0	1 12E-03 58E-04	2.0E-01 7.2E-02 1.4E-03	6.9E-04 1.8E-0	2 6.4E-03	2.5E-04	1 2E 04
OTHER SUDBURY RIVER CONTAMINANTS	HAZARD INDEX	3 1E-02 1 5E-0	2 7 8E-05 4 9E-05	19E-02 9.3E-03 9.5E-05	8.0E-05 1.7E-0	8 2E ~ 04	1.7E-05	1.1E-05
NYANZA SITE CONTAMINANTS	 HAZARD INDEX	3 0E-01 1.0E-0	1 1.1E-03 5 1E-04	16E-01 63E-02 1.4E-03	6.3E-04 1.6E-0	2 5.5E-03	2.4E-04	1.1E-04

BEPH:BIS(2-ETHYL HEXYL)PHTHALATE

TABLE 6-30B RISK RESULTS FOR SEDIMENT EXPOSURE REACH NO. 4 NYANZA OPERABLE UNIT 3

MIDDLESEX CO., MASSACHUSETTS

				CARCINOGE	ENIC RISK A	WALYSIS I	RESULTS		CANCER
CONTAMINANTS of		TRATION KG	EXPOSURE	FACTOR			RISKS		SLOPE FACTOR
CONCERN	MAX	AVG	INGESTION	DERMAL CONTACT	ACCIDEN	TAL DN	DERM CONT/	AL ACT	(MG/KG/D)-1 / WEIGHT
					MAX	AVG	MAX AVG		OF EVIDENCE
ARSENIC*	32 3	10 68	3 64E - 07	0.00E + 00	2.1E-05	7.0E-06	0.0E+00	0.0E+00	1.80E+00 A
CADMIUM*	14 9	4 81	3 64E - 07	0 00E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
CHROMIUM*	224	71 61	3 64E - 07	0 00E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
LEAD*	219	93 96	1 82E - 07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MERCURY*	73	3 38	3 64E - 07	1.51E-08	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MONOMETHYLMERCURY	0 0793	0 0267	3 64E - 07	1.16E - 06	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ACETONE	2.6	1.362	3 64E - 07	7.56E - 07	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
PHENANTHRENE	0.17	0 17	3 64E - 07	7.56E - 08	0.0E+00	0 0E+00	0.0E+00	0.0E+00	
FLUORANTHENE	0 27	0 26	3 64E ~ 07	7.56E - 08	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
PYRENE	0 26	0 255	364E-07	7.56E - 08	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
BENZO(A)ANTHRACENE	0.11	0 11	3 64E - 07	7.56E-08	2.3E-07	2.3E-07	4.8E-08	4.8E-08	5.80E+00 E
CHRYSÈNE	0 22	0 205	364E-07	7.56E-08	4.6E-07	4.3E-07	9.6E-08	9.0E-08	5.80E+00 E
BENZO(B)FLUORANTHEN	E 04	0 275	3.64E-07	7.56E-08	8.4E-07	5.8E-07	1.8E-07	1.2E-07	5.80E+00 E
BENZO (K) FLUORANTHEN	E 015	0 15	3 64E - 07	7.56E-08	3.2E-07	3.2E-07	6.6E-08	6.6E-08	5.80E+00 E
BENZO(A) PYRENE	0 15	0.14	3 64E - 07	7.56E - 08	3.2E-07	3.0E-07	6.6E-08	6.1E-08	5.80E+00 E
INDENO(123-CD)PYRENE	0 094	0 094	3 64E - 07	7.56E-08	2.0E-07	2.0E-07	4.1E-08	4.1E-08	5.80E+00 E
BENZO(GHI)PERYLENE	0 092	0 092	3.64E - 07	7.56E-08	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
BARIUM	135	69.91	3 64E - 07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
BERYLLIUM	36	08	3 64E - 07	0.00E+00	5.6E-06	1.3E-06	0.0E+00	0.0E+00	4.30E+00 A
COPPER	332	115.85	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
NICKEL .	63	26 69	3 64E - 07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
SELENIUM	4	1.74	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	1
VANADIUM	67.1	34.63	3.64E - 07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	

NYANZA SITE CONTAMINANTS	CANCER RISK	2.1E-05	7.0E-08	0.0E+00	0.0E+00
OTHER SUDBURY RIVER CONTAMINANTS	CANCER RISK	8.0E-06	3.3E-06	4.9E-07	4.3E-07
ALL CHEMICALS OF CONCERN	CANCER RISK	2.9E-05	1.0E-05	4.9E-07	4.3E-07

BEPH:BIS(2-ETHYL HEXYL)PHTHALATE

6-116

1 6-31A
RISK ASSESSMENT RESULTS FOR SURFACE WATER EXPOSIONE SCENARIOS
REACH NO. 4
NYANZA OPERABLE UNIT 3
MIDDLESEX COUNTY, MASSACHUSETTS

.

· · · · ·		NONCARCINOGENC RISK ANALYSIS RESULTS																
CONTAMINANTS of CONCERN	CONCENT	ATION I KG	EXPOSUAL RECEPTO	e factor	A¥FD (MG - KG -	HAZA		ENTS: CH		HAZA	rid quotiei	NTS: TEEN		HAZAR	D QUOTIEN	TS: ADULT		TOXICITY END- POINT
	MAX	AVG	NOESTO	DERMAL	DAY)	ACCIDEN	TAL ON	DERM	AL NCT	ACCIDEN	TAL	DERMA	L DT	ACCIDENT	AL N	DERMA	L	
				1		MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	
CHROMIUM* LEAD*	0 000 0 01 2 0	0 0045	1 11E-03	1 38E-04 1 38E-04	1 00E+00	8 5E - 08 0 0E + 00	4.1E-08 0.0E+00	38E-07 00E+00	2 9E-07 0 0E+00	6.7E-08 0.0E+00	5.0E-08 0.0E+00	0 3E-07 0.0E+00	6 2E - 07 0.0E+00	1.2E08 0.0E+00	8 8E-07 0.0E+00	4 6E-07 0.0E+00	3.4E-07 0.0E+00	LIVER CNS
BARIUM MANGANESE	0 0241	0.0129	1 11E-03	138E-04	1 38E-04 8 00E-02 4 5E- 1 38E-04 1.00E-01 1 9E-		45E-04 23E-04 19E-03 12E-03		1 6E - 05 6 2E - 05	5.5E-04 2.3E-03	2.9E-04 1.4E-03	6.8E-05 2.9E-04	3.5E-05 1.8E-04	9.7E-05 5.0E-05 4.1E-04 2.5E-04		3.6E-05 1.6E-04	2.0E05 9.8E05	BLÓCD CNS
NYANZA SITE CONTAMI	NANTS			HAZARD IN	DEX	8 5E - 08	4.1E-08	3 8E - 07	2 9E-07	67E-06	5.0E-08	8.3E-07	6.2E-07	1.2E-08	8.8E-07	4.6E-07	3.4E-07]
OTHER SUDBURY RIVER	CONTAMINA	NTS		HAZARD IN	DEX	2 4E - 03	1 4E-03	1 OE - 04	9 9E - 05	2.9E-03	1.7E-03	3.6E-04	2.1E-04	5.1E-04	3.0E-04	2.0E-04	1.2E-04	j
ALL CHEMICALS OF CO	NCERN		· · _ - · · · · · · ·	HAZARD IN	DEX	2 4E-03	1 4E-03	1 7E - 04	9 9E - 06	2.9E-03	1.7E-03	3.6E-04	2.1E-04	5.1E-04	3.0E-04	2.0E-04	1.2E-04	j
BEPH: BIS(2ETHYL HEXY	L)PHTHALATE				NDPOINTS A	BBREVIATIO	ONS.	NS NOTS	IPECIFIED	L/K: LIV DUS SYSTE	(er and kie M	INEY						

W92194F

(

TABLE 6-31B RISK RESULTS FOR SURFACE WATER REACH NO 4 NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

				CARCINOGENIC RISK RESULTS									
CONTAMINANTS of CONCERN	CONCENTR/ MG/	ATION KG	EXPOSURE	FACTOR		CANCER	RISKS		CANCER SLOPE FACTOR				
	MAX	AVG	INGESTION	DERMAL CONTACT	ACCIDEN	TAL DN	DERMAL CONTACT		(MG/KG/D) /				
					MAX	AVG	MAX	AVG	WEIGHT				
									EVIDENCE				
CHROMIUM*	0.006	0.0045	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00					
LEAD*	0.0126	0.0069	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00					
BARIUM	0.0247	0.0129	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00					
MANGANESE	0.209	0.1289	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00					
NYANZA SITE CONTAM	NANTS*		CANCER R	ISK	0.0E+00	0.0E+00	0.0E+00	0.0E+00	7				

ļ	NYANZA SITE CONTAMINANTS"	CANCERHISK	U.UE+00	0.0E+00	U.UE+00	0.0E+00
	OTHER SUDBURY RIVER CONTAMINANTS	CANCER RISK	0.0E+00	0.0E+00	0.0E+00	0.0E+00
	ALL CHEMICALS OF CONCERN	CANCER RISK	0.0E+00	0.0E+00	0.0E+00	0.0E+00

BEPH: BIS(2ETHYL HEXYL)PHTHALATE

FINAL

W92194F

::

6-118

surface water and sediments under the conditions of the recreational exposure scenarios defined in Section 6.4.

The excess lifetime cancer risks estimated for the sediment exposures are 2.9×10^{-5} and 1.0×10^{-5} (risks summed for the accidental-ingestion and dermal-contact exposure routes) when maximum and average contaminant concentrations are evaluated, respectively. As a point of reference, cancer risks estimated for background sediment concentrations were 1.8×10^{-5} and 8.2×10^{-6} for the reasonable maximum and average case, respectively. The principal contaminants contributing to the risk are beryllium, the carcinogenic PAHs, and arsenic, a Nyanza Site contaminant.

<u>Risk Assessment Results for Fish Tissue Samples - Reservoir No. 1</u>

Hazard indices calculated for all COCs and the Nyanza Site contaminants exceed unity in all cases presented in Table 6-32 except when average contaminant concentrations are evaluated and a sport fishers man is the receptor of concern. The principal contaminants contributing to the noncarcinogenic risk are mercury, methyl mercury, zinc, and arsenic. The hazard quotients calculated for mercury and methyl mercury exceed unity in all cases except when average contaminant concentrations are evaluated and the sports fishermen is considered the receptor of concern. The hazard quotient calculated for arsenic, another Nyanza Site contaminant, exceeds 0.5 when the reasonable maximum-case scenario is evaluated and the subsistence fisherman is the receptor of concern. Hazard indices calculated for "other Sudbury River contaminants" exceed unity only when maximum contaminant concentrations are evaluated and the subsistence fisherman is considered the receptor of concern. The hazard quotient calculated for zínc exceeds unity (1.8). Hazard quotients calculated for bis(2-ethylhexyl)phthalate, copper, manganese, and vanadium all exceed 0.1 when the reasonable maximum-case scenario is evaluated and the receptor is the subsistence fisherman. Hazard indices calculated on a target organ-specific basis exceed unity in one or more cases when the central nervous system, kidney, or the blood system is considered the target organ of concern. These results suggest that adverse noncarcinogenic health effects would be anticipated for the receptor routinely consuming fish taken from Reservoir No. 1.

Cancer risks estimated for the sports fishermen and subsistence fishermen exceed 1×10^{-5} and 1×10^{-4} , respectively. Several pesticides, bis(2-ethylhexyl)phthalate, and arsenic contribute to the estimated risk. These cancer risk levels are similar to those reported for fish tissue samples taken from the Sudbury Reservoir.

W92194F

TABLE 8-32 RISK ASBE SSMENT RESULTS FOR FISH INDESTION EXPOSURE SCENARIOS RESERVOIR NO 1 NYANZA OPERABLE UNIT 3 MODLESEX COUNTY, MASSACHUSETTS

	T					NONCARCIN	OGENIC RISK AN	ALYSIS RESULT	19		1		CARCINOG	ENIC RISK /	NALYSIS RI	ESULTS	
CONTAMINANTS OF CONCERN		KG	EXPOSURE	FACTOR	PFD MG		HAZARD	QUOTIENTS		TOXICITY END-	EXPOSURE	FACTOR-	CA	NCER RISK	3		
	MAX	AVO	-	AURON	DAM		·····	- T			SPORTS	SUBSIS-	SPORTE	SHERMEN	BUBBIST	ENCE	
		1		-TENCE	,	BPORTHE	RHERMEN	SUBSE	TENCE		FISHER-	TENCE			FISHER	MEN	/
	Ì	1	FIGHER-	FIBHER-				FISHEF	MEN		MEN	FISHER-		·		T	WEIGHTOF
	_	I	MEN	MEN		MAXIMUM	AVERAGE	MAXIMUM	AVERAGE	-		MEN	MAXIMUM	AVERAGE	MAXIMUM	AVERAGE	EVIDENCE
PHENOL*	23	0 4 1035	1857-04	1 36E - 03	6 00E - 01	7 1E-04	13E-04	5 2E - 03	9 3E - 04	FETUS	7.93E-05	5.81E04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ARSENIC	0 114	0.08	1 8E - 04	1 3E - 03	300E-04	7 0E - 02	4 E - 02	52E-01	36E-01	SKIN	7.93E-05	5.81E-04	1.6E-05	1.1E-05	1.2E-04	8.4E-05	1.80E+00 A
CHROMIUM*	0.310	1 000	185E-04	1 36E - 03	100E+00	1 E -03	2 0E - 04	1 1E-02	15E-03	LIVER	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
LEAD*	2 157	0 042	1 85E-04	1 36E - 03		0.000 + 000	0 0E + 00	0 0E + 00	0 0E + 00	CNS	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MERCURY*	4 185	0 748	1 85E-04	1 38E - 03	3 00E 04	2 6€+00	4 6E - 01	1 97E + 01	3.4E+00	CNS	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0 0E+00	0.0E+00	}
METHYL MERCURY*	3 78	0 99075	1 8CE - 04	1 38E - 03	3 OCE - 04	2 3E + 00	6 1E-01	1	4 5E + 00	CNS	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
верн	33	0 85855	1 85E - 04	1 36E 03	2 00E - 02	3 1€-02	6 1E-03	2 Æ -01	4 9E - 02	LIVER	7.93E-05	5 81E-04	3.7E-08	7.3E-07	2.7E-05	5.4E-06	1.40E-02 E
4,4-DDD	0 00	0 00131	1 85E - 04	1 3E - 03		0 0E +00	0 0E + 00	0 0E + 00	0 0E + 00	LIVER	7.93E-05	5.81E-04	3.6E −07	2.5E-08	2.8E-08	1.6E−07	2.40E-01 E
4,4-D0E	0 032	0 01773	1 85E - 04	1 3E - 03		0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	LIVER	7.93E-05	5.81E-04	6.6E07	4.8E-07	6.3E-08	3.5E-06	3.40E-01 8
4,4-DOT	0 001	0 00074	1 85E-04	1 36E - 03		0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	LIVER	7.93E-05	5.81E-04	2.7E-08	2.0E 08	2.0E-07	1.5E07	3.40E-01 E
ALPHA-CHLORDANE	0 0025	0.00164	1 BEE-04	1 3EE - 03	8 00E - 05	7 7E -03	5 1E - 03	5 7E - 02	37E-02	LIVER	7.93E-05	5.81E-04	2.6E – 07	1.7E-07	1.9E-08	1.2E-08	1.30E+00 E
GAMMA-CHLORDANE	0 0015	0 001	1 BEE - 04	1 36E - 03	8 00E - 05	4 6E - 03	3 1€-03	3 € - 02	2.3E~02	LIVER	7.93E-05	5.81E04	1.5E-07	1.0E-07	1.1E-08	7.6E-07	1.30E+00 E
AROCLOR-1254	0 034	0 00636	1 85E-04	1 3E -03		0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	NS	7.93E-05	5.81E-04	2.Æ-05	5.1E-08	1.7E-04	3 6E - 05	7.70E+00 E
AROCLOR-1280	0.086	0 0088	1 85E - 04	1 30E -03		0 0E + 00	0 0E+00	0 0E+00	0 0E + 00	NS	7.93E-05	5.81E-04	4.2E~05	4.1E-08	3.1E-04	3.0E - 05	7.70E+00 8
BARIUM	0 167	0 056	1 80E - 04	1 36E - 03	8 00E - 02	8 2E - 04	2 1E-04	4 5E - 03	16E-03	BLOOD	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
COPPER	6313	0 57	1 85E-04	1 36E - 03]	0 0E+00	0 0E + 00	0 0E + 00	0 0E+00	NS	7.93E-05	5.81E04	0.0E+00	0.0E+00	0.0E+00	0.0E+00]
MANGANESE	10 063	2 264	1 85E-04	1 36E - 03	1 00E - 01	2 Œ -02	4 2E - 03	1 € −01	3. 1E ~ 02	CNS	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
VANADIUM	1 572	0 407	1 8E-04	1 36E - 03	7 00E - 03	4 22 - 02	11E-02	3 OE - 01	9.6E - 02	NS	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	ĺ
ZINC	202 40	31 715	1 8E - 04	1 3EE - 03	2 00E -01	2 € -01	2 9E - 02	1 BE + 00	2 2E - 01	BLOOD	7.93E - 05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
NYANZA SITE CONTAMIN	ANTS"		HAZARD IN	DEX		2.7E+00	5 1E-01	1 GE+01	3.7E+00		CANCER R	ISKS:	1.6E-05	1.1E-05	1.2E-04	8.4E-05	7
OTHER SUDBURY RIVER	CONTAMIN	ETIN	HAZARD IN	DEX		3 5E - 01	6.1E-02	2 5E+00	4.5E-01		CANCER R	ISKS:	7.1E-05	1.1E-05	5.2E-04	7.9E05	
ALL CHEMICALS OF CON	CERN		HAZARDIN	DEX		3 OE + 00	5 7E 01	2 2 + 01	4 2E+00		CANCER RI	SKS:	6.8E-05	2.2E-05	6.4E -04	1.6E-04	

BEPH:BIS(2-ETHYL HEXYL)PHTHALATE

TOXCITY ENDPOINT ABBREVIATIONS.

NG. NOT SPECIFIED BW: BODY WEIGHT CNS: CENTRAL NERVOUS SYSTEM

6.5.3.11 <u>Risk Assessment Results for Reach 5</u>

As described in Section 1.0, Sudbury River Reach 5 extends from Dam No. 1 to the Massachusetts Turnpike (Interstate 90) overpass where the River widens. The river flow velocities are higher than in the impounded reservoirs; however, there are low-flow velocity areas present throughout Reach which are conducive to sediment deposition. Four Nyanza Site contaminants where detected in Reach 5 sediments and surface waters:

0	Arsenic	0	Lead
0	Chromium	ο	Mercury

The average mercury concentration detected in Reach 5 sediments is greater than three times the concentration detected in background sediment samples.

Tables 6-33 and 6-34 present risk assessment results for COC concentrations detected in sediments and surface waters of Reach 5. The accidental-ingestion and dermal-contact routes of exposure (surface waters and sediments) were evaluated assuming recreational land/water use scenarios. The hazard quotients and hazard indices do not exceed unity in any case presented. None of the hazard quotients presented exceed 0.1. If hazard indices are nazara quotients presented exceed 0.1. If hazard indices are summed for the accidental-ingestion and dermal-contact exposure routes and then combined for surface water and sediment exposures, the total hazard index does not exceed unity. Thus, adverse noncarcinogenic health effects are not anticipated under the conditions of the recreational exposure scenarios defined in Section 6.4.

Excess lifetime cancer risks estimated for maximum and average COC concentrations detected in Reach 5 sediments are similar to those estimated for background COC concentrations. Arsenic was the only Nyanza Site contaminant contributing to the risk. Background arsenic sediment concentrations exceed those detected in Reach 5.

As summarized on Table 6-34, cancer risks estimated for surface water exposure scenarios do not exceed 1×10^4 (risks summed for the accidental-ingestion and dermal-contact exposure routes). Arsenic is the only COC contributing to the estimated risk. As a point of reference, the 1×10^4 to 1×10^4 cancer risk range is often evaluated in the development of health-based standards/criteria and in the determination of cleanup goals at hazardous waste sites.

W92194F

6-121

W92194F

6-122

TABLE 6-33A RISK ASSESSMENT RESULTS FOR SEDIMENT EXPOSURE SCENARIOS REACH NO 5 NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

					T		NONOANC	ANOGENE	INSA AN	Erdid HEde								·······
CONTAMINANTS of CONCERN	CONCEN	TRATION KG	EXPOSURE RECEPTO	FACTOR	RED	HAZ		NENTS CH	ŧLD	HAZ	ARD QUO	NENTS: TE	EN	HAZAR		S. ADULT		TOXIC END- POINT
	MAX	AVG	INGESTION		(MG - KG -			DE	RMAL NTACT			DERM. CONTA	AL NCT		ENTAL STION	DERI		
					DAY	MÁX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	l
ARSENIC		3 63	1 11E-08	0 00E + 00	3 00E - 04	5 8E - 02	2 3E - 02	0 0E + 00	0 0E + 00	3 4E - 02	1.4E-02	0.0E + 00	0.0E+00	3.0E-03	1.2E-03	0.0E + 00	0.0E + 00	SKIN
CHROMIUM*	60.6	17 85	1 11E-08	0 00E + 00	1 00E + 00	1 1E-04	3 3E - 05	0 0E + 00	0 0E + 00	87E-05	2.0E-05	0.0E + 00	0.0E+00	5.9E-06	18E-06	0.0E + 00	0.0E + 00	UVER
LEAD*	809	237 99	5 55E - 07	0 00E + 00		0 0E + 00	0 0E + 00	0.0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0 0E + 00	0.0E + 00	0.0E+00	0 0E + 00	0 0E + 00	0.0E + 00	CNS
MERCURY*	41	0 98	1 11E-08	5 55E - 08	3 00E - 04	2 5E - 02	6.0E-03	6 2E - 04	1 5E - 04	1.5E-02	3.6E-03	7.6E – 04	1.6E-04	1.3E-03	3.2E - 04	1.3E-04	3.2E-05	CNS
ВЕНР	0 33	0 33	1 11E-06	5 55E - 07	2 00E - 02	3 OE - 05	3 0E - 05	7.5E - 08	7 5E - 06	1.8E-05	1.8E-05	9.2E - 06	9.2E-06	1.6E-08	1.6E-06	1.6E~06	1.8E-08	UVER
3/4METHYLPHENOL	0 13	0 13	1 11E-08	5 55E - 07	5 00E - 02	47E-08	47E-08	1 2E - 08	1 2E - 06	2 9E - 06	2 9E~08	1.4E-08	1.4E-06	2.5E-07	2.5E - 07	2 5E - 07	2.5E - 07	BW,CNS
ACENAPTHYLENE	0 16	0 18	1 11E-06	2 78E -07	4 00E - 03	7 3E 05	7 3E - 05	9 1E - 06	9 1E - 08	4 4E - 05	4.4E-05	1.1E-05	1.1E-05	3.9E - 06	3.9E - 06	2.0E - 06	2.0E - 08	NS
PHENANTHRENE	0 54	0 54	1 11E-06	2 78E -07	4 00E - 03	2 5E ~04	2 5E - 04	3 1E ~ 05	3.1E-05	1.5E-04	1.5E-04	3.7E - 05	3.7E-05	1.3E-05	1.3E-05	6.6E~06	6.6E ~ 06	BW
FLUORANTHENE	11	11	1 11E-08	2 78E -07	4 00E - 02	5 OE - 05	5 OE - 05	8 3E - 08	8 3E - 08	3.1E-05	3.1E-05	7.6E-08	7.8E-08	2.7E-08	2.7E-06	1.3E06	1.3E - 06	L/K
PYRENE	11	11	111E-08	2 78E -07	3 00E - 02	8 7E - 05	87E-05	84E-06	8.4E-08	4 1E-05	4.1E-05	1.0E – 05	1.0E-05	3.8E-08	3.6E - 06	1.8E-08	1.8E~06	KIDNEY
BENZO(A)ANTHRACENE	0 47	0 47	1 11E-06	2 78E - 07		0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	NS			
CHRYSENE	07	07	1 11E-08	2 78E -07		0 0E + 00	0.0E + 00	0.0E + 00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	NS			
BENZO(B)FLUORANTHEN	14	1.4	111E-08	2 78E -07		0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0 0E + 00	0.0E+00	0.0E+00	NS			
BENZO(A)PYRENE	0 32	0 32	1 11E-08	2 78E - 07		0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E + 00	0.0E + 00	0.0E + 00	0.0E+00	NS			
INDENO(123 - CD) PYRENE	0 24	0 24	1 11E-08	2 78E - 07		0 0E + 00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E + 00	NS			
BENZO(GHI)PERYLENE	027	0 27	1 11E-06	2 78E -07	4 00E 03	1 2E - 04	1 2E - 04	1 5E - 05	1 5E - 05	75E-05	7.5E~05	1 9E - 05	1.9E-05	6.6E - 06	0.6E~06	3.3E-06	3.3E - 08	NS
BARIUM	225	73 18	1 11E-08	0 00E + 00	5 00E - 02	8 2E - 03	2.7E-03	0.0E+00	0 0E + 00	5.0E - 03	1.6E-03	0.0E + 00	0.0E+00	4.4E-04	1.4E-04	0.0E + 00	0.0E+00	BLOOD
BERYLUUM	0 25	0 25	1 11E-08	0 00E + 00	5 00E - 03	# 1E - 05	9 1E - 05	0 0E + 00	0 0E + 00	5 6E - 05	5.6E - 05	0 0E + 00	0.0E+00	4.9E-06	4.9E ~ 08	0.0E+00	0 0E + 00	NS
COPPER	158	41 67	1 11E-06	0 00E + 00		0 0E + 00	0 0E + 00	0.0E+00	0 0E + 00	0.0E+00	0 0E + 00	0.0E+00	0.0E+00	0.0E + 00	NS			
MANGANESE	1935	698 83	1 11E-00	0 00E + 00	1 00E - 01	3 5E - 02	1 JE - 02	0 0E + 00	0 0E + 00	2.1E~02	7 8E - 03	0 0E + 00	0.0E+00	1.9E-03	8.8E-04	0.0E + 00	0.0E+00	CNS
NICKEL	18 1	8 78	1116-08	0 00E + 00	2 00E - 02	1 5E - 03	8 2E - 04	0 0E + 00	0 0E + 00	8.9E - 04	3.8E-04	0 0E + 00	0.0E+00	7.9E-05	3 3E - 05	0.0E+00	0 0E + 00	BW
SILVER	64	1.0	1 11E-08	0 00E + 00	3 00E - 03	3 9E - 03	1 2E - 03	0.0E+00	0 0E + 00	2.4E-03	7.0E-04	0.0E + 00	0.0E+00	2.1E-04	6 2E - 05	0.0E + 00	0.0E + 00	ARGYRIA
SELENUM	0 44	0 43	111E-08	0 00€ + 00	5 00E - 03	1 8E - 04	1 8E - 04	0.0E+00	0.0E+00	9.8E - 05	9.6E - 05	0.0E + 00	0.0E+00	8.6E06	8.4E-08	0.0E + 00	0.0E+00	SELENOSIS
VANADIUM	30 7	17 05	1 11E-00	0 00E + 00	7 00E - 03	# 0E - 03	4 4E - 03	0.0E+00	0.0E+00	4.9E - 03	2.7E-03	0.0E + 00	0.0E+00	4.3E-04	2.4E-04	0.0E+00	0.0E + 00	NS
ZINC	785	199 88	1 11E-00	0 00E + 00	2 00E - 01	7 OE ~03	1.8E-03	0.0E+00	0.0E+00	4.2E-03	1.1E-03	0.0E + 00	0.0E+00	3.7E-04	9.8E - 05	0.0E+00	0.0E+00	BLOOD

	TOXICITY ENDPOINTS AB	BREVIATIONS	NS: NOT S	PECIFIED	L/K: UVER		IEY	BW: BODY WE	GHT	CNS: CENTRAL	NERVOUS S	YSTEM	
ALL CHEMICALS OF CONCERN		1 BE-01 B 4E	-02 7.0E-04	2 3E - 04	8.9E-02	3.3E - 02	8.5E - 04	2.6E-04	7.8E-03	2.9E 03	1.5E-04	4.9E - 05	
	HAZARD INDEX	• 5E - 02 2 4E	-02 7 8E-05	7.9E - 05	3 OE - 02	1.5E-02	9 8E ~ 05	9.6E - 05	3.5E-03	1.3E - 03	1.7E-05	1.7E~05	
NYANZA SITE CONTAMINANTS	HAZARO INDEX	0 1E-02 29E	-02 0 2E-04	1 6E - 04	4 9E - 02	1.6E - 02	7.6E - 04	1.8E-04	4.3E-03	1.6E-03	1.3E-04	3.2E-05	

BEHP.BIS(2-ETHYL HEXYL)PHTHALATE

6-123

TABLE 8-338 RISK REBULTS FOR SEDIMENT EXPOSURE REACH NO 5 NYAN7A OPERABLE UNIT 3

1

MIDDLESEX COUNTY, MASSACHUSETTS

					FNIC RISK A		RESULTS		
CONTAMINANTS of	CONCENTRATION MG/KG		EXPOSURE	FACTOR		CANCER	NSKS		CANCER SLOPE FACTOR
CONCERN	MAX	AVG	INGESTION	DERMAL CONTACT	ACCIDEN		DERM Cont	AL ACT	(MG/KG/D) - 1 / WEIGHT
	. 1				MAX	AVG	MAX	AVG	EVIDENCE
	92	3 83	3 64E - 07	0 00E + 00	6 0E - 06	2.5E-06	0 0E + 00	0 0E + 00	1.80E+00 A
CHROMIUM*	60 6	17 95	3 64E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	
LEAD*	809	237 99	1 62E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
MERCURY*	41	0 98	3 64E - 07	1 51E-08	0 0E + 00	0.0E+00	0.0 E + 00	0.0E+00	
ВЕНР	0 33	0 33	3 64E - 07	1 51E - 07	1 7E - 09	1.7E-09	7 OE - 10	7.0E - 10	1.40E-02 B2
3/4METHYLPHENOL	0 13	0 13	3 64E - 07	1 51E - 07	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	
ACENAPTHYLENE	0 16	0 16	3 84E - 07	7 58E - 08	0 0F + 00	0 0E + 00	0 0E + 00	0.0E + 00	
PHENANTHRENE	0 54	0 54	3 84E - 07	7 56E - 06	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	
FLUORANTHENE	11	1 1	3 64E - 07	7 56E - 06	0 0E + 00	0 0E + 00	0.0E + 00	0.0E + 00	
PYRENE	11	1 1	3 64E - 07	7 56E - 08	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	
BENZO(A)ANTHRACENE	0 4 7	0 47	3 84E - D7	7 56E - 06	9 9E - 07	9 9E - 07	2 1E - 07	2. 1E - 07	5.80E+00 B2
CHRYSENE	07	07	3 64E - 07	7 56E - 06	1 5E - 08	1 5E - 06	3.1E-07	3.1E-07	5.80E+00 B2
BENZO(B)FLUORANTHEN	14	14	3 64E - 07	7 56E - 08	3 0E - 06	3 OE - 06	6.1E-07	6.1E-07	5.80E + 00 B2
BENZO(A)PYRENE	0 32	0 32	3 64E - 07	7 56E - 06	6 BE - 07	6 8E - 07	1 4E - 07	1.4E-07	5.80E+00 B2
INDENO(123-CD)PYRENE	0 24	0 24	3 84E - 07	7 56E - 06	5 1E - 07	5 1E - 07	1 1E - 07	1.1E-07	5.80E+00 B2
BENZO(GHI)PERYLENE	0 27	0 27	3 64E - 07	7 56E - 08	0 0E + 00	0 0E + 00	0.0E + 00	0.0E + 00	
BARIUM	225	73 18	3 64E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	
BERYLLIUM	0 25	0 25	3 64E - 07	0 00E + 00	3 9E - 07	3.9E - 07	0.0E + 00	0.0E + 00	4.30E+00 A
COPPER	158	41 67	3 64E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E + 00	
MANGANESE	1035	999 83	3 64E - 07	0 00E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E + 00	
NICKEL	16 1	6 79	3 64E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
SILVER	64	1.9	3 64E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E + 00	
SELENIUM	0 44	0 4 3	3 64E - 07	0 00E + 00	0 0E + 00	0.0E + 00	0.0E+00	0.0E+00	
VANADIUM	30 7	17 05	3 64E - 07	0 00E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E + 00	
ZINC	765	199 66	3 64E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0.0E + 00	0.0E + 00	

NYANZA SITE CONTAMINANTS	CANCER RISK	6 0E - 06	2.5E-06	0.0E + 00	0.0E + 00
OTHER SUDBURY RIVER CONTAMINANTS	CANCER RISK	7 OE - 06	7.0E-06	1.4E - 06	1.4E-06
ALL CHEMICALS OF CONCERN	CANCER RISK	1.3E - 05	9.5E-06	1.4E-06	1.4E-06

BEHP:BIS(2-ETHYL HEXYL)PHTHALATE

TABLE 6 - 34A RISK ASSESSMENT RESULTS FOR SURFACE WATER EXPOSURE SCENARIOS REACH NO 5 NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

······	1						NONCARC	NOGENC	PISK ANAL	YSIS RESUL	L TS							
CONTAMINANTS of CONCERN	CONCENTR/ MC/	KO KO	EXPOSURE RECEPTOR	FACTOR	AFD (MG - KG - DAY)	HAZA	RD QUOTI	ENTS: CHI	ம	HAZAF	id quotier	NTS: TEEN		HAZARI		ts: Adult		TOXICITY END - POINT
		~~~	INGESTION	DERMAL		ACCIDEN	TAL DN	DERM CONT/	AL VCT	ACCIDEN	TAL	DERMA	L ST	ACCIDENT	AL N	DERMAI	<u>.</u>	
						MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	
ARSENIC*	0 0011	0 00 105	1 11E-03	1 30E-04	3 00E-04	3 3E-03	3.2E-03	2.3E-04	2.2E-04	4.1E-03	3.9E~03	5.1E-04	4.8E-04	7.2E-04	6.8E04	2.8E-04	2.7E-04	SKIN
BARIUM MANGANESE	0 01275	0 01275 0 0 <b>59</b> 45	1 11E-03 1 11E-03	1 38E-04 1 38E-04	5 00E - 02 1 00E - 01	2 3E-04 9 7E-04	2.3E-04 8 3E-04	16E-05 86E-05	1.6E-05 4.4E-05	2 6E - 04 1.2E - 03	2.8E-04 7.7E-04	3.5E-05 1 5E-04	3.5E - 05 9.6E - 05	5.0E05 2.1E04	5.0E05 1.4E04	1.9E-05 8 0E-05	1.9E-05 5.3E-05	BLOOD CNS
NYANZA SITE CONTAMIN	I IANTS"		1	HAZARD IN	DEX	3 3E-03	3 2E - 03	2 38-04	2 2E-04	4.1E-03	3.9E-03	5.1E-04	4.8E-04	7.2E-04	6.8E-04	2.8E-04	2.7E-04	
OTHER SUDBURY RIVER	CONTAMINAN	ITS			DEX	1 2E-03	8 7E - 04	84E-05	01E-05	1 5E-03	1.1E-03	1.8E-04	1.3E-04	2.6E - 04	1.9E04	1.0E-04	7.2E-05	
ALL CHEMICALS OF COM	ICERN			HAZARD IN	DEX	4 5E - 03	4 1E-03	3 2E - 04	28E-04	5.5E-03	4.9E-03	6.9E-04	6.1E-04	9.7E-04	8.7E-04	3.8E-04	3.4E-04	j
BEPH: BIS/2ETHYL HEM	.)PHTHALATE				NOPOINTS A	BBREVIATIO	ONS:		PECIFIED	L/K: LIV OUS SYSTE	ier and Kid	NEY						

BEPH: BIS(2ETHYL HEXYL)PHTHALATE

W92194F

## TABLE 6-34B RISK RESULTS FOR SURFACE WATER REACH NO 5 NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

					CARCINO	GENIC RIS	<b>KRESULTS</b>	5	
CONTAMINANTS of CONCERN	CONCENTRA MG/	TION KG	EXPOSURE	FACTOR		CANCER	RISKS		CANCER SLOPE FACTOR
	MAX	AVG	INGESTION	DERMAL CONTACT	ACCIDEN INGESTIC	TAL DN	DERM CONTA	AL \CT	(MG/KG/D) /
					MAX	AVG	MAX	AVG	WEIGHT OF
ARSENIC*	0.0011	0.00105	3.02E-04	4.21E-05	6.0E-07	5.7E-07	8.3E-08	8.0E-08	EVIDENCE 1.80E+00 A
BARIUM	0.01275	0.01275	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
		0.00945	J.UZE-04	4.21E-00		0.0E+00	0.00+00		
NYANZA SITE CONTAMIN	W12-			SK	6.UEU/	5./E-0/	8.3E-08	8.UE08	
OTHER SUDBURY RIVER (	CONTAMINAN	TS		SK	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ALL CHEMICALS OF CON	CERN	<u></u>	CANCER R	SK	6.0E-07	5.7E-07	8.3E-08	8.0E-08	

BEPH: BIS(2ETHYL HEXYL)PHTHALATE

FINAL

#### 6.5.3.12 <u>Risk Assessment Results for Reach 6</u>

Sudbury River Reach 6 extends from the Massachusetts Turnpike to the Saxonville Dam at the Central Street crossing. Reach 6 includes ponded areas (depositional areas) behind the Saxonville Dam. Five Nyanza Site contaminants were detected in Reach 6 surface waters and sediments:

0	Arsenic	0	Lead
0	Cadmium	0	Mercury

o Chromium

The average mercury concentration detected in Reach 6 sediment samples is greater than 12 times the concentration detected in background samples.

Tables 6-35 and 6-36 present risk assessment results for COC concentrations detected in sediments and surface waters of Reach 6. The accidental-ingestion and dermal-contact routes of exposure (surface waters and sediments) were evaluated assuming recreational land/water use scenarios. The hazard quotients and hazard indices do not exceed unity in any of the cases presented. Only the hazard quotients presented for arsenic and mercury exceed 0.1 (accidental-exposure route, maximum concentrations, child receptor). If hazard indices are summed for the accidental-ingestion and dermal-contact exposure routes and then combined for surface water and sediment exposures, the total hazard index does not exceed unity. Thus, adverse noncarcinogenic health effects are not anticipated under the conditions of the recreational exposure scenarios defined in Section 6.4.

The excess lifetime cancer risks estimated for sediment exposures are 4.2 x 10⁻⁵ and 2.4 x 10⁻⁵ (risks summed for the accidentalingestion and dermal-contact exposure routes) when maximum and average contaminant concentrations are evaluated, respectively. As a point of reference, cancer risks estimated for background sediment concentrations were 1.8 x 10⁻⁵ and 8.2 x 10⁻⁶ for the reasonable maximum and average case, respectively. The COCs contributing to the risk are bis(2-ethylhexyl)phthalate, the carcinogenic PAHs, 4,4-DDD, beryllium, and arsenic, a Nyanza Site contaminant. The arsenic concentrations detected in Reach 6 sediments (average concentration ( $C_{re}$ =11.1 mg/kg)) are similar to those detected in background samples.

#### TABLE 8 - 35A RISK ASSESSMENT RESULTS FOR SEDIME: POSURE SCENARIOS REACH NO 8 NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

			-			NO	NCARC	INOGENIC RISK	NALYDIS RES	ULTS							
CONTAMINANT8 of CONCERN	CONCEN MG/	TRATION	EXPOSURE RECEPTO	FACTOR R = TEEN	RFD	HAZARD		IENTS CHILD	HA	ZARD QUO	TIENTS: TE	EN	HAZAR	DQUOTIENT	S: ADULT		TOXIC END- POINT
	MAX	AVG	INGESTION		(MG - KG -	ACCIDEN		DERMAL			DERM		ACCIE		DERI		]
			l		DAY	MAX	AVO	MAX AV	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	
ARSENIC*	24		1 11E-08	0 00E + 00	3 00E - 04	15E-01 68	IE - 02	0 0E+00 0 0E+	00 8.9E-02	4.1E-02	0.0E+00	0.0E+00	7.8E - 03	3.6E - 03	0.0E+00	0.0E + 00	SKIN
CADMIUM*	13.6	3 83	111E-08	0 00E + 00	5 00E - 04	50E-02 14	IE - 02	0 0E + 00 0 0E -	00 3.0E-02	8.5E-03	0.0E+00	0.0E+00	2.7E-03	7.5E-04	0.0E + 00	0.0E+00	KIDNEY
CHROMIUM*	281	86 6	1 11E-08	0 00E + 00	1 00E + 00	51E-04 12	E-04	0 0E + 00 0 0E -	00 3.1E-04	7.4E-05	0.0E + 00	0.0E+00	2.7E-05	6.5E - 08	0.0E+00	0.0E + 00	UVER
LEAD*	876	285 85	5 55E - 07	0 00E + 00		00E+00 00	)E + 00	0 0E+00 0 0E+	00 0 0E+00	0 0E + 00	0.0E + 00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E + 00	CNS
MERCURY*	178	33	1 11E-08	5 55E - 00	3 00E - 04	11E-01 20	)E - 02	2.7E-03 5.0E	04 8.5E - 02	1.2E-02	3.3E - 03	6.1E-04	5.7E-03	1.1E-03	5.7E – 04	1.1E-04	CNS
ACETONE	0 57	0 4013	1 11E-08	2 78E - 06	1 00E - 01	10E-05 73	E - 08	1 3E-05 9 2E-	08 8.3E-06	4.5E-08	1.6E - 05	1.1E-05	5.6E-07	3.9E 07	2.8E-08	2.0E - 08	L/K
ВЕНР	1.9	1 4187	1 11E-06	5 55E - 07	2 00E - 02	17E-04 1.3	E-04	4.3E-05 3 2E-	05 1.1E-04	7.9E-05	5.3E - 05	3.9E - 05	9.3E - 08	6.9E - 06	9.3E-06	6.9E - 08	UVER
PHENANTHRENE	18	1 1713	111E-08	2 70E - 07	4 00€ -03	73E-04 53	1E - 04	91E-05 87E-	05 4 4E - 04	3 3E - 04	1.1E-04	8.1E-05	3 9E ~ 05	2 9E - 05	2 0E - 05	1.4E-05	BW
FLUORANTHENE	28	1 875	1 11E-08	276E-07	4 00E - 02	12E-04 90	DE - 05	15E-05 11E-	05 7 2E - 05	i 5.5E-05	1.8E-05	1.4E05	6.4E-08	4 8E - 08	3 2E - 06	2.4E-08	L/K
PYRENE	2 🕈	2 075	1 11E-00	2 70E - 07	3 00E - 02	18E-04 1.3	BE - 04	2.2E-05 1 0E	05 1.1E-04	7.7E-05	2.7E-05	1.8E-05	9 5E - 06	6.8E - 06	4.7E-06	3.4E - 06	KIDNEY
BENZO(A)ANTHRACENE	15	1 0175	111E-08	2 70E - 07		00E+00 00	DE + 00	0 0E + 00 0 0E -	00 0.0E+00	0.0E+00	0 0E + 00	0.0E+00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	NS
CHRYSENE	15	1 3067	1 11E - 08	2 70E - 07		00E+00 00	DE + 00	0 0E+00 0 0E+	00 00E+00	0 0E+00	0 0E + 00	0.0E+00	0.0E+00	0 0E + 00	0 0E + 00	0.0E+00	NS
BENZO(B)FLUOHANTHEN	34	1 6263	1116-00	2786-07		00E+00 00	DE + 00	000000000000000000000000000000000000000		0.0E+00	0.0E+00	0.0E+00	0.000	0.000	0.05+00	0.0E+00	NS
BENZUKIF LUUHAN THEN		0 2003	1116-00	2700-07		000000000000000000000000000000000000000					0.02+00	0.02+00	0.02+00	0.02+00	0.02 +00	0.00 + 00	Ne
INDENO(122 - COLPARENE	074	0 6317	1116-08	2785-07		0.05+00 0.0		0.05+00 0.05		0.0E+00		0.02+00	0.0E + 00	0.0E + 00	0.0E+00	0.00+00	NG
BENZO/OHIPEBYLENE	0.82	0 8783	1115-00	2 78E - 07	4 00F - 03	37E-04 24	E - 04	47E-05 33E	05 23E-04	16F-04	5 7E-05	4 0F-05	2 0E - 05	14E-05	1 OF 05	7 1F-08	NS
BARIUM	143	85.6	1116-08	0 00E + 00	5 00E - 02	52E-03 3.1	E-03	0 0E+00 0 0E	00 3.2E-03	1.9E-03	0.0E+00	0.0E+00	2.8E-04	1.7E-04	0.0E+00	0.0E+00	BLOOD
BERYLLIUM	2	0 49	1 11E-08	0 00E + 00	5 00E - 03	73E-04 16	DE - 04	0 0E + 00 0.0E	00 4.4E-04	1.1E-04	0 0E + 00	0.0E+00	3.9E-05	9.6E - 06	0.0E+00	0.0E+00	NS
COPPER	303	113 5	1 11E-08	0 00E + 00		0 0E + 00 0.0	)E + 00	0 0E+00 0 0E+	00 0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	NS
MANGANESE	1415	589 3	1 11E-08	0 00E + 00	1 00E - 01	28E-02 1.1	E - 02	0 0E+00 0 0E+	00 1.8E-02	0.5E-03	0.0E+00	0.0E+00	1.4E-03	5.0E-04	0.0E + 00	0.0E + 00	CNS
NICKEL	78.4	20 3	1 11E-08	0 00E + 00	2 00E - 02	7 2E - 03 1 9	)E - 03	0 0E+00 0 0E+	00 4.4E-03	1.1E-03	0.0E+00	0.0E+00	3.6E-04	9.9E-05	0.0E + 00	0.0E+00	BW
SELENIUM	6 1	1 37	1 11E-06	0 00E + 00	5 00E - 03	2 2E - 03 5 0	DE - 04	0 0E + 00 0 0E -	00 1.4E~03	3.0E-04	0.0E+00	0.0E+00	1.2E-04	2.7E - 05	0.0E + 00	0 0E + 00	SELENOS
VANADIUM	\$3	55 🖷	1 11E-08	0 00E + 00	7 00E - 03	2 2E - 02 1.5	5E - 02	00E+00 00E+	00 1.3E-02	8 9E - 03	0.0E + 00	0.0E+00	1 2E - 03	78E-04	0.0E+00	0 0E + 00	NS
ZINC	564	292 3	1 11E-08	0 00E + 00	2 00E -01	52E-03 27	/E - 03	0 0E + 00 0 0E -	00 3.1E-03	1 8E-03	0.0E + 00	0.0E+00	28E-04	1 4E - 04	0 0E + 00	0.0E + 00	BLOOD
4.4-DOD	07	0 3994	3 33E - 07	2 78E - 07		0 0E + 00 0.0	DE + 00	0 0E+00 0 0E+	00 0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E + 00	0.0E + 00	0.0E+00	0.0E + 00	NS
NYANZA BITE CONTAMINA	NTS		• · · ·	HAZARD IN	<del>XEX</del>	30E-01 10	DE - 01	27E-03 50E-	04 1 8E-01	6 2E - 02	3 3E - 03	8.1E-04	1.6E 02	5 5E - 03	5.7E-04	1.1E-04	]
OTHER SUDBURY RIVER C		ANTS		HAZARD IN	жx	7 0E - 02 3.5	5E - 02	2 3E-04 1 7E-	04 4.2E-02	2.1E-02	2.6E-04	2.0E - 04	3.7E-03	1 9E - 03	5.0E - 05	3.6E - 05	
ALL CHEMICALS OF CONC	ZAN			HAZARD IN	жх	37E-01 1.4	E-01	2 8E-03 8.7E-	04 2.3E-01	8.3E-02	3.5E-03	8.2E-04	2.0E - 02	7.3E - 03	6.2E-04	1.4E-04	

BEHP:BIS(2-ETHYL HEXYL)PHTHALATE

W92194F

(

6-127

FINAL

Ţ

6-128

### TABLE 6 - 358 RISK RESULTS FOR SEDIMENT EXPOSURE REACH NO 6 NYANZA OPERABLE UNIT 3

#### MIDDLESEX COUNTY, MASSACHUSETTS

ļ				CARCINOGE	NIC RISK A	NALYSIS I	RESULTS		
CONTAMINANTS of	CONCEN		EXPOSURE	FACTOR			asks		SLOPE FACTOR
CONCERN	MAX	AVG	INGESTION	DERMAL CONTACT	ACCIDEN	TAL DN	DERM CONT/	AL ACT	(MG/KG/D) - 1 / WEIGHT
					MAX	AVG	MAX	AVG	OF EVIDENCE
ARSENIC	24		3 64E - 07	0 00E + 00	1.6E - 05	7.3E-06	0.0E+00	0 0E +00	1.80E+00 A
CADMIUM*	136	3 83	3 64E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E + 00	
CHROMIUM*	281	66 6	3 64E - 07	0 00E + 00	0.0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
LEAD*	876	285 65	1 82E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	
MERCURY*	176	33	3 64E -07	1 51E-08	0 0E + 00	0 0E +00	0.0E+00	0.0E+00	
ACETONE	0 57	0 4013	3 64E -07	7 56E - 07	0 0E + 00	0 0E + 00	0 0E + 00	0.0E +00	
ВЕНР	19	1 4 187	3 64E - 07	1 51E-07	97E-09	7 2E ~ 09	4.0E - 09	3 OE - 09	1.40E-02 B2
PHENANTHRENE	16	1 1713	3 64E - 07	7 56E - 08	0 0E + 00	0.0E+00	0.0E+00	0 0E + 00	
FLUORANTHENE	26	1 975	3 64E -07	7 56E - 06	0 0E + 00	0 0E + 00	0.0E+00	0 0E + 00	
PYRENE	29	2 075	3 64E - 07	7 56E - 08	0 0E + 00	0 0E + 00	0.0E+00	0.0E + 00	
BENZO (A) ANTHRACENE	15	1 0175	3 64E - 07	7 56F - 08	3 2E - 06	2 1E-06	6 6E - 07	4.5E-07	5.80E+00 B2
CHRYSENE	15	1 3067	364E 07	7 56E - 08	3.2E - 06	2 8E - 06	6.6E-07	5.7E-07	5.80E+00 B2
BENZO (B) FLUORANTHEN	34	1 6263	3 64E - 07	7 56E - 08	7 2E - 08	3.4E-08	1.5E-06	7.1E-07	5.80E+00 B2
BENZO(K)FLUORANTHEN	0 96	0 8833	364E-07	7 56E - 08	2 1E - 06	1.9E-06	4.3E-07	3 9E - 07	5.80E+00 B2
BENZO (A) PYRENE	11	0 7883	3 84E - 07	7 56E - 08	2 JE - 06	1.7E-06	4.8E-07	3.5E-07	5.80E+00 B2
INDENO(123-CD)PYRENE	0 74	0 5317	3 64E - 07	7 56E - 08	16E-08	1.1E-06	3 2E - 07	2.3E-07	5.80E+00 82
BENZO(GHI)PERYLENE	0 82	0 5763	3 64E - 07	7 56E - 08	0 0E + 00	0 0E + 00	0.0E+00	0 0E + 00	
BARIUM	143	85 8	3 64E - 07	0 00E + 00	0 0E + 00	0.0E+00	0.0E + 00	0.0E+00	
BERYLLIUM	2	0 49	3 64E - 07	0 00E + 00	3 1E - 06	77E-07	0.0E+00	0.0E+00	4.30E+00 A
COPPER	303	113 5	3 64E - 07	0 00E + 00	0.0E +00	0 0E + 00	0.0E+00	0.0E+00	
MANGANESE	1415	589 3	3 64E - 07	0 00E + 00	0.0E + 00	0 0E + 00	0.0E+00	0.0E+00	
NICKEL	78.4	20 3	3 64E - 07	0 00E + 00	0.0E +00	0 0E + 00	0.0E+00	0.0E+00	
SELENIUM	8 1	1 37	3 64E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	
VANADIUM	63	55 9	3 64E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E +00	0.0E+00	
ZINC	564	292 3	3 64E - 07	0 00E + 00	0.0E +00	0.0E+00	0.0E+00	0.0E+00	
4,4-DDD	07	0 3994	1 09E - 07	7 56E - 08	1.8E - 08	1.0E-08	1.3E-08	7.2E-09	2.40E-01 B2

NYANZA SITE CONTAMINANTS	CANCER RISK	1.8E-05	7 3E-06	0.0E+00	0.0E+00
OTHER SUDBURY RIVER CONTAMINANTS	CANCER RISK	2.3E-05	1.4E-05	4.1E-06	2.7E-06
ALL CHEMICALS OF CONCERN		3.8E-05	2.1E-05	4.1E-06	2.7E-08

BEHP:BIS(2-ETHYL HEXYL)PHTHALATE

(

1

·····	1						NONCAR	INOGENC	HISK ANAL	YSIS RESUL	.19							
CONTAMINANTS of CONCERN	CONCENTR MOV	ATION KQ	EXPOSURE RECEPTOR	FACTOR	AFD (MG KG -	HAZA	RID QUOTII	ENTS. CHI	ם	HAZAF	RD QUOTIEN	ITS: TEEN		HAZARI		TS: ADULT		TOXICITY END- POINT
	MAX	AVO.	INGESTION	DEFINAL	DAY)	ACCIDEN	ITAL ON	DERM	AL ICT	ACCIDEN	TAL	DERMA	L ST	ACCIDENT	AL N	DERMA		
						MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	
LEAD*	0 0024	0 0014	1 11E-03	1 38E-04		0 0E+00	0 0E+00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0 0E + 00	0.0E+00	0.0E+00	CNS
BARIUM MANGANESE	0 0162	0 0048 0 0964	1 11E-03 1 11E-03	1 30E-04 1 30E-04	5 00E-02 1 00E-01	3 0E - 04 1 5E - 03	8 8E ~ 05 8 8E ~ 04	2.1E-05 1.1E-04	8.1E-06 8.1E-05	3.6E-04 1.9E-03	1.1E-04 1.1E-03	4.5E-05 2.3E-04	1.3E-05 1.3E-04	€.3E~05 3.3E-04	1.9E05 1.9E04	2.5E-05 1.3E-04	7.3E08 7.3E05	BLOOD CNS
NYANZA SITE CONTAMIN	i NANTS		I	HAZAPO IN	DEX	0 0E+00	0 0E+00	0 0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
OTHER SUDBURY RIVER	CONTAMINA	11		HAZARD IN	DEX	1 8E - 03	97E-04	1 3E-04	6 8E - 05	2 2E - 03	1.2E-03	2.7E-04	1.5E-04	3.9E-04	2.1E-04	1.5E-04	8.0E-05	
ALL CHEMICALS OF COM	ICERN			HAZARD IN	DEX	1 8E - 03	97E-04	1 3E-04	6 8E - 05	2 2E - 03	1.2E03	2.7E-04	1.5E-04	3.9E-04	2.1E-04	1.5E-04	8.0E~05	
BEPH: BIS(2ETHYL HEXM	L)PHTHALATE			TONCITY E	NDPOIN 19 A	BOREVIATK	ONS.	NS: NOT S	PECIFIED	L/K: LIV OUS SYSTE	er and Kid M	NEY						

BEPH: BIS (2ETHYL HEXYL)PHTHALATE

.

## TABLE 6-36B RISK RESULTS FOR SURFACE WATER REACH NO 6 NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

					CARCINO	GENIC RIS	K RESULTS	S	
CONTAMINANTS of CONCERN	CONCENTR/ MG/	ATION KG	EXPOSURE	FACTOR		CANCER	RISKS		CANCER SLOPE FACTOR
	MAX	AVG	INGESTION	DERMAL CONTACT	ACCIDEN	TAL DN		AL ACT	(MG/KG/D)
					MAX	AVG	MAX	AVG	WEIGHT OF EVIDENCE
LEAD*	0.0024	0.0014	3.02E-04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
BARIUM	0.0162	0.0048	3.02E - 04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MANGANESE	0.167	0.0 <b>964</b>	3.02E - 04	4.21E-05	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
NYANZA SITE CONTAMI	ANTS*		CANCER R	ISK	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
	CONTAMINAN	ITS	CANCER R	ISK	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ALL CHEMICALS OF CON			CANCER R	ISK	0.0E+00	0.0E+00	0.0E+00	0.0E+00	

BEPH: BIS(2ETHYL HEXYL)PHTHALATE

#### FINAL

The Saxonville impoundment is located within Reach 6 of the Sudbury River basin. The maximum concentration of mercury in the fish tissue samples taken from the impoundment slightly exceeds the FDA Action Level for mercury in fish. Risk assessment results for the fish tissue samples taken are presented in Table 6-37. Hazard quotients calculated for mercury, methyl mercury, and thallium exceed unity in one or more cases except when average contaminant concentrations a sport fishers man is the receptor of concern. presented for all COCs exceed unity for all cases presented are evaluated and Hazard indices presented. It should be noted; however, that thallium was detected very infrequently in fish tissue samples collected from the Sudbury River Study Area. Based on the results presented on Table 6-37, adverse noncarcinogenic health effects are anticipated for the receptors of concern due principally to the mercury and methyl mercury concentrations detected in the fish tissue samples. Adverse noncarcinogenic health effects are anticipated for the sports fisherman only when maximum COC concentrations are evaluated.

Several pesticides and PCBs contribute to the cancer risks estimated for the fish tissue samples collected from the Saxonville impoundment. Cancer risks summed for all COCs exceed 1 x  $10^{-5}$  in all cases presented and approach 1 x  $10^{-3}$  when the subsistence fisherman is evaluated as the receptor of concern. However, the majority of the risk is attributable to COCs that are not Nyanza site-specific contaminants. Cancer risks associated with arsenic, the only Nyanza Site contaminant contributing to the overall cancer risk, do not exceed 5 x  $10^{-5}$  in any case presented in Table 6-37.

#### 6.5.3.13 Risk Assessment Results for Reach 7

As described in Section 1.0, Reach 7 extends from the Saxonville Dam downstream to the Route 20 overpass in Wayland. Downstream of the Saxonville Dam, the low stream gradient results in a slow meandering river with high depositional potential. Five Nyanza Site contaminants were detected in the Reach 7 sediments:

0	Arsenic	0	Lead
0	Cadmium	0	Mercury
0	Chromium		-

The average mercury concentration detected in Reach 7 sediments is greater than five times the concentration detected in background sediment samples.

#### TABLE 0-37 RISK ASSESSMENT RESULTS FOR FISH INGESTION EXPOSURE SCENARIOS SAXONVILLE IMPOUNDMENT NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

						NONCARCIN	IOGENIC RISK AN	VALYSIS RESULT	19		Ţ		CARCINOG	ENIC RISK A	NALYSIS RE	SULTS	
CONTAMINANTS OF CONCERN		FATION MG	EXPOSURE	FACTOR	RFD (MG		HAZARD QU	OTIENTS		TOXICITY END- POINT	EXPOSURE	FACTOR	CA	NCER RISKS	•		CANCER SLOPE
	MAX	AVG		80896-	DAY)		·····				SPORTS	SUBSIS-	SPORT FI	SHEPIMEN	SUBSIST	ENCE	(MG/KG/D)-1
				TENCE	1	<b>BPORTBF</b>	ISHERMEN	SUBS4	STENCE		FISHER-	TENCE		<b></b>	FISHERI	WEN	/
	1	ļ	MEN	MEN		MAXIMUM	AVERAGE	MAXIMUM	AVERAGE		MEN	MEN	MAXIMUM	AVERAGE	MAXIMUM	AVERAGE	EVIDENCE
		•						·					· · · · ·				
NAPHTHALENE	10	0 12925	1 KE-04	1 36E - 03	4 00E - 03	8 E - 02	6 0€ - 03	64E-01	4 € - 02	BW	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
PHENOL*	0 12	0 04475	1475-04	1 3E - 00	6.00E-01	376-05	1.42 -00	276-04	1.0E-04	FETUS	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ANTIMONY	0 14	0 14	1 #26 - 04	1 302 -03	4 00E - 04	6 GE - 02	612-02	4 72-01	4 7E-01	BLOOD	7.93E -00	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ARSENIC"	0 048	0.04	1000 - 00	1 38E - 03	3 00E - 04	2 0E - 02	2 52 -02	216-01	1 6E - 01	SKIN	7.93E-00	5.81E-04	8.6E-08	5.7E08	4.6E-05	4.2 - 05	1.80E+00 A
CHROMIUM*	4 278	0 663	1852-06	1 3E -03	1 00E + 00	7 6E - 04	12E-04	56E-03	8 9E 04	LIVER	7.93E-00	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
LEAD*	0 69	0 031	1 85E - 04	1 34E - 03		0 0E+00	0 0E + 00	0 0E + 00	0.0E+00	CNS	7.93E-05	5.81E04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MERCURY*	1.	0 63	1 80E - 04	1 36E ~ 03	300E-04	11€+00	362-01	8 1E+00	2.6E+00	CNS	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
METHYL MERCURY*	1 37	0 40285	1 80E - 04	1 30E - 03	3 00E - 04	●Æ-01	2 5E -01	8 2E + 00	1.6E+00	CNS	7.93E05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
верн	0.00	0 10000	1 85E-04	1 3E - 03	2 00E-02	€ <b>∑</b> -03	1 6E - 03	4 EE - 02	1 1E-02	LIVER	7.93E05	5.81E-04	7.5E-07	1.9E-07	5.5E-08	1.4E-08	1.40E~02 B2
ENDOSULFAN I	0 0015	0 00125	1 85E-04	1 36E - 03	5 00E-05	6 5E - 03	4 <del>CE</del> - 03	4 1E-02	3.4E-02	KIDNEY	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ENDOSULFAN SULFATE	0 001	0 001	1 85E - 04	1 3E -03	5 00E-06	376-03	3 Æ - 03	27E-02	27Ë-02	NS	7.93E - 05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
DIELORIN	0 0005	0 00085	1 85E-04	1 3E - 03	5 00E-05	1 6E~03	9 2E - 04	1 4E - 02	6 6E - 03	LIVER	7.93E-05	5.81E-04	6.3E-07	3.2E-07	4.6E-08	2.3E-08	1.60E+01 B2
4,4-000	0 014	0 00466	1 85E - 04	1 3E - 03	1	0 0E+00	0 0E + 00	0 0E + 00	0 0E + 00	LIVER	7.93E-05	5.81E-04	2.7E-07	8.9E-08	2.0E08	6.5E-07	2.40E-01 82
44-DDE	0 02	0 0074	18E-04	1 36E - 03	1	0 0E+00	0 0E + 00	0 0E+00	0 0E+00	LIVER	7.93E-05	5.81E-04	5.4E-07	2.0E-07	4.0E08	1.5E-08	3.40E-01 B2
4,4DOT	0 002	0 00133	1 80E - 04	1 3E - 03	1	0 0E+00	0 0E + 00	0 0E + 00	0 0E+00	LIVER	7.93E-05	5 5.81E-04	5.€-08	3.6E~08	4.0E-07	2.6E07	3.40E-01 Ba
ALPHA-CHLORDANE	0 002	0 002	1 85E-04	1 36E - 03	0 00E - 05	0 <b>₹</b> -03	62E-03	4 5E - 02	4 5E-02	LIVER	7.93E-05	5 5.81E-04	2.1E-07	2.1E-07	1.5E08	1.5E-08	1.30E+00 B
GAMMA-CHLORDANE	0 001	0 00084	1 85E - 04	1 36E - 03	6 00E - 05	3 1€ - 03	2 Œ -03	2 3E - 02	1 € -02	LIVER	7.93E-06	5.81E-04	1.0E-07	8.6E-08	7.6E-07	6.3E-07	1.30E+00 Ba
HEPTACLOR	0 000	0.00100	1 80E -04	1 3E -03	5 00E-04	2 E-03	8 Æ -04	1 E-02	4 6E - 03	LIVER	7.93E-05	5.81E-04	2.1E-08	8.0E 07	1.6E05	4.4E~06	4.50E+00 Ba
AROCLOR-1254	0.04	0 0201	1 80E -04	1 36E - 03		0 0E+00	0 0E + 00	0 0E + 00	0 0E+00	NS	7.93E-05	5.81E-04	2.4E-05	1.22-05	1.6E-04	9.0E-05	7.70E+00 82
AROCLOR-1280	0 11	0 04667	1 85E -04	1 36E - 03		0 0E+00	0 0E + 00	0 0E + 00	0 0E+00	NS	7.93E-05	5.81E04	6.7E~05	2.9E-05	4.9E-04	2.1E-04	7.70E+00 82
COPPER	01	0 384	1 80E -04	1 36E - 03		0 0E+00	0 0E + 00	0 0E+00	0.0E+00	NS	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MANGANESE	107	2 108	1 8E-04	1 36E - 03	1 00E-01	3@-02	3 9E - 03	2 TE-01	29E-02	CNS	7.93E05	5 6.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
NICKEL	1 24	0 23	1 85E-04	1 SE-03	200E-02	1 1E-02	2.1E-03	84E-02	1.6E-02	8W	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	1
SELENIUM	0 62	0 184	1 85E-04	1 36E - 03	5 00E - 03	8 E-02	6.1E-03	1. 元~01	4.4E-02	SELENOS	S 7.93E-05	5 5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
SILVER	0 24	0 082	1 80E - 04	1 JE - 03	3 00E - 03	1 52-02	3 Æ-03	1.1E-01	2 € -02	ARGYRIA	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
THALLIUM	1 10	0 065	1 BE -04	1 36E - 03	7 00E - 05	3 1€+00	1.EE-01	2 3E+01	1.1E+00	NS	7.93E-05	5 5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
VANADIUM	1 218	0 427	1 8E-04	1 3E -03	7 00E03	325-02	1.1E-02	2.4-01	0.3E-02	NS	7.93E-05	5.61E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ZINC	84 783	18 016	1 BEE - 04	1 345 - 03	2 00E-01	8 E - 02	1.7E-02	<b>8 €</b> -01	1. <b>Z</b> -01	BLOOD	7.93E-08	5.81E-04	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	
NYANZA SITE CONTAMIN	ANT8*		HAZAPID IN	DEX		150	4 <b>E</b> -01	● <b>5E</b> + 00	5 4E+00		CANCERF	19K9:	8.6E-08	5.7E-08	4.E-05	4.2E-05	7
OTHER SUDBURY RIVER	CONTAMIN	etru	HAZARDIN	DEX		3€+00	<b>2 1€</b> -01	2 E + 01	1.5E+00			ISKS:	9.6E – 05	4.3E-05	7.1E-04	3.1E-04	
ALL CHEMICALS OF CON	CERN	-	HAZARD IN	DEX		4 7 =+ 00	€ Œ -01	3 Æ +01	8 9E+00		CANCER R	19K9:	1.0E - 04	4.6E-05	7.5E04	3.5 <u>E</u> -04	i n
BEPH:BIB12-ETHYL HEX	ሲምተጡ <del>ለል</del> Lል	TĒ	TONOTTY E	NDPOINT AI	DEFENATION	• N U	8 NOT SPECIFIE IK LIVER AND KID	d Bw Body DNEY	WEIGHT C	NS: CENTRAL	NERVOUS 8	YSTEM					INAL

W92194F

6-132

Tables 6-38 and 6-39 present risk assessment results for COC concentrations detected in sediments and surface waters of Reach 7. The hazard quotients and hazard indices do not exceed unity in any case presented. Also, if hazard indices are summed for the accidental-ingestion and dermal-contact exposure routes and then combined for surface water and sediment exposures, the total hazard index does not exceed unity. Thus, adverse noncarcinogenic health effects are not anticipated under the conditions of the recreational exposure scenarios defined in Section 6.4.

Excess lifetime cancer risks estimated for maximum and average COC concentrations in Reach 7 sediments are approximately two times those estimated for background. The carcinogenic PAHs, beryllium, Aroclor-1254, and arsenic are the principal COCs contributing to the estimated cancer risk. Arsenic is the only Nyanza Site contaminant contributing to risk. However, the average arsenic concentration in Reach 7 sediments ( $C_{avg}$ =10.13 mg/kg) is similar to background arsenic concentrations ( $C_{avg}$ =8.74 mg/kg).

## 6.5.3.14 Risk Assessment Results for Reach 8

Sudbury River Reach 8 is a meandering stretch of river with extensive bordering swamps and marshes. The Reach extends from the Route 20 overpass in Wayland to the area of the Route 117 overpass, upstream of the inlet at Fairhaven Bay. Reach 8 includes the Great Meadows National Wildlife Refuge. Five Nyanza Site contaminants were detected in the Reach 8 sediments:

0	Arsenic	0	Lead
0	Cadmium	0	Mercury
0	Chromium		-

The average mercury concentration detected in Reach 8 sediments is approximately six times the concentration detected in background samples. Organic contaminant analyses were not conducted in Reach 8 sediment samples.

Table 6-40 presents risk assessment results for COC concentrations detected in sediments of Reach 8. The accidental-ingestion and dermal-contact routes of exposure (surface waters and sediments) were evaluated assuming recreational land/water use scenarios. (No surface water samples were collected in this Reach.) The hazard quotients and hazard indices calculated do not exceed unity in any of the cases presented. Also, if hazard indices are summed for the accidental-ingestion and dermal-contact exposure routes and then combined for surface water and sediment exposures, the total hazard

6-133

#### TABLE 6-36A RISK ASSESSMENT RESULTS FOR SEDIMENT EXPOSURE SCENARIOS REACH NO 7 NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

							NONCARC	NOGENC	RISK ANA	LYSIS RESI	JLTS							
CONTAMINANTS of CONCERN			EXPOSURE RECEPTO	FACTOR R = TEEN	RFD	HAZ		TIENTO: CH	\$LD	HAZ	ARD QUO'	TIENTS: TE	EN	HAZAF	ID QUOTIENT	S: ADULT		TOXIC END POINT
	MAX	AVG	INGESTION	DERMAL	(MG KG			DE	RMAL NTACT	ACCIDEN	TAL DN	DERM	IAL ACT	ACCII INGI	DENTAL	DERI	MAL ACT	
		L		1	DAY	MÁX	AVÓ	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	1
	40.0	10.13	1.115-04	0.005.00	3.005-04	2 66 - 01	A 25	0.05 + 00	0.05+00	1.6501	3.05-03	0.05+00	0.05 + 00	1 35 02	3 35 - 03	0.05+00	0.05 .00	OKIN
	17.0	2 2 2 1	1116-00	0.002+00	5 00E -04	# 5E _ 02	815-02	0.05+00	0.05+00	4 0E -02	4 0E 02	0.02+00	0.02+00	1.3E-02	4 3E-04	0.00 + 00	0.02+00	KIDNEY
CHROMILIM	200	47.04	1116-00	0 000 + 00	1.005+00	3 65 - 04	8.1E-05	0.00 + 00	0.00 + 00	235-04	5 25 - 05	0.02+00	0.00400	2.00-05	4.75_04	0.00 + 00	0.00 + 00	
	804	117 04	8 885 - 07	0.005+00	1.002 + 00	0.05+00	0.0E + 00	0.02+00	0.00 + 00	2.3E-04	0.0E±00	0.02+00	0.00+00	2.02-03	4.7C-00	0.02+00	0.02+00	CNR
MERCURY*		1 62	1115-06	5 55E - 08	3 00E - 04	3 3E - 02	9 3E -03	8 45 -04	2 35 - 04	2.0E-02	5.6E - 03	105-03	2 8E - 04	1.8E_03	5.0E -04	1.8E-04	5 0E - 05	CNB
MERCONT	1 33			0 000 - 00		000-01	U.UL - UJ	0.42 - 04	1.0L - 04		0.0L - 00	1.02-03	2.02-04	1.02 -03	0.02 04	1.02 -04	5.02-05	0.00
ACETONE	1.0	0 871	111E-08	2 78E - 08	1 00E-01	2 9E - 05	1 8F - 05	3 7E - 05	2 OF 05	1 8F - 05	97E-08	4 4E-05	2 4E - 05	1 6F - 08	8 5E - 07	7 8F 08	4 3E-08	
BEHP	1 11	0 846	1 11E-06	5 55E - 07	2 00E - 02	1 OE -04	7.7E-05	2 5E ~05	19E-05	6.1E-05	4.7E-05	3.1E-05	2.3E-05	5.4E-06	4.1E-08	5.4E-08	4.1E~08	UVER
2 - METHYLNAPHTHALENE	0 076	0 078	1 11E-08	2.78E-07		0 0E + 00	0 0E+00	0 0E + 00	0 0E + 00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	BW
PHENANTHRENE	12	0 593	1 11E-08	2 78E-07	4 00E - 03	5 5E - 04	2.7E-04	8 8E - 05	3.4E-05	3.3E-04	1.8E-04	8.3E05	4.1E-05	2.9E-05	1.5E-05	1.5E + 05	7.3E-08	BW
FLUORANTHENE	1 10	0 853	1 11E-08	2 78E - 07	4 00E-02	87E-05	4.4E-05	1.1E-05	5.4E-06	5.3E-05	2.6E-05	1.3E - 05	6.6E-06	4.6E-06	2.3E-06	2.3E-08	1.2E-08	L/K
PYRENE	1 17	0 676	1 11E-06	2 78E - 07	3 00E - 02	1 OE 04	5 3E - 05	1.3E-05	67E-06	8.3E~05	3.2E - 05	1.8E-05	8.1E-06	5.5E-08	2.9E ~ 06	2.8E - 06	1.4E-06	KIDNEY
BENZO(A)ANTHRACENE	0 88	0 43	111E-08	2 78E - 07		0 0E + 00	0 0E + 00	0.0E+00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	NS
CHRYSENE	12	0 64	1 11E-08	2 78E - 07		0 0E + 00	0 0E + 00	0.0E+00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	0.0E + 00	NB
BENZO(B)FWORANTHEN	E 0.91	0 518	1 11E-08	2.78E-07		0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E + 00	0.0Ë+00	0.0E + 00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	NS
BENZOKOFLUOHANTHEN	È 085	0 448	1 11E-06	2 78E - 07		0 0E + 00	0.0E + 00	0.0E + 00	0.0E + 00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	NS
BENZO(A)PYRENE	0 86	0 72	1 11E-06	278E-07		0 0E + 00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	NS
INDENO(123-CD)PYRENE	0 64	0 54	1.11E-08	2.78E-07		0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0 <b>E + 00</b>	0.0E+00	0.0E+00	0.0 <b>E + 00</b>	0.0E+00	0.0E + 00	0.0E+00	NS
BENZO(GHI)PERYLENE	0 61	061	1.11E-08	2 70E - 07	4 00E - 03	2 8E - 04	2 8E ~ 04	3 5E - 05	3.5E-05	1.7E-04	1.7E-04	4.2E-05	4.2E-05	1.5E-05	1.5E-05	7.5E-08	7.5E-08	
BARIUM	272	82 42	1.11E-08	0 00E + 00	6 00E - 02	● 9E – 03	3 4E - 03	0.0E+00	0 0E + 00	8.0E - 03	2 1E-03	0.0E + 00	0.0E+00	5.3E-04	1.8E-04	0.0E+00	0.0E+00	BLOOD
BERYLUUM	0.01	0 57	1.11E-08	0 00E + 00	8 00E - 03	2 2E - 04	2.1E-04	0 0E + 00	0 0E + 00	1.4E-04	1.3E-04	0.0E + 00	0.0E+00	1.2E-05	1.1E05	0.0E + 00	0.0E + 00	N9
COPPER	270	85 84	1 11E-08	0 00E + 00		0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	0.0E + 00	0.0E + 00	0.0E+00	0.0E + 00	0.0E + 00	0.0E + 00	0.0E + 00	NS
MANGANESE	1430	300 20	1 11E-08	0 00E + 00	1 00E - 01	2 6E - 02	6.7E-03	0 0E + 00	0.0E + 00	1.8E-02	4.1E-03	0.0E+00	0.0E+00	1.4E-03	3.6E-04	0.0E+00	0.0E+00	CNS
NICKEL	44 1	11 29	1 11E-08	0 00E + 00	2 00E - 02	4 0E - 03	1 OE - 03	0 0E + 00	0 0E + 00	2.4E-03	8.3E-04	0.0E + 00	0.0E+00	2.2E - 04	5.5E – 05	0.0E+00	0.0E+00	BW
SELENUM	72	1.18	1 11E-06	0 00E + 00	5 00E - 03	2 8E - 03	4.3E ~ 04	0 0E + 00	0 0E + 00	1.6E-03	2.6E - 04	0.0E + 00	0.0E+00	1.4E-04	2.3E-05	0.0E + 00	0.0E + 00	SELENOSIS
VANADIUM	47	18 55	1.11E-08	0 00E + 00	7 00E - 03	1 2E - 02	5.1E-03	0.0E + 00	0 0E + 00	7 5E - 03	3.1E-03	0.0E + 00	0.0E+00	6.8E-04	2.7E-04	0.0E + 00	0.0E+00	NS
ZINC	040	170 06	1 11E-06	0 00E + 00	2 00E - 01	6 9E - 03	1 8E - 03	0 0E + 00	0 0E + 00	3.6E - 03	9.4E - 04	0.0E + 00	0.0E+00	3.2E-04	8.3E-05	0 0E + 00	0.0E + 00	BLOOD
CHLORDANE	0.010	0.018	3 33E - 07	2 70E - 07	6 OOE - 05	1 0E - 04	1 8E - 04	6 8E - 05	8 8E - 05	1 0E - 04	1.0E-04	8 3E - 05	8.3E-05	6.0E-06	8.6E-08	1.5E-05	1.5E-05	
AROCLOR 1254	L0 <u>61</u>	0 51	3 33E -07	0 00E + 00	L 1	0 0E + 00	0.0E+00	0 0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	โฟล
NYANZA SITE CONTAMIN						3.85 - 01	7.05 02	8 45 . 04	2 15 - 04	2 15 - 01	4.85	1.05-03	285-04	1 05 - 02	425-02	1.95-04	5.05-05	ו
INTANZA SITE CONTAMINA	10			TIAL AND IN	UG A	3 32 -01	1.02-02	0.42 -04	# JE - 04	2.1E-01	4.0C - UZ	1.02-03	2.85,~04	1.85-02	<b>₹.2C ~ U3</b>	1.05 - 04	5.02-05	

	TOXICITY ENDPOINTS A	88REVIATIONS	NS: NOT SPECIFIED		Y BW: BODY	WEIGHT	CNS: CENTRA	LNERVOUS	SYSTEM
ALL CHEMICALS OF CONCERN	HAZARD INDEX	4 1E-01 0.8E-	02 1 1E - 03 4 2E - 04	25E-01 6.0E-02 1.	3E-03 5.1E-04	2.2E - 02	5.3E-03	2.3E-04	9.0E - 05
OTHER SUDBURY RIVER CONTAMINANTS	HAZARD INDEX	0 2E-02 1 9E-	02 28E-04 19E-04	3.8E-02 12E-02 3.	1E-04 2.3E-04	3 3E - 03	1.0E-03	5.5E - 05	4.0E-05

BEHP:BIS(2-ETHYL HEXYL)PHTHALATE

W92194F

6-134

6-135

#### TABLE 6 - 388 RISK RESULTS FOR SEDIMENT EXPOSURE REACH NO 7 NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

1				CARCINOGE	INC FISK A	NALYSIS	RESULTS	,	
CONTANUNANTO	CONCEN	TRATION	EVECAUDE	FACTOR			Neve		CANCER
of	MON	1001	EAFOBUIE	TACION			4313		FACTOR
CONCERN		<u></u>		r+		• • • • • •		··· ·	(MG/KG/D)-1
			INGESTION	DERMAL	ACCIDEN	TAL	DERM	AL	1
	MAX	AVQ		CONTACT	INGESTIC	N .	CONT	ACT	WEIGHT
				{					OF
	ł			!	MAX	AVG	MAX	AVG	EVIDENCE
	40 6	10 13	3 64E-07	0 00E + 00	27E-05	6 6E-06	0 0E + 00	0 0E + 00	1.80E+00 A
CADMIUM*	17 9	221	3 64E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
CHROMIUM*	209	47 95	3 64 [-07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
LEAD*	526	117 94	1 82E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
MERCURY.	55	1 52	3 64E - 07	1 51E-08	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
ACETONE	16	0 871	3 64E - 07	7 56E-07	4 4E-09	2 4E-09	9 1E-09	4 9E - 09	7.50E-03 B
BEHP	11	0 646	364E-07	1 518-07	5 6E-09	4 3E-09	2 3E-09	1 8E-09	1.40E-02 B
2-METHYLNAPHTHALENE	0 078	0 078	3 64E - 07	7 56E - 08	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
PHENANTHRENE	12	0 593	3 64E-07	7 56E - 08	0 OE + 00	0 0E+00	0 0E + 00	0 0E + 00	
FLUORANTHENE	19	0 953	3 64E - 07	7 56E-08	0 0E + 00	0 0E+00	0 0E + 00	0 0E + 00	
PYRENE	17	0 876	3 64E - 07	7 56E - 08	0 0E + 00	0 0E + 00	0 0E+00	0 0E + 00	
BENZO(A)ANTHRACENE	0 88	0 43	3 64E - 07	7 56E-00	19E-08	91E-07	3 9E-07	19E-07	5 80E + 00 B
CHRYSÈNE	12	0 64	3 64E - 07	7 56E-08	2 5E-06	14E-06	5 3E-07	2 8E-07	5.60E+00 Ba
BENZO(B)FLUORANTHEN	0 91	0 516	3 64E - 07	7 56E-08	1 9E-06	1 1E-06	4 0E - 07	2 3E-07	5.80E+00 B
BENZO(K)FLUORANTHEN	0 85	0 4 4 6	2 15E - 07	5 15E-08	1 1E-06	56E-07	2 5E-07	1.3E-07	5.80E + 00 B2
BENZO(A)PYRENE	0 66	0 72	3 64E-07	7 56E-06	19E-06	1 SE-06	3 9E - 07	3 2E - 07	5.80E+00 B
INDENO(123-CD)PYRENE	0 54	0 54	3 64E - 07	7 56E - 08	1 1E-08	1 1E-08	2 4E-07	24E-07	5 80E+00 B
BENZO(GHI)PERYLENE	061	0 61	3 64E - 07	7 56E - 08	0 OE + 00	0 0E + 00	0 0E + 00	0 0E + 00	
BARIUM	272	92 42	3 64E-07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
BERYLUUM	061	0 57	<b>J 64</b> E - 07	0 00E + 00	9 5E-07	8 9E-07	0 0E + 00	0 0E + 00	4.30E+00 A
COPPER	278	65 94	3 64E - 07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	
MANGANESE	1430	366 29	3 64E-07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	
NICKEL	44-1	11 29	3 64E-07	0 00E + 00	0 0E+00	0 0E+00	0 0E + 00	0.0E+00	
SELENIUM	72	1 10	3 64E-07	0 00E + 00	0 0E+00	0 0E + 00	0 0E + 00	0 0E + 00	
VANADIUM	47	19 55	3 64E-07	0 00E + 00	0 0E+00	0 0E + 00	0 0E + 00	0.0E+00	
ZINC	646	170 06	3 64E-07	0 00E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	
CHLORDANE	0 018	0 016	1 00E-07	7 56E-08	285-09	2 8E-09	1 8E-09	1 8E~09	1.30E+00 B
AROCLOR 1254	0 51	0 51	1 09E-07	0 00E + 00	4 3E-07	4 3E-07	0 0E + 00	3.0E-07	7.70E+00 B2

NYANZA SITE CONTAMINANTS	CANCER RISK	27E-05	8 6E-06	0 0E + 00	0 0E + 00
OTHER SUDBURY RIVER CONTAMINANTS	CANCER FISK	1 2E-05	7.9E-06	2.2E-06	17E-08
ALL CHEMICALS OF CONCERN		3 8E-05	1.5E-05	2.2E-06	1.7E~06

BEHP:BIS(2-ETHYL HEXYL)PHTHALATE

,

## TABLE 0 - 39A RISK ASSESSMENT RESULTS FOR SURFACE WATER EXPOSURE SCENARIOS REACH NO 7 NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

							NONCARC	NÓGENC	RISK ANAL	YSIS RESUL	.19							
CONTAMINANTS of CONCERN	CONCENTRI MGV	KG	EXPOSURE RECEPTOR	FACTOR	AFD (MG KG	HAZA		ENTS: CHI	ш	HAZAP		NTS: TEEN		HAZAR		rs: Adult		TOXICITY END- POINT
	MAX	AVG	INGESTION	DERMAL	DAY)	ACCIDEN	TAL	DERM CONT/	AL VCT	ACCIDEN	TAL DN	DERMA	L	ACCIDENT	AL N	DERMA	Т	
		ĺ				MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	
BARIUM	0.0213	0 0074	1 11E-03	1 30E-04	5 00E - 02 1 00E - 01	39E-04	14E-04	27E-05	94E-08 47E-05	4.7E-04 1.3E-03	1.6E-04 8.3E-04	5.9E-05 1.6E-04	2.0E-05	8.3E-05 2.3E-04	2.9E-05	3.2E-05 9.0E-05	1.1E-05	BLOOD CNS
SILVER	0 0005	0 0035	1 11E-03	1 38E-04	3 00E - 03	20E-03	1.1E-03	1 4E-04	7.4E-05	2.4E-03	1 3E-03	3.0E-04	1.6E-04	4.2E-04	2.3E-04	1.6E-04	8.9E-05	ARGYRIA
NYANZA SITE CONTAMIN	ÁNTS*		·	HAZARD IN	DEX	0 0E+00	0 0E+00	0 0E+00	0.0E+00	0.0E+00	0 0E+00	0.0E+00	0.0E+00	0 0E+00	0.0E+00	0.0E+00	0.0E+00	
OTHER SUDBURY RIVER (	CONTAMINAN	118		HAZARO IN	DEX	3 5E-03	1.9E-03	2.4E-04	1.3E-04	4.2E-03	2.3E-03	5.2E-04	2.8E-04	7.4E-04	4.0E-04	2.9E-04	1.6E-04	
ALL CHEMICALS OF CON	CERN			HAZARD IN	Dex	3 5E - 03	1.9E-03	2.4E-04	1.3E-04	4.2E-03	2.3E-03	5.2E-04	2.0E-04	7.4E-04	4.0E-04	2.9E-04	1.6E-04	
BEPH: BIS(2ETHYL HEXYL)	PHTHALATE			TOXICITY E	NDPOINTS A	BBREVIATIC	NS:	NS NOT S	PECIFIED	L/K: LIV OUS SYSTE	er and Kid M	NEY						

BEPH: BIS (2ETHYL HEXYL)PHTHALATE

.

W92194F

## TABLE 6-39B RISK RESULTS FOR SURFACE WATER REACH NO 7 NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

					CARCINO	GENIC RIS	K RESULTS	5	
CONTAMINANTS of CONCERN	CONCENTRA MG/	ATION KG	EXPOSURE	FACTOR		CANCER	RISKS		CANCER SLOPE FACTOR
	MAX	AVG	INGESTION	DERMAL CONTACT	ACCIDEN INGESTIC	TAL DN	DERM CONT/	AL \CT	(MG/KG/D) /
					MAX	AVG	MAX	AVG	WEIGHT OF EVIDENCE
BARIUM MANGANESE SILVER	0.0213 0.119 0.0065	0.0074 0.0745 0.0035	3.02E-04 3.02E-04 3.02E-04	4.21E-05 4.21E-05 4.21E-05	0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00	
NYANZA SITE CONTAMIN	ANTS*	4	CANCER R	ISK	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
OTHER SUDBURY RIVER	CONTAMINAN	ITS	CANCER R	ISK	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ALL CHEMICALS OF CON	CERN		CANCER R	ISK	0.0E+00	0.0E+00	0.0E+00	0.0E+00	

BEPH: BIS(2ETHYL HEXYL)PHTHALATE

6-138

#### TABLE 0 - 40A

#### PISK ASSESSMENT RESULTS FOR SEDIMENT EXPOSURE SCENARIOS

#### REACH NO 8

## NYANZA OPERABLE UNIT 3

#### MIDDLESEX COUNTY, MASSACHUSETTS

CONTAMINANTS of CONCERN	CONCENTRATION MG/KQ		N EXPOSURE FACTOR RECEPTOR = TEEN		CTOR RED TEEN		HAZARD QUOTIENTS CHILD				HAZARD QUOTIENTS: TEEN			HAZARD QUOTIENTS: ADULT				TOXIC END- POINT
	MAX	AVG	INGESTION	DERMAL CONTACT	(MG - KG -	ACCI		DE CO	RMAL NTACT						DENTAL			
		1	• ·-·	k			AVG	MAA	L AVG	MAX	AVG	MAA	NAG 1	MAX	AVG	MAX		ł
ARSENIC	30	12 13	1 11E-00	0 00€ + 00	3 00E - 04	1 8E - 01	7 4E - 02	0 0E + 00	0 0E + 00	1.1E-01	4 5E - 02	0.0E+00	0.0E+00	9 8E ~ 03	4.0E - 03	0.0E + 00	0.0E+00	SKIN
CADMIUM*	4 3	2 06	1 11E-08	0 00E + 00	8 00E - 04	1 8E - 02	7 6E - 03	0 0E + 00	0 0E + 00	9 8E - 03	4 6E 03	0.0E + 00	0.0E+00	8.4E-04	4.1E-04	0.0E + 00	0.0E+00	KIDNEY
CHROMIUM*	38 4	22	1 11E-00	0 00E + 00	1 00E + 00	8 8E - 05	4 OE - 05	0 0E + 00	0 0E + 00	4 0E - 05	2.4E-05	0 0E + 00	0.0E+00	3.6E - 08	2.2E~08	0.0E + 00	0.0E+00	UVER
LEAD*	100	<b>00</b> 7	5 55E - 07	0 00E + 00		0 0E + 00	0 0E + 00	0 0E + 00	0 0E 4 00	0 0E + 00	0 0E + 00	0.0E + 00	0.0E+00	0.0E+00	0.0E + 00	0.0E + 00	0.0E + 00	CNS
MERCURY.	21	1 62	1 11E-00	5 55E - 08	3 00E - 04	1 3E - 02	9 9E - 03	3 2E - 04	2 5E - 04	7 8E - 03	0 OE - 03	3 9E - 04	3.0E-04	6 8E ~ 04	5.3E-04	6.8E-05	5.3E - 05	CNS
BARIUM	222	113 41	1 11E-00	0 00E + 00	5 00E - 02	# 1E-03	4 1E -03	0 0E + 00	0 0E + 00	4 9E - 03	2.5E - 03	0.0E + 00	0.0E+00	4 3E 04	2.2E-04	0.0E + 00	0.0E + 00	BLOOD
COPPER	99	87 61	1 11E-00	0 00E + 00		0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E+00	0.0E+00	0.0E + 00	0.0E+00	0 0E + 00	0.0E+00	0.0E + 00	0.0E + 00	NS
MANGANESE	5860	1880 1	1 11E-00	0 00E + 00	1 00E - 01	1 1E - 01	3 1E - 02	0 0E + 00	0 0E + 00	8 5E - 02	1 9E - 02	0.0E + 00	0.0E+00	5.7E - 03	1.6E – 03	0.0E + 00	0.0E+00	CNS
NICKEL	13 1	7 00	1 11E-00	0 00E + 00	2 00E - 02	1 2E - 03	6 5E - 04	0 0E + <b>00</b>	0 0E + 00	7 3E - 04	3.9E – 04	0.0E + 00	0.0E+00	6.4E - 05	3.5E - 05	0.0E + 00	0.0E + 00	BW
SELENIUM	10	075	1 11E-08	0 00E + 00	5 00E - 03	<b>8 8E -</b> 04	2 7E - 04	0 0E + 00	0 0E + 00	4 0E - 04	1.7E-04	0.0E + 00	0.0E+00	3.5E-05	1.5E-05	0.0E + 00	0.0E+00	SELENOSIS
VANADIUM	12.8	8 72	1 11E-08	0 00€ + 00	7 00E - 03	3 3E - 03	2 5E - 03	0 0E + 00	0 0E + 00	2 OE 03	1.5E-03	0.0E + 00	0.0E+00	1.8E-04	1 4E - 04	0.0E + 00	0.0E + 00	NS
ZINC	] 326	187.38	] 1 11E- 00	0 00E + 00	2 00E - 01	3 OE - 03	17E-03	_0 0E + 00	0 0E + 00	1.8E-03	1.0E-03	0.0E + 00	0.0E+00	1.6E-04	9.2E-05	0.0E + 00	0 0E + 00	BLOOD
INVANZA BITE CONTAN	NANTR			HĂZĂBŌIM	DE X		9 1Ê - 02	1 2F -04	2 SE -04	1 1E - 01	5 AF -02	19F-04	10E-04	1 15 - 02	4 8E - 03	8.8E_05	5.3E - 05	ר
NTANZA SHE CONTAN	in Anti S							J 6L - 04			J UL - V2	3.8L - 04	3.02 -04	1.10 - 02	4.02 .03	0.05-00	J.32 03	
OTHER SUDBURY RIVE		IANTS		HAZARD IN	DEX	1 2E -01	4 OE - 02	0 0E + 00	0 0E + 00	7.5E-02	2 4E - 02	0.0E+00	0.0E+00	6.6E - 03	2.1E-03	0.0E + 00	0.0E+00	
ALL CHEMICALS OF CO	NCERN		_	HAZARD INDEX		3 3E - 01	1 3E - 01	3 2E - 04	2 5E - 04	2 0E-01	8.0E-02	3.9E - 04	3.0E-04	1.8E - 02	7.0E - 03	6 8E - 05	5.3E - 05	]
				TOXICITY E		BAREVIATIO	DNS	NS NOT	SPECIFIED	L∕K∙ UVE		NEY	BW: BODY N	<b>WEIGHT</b>	CNS: CENTR	AL NERVOUS	SYSTEM	

FINAL

۰,

## TABLE 6-40B

## RISK RESULTS FOR SEDIMENT EXPOSURE

## REACH NO. 8

## NYANZA OPERABLE UNIT 3

## MIDDLESEX COUNTY, MASSACHUSETTS

				CARCINOGENIC RISK ANALYSIS RESULTS								
CONTAMINANTS of			EXPOSURE	FACTOR			CANCER SLOPE FACTOR					
CONCERN	MAX	AVG	INGESTION	DERMAL CONTACT	ACCIDENTAL INGESTION		DERMAL CONTACT		(MG/KG/D)-1 / WEIGHT			
					MAX	AVG	MAX	AVG	OF EVIDENCE			
ARSENIC* CADMIUM* CHROMIUM* LEAD* MERCURY*	30 4.3 36.4 100 2.1	12.13 2.08 22 66.7 1.62	3.64E-07 3.64E-07 3.64E-07 1.82E-07 3.64E-07	0.00E +00 0.00E +00 0.00E +00 0.00E +00 1.51E -08	2.0E-05 0.0E+00 0.0E+00 0.0E+00 0.0E+00	7.9E-06 0.0E+00 0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	1.80E+00 A			
BARIUM COPPER MANGANESE NICKEL SELENIUM VANADIUM ZINC	222 96 5860 13.1 1.8 12.8 329	113.41 67.61 1680.1 7.08 0.75 9.72 187.36	3.64E - 07 3.64E - 07 3.64E - 07 3.64E - 07 3.64E - 07 3.64E - 07 3.64E - 07	0.00E + 00 0.00E + 00 0.00E + 00 0.00E + 00 0.00E + 00 0.00E + 00 0.00E + 00	0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00				

NYANZA SITE CONTAMINANTS	CANCER RISK	2.0E-05	7.9E-06	0.0E+00	0.0E+00
OTHER SUDBURY RIVER CONTAMINANTS	CANCER RISK	0.0E+00	0.0E+00	0.0E+00	0.0E+00
ALL CHEMICALS OF CONCERN	CANCER RISK	2.0E-05	7.9E-06	0.0E+00	0.0E+00

FINAL

1

index does not exceed unity. Thus, adverse noncarcinogenic health effects are not anticipated under the conditions of the recreational exposure scenarios defined in Section 6.4.

Arsenic is the only COC contributing to the excess lifetime cancer risk calculated for the sediment exposure scenarios. The arsenic concentrations detected in Reach 8 sediments and the associated estimated cancer risks are similar to those estimated for background arsenic concentrations.

### 6.5.3.15 Risk Assessment Results for Reach 9

Sudbury River Reach 9 includes Fairhaven Bay. The following five Nyanza Site contaminants were detected in Reach 9 sediments:

0	Arsenic	0	Lead
0	Cadmium	0	Mercury
0	Chromium		

The average mercury concentration detected in Reach 9 sediments is greater than 11 times the concentrations detected in background samples. Organic contaminant analyses were not conducted in Reach 9 sediment samples. No surface water samples were collected from this Reach.

Table 6-41 presents risk assessment results for COC concentrations detected in the sediments of Reach 9. The accidental-ingestion and dermal-contact routes of exposure (surface waters and sediments) were evaluated assuming recreational land/water use scenarios. The hazard quotients and hazard indices do not exceed unity for any of the cases presented. Also, if hazard indices are summed for the accidental-ingestion and dermal-contact exposure routes and then combined for surface water and sediment exposures, the total hazard index does not exceed unity. Thus, adverse noncarcinogenic health effects are not anticipated under the conditions of the recreational exposure scenarios defined in Section 6.4.

Arsenic is the only COC contributing to the excess lifetime cancer risk estimated for Reach 9 sediments. The arsenic concentrations detected in the Reach 9 sediments and the associated estimated cancer risks are approximately 3.5 times those estimated for background.

6-140

6-141

1

					·		NONCAR	DINOGENC	RISK ANA	LYSIS RESU	ILTS						. <u> </u>	
CONTAMINANTS of CONCERN		TRATION KG	EXPOSURE RECEPTO	XPOSURE FACTOR RECEPTOR = TEEN		HAZARD QUOTIENTS: CHILD				HAZARD QUOTIENTS: TEEN			HAZARD QUOTIENTS: ADULT				TOXIC END- POINT	
	MAX	AVG	INGESTION	DERMAL	(MG - KG -	ACCI	DENTAL	DE	RMAL NTACT			DERM	AL ACT	ACCID	STION	DERI CON1	MAL	
	<b>.</b>				_DAY_	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	1
ARSENC* CADMIUM* CHROMIUM* LEAD* MERCURY* BARIUM COPPER MANGANESE NICKEL VANADIUM ZINC	64 6 0 6 78 184 3 0 151 135 516 201 231 310	32 7 05 50 24 115 26 3 15 518 83 33 428 33 24 85 10 4 197 13	1 11E - 08 1 11E - 08 5 55E - 07 1 11E - 09 1 11E - 09	0 00E + 00 0 00E + 00 0 00E + 00 0 00E + 00 9 55E - 00 0 00E + 00	3 00E - 04 5 00E - 04 1 00E + 00 3 00E - 04 5 00E - 02 1 00E - 01 2 00E - 03 2 00E - 03 2 00E - 01	3 8E - 01 3 1E - 02 1 4E - 04 0 0E + 00 2 4E - 02 5 5E - 03 0 0E + 00 9 4E - 03 2 6E - 03 2 6E - 03 2 6E - 03	1 9E - 01 2 6E - 02 9 2E - 05 0 0E + 00 1 9E - 02 4 2E - 03 0 0E + 00 7 6E - 03 2 3E - 03 1 8E - 03	$\begin{array}{c} 0.0E + 00\\ 0.0E + 00\\ 0.0E + 00\\ 0.0E + 00\\ 5.9E - 04\\ 0.0E + 00\\ \end{array}$	0 0E + 00 0 0E + 00 0 0E + 00 0 0E + 00 4 8E - 04 0 0E + 00 0 0E + 00	2 4E - 01 1 9E - 02 8 7E - 05 0 0E + 00 1 4E - 02 3 4E - 03 0 0E + 00 5 7E - 03 1 .6E - 03 3 .7E - 03 1 6E - 03	1 2E - 01 $1 8E - 02$ $5.8E - 05$ $0.0E + 00$ $1 2E - 02$ $2.8E - 03$ $0 0E + 00$ $4 8E - 03$ $1 4E - 03$ $2 9E - 03$ $1 .1E - 03$	0.0E + 00 0.0E + 00 0.0E + 00 7.2E - 04 0.0E + 00 0.0E + 00 0.0E + 00 0.0E + 00 0.0E + 00 0.0E + 00 0.0E + 00	0.0E + 00 0.0E + 00 0.0E + 00 0.0E + 00 5.8E - 04 0.0E + 00 0.0E + 00 0.0E + 00 0.0E + 00 0.0E + 00 0.0E + 00	2.1E - 02 $1.7E - 03$ $7.8E - 08$ $0.0E + 00$ $1.3E - 03$ $3.0E - 04$ $0.0E + 00$ $5.0E - 04$ $1.4E - 04$ $3.2E - 04$ $1.6E - 04$	1.0E - 02 $1.4E - 03$ $4.9E - 06$ $0.0E + 00$ $1.0E - 03$ $2.3E - 04$ $0.0E + 00$ $4.2E - 04$ $3.2E - 04$ $9.6E - 05$	0.0E + 00 0.0E + 00 0.0E + 00 1.3E - 04 0.0E + 00 0.0E + 00 0.0E + 00 0.0E + 00 0.0E + 00 0.0E + 00	0.0E+00 0.0E+00 0.0E+00 1.0E-04 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	SKIN KIDNEY LIVER CNS CNS BLOOD NS CNS BW NS BLOOD
NYANZA SITE CONTAMI OTHER SUDBURY RIVER	l Ñânts RCONTAMN NCERN	ANTS	1	HĂZĂPÔ IN HĂZĂPÔ IN HAZAPO IN HAZAPO IN	JEX JEX JEX	4 5E - 01 2 8E - 02 4 8E - 01	2 4E - 01 2 1E - 02 2 9E - 01	5 9E - 04 0 0E + 00 5 9E - 04	4 8E - 04 0 0E + 00 4 8E - 04	2.7E-01 1 6E-02 2 9E-01	1 5E - 01 1.3E - 02 1 6E - 01	7.2E-04 0 0E+00 7.2E-04	5.8E-04 0.0E+00 5.8E-04	2.4E - 02 1.4E - 03 2.5E - 02	1.3E-02 1.1E-03 1.4E-02	1.3E-04 0.0E+00 1.3E-04	1.0E - 04 0.0E + 00 1.0E - 04	]

W92194F

## TABLE 6-41B

## RISK RESULTS FOR SEDIMENT EXPOSURE

## **REACH NO. 9**

## NYANZA OPERABLE UNIT 3

## MIDDLESEX COUNTY, MASSACHUSETTS

				CARCINOG	ENIC RISK A	NALYSIS F	RESULTS		
CONTAMINANTS of	CONCENTRATION MG/KG		EXPOSURE FACTOR				SLOPE FACTOR		
CONCERN	мах	AVG	INGESTION	DERMAL CONTACT	ACCIDEN INGESTIC	TAL DN	DERM CONT/	AL ACT	(MG/KG/D) 1 / WEIGHT
					MAX	AVG	МАХ	AVG	
ARSENIC*	64.6	32	3.64E-07	0.00E+00	4.2E-05	2.1E-05	0.0E+00	0.0E+00	1.80E+00 A
CADMIUM*	8.6	7.05	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
CHROMIUM*	78	50.26	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
LEAD*	184	115.26	1.82E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MERCURY*	3.9	3.15	3.64E-07	1.51E-08	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
BARIUM	151	116	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
COPPER	135	83.33	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MANGANESE	516	429.33	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
NICKEL	28.1	24.95	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
VANADIUM	23.1	18.4	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ZINC	319	197.13	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
						0.15.00	0.05.00		L,,,,,
NYANZA SITE CONTAMI	NANIS		CANCER RIS		4.2E-05	2.1E-05	U.UE+00	U.UE+00	
OTHER SUDBURY RIVER	OTHER SUDBURY RIVER CONTAMINANTS				0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ALL CHEMICALS OF CONCERN			CANCER RISI	<	4.2E-05	2.1E-05	0.0E+00	0.0E+00	

BEPH:BIS(2-ETHYL HEXYL)PHTHALATE

6-142

FINAL

(

## 6.5.3.16 <u>Risk Assessment Results for Reach 10</u>

Sudbury River Reach 10 extends from the area of the outlet of Fairhaven Bay to the confluence of the Sudbury and Assabet Rivers. The following four Nyanza Site contaminants were detected in Reach 10 sediments:

0	Arsenic	ο	Lead
0	Chromium	0	Mercury

The average mercury concentration detected in Reach 10 sediments ( $C_{avg} = 0.16 \text{ mg/kg}$ ) is similar to mercury concentrations detected in background samples ( $C_{avg} = 0.27$ ). No organic contaminant analyses were conducted in Reach 10 sediments. No surface water samples were collected in the Reach.

Table 6-42 presents risk assessment results for COC concentrations detected in the sediments of Reach 10. The accidental-ingestion and dermal-contact routes of exposure (surface waters and sediments) were evaluated assuming recreational land/water use scenarios. The hazard quotients and hazard indices do not exceed unity for any of the cases presented. Also, if hazard indices are summed for the accidental-ingestion and dermal-contact exposure routes, the total hazard index does not exceed unity. Thus, adverse noncarcinogenic health effects are not anticipated under the conditions of the recreational exposure scenarios defined in Section 6.4.

Excess lifetime cancer risks estimated for maximum and average COC concentrations in Reach 10 sediments are  $8 \times 10^{6}$  and  $3.9 \times 10^{6}$ , respectively. These risk values are similar to those estimated for background. Arsenic, the only COC contributing to risk, was detected at concentrations similar to those observed in background samples.

## 6.5.3.17 <u>Risk Assessment Results for Fairhaven Bay</u>

This section presents the results of the risk assessment of COC concentrations detected in fish tissue samples taken from Fairhaven Bay. Hazard quotients, hazard indices, and cancer risks estimated by the risk assessment are presented in Table 6-43.

Hazard indices calculated for all COCs exceed unity in all cases presented except when average contaminant concentrations are evaluated and the sport fisherman is the receptor of concern. The majority of the noncarcinogenic risk is attributable to Nyanza Site contaminants; hazard indices calculated for other Sudbury River

#### TABLE 8-42A RISK ASSEBSMENT REBULTS FOR SEDIMENT EXPOSURE SCENARIOS REACH NO 10 NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

....

.....

. ....

							NONCAR	CINOGENC RISK AN	ALYSIS RESUL	LTS			<u>.                                    </u>		· · · · · · · · · · · · · · · · · · ·		
CONTAMINANTS of CONCERN			ICENTRATION EXPOSURE		CTOR RED REEN	HAZARD QUOTIENTS: CHLD			HAZARD QUOTIENTS: TEEN				HAZARD QUOTIENTS: ADULT				TOXIC END- POINT
	MAX	AVG	INGESTION		(MG - KG -	ACCI	DENTAL	DERMAL CONTACT	ACCIDENT	AL N	DERM	IAL ACT		ESTION	DER CON	MAL	
		!	ļ	I	DAY	MAX	AVG	MAX AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	1
ARSENIC*	122	5 90	1 11E - 06	0 00E + 00	3 00E - 04	7 4E - 02	3 8E - 02 2 15 - 05	0 0E+00 0 0E+0	0 4 5E - 02	2.2E - 02	0 0E + 00	0.0E+00	4.0E-03	2.0E - 03	0.0E + 00	0.0E+00	SKIN
LEAD*	33 0	22 48	5 55E - 07	0 00E + 00	100E - 04	0 0E+00	0 0E+00	0 0E+00 0 0E+0	0 0 0E + 00	0.0E+00	0.0E+00	0 0E+00	0.0E+00	0.0E+00 5.3E=05	0.0E + 00	0.0E+00 5.3E-08	CNS
BARIUM	84.3	33 61	1 11E-00	0 00E + 00	5 00E - 02	2 3E - 03	1 2E - 03	0 0E+00 0 0E+0	0 1.4E-03	7.5E-04	0.0E+00	0.0E+00	1.3E-04	6.6E - 05	0.0E+00	0.0E + 00	BLOOD
	31	14 115	1 11E-00 1 11E-00	0 00E + 00 0 00E + 00	1 00E - 01	0 0E+00 0 1E-03	0 0E + 00 3 3E - 03	0 0E+00 0 0E+0 0 0E+00 0 0E+0	0.0E+00 3.7E-03	0.0E+00 2.0E-03	0.0E+00 0.0E+00	0.0E + 00 0.0E + 00	0.0E+00 3.3E-04	0.0E+00 1.7E-04	0.0E + 00 0.0E + 00	0.0E + 00 0.0E + 00	NS CNS
	111	7 37	1 11E - 08	0 00E + 00 0 00E + 00	2 00E - 02 7 00E - 03	1 0E-03 4 1E-03	87E-04 24E-03	0 0E + 00 0 0E + 0 0 0E + 00 0 0E + 0	0 62E-04 2.5E-03	4.1E-04 1.5E-03	0.0E+00 0.0E+00	0.0E+00 0.0E+00	5.4E-05 2.2E-04	3.6E-05 1.3E-04	0.0E + 00 0.0E + 00	0.0E + 00 0.0E + 00	BW NS
ZINC	53.4	38.35	1 11E - 00	0 00E + 00	2 00E - 01	4 9E - 04	3 5E - 04	0 0E + 00 0 0E + 0	3 0E - 04	2.1E-04	0.0E + 00	0.0E + 00	2.6E - 05	1.9E-05	0.0E + 00	0.0E + 00	BLOOD
NYANZA SITE CONTAMI	NANTS		l	HAŻAŔŐ IN	DEX	7 0E - 02	3 7E - 02	6 1E - 05 2 5E - 0	5 4 7E - 02	2 3E - 02	9 8E - 05	3 0E - 05	4.2E-03	2 0E - 03	1.7E-05	5 3E - 06	L
OTHER SUDBURY RIVER		ANTS		HAZAPO IN	DEX	1 4E - 02	7 9E - 03	0 0E + 00 0 0E + 0	0 66E-03	4 8E - 03	0.0E + 00	0 0E + 00	7.6E - 04	4.2E-04	0.0E + 00	0.0E + 00	
ALL CHEMICALS OF COI	NCERN			HAZARD IN	DEX	0 2E - 02	4 5E - 02	8 1E-05 2 5E-0	5 5 6E - 02	2.8E - 02	9.8E - 05	3.0E 05	4 9E - 03	2.4E-03	1.7E-05	5.3E - 06	
				TOXICITY		BBREVIATIO	ONS.	NS NOT SPECIFIE	J L/K: UVER		NEY	BW: BODY V	VEIGHT	CNS: CENTRA	L NERVOUS	SYSTEM	

6-144

----- ·

1

. .

## TABLE 6-42B RISK RESULTS FOR SEDIMENT EXPOSURE REACH NO. 10

## NYANZA OPERABLE UNIT 3

## MIDDLESEX COUNTY, MASSACHUSETTS

				CARCINOG	ENIC RISK A	NALYSIS F	RESULTS		
CONTAMINANTS of	CONCENTRATION MG/KG		EXPOSURE	FACTOR			CANCER SLOPE FACTOR		
CONCERN	МАХ	AVG	INGESTION	DERMAL CONTACT	ACCIDENTAL INGESTION		DERMAL CONTACT		(MG/KG/D) – 1 / WEIGHT
					MAX	AVG	MAX	AVG	
ARSENIC*	12.2	5.99	3.64E – 07	0.00E+00	8.0E-06	3.9E-06	0.0E+00	0.0E+00	1.80E+00 A
CHROMIUM*	17.1	11,24	3.64E 07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
LEAD*	33.8	22.48	1.82E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MERCURY*	0.53	0.161	3.64E-07	1.51E-08	0.0E+00	0.0E+00	0.0E+00	0.0E+00	ĺ
BARIUM	64.3	33.61	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
COPPER	31	14.115	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MANGANESE	336	178.04	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
NICKEL	11.1	7.37	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
VANADIUM	15.9	9.15	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ZINC	53.4	38.35	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
NYANZA SITE CONTAMINA				<u> </u>	8 0F - 06	39E-06	0.05+00	0.0E±00	
				×	0.02-00	0.0E -00			
OTHER SUDBURY RIVER CONTAMINANTS			CANCER RISH	ĸ	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ALL CHEMICALS OF CONCERN			CANCER RISH	<b>K</b>	8.0E-06	3.9E~06	0.0E+00	0.0E+00	

FINAL

6-145

# W92194F
#### TABLE 8-43 RISK ASSESSMENT RESULTS FOR FISH INGESTION EXPOSURE SCENARIOS FAIRHAVEN BAY NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

						NONCARCIN	OGENIC RIBK AN	ALYSIS RESULT	18		1		CARCINOG	ENIC RISK A	NALYSIS RI	ESULTS	
CONTAMINANTS OF CONCERN	CONCEN MO	KG	EXPOSURE	FACTOR	AFD MG-		HAZARD	QUOTIENTS		TOXICITY END-	EXPOSURE	FACTOR	CA	NCER RISKS	3		CANCER SLOPE
	MAX	<b>EVA</b>	SPORTS	BUBBIS- -TENCE	KG- DAY)	8PORTEF	ISHERMEN	5U88IST	ENCE		Sports Fisher-	SUBSIS - TENCE	SPORT FI	SHERMEN	SUBSIST FISHERI	ence Men	FACTOR (MG/KG/D)~1 /
		Ì	FUENER-	FIGHER-		MAXONAIN	AVERAGE	MAXIMUM	AVERAGE	4	MEN	FISHER-	MAXIMUM	AVERAGE	MAXIMUM	AVERAGE	EVIDENCE
	<b>f</b>	•			_		_					<u> </u>					
PHENOL*	82	1 079	185E-04	1 JEE - 03	0 00E -01	2 9E - 03	3 3E - 04	1 GE - 02	2.4E-03	FETUS	7.93E ~ 08	5.81E04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
NITROBENZENE	0.024	0 024	1 80E - 04	1 36E - 03	8 00E - 04	8 EE - 03	6 9E -03	0 5E - 02	6.5E-02	L/K	7.93E-05	5.81E04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
CHROMIUM*	34	0 453	1 05E - 04	1 30E - 03	100E+00	0 3E - 04	8 € ~05	4 <del>CE</del> - 03	6 1E-04	LIVER	7.93E~05	5.81E-04	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	
LEAD*	56	0 270	1 85E - 04	1 3E - 03		0 0E+00	0 0E + 00	0 0E + 00	0 0E+00	CNS	7.93€05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MERCURY*	32	1 054	1 8 E-04	1 3E - 03	3 00E - 04	2 0E+00	6 5E - 01	1 € +01	4 6E+00	CNS	7.93E~05	5.81E~04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	· ·
METHYL MERCURY*	12	0 56293	1 85E - 04	1 36E - 03	3 00E - 04	7∉-01	<b>3 ∰</b> -01	5 € +00	2.5E+00	CNS	7.93E-05	5.61E04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
3/4 METHYL PHENOL	1 1	0 138	1 85E-04	1 38E - 03	5 00E - 02	7 0E - 03	5 1E - 04	5 2E - 02	3 8E - 03	BW,CNS	7.93E - 05	5.81E04	0.0E+00	0.0E+00	00E+00	0.0E+00	1
METHYLENE CHLORIDE	14	0 5074	1 85E - 04	1 36E - 03	00E-02	4 3E - 02	1 6E - 03	3 2E - 01	1.1E-02	LIVER	7.93E-05	5.81E04	8.3E-06	3.0E - 07	6.1E-05	2.2E-06	7.50E-03 Ba
ACETONE	1.0	0 30820	1 85E - 04	1 36E - 03	1 00E - 01	3 0E - 03	7 Æ-04	2 Æ -02	5.4E-03	L/K	7.93E-05	5.81E04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
BEPH	0 003	0 0325	181E-04	1 3E - 03	\$ 00E -02	3€-04	3 Œ -04	2 🖭 - 03	22E-03	LIVER	7.93E-05	5.81E04	4.1E-08	3.6E~08	3.0E-07	2.6E-07	1.40E-02 82
BENZO (B)FLUORANTHEN	¢ 0.087	0 07783	1 85E - 04	1 36E - 03	ł	0.0€+00	0 0E + 00	0 0E + 00	0.0E+00	NS	7.93E-05	5.81E04	4.0E-05	3.6E~05	2.9E-04	2.6E-04	5.80E+00 B2
BENZO (A) PYRENE	0 021	0 020	1 85E-04	1 3EE - 03		0 0E + 00	0 0E + 00	0 0E + 00	0.0E+00	NS	7.93E-05	5.81E04	1.2E05	1.2E-05	8.8E~05	8.6E-05	5.80E+00 B2
4,4DDE	0.034	0 01782	1 85E - 04	1 3E - 03		0 0E + 00	0 0E + 00	0 0E + 00	0 0E+00	LIVER	7.93E05	5.81E - 04	9.7E-07	4.8E-07	7.1E-08	3.5E-08	3.40E-01 82
AROCLOR-1254	04	0 18753	1 80E-04	1 JE -03		0 0€+00	0 0E + 00	0 0E+00	0.0E+00	NB	7.93E05	5.61E04	2.6E-04	1.2E04	2.1E-03	8.6E 04	7.70E+00 B2
AROCLOR-1200	000	0 0 1016	1 85E - 04	1 36E - 03		0 0E+00	0.0E+00	0 0E + 00	0.0E+00	NS	7.93E-05	5.61E04	1.6E-05	6.2E-06	1.3E-04	4.6E-05	7.70E+00 B2
BARIUM	1 1	0 637	1 80E-04	1 30E -03	\$ 00E-02	4 ¥€ - 03	2.€-03	3 0E - 02	1.7E-02	8LOOD	7.93E06	5.61E04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
COPPER	34	0 884	1 80E-04	1 30E-03	1	0 0E+00	0 0E+00	0 0E + 00	0.0E+00	NS	7.93E-06	5.81E04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MANGANESE	33 2	2 858	1 85E-04	1 36E - 03	1 00E 01		5 3E -03	4 5E - 01	3 GE 02	CNS	7.93E-05	5.81E-04	0 0E+00	0.0E+00	0.0E+00	0.0E+00	
NICKEL	1 1	1 0 32	185 -04	1 3E - 03	2 00E - 02	166-02	3 0E - 03	1 € -01	2 2E-02	BW	7.93E-05	5.81E04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
SELENIUM	0 27	0 084	1 85E-04	1 30E - 03	5 00E 03	10E-02	3 Æ -03	7.3E-02	2.5E-02	SELENOSIS	7.93E-06	5.81E-04	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	
ZINC	21 2	: 0 963	1 85E-04	1 30E - 03	2 00E -01	2 0E - 02	0 JE - 03	1.€-01	6.1E+02	BLOOD	7.93E-05	5.81E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	 
NYANZA SITE CONTAMIN	ANTS"		HAZAPID IN	DEX		2 (E+00)	€ Œ-01	1.5E+01	4 6E + 00		CANCERR	ISKS:	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	]
OTHER SUDBURY AMER	CONTAMIN	NTS	HAZAFID IN	DEX		1 モ-01	2 E-02	12E+00	1.9E-01		CANCER R	<b>ISKS</b> :	3.6E - 04	1.8E-04	2.6E-03	1.3E03	
ALL CHEMICALS OF CON	CEAN		HAZARDIN	DEX		2 2E+00	6 6E - 01	1 <b>€</b> +01	5.0E+00		CANCER R	ISKS:	3.6E-04	1.8E-04	2.6E - 03	1.3E-03	
BEPH: BIS(2-ETHYL HEXY	UPHTHALA	TE			<b>HEWATION</b>	49 HK	NOT SPECIFIED	BW-BODY	WEIGHT CH	S: CENTRAL	NERVOUS SY	STEM					

LAK LIVER AND KIDNEY

FINAL

6-146

contaminants do not exceed unity. Hazard quotients calculated for mercury and methyl mercury exceed unity when the reasonable maximum-case and average-case scenarios are evaluated and the subsistence fisherman is the receptor of concern. The hazard quotient calculated for mercury also exceeds unity when maximum contaminant concentrations are evaluated and a sports fisherman is considered the receptor of concern. Because mercury compounds affect the central nervous system and the kidney (they are primary target organs), hazard indices calculated for the kidney and/or central nervous system exceed unity in one or more cases presented in Table 6-43. The maximum and average mercury and the maximum methyl mercury concentrations detected in the Fairhaven fish tissue samples exceed the current FDA Action Level for mercury in fish (1 mg/kg). These results suggest that adverse noncarcinogenic health affects would be anticipated for the receptor routinely ingesting fish taken from Fairhaven Bay at the fish ingestion rate and frequency assumed in Section 6.4.

Although the excess lifetime cancer risk estimated for the Fairhaven Bay fish tissue samples exceeds  $1 \times 10^4$  in all cases presented, it should be noted that Nyanza Site contaminants are not contributing to the estimated risks. (Cancer risks exceed  $1 \times 10^3$  when the subsistence fisherman is evaluated as the receptor of concern.) Two PCB compounds, Aroclor-1254, and Aroclor-1260, are the principal contaminants contributing to the estimated excess lifetime cancer risk. As a point of reference, cancer risks estimated for subsistence fishermen equal  $5 \times 10^5$  when maximum and average COC concentrations in the Sudbury Reservoir fish tissue samples are evaluated (the Sudbury Reservoir is considered as part of the background area for the Study Area).

#### 6.5.3.18 <u>Risk Assessment Results for the Bordering Wetlands</u>

A number of wetlands along the Sudbury River adjoin or are located in the general vicinity of the residential areas. Consequently, this section presents risks assessment results for residential as well as recreational land-use scenarios. As discussed in Section 6.4, the exposure assumptions for the residential and recreational scenarios are similar except that a receptor is expected to be exposed 270 days/year under a residential land-use scenario instead of the 50 to 150 days/year assumed for receptors for the recreational land-use scenarios.

Tables 6-44 and 6-45 present the noncarcinogenic risk assessment results for the recreational and residential land-use scenarios, respectively. The accidental-ingestion and dermal-contact routes

#### TABLE 6 - 44A RISK RESULTS FOR SEDIMENT EXPOSURE BORDERING WETLANDS RECREATIONAL NYANZA OPERABLE UNIT 3 MODLESEX COUNTY, MASSACHUSETTS

			• ••••• • • • • •				NONCAR	NOGENC	RISK ANA	LYDIB RESI	JLTS							
CONTAMINANTS d1 CONCERN	CONCEN	KG	EXPOSURE RECEPTO	FACTOR	PFD	HA		NENTS CH	eLD	HAZ	ARD QUO	NENTS: TE	EN	HAZARI	DQUOTIENT	S: ADULT		TOXIC END- POINT
	MAX	AVG	INGESTION	DERMAL CONTACT	(MC) - KC) -	ACC	DENTAL STION	DE		ACCIDEN		DERN			ENTAL	DERI		
		I			DAY	MAX	AVG .	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	1
ARSENIC*	11.0	8 22	1 11E-00	0 00E + 00	3 00E - 04	7 2E - 02	38E - 02	0 0E + 00	0 0E + 00	4 4E - 02	2 3E - 02	0.0E + 00	0 0E + 00	3 9E - 03	2.0E-03	0 0E + 00	0.0E+00	SKIN
CHROMIUM*	101	24 01	1 11E - 00	0 00E + 00	1 00E + 00	1 UE - 04	442-05	0.05+00	0.05+00	1.1E-04	2./E-05	0.0E+00	0.0E+00	9.9E-06	2.31-06	0.0E+00	0.0E+00	ONB
MERCURY	16	1 071	1116-00	5 55E - 00	3 00E - 04	4 8E - 01	8 5E - 03	1 2E - 03	1 6E - 04	2 8E-02	4.0E-03	1.4E-03	2.0E-04	2.5E-03	3.5E-04	2 5E - 04	3.5E-05	CNS
BARIUM	67 5	38 125	1 11E-00	0 00E + 00	5 00E - 02	2 SE - 03	14E-03	0 0E + 00	0 0E + 00	1.5E - 03	8.5E - 04	0.0E + 00	0.0E+00	1.3E - 04	7.5E-05	0.0E+00	0.0E+00	BLOOD
BERYLLIUM	10	0 816	1116-08	0 00E+00	5 00E - 03	0 9E - 04	1 3 0E - 04	0 0E + 00	0 0E + 00	4.2E-04	1.8E-04	0.0E+00	0.0E+00	3.7E-05	1.6E - 05	0.0E + 00	0.0E+00	INS
MANGANESE	1070	18 013 297 82	1 11E-00	0 00E + 00	1 00F - 01	2 OE - 02	2 54E-03	0 0E + 00	0 0E + 00	1 2E - 02	3.3E-03	0.0E+00	0.0E+00	1.0E - 03	2.9E~04	0 0E + 00	0.0E+00	CNS
NICKEL	21 8	7 18	1 11F-00	0 00E + 00	2 00E - 02	2 OE - 03	0 0E - 04	0 0E + 00	0 0E + 00	1 2E - 03	4.0E-04	0.0E + 00	0.0E+00	1.1E-04	3.5E-05	0.0E+00	0.0E+00	BW
SELENIUM	12	0 07	1 11E - 00	0 00E + 00	5 00E - 03	4 4E - 04	2 6E - 05	0 0E + 00	0 0E + 00	2.7E 04	1.6E - 05	0.0E + 00	0.0E+00	2.3E - 05	1.4E-06	0.0E + 00	0.0E + 00	SELENOS
VANADIUM	57 5	28 085	1 11E - 00	0 00E + 00	7 00E -03	1 5E - 02	7 3E - 03	0 0E + 00	0 0E + 00	9.1E-03	4.5E-03	0.0E+00	0.0E+00	8.0E-04	3.9E-04	0.0E + 00	0.0E+00	NS
ZINC	.] 127	41 078	1 1 1 1E - 00	0 00E + 00	2 00E - 01	1 7E - 03	J 8E - 04	0 UE + 00	0 0E + 00	71E-04	2.3E-04	0.0E+00	0.05+00	6.2E - 05	2.06-05	0.0E+00	0.0E + 00	BLOOD
NYANZA SITE CONTAM	NANTB			HAZARD IN	DEX	1 2E - 01	4 4E - 02	1 2E - 03	1.8E - 04	7.2E-02	2.7E-02	1.4E-03	2.0E-04	6.4E-03	2.4E - 03	2.5E - 04	3.5E-05	] .
OTHER SUDBURY RIVE		ANTS			DEX	4 1E - 08	1 6E - 02	0 0E + 00	0 0E + 00	2 5E - 02	9.4E-03	0.0E + 00	0.0E+00	2.2E - 03	8.3E-04	0.0E + 00	0.0E+00	
ALL CHEMICALS OF CO	NCERN			HAZARD IN	DEX	1 8E - 0	6 OE - 02	1 2E - 03	1 6E - 04	9.7E-02	3 8E - 02	1.4E-03	2.0E-04	8.6E-03	3.2E - 03	2.5E - 04	3.5E - 05	

Ĵ,

L

(

# TABLE 6¹,B

# RISK RESULTS FOR SEDIMENT EXPOSURE BORDERING WETLANDS: RECREATIONAL NYANZA OPERABLE UNIT 3

## MIDDLESEX COUNTY, MASSACHUSETTS

				CARCINOGE	ENIC RISK A	NALYSIS F	RESULTS		
CONTAMINANTS of	CONCEN MG/I		EXPOSURE	FACTOR		CANCER F	risks		CANCER SLOPE FACTOR
CONCERN	мах	AVG	INGESTION	DERMAL CONTACT		TAL DN	DERM CONT	AL	(MG/KG/D) – 1 / WEIGHT
					MAX	AVG	МАХ	AVG	OF EVIDENCE
ARSENIC* CHROMIUM* LEAD* MERCURY*	11.9 101 254 7.6	6.22 24.01 71.19 1.071	3.64E-07 3.64E-07 1.82E-07 3.64E-07	0.00E+00 0.00E+00 0.00E+00 1.51E-08	7.8E-06 0.0E+00 0.0E+00 0.0E+00	4.1E-06 0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00 0.0E+00	1.80E+00 A
BARIUM BERYLLIUM COPPER MANGANESE	67.5 1.9 83.4 1070	38.125 0.818 18.013 297.92	3.64E-07 3.64E-07 3.64E-07 3.64E-07	0.00E+00 0.00E+00 0.00E+00 0.00E+00	0.0E+00 3.0E-06 0.0E+00 0.0E+00	0.0E+00 1.3E06 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00 0.0E+00	4.30E+00 A
NICKEL SELENIUM VANADIUM ZINC	21.8 1.2 57.5 127	7.18 0.07 28.065 41.078	3.64E-07 3.64E-07 3.64E-07 3.64E-07	0.00E+00 0.00E+00 0.00E+00 0.00E+00	0.0E+00 0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00 0.0E+00	

.

NYANZA SITE CONTAMINANTS	CANCER RISK	7.8E-06	4.1E-06	0.0E+00	0.0E+00
OTHER SUDBURY RIVER CONTAMINANTS	CANCER RISK	3.0E06	1.3E-06	0.0E+00	0.0E+00
ALL CHEMICALS OF CONCERN	CANCER RISK	1.1E-05	5.4E-06	0.0E+00	0.0E+00

BEPH:BIS(2-ETHYL HEXYL)PHTHALATE

6-149

#### TABLE 6-45A RISK ASSESSMENT RESULTS FOR SEDIMENT EXPOSURE SCENARIOS BORDERING WETLANDS: RECREATIONAL (270 DAYS / YEAR) NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

							NONCAR	INOGEN	C RISK ANA	LYSIS RESI	JL 19					<b></b>		·····
CONTAMINANTS of CONCERN	CONCEN MG/	TRATION KG	EXPOSURE RECEPTO	FACTOR R = TEEN	rfd	HAZ		TIENTS: C	HLD	HAZ	ARD QUO	TIENT <b>S</b> : TE	EN	HAZAR	DQUOTIENT	S: ADULT		TOXIC END POINT
	MAX	AVG	INGENTION	DERMAL CONTACT	(MG - KG -	ACCI	DENTAL		RMAL NTACT	ACCIDEN	TAL	DERM	ML ACT	ACCID	ENTAL STION	DERI	AAL ACT	
					DAY	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	
ARSENIC* CHROMIUM* LEAD* MERCURY*	11 9 101 254 7 6	6 22 24 01 71 19 1 071	2 00E - 06 2 00E - 06 1 00E - 06 2 00E - 06	0 00E + 00 0 00E + 00 0 00E + 00 1 00E - 07	3 00E - 04 1 00E + 00 3 00E - 04	39E-01 10E-03 00E+00 26E-01	2 0E - 01 2 4E - 04 0 0E + 00 3 5E - 02	0 0E+00 0 0E+00 0 0E+00 6 2E-03	0 0E+00 0 0E+00 0 0E+00 6 6E-04	7.9E-02 2.0E-04 0.0E+00 5.1E-02	4.1E-02 4.8E-05 0.0E+00 7.1E-03	0.0E+00 0.0E+00 0.0E+00 2.5E-03	0.0E+00 0.0E+00 0.0E+00 3.6E-04	2.1E-02 5.3E-05 0.0E+00 1.3E-02	1.1E - 02 1.3E - 05 0.0E + 00 1.9E - 03	0.0E+00 0.0E+00 0.0E+00 1.3E-03	0.0E+00 0.0E+00 0.0E+00 1.9E-04	SKIN LIVER CNS CNS
BARIUM BERYLUUM COPPER MANGANESE NICKEL SELENIUM VANADIUM ZINC	67 5 19 63 4 1070 21 6 12 57 5 127	38 125 0 818 18 013 297 92 7 18 0 07 28 085 41 078	2 00E - 08 2 00E - 08	0 00E + 00 0 00E + 00	5 00E - 02 5 00E - 03 1 00E - 01 2 00E - 02 5 00E - 03 7 00E - 03 2 00E - 01	1 3E - 02 3 7E - 03 0 0E + 00 1 1E - 01 1 1E - 02 2 4E - 03 6 1E - 02 6 3E - 03	7 5E - 03 1 6E ~ 03 0 0E + 00 2 9E - 02 3 5E - 03 1 4E - 04 4 0E - 02 2 0E - 03	0 0E + 00 0 0E + 00	0 0E+00 0 0E+00 0 0E+00 0 0E+00 0 0E+00 0 0E+00 0 0E+00 0 0E+00	2.7E-03 7.6E-04 0.0E+00 2.1E-02 2.2E-03 4.6E-04 1.6E-02 1.3E-03	1 5E - 03 3.3E - 04 0.0E + 00 6.0E - 03 7.2E - 04 2.8E - 05 8.0E - 03 4.1E - 04	0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	7.1E-04 2.0E-04 0.0E+00 5.7E-03 5.6E-04 1.3E-04 4.3E-03 3.4E-04	4.0E - 04 8.8E - 05 0.0E + 00 1.6E - 03 1.9E - 04 7.4E - 06 2.1E - 03 1.1E - 04	0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	BLOCD NS CNS BW SELENOSIS NS BLOCD

	TORCITY ENDPOINTS	ABBREVIATIONS:	NS NOT SPECIFIED	LAC LIVER AND KIDNEY	BW: BODY WEIGH	CNS: CENTR/	L NERVOUS	SYSTEM
ALL CHEMICALS OF CONCERN	HAZAFED INDEX	87E-01 32E-	01 82E-03 88E-04	1.6E-01 6.6E-02 2.5E-	03 3.6E-04 4.6	E-02 1.7E-02	1.3E-03	1.9E-04
OTHER SUDBURY RIVER CONTAMINANTS	HAZARD INDEX	2 2E-01 84E-	02 0 0E+00 0 0E+00	4.5E-02 1.7E-02 0.0E+	00 0.0E+00 1.2	E-02 4.5E-03	0.0E+00	0.0E+00
NYANZA SITE CONTAMINANTS	HAZARD INDEX	04E-01 24E-	01 0 2E-03 8 6E-04	1.3E-01 4.9E-02 2.5E-	03 3.6E-04 3.4	E-02 1.3E-02	1.3E-03	1.9E-04

6-150

OTHER SUDBURY RIVER CONTAMINANTS

NYANZA SITE CONTAMINANTS

ALL CHEMICALS OF CONCERN

RISK ASSESSMENT RESULTS FOR SEDIMENT EXPOSURE SCENARIOS
BORDERING WETLANDS: RECREATIONAL (270 DAYS / YEAR)
NYANZA OPERABLE UNIT 3
MIDDLESEX COUNTY MASSACHUSETTS

2.7E-05 1.4E-05 0.0E+00 0.0E+00

1.0E-05 4.5E-06 0.0E+00 0.0E+00

3.8E-05 1.9E-05 0.0E+00 0.0E+00

				CARCINOGE	ENIC RISKA	NALYSIS	RESULTS		
CONTAMINANTS of		TRATION	EXPOSURE	FACTOR		CANCER F	NSKS		CANCER SLOPE FACTOR
CONCERN	мах	AVG	INGESTION	DERMAL CONTACT	TOR         CANCER RISKS           IMAL         ACCIDENTAL INGESTION         DERMAL CONTACT           MAX         AVG         MAX         AVG           E+00         2.7E-05         1.4E-05         0.0E+00         0.0E+00           E+00         0.0E+00         0.0E+00         0.0E+00         0.0E+00           E+00         1.0E-05         4.5E-06         0.0E+00         0.0E+00           E+00         1.0E-05         4.5E-06         0.0E+00         0.0E+00	(MG/KG/D) – 1 / WEIGHT			
					MAX	AVG	MAX	AVG	EVIDENCE
ARSENIC* CHROMIUM* LEAD* MERCURY*	11.9 101 254 7.6	6.22 24 01 71.19 1.071	1.28E 06 1.28E 06 6.39E 07 1.28E 06	0.00E + 00 0.00E + 00 0.00E + 00 4.73E - 08	2.7E-05 0.0E+00 0.0E+00 0.0E+00	1.4E-05 0.0E+00 0.0E+00 0.0E+00	0.0E + 00 0.0E + 00 0.0E + 00 0.0E + 00	0.0E+00 0.0E+00 0.0E+00 0.0E+00	1.80E+00 A
BARIUM BERYLLIUM COPPER MANGANESE NICKEL SELENIUM VANADIUM ZINC	67 5 1.9 83.4 1070 21.8 1.2 57.5 127	38.125 0 818 18.013 297.92 7.18 0.07 28.065 41.078	1.28E - 06 1.28E - 06	0.00E + 00 0.00E + 00	0.0E+00 1.0E-05 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	0.0E+00 4.5E-06 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	4.30E + 00 A

CANCER RISK

CANCER RISK

CANCER RISK

## TABLE 6-45B

6-151

- {

of exposure (sediments) were evaluated. The chemical-specific hazard quotients and the hazard indices calculated for each exposure route do not exceed unity for either land-use scenario even when maximum contaminant concentrations are evaluated. Also, if hazard indices are summed for the accidental-ingestion and dermal-contact exposure routes, the total hazard index does not exceed unity. Thus, adverse noncarcinogenic health effects are not anticipated under the conditions of the recreational or residential exposure scenarios defined in Section 6.4. If the days per year of exposure for the residential exposure scenario are increased from 270 to 350 (Table 6-46), the hazard index calculated for the accidental-ingestion exposure route exceeds unity when a young child is considered as the receptor of concern and maximum contaminant concentrations are evaluated. However, hazard indices calculated on a target organ or target effect basis would not exceed unity.

Cancer risks presented on Tables 6-44, 6-45, and 6-46 do not exceed  $1 \times 10^4$  in any of the cases presented. Arsenic and beryllium are the only COCs contributing to the estimated excess lifetime cancer risks. It should be noted that the maximum and average arsenic and beryllium concentrations detected in the sediment samples collected from the bordering wetlands are similar to or less than background arsenic and beryllium concentrations.

#### 6.5.3.19 Risk Assessment Results for Heard Pond

Heard Pond is located in Reach 7. It was evaluated as part of the remedial investigation because the Pond is reported to be occasionally flooded by the Sudbury River. These two surface water bodies are not otherwise connected. Four Nyanza Site contaminants were detected in the sediment samples collected during the RI: arsenic, chromium, mercury, and lead. The organic COCs were not detected. Surface water samples were not collected from Heard Pond.

Table 6-47 presents risk assessment results for COC concentrations detected in Heard Pond sediment samples. The accidental-ingestion and dermal-contact routes of exposure (sediments) were evaluated assuming recreational land/water use scenarios. The hazard quotients calculated for each COC and hazard indices calculated for each exposure route do not exceed unity. If hazard indices are summed for the accidental-ingestion and dermal-contact exposure routes, the total hazard index does not exceed unity. These results indicate that adverse noncarcinogenic health effects would not be anticipated for a receptor contacting the sediments under

W92194F

6-152

#### TABLE 0-40A RISK ASSESSMENT RESULTS FOR SEDIMENT EXPOSURE SCENARIOS BORDERING WETLANDS RESIDENTIAL NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

_____

							NONCAR	CINOGEN	C RISK ANA	V YSIS BESI	ULTS							
CONTAMINANTS of CONCERN		ITFATION NG	ECEPTO	FACTOR	<b>P</b> ₽FD	HAZ	ARD QUO	TIENTS: C	ню	HAZ	ARD QUO	TIENTS: TE	EN	HAZARI	DQUOTIENT	S: ADULT		TOXIC END~ POINT
	MAX	AVG	INGESTION	DEPMAL	(MG - KG -	ACCI				ACCIDEN	TAL ON	DERM	IAL. ACT	ACCID INGE	STION	DERI	MAL ACT	
				[	DAY	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	1
ARSENIC* CHROMIUM* LEAD* MERCURY* BARIUM BERYLLIUM COPPER MANGANESE NICKEL SELENIUM VANADIUM	11 9 101 254 7 6 67 5 19 83 4 1070 21 8 1 2 57 6	6 22 24 01 71 19 1 071 36 125 0 816 18 013 297 92 7 18 0 07 28 086	2 59E - 08 2 59E - 08 1 30E - 08 2 59E - 08	0 00E + 00 0 00E + 00 0 00E + 00 1 30E - 07 0 00E + 00 0 00E + 00	3 00E - 04 1 00E + 00 3 00E - 04 5 00E - 02 5 00E - 02 5 00E - 03 1 00E - 02 8 00E - 03 7 00E - 03	8 1E -01 1 3E -03 0 0E +00 3 2E -01 1 7E -02 4 9E -03 0 0E +00 1 4E -01 1 4E -02 3 1E -03 1 1 E -01	2.7E - 01 3.1E - 04 0.0E + 00 4.6E - 02 9.7E - 03 2.1E - 03 0.0E + 00 3.6E - 02 4.6E - 03 1.6E - 04 5.1E - 04	0 0E+00 0 0E+00	0 00E+00         0 00E+00         0 00E+00         1.1E-03         0 00E+00         0 00E+00	1.0E-01 2.6E-04 0.0E+00 6.6E-02 3.5E-03 9.6E-04 0.0E+00 2.6E-02 2.6E-03 6.2E-04 2.1E-02	5.4E-02 6.2E-05 0.0E+00 9.3E-03 2.0E-03 4.2E-04 0.0E+00 7.7E-03 9.3E-04 3.6E-05 1.0E-02	0.0E+00 0.0E+00 0.0E+00 3.3E-03 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 4.8E-04 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	2.7E-02 6.9E-05 0.0E+00 1.7E-02 9.2E-04 2.6E-04 0.0E+00 7.3E-03 7.5E-04 1.6E-04 5.6E-03	1.4E-02 1.8E-05 0.0E+00 2.4E-03 5.2E-04 1.1E-04 0.0E+00 2.0E-03 2.5E-04 9.8E-06 2.7E-03	0.0E+00 0.0E+00 1.7E-03 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 2.4E-04 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	SKIN LIVER CNS CNS BLOOD NS NS CNS BW SELENOSIS NS SLENOSIS
ZINC	127	41 078	2 59E - 08	0 00E+00	2 00E - 01	8 1E -03	2 0E - 03	0.0E+00	0 0E+00	1.6E-03	5.3E-04	0.0E+00	0.0E+00_	4.3E-04	1.4E-04	0.0E+00	0.0E+00	BLOOD
NYANZA SITE CONTÂMI OTHER SUDBURY RIVER ALL CHEMICALS OF CO	NANTS CONTAMIN NCERN	IANTS	. <u> </u>	HAZAPO IN HAZAPO IN HAZAPO IN	dex Dex	83E-01 29E-01 11E+00	3.1E-01 1.1E-01 4.2E-01	0.1E-03 0.0E+00	0 00E+00	1.7E-01 5 9E-02 2.3E-01	6.3E-02 2.2E-02 6.5E-02	3.3E-03 0.0E+00 3.3E-03	4.6E-04 0.0E+00 4.6E-04	4.5E-02 1.5E-02 6.0E-02	1.7E-02 5.8E-03 2.2E-02	1.7E-03 0.0E+00 1.7E-03	2.4E-04 0.0E+00 2.4E-04	
BEPH:BIS(2-ETHYL HEX	YL)PHTHAL	ATE		TOIOCITY E	NDPOIN TS A	BREVIATIO	ONS [.]	NS NOT	SPECIFIED	LAK: LIVE	R AND KID	INEY	BW: BODY	WEIGHT C	CNS: CENTR	AL NERVOUS	SYSTEM	

6-153

.

1

_____

# TABLE 6~46B RISK ASSESSMENT RESULTS FOR SEDIMENT EXPOSURE SCENARIOS BORDERING WETLANDS: RESIDENTIAL NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

				CARCINOGE	ENIC RISK A	NALYSIS F	RESULTS		
CONTAMINANTS of		TRATION (G	EXPOSURE	FACTOR		CANCER F	IISKS		CANCER SLOPE FACTOR
CONCERN	мах	AVG	INGESTION	DERMAL CONTACT	ACCIDEN	TAL DN	DERM CONT/	AL ACT	(MG/KG/D)1 / WEIGHT
······································					MAX	AVG	MAX	AVG	OF EVIDENCE
ARSENIC*	11.9	6.22	1.66E-06	0.00E+00	3.6E-05	1.9E-05	0.0E+00	0.0E+00	1.80E+00 A
CHROMIUM*	101	24.01	1.66E-06	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
LEAD*	254	71.19	8.29E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MERCURY*	7.6	1.071	1.66E-06	6.14E-08	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
BARIUM	67.5	36.125	1.66E-06	0.00E+00	0. <b>0E+0</b> 0	0.0E+00	0.0E+00	0. <b>0E+00</b>	
BERYLUUM	1.9	0.818	1.66E-06	0.00E+00	1.4E05	5.8E-06	0.0E+00	0.0E+00	4.30E+00 A
COPPER	83.4	18.013	1.66E-06	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MANGANESE	1070	297.92	1.66E-06	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
NICKEL	21.8	7.18	1.66E-06	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0. <b>0E+0</b> 0	
SELENIUM	1.2	0.07	1.66E-06	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
VANADIUM	57.5	28.065	1.66E-06	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
ZINC	127	41.078	1.66E-06	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	]

NYANZA SITE CONTAMINANTS	CANCER RISK	3.6E-05	1.9E-05	0.0E+00	0.0E+00
OTHER SUDBURY RIVER CONTAMINANTS	CANCER RISK	1.4E-05	5.8E-06	0.0E+00	0.0E+00
ALL CHEMICALS OF CONCERN	CANCER RISK	4.9E-05	<b>2.4E-0</b> 5	0.0E+00	0.0E+00

BEPH:BIS(2-ETHYL HEXYL)PHTHALATE

W92194F

6-154

#### TABLE 0 - 4( RISK ASSESSMENT RESULTS FOR SEDIME... EXPOSURE SCENARIOS HEARD POND NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

						NONCAR	NOOEN	RISK ANA	LYSIS RESL	ILTS						·	
CONCEN	TRATION	EXPOSURE RECEPTO	FACTOR R = TEEN	R¢D	HAZ	ARD QUO	NENTS CH	•LD	HAZ	ARD QUOT	IENTS: TE	EN	HAZARI	D QUQTIENT	S: ADULT		· TOXIC END- POINT
MAX	AVG	INGESTION	DERMAL CONTACT	(MG - KG - DAY)	ACCIL INGES	DENTAL TION	DE CO MAX	RMAL NTACT		AL DN AVG	DERM CONT/ MAX	AL NGT AVG	ACCID INGE	ENTAL STION AVG	DERM CONT MAX	AAL ACT AVG	
10.85 40.2 149 3.5	10 89 40 2 149 3 5	1 11E - 06 1 11E - 06 9 55E - 07 1 11E - 08	0 00E + 00 0 00E + 00 0 00E + 00 5 55E - 06	3 00E - 04 1 00E + 00 3 00E - 04	8 8E - 02 7 3E - 03 0 0E + 00 2 1E - 02	6 6E - 02 7 3E - 05 0 0E + 00 2 1E - 02	0 0E + 00 0 0E + 00 0 0E + 00 5 3E - 04	0 0E + 00 0 0E + 00 0 0E + 00 5 3E - 04	4 0E - 02 4 5E - 05 0 0E + 00 1 3E - 02	4 0E - 02 4.5E - 05 0 0E + 00 1.3E - 02	0 0E + 00 0.0E + 00 0 0E + 00 6 5E - 04	0.0E + 00 0.0E + 00 0.0E + 00 6.5E - 04	3.5E - 03 3.9E - 0 <del>0</del> 0.0E + 00 1.1E - 03	3.5E ~ 03 3.9E ~ 06 0.0E + 00 1.1E ~ 03	0.0E + 00 0.0E + 00 0.0E + 00 1.1E - 04	0.0E+00 0.0E+00 0.0E+00 1.1E-04	SKIN UVER CNS CNS
89 6 136 333 5 11 3 21 7 302	00 6 136 333 5 11 3 21 7 307	1 11E - 08 1 11E - 08	0 00E + 00 0 00E + 00	5 00E - 02 1 00E - 01 2 00E - 02 7 00E - 03 2 00E - 01	3 3E - 03 0 0E + 00 0 1E - 03 1 0E - 03 8 7E - 03 2 0E - 03	3 3E - 03 0 0E + 00 6 1E - 03 1 0E - 03 5 7E - 03 2 8E - 03	0 0E + 00 0 0E + 00	0 0E + 00 0 0E + 00	2 0E -03 0.0E +00 3 7E -03 6 3E -04 3 4E -03 1.7E -03	2.0E - 03 0.0E + 00 3.7E - 03 8.3E - 04 3.4E - 03 1.7E - 03	0 0E + 00 0.0E + 00 0.0E + 00 0.0E + 00 0 0E + 00 0.0E + 00	0.0E + 00 0.0E + 00 0.0E + 00 0.0E + 00 0.0E + 00 0.0E + 00	1.8E - 04 0.0E + 00 3.3E - 04 5.5E - 05 3.0E - 04 1.5E - 04	1.8E ~ 04 0.0E + 00 3.3E ~ 04 5.5E ~ 05 3.0E ~ 04 1.5E ~ 04	0.0E + 00 0.0E + 00 0.0E + 00 0.0E + 00 0.0E + 00 0.0E + 00	0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	BLOOD NS CNS BW NS BLOOD
NTS	ANTS		HAZARD IN	DEX	87E-02 19E-02	6 7Ē - 0Ž 1 9E - 02	5 3È-04 0 0E+00	5 3E - 04 0 0E + 00	5 3E - 02 1.1E - 02	5.3E - 02 1.1E - 02	8.5E - 04 0.0E + 00	8.5E-04 0.0E+00	4.7E-03 1.0E-03	4.7E ~ 03 1.0E ~ 03	1.1E-04 0 0E+00	1.1E-04 0.0E+00	]
	CONCEN MG/ MAX 10.85 40.2 140 35 40.2 140 35 10.8 3335 11.3 21.7 307 NTS ONTAMIN ERN	CONCENTRATION MG/KG MAX AVG 10.85 10.89 40.2 40.2 140 140 3.5 3.5 89.6 99.6 136 136 333.5 333.5 11.3 11.3 21.7 21.7 302 302 NTS ONTAMINANTS ERN	CONCENTRATION EXPOSURE MG/KG RECEPTON MAX AVG INGESTION 10.85 10.89 1 11E-08 40.2 40.2 1 11E-08 140 140 9 55E-07 3.5 3.5 1 11E-08 136 136 1 11E-08 133 135 1 11E-08 113 113 11E-08 217 217 1 11E-08 302 302 1 11E-08 NTS DNTAMINANTS ERN	CONCENTRATION         EXPOSURE         FACTOR           MG/KG         RECEPTOR = TEEN           MAX         AVG         INGESTION         DERMAL CONTACT           10.85         10.81         1.11E-06         0.00E+00           40.2         40.2         1.11E-06         0.00E+00           40.2         40.2         1.11E-06         0.00E+00           140         140         9.55E-07         0.00E+00           135         3.5         1.11E-08         0.00E+00           136         1.36         1.11E-08         0.00E+00           131         1.12         1.00         0.00E+00           21.7         2.17         1.11E-08         0.00E+00           302         302         1.11E-08         0.00E+00           307         302         1.11E-08         0.00E+00           307         302         1.11E-08         0.00E+00           NTS         HAZAPD INT         HAZAPD INT           DNTAMINANTS         HAZAPD INT	CONCENTRATION MG/KG         EXPOSURE RECEPTOR = TEEN         RED           MAX         AVG         INGESTION CONTACT         DERMAL CONTACT         (MG - KG - DAY)           10.85         10.83         1.11E - 06         0.00E + 00         3.00E - 04           40.2         40.2         1.11E - 06         0.00E + 00         1.00E + 00           140         140         9.55E - 07         0.00E + 00         1.00E + 00           135         3.5         1.11E - 08         0.00E + 00         1.00E - 04           136         1.36         1.11E - 08         0.00E + 00         1.00E - 01           131         1.11E - 08         0.00E + 00         1.00E - 01         1.00E - 01           1331         3.113         1.11E - 08         0.00E + 00         2.00E - 02           1335         3.335         1.11E - 08         0.00E + 00         2.00E - 01           113         1.11E - 08         0.00E + 00         2.00E - 02         2.00E - 02           207         302         1.11E - 08         0.00E + 00         2.00E - 01           307         302         1.11E - 08         0.00E + 00         2.00E - 01           NTS         HAZAPD INDEX         HAZAPD INDEX         ERN         HAZAPD INDEX	CONCENTRATION MG/KG         EXPOSURE PECEPTOR = TEEN         PFD           MAX         AVG         INGESTION DERIMAL CONTACT         IMG - KG - INGES         ACCID KG - INGES           MAX         AVG         INGESTION DERIMAL CONTACT         IMG - KG - INGES         ACCID KG - INGES           10.85         10.85         10.85         111E - 06         0.00E + 00         3.00E - 04         8.8E - 02           40.2         40.2         1.11E - 06         0.00E + 00         1.00E + 00         7.3E - 05           140         149         5.95E - 07         0.00E + 00         1.00E + 00         7.3E - 05           135         3.5         1.11E - 08         0.00E + 00         3.00E - 04         8.8E - 02           89.6         9.6         1.11E - 08         0.00E + 00         3.00E - 04         2.1E - 02           89.6         9.6         1.11E - 08         0.00E + 00         1.00E - 01         0.00E + 00           333.5         3.33.5         1.11E - 08         0.00E + 00         2.00E - 02         0.0E - 03           11.3         1.12         0.00E + 00         2.00E - 01         2.0E - 03         8.7E - 03           3027         302         1.11E - 08         0.00E + 00         2.00E - 01         2.0E - 03     <	CONCENTRATION         EXPOSURE         FACTOR         RFD           MG/KG         RECEPTOR = TEEN         HAZARD QUOT           MAX         AVG         INGESTION         DERIMAL CONTACT         (MG - DAY)         ACCIDENTAL INGESTION           10.85         10.89         1.11E-06         0.00E+00         3.00E-04         8.8E-02         8.8E-02         8.8E-02           40.2         40.2         1.11E-06         0.00E+00         1.00E+00         7.3E-05         7.3E-05           140         149         5.95E-07         0.00E+00         1.00E+00         7.3E-05         7.3E-05           136         1.36         1.11E-08         0.00E+00         3.00E-04         8.8E-02         2.8E-02           89.6         99.6         1.11E-08         0.00E+00         3.00E-04         2.1E-02         2.1E-02           136         1.36         1.11E-08         0.00E+00         5.00E-02         3.2E-03         3.2E-03           133         1.13         1.1E-08         0.00E+00         1.00E-01         8.7E-03         8.7E-03           302         302         1.11E-08         0.00E+00         2.00E-01         2.0E-03         2.8E-03           303         1.13         1.1E-08	CONCENTRATION         EXPOSURE         FACTOR         RFD           MG/KG         RECEPTOR = TEEN         HAZARD QUOTIENTS CH           MAX         AVG         INGESTION         DERMAL CONTACT         (MG - INGESTION         ACCIDENTAL CONTACT         DERMAL CONTACT           10.85         10.83         1.11E-06         0.00E+00         3.00E-04         8.8E-02         8.8E-02         0.0E+00           40.2         40.2         1.11E-06         0.00E+00         1.00E+00         7.3E-05         0.0E+00           140         149         5.95E-07         0.00E+00         3.00E-04         8.8E-02         8.8E-02         0.8E+00           3.5         3.5         1.11E-08         0.00E+00         3.00E-04         8.8E-02         2.1E-02         5.3E-04           88.6         9.8         1.11E-08         0.00E+00         3.00E-04         2.1E-02         2.1E-02         5.3E-04           3.35         3.35         1.11E-08         0.00E+00         3.00E-04         2.1E-02         0.0E+00         0.0E+00           3.35         3.35         1.11E-08         0.00E+00         1.00E-01         0.0E+00         0.0E+00         0.0E+00         0.0E+00         0.0E+00         0.0E+00         0.0E+00         0.0E	CONCENTRATION         EXPOSUME         FACTOR         RED           MG/KG         RECEPTOR = TEEN         RED         HAZARD QUOTIENTS CHILD           MAX         AVG         INGESTION         DERMAL CONTACT         (MG - LONTACT         ACCIDENTAL INGESTION         DERMAL CONTACT           10.85         10.81         1.11E-06         0.00E+00         3.00E-04         8.8E-02         8.8E-02         0.0E+00         0.0E+00           40.2         40.2         1.11E-06         0.00E+00         3.00E-04         8.8E-02         8.8E-02         0.0E+00         0.0E+00           148         149         5.5E-07         0.00E+00         1.00E+00         7.3E-05         0.0E+00         0.0E+00           3.5         3.5         1.11E-08         0.00E+00         3.00E-04         8.8E-02         3.8E-03         0.0E+00         0.0E+00           3.5         3.5         1.11E-08         0.00E+00         3.00E-04         2.1E-02         2.1E-02         5.3E-04         5.3E-04           136         1.36         1.11E-08         0.00E+00         1.00E-01         1.0E-03         0.0E+00         0.0E+00           133.5         3.35         1.11E-08         0.00E+00         2.00E-02         1.0E-03         0.0E+00	CONCENTRATION         EXPOSURE         FACTOR         RED           MG/KG         RECEPTOR = TEEN         RED         HAZARD QUOTIENTS CHILD         HAZARD           MAX         AVG         INGESTION         DERMAL CONTACT         (MG - KG - DAY)         ACCIDENTAL INGESTION         DERMAL CONTACT         ACCCIDENTAL INGESTION           10 85<10 R5	CONCENTRATION         EXPOSUME         FACTOR         PFD           MG/KG         RECEPTOR = TEEN         PFD         HAZARD QUOTENTS CHILD         HAZARD QUOT           MAX         AVG         INGESTION         DERMAL CONTACT         (MG - KG - INGESTION         ACCIDENTAL INGESTION         DERMAL CONTACT         ACCIDENTAL INGESTION         ACCIDENTAL INGESTION         ACCIDENTAL INGESTION         ACCIDENTAL INGESTION         ACCIDENTAL INGESTION           10 B5         10 B1         111E - 06         0 00E + 00         3 00E - 04         8E - 02         6 E - 02         0 DE + 00         0 DE + 00         4 0E - 02         4 0E -	CONCENTRATION MG/KG         EXPOSURE RECEPTOR = TEEN         PFD         HAZARD QUOTIENTS CHLD         HAZARD QUOTIENTS: TE           MAX         AVG         INGESTION CONTACT         DERMAL CONTACT         IMG - KG - DAY         ACCIDENTAL KG - DAY         DERMAL MAX         ACCIDENTAL CONTACT         DERMAL INGESTION         ACCIDENTAL CONTACT         DERMAL INGESTION         ACCIDENTAL CONTACT         DERMAL INGESTION         ACCIDENTAL CONTACT         DERMAL INGESTION         ACCIDENTAL INGESTION         DERMAL CONTACT         ACCIDENTAL INGESTION         DERMAL CONTACT         ACCIDENTAL INGESTION         DERMAL CONTACT         ACCIDENTAL INGESTION         DERMAL CONTACT           10 85         10 83         111E - 08         000E + 000         3 00E - 04         8 8E - 02         0 8E + 02         0 8E + 00         0 0E + 00         0 4 0E - 02         4 0E - 02         0 0E + 00           35         35         111E - 08         0 00E + 00         1 00E + 00         0 0E + 00	NORCARICINO DENDE NAL LEG PESULIS           CONCENTRATION MG/KG         EXPOSURE FACTOR RECEPTOR = TEEN         RFD         HAZARD QUOTIENTS CHLD         HAZARD QUOTIENTS: TEEN           MAX         AVG         INGESTION CONTACT         DERMAL CONTACT         (MG - CONTACT         ACCIDENTAL CONTACT         DERMAL CONTACT         CONTACT MAX         AVG         MAX         AVG         MAX         AVG           10.85         10.81         111E - 08         0.00E + 00         3.00E - 04         8.0E - 02         0.0E + 00         0.0E + 00	CONCENTRATION MORCA         EXPOSURE FACTOR RECEPTOR = TEEN         RED HE         HAZARD QUOTIENTS         CHLD         HAZARD QUOTIENTS: TEEN         HAZARD HAZARD QUOTIENTS: TEEN           MAX         AVG         INGESTION CONTACT         DERMAL CONTACT         (MG - CONTACT         ACCIDENTAL INGESTION CONTACT         DERMAL CONTACT         ACCIDENTAL INGESTION         DERMAL CONTACT	CONCARCINOLERC. NOR ANALISS INSTANCTION FROM TO THE OFFICE TO RESULTS           CONCARCINOLERC. NOR ANALISS INSTANCTION FROM TO THE PROPERTY OF THE OFFICE TO RESULTS           MAX         AVG         INGESTION         DEFINAL CONTACT         NOR CARCINOLERC. NOR ANALISS INSTANCTION         HAZARD QUOTIENTS: TEEN         HAZARD QUOTIENTS: HAZARD QUOTIENTS: TEEN           MAX         AVG         INGESTION         DERMAL CONTACT         INGESTION         CONTACT         INGESTION         CONTACT         INGESTION           10 85         10 85         1 11E - 06         0 00E + 00         3 00E - 04         INGESTION         CONTACT         INGESTION         CONTACT         INGESTION           10 85         10 85         1 11E - 06         0 00E + 00         3 00E - 04         INGESTION         AVG         MAX         AVG         MAX         AVG           10 85         10 85         9 05E - 07         0 00E + 00         3 00E - 04         I 00E + 00         0 0E + 0	CONCENTRATION         EXPOSURE         FACTOR         RED         HAZARD QUOTIENTS         CHLD         HAZARD QUOTIENTS:         TEEN         HAZARD QUOTIENTS:         ADULT           MAX         AVG         INGESTION         DEFIMAL CONTACT         INGESTION         CONTACT         INGE         AVG         MAX         AVG </td <td>CONCENTRATION         EXPOSUME         FACTOR         RED           MGX.Q         RECEPTORI = TEEN         HAZARD QUOTIENTS         CHLD         HAZARD QUOTIENTS: TEEN         HAZARD QUOTIENTS: ADULT           MAX         AVG         INGESTION         DERMAL CONTACT         MAX         AVG         HAZARD QUOTIENTS: TEEN         HAZARD QUOTIENTS: ADULT           MAX         AVG         INGESTION         DERMAL CONTACT         MAX         AVG         M</td>	CONCENTRATION         EXPOSUME         FACTOR         RED           MGX.Q         RECEPTORI = TEEN         HAZARD QUOTIENTS         CHLD         HAZARD QUOTIENTS: TEEN         HAZARD QUOTIENTS: ADULT           MAX         AVG         INGESTION         DERMAL CONTACT         MAX         AVG         HAZARD QUOTIENTS: TEEN         HAZARD QUOTIENTS: ADULT           MAX         AVG         INGESTION         DERMAL CONTACT         MAX         AVG         M

6-155

W92194F

1

;

# TABLE 6-47B RISK RESULTS FOR SEDIMENT EXPOSURE HEARD POND NYANZA OPERABLE UNIT 3 MIDDLESEX CO., MASSACHUSETTS

				CARCINOG	ENIC RISK A	NALYSIS F	RESULTS		CANCER
CONTAMINANTS of	CONCEN MG/I	TRATION	EXPOSUR	FACTOR	_	CANCER F	ISKS	_	SLOPE
CONCERN	МАХ	AVG	INGESTIO	DERMAL CONTACT		TAL DN		AL ACT	(MG/KG/D)-1 / WEIGHT
					MAX	AVG	МАХ	AVG	OF EVIDENCE
ARSENIC* CHROMIUM*	10.85 40.2	10.85 40.2	3.64E-07 3.64E-07	0.00E+00 0.00E+00	7.1E-06 0.0E+00	7.1E-06 0.0E+00	0.0E+00 0.0E+00	0.0E+00 0.0E+00	1.80E+00 A
LEAD* MERCURY*	149 3.5	149 3.5	1.82E-07	0.00E+00 1.51E-08	0.0E+00 0.0E+00	0.0E+00 0.0E+00	0.0E+00 0.0E+00	0.0E+00 0.0E+00	
BARIUM	89.6	89.6 136	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
MANGANESE	333.5	333.5	3.64E-07	0.00E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	
VANADIUM ZINC	21.7 302	21.7 302	3.64E-07 3.64E-07	0.00E+00 0.00E+00	0.0E+00 0.0E+00	0.0E+00 0.0E+00	0.0E+00 0.0E+00	0.0E+00 0.0E+00	

NYANZA SITE CONTAMINANTS	CANCER RISK	7.1E-06	7.1E-06	0.0E+00	0.0E+00
OTHER SUDBURY RIVER CONTAMINANTS	CANCER RISK	0.0E+00	0.0E+00	0.0E+00	0.0E+00
ALL CHEMICALS OF CONCERN	CANCER RISK	7.1E-06	7.1E-06	0.0E+00	0.0E+00

١

W92194F

FINAL

the conditions of the recreational exposure scenarios defined in Section 6.4.

Arsenic is the only COC contributing to the estimated cancer risk which approaches 1E-05. Arsenic concentrations detected in the sediments ( $C_{avg}$ =10.85 mg/kg) are similar to those detected in background sediment samples ( $C_{avg}$ =8.74).

#### 6.6 <u>Summary of Risk Assessment Results</u>

This section summarizes the results of the risk assessment conducted for the Sudbury River Study Area. The baseline public health assessment risk evaluated the carcinogenic and noncarcinogenic risks associated with human exposures to contaminated surface waters and sediments of the Sudbury River Study Area. Risks associated with the routine ingestion of fish taken from surface water bodies within the Study Area were also evaluated. Mercury and methyl mercury are the principal contaminants of concern. However, several other organic and inorganic contaminants, identified by the EPA as Nyanza Site contaminants (arsenic), were also detected in surface water, sediment, and/or fish tissue samples collected from the Study Area. These contaminants and other Sudbury River COCs (several pesticides, PCBs, metals) contribute significantly to the carcinogenic and noncarcinogenic risks estimated for some of the river reaches and surface water bodies evaluated.

Table 6-48 presents a summary of risk results for each compound in each reach of the study area these summary values are specific to media, but not specific to exposure pathway.

The following items summarize the results of the risk assessment of COC concentrations detected in fish tissue samples collected in the Study Area:

o The fish-ingestion exposure scenarios presented in the risk assessment considered a sports fisherman and subsistence fisherman as receptors of concern. Hazard indices calculated for all COCs detected in fish tissue samples collected during the RI exceed unity in at least one of the cases presented for each surface water body evaluated. The cases presented include an evaluation of maximum and average COC concentrations assuming that the sports fisherman and the subsistence fisherman are the receptors of concern.

W92194F

6-157

#### TABLE 6-48 SUMMARY OF RISK RESULTS NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

· · · · · · · · · · · · · · · · · · ·	<u> </u>		·				AISK VALU	9: FISH #	GESTION							
PARAMETER				BUDBUAY	AEBERVOIR					,		CEDAR SW	MP POND			
	8U85151	ENCE FIEH	ERMAN		6	iport fibh	EFMAN		SUBSIST	ENCE FISH	ERMAN		S	PORT FISH	ERMAN	
	HAZARO O	JOTIENT	CANCERIA	SK	HAZARO O	UOTIENT	CANCERR	19K	HAZARD QL	JOTIENT	CANCER R	ISK	HAZARD O	JOTIENT	CANCER R	ISK .
	MAX	AVG	MAX	<b>BVA</b>	MAX	EVA	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG
NAPHTHALENE* PHENOL* NITROBEN ZENE*	1 JE-04	1 <b>3E</b> - 04	0 0E + 00	0 QE + 00	1 7E-05	1.7E-05	0 0E + 00	0.0E + 00	1							
ANTIMONY" ARSENIC" CADMIUM"	2 7E-01	27E-01	e 3E-05	8 JE - 05	37E-02	3 7E-02	8 8E-06	6 6E-06	2.8E+00	1.9E+00	0.0E+00	0 0E + 00	3.8E-01	2.6E-01	0.0E + 00	0.0E + 00
CHROMIUM*	3 0E - 03	1 5E-03	0 OE + 00	0 0E + 00	5 3E-04	2 1E-04	0 0E + 00	0 0E + 00	87E-04 0.0E+00	2.9E-04 0.0E+00	0.0E + 00 0.0E + 00	0 0E + 00 0.0E + 00	1.2E-04 0.0E+00	3.9E-05 0.0E+00	0.0E+00 0.0E+00	0.0E+00 0.0E+00
MERCURY. METHYL MERCURY.	5 5E + 00 2 9E + 00	4 0E + 00 2 2E + 00	0 0E + 00 0 0E + 00	0 DE + 00 0 DE + 00	7 3E-01 4 0E-01	5 5E-01 3 0E-01	0 DE + 00 0 OE + 00	0 0E + 00 0 0E + 00	4.3E+01	3.3E+00	0 0E + 00	0 0E + 00	5.9E+00	4.5E-01	0 0E + 00	0.0E + 00
3/4 METHYL PHENOL METHYLENE CHLORIDE ACETONE BENZO(B)FLUORANTHENE BENZO(A)PYRENE ENDOSULFAN I ENDOSULFAN II ENDOSULFAN II	125-01	7 <del>8</del> 6 - 02	0 0E + 00	0 0E + 00	1 7E-02	1 OE - 02	0 DE + 00	0 0F + 00								
DIELDRIN 4.4-DDD	1 4E-02 0 0E+00	1 4E-02 6 0E+00	4 8E-08 8 8E-07	4 8E - 08 5 2E-07	18E-03	1 8E-03 0 0E+00	8 3E-07 1 3E-07	8 3E-07 7 1E-08								
4,4-DDE	0.0E+00	0 0E+00	8 1E-08 4 0E-07	3 0E-00	0 0E+00	0 0E + 00	1 1E-06 5 4E-05	5 3E-07								
ALPHA-CHLORDANE GAMMA-CHLORDANE ALDRW HEPTACLOR AROCHLOR 1240	2 3E-02	8 36-05	7 <b>6</b> E - 07	7 <b>6</b> E-07	3 1E-03	3 1E-03	1 OE-07	1 0E-07								
AROCLOR - 1254 AROCLOR - 1260 BARIUM COPPER	0 DE + 00	0 DE + 00	4 3E-04	<b>2 0</b> E-04	0 DE + 00	0.0E + 00	8 8E-05	3.9E-05								
MANGANESE	3 3E~05	1 4E-02	0 OE + 00	0 0E + 00	4 5E-03	2 0E - 03	0 0E + 00	0 0E + 00	A 15 OI	1 15	0.05.00	0.05.00	<b>1 5</b> 00	4 45 43	0.05.00	0.05.00
SELENIUM SILVER THALLUM	4 0E-01	1 SE-01	0 QE + 00	0 OE + 00	6 7E-02	1.7E-02	0 0E + D0	0 0E + 00	4.1E-01 1.1E+00 9.9E-01 1.5E+00	3.3E-02 8.5E-02 7.4E-02 9.5E-01	0.0E+00 0.0E+00 0.0E+00	0.0E+00 0.0E+00 0.0E+00	5.5E-02 1.5E-01 1.4E-01 2.1E-01	1 2E-02 1 0E-02 1 3E-01	0 0E + 00 0 0E + 00 0 0E + 00	0.0E+00 0.0E+00 0.0E+00 0.0E+00
VANADIUM ZINC	1 8E-01 4 2E-01	0 3E-02 2 2E-01	0 0E + 00 0 0E + 00	0 0E + 00 9 0E + 00	2 DE - 02 5 7E - 02	e eE-03 3 0E-02	0 0E + 00 0 0E + 00	0 0E + 00 0 0E + 00	3 9E-01	6 8E-02	0.0E+00	0.0E+00	5.3E-02	0 2E-03	0 0E +00	0.0E+00
SUM	1 0 0E + 001	4 66 + 90	8 QE-04	3 0E-04	0 3E-01	0 8E-01	0.0E-03	4 9E-05	5.1E+01]	6.4E+00	0.0E+00	0 0E + 00	0.0E+00	6.8E-01	0.0E + 00	0.0E+00

(

.

BOXED VALUES ARE HAZARD QUOTIENTS WHICH EXCEED UNITY

W92194F

6-158

ł

#### TABLE 6-SUMMARY OF RISK RESULTS NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS PAGE 2

	PARAMETER			<b>BOUTHVILL</b>	E PONÖ			
		808961		ERMAN		8		ERMAN
		HAZANO O	UCTIENT	CANCERR	15K	HAZARD O	UOTIENT	CANC
		MAX	AVG	MAX	DVA .	MAX	AVG	MAN
6-159	NAPHTHALENE* PHENOL* NITROBENZENE* ANTIMONY* ARSENIC* CADMIUM* CHROMIUM* LEAO* MERCURY* METHYL MERCURY* 3/4 METHYL PHENOL METHYL ENE CHLORIDE ACETONE BEPH BENZO(B)FLUORANTHENE BENZO(A)PYRENE ENDOSULFAN I ENDOSULFAN I ENDOSULFAN I ENDOSULFAN BULFATE DIELDRIN 4.4-DDD 4.4-DDD 4.4-DDD 4.4-DDT ALPHA-CHLORDANE GAMMA-CHLORDANE ALDRIN HEPTACLOR AROCHLOR 1248 AROCLOR - 1250 BARILM COPPER MANGANESE NICKEL BELENIUM	● 5E - 04 0 0E + 00 ( 4 0E + 00) ( 4 0E + 00) S 2E - 03 4 1E - 02	3 4E - 04 0 0E + 00 1 4E - 00 1 4E - 03 2 4E - 03 2 4E - 03	0 0E + 00 0 0E + 00 0 0E + 00 0 0E + 00 0 0E + 00	0 0E + 00 0 0E + 00 0 0E + 00 0 0E + 00 0 0E + 00	t 3E - 04 0 0E + 00 8 5E - 01 7 0E - 04 8 5E - 03	4 7E-05 0 0E+00 1 0E-01 7 0E-04 3 0E-03	0 OE 0 OE 0 OE 0 OE 0 OE
	NICKEL SELENIUM SILVER	4 1E-02	2 OE-02	0 0E + 00	0 0E + 00	8 5E-03	3 6E-03	

1 5E-01 8 3E-02

BOXED VALUES ARE HAZARD QUOTIENTS WHICH EXCEED UNITY

0000+00 0000+00

1

CANCER RISK

MAX

RISK VALUES FIGH INGESTION

AVG

0 0E + 00 0 0E + 00 0 0E+00 0 0E+00

0 0E+00 0 0E+00

0 0E+00 0 0E+00

20E-02 86E-03 00E+00 0.0E+00

**SUBSISTENCE FIGHERMAN** 

**AVG** 

HAZARD QUOTIENT

MAX

W92194F

0.0E+00

0.0E + 00

0.0E+00

0.0E+00

0.0E+00

4.4E-05

FINAL

ZINC

THALLIUM VANADIUM CANCER RISK

AVG

0.0E+00 0.0E+00

MÁX

SPORT FISHERMAN

AVG

HAZARD QUOTIENT

4.5E-01 1.3E-01

0.0E+00 0.0E+00 3 1E-04

5.6E-04 2.3E-04

15E-03 15E-03

1.2E-03 0 2E-04

3.8E-03 3.0E-03

MAX

0 0E+00 0 0E+00 5 9E-08 2.1E-08 0 0E+00 0 0E+00 8.1E-07 2.9E-07

MILL POND

AVG

CANCER RISK

0 0E+00 14E+00 0.0E+00 0.0E+00 1.2E+00 1.0E-01

0.0E+00 0.0E+00

0.0E+00 0.0E+00

0.0E+00 0.0E+00

MAX

3.3E+00 0.7E-01 0.0E+00 0.0E+00

0 0E+00 0 0E+00 2 2E-03 3 2E-04

0 0E+00 0 0E+00 4 2E-03 1.7E-03 0.0E+00 0 0E+00

1.1E-02 1.1E-02

89E-03 6.8E-03

2.7E-02 22E-02

425 +00 1 55 +00 * 0 05 +00 * 0 05 +00 * 0 05 +00 * 0 05 +00 * 0 05 +00 * 0 05 +00 * 0 14E +00 * 2.25-03 * 3.25-04 * 1.25 +00 * 2.05-01 ** 3.15-04 * 4.45-05

# TABLE 6–48 SUMMARY OF RISK RESULTS NYANZA OPERABLE UMIT 3 MIDDLESEX COUNTY, MASSACHUSETTS PAGE 3

W92194F

•	1						<b>AISK VALU</b>	es: Fish N	IGESTION				• •			
PARAMETER	·		1	<b>NËS E AVON</b>	1NO 2		···				1	RESERVOIR	11			
	\$U896				1	PORT FISH	EPMAN		SUBSIST	IENCE FIBH	ERMAN		s	iport fish	EFMAN	
	HAZAND G	DOTIENT	CANCERIA	isk.	HAZANO O	UOTIENT	CANCERR	IÐK	HAZARD Q	UOTIENT	CANCER R	ISK	HAZARD Q	UOTIENT	CANCER R	IBK /
	MAX	AVO	-MÁX -	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG
NAPHTHALENE*																
PHENOL*									5 2E-03	9.3E-04	0.0E + 00	0.0E+00	7.1E-04	1 3E-04	0.0E+00	0.0E+0
NITHOBENZENE"	1 4 28 - 011		0.05 + 00	0.05.400	1718.00	2 AF-01	0.0E ± 00	0.0E + 00								
ARS ENIC"	7 12 100	476-01	1 #F-03	1 1E-04	11111100	6 5E-02	2 5E-04	1 5E-05	5 2E-03	9 3E-04	0.0F+00	0 0E + 00	7 1E-04	1.3E~04	0.0F+00	0.0E+0
CADMIUM*	1 4E-01	8 1E-02	0 0E+00	0 0E+00	1 0E-02	1 1E-02	0 0E + 00	0 0E+00	5 2E-01	3 8E-01	1 2E-04	8.4E-05	7.0E-02	4 9E-02	1 6E-05	1.1E-0
CHROMIUM*	7 7E-03	7 9E - 04	0 0E + 00	0 0E + 00	1 1E-03	1 IE-04	0 0E + 00	0 0E + 00								
LEAD*	0 OE + 00	0 0E + 00	0 0E + 00	0 OE + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E + 00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0 0E + 00	0.0E+00	0 0E + 00
MERCURY*	3 4E+01	9 9E + 00	0 OE + 00	0 OE + 00	47E+00	1 3E + 00	0 0E + 00	0 0E + 00	1.0E+01	3.4E+00	0.0E+00	0 0E + 00	2.8E+00	4.6E~01	0.0E+00	0.0E + 00
METHYL MERCURY*	[ 1 0 <u>E + 0</u> 1]	7 DE +00	0 OE + 00	0 OE + 00		[ <u>1 1E + 00</u> ]	0 0E + 00	0 0E + 00	1.7E+01	4.5E+00	0.0E+00	0.0E+00	23E+00	0.1E-01	0.0E+00	0.0E + 00
3/4 METHYL PHENOL	1															
METHYLENE CHLORIDE																
ACETONE																
BEPH									2 2E-01	4 SE-02	2.7E-05	5.4E-08	3.1E-02	0.1E03	3.7E-08	7.3E-07
BENZO(B)FLUORANTHENE					1								ļ			
BENZO(A)PYRENE																
ENDOSULFAN 1								0.05.00								
	272-02	206-05	0.05+00	0.05+00	378-03	272-03	0.05+00	U VE + UU								
	146-02	145-02	4.05-00	A 85-08	1.05-03	1.85-03	A 16-07	# 3E-07								
	0.05+00	0.06+00	8 8E - 07	8 5E-07	0.000	0 0E + 00	1 3E-07	7.5E-08	0.0E+00	0.0E+00	2 8E-08	1.8E-07	0.0E+00	0.0E+00	3.8E07	2.5F-06
4.4-DDE	0 0E+00	0 0E + 00	1 36-05	3 8E-08	0 0E+00	0 0E+00	1 8E-00	4.9E-07	0.0E+00	0.0E+00	6.3E-08	3.5E-08	0.0E+00	0.0E+00	8.8E-07	4 8E-07
4,4-DDT	0 0E+00	8 0E + 00	4 OE - 07	2 4E-07	0 06 + 00	0.0E+00	5.4E-08	3 2E-08	0.0E+00	0.0E+00	2.0E-07	1.5E-07	0.0E+00	0 0E + 00	27E-08	2.0E-08
ALPHA-CHLORDANE	2 3E-02	1 9E-02	7 OE - 07	\$ JE-07	31E-03	2 6E - 03	1 OE - 07	8 6E-08	5.7E-02	3.7E-02	1.9E-08	1.2E-08	7.7E-03	5.1E~03	2.6E-07	1.7E-07
GAMMA-CHLORDANE	4 5E - 02	2 4E - 02	1 5E-08	8 OE-07	● 2E-03	3 3E-03	2 1E-07	1 1E-07	3.4E-02	2 JE~02	1.1E-08	7.6E-07	4.8E-03	3 1E-03	1 5E-07	1.0E-07
ALDRIN	4 5E-02	1 5E-02	9 9E - 08	3 2E - 00	• 2E-03	<b>2 0E</b> -03	1 3E-08	4.4E-07								
HEPTACLOR					1								1			
	0.05.00		3.36 .03	415.04	1	0.05.00	4.85 .04		0.05.00	0.0E .00	1 75 04	3.85 .06	0.05.00	0.05.00	2 4E 05	A.1.C. 04
AROCI OR - 1280	0.05.00		4 25 - 04	3 4F-04	0.05+00	0.0E+00	5 7E-04	47E-05	0.02+00	0 0E + 00	3 16-04	3.0E-05	0.0E+00	0.0E+00	4.9E-05	416-00
BARILM	1		~			0.05.00	07L-0J		45E-03	1.6E-03	0.0E+00	0 0E+00	62E-04	2 1E~04	0.0E+00	0.0E+00
COPPER											••••					
MANGANESE	1.00-01	1 IE-02	0 OE + 00	0 OE + 00	8 5E-02	1 5E-03	0 0E + 00	0 0E + 00	14E-01	3.1E-02	0 0E + 00	0 0E+00	2.0E-02	4.2E~03	0.0E+00	0.0E+00
NICKEL	1															
SELENIUM					1								1			
SILVER	2 0E-01	\$ 7E-02	0 OE + 00	0 0E + 00	2 7E-02	37E-03	0 0E + 00	0 0E + 00								
THALLIUM			0 OE + 00	0 0E + 00	1 3 0E+00	2 1E-01	0 0E + 00	0 0E + 00								
VANADIUM	7 2E - 01	• 2E - 02	0 DE + 00	0 0E + 00	31E-02	85E-03	0 0E + 00	0 0E + 00	3.0E-01	9.0E-02	0.0E+00	0 0E + 00	4.2E-02	1.3E~02	0.0E+00	U.0E+00
ZING	[[īsē +òó]	1 36 - 01	V 0E +00	V UE + 00	1 /E-01	1/6-02	U UE +00	U UE + 00	L1.82 +00]	2 26-01	0.02+00	U.UE + 00	2.40-01	2 85~02	U.UE + 00	U.UE +00
<b>A</b> 1 <b>A</b> 1	التتريقية الم			<b>-</b>	$L : \mathcal{L} \to \mathcal{D}$			_	r	(	_	<b>.</b>	· · · · · · · · · · · · · · · · · · ·			

(

FINAL

.

BOXED VALUES ARE HAZARD QUOTIENTS WHICH EXCEED UNIT Y

١

6-160

•

(

.

ì

#### TABLE 6~48 SUMMARY OF RISK RESULTS NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS PAGE 4

		·					RISK VALU	9. FIGH F	GESTION							
PARAMETER				AXONVILL	ê Mebêrvîl	A				- +	<del>ا</del> ''	AIPHAVEN	BAY			
		ENCE FOH	ERMAN		8	PORT FIBH	EFIMAN		8088181	ENCE FIBH	ERMAN		3	PORT FISH	FMAN	
	HAZAND OL	ROTHENT	CANCER R	SK.	HAZARO OL	OTIENT	CANCER A	8K	HAZARD O	JOTIENT	CANCER A	ŠK	HAZARD QL	JOTIENT	CANCER R	ak .
	MAX	AVO	MAX	<b>DVA</b>	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG
NAPHTHALENE*	0 4E - 01	4 46 - 02	9 0E + 00	0 0E + 00	8 8E-02	6 0E-03	0 0E + 00	0 0E + 00								
PHENOL* NITROBENZENE*	27E-04	1 OE - 04	0 OE + 00	0 0E + 00	37E-05	1 4E-05	0 0E + 00	0 0E + 00	1.9E-02 8 5E-02	2.4E-03 0.5E-02	0.0E+00 0.0E+00	0.0E+00 0.0E+00	2.5E-03 8.9E-03	3 3E-04 8.8E-03	0.0E+00 0.0E+00	0.0E+00 0.0E+00
ANTIMONY*	4 7E - 01	4 7E-01	6 DE + 00	0 0E + 00	0 5E-02	0 5E-02	0 0E + 00	0 0E + 00								
ARSENIC*	2 12-01	1 BE-01	4 8E - 05	4 ZE - 05	285-02	2 5E-02	8 8E-08	57E-08								
	8.05.01	A 95 - 64	0.05.00	0.05 + 00	7.06.04	125-04	0.05.400	0.05.00	4.85-03		0.05.400	0.05+00	0.35.04	A 45-05	0.05.00	0.05+00
	0.05.00	0.05+00	0.05+00	0.00 + 00	0 0F+00	0 0E+00	0 0E + 00	0 0E + 00	0 0E + 00	0.1E-04	0 0E+00	0 0E +00	0.05+00	0 0E+00	0.0E+00	0.0E+00
MERCURY*			0 0E+00	0 0E+00		3 0E-01	0 0E + 00	0 0E + 00	1.4E+01	4 8E+00	0 0E + 00	0.0E+00	2.0F+00	6 5E-01	0.0E+00	0.0E+00
METHYL MERCURY	6 2E + 00	1 8C + 00	0 0E + 00	0 0E + 00	4E-01	2 5E-01	0 0E + 00	0 0E + 00	54E+00	2.5E+00	0.0E+00	0.0E+00	7.4E-01	3.5E-01	0.0E+00	0.0E+00
3/4 METHYL PHENOL									6 2E-02	3 8E-03	0 0E + 00	0 0E + 00	7 0E-03	5.1E-04	0.0E+00	0.0E+00
METHYLENE CHLORIDE									3 2E-01	1 1E-02	6.1E-05	2 2E-08	4.3E-02	16E-03	8.3E-08	3.0E-07
ACETONE									2.2E-02	5.4E-03	0.0E+00	0 0E+00	3.0E-03	7.4E-04	0.0E+00	0.0E+00
BEPTI	4 98 - 92	1 16-05	3 3E - CO	146-00	# 3E-03	196-03	/ JE-0/	1.06-07	2.5E-UJ	2.2E-03	306-07	265-07	3.4E-04	3.0E-04	4 12-08	3.65-06
BENZU(B)FLUOMANTHEME	}								0.02+00		2.90-04	200-04	0.0E+00	0.05+00	4.06-05	J.0E-03
	41E-02	1 4F - 02	0.0F+00	0 OF + 00	8.5E-01	4 8E-03	0.0F + 00	0.0F + 00	002400	0 02 4 00	0.0L~UJ	8.0C-03	002400	0.02 400	1.22-03	1.20~03
ENDOBULEAN								002000								
ENDOBULFAN SULFATE	276-02	2 7E - 02	0 0E + 00	0 0E + 00	37E-03	37E-03	0 0E + 00	0 0E + 00								
DIBLORIN	14E-02	0 0E-03	4 8E - 08	2 JE - 08	10E-03	82E-04	0 3E-07	3 2E-07								
4,4-DDD	0 0€ + 00	0 0E + 00	2 OE - OE	6 5E-07	0 0E + 00	0 0E + 00	27E-07	6 9E-08								
4,4-DDE	0 0€+00	0 OE + 00	4 DE - 08	1 SE - 08	0 0E + 00	0 0E + 00	54E-07	2 0E - 07	0.0E+00	0.0E+00	7.1E-08	3 5E-08	0.0E+00	0 0E + 00	97E-07	4.8E~07
4,4-DDT	0.0€+00	0 DE + 00	4 OE - 07	2 OE - 07	0 0E + 00	0 0E + 00	54E-08	3 6E-08								
ALPHA-CHLORDANE	4 56 - 02	4 9E - 02	1 5E-00	1 5E-08	0 2E-03	8 2E-03	2 1E-07	2 1E-07								1
GAMMA-CHLORDANE ALDRIN	1 X - 01	1 9E-02	7 OE - 07	8 3E - 07	3 1E-03	2 OE-03	1 OE-07	0 8E-08								
HEPTACLOR	1 86-02	4 0E-03	1 8E-05	4 4E - 08	2 2E-03	0 2E-04	21E-00	0 0E-07								
AROCLOR-1254	0.0€+00	6 0E + 00	1 8E - 04	0 OE - 05	0 05 + 00	0 0E + 00	2 4E-05	t 2E-05	0.0E+00	0.0E+00	2.1E-03	6 6E-04	0.0E+00	0 0E+00	28E-04	1.2E-04
AROCLOR-1200	0 0E + 00	8 0E+00	4 BE-04	21E-04	0 DE + 00	0 0E+00	07E-05	2 8E-05	0 0E+00	0.0E+00	1.3E-04	4.0E-05	0.0E+00	0.0E+00	1.8E-05	8.2E-08
BARILIM				•					3.0E-02	1.7E-02	0.0E+00	0.0E+00	4.1E-03	2.4E-03	0.0E+00	0.0E+00
COPPER .																
MANGANEBE	27E-01	2 0E - 02	0 0E + 00	0 0E + 00	3 8E-02	3 0E-03	0 0E + 00	0 0E + 00	4.5E-01	3 9E-02	0.0E+00	0 0E + 00	6.1E-02	5.3E-03	0.0E+00	0.0E+00
NICKEL	0 4E - 02	1 8E - 02	0 0E + 00	0 0E + 00	1 1E-02	2 1E-03	0 0E + 00	0 0E + 00	1.4E-01	2.2E-02	0 0E + 00	0 0E + 00	1 8E-02	3.0E-03	0.0E + 00	0.0E+00
SELENIUM	17E-01	4 4E-02	0 0E + 00	0 OE + 00	2 3E-02	01E-03	0 0E + 00	0 0E + 00	7.3E-02	2 5E-02	0.0E+00	0.0E+00	1.0E-02	3.5E-03	0.0E+00	0.0E+00
SILVER	1 1E-01	2 4E-02	0 0E + 00	0 0E + 00	1 6E-02	3 2E - 03	0 0E + 00	0 0E + 00								
THALLIUM	2 3E+01	1 1E+00	0 0E + 00	0 0E + 00	3 1E+00	1 6E-01	0 0E + 00	0 0E + 00								
VANADIUM	2 4E - 01	1 JE - 02	0 0E + 00	0 0E + 00	1 2E-02	1 1E-02	0 0E + 00	0 0E + 00								
ZINC	0 4E - 01	1 2E-01	0 0E + 00	0 0€ + 00	8 8E-02	1 7E-02	0 0E +00	0.0E+00	1.4E~01	0.1E-02	0.0E+00	0.0E+00	2.0E-02	8.3E-03	0 0E + 00	0.0E+00
8UM	<u>H 2.46+01</u> H	8.1E+007	7 86-94 *	2 DE-04	47E+00	6.9E-01 *	• 1.0E-04 •	4.0E-05	1.0E+01	5.0E+00*	• 2.0E-03 •	1.3E-03	2 2E+00	6.8E-01 **	3.6E-04 *	1.8E-04

W92194F

FINAL

{

#### TABLE 6 - 48 SUMMARY OF RISK RESULTS NYANZA OPERABLE UMIT 3 MIDDLESEX COUNTY, MASSACHUSETTS PAGE 5

							RISK VALUE	S: BEDIME	NT EXPOSUR	E						
PARAMETER	•	NEACH IND BACKO	POUND			REACH	1		1	EASTERN W	ETLANDS			ASTERN W DRILLING	ETLANDS	
	HAZANDOL	DOTIENT		ME	HAZARD Q	UOTIENT D	CANCER	ALSK ME	HAZARD QI CHIL		CANCER	AISK IME	HAZARDQ	DOTIENT	CANCER	RISK ME
	HAX I	AYQ	- KAX	DYA	MAX	AVQ	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG
TRICHLOBOETHENE									0.0E+00	0.0E+00	2 1E - 09	115-09	0.05+00	0.0E+00	4 2E - 08	2.750(
12DICHLOROETHENE	1				1 JE-05	3 8E - 08	0 0E+00	0 0E + 00	5 3E-05	3 0E-05	0.0E+00	0.0E+00	2.3E-03	1.1E-04	0.0E+00	0.0E+00
CHLOROBENZENE*					6 5E - 08	19E-00	0 0E + 00	0 0E + 00	3 3E-04	1.7E-04	0.0E+00	0.0E+00	7.0E-03	5.4E-04	0.0E+00	0.0E+00
NITROBENZENE"									30E-03	3.0E-03	0.0E+00	0.0E+00	6.4E-04	4.6E-04	0.0E+00	0.0€+00
1,2-DICHLOROBENZENE	6			1	4 0E - 05	2 2E - 05	0 0E + 00	0 0E + 00	1.6E04	1.0E-04	0.0E+00	0.0E+00	3 3E - 04	4.6E-05	0.0E+00	0.0E+00
1,3-008*													2.2E-05	2.1E-05	0.0E+00	0.0E+00
1,4-DCB*					0 0E +00	0 0E + 00	14E-08	79E-08	00E+00	0.0E+00	2.0E-08	1.5E-00	0.0E+00	0.0E+00	3.8E-00	7.2E-06
1,2,4- THICHLOHOUENZENE					1 36 - 03	115-04	0.05+00	0.05+00	54E-03	346-03	0.06+00	0.0E+00	0.0E-04	0.36-04	0.0E+00	0.000 + 00
PHENOI®					. IE - 04	3.12-04	0.05 4.00	002400	1.20~00	0.02-04	0.02 400	0.05+00	4.2E-05	4.20-00	0.02 +00	0.00 + 00
ARSENIC*	136-01	6 3E - 02	146-05	6 7E - 08	0 1E-02	4 2E - 02	9 8E - 06	4 6E - 06	7.7E-02	4 OE - 02	8.3E-08	4 3E-06	4 3E - 02	1.0E-02	4.6E - 08	1.1E-0
ANTIMONY	37E-02	2 0E - 0E	0 0E+00	0 0E + 00												
CADMIUM*													1.2E-02	2.4E-03	0.0E+00	0.0E+00
CHROMIUM*	10E-04	4 1E - 05	0 0E + 00	0 0E + 00	3 8E - 04	8 2E - 05	0 0E + 00	0 0E + 00	8.4E-04	2.3E-04	0.0E+00	0.0E+00	7.7E-04	7.8E-05	0.0E+00	0.0E+00
LEAD	8 0E + 00	0 OE + 00	0 0E + 00	0 0E + 00	0 0E + 00	0 OE + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00
MERCURY	0 0E - 03	1 7E - 03	0 0E + 00	0 0E + 00	1 BE - 01	2 4E - 02	0 0E+00	0.0E+00	9.5E-01	2.2E-01	0.0E+00	0 0E + 00	5.7E-01	3.8E - 02	0.0E+00	0.0E+00
MONOMETHYLHG* DIMETHYLHG*					5 ØE - 03	1 0E-03	0 0E + 00	0 0E + 00	4.1E~03	1.2E-03	0.0E+00	0.0E+00				
VINYL CHLORIDE	1												1			
BENZENE													0.0E+00	0.0E+00	1.3E - 10	1.3E-10
DICHLOROMETHANE					6 6E - 07	4 1E-07	€7E-11	6 0E - 11					1.1E-04	9.6E - 08	1.3E08	1.2E-06
ACETONE	4 1E - 07	4 1E - 07	0 0E + 00	0 0E + 00	1 3E - 05	5 OE - OS	0 0E + 00	0 0E + 00	6.6E05	3.7E - 05	0.0E+00	0.0E+00	4.5E-08	1.4E-08	0.0E + 00	0.0E+00
BEHP				1	2 3E - 04	7 4E - 05	14E-08	4 7E - 09					4.1E-08	2.6E-08	2.6E - 10	1.6E - 10
J/4 - METHYLPHENOL									12E-05	1.2E-05	0.0E+00	0.0E+00	3.4E-09	3.4E-08	0.0E + 00	0.0E+00
					9.0E 04		0.05.00	0.05.00					0.05 OF	4.45 05	0.05.00	0.0E . 04
AUENAPIPITLENE Disenantindene	1	7 75 - 06	0.05 + 00	0.05.00	200-04	2.0E-04	0.000	0.05+00	125-04	3 26 - 04	0.0E + 00	0.05.00	236-05	1.05-03	0.000+00	0.000+00
FILIOBANTHENE	1.00.00	115-05	0.05 +00		4 86 - 04	10F-04	0.02+00	0.00 + 00	0.2E-04	3.20-04	0.00 + 00	0.0E+00	1.12-05	4.02-03	0.0E+00	0.000+00
PYRENE	1.05-05	125-06	0.00 + 00	0.05.00	6 1E-04	1 1E-04	0.0E+00	0.0E+00	12E-04	B 6E - 05	0.0E+00	0.0E+00	175-08	17E-08	0.00 + 00	0.05+00
BENZO(A)ANTH					0 0E+00	0 0E+00	1.1E-05	2 BE-08	1.66 - 04	0.02, -00	0.02 100	002100	0.0E+00	0.0E+00	4.8E-07	3 2F - 07
CHRYSENE	0.02+00	0 0E+00	4 ØE - 07	4 1E-07	0 0E+00	0 0E + 00	1.1E-05	2 6E-08	0.0E+00	0.0E+00	3.1E~08	2.6E - 06	0.0E+00	0.0E+00	2.5E-07	2.0E-07
BENZO(B)FLUOR	8 0E + 00	0 0E + 00	\$ 1E-07	4 7E - 07	0 0E+00	0 0E + 00	8 0E - 08	2.6E-08	0.0E+00	0.0E+00	0.4E-00	7.1E-08				
BENZORIFLUOR				_	0 0E+00	0 0E + 00	\$7E-08	2.5E-08					0.0E+00	0.0E+00	2.2E - 08	7.18-07
BENZO (A) PYRENE	8 0E+00	0 0E + 00	2 3E - 07	2 3E - 07	0 0E + 00	0 0E + 00	0 0E - 08	2.8E-08	0.0E+00	0.0E+00	2.2E-08	2.1E-00	0.0E+00	0.0E+00	3.1E-07	3.1E-07
IN(123-CO)PYPENE				i	0 0E+00	0 0E+00	2 3E-00	1.0E-08								
DIBENZ(AH)ANTH					0 0E + 00	0.0E+00	2.4E-07	2.2E-07					0.0E+00	0.0E + 00	3.1E-07	3.1E-07
BENZO(GHI)PERYL					3 3E - 04	3 OE - 04	0 0E + 00	0 0E + 00								
BARIUM	0 86 - 03	2 4E - 03	0 02 + 00	0 0E + 00	7 6E-03	2 0E-03	0 0E+00	0 0E+00	21E-03	1.2E-03	0.0E+00	0.0E+00	2.4E-03	1.2E-0J	0.0E+00	0.0E+00
	0 00 - 04	3 1E - 04	3 WE - CU	1 JE - 00	6 9E ~ 04	3 ZE - 04	J 0E - 06	1.4E ~ 05	1.5E-03	6.1E-04	6.3E - 06	3.5E - 08	3.0E-03	8.8 <u>2</u> - 04	1.36 - 05	2.8E-00
		7.66	A 68 - M		7 84 _ 84	1.05-00	0.0E + 00	0.0E . 00	2 4E - 00	8 3E A4	0.05 + 00	0.05.00	A 75 _ 01	1.4502	0.0E + 00	0.05.00
NICKEI	475.00	105-01	0.06+00		175-02		0.05+00	0.05+00	3.75-02	0.36-03	0.02 +00	0.06+00	215-03	8 8F-04	0.0E+00	0.05+00
BILVER						• ••• - ••		U UL V UU	J., L - 00	w.w., - w4	0.02 700	0.06 7 00				
SELENUM	1 16-65	3 8E - 84	0 0E + 00	0 0E+00					24E-03	5 2E-04	0 0E + 00	0.0E+00				
THALLIUM													3.7E-02	1.2E-02	0.0E+00	0.0E+00
VANADIUM	136-08	6 XE - 03	0 0E + 00	0 0E + 00	1 2L - 02	6 2E - 03	0 0E + 00	0 0E + 00	9.6E - 03	4.6E-03	0.0E+00	0 0E + 00	5.9E-03	3 0E - 03	0.0E+00	0.0E+00
ZINC	\$ 7E-03	1 2E - 03	0 0E + 00	0 0E + 00	3 0E - 03	1 2E - 03	0 0E + 00	0 0E + 00	1.6E - 03	6 0E - 04	0.0E+00	0.0E + 00	2.6E-03	3 9E - 04	0.0E + 00	0.0E+00
4,4-DDE	J				0 0E + 00	0 0E + 00	3 6E - 09	2 5E - 09								
4,4-DDD					0 0E + 00	0 0E + 00	1 4E - 00	14E-09	0.0E + 00	0 0E + 00	7.8E - 09	3.3E - 09				
4,4-DDT					0 0E + 00	0 0E + 00	8 6E - 08	1.1E-08								
CHLORDANE	1															
AHOCLOH 1254	1															
8UM	1 35 - 61	1 OF -01	1 <b>BF</b> - 06	6 26 - na	3 OF - 01	0 5E - 02	8 8E - 05	215-00		2 BE-01	2.6E-05	2.0E-05	7.0E-01	7.3E-02	2.1E-05	5.8E - 04
Yem .																

l

BOXED VALUES ARE HAZARD QUOTIENTS WHICH EXCEED UNITY

(

FINAL

.

• (

#### TABLE 6 -48 SUMMARY OF RISK RESULTS A PERABLE UMT 3 M. LESEX COUNTY, MASSACHUSETTS PAGE 6

W92194F

6-163

								(): BEDIME		5						
PARAMETER		OULVENT				SUTFALL C	ÆEK			PACEW	AY		COLD	PRING BR	DOK	
	HAZANDO	UOTNENT	CANCER	August Me	HAZARD Q	DOTIENT	CANCER	<b>AISK</b>	HAZARD QI		CANCER	AISK ME	HAZARDQI	JOTIENT	CANCER	AISK .
	HAX	AVG	MAX	DYA	MAX	AYG	MAX	AYG	MAX	DYA.	MAX	AVG	MAX	AVG	MAX	AVG
	0.05.00	0 0E + 00	8 8E - 10	4 BE - 10	0.0E+00	0 0E + 00	4 BE - 11	3 2E-08	0.0F+00	0.0F+00	5 5E~07	215-07				
12DICHLOROETHENE	185-05	1 OE - 05	0 0E+00	0 0E+00	4.1E-07	4.1E-04	0 0E+00	0 0E + 00	5.3E-04	2.3E-04	0.0E+00	0.0E+00	1			
CHLOROBENZENE*	1 0E - 05	1 1E - 05	0 0E + 00	0 0E + 00					4.0E-03	1.2E-03	0.0E+00	0.0E+00				
NITROBENZENE*	146-03	1 4E - 03	0 0E + 00	0 0E + 00	2.3E-03	1.6E - 03	0 0E + 00	0.0E+00					1			· · ·
1,2-DICHLOROBENZENE	4 8E-05	4 2E - 05	0 0E + 00	0 0E + 00	7.4E-08	6.3E-06	0.0E+00	0 0E + 00	1.6E04	1.1E-04	0.0E+00	0.0E+00	1			
1,3-DCB*	7 0E - 00	7 6E - 08	0 0E + 00	0 OE + 00								•				
1,4-DCB*	0 0E + 00	0 OE + 00	7 3E - 08	5 7E - 00									1			
1,2,4-TRICHLOROBENZENE	1 6E - 03	1 6E - 03	0 0E + 00	0 OE + 00	8 3E - 03	1.4E - 03	0 0E + 00	0 0E + 00					·			
NAPHTHALENE*	3 1E - 04	2 0E - 04	0 OE + 00	0 OE + 00	1.7E-04	1.1E-04	0 OE + 00	0 0E + 00	3.4E03	2.1E-03	0.0E+00	0.0E+00				
PHENOL*																
ARSENIC*	4 06 - 02	2 E - 02	4 3E-00	2 8E - 06	\$ 0E - 02	1.8E-02	2.7E-08	1.9E-08	2.3E-01	1.6E - 01	2.4E-05	1.6E-05	2.6E-02	2.1E-02	2.6E-08	2.3E-08
ANTIMONY																
CADMIUM"	J								2.0E-02	8.3E - 03	0.0E+00	0.0E+00	]			
CHROMIUM*	2 9E - 04	1 BE - 04	0 OE + 00	0 0E + 00	1 0E-03	8 2E - 04	0 0E + 00	0.0E+00	3.8E - 04	2.8E-04	0.0E+00	0 0E+00	3.1E-05	2.8E-05	0.0E + 00	0.0E+00
LEAD*	0 0E+00	0 OE + 00	0 0E + 00	0 0E + 00	0 0E+00	0 0E + 00	0 0E + 00	0 0E + 00	0 0E+00	0 0E + 00	0.0E + 00	0 0E + 00	0.0E+00	0.0E+00	0.0E+00	0.0E+00
MERCURY	446-08	4 2E - OE	0 0E + 00	0 0E + 00	0.2E-01	2 2E - 01	0 0E + 00	0 0E + 00	6.1E-03	4.4E - 03	0.0E+00	0.0E + 00	1			
Monomethylhg" Dimethylhg"																
VINYL CHLORIDE																
BENZENE	0.05+00	0 OE + 00	6 5E - 11	6 DE 11												
DICHLOROMETHANE					3 0E-08	1.6E - 06	4 6E 10	1.9E - 10	4.2E-06	4.2E - 06	5.2E-10	5.2E - 10				
ACETONE	1				1.4E-08	€ 5E - 07	0 0E+00	0 0E + 00	31E-05	2.0E - 05	0.0E+00	0.0E+00	1			
BEHP	2 3E - 06	1 7E - 05	1 4E - 08	1 16-09	2 3E-04	1.1E-04	14E-08	0.9E - 09	29E-04	2 9E - 04	1.8E - 08	1.8E-08				
1/4-METHYLPHENOL 7-METHYLNAPH																
ACENAPTHYLENE	7 26 - 06	7 2E - 06	0 0E + 00	0 0E + 00	1 8E - 04	14E+04	0.0E+00	0 0E + 00								
PHENANTHRENE	418-04	3.86-04	0 0E+00	0 0E + 00	4 BE-04	3 2E - 04	0 0E+00	0 0E + 00	8 2F - 03	6 2E - 03	0.0F+00	0.0E+00				
LUORANTHENE	726-06	8 7E - 05	0 0E+00	0 0E+00	6 2E-05	67E-05	0 0E+00	0 0E + 00	10E-03	7.3E-04	0.0E+00	0 0E+00				
PYRENE	1 2E - 05	8 2E - 05	0 0E+00	0 0E + 00	1.7E-04	1.2E-04	0 0E+00	0 0E + 00	8 9E-04	6 4E - 04	0.0E+00	0 0E+00				
BENZOAIANTH	006+00	0 0E + 00	2 1E-00	1 8E-06	0 0E+00	0 0E+00	2 7E-08	2 0E - 08	0.0E+00	0 0E+00	2.6E-05	1.5E-05				
CHRYBENE	0 0E+00	0 0E + 00	1 0E - 00	1 8E-08	0 0E + 00	0 0E + 00	3 6E - 06	27E-08	0.0E+00	0.0E+00	2.2E-05	1.5E-05				
BENZORBIFLUOR	0.0E+00	0 0E + 00	2 1E-00	2 1E-08	0 0E + 00	0 0E + 00	4 6E - 08	3.1E-06	0.0E+00	0.0E+00	1.3E-05	1.1E-05				
BENZOKIFLUOR	0 0E+00	0 0E + 00	2 OE - 08	17E-08	0 0E + 00	0 0E+00	2 6E - 08	1 2E-08	0.0E+00	0.0E+00	1.6E - 05	1.1E-05				
BENZO(A)PYRENE	0 0E+00	0 0E + 00	2 0E - 08	1 0E-00	0 0E + 00	0 0E+00	3.1E~08	2 3E - 08	0 0E+00	0.0E+00	1.2E-05	9.9E - 08				
N(123-CD)PYRENE	0 06+00	0 0E + 00	8 4E - 07	6 4E - 07	0 0E + 00	0 0E + 00	1.0E00	1.2E-08								
DIBENZIAHIANTH																
BENZO(GHI)PERYL	146-04	1 4E - 04	0 0E + 00	0 0E + 00	3 8E - 04	2.6E-04	0 0E + 00	0 0E + 00								
BARIUM	0 SE - 04	0 0E - 04	0 0E + 00	0 0E + 00	21E-03	1.2E-03	0 0E + 00	0 0E + 00	2.6E-03	1.6E-03	0.0E+00	0.0E+00	2.4E-03	2.2E-03	0.0E+00	0.0E+00
BERYLLIUM	2 08-04	1 0E - 04	8 BE - 07	7 OE - 07	8 0E-04	3.9E-04	3.4E-08	1.7E-08	3.7E - 03	2.6E - 03	1.6E - 05	1.1E+05	0.9E-04	3.7E-04	3.0E-08	1.8E-08
DOPPER													ł			
MANGANESE	1 HE-00	1 7E-03	0 0E + 00	0 0E + 00	6 E - 03	3 OE - 03	0 0E + 00	0.0E+00	1.8E-02	6.3E-03	0.0E+00	0 0E + 00	1.0E-02	1.2E-02	0.0E+00	0.0E+00
NICKEL									1.7E-02	8.2E - 03	0.0E+00	0.0E+00				
BILVER	1															
BELENIUM								1								
THALLIUM	1															
ANADIUM	47E-03	3 IE-03	0 0E + 00	0 0E + 00	94E-03	4 6E - 03	0.0E+00	0.0E+00	1.3E-02	1.2E-02	0.0E+00	0 0E + 00	1.1E-02	8.1E-03	0.0E+00	0.0E+00
ZINC	84€-04	8 5E - 04	0 0E + 00	0 0E + 00	3 6E - 03	1.6E - 03	0 0E + 00	0.0E+00	4 9E - 03	3.3E - 03	0.0E+00	0.0E+00	1.5E-03	1.4E03	0.0E + 00	0.0E+00
1,4-DDE	I .								0 0E + 00	0.0E+00	1.3E - 09	1.6E - 09				
1,4-DDD	1								0 0E+00	0.0E+00	4.8E-09	1.4E-09				
I,4-DOT								1								
CHLORDANE																l
ROCLOR 1254									0.0E+00	0.0E+00	4.9E-07	2 8E - 07				
		_	_	_			_									
JUM	1 8 8E - 02	7.6E-02	1 6E - 05	1.4E-05	67E-01	2.5E-01	2.4E-05	1.6E-05	3.3E - 01	2.1E-01	1,3E-04	8.9E-05	5.7E-02	4.6E-02	5.6E-08	3.9E - 08

Ę

1

•

#### TABLE 6 – 40 SUMMARY OF NSK RESULTS NYANZA OPERABLE UMIT 3 MIDDLESEX COUNTY, MASSACHUSETTS PAGE 7

						1	PIBK VALUE	s: BEDIME	NT EXPOBUR	E				:		
PARAMETER		TREACH	3			REACH	•			REACH	â			REACH	16	· · · ·
	HAZARDO	DOTIENT	CANCER	Ansix Me	HAZARD O	DOTIENT	CANCER	AISK ME	HAZARD Q	UOTIENT	CANCER	FISK ME	HAZARD QI	UOTIENT	CANOER	ANGK
		DYA	MAX	DVA	MAX I	DVA	MAX	AVQ	MAX	AVG	MAX	AVG	MAX	AVQ	MAX	AVG
TRICHLOROETHENE 120ICHLOROETHENE CHLOROBENZENE* NITROBENZENE* 1,2-DICHLOROBENZENE 1,3-DCB*																
1,4-DC8* 1,2,4-TRICHLOROBENZENE* NAPHTHALENE* PHENDL*	9 0E + 00 3 2E - 04 6 2E - 05	0 DE + 00 2 2E - 04 8 2E - 05	1 6E - 08 8 0E + 00 8 0E + 00	1 6E - 09 0 0E + 00 0 0E + 00												
ARSENIC*	1 3E-01	4 66 - 02	146-05	6 OE - 06	\$ 0E-01	0 6E - 02	2.1E~05	7.0E - 08	6 6E - 02	2.3E02	0.0E - 00	2.5E-06	1.5E-01	6.8E - 02	1.6E-05	7.3E~08
CADMILM	734-04	1 95-02	0 0E +00	0 0E +00	\$4E-02	1 OE - 02	0 0E + 00	0 0E + 00					5.0E-02	1.4E-02	0.0E+00	0.0E+00
	4 86-00	8 XE - 04	0 0E+00	0 0E + 00	4 1E-04	1 3E - 04	0 0E+00	0 0E+00	1.1E-04	3.3E-05	0.0E+00	0.0E+00	5.1E-04	1.2E-04	0.0E+00	0.0E+00
MERCURY	34€-01	1 0E - 01	0 0E + 00	0 0E + 00	4 66-02	216-02	0 0E+00	0.0E+00	2.02-02	8.1E-03	0.0E+00	0.0E+00	1.1E-01	2.1E-02	0.0E+00	0.0E+00 0.0E+00
MONDMETHYLHG. DIMETHYLHG.	47E-04 34E-04	4 7E - 04 1 JE - 04	0 0E + 00 0 0E + 00	0 0E + 00 0 0E + 00	1.4E-03	4 8E - 04	0 0E + 00	0 OE + 00								
VINYL CHLORDE BENZENE DICHLOROMETHANE ACETONE	1 JE - 00	84E-08	0 0E+00	0.0E+00	1.1E-04	6 0E - 05	0 0E + 00	0.0E+00	0.4F . 05	1.45 .05	0.4E - 60		2.3E-05	1.6E-05	0.0E+00	0.0E+00
34-METHYLPHENOL 2-METHYLNAPH	102-04	W UE - 08	U UC - 00	57E-UU					3 8E - 08 5 9E - 08	3.8E-08 5.9E-08	2.4E-09 0.0E+00	2.4E-08 0.0E+00	2.25-04	1.6E-04	1.46-08	1.0E-06
ACENAPTHYLENE	4 92 - 04	4 2E - 04 7 0E - 04	0 0E+00	0 0E+00	A 75-05	8.75 - 05	0.05+00	0.05+00	8.2E-05	8.2E~06	0 0E+00	0 0E+00	# 2E_04	A 05 - 04	0.05+00	0.05+00
FLUORANTHENE	\$ 2E-04	1 SE-04	0 0E + 00	0 0E +00	1 4E-05	1 3E - 05	0 0E+00	0 0E + 00	5.7E-05	8.7E-05	0.0E+00	0.0E+00	1.3E-04	1.0E-04	0.0E+00	0.0E+00
PYRENE	7 06-04	17E-04	0 0E+00	0 0E+00	1 0E-05	17E-05	0 0E + 00	0.0E+00	7.5E-05	7.5E-05	0.0E+00	0.0E+00	2.0E-04	1.4E-04	0.0E+00	0.0E+00
CHRYSENE	0 0E+00	0 0E + 00	2 DE - 06	4 4E - 00	0 0E+00	0 0E+00	5 6E-07	5 2E - 07	0.0E+00	0.0E+00	1.86-08	1.8E-06	0.0E+00	0.0E+00	3.8E-08	2.0C-08
BENZD(B)FLUOR	0 0E+00	8 OE + 00	1 1E - 05	3 8E - 08	0 0E+00	0 0E+00	1 OE-08	7.0E-07	0 0E+00	0.0E+00	3.6E-06	3.6E-08	0.0E+00	0.0E+00	8.7E-08	4.1E-08
BENZO(K)FLUOR		0.06+00	9 EE - 08	302-00	0.0000+000	0.0E+00	3 8E - 07	3.6E-07	0.05+00	0.05.00	a 25_07	8 25 - 07	0.0E+00	0.0E+00	2.5E00	2.3E-06
IN(123-CD)PYPENE	00 + 30 P	0 DE +00	4 66 - 08	20E-00	0 0E+00	0 0E+00	2 4E-07	2.4E-07	0.0000	0.0E+00	8.1E-07	8.1E-07	0.0E+00	0.02+00	1.9E-08	1.4E-08
DIBENZ(AH)ANTH	0 0E + 00	0 0E + 00	7 8E - 07	6 6E - 07	A 70 AM	A 76	A.0E	0.0E		145	0.0E - 00	0.05.00	4.95	205 21	0.0E · **	0.05.00
BARUM	411-00	10E-00	0 02 + 00 0 02 + 00	0.02+00	406-03	4 /E - 00 2 8E - 00	0 0E+00	0 0E+00	140 UL €21 01	1.4E-04 2.7E-03	0.0E+00	0.0E+00 0.0E+00	4.2E-04 5.2E-03	3.0E-04 3.1E-03	0.0E+00	0.0E+00
BERYLLIUM	100-00	\$ ØE - 04	0 7E - 00	2 4E-00	1 3E-03	2 9E - 04	6 0E - 06	1.3E-08	●11 3	9.1E-05	3.9E-07	3.9E - 07	7.3E-04	1.6E-04	3 1E-08	7.7E-07
MANGANESE	106-02	7 18 - 03	0 0E + 00	0 0E + 00					3.6E - 02	1.3E02	0.0E+00	0.0E+00	2.6E-02	1.1E-02	0 0E + 00	0.0E + 00
NICKEL	0 1E-60	2 46 - 03	0 OE + 00	0 OE + 00	8 8E - 03	2 0E - 03	0 0E + 00	0 0E + 00	1.6E - 03	6.2E - 04	0.0E+00	0.0E+00	7.2E-03	1.0E - 03	0.0E+00	0.0E+00
BILVER SELENIUM THALLIUM	1 92 - 90	3 0E - 04	0 0E + 00	0 CE + CO	1 BE - 03	8 4E - 04	0 0E + 00	0.0E+00	3 9E - 03 1 8E - 04	1.2E-03 1.8E-04	0.0E+00 0.0E+00	0.0E+00 0.0E+00	2.2E-03	5.0E-04	0.0E+00	0 0E + 00
VANADIUM	102-02	0 1E - 03	0 0E + 00	0 0E + 00	1 #E02	0 OE - 03	0 0E + 00	0.0E+00	8 0E - 03	4.4E-03	0.0E+00	0.0E+00	2.2E-02	1.5E-02	0.0E+00	0.0E+00
	402-00	101 - 03	0 0E+00 3 mF-04	0 0E+00					7.0E - 03	1.8E-03	0.0E+00	0.0E+00	5.2E-03	2.7E - 03	0.0E+00	0.0E + 00
4,4-DDD 4,4-DDT	0	0 CE + 00	1 96 - 00	3 9E - 00									0.0E + 00	0 0E + 00	3.1E-06	1.8E - 08
AROCLOR 1254																
OUM	1.1.1.1.1.1.	2.1E-91	9 QE - Q6	1 16-00	3 3E - 01	1 2E - 01	3 0E - 05	1.1E-05	1.5E-01	5 4E-02	1.4E-05	1.1E-05	3.6E-01	1.4E-01	4.2E-05	2.4E-06

ł

BOXED VALUES ARE HAZAND QUOTIENTS WHICH EXCEED UNITY

Ĺ

•

FINAL

.

į

1

# 

W92194F

6-165

						. 1	<b>FINSK VALUE</b>	St SEDIME	INT EXPOSUR	E						:
PARAMETER		REACH	7			REACH				REACH	0		· ·	REACH	10	:
	HALANDO	DOTIENT	CANCER	And K ME	HAZARD Q CHIL	UOTIENT	CANCER	PISK ME	HAZAROQ		CANCE	AISK ME	HAZARDQ	UOTIENT	CANCER	ALSK ME
	HAX	AVG	MAX	DVA	MAX	AVG	MAX	AYG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG
TRICHLOROETHENE 120ICHLOROETHENE CHLOROBENZENE* NITROBENZENE* 1,2-DICHLOROBENZENE 1,3-DCB* 1,4-DCB* 1,4-DCB* 1,2,4-TRICHLOROBENZENE*																
PHENOL*	26E-01	# 26 - OE	27E-06	8 8E - 08	1 <b>BE</b> -01	7 4E - 02	2.0E05	7.9E-08	3.9E-01	1.9E-01	4.2E-05	2.1E-05	7.4E~02	3.6E-02	6.0E-06	3.9E-0
ANTIMONY																
CADMIUM*	4 96 - 02	8 1E-03	0 0E+00	0 0E+00	1.0E-02	7 6E - 03	0.0E+00	0.0E+00	3.1E-02	2.6E-02	0.0E+00	0 0E+00	A 15 05	A 45 AR	·	0.05.4
	0 0E+00	0.05+00	0 0E+00	0.02+00	0.000+000	4 UE-US 0 0E+00	0 0E+00	0.000+00	1.4E-04	0.0E+00	0.0E+00	0.0E+00	0.0E+00	2.1E-05	0.0E+00	0.0E+0
MERCURY	34E-02	0 5E - 03	0 0E + 00	0 0E+00	1.3E - 02	1.0E-02	0 0E + 00	0 0E + 00	2.4E-02	2.0E-02	0.0E + 00	0.0E+00	3.3E03	1.0E-03	0.0E+00	0.0E+0
MONOMETHYLHG" DIMETHYLHG"																
VINYL CHLORIDE BENZENE DICHLOROMETHANE ACETONE BENR	€ 8E - 00	3年-05 877-08	13E-00	7 3E - 00												
DETIF 3/4-METHYLPHENGL 2-METHYLNAPH	12-04	• / 2 - 00	/ 02 ~ 08													
PHENANTHENE	0.75-04	308-04	0 0E + 00	0 0E+00												
FLUOPANTHENE	1 mt - 05	496-05	0 0E + 00	0 0E + 00												
PYRENE	1 2E - 04	6 OE - 05	0 0E + 00	0 0E+00												
BENZD(A)ANTH	0 0E+00	0.000 + 00	2 2E - 0#	1 1E-08												
RENZORDELLOR		6.04 + 00	3 12~00 2 36~00	126-08												
BENZONFLUOR	0 0E + 00	0.00 + 00	1 3E-00	6 0E-07												
BENZO (A) PYRENE	0 0E+00	0 CE + CO	8 2E ~ 00	1 0E - 00												
IN(123-CO)PYRENE	0 OE + 00	0 CE + CO	1 4E ~ 08	14E-00												
BENZOIGHDPERM	316-04	318-04	8 OF 4 00	0.05+00												
BARIUM	0 0E - 03	14-0	0 0E + 00	0 0E+00	6.1E-03	4.1E-03	0.0E+00	0.0E+00	5.5E-03	4.2E-03	0.0E+00	0.0E+00	2.3E~03	1.2E-03	0.0E + 00	0.0E+0
BERYLLIUM	2 2E - 04	2.1E-04	9 使 - 07	8 0E-07												
COPPER																
MANGANEUR NICKEL		105-03		0.06+00	1.1E-01	3.1E-02	0 0E+00	0.00 + 00	9.45-03	7.0E-03	0.0E+00	0.0E+00	6.1E-03	3.3E-03	0.0E+00	0.02+0
BILVER								0.00.000	2.00-00	8.JC-0J	0.95 400	0.02 400	1.02~03	0.76 - 04		
BELENUM	2 CE-03	4 X - 04	6 0E + 00	0 0E+00	0.0E~04	£.7E-04	0.0E+00	0.0E+00					i i			
THALLIUM																
		1.00-00	0.06+00	0.02+00	3 35 - 03	2.0E-03	0.05+00	0.06+00	0.0E-03	4.02-03	0.0E+00	0.0E+00	4.16-03	2.46-03	0.0E+00	0.02+0
4.4-DDE	·····		v vi. 1 W	50.700	0.02 °W	1.72-00	00.700	0.002 ¥ 00	2.VE-W	1.02-03	V.U2 ¥ UU	V.VE TUU	4.65 - 64	0.02-04	0.02700	U.UE TU
4,4-000								- 1								
4,4-DDT																
CHLDRDANE AROCLOR 1854	8 3E - 04 9 0E + 00	11 JE - 04 0 DE + 00	4 3E - 08 4 3E - 07	4 王 - 09 7 3E - 07					1							
	416-01	8 65 - 02	4 1E-08	1 0E - 05	3.3E~01	1.3E-01	2 OE - 05	7 8E - 08	4.6E-01	2 8E-01	4.2E05	2.1E-05	9.2E-02	4.5E-02	8.0E-08	3.0E-0

i

____

FINAL

1

•.

BOXED VALUES ARE HAZARD QUOTIENTS WHICH EXCEED UNITY

# 

							PIBK VALUE	19: BEDIME	ent exposur	£		FISK SUMM	APLEB: SEC	XMENT EXP	OSURE
PARAMETER	604	DERING W	ETLANDS MAL		BORDE RECREATIO	PING WET	LANDS MYS PER YE	AFQ	BORD	ERING WE	ILANDS AL	1	HEARD PON	1D	
	HAZANDO	JOTIENT	CANCER	fesk Me	HAZAPO Q CHIL	UOTIENT	CANCER	i fusik Ime	HAZARDQ CHIL	UOTIENT D	CANCER RISK LIFETIME	HAZARD QI CHIL	uotient D	CANCER	<b>AISK</b> Me
l	I MAX	AVG	MAX	AVG.	MAX	AYO	MAX	AYG	MAX	AVG	MAX AVG	MAX	AVG	MAX	<b>DVA</b>
TRICHLOROETHENE 12DICHLOROETHENE CHLOROBENZENE* NITROBENZENE* 1.2-DICHLOROBENZENE 1.3-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.4-DCB* 1.	7 5E - 02 1 8E - 04 8 0E + 00 4 7E - 02	3 8E - 02 4.4E - 05 8 0E - 03 8 7E - 03	7 6E - 00 6 0E + 00 8 0E + 00 9 0E + 00	4. 1E - 00 0 DE + 00 0 DE + 00 0 DE + 00	3 0E - 01 1 0E - 03 0 0E + 00 2 0E - 01	2.0E - 01 2.4E - 04 0.0E + 00 3.6E - 02	2.7E - 08 0 0E + 00 0 0E + 00 0 0E + 00	1.4E - 05 0.0E + 00 0.0E + 00 0.0E + 00	5.1E-01 1.3E-03 0.0E+00 3.3E-01	2.7E - 01 3.1E - 04 0.0E + 00 4.7E - 02	3.0E - 05 1.9E - 0 0.0E + 00 0.0E + 0 0.0E + 00 0.0E + 0 0.0E + 00 0.0E + 0	6.8E-02 7.3E-05 0.0E+00 2.2E-02	6.6E - 02 7.3E - 05 0.0E + 00 2 21 ° 02	7.1E-06 0.0E+00 0.0E+00 0.0E+00	7.1E - 00 0.0E + 00 0.0E + 00 0.0E + 00
VINYL CHLORIDE BENZENE DICHLORDMETHANE ACETONE BEHP 3/4 - METHYLINAPH ACENAPTHYLENE PHENANTHRENE FILIORANTHRENE BENZD(A)ANTH CHRYBENE BENZD(A)ANTH CHRYBENE BENZD(A)FILIOR BENZD(A)FILIOR BENZD(A)FILIOR BENZD(A)FILIOR BENZD(A)FILIOR BENZD(A)FILIOR BENZD(A)FILIOR BENZD(A)FILIOR BENZD(A)FILIOR BENZD(A)FILIOR BENZD(A)FILIOR BENZD(A)FILIOR BENZD(C)FILIOR BENZD(C)FILIOR BENZD(C)FILIOR BENZD(C)FILIOR BENZD(C)FILIOR BERYLLIUM COPPER MANGANESE NICKEL BILVER BELENIUM THALLIUM VANADUM	2.00 - 00 2.00 - 04 2.00 - 04 2.00 - 03 4.45 - 04 1.05 - 05 1.35 - 03	1.42 - 03 3.05 - 04 8.42 - 03 6.62 - 04 2.65 - 04 7.32 - 03 3.65 - 04	8 0E + 00 3 0E - 08 9 0E + 00 9 0E + 00 9 0E + 00 6 0E + 00 6 0E + 00	90E + 00 13E - 00 90E + 00 90E + 00 90E + 00 90E + 00 80E + 00	1.3E-09 3.7E-03 1.1E-01 1.1E-02 2.4E-03 8.1E-02 9.3E-03	7.5E - 03 1.6E - 03 2.9E - 03 3.5E - 03 1.4E - 04 4.0E - 02 2.0E - 03	0 0E + 00 1.0E - 08 0 0E + 00 0 0E + 00 0 0E + 00 0 0E + 00 0 0E + 00	0.0E+00 4.5E-08 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	1.7E-02 4.9E-03 1.4E-01 1.4E-02 3.1E-03 1.1E-01 0.1E-03	9.7E - 03 2.1E - 03 3.8E - 02 4.8E - 03 1.8E - 04 6.1E - 02 2.8E - 03	0.0E+00 0.0E+00 1.4E-05 5.8E-00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	3.3E-03 6.1E-03 1.0E-03 2.0E-03 2.0E-03	3.3E - 03 6.1E - 03 1.0E - 03 5.7E - 03 2.6E - 03	0.0E+00 0.0E+00 0.0E+00 0.0E+00 0.0E+00	0.0E + 00 0.0E + 00 0.0E + 00 0.0E + 00 0.0E + 00
4.4-DDE 4.4-DDD 4.4-DDT CHLORDANE AROCLOR 1254	1.00-01	<b>•</b> 0 <b>F</b> - 0 <b>•</b>	1.16-04	1 4F - 04	8.7E-01	3.26-01	3 <b>8</b> F - 05	1.0F-nA		42F-01	4.9E-05 2.4F-M	1.15-01	1.16-01	7.1E-04	7.1E-0 <b>4</b>

ί

BOXED VALUES ARE HAZARD QUOTIENTS WHICH EXCEED UNITY

Ĺ

FINAL

N

#### IADLE 5-35 SUMMARY OF RISK RESULTS NYANZA (* TABLE UNIT 3 MIDDLES), JUNTY, MASSACHUSETTS PAGE 10

							RISK VALL	jes: Surf	ACE WATE	B			· ·			
PARAMETER	, <u> </u>	REACH I BACKORO	AND			REACH 2				Eastern W	etiands			OUTFALL	CREEK	
	HAZAHD	UOTIENT	CANCER	<b>HISK</b>	HAZARD	UOTIENT	CANCER	<b>Ri</b> sk	HAZARD C	JUOTIENT	CANCER	AISK	HAZARD	NOTIENT	CANCER	<b>FISK</b>
	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVQ	MAX	AVG	MAX	AVG	MAX	AYG	MAX	AVG
TCE*									0.0E+00	0. 00	6.9E-08	4.8E-08	0.0E+00	0.0E+00	1.4E-07	1.4E-07
1.2-DCE					1.9E-04	1.9E-04	0.0E+00	0.0E+00	1.5E-03	9.9E-04	0.0E+00	0.0E+00	2.2E-03	2.2E-03	0.0E+00	0.0E+00
1.4-DC8*													0.0E+00	0.0E+00	9.0E-08	9.0E-06
1.2-DC8*									2.0E-04	2.0E-04	0.0E+00	0.0E+00	2.7E-04	2.7E-04	0.0E+00	0.0E+00
ARSENIC*					9.8E-03	9.8E-03	1.9E-08	1.9E-06								
CADMIUM					1.4E-02	5.5E-03	0.0E+00	0.0E+00								
CHROMIUM					6.8E-06	5.2E-06	0.0E+00	0.0E+00	7.7E-05	7.7E-05	0.0E+00	0.0E+00	5.5E-06	5.5E-08	0.0E+00	0.0E+00
LEAD*	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00	0.0E+00
MERCURY*									1.2E-02	6.8E-03	0.0E+00	0.0E+00	1.6E-03	1.6E-03	0.0E+00	0.0E+00
1,1-DCE	6.4E04	5.6E-04	1.7E-08	1.5E-08												
BEHP									1				4.8E-05	4.8E-05	4.7E-09	4.7E-06
BARIUM	4 5E-04	3.1E-04	0.0E+00	0.0E+00	2.6E-03	6.6E-04	0.0E+00	0.0E+00	1.8E-04	1.7E-04	0.0E+00	0.0E+00	2.1E-04	2.1E-04	0.0E+00	0.0E+00
BERYLLIUM					1.5E-03	5.1E-04	1.1E-05	3.9E-06					]			
COPPER									ł							
MANGANESE	1.1E-03	7.9E-04	0 0E+00	0 0E+00	9.8E-02	1.5E-02	0.0E+00	0.0E+00	1.1E-03	9.8E-04	0.0E+00	0.0E+00	1.4E-03	1.4E-03	0.0E+00	0.0E+00
NICKEL	8 8E-04	2.5E-04	0 0E+00	0.0E+00	3.8E-03	1.1E-03	0.0E+00	0.0E+00								
SELENIUM									1							
SLVER	5 5E-03	1.5E-03	0.0E+00	0.0E+00	2.2E-02	3.1E-03	0.0E+00	0.0E+00								
VANADIUM					2.2E-03	2.0E-03	0.0E+00	0.0E+00								
ZINC	3.0E-05	3.0E-05	0.0E+00	0.0E+00	6.1E-04	1.2E-04	0.0E+00	0.0E+00					2.3E-04	2.3E-04	0.0E+00	0.0E+00
SUM HAZARD QUOTIENTS	8.6E-03	3.5E-03	1.7E-08	1.5E-06	1.5E-01	3.8E-02	1.3E05	5.7E-06	1.5E02	9.3E03	6.9E-06	4.8E08	5.9E-03	5.9E03	2.3E07	2.3E-07

W92194F

۰.

•

# TABLE 6-46 SUMMARY OF RISK RESULTS NYANZA OPERABLE UMT 3 NIDDLESEX COUNTY, MASSACHUSETTS PAGE 11

# W92194F -

2194		RISK VALUES: SURFACE WATER															
Ξ	PARAMETER	PACEWXY			COLD SPRING BROOK			AEACH 3			REACH 4						
		HAZARD	JUCTIENT	CANCER	<b>HEK</b>	HAZAND	QUOTIENT	CANCER	RISK	HAZARD	DUOTIENT	CANCER	AISK	HAZARD C	NOTIENT	CANCER	HISK
		MAX	AVG	MAX	AVG	MAX	<u>DVA</u>	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG	MAX	AVG 🛛
	TCE* 1,2-DCE* 1,4-DCB* 1,2-DCB* ARSENIC* CADMILM* CHROMILM* LEAD* MERCURY*	0.0E+00 3.7E-04	0.0E+00 3.7E-04	2.1E-08 0.0E+00	2.1E-08 0.0E+00					0.0E+00	0.0E+00	0.0E+00	0.0E+00	5.9E-06 0.0E+00	4.4E-06 0.0E+00	0.0E+00 0.0E+00	0.0E+00 0.0E+00
6-168	1,1-DCE BEHP BARIUM BERYLLIUM COPPER MANGANESE NICKEL SELENIUM SILVER	3.2E-04 2.8E-03	3.2E-04 9.8E-03	0.0E+00	0.0E+00	2.8E-03 1.1E-03 4.5E-03	3 2.8E-03 3 1.1E-03 3 4.5E-03	2.7E-0 0.0E+0	7 2.7E-0 0 0.0E+0 0 0.0E+0	4.8E-05 2.5E-04 8.2E-04 3.8E+00 3.9E-03	4.8E-05 1.1E-05 6.2E-04 7.5E-01 2.3E-03	4.7E-09 0.0E+00 0.0E+00 0.0E+00 0.0E+00	4.7E-09 0.0E+00 0.0E+00 0.0E+00 0.0E+00	4.8E-04 2.0E-03	2.5E-04 1.3E-03	0.0E+00 0.0E+00	0.0E+00 0.0E+00
	SUM HAZARD QUOTIENTS	3.3E-03	3.3E-03	2.1E-08	2.1E-06	6.4E-03	3 8.4E03	2.7E-07	7 2.7E-07	3.9E-05 3.8E+00	3.9E-05	0.0E+00 4.7E-09	0.0E+00 4.7E-09	2.5E-03	1.5E-03	0.0E+00	0.0E+00

(

١

t

1 Ń

#### IABLE 6–48 SUMMARY OF RISK RESULTS NYANZA O ABLE UNIT 3 MIDDLESE JUNTY, MASSACHUSETTO PAGE 12

	RISK VALUES: BURFACE WATER							ER					
PARAMETER	REACH 8				REACH 6				REACH 7				
	HAZAND	UOTIENT	CANCER RISK		HAZARD QUOTIENT		CANCER RISK		HAZARD QUOTIENT		CANCER HISK		
	MAX	AVO	MAX	AVG	MAX	AVG	MAX	AVQ	MAX	AVG	MAX	AVG	
TCE* 1,2-DCE* 1,4-DCB* 1,2-DCB* ARSENIC* CADMIUM* CHROMIUM* LEAD* MERCURY*	3.6E-03	3.4E-03	6.6E-07	6.5E-07	0.0E+00	0.0E+00	0.0E+00	0.0E+00				, ,	
1,1-DCE BEHP BARIUM BERYLLIUM COPPER	2.5E-04	2.5E-04	0.0 <b>E+00</b>	0.0E+00	3.2E-04	9.4E-05	0.0E+00	0.0E+00	4.2E-04	1.4E-04	0.0E+00	0.0E+00	
MANGANESE NICKEL SELENIUM SILVER VANADIUM ZINC	1.0E-03	6.8E-04	0.0E+00	0.0E+00	1.8E-03	9.4E-04	0.0E+00	0.0E+00	1.2E-03 2.1E-03	7.3E-04	0.0E+00 0.0E+00	0.0E+00 0.0E+00	
SUM HAZARD QUOTIENTS	4.9E-03	4.3E-03	6.8E07	6.6E-07	1.9E03	1.0E-03	0.0E+00	0.0E+00	3.7E03	2.0E-03	0.0E+00	0.0E+00	

FINAL

- o The hazard index calculated for <u>all COCs</u> ranged from 1.5 (Southville Pond, average COC concentrations) to 120 (Reservoir No. 2, maximum COC concentrations) when the subsistence fisherman is considered the receptor of concern. The hazard index calculated for <u>all COCs</u> ranged from 0.21 (Southville Pond, average COC concentrations) to 17 (Reservoir No. 2, maximum COC concentrations) when the sports fisherman is considered the receptor of concern. The HQ calculated for mercury and/or methyl mercury exceeds unity in every case that the hazard index exceeds unity.
- Hazard indices calculated for fish tissue sample results from surface water bodies downstream of the Nyanza Site exceed 10 in one or more cases presented. With the exception of Cedar Swamp Pond, hazard indices calculated for surface water bodies upstream of the Nyanza Site do not exceed 10.
- With the exception of Southville Pond, maximum and/or average mercury concentrations detected in fish tissue samples collected downstream of the Nyanza Site exceed the FDA Action Level for mercury in fish.
- o The fact that HQs and HIs exceed unity and the fact that mercury concentrations exceed FDA Action Levels in one or more cases presented for each surface water body evaluated suggests that adverse noncarcinogenic health effects are anticipated for the sports fisherman and subsistence fisherman under the conditions of the exception of the Saxonville impoundment and the Sudbury Reservoir, hazard indices calculated for Nyanza Site contaminants exceed those calculated for "other Sudbury River contaminants in all cases presented."
- o Cancer risks estimated for the fish-ingestion exposure scenarios range from not calculated (Cedar Swamp Pond and Southville Pond, no risk calculated; CSFs not available for COCs) to  $5.5 \times 10^{-3}$  (Reservoir No. 2, maximum COC concentrations, subsistence fisherman). As a point of reference, cancer risks estimated for samples collected in the Sudbury Reservoir are  $5 \times 10^4$  and  $3.6 \times 10^4$  when maximum and average COC concentrations are evaluated and the subsistence fisherman is considered the receptor of concern. Cancer risks estimated for surface water bodies

6-170

upstream of the Nyanza Site do not exceed  $1 \times 10^{-3}$  in any case presented. Cancer risks estimated for Mill Pond, Reservoir No. 2, and Fairhaven Bay exceed  $1 \times 10^{-3}$  in at least one case presented. The principal COCs cont~ibuting to the estimated excess lifetime cancer risks are arsenic, several pesticides (e.g., 4,4-DDT), and the PCBs. In all cases presented, risks associated with "other Sudbury River contaminants" exceed those estimated for Nyanza Site contaminants.

The following items summarize the results of the risk assessment of COC concentrations detected in surface water and sediment samples collected in the Study Area:

- ο In all cases presented for the sediment exposure scenarios, HQs and HIs calculated for the accidentalingestion exposure route exceed those calculated for the dermal-absorption exposure route. The hazard index calculated for the bordering wetlands (residential exposure scenario) exceeds unity when maximum concentrations are evaluated and a child is evaluated as the receptor of concern. However, hazard indices calculated on a target organ-specific basis do not exceed Adverse noncarcinogenic health effects are unity. anticipated when HQs or HIs (calculated on a target organ specific basis) exceed unity.
- With few exceptions, hazard quotients and hazard indices calculated for the recreational sediment exposure scenarios do not exceed unity. However, hazard indices calculated for the COC concentrations detected in the Eastern Wetlands sediments (recreational exposure scenarios) exceed unity when maximum contaminant concentrations are evaluated and a small child is considered the receptor of concern. (In this case, the hazard index calculated for chemicals affecting the kidney and/or central nervous system equals unity (>0.95.))

W92194F

6-171

- o The hazard index calculated for COC concentrations detected in background sediment samples does not exceed 0.3 in any of the cases presented. Hazard indices calculated for sediment exposure scenarios for the following reaches/surface water bodies exceeds 0.5 (but not unity) in one or more cases presented:
  - Outfall Creek
  - Reach No. 3

The hazard index calculated for the following reaches/surface water bodies (sediment exposure scenarios) does not exceed 0.5:

-	Reach	No.	2	-	Reach No. 9
-	Reach	No.	4	-	Reach No. 10
-	Reach	No.	5	-	Chemical Brook Culvert
-	Reach	No.	6	-	Raceway
-	Reach	No.	7		Cold Spring Brook
-	Reach	No.	8		

As stated previously, adverse noncarcinogenic health effects are not anticipated when HQs or HIs do not exceed unity.

- o Cancer risk estimates for the background sediments (recreational sediment exposure scenario) do not exceed  $2 \times 10^{-5}$  in any case evaluated. In contrast to the fishingestion exposure scenarios, cancer risks estimated for the sediment exposure scenarios do not exceed  $2 \times 10^{-4}$  in any case presented for any river reach or surface water body evaluated. As a point of reference, the  $1\times10^{-4}$  to  $1\times10^{-6}$  cancer risk range is often evaluated in the development of health-based standards/criteria and in the determination of clean-up goals at hazardous waste sites.
- Generally, the cancer risks estimated for maximum COC 0 (sediment exposure scenarios) concentrations in downstream surface water bodies range from  $1 \times 10^{-5}$  to  $1 \times 10^4$ . Risks estimated for the average-case scenario range from  $1 \times 10^{-6}$  to  $1 \times 10^{-5}$ . In most cases, risks associated with "other Sudbury River contaminants" exceed those estimated for the Nyanza Site contaminants. The principal contaminants contributing to the estimated risk are arsenic, beryllium, and the carcinogenic PAHs. Risks associated with the accidental-ingestion exposure route predominate over those estimated for the dermal-contact exposure route.

W92194F

6-172

- Cancer risks estimated for the following river reaches/surface water bodies (sediment exposure scenarios) exceed 5 x  $10^{-5}$  in one or more cases presented:
  - Reach No. 2
  - The Raceway
  - Reach No. 3

As a point of reference, the  $1\times10^4$  to  $1\times10^6$  cancer risk range is often evaluated in the development of healthbased standards/criteria and in the determination of clean-up goals at hazardous waste sites.

Hazard quotients and hazard indices estimated for surface water exposure scenarios are minimal (all values <0.1) when compared to the sediment exposure scenarios (the selenium concentration in Reach 3 appears to be an anomaly). None of the cancer risks estimated for the surface-water exposure scenarios exceeds 2 x 10⁶.

#### 6.7 <u>Uncertainty Analysis</u>

The carcinogenic and noncarcinogenic health risks presented in this risk assessment are estimated using various assumptions, and the results presented are subject to a certain amount of uncertainty. These uncertainties can be categorized into two major groups, uncertainty with toxicological data and uncertainty with exposure parameters and estimations.

The toxicological data that form the basis for the risk assessment are subject to uncertainty in the following areas:

- o The extrapolation of nonthreshold (carcinogenic) effects from the high doses administered to laboratory animals to the low doses received in the application of exposure scenarios.
- o The extrapolation of the results of laboratory animal studies to human or environmental receptors (described in the toxicity profiles).
- o The inter-species variation in toxicological endpoints used in characterizing potential health effects resulting in exposure to a chemical.
- The variations in sensitivity among individuals of any species.

W92194F

6-173

ο

 Some of the toxicity criteria used for the risk assessment are currently under review (arsenic, mercury).
 If toxicity criteria change radically, the conclusions of this risk assessment could be effected.

Toxicity criteria are not currently available Ο for parameters such as copper and 2-methyl naphthalene. Conservatively, these compounds are carried through the risk assessment although quantitative results were not presented. Because neither copper or 2 methyl napthalene were detected frequently in Study Area environmental samples, it is unlikely that the conclusions drawn by the risk assessment would be altered if toxicity criteria were available for either compound. It should be noted that the maximum concentration of copper detected in Study Area ( $C_{max}$ =454 mg/kg) sediment samples is less than the maximum concentration detected in background samples  $(C_{max} - background = 340.4 mg/kg).$ The maximum concentration of copper in Study Area surface water samples is less than the current Federal SDWA MCL.

The exposure parameters and data consolidation used in this risk assessment create some amount of uncertainty, as described below:

- o The exposure scenarios assume chronic exposure to contaminant levels which are constant. In reality, contaminant levels can change with weather conditions, water levels, and chemical changes in the waters.
- l,l-Dichloroethene and 1,4-Dichlorobenzene are Class C carcinogens. Although Class C carcinogens are evaluated in the risk assessment, there is limited evidence supporting the classification of these compounds as carcinogens. Therefore, the carcinogenic risk totals must be viewed with this information in mind.
- o Data gaps exist in the risk assessment. Toxicity criteria are not available for certain chemicals detected in the Study Area (lead, aluminum, cobalt). If these data gaps could be filled, the risk results which are presented in Sections 6.5 and 6.6 could be effected.
- Although EPA guidelines were followed in the development of the exposure scenarios, these scenarios assume certain activity frequencies. The exposure dose assumptions are based on available information and are conservative in nature.

W92194F

6-174

FINAL

- Permeability constants necessary to evaluate dermal absorption are not currently available for many chemicals. Default values have been used in this risk assessment as necessary. Absorption of VOCs is predicted by researchers to be significant; absorption of metals may approach zero.
- o The chemical database has limitations in the areas of sample location and representativeness of the actual situation. These uncertainties are present in all environmental sampling databases and cannot reasonably be avoided.

ο

W92194F



FINAL

#### 7.0 ECOLOGICAL RISK ASSESSMENT

#### 7.1 <u>General Approach</u>

This ecological risk assessment consists of four primary components: (1) the selection of chemicals of concern, (2) the exposure assessment, (3) the toxicity or hazard assessment, and (4) the risk characterization. Sections 7.3, 7.4, 7.5, and 7.6, respectively, describe these primary elements.

The selection of media-specific chemicals of concern (COCs) is based on selection criteria that will provide an appropriate level of conservatism. COCs selected for each media are based on concentration, frequency of detection, toxicity, bioconcentration potential, or environmental persistence. The contaminants evaluated in the ecological risk assessment include both Nyanza site-related contaminants as identified by EPA and other Study Area contaminants.

The exposure assessment includes estimated environmental concentrations (EECs) of COCs in each media of concern (surface water, sediment, and biota). EECs are based on measured concentrations and estimates of chemical fate and transport, which are described in Section 5.0. Estimates of chemical fate and transport for this assessment are based primarily on simple algebraic models, such as partitioning coefficients. Estimates of chemical concentrations in biota use a more complex computer-based model (Thomann's Bioaccumulation Model). Average and maximum EECs were calculated for each COC within each reach and each media.

The hazard assessment, also known as toxicity assessment, evaluates concentrations of COCs that are known to or are likely to result in adverse effects to biota. Species that might be of concern are those that are known to or are likely to inhabit or use the study area; these include plants, aquatic animals (invertebrates and vertebrates), terrestrial animals, and birds. Indicator species were selected to represent specific trophic levels in the assessment of impacts to food chains. Toxicity data for species that are known to exist or likely to be present in the study area are sparse; therefore, most toxicity data are based on standard test species that are representative of similar, related species that might exist within the Study Area.

Risk characterization is primarily the integration of exposure data with toxicity or hazard assessments; that is, estimated exposure concentrations for media-specific COCs are compared to toxic or hazardous concentrations of those COCs. For assessing impacts to food chains, bioaccumulation data are more appropriate than toxicity data. Although several methods have been developed to

W92194F

accomplish the integration of toxicity and exposure evaluations, the quotient method is the most frequently used and accepted approach. This method divides the EEC by the selected toxicity benchmark value. The resulting quotient enables the evaluation of relative toxicity between individual COCs; higher quotients are associated with greater potential toxicity. Cumulative toxicity, or the toxicity associated with chemical mixtures, is also an important component of risk characterization. This assessment addresses cumulative toxicity by summing all exposure/toxicity quotients for each media and each reach, resulting in a reachspecific total risk estimate.

Secondary components of the ecological risk assessment include the identification of potential ecological receptors, and the analysis of uncertainty. The potential ecological receptors primarily include plants and animals that are known to or are likely to inhabit or use the Study Area; among these plant and animal species, threatened and endangered species are also included. Section 7.2 identifies potential ecological receptors, or populations potentially at risk. This assessment includes uncertainty analysis as part of the discussions of exposure assessment, toxicity assessment, and risk characterization.

Figure 7-1 shows the major components of this ecological risk assessment.

## 7.2 <u>Biological Description</u>

This section describes the major plant and animal species, both aquatic and terrestrial, that might be exposed to the Study Area and Site-related contaminants. These organisms comprise the potentially exposed populations for this risk assessment. Table 7-1 lists the plant and animal species observed in the study area, and those plants and animals considered of special concern by the State of Massachusetts.

#### 7.2.1 Vegetation

Site vegetation includes common aquatic and terrestrial species of herbaceous plants, ferns, sedges, wildflowers, woody shrubs, and trees. Observed vegetation did not appear visibly stressed; however, community effects, such as species diversity, were not evaluated. Table 7-1 lists plant species recorded during the 1989-1991 field observations.





## Figure 7-1. Ecological Risk Assessment Components

FINAL

(

#### TABLE 7-1

#### SUMMARY OF STUDY AREA PLANT AND ANIMAL SPECIES (1) NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

#### STATUS COMMON NAME SCIENTIFIC NAME **OBSERVED PLANTS** NORTHERN RED OAK Quercus borealis **COMMON - CLIMAX FOREST** Pinus strobus **COMMON - CLIMAX FOREST** WHITE PINE Acer seccharum COMMON - CLIMAX FOREST SUGAR MAPLE RED MAPLE Acer rubrum COMMON - CLIMAX FOREST EASTERN COTTONWOOD Populus deltoides **COMMON - CLIMAX FOREST** Betula populifolia **GREY BIRCH** OCCASIONAL - UPLAND CANOPY AMERICAN ELM Ulmus emericana COMMON ~ NON-DOMINANT CANOPY Quercus bicolor SWAMP WHITE OAK COMMON - NON-DOMINANT CANOPY Pinus resinose PLANTED - RESERVOIR SHORES RED PINE EASTERN WHITE CEDAR Thuja occidentalis PLANTED - RESERVOIR SHORES WATER LILY Nymphaes odorata COMMON - AQUATIC ARROW ARUM Peltandra virginica COMMON - AQUATIC DUCKWEED Lemna sp. COMMON - AQUATIC ALDER BUCKTHORN Rhamnus frangula COMMON - DOMINANT SHRUB **RED OSIER DOGWOOD** Comus stolonifera COMMON - DOMINANT SHRUB COMMON - DOMINANT SHRUB SPECKLED ALDER Anus rugosa HIGHBUSH BLUEBERRY Vaccinium corymboaum COMMON - DOMINANT SHRUB Vibumum dentatum **COMMON - DOMINANT SHRUB** ARROWOOD VIBURNUM Clethra alnifolia COMMON - NON - DOMINANT SHRUB SWEET PEPPERBUSH SWAMP AZALEA **Rhododendron viscosum** COMMON - NON-DOMINANT SHRUB COMMON - NON-DOMINANT SHRUB BUTTONBUSH Cephalanthus occidentalis OCCASIONAL - WILDFLOWER ASTER Aster sp. GOLDENROD Solidago sp. OCCASIONAL - WILDFLOWER WINTERGREEN Gaultheria procumbens **OCCASIONAL - WILDFLOWER FALSE NETTLE** Boehmeria cylindrica OCCASIONAL - WILDFLOWER **CINNAMON FERN** Osmunda cinnamonea **COMMON - GROUNDCOVER ROYAL FERN** Osmunda regalis COMMON - GROUNDCOVER **Onocies sensibilis** COMMON - GROUNDCOVER SENSITIVE FERN MARSH FERN Thelypteris palustris COMMON - GROUNDCOVER POISON IVY Toxicodendron radicans COMMON - WOODY VINES COMMON GREENBRIER Smilax rotundifoila **COMMON - WOODY VINES COMMON - WOODY VINES** BRAMBLES Rubus spp.

W92194F

1

#### TABLE 7 - 1 SUMMARY OF STUDY AREA PLANTS AND ANIMAL SPECIES (1) NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS PAGE 2

		STATUS
SEDGES	Carex and	COMMON - IMPOUNDMENT SHORES
BUSHES		COMMON - IMPOUNDMENT SHORES
LOOSESTREE	l vibrum selicaria	COMMON - CLEARED UPLANDS
IEWELWEED	Impatiens capenals	
LADYSTHMB	Polygonum persicaria	LOCALLY DOMINANT - WILDFLOWER
ABROW LEAVED TEARTHUMB	Polygonum sagittatum	LOCALLY DOMINANT - WILDFLOWER
ARROW ARUM	Peltandra virginica	COMMON - AQUATIC
WATER WILLOW	Decodon verticillatus	DOMINANT - IMPOUNDMENT SHORES
COMMON COONTAIL	Ceratophyllum demursum	COMMON - IMPOUNDMENT COVES
PLANTS OF SPECIAL CONCERN		
ENGELMANN'S UMBRELLA SEDGE	Cyperus engelmannii	STATE SPECIAL CONCERN
RIVER BULRUSH	Scirpus fluviatilis	STATE SPECIAL CONCERN
LINEAR-LEAVED MILKWEED	Asciepias verticiliata	STATE THREATENED
CLIMBING FUMITORY	Adiumia fungosa	STATE THREATENED
SWAMP OATS	Sphenopholis pensylvanica	STATE THREATENED
OBSERVED ANIMALB		
RACCOON	Procyon lotor	COMMON
EASTERN CHIPMUNK	Tamias striatus	COMMON
BLUE JAY	Cyanocitta cristata	COMMON
CROW	Corvus brachyrhynchos	COMMON
HAWK	Buteo sp.	COMMON
DOCKCREST CHICKADEE	Рагиз вр.	COMMON
OSPREY	Pandion haliaetus	COMMON
GREAT BLUE HERON	Ardea herodias	COMMON
MALLARD DUCK	Anas playnynchos	COMMON
CANADA GOOSE	Branta canadensis	COMMON
PAINTED TURTLE	Chrysemys picta	COMMON
GREEN FROG	Hana clamitana	
AMERICAN EEL	Anguille rostrate	
BULLHEAD	Ictaturus spp.	COMMON
BLACK CRAPPIE	Pomoxis nigromaculatus	COMMON
#### TABLE 7 - 1 SUMMARY OF STUDY AREA PLANTS AND ANIMAL SPECIES (1) NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS PAGE 3

COMMON NAME	SCIENTIFIC NAME	STATUS
BROOK TROUT	Salvelinus fontinalis	UNCOMMON
CARP	Cyprinus carpio	UNCOMMON TO COMMON
CREEK CHUBSUCKER	Erimyzon oblongus	UNCOMMON
CHAIN PICKEREL	Esox niger	UNCOMMON TO COMMON
GOLDEN SHINER	Notemigonus crysoleucas	COMMON
LARGEMOUTH BASS	Micropterus salmoides	COMMON
REDFIN PICKEREL	Esox americanus americanus	UNCOMMON
SUNFISH	Lepomis spp.	COMMON
WHITE PERCH	Morone americana	UNCOMMON TO COMMON
YELLOW PERCH	Perca flavescens	COMMON
ANIMALS OF SPECIAL CONCERN		
GOLDEN - WINGED WARBLER	Vermivora chrysoptera	STATE ENDANGERED
LEAST BITTERN	hobrychus exilis	STATE THREATENED
AMERICAN BITTERN	Botaurus lentiginosus	STATE SPECIAL CONCERN
KING RAIL	Raikus elegans	STATE THREATENED
COMMON MOOFHEN	Gallinula chloropus	STATE SPECIAL CONCERN
BLUE - SPOTTED SALAMANDER	Ambystoma laterale	STATE SPECIAL CONCERN
EASTERN BOX TURTLE	Terrapene carolina	STATE SPECIAL CONCERN

(1) INCLUDES SPECIES OBSERVED FROM 1989 TO 1991 AND SPECIES OF SPECIAL CONCERN

W92194F

#### 7.2.2 Animals

Animal species recorded in the Study Area primarily include those easily observed because of size, habits, or recognizable sounds. Many additional species of invertebrates, fish, amphibians, reptiles, birds, and mammals probably inhabit the site either permanently or seasonally. Sufficient cover and food resources are available in the Study Area to provide habitats for a large and diverse aquatic and terrestrial community. Table 7-1 lists some of the common, easily observed species recorded from 1989 to 1991; however, this list is not intended to include all species present at the site.

#### 7.2.3 Sensitive, Threatened, or Endangered Species

Several animal and a few plant species that are known to use Study Area habitats are threatened, endangered, or of special concern to the Commonwealth of Massachusetts. Table 7-1 lists species that meet Commonwealth criteria to be considered as Plants and Animals of Special Concern.

#### 7.2.4 Indicator Species for Evaluating Food Chain Effects

This assessment considered several trophic levels for food evaluation. These levels included chain/web detritus, phytoplankton, zooplankton, aquatic invertebrates, small fish (defined as primarily zooplanktivores), large fish (defined as primarily piscivorous species), and top-level predators likely to consume fish. The terms "small fish" and "large fish" are based on terminology associated with the simplified food chain model used to estimate body burdens of biota COCs in organisms representing lower trophic levels in the aquatic food chain. Representative top-level predators that were selected as indicator species included raccoons, blue herons, and ospreys. Chemical analyses of tissues from small and large individuals of three species of fish (largemouth bass, yellow perch, and bullheads), were conducted. In addition, some caddis fly larvae were analyzed, but the data were Figure 7-2 shows simplified food webs for of limited value. sediment and water-column source contamination.



Legend

- * = Sampled
- = Modeled



#### 7.3 <u>Selection of Chemicals of Concern (COCs)</u>

This section discusses the processes used to identify the primary study area contaminants, including those that are naturally occurring, and both site related and non-site related anthropogenic contaminants. Analytical results of environmental sampling performed for this risk assessment are presented in Appendix A, and are summarized in Section 4.0.

#### 7.3.1 Potential Chemicals of Concern (PCOCs)

The Technical Directive Memorandum for Phase I Activities of the Remedial Investigation/Feasibility Study (NUS, 1990), the history of site activities, and several rounds of environmental sampling in the Study Area suggest that the risks associated with several potential COCs might require assessment. Tables 7-2, 7-3, and 7-4 list chemicals that can be considered potential COCs in each media (surface water, sediment, and biota). Figure 7-3 shows the process used for the selection of the final media-specific COCs.

#### 7.3.2 Selection of Surface Water COCs

The criteria used for the selection of surface water COCs included frequency of detection, bioconcentration potential, and toxicity. The selection process was based on a tiered approach. The first tier consisted of evaluating the frequency of detection, and the second tier consisted of evaluating both bioconcentration potential and toxicity.

With this approach, chemicals are retained for consideration as surface water COCs if they were detected in more than five percent of the surface water samples, and <u>either</u> the maximum measured surface water concentration exceeded benchmark toxicity values, <u>or</u> the potential COCs are associated with bioconcentration factors (BCFs) greater than 3,000. Chemicals associated with BCFs of less than 3,000, or chemicals with log octanol/water partitioning coefficients (Kow) of less than 3.5, have low bioconcentration potential (EPA, 1991).

The first criterion for selecting surface water COCs was frequency of detection. Chemicals that were measured above detection limits in site surface waters in five percent or more of either filtered or unfiltered surface water samples were considered to be potential surface water COCs. Because of the relatively small number of filtered surface water samples collected, both filtered and unfiltered samples were evaluated by the frequency of detection criterion at this screening stage of

- W92194F

#### TABLE 7–2 SURFACE WATER COC BASED ON FREQUENCY OF DETECTION, TOXICITY, AND BIOCONCENTRATION POTENTIAL NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

E	BENCHMARK				(MAX SW					
	TOXICITY			MAX	CONC)/		GEOMETRI	С		
POTENTIAL	VALUE (TV)	тох		SW+ CONC	(TV/10)	TOXICITY	MEAN	BCF	BICON	FINAL
SURFACE WATER COCS	(ug/L)	REF*	TV/10	(ug/L)	RATIO	SW COC	BCF	REF*	SW COC	SW COC
ALUMINUM	87	1	8.7	3220.0	370.1	YES	82	3	NO	YES
ARSENIC	190	1	19	3.0	0.2	NO	8	4	NO	NO
CHROMIUM	11	1	1.1	7.0	6.4	YES	23	3	NO	YES
COPPER	3.6	1	0.36	27.7	76.9	YES	856	3	NO	YES
LEAD	0.54	1	0.054	21.1	<b>390.7</b>	YES	403	3	NO	YE6
MERCURY (INORGANIC)	0.23	1	0.023	3.8	165.2	YES	1852	3	NO	YEB
NICKEL	49	1	4.9	17.7	3.6	YES	926	4	NO	YES
SILVER	0.12	1	0.012	68.9	5741.7	YES	36	3	NO	YES
ZINC	33	1	3.3	47.9	14.5	YES	378	3	NO	YES
1,1-DICHLOROETHENE	22400	2	2240	3.0	< 0.1	NO	25	5	NO	NO
1,2-DICHLOROETHENE	22400	2	2240	12.0	< 0.1	NO	25	5	NO	NO
BIS-(2-ETHYLHEXYL) PHTHALATE	03	2	0.03	58.0	1933.3	YES	2680	3	NO	L YES
TRICHLOROETHENE	2190	2	219	13.0	0.1	NO	17	5	NO	NO

* REFERENCES

+ SURFACE WATER

1: TV = CHRONIC AMBIENT WATER QUALITY CRITERIA (EPA, 1980–88). ARSENIC TV BASED ON As III, CHROMIUM ON Cr VI. BASED ON WATER HARDNESS = 25 mg CaCO3/L FOR HARDNESS-DEPENDENT METALS

2: TV = LOEC (LOWEST OBSERVED EFFECT CONCENTRATION) / 10

LOEC/10 = ESTIMATED NOEC (NO OBSERVED EFFECT CONCENTRATION)

3: AMBIENT WATER QUALITY CRITERIA DOCUMENT (EPA, 1980-88)

4: ACQUIRE DATABASE, (EPA, 1990)

5: SYRACUSE RESEARCH CORPORATION, CHEMFATE DATABASE (SRC, 1990)

#### TABLE 7-3

#### SEDIMENT COC BASED ON THE FREQUENCY OF DETECTION, TOXICITY, AND BIOCONCENTRATION POTENTIAL NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

	ER-L								
	OR		MAX	MAX SED		GEOMETRIC			
POTENTIAL	SURROGAT	TOX	SED CONC	CONC/ER-L	TOXICITY	MEAN	BCF	BIOCON	FINAL
SEDIMENT COCS	(ug/kg)	REF*	(ug/kg)	RATIO	SED COC	BCF	REF*	SED+ COC	SED COC
ALUMINUM	174000	1	22900000	131.6	YES	82	6	NO	YES
ARSENIC	33000	2	64600	2.0	YES	8	7	NO	YES
BERYLLIUM	10600	t	4000	0.4	NO	19	6	NO	NO
CADMIUM	5000	2	19900	4.0	YES	404	6	NO	YES
CHROMIUM	80000	2	968000	12.4	YES	23	6	NO	YES
COPPER	70000	2	454000	6.5	YES	856	6	NO	YES
LEAD	35000	2	876000	25.0	YES	403	6	NO	YE8
MERCURY (INORGANIC)	150	2	152000	1013.3	YES	1852	6	NO	YES
NICKEL	30000	2	88800	3.0	YES	926	7	NO	YES
SELENIUM	6000	1	7200	12	YES	120	6	NO	YES
ZINC	120000	2	765000	6.4	YES	378	6	NO	YE8
1,2,4-TRICHLOROBENZENE	10000	1	3100	0.3	NO	1460	8	NO	NO
1,2-DICHLOROBENZENE	10000	1	7200	0.7	NO	278	7	NO	NO
1,2-DICHLOROETHENE	10000	1	130	<0.1	NO	25	8	NO	NO
1,4-DICHLOROBENZENE	10000	1	1600	0.2	NO	71	8	NO	NO
2-METHYLNAPHTHALENE	110000	4	820	< 0.1	NO	300	9	NO	NO
4,4'-DDD	2	3	700	350.0	YES	79439	8	YES	YEB
4,4'-DDE	2	3	60	30.0	YES	2166	8	NO	YES
4,4' – DDT	1	3	1400	1400.0	YES	93332	8	YES	YES
ACENAPHTHENE	150	3	1235	8.2	YES	389	7	NO	YES
ACENAPHTHYLENE	60	3	880	14.7	YES	575	8	NO	YES
ACETONE	1000000	5	2600	< 0.1	NO	NA	10	NO	NO
ANTHRACENE	85	2	2900	34.1	YES	1458	8	NO	YES
BENZOIC ACID	18000000	5	1 300	< 0.1	NO	NA	10	NO	NO
BENZO(A)ANTHRACENE	230	2	4500	19.6	YES	2099	7	NO	YES
BENZO(A)PYRENE	400	2	4400	11.0	YES	920	8	NO	YE8
BENZO(B)FLUOPANTHENE	60	3	4400	73.3	YES	23990	8	YES	YES
BENZOBUTYLPHTHALATÉ	44000	1	630	< 0.1	NO	663	6	NO	NO
BENZO(G,H,1)PERYLENE	60	3	1600	26.7	YES	147752	11	YES	YES
BENZO(K)FLUORANTHENE	60	3	3900	65.0	YES	33887	8	YES	YES

W92194F

#### TABLE 7-3 SEDIMENT CHEMICALS OF CONCERN

W92194F

NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

PAGE 2

	ER-LOR		MAX	MAX SED		GEOMETRIC			
POTENTIAL	SURROGAT	τοχ	SED CONC	CONC/ER-L	TOXICITY	MEAN	BCF	BIOCON	FINAL
SEDIMENT COCS	(ug/kg)	REF*	(ug/kg)	RATIO	SW COC	BCF	REF*	SED COC	SED COC
BIS(2-ETHYLHEXYL)PHTHALATE	600	1	2000	3.3	YES	378	6	NO	YES
CHLOROBENZENE	10000	1	1600	0.2	NO	1768	7	NO	NO
CHRYSENE	60	3	7700	128.3	YES	5755	7	YES	YES
DIBENZOFURAN	60	3	1060	17.7	YES	716	11	NO	YES
DIBENZO(A,H)ANTHRACENE	60	2	310	5.2	YES	50122	7	YES	YES
DI-N-BUTYLPHTHALATE	88000	1	5200	0.1	NO	10472	11	YES	YES
FLUORANTHENE	600	2	12000	20.0	YES	1738	7	NO	YES
FLUORENE	600	2	1500	2.5	YES	501	7	NO	YES
INDENO(1,2,3-CD)PYRENE	60	3	1800	30.0	YES	32362	8	YES	YES
METHYLENE CHLORIDE	19300000	5	55	<0 1	NO	5	8	NO	NO
METHYLETHYLKETONE	56000000	5	450	<0.1	NO	NA	10	NO	NO
METHYLMERCURY	150	2	312	2.1	YES	26919	6	YES	YES
NAPHTHALENE	340	2	2300	6.8	YES	14	7	NO	YE8
NITROBENZENE	2700000	4	650	< 0.1	NO	19	8	NO	NO
N-NITROSODIPHENYLAMINE	585000	4	290	< 0.1	NO	217	6	NO	NO
PHENANTHRENE	225	2	10650	47.3	YES	2630	7	NO	YES
PYRENE	350	2	11000	31.4	YES	457	8	NO	YEA
TRICHLOROETHENE	43800000	4	170	<01	NO	17	8	NO	NO
(3-AND/OR 4-)METHYLPHENOL	350000	5	260	< 0.1	NO	NA	10	NO	NO

REFERENCES

+ SEDIMENT

1: SURROGATE VALUE BASED ON AQUEOUS TOXICITY [CAWQC OR (LOEC/10) / 0.0005] (EPA 1980-88)

0.0005 - MEDIAN CAWQC/ER-L FOR ALL CHEMICALS WITH BOTH CAWQC AND ER-L

2: LONG AND MORGAN, 1989. ARSENIC VALUE BASED ON As III, CHROMIUM ON Cr VI.

3: LOWEST PAH ER-L (LONG AND MORGAN, 1989)

4: SURROGATE VALUE BASED ON [(LOWEST ACUTE LC50/ACR) / 0.0005] (EPA 1980-88) ACR-20 FOR NONPERSISTENT CHEMICALS, 100 FOR PERSISTENT CHEMICALS (EPA 1985))

5: [(LOWEST ACUTE LC50/ACR) / 0.005)], (LC50, VERSCHUEREN, 1983)

6: AMBIENT WATER QUALITY DOCUMENT (EPA, 1980-86)

7: ACQUIRE DATABASE, 1990

8: SYRACUSE RESEARCH CORPORATION, CHEMFATE DATABASE (SRC, 1990)

9: VERSCHUEREN, 1983

10: NO AVAILABLE BCF, LOG KOW <3 5

11: CALCULATED FROM EQUATION (VEITH AND KOSIAN, 1983)

7-12

# TABLE 7-4

**BIOTA COC** 

BASED ON FREQUENCY OF DETECTION AND BIOCONCENTRATION POTENTIAL

	GEOMETRIC		FINAL
POTENTIAL	MEAN	BCF	BIOTA
BIOTA COC	BCF	REF*	00
ALUMINUM	82	1	NO
ARSENIC	8	2	NO
CADMIUM	404	1	NO
CHROMIUM	23	1	NO
COPPER	856	1	NO
MERCURY (INORGANIC)	1852	1	NO
SELENIUM	120	t	NO
ZINC	378	1	NO
4,4'-DDD	79439	3	YES
4,4'-DDE	2166	3	NO
4,4'DDT	93332	3	YES.
ACETONE	NA	4	NO
AROCLOR 1254	18803	1	
AROCLOR 1260	270000	1	
BIS(2-ETHYLHEXYL)PHTHALATE	378	1	NO
CHLORDANE (TOTAL)	7350	3	YES
DIELDRIN	4467	3	YES
DI-N-BUTYLPHTHALATE	748	1	NO
METHYLMERCURY	26919	1	YES
PHENOL	2	2	NO
4-METHYLPHENOL	NA	4	NO

# * REFERENCES

1: AMBIENT WATER QUALITY DOCUMENT (EPA, 1980-88). CHROMIUM BCF BASED ON Cr VI.

2: ACQUIRE DATABASE (EPA, 1990). ARSENIC BCF BASED ON As III.

3: SYRACUSE RESEARCH CORPORATION, CHEMFATE DATABASE (SRC. 1990)

4: NO AVAILABLE BCF, LOG KOW < 3.5



7-14

ER-L - Effects Range-Low (10th percentile)

CAWQC - Chronic Ambient Water Quality Criteria or Surrogate

BCF - Highest Bioconcentration Factor for Fish or Invertebrates

Figure 7-3. Process for Selection of Media-Specific Chemicals of Concern (COCs)

FINAL

the assessment. Rejection of data associated with unfiltered samples was considered inappropriate at this stage, even though unfiltered samples probably overestimate actual exposure concentrations of surface water chemicals. Section 7.4.4 discusses the differences in bioavailability between filtered and unfiltered samples.

The second criterion used for selection of surface water COCs was toxicity. The selected benchmark toxicity value (TV) for screening purposes was one-tenth of the chronic ambient water quality criteria (CAWQC/10). This value was selected as a conservative and appropriate screening criterion. Surrogate values for CAWQC were used for chemicals for which CAWQC are unavailable. Surrogate values for CAWQC include lowest observed effects concentration (LOEC) divided by 10.

LOEC/10 is an estimate of the no-observed-effects concentration (NOEC). The calculation of CAWQC that are hardness dependent were based on a surface water hardness value of 25 mg  $CaCO_3/L$ .

The final criterion for selection of surface water COCs was bioconcentration potential. In most cases, and specifically if sufficient data were available, the geometric mean freshwater BCF was used to represent the chemical-specific BCF for selecting surface water COCs. If insufficient bioconcentration data using freshwater species were available, BCFs based on saltwater species were included. BCFs were unavailable for a few chemicals, and were estimated by using appropriate Kow/BCF relationships. Surface water chemicals associated with appropriately derived BCFs greater than 3,000 were considered surface water COCs.

Surface water samples were analyzed for inorganic and organic chemicals, which are presented in Section 4.0. The five percent frequency of detection criterion eliminated antimony, beryllium, cadmium, cobalt, selenium, thallium, vanadium, chloromethane, methylethyl ketone, 1,4-dichlorobenzene, 1,2-dichlorobenzene, and gamma-BHC (lindane) from consideration as COCs in surface waters.

After being subjected to the frequency of detection screen, several of the 20 chemicals that remained under consideration as surface water COCs are generally considered to be nontoxic at concentrations present in the Study Area. These chemicals include barium, calcium, iron, magnesium, manganese, potassium, and sodium). Although very high concentrations of some of these chemicals can result in adverse effects to aquatic life (e.g., by upsetting acid-base balance), they were eliminated from further consideration because of their relatively nontoxic nature compared to the large number of potentially toxic chemicals requiring further evaluation.

W92194F

The 13 chemicals retained for consideration as surface water COCs were subjected to toxicity and bioconcentration screens. To eliminate any of these chemicals from consideration as a surface water COC, the maximum concentration of the chemical measured in surface waters had to be less than the selected benchmark toxicity value (CAWQC/10), and the bioconcentration factor associated with the chemical had to be less than 3,000. Table 7-2 lists the results of the toxicity and bioconcentration screening process for surface water contaminants, and identifies the final nine surface water COCs.

#### 7.3.3 Selection of Sediment COCs

This assessment based the selection of sediment COCs on criteria generally similar to those used for the selection of surface water COCs; these included frequency of detection, toxicity, and bioconcentration potential. Again, a tiered approach was used for the selection of sediment COCs. The first tier consisted of evaluating frequency of detection, and the second tier consisted of evaluating toxicity and bioconcentration potential. Environmental persistence, a common screening criteria, was evaluated separately within the exposure assessment. Persistence is most appropriately with exposure duration, and it was associated considered inappropriate to eliminate chemicals from consideration as sediment COCs based on environmental persistence.

The first tier in screening potential sediment COCs consisted of evaluating sediment source chemicals that were measured above detection limits in five percent or more of total sediment samples analyzed. Chemicals for which the frequency of detection exceeded five percent were considered potential sediment COCs.

The second tier in screening potential sediment COCs consisted of evaluating toxicity and bioconcentration potential. The toxicity evaluation was based on the biological effects data of Long and Morgan (1989), who derived toxicant concentrations in sediments that are associated with observed adverse biological effects.

For toxicants present in sediments, Long and Morgan sorted in ascending order the concentrations of specific toxicants that are associated with observed adverse biological effects, and derived the 10th and 50th percentile concentrations, described as the (ER-L) and Effects Range-Median Effects Range-Low (ER-M), respectively, for each chemical evaluated. ER-L values represent concentrations of a chemical in sediment (dry weight) that is equivalent to the lower 10th percentile of the screened available Similarly, ER-M values are equivalent to the median (50th data. percentile) of the screened available data. Although the database apparent effects is based predominantly on marine life, of freshwater organisms are not expected to exhibit markedly different

W92194F

responses; therefore, the entire database was considered useable for screening purposes.

The ER-L value was selected as the criterion value for selecting sediment COCs. ER-L values were unavailable for several chemicals measured in sediment samples. For such chemicals, surrogate values were used as screening criteria for toxicity. Surrogate values included (1) CAWQC (or surrogate) divided by 0.0005, which is the median value of the ratio of CAWQC/ER-L computed for all chemicals for which ER-L values and CAWQC are available, and (2) ER-L values for closely related chemicals that are expected to exhibit similar toxicities. Chemicals measured in sediments at concentrations above ER-L values (or surrogate values) were considered to be sediment COCs. Table 7-3 lists the ER-L values and the surrogate ER-L values for chemicals for which ER-L values have not been derived.

This ecological risk assessment evaluated bioconcentration potential, the final criterion for selecting sediment COCs, by determining appropriate BCFs for each potential sediment COC. The selection of appropriate BCFs for sediment COCs was based on the same criteria used for determining appropriate BCFs for potential surface water COCs. Chemicals measured in sediment samples above detection limits with BCFs greater than 3,000 (or log Kow >3.5) were considered to be sediment COCs.

Sediment samples were analyzed for inorganic and organic chemicals, which are presented in Section 4.0. The first tier of the screening process for the selection of sediment COCs, which involved the evaluation of the frequency of detection, resulted in the elimination of 11 chemicals (antimony, silver, benzene, tetrachloroethene, toluene, 1,3-dichlorobenzene, diethyl phthalate, di-n-octylphthalate, gamma-chlordane, PCB-1254 or Aroclor 1254, and dimethylmercury).

Nine of the 58 chemicals that remained under consideration as sediment COCs, after being subjected to the frequency of detection screen, are generally considered to be associated with low toxicity. In general, toxicity data are unavailable for these nine chemicals which included barium, calcium, cobalt, iron, magnesium, manganese, potassium, sodium, and vanadium.

____

FINAL

These chemicals were eliminated from further consideration because of their relatively low toxicity in comparison to the large number of potentially toxic chemicals requiring further evaluation.

The remaining 11 inorganic and 38 organic chemicals were subjected to toxicity and bioconcentration screens. Table 7-3 lists the results of the toxicity and bioconcentration screens, as well as the 33 final sediment COCs.

#### 7.3.4 Selection of Biota COCs

This assessment based the selection of COCs for biota samples on frequency of detection in fish tissue samples and on bioconcentration potential. Chemicals detected in fish tissue at a frequency of five percent or more were considered potential biota COCs. Eight inorganic and 13 organic chemicals exceeded the 5 percent frequency of detection criterion; therefore, 21 chemicals were considered to be potential biota COCs. Potential biota COCs with BCFs of 3,000 or greater became final biota COCs. Of the 21 potential biota COCs evaluated (Table 7-4), 14 were eliminated due to low bioconcentration potential. The seven organic chemicals remaining were selected as biota COCs; these COCs are listed in Table 7-4.

#### 7.4 Exposure Assessment

The major objective of the exposure assessment is to estimate, as accurately as possible, the media-specific chemical concentrations to which site biota might be exposed. Estimated environmental concentrations (EECs) are based most appropriately on measured site-specific data, and should be based on the bioavailable, or potentially bioavailable, portion of the total media-specific chemical concentration. The bioavailable portion of a chemical is the portion that is in a form that is known or likely to cause adverse effects to biota under likely exposure scenarios. The important components of exposure scenarios include not only EECs, but also frequency and duration of exposure. For aquatic life, exposures are constant or nearly constant for water-column contaminants, while they can vary for sediment-source contaminants, depending on species. For semiaguatic and terrestrial biota and birds, exposure durations and frequencies are likely to vary with season, species, and life stage.

If site-specific data on a chemical are insufficient for predicting chemical fate and transport, appropriate models are often used. The models used in this exposure assessment described the partitioning behavior of contaminants between various media. The relationships between media-specific contaminant concentrations are expressed by partitioning coefficients. Because sufficient site-

W92194F

specific data was available, the use of complex computer modeling for chemical fate and transport was not appropriate with the exception of its use for food chain modeling.

#### 7.4.1 Fate and Transport of COCs

Section 5.0 describes chemical fate and transport mechanisms in detail. Most Site related contaminants probably are transported primarily by sediment transport. Atmospheric transport and deposition is likely for some offsite contaminants, such as some polycyclic aromatic hydrocarbons (PAHs). The uptake of certain chemicals that are likely to biomagnify within food chain organisms is a probable fate process; the transport of such chemicals through food chains via ingestion of contaminated prey by upper level predators is also likely.

Environmental persistence of media-specific COCs is a critical component of exposure duration. Chemicals that are toxic and environmentally persistent probably provide a long-term source of hazard to exposed biota. This assessment determined environmental persistence by evaluating the maximum reported soil or surface water aerobic half-life of a chemical. The categories of persistence are high (half-life > 1 year), moderate (half-life 6 months to 1 year), and low (half-life less than 6 months). These categories are based on generally accepted definitions of high, moderate, and low biodegradation rates (Howard et al., 1991).

Table 7-5 lists environmental half lives for the media-specific organic and organometalic COCs. Of the 39 COCs derived for all media, 12 (all) inorganic and 17 organic COCs are associated with high persistence. Of the remaining 10 COCs, 8 are considered to have low persistence, and the final 2 COCs are designated moderate in persistence. The fact that the majority of COCs are highly persistent is not unexpected. COCs detected in sediments are most often environmentally persistent, and 33 of the 39 total COCs are sediment COCs.

#### 7.4.2 Exposure Pathways

The four major elements of an exposure pathway are (1) contaminant sources and release mechanisms, (2) retention or transport media, (3) points of potential contact, and (4) exposure routes (EPA, 1989a). Figure 7-4 shows these primary exposure pathway elements.

FINAL

	· · · · · · · · · · · ·			
CHEMICAL	MEDIA COC	(DAYS)	REF*	PERSISTENCE
ALUMINUM	SW/SED	NA		HIGH
ARSENIC	SED	NA		HIGH
BERYLLIUM	SW	NA		HIGH
CADMIUM	SED	NA		HIGH
CHROMIUM	SW/SED	NA		HIGH
COPPER	SW/SED	NA		HIGH
LEAD	SW/SED	NA		HIGH
MERCURY (INORGANIC)	SW/SED	NA		HIGH
NICKEL	SW/SED	NA		HIGH
SELENIUM	SED	NA		HIGH
SILVER	SW	NA		HIGH
ZINC	SW/SED	NA		HIGH
4,4'-DDD	SED/BIOTA	5694	1	HIGH
4,4'DDE	SED	6	1	LOW
4,4'-DDT	SED/BIOTA	350	1	MOD
ACENAPHTHENE	SED	102	1	LOW
ACENAPHTHYLENE	SED	60	1	LOW
ANTHRACENE	SED	460	1	HIGH
AROCLOR 1254	BIOTA	5475	2	HIGH
AROCLOR 1260	BIOTA	5475	2	HIGH
BENZO(A)ANTHRACENE	SED	679	1	HIGH
BENZO(A)PYRENE	SED	529	1	HIGH
BENZO(B)FLUORANTHENE	SED	610	1	HIGH
BENZO(G,H,I)PERYLENE	SED	650	1	HIGH
BENZO(K)FLUORANTHENE	SED	2139	1	HIGH
BIS(2-ETHYLHEXYL)PHTHALATE	SW/SED	23	1	LOW
CHLORDANE	BIOTA	1387	1	HIGH
CHRYSENE	SED	<b>99</b> 3	1	HIGH
DIBENZOFURAN	SED	28	1	LOW
DIBENZO(A, H)ANTHRACENE	SED	942	1	HIGH
DIELDRIN	BIOTA	1095	1	HIGH
DI-N-BUTYLPHTHALATE	SED	23	1	LOW
FLUORANTHENE	SED	440	1	HIGH
FLUORENE	SED	60	1	LOW
INDENO(1,2,3-CD)PYRENE	SED	730	1	HIGH
METHYLMERCURY	SED/BIOTA	NA	3	HIGH
NAPHTHALENE	SED	48	1	LOW
PHENANTHRENE	SED	200	1	MOD
PYRENE	SED	1898	1	HIGH

#### TABLE 7-5 ENVIRONMENTAL PERSISTENCE FOR FINAL COCS (ALL MEDIA) NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

- -

NA: NOT APPLICABLE

* REFERENCES

1. BASED ON MAXIMUM SOIL OR SURFACE WATER AEROBIC HALF-LIFE (HOWARD ET AL., 1991)

2. BASED ON MAXIMUM PCB PERSISTENCE IN SEDIMENTS (USFWS. 1986b)

3: BASED ON INORGANIC MERCURY

PERSISTENCE: HIGH:>1 YR, MOD: 6 MOS - 1 YR, LOW: < 6 MOS (HOWARD ET AL., 1991)

W92194F

#### 7.4.2.1 Exposure Sources and Release Mechanisms

Exposure sources include the source of the contaminant, such as contaminated sediments. In some cases the exposure source can be the exposure point, and no release mechanism is identified. In other cases, however, release mechanisms can be identified (spills, leaks, leaching, volatilization, etc.).

Exposure sources can be point or nonpoint sources, or a combination of both. Point sources for contaminants include the original site location and adjoining wetlands and creeks.

In addition, present industrial facilities contributing to the contamination of the Sudbury River and its tributaries can be point sources of pollutants. Nonpoint sources can include surface runoff, atmospheric deposition, transported contaminants from upstream waters, and transported sediments.

#### 7.4.2.2 <u>Exposure Points</u>

Exposure points are locations of potential contact between an organism and the chemical agent of concern. Exposure points that require evaluation include locations with the highest levels of potential contamination.

#### 7.4.2.3 <u>Exposure Routes</u>

Exposure routes are ways by which a COC comes in contact with an organism; they can include ingestion, inhalation, and dermal contact. Table 7-6 summarizes the plant and animal exposure pathways considered in the Study Area.

#### 7.4.3 Future Exposures

Present and future land use scenarios often produce similar EECs of COCs. In other cases, however, exposures estimated for future land use scenarios could differ from exposures based on present land uses. Even when land uses remain essentially unchanged, the chemical structure or concentration of a contaminant can change over time due to physical, chemical, or biological processes. For most COCs and under most conditions, the present exposure posed to site organisms will be the most hazardous, with hazards decreasing over time. For other COCs and under other conditions, however, exposures can increase over time.



Figure 7-4. Exposure Pathways

7-22

# 

Potentially Exposed Organisms	Exposure Route, Medium, and Exposure Point	Pathway Selected for Evaluation	Reason for Selection or Exclusion
Water-column biota (e.g., phytoplankton,	Direct contact with aqueous chemicals	Yes	<ul> <li>Toxicity to aquatic life is evaluated based on constant exposure that considers both direct</li> </ul>
zooplankton, fish)	<ul> <li>Ingestion of aqueous chemicals</li> </ul>	Yes	contact and aqueous ingestion
	Ingestion of contaminated biota	Yes	<ul> <li>Potential for significant exposure via this pathway is high for chemicals that tend to bioaccumulate</li> </ul>
	Direct contact with contaminated sediments	No	<ul> <li>Potential for significant exposure via this pathway is low compared to aqueous ingestion</li> </ul>
<u>Benthic biota</u> (e.g., Insect Iarvae, craylish, benthic fish)	<ul> <li>Ingestion of contaminated sediments</li> </ul>	Yes/No	<ul> <li>Potential for significant exposure via this pathway is high for chemicals that tend to bioaccumulate</li> </ul>
	<ul> <li>Ingestion of contaminated biota</li> </ul>	Yes	<ul> <li>Potential for significant exposure via this pathway is high for chemicals that tend to bioaccumulate</li> </ul>
	Ingestion of aqueous chemicals     in interstitial water	Yes	Actual exposure to benthic organisms is best estimated by interstitial water concentrations
- <u></u>	<ul> <li>Direct contact with aqueous chemicals</li> </ul>	Yes	<ul> <li>Toxicity to aquatic life is evaluated based on constant exposure that considers both direct contact and aqueous ingestion</li> </ul>
Semi-aquatic biota	<ul> <li>Ingestion of contaminated biota</li> </ul>	Yes	<ul> <li>Potential for significant exposure via this pathway is high for chemicals that tend to bioaccumulate</li> </ul>
(e.g., amphibians, turtles, snakes, ducks, beavers)	<ul> <li>Direct contact with contaminated sediments</li> </ul>	Νο	<ul> <li>Potential for significant exposure via this pathway is low compared to other pathways</li> </ul>
	<ul> <li>Ingestion of contaminated sediments</li> </ul>	Yes/No	<ul> <li>Potential for significant exposure via this pathway is high for chemicals that tend to bioaccumulate</li> </ul>
	<ul> <li>Ingestion of aqueous chemicals</li> </ul>	Yes	<ul> <li>Toxicity to aquatic life is evaluated based on constant exposure that considers both direct contact and aqueous ingestion</li> </ul>
<u>Terrestrial/avian biota</u>	Ingestion of aqueous chemicals	No	<ul> <li>Potential for significant exposure via this pathway is low for chemicals that tend to bioaccumulate</li> </ul>
(e.g., invertebrates, reptiles, birds, mammals)	Direct contact with aqueous chemicals     Ingestion of aqueous chemicals     Ingestion of contaminated biota     Direct contact with contaminated sediments     Ingestion of contaminated sediments     Ingestion of contaminated biota     Ingestion of aqueous chemicals in interstitial water     Oirect contact with aqueous chemicals     Ingestion of contaminated biota     Ingestion of aqueous chemicals in interstitial water     Oirect contact with aqueous chemicals     Ingestion of contaminated biota     Ingestion of aqueous chemicals     Ingestion of contaminated aquatic biota     Uptake of aqueous chemicals     Uptake of chemicals from aodis	Yes/No	<ul> <li>Potential for significant exposure via this pathway is high for chemicals that tend to bioaccumulate</li> </ul>
Floating aquatic plants	Uptake of aqueous chemicals	Yes	Potential for significant exposure via this pathway is     high for water-column source contaminants
Rooted aquatic plants	Uptake of chemicals from     sediments or interstitial water	Yes/No	<ul> <li>Potential for significant exposure via this pathway is evaluated for chemicals that tend to bioaccumulate</li> </ul>
<u>Terrestriai plants</u>	Uptake of chemicals from     solis	No	Potential for significant exposure via this pathway     Is low for aqueous/sediment contaminants

(

FINAL

1

#### FINAL

Examples of situations that can result in increased bioavailable exposures over time include (1) the presence of COCs for which the degradation products are potentially more toxic than the parent chemical, and (2) changes in exposure locations (e.g., increasing sediment deposition and contamination downstream of the source over time). Usually, future changes in exposure scenarios are best predicted using computer-based fate and transport models, but because this particular risk assessment is focused on present land use only, future land uses and future exposure scenarios are not considered.

#### 7.4.4 Surface Water Exposure Assessment

The surface water exposure assessment is based on both the average and maximum concentration of COCs measured in surface waters. The use of the maximum measured value is a conservative approach that estimates maximum likely exposure. In contrast, the use of average concentrations of surface water COCs allows a more realistic estimate of exposure for potentially exposed organisms. The value used to represent the average surface water concentration is based on the average reported concentration of each surface water COC.

In the average reported concentration, COCs measured below detection limits (BDLs), have been considered by using half of the detection limit to represent their actual concentration.

Both filtered and unfiltered samples were used for selecting surface water COCs. The use of unfiltered samples is an appropriate and conservative approach for this selection. However, such an approach is inappropriate for estimating bioavailable, or potentially hazardous, exposure concentrations of surface water Therefore, the average and maximum concentrations of COCs. inorganic COCs presented in the surface water exposure are based on filtered samples only. Filtered samples express concentrations of chemicals dissolved in water, and samples filtered at ambient pH probably best represent potentially hazardous aqueous exposure concentrations. Chemicals bound to particulate matter, which are measured in unfiltered samples, are not generally considered to be bioavailable (US EPA, 1985h). Such chemicals should eventually settle out of the water column and become sediments. Section 7.4.5 discusses the assessment of sediment COC exposures. Table 7-7 lists the average reported and maximum concentrations of the 10 surface water COCs in filtered samples for each location sampled.

	MIDDLSEX COUNTY, MASSACHUSETTS												
LOCATION: CHEMICAL (ug/L)	BACKGF MEAN		R2 MEAN	R2 MAX	R3 MEAN	R3 MAX	R4 MEAN	R4 MAX	R5 MEAN	R5 MAX			
ALUMINUM	ND	-	29.28	67.30	80.80	80.80	107.00	107.00	ND	-			
BERYLLIUM	ND	~	ND	-	ND	-	ND	-	ND	_			
CHROMIUM	ND	-	ND	-	ND	-	ND	-	ND	_			
COPPER	ND	-	ND	-	ND	-	ND	-	ND	_			
LEAD	ND	-	ND	-	ND		1.50	1.50	NÐ	_			
MERCURY	ND		ND	-	ND	-	ND	-	ND	-			
NICKEL	3.80	7.00	ND	-	ND	-	ND	_	ND	-			
SILVER	ND	~	ND	-	ND	-	ND	-	ND	-			
ZINC	12.11	21.60	ND	-	18.50	18.50	22.80	22.80	ND	-			
BIS-(2-ETHYLHEXYL)PHTHALATE	ND	-	ND	-	NV	1.00	ND	-	ND	-			
CHEMICAL (ug/L)	R6 MEAN	R6 MAX	R7 MEAN	R7 MAX	EW MEAN	EW MAX	OC MEAN	OC MAX	CSB MEAN	CSB MAX			
	110 57	174.00	72 34	155.00	ND	_	67 00	67.00	NS				
REBULLIUM	ND	-	ND	-	ND	_	ND	-	NS				
CHROMIUM	ND	-	ND	<b></b>	ND	_	ND	_	NS	_			
COPPER	ND	-	ND	-	ND	_	ND	_	NS	_			
IFAD	ND	-	ND	-	6 80	8 70	2 40	2 40	NS	_			
MERCURY	ND	-	ND	-	0.43	0.49	0.42	0.42	NS	-			
NICKEI	ND	_	ND	_	ND	-	ND	_	NS				
SILVER	ND	-	3 88	8 00	ND	-	ND	-	NS				
ZINC	ND	_	ND	-	ND	-	46.10	46.10	NS	_			
BIS-(2-ETHYLHEXYL)PHTHALATE	ND	-	ND	-	ND	_	1.00	1.00	58.00	58.00			

# TABLE 7 – 7 REPORTED MEAN AND MAXIMUM CONCENTRATIONS OF SURFACE WATER COC (FILTERED INORGANICS), BY LOCATION NYANZA OPERABLE UNIT 3

REPORTED MEAN: INCLUDES BELOW DETECTION LIMIT (DL) VALUES SET TO 1/2 DL

LOCATIONS SAMPLED: R=REACH 2-7, EW=EASTERN WETLANDS, OC=OUTFALL CREEK, CSB=COLD SPRING BROOK

ND: NOT DETECTED NS: NOT SAMPLED NV: NO VALUE – MEAN EXCEEDS MAXIMUM

7-25

FINAL

#### 7.4.5 Sediment Exposure Assessment

The sediment exposure assessment is based on both the reported average and the maximum concentration of COCs in sediment. An assessment of sediment exposures can present concentrations of COCs in sediment directly as sediment concentrations (ug COC/kg sediment, dry weight), as listed in Table 7-8. As an alternative, sediment COC concentrations can be extrapolated to interstitial water through the use of partitioning coefficients.

Interstitial water concentrations are estimates of actual bioavailable exposure concentrations of chemicals associated with contaminated sediments. Most modeling methods for determining interstitial water concentrations (e.g., equilibrium partitioning, or EP) are limited to organic chemicals, and are not applicable to metals because of their complex partitioning behavior. Metal adsorption-desorption from sediments can depend on ionic strength, pH, Eh, and, for at least some metals, the concentration of acid volatile sulfides (DiToro et al., 1986; DiToro, 1989). Much of the research on metal partitioning behavior is based on laboratory studies in which a specific, well-characterized sediment was spiked with a single contaminant. There have been few field studies investigating complex interactions between multiple contaminants and various sediment types, and the collection of sufficient sitespecific information about the partitioning behavior of metals is difficult for large, complex sites such as the subject study area.

Another alternative for evaluating exposures of COCs in sediment is the measurement of COC concentrations in interstitial water samples.

This approach has many limitations, including difficulties obtaining representative samples that provide acceptably accurate and precise analytical results. Analytical data based on COC concentrations in interstitial water were unavailable for this assessment.

This assessment employed a separate and distinct methodology for organic and inorganic sediment COCs. For inorganic sediment COCs, exposures were based on concentrations of COCs in sediment without extrapolation to interstitial water (ug COC/kg sediment). These exposure concentrations of sediment COCs were compared directly to biological effects data, as described in Section 7.6. Biological effects data are based on concentrations of chemicals in sediments that are associated with known biological effects, including mortality, impaired growth and reproduction, and ecologically significant behavioral changes. Table 7-8 lists the reported average and the maximum concentration of each sediment COC for each reach.

For organic sediment COCs, equilibrium partitioning (EP) was used to extrapolate from concentrations of COCs in sediment (ug/kg sediment, dry weight) to concentrations in interstitial water interstitial water (uq/L). Location-specific estimates of concentrations are based on location-specific concentrations of COCs and organic carbon in sediments. Partitioning coefficients used (Table 7-9) include Foc (fraction organic carbon), Kow coefficient), partitioning Koc (octanol/water (organic carbon/chemical partitioning coefficient), and Kd (sediment/water partitioning coefficient). Section 5.3 presents definitions of, and the relationships between, these input parameters.

EP is appropriate for single non-ionic organic chemicals; however, EP is associated with several assumptions. The primary assumptions are that equilibrium conditions exist between sediment and interstitial waters, and that the process determined for single chemicals are not affected by other chemicals while no single approach appears to be applicable for deriving sediment quality criteria, EPA has selected EP because it "presents the greatest promise for generating defensible natural numerical chemicalspecific sediment quality criteria applicable across a broad range of sediment types" (EPA, 1991d, e, f).

The final result of the equilibrium partitioning approach is the derivation of the mean and maximum estimated interstitial water concentrations of organic sediment COCs for each sampled location (Table 7-10). These values were compared directly to aqueous concentrations of sediment COCs that are potentially hazardous to aquatic biota. The comparisons are made and evaluated in the risk characterization section of this assessment.

#### 7.4.6 Biota Exposure Assessment

The assessment of biota exposure consists of two approaches. The first approach evaluated the exposure assessment based on measured tissue concentrations of biota COCs in sampled fish. The second approach used a food chain model to estimate body burdens of biota COCs in various trophic levels.

TABLE 7–8
REPORTED MEAN AND MAXIMUM CONCENTRATIONS OF SEDIMENT COC, BY LOCATION
NYANZA OPERABLE UNIT 3
MIDDLESEX COUNTY, MASSACHUSETTS

_ _

---

_

LOCATION	BACKG	R1*	R2	R2	R3	R3	R4	R4	R5	R5
CHEMICAL (ug/kg)	MEAN	MAX	MEAN	MAX	MEAN	MAX	MEAN	MAX	MEAN	MAX
ALUMINUM	9200910	2E+07	7608330	2E+07	1E+07	2E+07	1E+07	2E+07	7023330	2E+07
ARSENIC	8740	21100	6950	14900	7610	21500	10680	32300	3830	9200
CADMIUM	ND	-	ND	-	4980	19900	4810	14900	ND	-
CHROMIUM	22400	55200	34000	216000	292620	2620000	71610	224000	17950	60600
COPPER	73020	340400	40190	184000	184470	454000	115850	332000	41670	158000
LEAD	85090	248000	58070	295000	13 <b>79</b> 70	285000	93960	219000	237 <b>9</b> 90	809000
MERCURY (INORGANIC)	270	1590	3810	30600	15 <b>980</b>	54600	3380	7300	980	4100
NICKEL	11350	51000	7470	19100	28430	88800	28690	63000	6790	16100
SELENIUM	1040	3100	ND	-	810	4000	1740	4000	NV	440
ZINC	133500	629000	126440	330000	177060	435000	161050	327000	199880	765000
4,4'-DDD	ND		NV	26	85	370	ND	-	ND	-
4,4'-DDE	ND	-	17	58	32	6 <b>0</b>	ND	_	ND	-
4,4'-DDT	ND	-	172	1400	ND		ND	<b>→</b>	ND	-
ACENAPHTHENE	ND	-	544	1235	NV	100	ND	-	ND	-
ACENAPHTHYLENE	ND	-	486	500	819	880	ND	-	160	160
ANTHRACENE	ND	-	754	2900	NV	240	NV	140	150	150
BENZO(A)ANTHRACENE	ND	-	1105	4400	1346	4500	NV	110	470	470
BENZOAPYRENE	NV	91	1028	3750	1209	4400	140	150	320	320
BENZO(B)FLUORANTHENE	NV	200	1034	3550	1496	4400	275	400	1400	1400
BENZO(G.H.I)PERYLENE	ND	-	594	650	773	1600	NV	92	270	270
BENZOKOFLUORANTHENE	ND		990	3800	1494	3900	NV	150	ND	_
BIS(2-ETHYLHEXYL)PHTHALATE	ND	-	651	2000	791	910	ND	_	330	330
CHRYSENE	NV	180	1092	4200	1737	7700	205	220	700	700
DIBENZOEURAN	ND	_	556	1060	ND	-	ND		ND	-
DIBENZO/A HIANTHRACENE	ND	-	NV	80	NV	310	ND		ND	-
DI-N-BUTYI PHTHALATE	ND	-	ND		1645	5200	ND	-	ND	_
FLUORANTHENE	NV	320	2014	9250	2332	12000	260	270	1100	1100
FLUORENE	ND	_	607	1500	NV	200	ND		ND	_
INDENO(1.2.3-CD)PYRENE	ND	-	626	900	786	1800	NV	94	240	240
METHYLMERCURY	ND	-	109	312	NV	26	27	79	ND	_
NAPHTHALENE	ND	-	599	1180	NV	120	ND	-	ND	-
PHENANTHRENE	NV	160	1635	10650	1368	7300	NV	170	540	540
PYRENE	NV	270	1669	8900	2461	11000	255	260	1100	1100

FINAL

7-28

# TABLE 7-8

## REPORTED MEAN AND MAXIMUM CONCENTRATIONS OF SEDIMENT COC, BY LOCATION

. . . . . . . .

____

NYANZA OPERABLE UNIT 3

MIDDLESEX COUNTY, MASSACHUSETTS

PAGE 2

LOCATION	R6	R6	R7	R7	R8	<b>R8</b>	R9	R9	R10	R10
CHEMICAL (ug/kg)	MEAN	MAX	MEAN	MAX	MEAN	MAX	MEAN	MAX	MEAN	MAX
ALUMINUM	1E + 07	2E+07	7736900	2E + 07	6636000	8310000	9806000	1E+07	6790000	2E+07
ARSENIC	11100	24000	10130	40600	12130	30000	32000	64600	5990	12200
CADMIUM	3830	13600	2210	17900	2080	4300	7050	8600	ND	_
CHROMIUM	66600	281000	47950	209000	22000	36400	50260	78000	11240	17100
COPPER	113500	303000	65940	278000	67610	96600	83330	135000	14115	31000
LEAD	285700	876000	117940	526000	66700	100000	115260	184000	22480	33800
MERCURY (INORGANIC)	3300	17600	1520	5500	1620	2100	3150	3900	161	530
NICKEL	20300	78400	11290	44100	7080	13100	24950	28100	7370	11100
SELENIUM	1370	6100	1180	7200	750	1800	ND		ND	-
ZINC	292300	564000	170060	646000	187360	329000	197130	319000	38350	53400
4,4'-DDD	399	700	ND	-	NS	-	NS		NS	-
4,4'-DDE	ND	-	ND	-	NS	-	NS	~	NS	-
4,4'-DDT	ND	-	ND		NS	_	NS		NS	-
ACENAPHTHENE	NV	74	490	110	NS	-	NS		NS	-
ACENAPHTHYLENE	ND	-	ND	-	NS		NS		NS	_
ANTHRACENE	NV	240	521	210	NS	-	NS	~	NS	-
BENZO(A)ANTHRACENE	1018	1500	430	880	NS	-	NS		NS	-
BENZO(A)PYRENE	NV	1100	720	880	NS	-	NS		NS	-
BENZO(B)FLUORANTHENE	1626	3400	516	910	NS	-	NS	-	NS	-
BENZO(G,H,I)PERYLENE	NV	820	655	610	NS	-	NS	-	NS	-
BENZO(K)FLUORANTHENE	2413	980	446	850	NS	-	NS	-	NS	-
BIS(2-ETHYLHEXYL)PHTHALATE	NV	1900	846	1100	NS	_	NS		NS	-
CHRYSENE	NV	1500	640	1200	NS	-	NS		NS	
DIBENZOFURAN	ND	-	ND	-	NS	-	NS	-	NS	-
DIBENZO(A,H)ANTHRACENE	ND	-	ND	-	NS	-	NS		NS	_
DI-N-BUTYLPHTHALATE	ND	-	ND	-	NS	-	NS		NS	-
FLUORANTHENE	1975	2600	<b>953</b>	1900	NS	-	NS		NS	-
FLUORENE	NV	160	496	140	NS	-	NS		NS	-
INDENO(1,2,3-CD)PYRENE	NV	740	631	540	NS	_	NS		NS	-
METHYLMERCURY	ND	-	ND	-	NS	-	NS	-	NS	-
NAPHTHALENE	ND	-	ND	-	NS	-	NS		NS	-
PHENANTHRENE	1171	1600	593	1200	NS	-	NS		NS	-
PYRENE	2075	2900	876	1700	NS	-	NS	~	NS	

7-29

#### TABLE 7-8

### REPORTED MEAN AND MAXIMUM CONCENTRATIONS OF SEDIMENT COC, BY LOCATION

NYANZA OPERABLE UNIT 3

MIDDLESEX COUNTY, MASSACHUSETTS

......

PAGE 3

		EW MAX	CBC MEAN	CBC MAX	OC MEAN	OC MAX		RW		CSB MAX
OT IL MICHE LUGHAN										
ALUMINUM	7216000	1E+07	3520000	3690000	5876670	1E+07	2E+07	3E+07	9225000	1E+07
ARSENIC	6590	12700	4300	6600	2930	4100	24960	37200	3500	4300
CADMIUM	ND	-	ND	-	ND	-	2280	5600	ND	_
CHROMIUM	123680	462000	82300	135000	341800	988000	155500	208000	15150	16700
COPPER	<b>63730</b>	120000	63350	92500	137830	359000	217750	306000	31500	32900
LEAD	72580	142000	38250	57500	105950	233000	435750	758000	234000	328000
MERCURY (INORGANIC)	35878	152000	6800	7100	35330	99200	710	970	ND	-
NICKEL	10143	40400	ND	-	ND		89890	186000	ND	-
SELENIUM	1430	6500	ND	-	ND	-	ND	-	ND	-
ZINC	65393	164000	92950	103000	166300	390000	365880	542000	148000	166000
4,4'-DDD	76	117	ND	-	ND	-	32	108	NS	-
4,4'-DDE	ND	-	ND	-	ND	-	11	21	NS	-
4,4'-DDT	ND	-	ND	-	ND	-	ND	-	NS	-
ACENAPHTHENE	ND	-	ND	-	ND	-	ND	-	NS	-
ACENAPHTHYLENE	ND	-	NV	140	263	310	ND	-	NS	-
ANTHRACENE	ND	-	215	250	NV	185	NV	3100	NS	_
BENZO(A)ANTHRACENE	ND	-	710	810	800	1050	5713	11000	NS	-
BENZO(A)PYRENE	643	860	720	770	900	1200	3888	4700	NS	-
BENZO(B)FLUORANTHENE	NV	2500	820	840	1233	1750	4263	5200	NS	-
BENZO(G,H,I)PERYLENE	ND	-	NV	270	492	730	ND	-	NŜ	-
BENZO(K)FLUORANTHENE	ND	-	680	780	473	1000	4300	6300	NS	-
BIS(2-ETHYLHEXYL)PHTHALATI	. ND	-	150	200	960	2000	NV	2500	NS	-
CHRYSENE	1013	1200	745	760	1077	1400	5975	8600	NS	-
DIBENZOFURAN	ND	-	NV	92	ND	-	ND	-	NS	-
DIBENZO(A, H)ANTHRACENE	ND	-	ND	-	ND	-	ND	-	NS	-
DI-N-BUTYLPHTHALATE	ND	-	ND	-	ND	-	ND	-	NS	-
FLUORANTHENE	1213	1600	1300	1400	1307	1600	14263	20000	NS	-
FLUORENE	ND	-	NV	84	ND	-	NV	2800	NS	-
INDENO(1,2,3-CD)PYRENE	ND	-	NV	330	453	635	ND	-	NS	-
METHYLMERCURY	68	229	ND	-	ND	-	ND	-	NS	-
NAPHTHALENE	1563	2300	565	600	218	340	4081	6700	NS	-
PHENANTHRENE	NV	620	760	790	630	930	12113	16000	NS	-
PYRENE	1263	1700	1200	1200	1800	2500	9400	13000	NS	

ĺ

_____

# REPORTED MEAN AND MAXIMUM CONCENTRATIONS OF SEDIMENT COC, BY LOCATION NYANZA OPERABLE UNIT 3

MIDDLESEX COUNTY, MASSACHUSETTS

PAGE 4

TABLE 7-8

LOCATION:	BW	BW	HP	HP
CHEMICAL (ug/kg)	MEAN	MAX	MEAN	MAX
	45 . 07	05 . 07	15 . 07	45 . 07
ADOFANO	1E+07	22 + 07	10050	10050
ARSENIC	6220	11900	10850	10050
	ND	-	ND	-
CHROMIUM	24010	101000	40200	40200
COPPER	18013	83400	136000	136000
	/1190	254000	149000	149000
MERCURY (INORGANIC)	1071	7600	3500	3500
NICKEL	7180	21800	11300	11300
SELENIUM	70	1200	ND	-
ZINC	41078	127000	302000	302000
4,4'-DDD	NS	-	NS	-
4,4'-DDE	NS	-	NS	-
4,4'~DDT	NS	-	NS	-
ACENAPHTHENE	NS	-	NS	-
ACENAPHTHYLENE	NS	-	NS	-
ANTHRACENE	NS	-	NS	-
BENZO(A)ANTHRACENE	NS	-	NS	-
BENZO(A)PYRENE	NS	-	NS	-
BENZO(B)FLUORANTHENE	NS	-	NS	-
BENZO(G,H,I)PERYLENE	NS	_	NS	-
BENZOKAFLUORANTHENE	NS	-	NS	-
BIS/2-ETHYLHEXYL)PHTHALATE	NS		NS	-
CHRYSENE	NS	_	NS	-
DIBENZOFURAN	NS	-	NS	-
DIBENZO (A H) ANTHRACENE	NS	-	NS	-
DI-N-BUTYLPHTHALATE	NS	-	NS	_
FLUORANTHENE	NS	-	NS	-
FLUORENE	NS	_	NS	_
INDENO(1.2.3-CD)PYRENE	NS	_	NS	-
METHYLMERCURY	NS	-	NS	-
NAPHTHALENE	NS	_	NS	_
PHENANTHRENE	NS	-	NS	_
PYRENE	NS	-	NS	_
	113	-	140	

FINAL

-

-----

#### TABLE 7-8 REPORTED MEAN AND MAXIMUM CONCENTRATIONS OF SEDIMENT COC, BY LOCATION NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS PAGE 5

#### REPORTED MEAN: INCLUDES BELOW DETECTION LIMIT (DL) VALUES SET TO 1/2 DL LOCATIONS SAMPLED: R=REACH 1-10, EW=EASTERN WETLANDS, CBC=CHEMICAL BROOK CULVERT, OC=OUTFALL CREEK RW=RACEWAY, CSB=COLD SPRING BROOK, BW=BORDERING WETLANDS, HP=HEARD POND

ND: NOT DETECTED NS: NOT SAMPLED NV: MEAN EXCEEDS MAXIMUM 7-33

#### PARTI IT KON ING COEFFICIENTS FOR ORDANIC BEDIMENT COCS IN YANZA OPERABLE UNIT 3 INIDDLEBEX COUNTY, MASSACHUSETTS

	-					-	-													
		100/	ATION	8450 +	NE	<b>P</b> D	Μ.	<u>No.</u>	<u> </u>	<u>N7</u>	<b>N</b>	No	A10	EW	CBC	oc	RW	C38	нр	BW
				ME AN	ME AN	ME AN	ME AN	MEAN	MEAN	MEAN	MEAN	MEAN	MEAN	MEAN	MEAN	MEAN	MEAN	MEAN	MEAN	MEAN
				FOC	FOC	FOC	FOC	FOC	FOC	FOC	FOC	FOC	FOG	FOC	FOC	FOC	FOC	FOC	FOC	FOC
				0 05	0 03	0.04	0.05	0 02	0.04	0.04	0 12	0.01	0.01	0 01	0.05	0.02	0 03	0 07	0 18	0.09
ORGANIC	100	KOW	100																	
BED COCS	KOW	MEF!	KOC	KD _	<b>KD</b>	10	KD	KD	<u></u>	KD	KD	KD	KD	KD	KD	KD	KD	KD	KD	KD
4,4'-DDD	e 07			1346	1346	1795	2243	887	1785	1795	5384	449	449	449	2243	897	1348	3141	7178	4038
4,4'-DDE	6 51	1	4 82	2487	2467	3315	4144	1058	3318	3315	9946	829	829	829	4144	1658	2487	5802	13261	7460
4.4'-DDT		2	4 84	2061	2061	2747	5434	1374	2747	2747	8242	687	687	667	3434	1374	2061	4808	10990	6182
ACENAPHTHENE	3 84	3	3 50	117	117	156	185	78	156	150	409	30	38	39	195	78	117	274	625	352
ACENAPHTHYLENE	4 07	•	3 81	183	193	257	321	128	257	257	771	64	64	64	321	126	183	449	1027	578
ANTHRACENE	4 45	- 1	4 18	437	437	563	729	292	583	583	1749	146	146	146	729	292	437	1020	2332	1312
BENZO (A) ANT H PACE NE	\$ 01	- 1	5 25	6342	\$342	7123	8804	3142	7125	7123	21300	1761	1781	1781	8904	3582	5342	12465	28492	16027
BENZOLAPYRENE	0 42	1	6.01	30000	20000	40882	\$1115	20448	40692	40692	122676	10223	10223	10223	51115	20446	30669	71561	163568	92007
BENZO(B)FLUOPANTHENE	6 32	•	5 92	24717	24717	32939	41155	18478	32958	32858	96609	6239	8239	6239	41195	16478	24717	57674	131625	74152
BENZO (Q.H.) PERVLENE	7 05	•	0.00	119403	119403	198304	199004	79902	150204	159204	477611	39801	39801	39601	199004	79602	119403	276606	636614	358208
BENZOROFLUORANTHENE	0.01	•	0 41	77558	77598	103407	12020	\$1704	103407	103407	\$10222	25852	25852	25852	129259	51704	77558	160983	413629	232687
BIS(2-EH)PHTHALATE	\$ 11	- 5	4 18	431	431	\$74	710	207	574	574	1722	144	144	144	718	287	431	1005	2296	1292
CHRYSENE	\$ 71	•	5 34	4629	06.29	8838	11048	4419	8636	6836	20515	2210	2210	2210	11048	4418	6629	15467	35353	19665
DIBENZOFURAN	4 18	•	3 85	215	215	200	356	143	296	288	854	72	72	72	358	143	215	501	1144	844
DIBENZO(AH)ANTHRACEN	0.50	•	0.00	38447	38447	48588	807 <b>4</b> 5	54546	48598	48598	145788	12149	12149	12140	60745	24298	30447	85043	194383	109341
DI-N-BUTYLPHTHALATE	5 80	1	4 48	705	785	1000	1320	\$ 30	1080	1080	3101	205	265	265	1326	530	795	1854	4242	2386
FLUORANTHEME	\$ 19		4 86	8190	2190	2070	3566	1430	2078	2678	8635	720	720	720	3596	1439	2150	5037	11513	6474
FLUORENE	4 27	4	2 00	207	297	305	494	190	395	395	1100		99	99	494	196	297	692	1562	800
INDENO(1,2,3 - CD)PYRENE	7 00	•	7 17	445.233	445233	883844	742056	200455	\$83844	593644	1780933	148411	148411	148411	742056	296622	445233	1035878	2374578	1335700
METHYLMERCURY	0.20	•	4 78	1864	1864	2485	3107	1243	2465	2485	7456	621	621	621	3107	1243	1864	4350	8942	5592
NAPHTHALENE	3 34	•	3 20		46	64	80	32	64	64	182	18	14	10	60	52	48	112	250	144
PHENANTHRENE	4 36	3	4 60	300	300	480	800	240	480	480	1441	120	120	120	600	240	360	840	1821	1060
PYRENE	4 00		4 87	1180	1100	1475	1843	737	1475	1475	4424	369	368	360	1643	737	1106	2580	5896	3318

+ BACKGROUND

* KOW PEF REFERENCE FOR LOG KOW

1. EPA 1998

- 2 SCHUURMANN AND RLEIN 1988
- 3 EPA 1001d+1
- 4 DUXBURY, 1988
- 5 HOWAND, 1989
- & ZAROOGIAN ET AL. 1996
- 7 EPA 19685

& ESTIMATED BY CALCULATION & YMAN &T AL 1988

LOCATION REPEACH EW-EASTERN WETLANDS COC+CHEMICAL BROOK GULVERT, DC+OUTFALL CREEK, RW+RACEWAY C88+COLD BRONK HP+HEARD POND BW+BORDERNG WETLANDS

.....

FOC = FRACTION ORGANIC CARBON LOG KOC = 0 544 LOG KOW + 1 377 (HON - PNHO) = 0 937 LOG KOW - 0 506 (PAHO) KD = KOC * FOC

#### TABLE 7-10

#### ESTIMATED LOCATION - SPECIFIC INTERSTITIAL WATER (W) CONCENTRATIONS OF ORGANIC SEDIMENT COC NYANZA OPERABLE UNIT 3 NIDDLEGEX COUNTY, MASSACHUSETTS

	- <u></u>		<b>MCKO</b>	IQUNQ		-		<u>. M2</u>					A3					R4		
		ME AN	MAX	ME AN	MAX		MEAN	MAX	MEAN	MAX		MEAN	MAX	MEAN	MAX		MEAN	MAX	MEAN	MAX
		πD	<b>BED</b>	RV .	NV		SE D	BE D	w	w		8E D	8E D	w	w		SED	8ED	w	w
ORGANIC COC		CONC	OOHC	CONC	CONC		CONC	CONC	CONC	CONC		CONC	CONC	CONC	CONC		CONC	CONC	CONC	CONC
IN SEDMENT			. unha	<u>ug4</u>	<u></u>	KO	uging _	ughg_	ug/L	ugl	KD	uaka	ugkg	ug/L	սց/Լ	_ <u>KD</u>	ug/kg	uaka	ug/L	<u> 49/L</u>
4.4'-000	1346	NO	-	-	-	1346	NV	24	0.00	0 02	1795	65	370	0 05	0 21	2243	ND	-	-	-
4.4'-DDE	2487	NO	-	-	-	2487	17	56	0 01	0 02	3315	32	80	0 01	0 02	4144	ND	-	-	-
4.4'-DOT	2081	NO	-	-	-	2061	172	1400	0.06	0 68	2747	ND	-	-	-	3434	ND	-	-	-
ACENAPHTHENE	117	NO	-	-	-	117	544	1236	4 64	10 83	150	NV	100	-	0 64	105	ND	-	-	-
ACENAPHTHYLENE	183	NO	-	-	-	193	466	500	2 52	2 60	257	819	880	3.19	3 4 3	321	ND	-	~	-
ANTHRACENS	437	ND	-	-	-	437	754	2900	1 72	6 63	583	NV	240	-	0 41	729	NV	140	-	0.19
BENZO(A) ANTHRACENE	\$342	NO	-	-	-	\$342	1105	4400	0 21	0 82	7123	1346	4500	0.18	0 63	8904	NV	110	-	0 01
BENZOLAIPYRENE	30000	NV	81	-	8 00	30969	1028	3750	0 03	0 12	40892	1209	4400	0 03	011	51115	140	150	0 00	0 00
BENZO B) FLUORANTHENE	84717	HV.	200	-	0 01	84717	1034	3550	0.04	0 14	32958	1498	4400	0 05	013	41195	275	400	0 01	0 01
BENZOIG H MPERYLENE	119403	NO	-	-	-	119403	594	650	0 00	0 01	159204	773	1600	0 00	0 01	199004	NV	92	-	0 00
BENZONGFLUORANTHENE	77598	NO	-	-	-	77556	890	3800	0 01	0.05	103407	1494	3900	0 01	0.04	129259	NV	150	-	0 00
BIS(2-EH)PHTHALATE	421	NO	-	-	-	431	851	2000	1 51	4 85	574	791	910	1 38	1 59	718	ND	-	-	-
CHRYSENE	0620	NV	100	) -	0 83	6629	1082	4200	0 10	0 63	6838	1737	7700	0 20	0 87	11048	205	220	0 02	0 02
DIBENZOFURAN	215	NO	-	-	-	215	550	1060	2 59	4 94	286	ND	-	-	-	358	ND	~	-	-
DIBENZOA,HIANTHRACENE	38447	NO	-	-	-	38447	NV	80	-	0 00	46596	NV	310	-	0 01	80745	ND	-	-	-
DI-N-BUTYLPHTHALATE	795	HO	-	-	-	795	ND	-	-	-	1060	1645	5200	1 55	4 90	1326	ND	-	-	-
FLUORANTHENE	2190	ITV	320	· -	0 15	2150	2014	8250	0 83	4 20	2878	2332	12000	0 61	4 17	3598	260	270	0 07	0.08
FLUOPENE	207	NO	-	-	-	207	807	1500	2 05	5 06	395	NV	200	-	0 5 1	494	ND	-	-	-
INDENO(1.2 3-COPYREME	445233	NO	-	-	-	445233	626	900	0.00	0.00	503044	766	1800	0 00	0 00	742056	NV.	84	-	0 00
METHYLMERCURY	1004	ND	-	-	-	1864	109	312	0.06	0 17	2465	NV	20	-	0 01	3107	27	79	0 01	0 03
NAPHTHALENE	46	ND	-	-	-	46	500	1180	12 48	24 54	64	NV	120	-	1 87	80	ND	-	-	-
PHENANTHRENE	380	NV	100	- 1	0 44	360	1835	10850	4 64	29 17	480	1368	7300	2 65	15 20	800	NV	170	-	0 24
PYRENE	1199	<u>MY</u>	279	!_ <u>-</u>	981	1100	1000	8800	111	8 05	1475	2461	11000	1 87	7 46	1643	255	200	0 14	014
			~					<b>PH6</b>					A7					EW		
		MEAN	MAX	MEAN	MAX		MEAN	MAX	MEAN	MAX		MEAN	MAX	MEAN	MAX	·····	MEAN	MAX	MEAN	MAX

		MEAN	MAX	MEAN	MAX		MEAN	MAX	MEAN	MAX		MEAN	MAX	MEAN	MAX	Pro	MEAN	MAX	MEAN	MAX
		SED.	BED		NV .		8E0	BED	w	w		SED	8ED	w	w		GED	8ED	w	w
ORGANIC COC		CONC	CONC	CONC	CONC		CONC	CONC	CONC	CONC		CONC	CONC	CONC	CONC		CONC	CONC	CONC	CONC
HI DEDMENT		. sebe	<u>unka</u>		<b>V01</b>	. 192	yoka	ugkg	ugi.		<u>_K</u>	<u>yaka</u>	ugkg	ug/L	ug/L	KD	ugkg	uaka	_ <u>uo/L</u>	<u>491.</u>
4.4'-000	887	ND	-	-	-	1786	300	700	0 22	0 30	1	795 NI	<b>)</b> –	-	-	449	78	117	0 17	0 26
4,4'-DOE	1050	HO	-	-	-	3318	ND	•	-	-	3	315 NI	) -	-	-	829	ND	-	-	-
4.4 - DOT	1374	NO	-	-	-	2747	ND	-	-	-	2	747 NI	- (	-	-	687	ND	-	-	-
ACENAPHTHENE	78	ND	-	-	-	156	NV	74	-	0 47		158 49	110		0.70	39	ND	-	-	-
ACENAPHTHYLENE	126	100	180	1 26	1 25	257	ND	-	-	-		257 NI	) -	-	-	64	NÐ	~	-	-
ANTHRACENE	202	180	190	0 51	0 51	583	NV	240	-	041		583 62	210	) -	0 36	146	ND	-	-	-
BENZOLAJANTHRACENE	3542	470	470	013	0 13	7123	1018	1500	0 14	0 21	7	23 43	660	0.06	0 12	1781	ND	-	~	-
BENZOLAIPYRENE	20448	820	170	8 02	0.02	40862	NV	1100	-	0 03	40	102 72		0 02	0 02	10223	843	860	0.08	0.06
BENZOB)FLUORAMTHENE	19478	1408	1400		0 00	32998	1626	3400	0.05	0 10	32	50 \$1	910	0 02	0 03	8239	NV	2500	0 00	0 30
BENZOIG H OFERYLENE	70002	270	270	0 00	0 00	188204	NV	820	-	0 01	159	204 65	610	-	0 00	39801	ND	-	-	-
BENZORGFLUORANTHENE	81704	HO	-	-	-	103407	2413	980	-	0 01	103	107 44	850	0.00	0 01	25652	ND	-	-	-
BIS(2-EH)PHTHALATE	267	330	\$30	1 15	1 18	674	NV	1900	-	3 31	1	574 64	1100	1 47	1 82	144	ND	-	-	-
CHRYBENE	4418	700	700		0 10	8638	NV	1500	-	0 17		38 64	1200	0 07	0 14	2210	1013	1200	0.48	0 54
DIBENZOFURAN	14\$	NO	-	-	-	200	ND	-	-	+	:	196 NC	) -	-	-	72	ND	-	-	-
DIBENZOM, HJANTHRACENE	94299	NO	-	-	-	40500	ND	-	-	-	48	590 NC	) -	-	-	12149	ND	-	-	-
DI-N-BUTYLPHTHALATE	\$30	ND	-	-	-	1080	NO	-	•	-	10	160 NC	) -	-	-	265	ND	-	-	-
FLUORANTHENE	1430	1100	1100	0.70	9.76	2678	1075	2000	0 66	0 80	2	78 95:	1900	0 33	0 88	720	1213	1000	1 69	2 22

Ć

FINAL

7-34

÷

.

(

7-35

			-					-					<b>A</b> 7					EW	•	
		MEAN	MAX	MEAN	MAX	<u> </u>	MEAN	MAX	MEAN	MAX		MEAN	MAX	MEAN	MAX		MEAN	MAX	MEAN	MAX
		SE D	CE D	•	w		BED	0ED	w	w		BED	8ED	w	w		8ED	8ED	w	w
ORGANIC COC		OONC	CONC	CONC	CONC		CONC	CONC	CONC	CONC		CONC	CONC	CONC	CONC		CONC	CONC	CONC	CONC
IN SEDIMENT	<u>.</u> KQ	_ 1921.0	veta.	yon.		_KD	uging _	ugka	<u>ug/</u>	ug/L	KD	vaka	uaka.	ug/L	ug/L	KD	ugkg	uaka	ug/L	<u>uo/L</u>
FLUOPENE	198	ND	-	-	-	305	NV	180	-	0 40	395	496	140	-	0 35	99	ND	-	-	-
INDENO(1 2 3-CD) PYPE NE	206422	240	240	0 00	0 00	593644	NV	740	-	0 00	593044	631	540	-	0 00	148411	ND	-	-	-
METHYLMERCURY	1243	ND	-	-	-	2485	ND	-	-	-	2485	ND	-	-	-	621	66	229	0 11	0 37
NAPHTHALENE	14	ND	-	-	-	64	ND	-	-	-	64	ND	-	-	-	16	1563	2300	97 52	143 51
PHENANTHRENE	240	540	\$40	2 25	2 25	480	1171	1800	2 44	3 33	480	593	1200	1 23	2 50	120	NV	820	-	5 16
PYRENE	131	1199			1.49	1975	2075		1.41	107	1475	876	1700	0 59	1.15	309	1203	1700	3 43	4 61

			CBC					oc					<b>HW</b>		
	<b></b> ·	MEAN	MAX GED	MEAN W	MAX W		MEAN	MAX	MEAN	MAX NV		MEAN	MAX SED	MEAN W	MAX
ORGANIC COC		CONC	CONC	CONC	COHC		CONC	CONC	CONC	CONC		CONC	CONC	CONC	CONC
IN SEDIMENT	<u>#9</u>	. 992-8 .	_ <u>uota</u> _	ugt	ugl	KD	ugkg	ugkg	ug/L	ug/L	<u>KD</u>	ug/kg	ugkg	սց/լ	ug/L
44'-000	2243	ND	-	-	-	007	ND	-	-	-	1346	32	106	0 02	0.06
4.4'-DDE	4144	ND	-	-	-	1658	ND	-	-	-	2487	11	21	0 00	0 01
4,4° DDT	3434	ND	-	-	-	1374	ND	-	-	-	2081	ND	-	-	-
ACENAPHTHENE	195	NO	-	-	-	76	ND	-	-	-	117	ND	-	-	-
ACENAPHTHYLENE	321	NV	140	-	0 44	120	563	310	2 05	2 41	103	ND	-	-	-
ANTHRACENE	729	215	210	0 20	0 34	202	NV	185	-	0 63	437	NV	3100	-	7.09
BENZO(A) ANTHRACENE	8804	710	810	8 06	0.00	3562	800	1050	0 22	0 29	5342	5713	11000	1 07	2 08
BENZO(A)PYRENE	\$1115	720	770	001	0 02	20446	800	1200	0.04	0.06	30669	3888	4700	0 13	0 15
BENZOBIFLUORANTHENE	41196	620	840	0 05	0 02	10470	1233	1750	0 07	0 11	24717	4263	5200	0 17	0 21
BENZOR HIPERYLENE	188004	NV	270	-	0 00	79602	492	730	0 01	0 01	119403	ND	-	-	-
BENZOROFLUORANTHENE	128258	880	700	0 01	0 01	61704	473	1000	-	0 02	77558	4300	6300	0.06	0.08
BIS(2-EH)PHTHALATE	716	180	200	021	0 20	207	960	2000	3 34	6 97	431	NV	2500	-	5 61
CHRYSENE	11048	748	780	0 07	0 07	4418	1077	1400	0 24	0 32	6629	5975	8600	0.90	1 30
DIBENZOFURAN	250	NV	82	-	0.26	149	ND	-	-	-	215	ND	-	-	-
DIBENZO(A.H)ANTHRACENE	80746	ND	-	-	-	24298	ND	-	-	-	36447	ND	-	-	-
DI-N-BUTYLPHTHALATE	1320	ND	-	-	-	\$30	ND	-	-	-	795	ND	-	-	-
FLUORANTHENE	3590	1 200	1400	0 36	0.30	1430	1307	1000	0 91	1 11	2158	14263	20000	6 61	9 27
FLUORENE	494	NV	M	-	0 17	188	ND	-	-	-	297	NV	2600	-	9 44
INDENO(1,2 3-CD)PYRENE	742064	NV	330	-	0 00	200422	453	635	0 00	0 00	445233	ND	-	-	-
METHYLMERCURY	3107	NÜ	-	-	-	1243	ND	-	-	-	1864	ND	-	-	-
NAPHTHALENE			808	7 06	7.40	22	218	340	8 80	10 61	48	4081	6700	64.66	139 35
PHENANTHRENE	808	780	780	1 27	1 32	240	630	030	2 62	3 07	360	12113	16000	<b>33 63</b>	44 43
PYRENE	1843	1200	1200		0 05	737	1800	2500	2 44	3 30	1108	6400	1 3000	8 50	11 76

SAMPLED LOCATIONS R-REACH EW-EASTERN WETLANDS COC+CHEMICAL BROOK CULVERT, RW-RACEWAY

IN CONC (LIGAL) = SED CONC (LIGAL) / KD

ND=NOT DETECTED NV=NO VALUE - MEAN EXCEEDS MAJOMUM NS=NOT BAMPLED (

.

(

FINAL

#### 7.4.6.1 <u>Measured Biota Exposure Assessment</u>

Except for dieldrin, biota COCs of a closely related nature were combined for evaluating exposures based on measured tissue concentrations in sampled fish. DDD and DDT were combined and evaluated as Total DDT.

Similarly, Aroclor 1254 and 1260 were combined as Total PCBs, alpha and gamma chlordane were combined as Total Chlordane. Mercury in fish was analyzed by cold vapor technique, which measures total mercury; in addition, methylmercury was analyzed be a method specific for methylmercury. Since the cold vapor analysis measures both inorganic and methylmercury, mercury concentrations based on this method were used to represent total mercury. The final biota COCs evaluated for exposure assessment consisted of Total DDT, Total PCBs, Total Chlordane, Total Mercury, and dieldrin.

The fish species sampled for this assessment included largemouth bass (<u>Micropterus salmoides</u>), yellow perch (<u>Perca flavescens</u>), and bullheads (<u>Ictalurus</u> spp.). In addition, caddis fly larvae were collected and analyzed for potential contaminants of concern. The usefulness of the caddis fly data was limited by technical problems associated with sample volume; therefore, these data were not included in the quantitative exposure assessment.

Fish were sampled in two phases. Phase I data were collected from Reaches 1, 3, 4, and 6. Phase II data were collected from Reaches 1, 2, 3, 4, 6, and 9. Phase I fish data consisted of many individual filet samples and relatively few composited viscera or offal samples. Viscera samples are the whole body minus the filet portions. No viscera samples were matched to filet samples for the same fish. Composited viscera samples were assumed to be representative of each particular fish species. The ingestion of only the filet portion of fish is not considered a likely route of exposure for nonhuman predators within food chains; therefore, COC whole-body concentrations were extrapolated from the concentrations measured in both filet and viscera samples.

The mean and maximum tissue concentrations of all biota COCs listed in Table 7-11, except for mercury, are based on estimated wholebody concentrations calculated from the viscera/filet (V/F) ratios derived for each species. Mercury concentrations listed on this table were based on total mercury in filet samples, which were considered to be representative of whole body total mercury. The basis for this assumption is explained below.

V/F ratios were determined for each fish species and for each biota COC other than mercury, and were based on average concentrations of each COC measured in viscera and filet samples taken from each fish species. The following table lists the V/F ratios determined for the three fish species:

W92194F

#### T/ 17-11

#### DETECTED MEAN AND MAXIMUM CONCENTRATION OF BIOTA COCS, BY SPECIES AND REACH NYANZA III REMEDIAL INVESTIGATION MIDDLESEX COUNTY, MASSACHUSETTS

	_			LAR	SEMOUTH E	BASS	]					
LOCATION	8	8	R2	R2	R3	R3	R4	R4	R6	R6	R9	R9
CHEMICAL	MEAN	MAX	MEAN	MAX	MEAN	MAX	MEAN	MAX	MEAN	MAX	MEAN	MAX
DDT (TOTAL) 1/	91	190	ND	-	116	293	95	114	57	88	73	137
PCBS (TOTAL) 2/	108	152	ND	-	249	709	553	553	177	208	405	750
CHLORDANE (TOTAL) 3/	1.2	12	ND	-	2.3	3.4	14.6	14.6	ND	-	0.5	0.5
DIELDRIN	2.8	2.8	ND	-	0.3	0.3	0.9	0.9	ND		0.3	0.3
MERCURY (TOTAL)	745	1180	1120	1120	2870	7600	1470	4190	940	1800	1320	3200

				YEL	LOW PERC	H						
LOCATION	В	8	R2	R2	R3	R3	R4	R4	R6	R6	R9	R9
CHEMICAL	MEAN	MAX	MEAN	MAX	MEAN	MAX	MEAN	MAX	MEAN	MAX	MEAN	MAX
					_							
DDT (TOTAL) 1/	NS	-	80	66	63	155	55	66	29	65	34	77
PCBS (TOTAL) 2/	NS	-	450	450	206	704	136	135	68	95	157	<b>29</b> 7
CHLORDANE (TOTAL) 3/	NS	-	ND	-	1.5	1.9	ND	-	2.1	2.3	2.3	2.3
DIELDRIN	NS	-	ND	-	ND	-	0.6	0.6	2.2	2.6	ND	-
MERCURY (TOTAL) 4/	3330	9600	1430	1980	1560	3640	760	760	700	1400	720	1800

					BULLHEAD		ר					
LOCATION	В	В	R2	R2	R3	R3	R4	R4	R6	R6	R9	R9
CHEMICAL	MEAN	MAX	MEAN	MAX	MEAN	MAX	MEAN	MAX	MEAN	MAX	MEAN	MAX
DDT (TOTAL) 1/	NS	-	NS	_	32	47	86	170	<b>6</b> 5	<del>96</del>	NS	-
PCBS (TOTÁL) 2/	NS	-	NS	-	127	150	127	216	127	190	NS	-
CHLORDANE (TOTAL) 3/	NS	-	NS	-	4.2	5.2	12	20.8	5.2	502	NS	_
DIELDRIN	NS	-	NS	-	2.6	2.6	ND	_	0	0	NS	_
MERCURY (TOTAL)	NS	-	NS		1990	2850	920	1200	740	880	NS	-

B = BACKGROUND (Includes Reach 1)

1/:4,4'-DDD + 4,4'-DDT 2/:AROCLOR 1254 + AROCLOR 1260 3/:ALPHA CHLORDANE + GAMMA CHLORDANE 4/:VALUES FOR YELLOW PERCH, INCLUDES REACH 1 PERCH WITH ABNORMALLY HIGH MERCURY CONCENTRATIONS ND:NOT DETECTED NS:NOT SAMPLED

CONCENTRATIONS (ug/kg, wet wi) UNADJUSTED FOR FISH AGE, LENGTH, OR WEIGHT VALUES BASED ON FILET OR VISCERA SAMPLES ADJUSTED TO WHOLE BODY CONCENTRATIONS WITH V/F RATIOS (EXCEPT MERCURY) MERCURY VALUES BASED ON TOTAL MERCURY IN FILET SAMPLES (V/F RATIO APPROXIMATELY = 1)

FINAL

	TOTAL TDDT	TOTAL TPCB	TOTAL CHLORDANE	DIELDRIN
Largemouth Bass	6.6	2.2	1.3	10
Yellow Perch	3.4	0.8	0.5	10
Bullhead	5.4	3.0	9.4	10

Because fish samples consisted of either composited viscera samples or individual filet samples, conversion of filet or viscera concentrations to estimated whole body concentrations required a different conversion formula for each sample type. The conversion formulas are based on the following assumption:

Whole-Body (WB) Weight = Filet Weight + Viscera Weight

Individual filet and viscera weights were unavailable. In general, the filet portions of food fish are approximately equal in weight to that of the viscera portions. For this study, filet weight was assumed to equal viscera weight. Based on this assumption, the following conversions were used, as appropriate to estimate wholebody tissue concentrations of biota COCs for all fish species:

#### <u>WB Conc = Filet Conc (measured) + (Filet Conc * V/F Ratio)</u> 2

# <u>WB Conc = Viscera Conc (measured) + (Viscera Conc / V/F Ratio)</u>

Concentrations of total mercury were evaluated because of potential methylation of inorganic mercury within biological tissues. Methylmercury is generally thought to partition preferentially to lipid (fat), while inorganic mercury is not expected to have a similar affinity for lipids. In addition, most mercury in fish tissues is generally accepted to be in the form of methylmercury. However, the results of this study do not fully support these assumptions.

Largemouth bass and yellow perch were sampled from six locations, and bullhead were sampled from three locations; resulting in 15 total cumulative samples. No single fish was analyzed for wholebody concentrations of total mercury; therefore, the relationships between filet and viscera samples are based on a comparison of the cumulative species- and location-specific mean and maximum values.

Total mercury V/F ratios for yellow perch and largemouth bass averaged approximately 1.0, suggestion that mercury does not differentially partition to lipids. In fact, studies have shown that methylmercury preferentially binds to sulfhydryl groups in proteins (Dr. D. Hinckley, personal communication, 1992). Therefore, similar concentrations of mercury in filet and viscera

W92194F

samples are not expected. In addition, viscera samples were based on relatively few (bass=3, perch=1, bullhead=2) composited samples, while filet samples were based on individual fish. Filet samples were therefore considered to be the most appropriate and most representative samples for estimating total mercury concentrations in individual fish. Table 7-11 lists estimated mean and maximum whole-body total mercury concentrations for each species and each

FINAL

The mean and maximum estimated whole-body mercury concentrations in yellow perch collected from Background locations (Tables 7-11, 7-20, and 7-21) may not be truly representative of that species or A filet sample from one individual yellow perch location. collected from Reach 1 contained extremely elevated concentrations of mercury (9,600 ug/kg) relative to other yellow perch collected from Reach 1. This value is therefore considered to be suspect. Because the source of this apparently non-representative value cannot be determined, the value was retained and evaluated. However, it is likely that this value represents a transcription or laboratory error that can not be identified. Risk estimates (Tables 7-20 and 7-21) associated with this sample (yellow perch, Reach 1) are likely to overestimate actual risks for this species and location.

Phase II data included estimates of fish age and measurements of length and weight; Table 7-12 summarizes these data, while Table 7-21 presents a more thorough analysis of Phase II data, in which estimated whole-body total mercury concentrations are normalized for fish age. The data summarized in Table 12 show no consistent relationships between location and fish age, length, or weight. However, it should be noted that, on average, yellow perch collected from Reach 3 were lighter in weight and shorter in length than yellow perch from other locations. Also, largemouth bass from Reach 3 were heavier and older than largemouth bass from other locations. Simple statistics generally indicate slower growth rates for both species in Reaches 3 and 4 as compared to less contaminated areas.

These relationships were not statistically analyzed for significance.

Length, weight, and age data were not collected during Phase I. Phase II samples consisted only of filet portions. There was some concern that the integrity of some Phase II fish samples might have been compromised after collection and before processing in the laboratory. Some decomposition of sampled fish was observed, and these samples might not have been treated properly to ensure acceptable sample integrity. Contaminants could leach from decomposing muscle tissue (filet), resulting in erroneous muscle concentrations of some contaminants. However, the primary biota

- W92194F

location.

~~

#### TABLE 7-12

- •

PHASE II	SPECIES-S	PECIFIC AND WEIGHT, AN NYANZA OPE DLESEX COUN	LOCATION- ND AGE DAT RABLE UNIT	-SPECIFIC FIS A I 3 CHUSETTS	H LENGTH,
LARGEMOUTH BASS	MEAN LENGTH (mm)	LENGTH PER YEAR OF AGE (mm)	MEAN WEIGHT (g)	WEIGHT PER YEAR OF AGE (g)	MEAN AGE (yr)
BACKGROUND	266	89	366	122	3
REACH 2	325	81	5 <b>68</b>	142	4
REACH 3	340	68	693	139	5
REACH 4	299	75	418	104	4
REACH 6	313	78	5 <b>3</b> 3	133	4
REACH 9	341	85	723	181	4
YELLOW	-				
BACKGROUND	NS	NS	NS	NS	NS
REACH 2	233	58	1 <b>8</b> 6	46	4
REACH 3	205	51	95	23	4
REACH 4	246	41	169	28	6
REACH 6	246	49	188	38	5
REACH 9	247	49	203	40	5

# NS: NOT SAMPLED

LENGTH, WEIGHT, AND AGE DATA UNAVAILABLE FOR BULLHEADS

____

COCs identified should sorb strongly to organic matter, probably decreasing the potential for contaminant leaching from filet samples.

#### 7.4.6.2 Food Chain Exposure Assessment

A computer-based food chain model was used to estimate body burdens of biota COCs in organisms occupying various trophic levels within the aquatic system. The following paragraphs describe this model.

Because not all trophic levels in a food chain or all organisms occupying a given trophic level in a food chain can be analyzed for tissue chemical concentrations, food chain models have been developed to estimate concentrations of COCs in specific food chain organisms. These models should reasonably predict the biotic concentrations of COCs, given adequate quality and quantity of input data.

One limiting assumption of most food chain models is that the food chain being evaluated is phytoplankton driven; however, many food chains are detritus driven and the Sudbury River is likely to be a detritus-based system. Estimates of tissue concentrations of organisms of a detritus-based food chain can be expected to have greater uncertainty in comparison to estimates based on organisms in a phytoplankton-based food chain. In spite of this limitation, the food chain models that have been the subject of most reviews and are the most accepted are phytoplankton driven. Although cautious interpretation of results is warranted when such models are applied to detrital systems, these models are assumed to provide reasonable estimates of tissue concentrations of detritalbased food chain organisms. This assumption can be validated with actual measured values of tissue chemical concentrations.

The food chain model selected for this assessment is the Bioaccumulation Model of Organic Chemical Distribution in Aquatic Food Chains, developed by R.V. Thomann (1989). This model is based generally on the relationship of chemical octanol/water partitioning coefficients (Kow) and uptake efficiency from water, excretion rate, and chemical assimilation.

As discussed in Section 5.0, bioaccumulation considers chemical uptake from both food and water, while bioconcentration considers uptake from water only. Bioconcentration factors (BCFs) generally approximate bioaccumulation factors (BAFs) for chemicals that do not biomagnify. For chemicals that biomagnify significantly in food chains/webs, such as mercury and DDT, chemical uptake is best expressed in terms of bioaccumulation rather than bioconcentration because food ingestion is considered to be the primary uptake process. However, BAFs are generally unavailable for most chemicals and most organisms, and BCFs are often used as surrogate

W92194F
BAFs for estimating chemical uptake by aquatic organisms (EPA, 1985h).

The outputs of this model include log BAFs for three trophic levels: Levels 2, 3, and 4. Level 1 is represented by phytoplankton, and at Kow values as high as approximately 6, phytoplankton BAF equals chemical Kow. Level 2 is represented by zooplankton, Level 3 by small fish, and Level 4 by large fish (top aquatic predator). Thomann's use of the terms "small fish" and "large fish" has been retained; however, these terms are considered synonymous with "zooplankton consumers" and "piscivorous fish," respectively. For the purposes of this risk assessment, the estimated log BAFs for Levels 2, 3, and 4 are the critical outputs from this model. Input variables for the model include chemical Kow, organism wet weight, and percent lipids in selected organisms. The model appears to be more sensitive to changes in the lipid content in comparison to the weight of organisms used to represent various trophic levels.

Because this model is based on chemical Kow, Kow must be accurately determined for the COCs. Kow values are readily available for most organic contaminants; however, Kows are not generally applicable to metals and, therefore, there is no readily available Kow for However, because methylmercury is an organic form of mercurv. estimations are possible based Kow mercury, on Kow/BCF This approach is based on generally accepted relationships. methods and equations, and is expected to reasonably estimate the partitioning behavior of methylmercury for the purpose intended. Using the equation of Veith (Lyman, Reehl, and Rosenblatt, 1990), the relationship between log Kow and log BCF is described as follows:

 $\log BCF = 0.76 \log Kow - 0.23$ 

Solving for log Kow results in the following equation:

Freshwater whole-body bioconcentration factors for methylmercury range from 11,000 to 85,700 (n=6, EPA, 1984g). Log BCFs for these same values range from 4.04 to 4.93. The calculated geometric mean of the log BCF for methylmercury is 4.54. Using the rearranged equation of Veith, the log Kow for methylmercury is 6.28 (rounded to 6.3), which was the primary input variable to the food chain model selected.

Log Kows for PCBs differ for each specific PCB species. In this study, two PCBs, Aroclor 1254 and Aroclor 1260, dominated fish

W92194F

7-42

tissue PCB concentrations. The following table lists the log Kows used in this risk assessment for the two dominant Aroclors, as well as for Chlordane, DDT and its metabolites (DDD, DDE), methylmercury, and dieldrin:

	Reported		Selected
<u>Chemical</u>	Log Kow	Reference	Log Kow
Aroclor 1254	6.03	U.S. EPA, 1979 [•]	6.5
	6.5	Thomann, 1989	
Aroclor 1260	7.14	U.S. EPA, 1979 [•]	6.5
	6.5	Thomann, 1989	
Chlordane	2.78	U.S. EPA, 1979	6.0
	6.0	Thomann, 1989	
DDT	4.88	U.S. EPA, 1979	6.0
	6.0	Thomann, 1989	
DDD	6.04	U.S. EPA, 1979	6.0
	6.0	Thomann, 1979	
DDE	5.74	U.S. EPA, 1979	6.0
	5.7	Thomann, 1989	
Methylmercury	6.28	estimated, see above	6.3
Dieldrin	5.48	Thomann, 1989	5.5

# * mean of reported values

The variability associated with log Kows for most COCs resulted in the decision to use Thomann's values, which are based on recently published values from a variety of sources. With the exception of Chlordane, Thomann's values are similar to values reported by EPA (1979). Reported log BCFs for chlordane (5.98-7.24; Thomann, 1989) suggest that the 1979 Kow value of 2.78 might be too low and insuffic: intly conservative; therefore, log Kow was set at 6.0 for Chlordane. The log Kow was set at 6.5 for total PCBs, 6.0 for DDT and its metabolites, and 5.5 for dieldrin.

Values for the other input variables (percent lipids and organism wet weight) were specific for organisms representing trophic Levels 1 through 4. Levels 1 and 2 input values remained constant, while weight values for Levels 3 and 4 varied within the ranges indicated below. The values of input variables used in this assessment are listed below:

Level	1	(phytoplankton):	Wet weight (g) = Percent lipid =	N/A 10
Level	2	(zooplankton):	Wet weight (g) = Percent lipid =	0.10 10

FINAL

Level	3	(small	fish):	Wet weight (g) = Percent lipid =	10 and 100 4.5
Level	4	(large	fish):	Wet weight (g) = Percent lipid =	500 and 1500 4.5

Reported mean percent lipids for fish species ranged from 3.6 (black bullhead) to 4.4 (yellow perch) to approximately 5.5 (freshwater bass, not <u>Micropterus</u>) (Sidwell, 1981). The range associated with the reported mean percent lipid values for the species overlapped considerably, three fish suggesting no significant differences in average percent lipids for these Species-specific values for percent lipids were species. unavailable for largemouth bass. Because of the uncertainty associated with them, the three mean values were averaged for all sampled fish species, resulting in a mean lipid value of 4.5 percent. Because of the extensive overlap in the range of percent lipids for sampled species, intraspecific differences in percent lipids are expected to be as much as the reported interspecific difference. Therefore, the use of a single value for all sampled fish species should not measurably impact the results of the food chain model.

The trophic levels described above (Levels 1-4) are considered to be representative of site biota. Level 1 is assumed to represent detritus as well as phytoplankton. Level 2 is considered to be representative of small aquatic invertebrates (e.g., zooplankton and caddis fly larvae) that might be consumed by Level 3 organisms, which are typically fish weighing 10 to 100g. Finally, Level 4 organisms are considered to be top-level aquatic predators, including large individuals of yellow perch and largemouth bass weighing from 500 to 1,500g. Variability fish weights did not produce measurable differences in projected BAFs modeled; therefore, the fact that the actual weight few largemouth bass exceeded the maximum modeled weight shou. It affect the model results.

Based on the results of the food chain model, Table 7-13 lists predicted BAFs for Levels 1 through 4, cumulative BAFs, and input parameters for biota COCs. Because the model is based predominantly on log Kow, and because the estimated log Kow for both groups of chemicals are the same (6.0), predicted BCFs for total DDT and total chlordane are equal. The results of this model suggest that log Kow provides a reasonable estimate of log BAF between any two trophic levels. For example, log BAFs for mercury (log Kow = 6.3) estimated between any two trophic levels range from 6.3 to 7.2. Similar results are predicted for PCBs, DDT, dieldrin, and chlordane, where all BAFs were estimated within the range of 5.5 to 7.3, while log Kows ranged from 5.5 to 6.5.

		<b></b>	NYANZ MIDDLESEX C	ZA OPERABLE OUNTY, MASS	UNIT 3 ACHUSETTS		
				BIOACCUMULA	TION FACTORS		
TROPHIC LEVEL	PERCENT LIPID	WEIGHT (g)	MERCURY (2)	PCB (3)	DDT (4)	CHLORDANE (4)	DIELDRIN (5)
1	10	NA	6.3	6.5	6.0	6.0	5.5
2	10	0.1	6.7	6.8	6.4	6.4	6.1
3	4.5	10 - 100	6.9	7.0	6.4	6.4	5.9
4	4.5	500 - 1500	7.2	7.3	6.5	6.5	5.9
CUMULATIVE BAF LEVELS 1-4			2097	2259	1597	1597	1168

(1) PREDICTIONS BASED ON RESULTS OF THOMANN'S MODEL, 1989
(2) LOG KOW = 6.3 (ESTIMATED)
(3) LOG KOW = 6.5 (AVERAGE OF REPORTED VALUES)
(4) LOG KOW = 6.0 (AVERAGE OF REPORTED VALUES)
(5) LOG KOW = 5.5 (THOMANN, 1989)
NA: NOT APPLICABLE

**TABLE 7-13** 

PREDICTED BIOACCUMULATION FACTORS (BAFS) FOR FOOD CHAIN ORGANISMS (1)

Predicted cumulative AFs, expressing uptake from the source (surface water or i erstitial water) to trophic level 4 (top aquatic predators) range from 1,168 (dieldrin) to 2,259 (total PCBs). Cumulative values for total mercury (2,097), total DDT (1,597), and total chlordane (1,597) fall within this range. These predicted cumulative BAF values are evaluated in the risk characterization section of this assessment.

# 7.4.7 Uncertainty Analysis, Exposure Assessment

The following paragraphs discuss some general areas of potential concern that are expected to contribute to exposure uncertainty. These statements are a means to identify the primary sources of potential uncertainty in the exposure assessment; they describe the methods used to decrease uncertainty that might be associated with the specific area of concern.

1. Sufficient quantities of environmental samples should be collected using appropriate sampling designs; the samples should be properly transported, and accurately and precisely analyzed.

Sample collection and transport methods generally followed accepted protocols. Possible deviations from standard protocols used for storage and transport of fish samples (Phase II) might have compromised some fish data, but the impacts of the deviations should be minimal for the COCs identified. In-laboratory quality control (QC) and quality assurance (QA) programs are expected to limit uncertainty associated with sample analysis. In general, sufficient quantities of both chemical and biological samples were collected and analyzed for use in this assessment. Additional samples of caddis fly larvae would have produced an increased volume of pooled samples, thereby improving the analytical recovery of contaminants.

2. Methods selected for analyzing environmental samples should be appropriate to the analyte and sufficiently sensitive and specific for the needs of the assessment.

Under the contract laboratory program (CLP), analytical laboratories must use EPA-recommended methods to analyze specific chemicals or groups of chemicals. All analyses in this assessment were performed using appropriate methods without deviation. Detection limits (DLs) and other QC results obtained in the analytical laboratory generally met predefined criteria. If QC criteria were not met, such data were appropriately qualified and considered to be conditionally acceptable or

W92194F

unacceptable, depending on the type of qualifier. Unacceptable environmental samples or environmental samples associated with unacceptable QC specimens were not used for this assessment.

3. Appropriate and reasonable assumptions should be made concerning exposure pathways (sources, points, routes).

All major pathways were considered for the ecological resources potentially at risk. Minor pathways were not considered if they were associated with insignificant contributions to overall exposure. For example, organisms occupying the upper trophic levels of the food chains examined (e.g., osprey) were assumed to obtain a relatively small amount of potentially hazardous chemicals through the water ingestion pathway in comparison to the ingestion of contaminated prey items. Therefore, the water ingestion pathway was not considered in this case.

4. Selection of COCs for each media should not eliminate potentially hazardous chemicals from consideration for full assessment.

Chemicals that were eliminated from further consideration included those that were relatively less toxic than retained chemicals, those that occurred in concentrations considered to be not toxic, and those that were detected in less than 5 percent of media-specific samples.

5. Appropriate and accepted sources of physical and chemical data should be used for evaluating fate and transport of COCs.

Physical and chemical data used in this assessment were obtained primarily from the U.S. Environmental Protection Agency (EPA), National Oceanic and Atmospheric Administration (NOAA), and the U.S. Fish and Wildlife Service (FWS). Other sources included peer-reviewed journals and books in the specific subject areas.

6. Relationships between COC concentrations in viscera and filet samples should not overestimate or underestimate actual values.

Filet samples were not correlated to viscera samples for any single fish. However, the composited viscera samples and individual filet samples offer sufficient information to establish reasonable relationships. Greater uncertainty is associated with relationships based on

W92194F

derived rather than measured values, but the consistent application of such relationships should not affect comparisons of relative differences in body burdens between species or between locations.

Sources of uncertainty in the exposure assessment fall into three major divisions: (1) analytical, (2) sampling, and (3) assumptions. The estimate of analytical uncertainty was relatively low due to stringent QA and QC procedures in CLP laboratories. In contrast, uncertainty associated with sampling procedures and especially sampling design, was expected to contribute significantly to overall uncertainty. This assumption was based on the fact that chemicals are not evenly distributed throughout any environmental study area, and even the best sampling design might not result in accurately describe the data that actual environmental concentrations or distributions of contaminants. Probably the greatest levels of uncertainty in the exposure assessment arise the uncertainty associated with exposure assumptions. from Assumptions are usually "best guess estimates," and certain levels of uncertainty are inherent in any assumption.

While potentially important areas of concern regarding exposure uncertainty have been identified, all efforts were made to limit the uncertainty as much as possible. For example, exposure assumptions were based on reasonable and accepted exposure scenarios where possible. If guidance was unavailable, reasonable assumptions describing "upper-bound likely" conditions were made. Analytical laboratories that were considered appropriate for the tasks assigned were used throughout the risk assessment process, ensuring that data quality objectives would be met. The sampling design, even though constrained by time and cost, was developed to provide site characterization that was as accurate as possible with respect to contaminant concentrations and distributions.

# 7.5 <u>Hazard Assessment</u>

Hazard assessment weighs evidence about the potential for COCs to cause adverse effects to exposed populations. Commonly assessed adverse effects include toxic effects (toxicity assessment) and effects (effects due to bioconcentration/ food chain In addition, hazard assessment biomagnification of chemicals). attempts to estimate the relationship between the duration and frequency of exposure to a chemical and the increased likelihood or severity of adverse effects (EPA, 1989a).

Duration of exposure is commonly expressed in terms of either acute or chronic toxicity. Acute exposures are relatively short-term duration exposures, often resulting in lethal or serious adverse effects to exposed organisms. Chronic exposures are considered long-term exposures, resulting in lethal or sublethal adverse

W92194F

effects to exposed organisms. Sublethal adverse effects include decreased growth and reproduction, teratogenic effects, and ecologically significant behavioral effects. No clear distinction exists between acute and chronic exposures in most cases, and exposure duration should be clearly defined. While most environmental exposures are considered to be chronic, exposures such as oil spills or runoff from the season's first rain event can pose an acute hazard to biota. Duration of exposure can be based on spatial and temporal components of a contaminated site or may be based on the behavior of affected organisms. For example, a "hot spot" of contaminated sediments can pose a continuous hazard to sessile benthic organisms. In contrast, mobile organisms are more likely to be exposed intermittently because contact is unlikely to be continuous, or because they might avoid the site.

Although evaluations of exposure and toxicity of <u>individual</u> chemicals are most often performed in risk assessments, chemicals rarely exist individually in the environment. In general, the toxicity of contaminant mixtures and the interactions of the contaminants making up the mixtures are poorly understood. Toxicant interactions can include synergism, antagonism, and additivity. In general, additivity best describes the type of interaction expected in toxicant mixtures, and it is assumed to be the predominant interaction in this risk assessment.

# Benchmark Toxicity Values

For comparing estimated exposure concentrations to potentially hazardous concentrations of COCs, benchmark toxicity values were established. Benchmark toxicity values are defined as concentrations of media-specific COCs that are known or likely to be hazardous to species of concern. Such concentrations are intended for the protection of sensitive species from sublethal effects. For this reason, benchmark toxicity values are most appropriately based on data derived from chronic toxicity tests utilizing sublethal endpoints, such as chronic ambient water quality criteria.

Actual COC exposures to study area organisms are assumed to be primarily long-term exposures, usually at sublethal concentrations. However, chronic toxicity data are not uniformly available and, in some cases, chronic values were estimated from acute toxicity data. The acute toxicity database is much larger than the chronic toxicity database for most chemicals and most organisms; if acute toxicity data were the only data available, they were used for estimating critical chronic toxicity values. Appropriate chronic toxicity values include chronic ambient water quality criteria (CAWQC), lowest observed effect concentration (LOEC), or no observed effect concentration (NOEC) values. The preferred chronic toxicity value for surface water evaluations is the CAWQC value.

W92194F

However, CAWQC values are unavailable for many chemicals, and toxicity values such as the lowest median lethal concentration, 50 percent (LC50)/100 or LOEC/10 were used as surrogate values.

Table 7-14 lists specific benchmark toxicity values used, and references the individual values.

The use of LC50/100 is based on laboratory data used to derive ambient water quality criteria for a variety of inorganic and organic chemicals. These data suggest that an LC50 for a standard sensitive test species divided by 100 provides a reasonable and adequate level of protection for sensitive untested species. Similarly, the use of LOEC/10 provides an estimate of NOEC, and LOEC/10 was selected as an appropriate chronic benchmark toxicity value for some chemicals.

Assumptions about the relationships between LOEC and NOEC values can be summarized as follows:

## LOEC > NOEC and LOEC/10 approximates NOEC

The relationship between LOEC/10 and NOEC is based on a conservative approach. The NOEC is probably estimated more accurately by LOEC/2 than by LOEC/10, based on experimentally derived toxicity data. For example, results of chronic toxicity tests with hexavalent chromium and rainbow trout (EPA, 1985c) revealed an LOEC of 105 ug/L and an NOEC of 51 ug/L. In another example, in which sheepshead minnows were tested with acenaphthene (EPA, 1980a), the LOEC and NOEC equalled 970 and 520 ug/L, respectively, supporting the general LOEC/2 approximation of NOEC relationship. Because chronic data are limited for most chemicals and most species, a conservative approach was considered to be appropriate for the protection of untested species that might not follo. the general relationships illustrated by these examples; therefore, LOEC/10 was used to estimate NOEC conservatively.

## Data Sources

Sources of toxicity data used in this risk assessment include EPA Ambient Water Quality Criteria Documents (EPA, 1980-1988), EPA AQUIRE database for aquatic organisms (ACQUIRE, 1990), Syracuse Research Corporation's Chemfate database (SRC, 1990), U.S. Fish and Wildlife Service Hazard Review Documents (FWS, 1986-1990), and recent peer-reviewed scientific literature. The determination of acceptable versus unacceptable data Was based on the appropriateness of endpoints, test design, and test species. EPA national quidelines contain appropriate criteria for data used in aquatic evaluations (EPA, 1985h).

#### 7.5.1 Benchmark Toxicity Values - Surface Water COCs

This assessment evaluated surface water COCs for potential impacts to freshwater aquatic animals, freshwater aquatic plants, and avian/terrestrial organisms. Benchmark toxicity values were derived for each group of organisms based on chronic effects data from the most sensitive tested species. For freshwater aquatic animals, the benchmark toxicity values for all surface water COCs, except Bis-(2-ethylhexyl)phthalate, were based on chronic ambient water quality criteria, adjusted to a water hardness of 25 mg  $CaCO_3/L$  where appropriate. Because a CAWQC value has not been derived for Bis-(2-ethylhexyl)phthalate, a surrogate value was used. The surrogate value was based on the lowest of all reported lowest observed effect concentrations for freshwater fish and invertebrates divided by a safety factor of 10 (LOEC/10). LOEC/10 estimates the no observed effect concentration (NOEL).

Although plant toxicity data are sparse compared to toxicity data on animals, sufficient data exist for some chemicals to predict the potential impacts associated with contamination. In general, plant toxicity data consist of concentrations estimated to produce significant growth reduction or mortality. Potential toxicity of surface water COCs posed to freshwater aquatic plants were evaluated by establishing the lowest concentration affecting 50 percent of tested organisms (effect concentration, 50 percent or EC50). For aquatic plants such as algae, EC50s are available for some chemicals. Because such data were unavailable for some surface water COCs, LOEC/10 was used to estimate NOEC.

Avian and terrestrial animals are exposed to surface waters primarily through drinking. Concentrations of surface water COCs that are considered safe for ingestion by birds or other wildlife have not been established; therefore, ambient water quality criteria for the protection of human health, adjusted for drinking water only, were used as benchmark toxicity values for avian and terrestrial biota. The use of values intended for the protection of humans is associated with uncertainty; however, this uncertainty is expected to primarily overestimate rather than underestimate the toxicity of surface water COCs.

Table 7-14 lists the benchmark toxicity values for the COCs for aquatic animals, aquatic plants, and avian and terrestrial biota based on the ingestion of surface water.

#### 7.5.2 Benchmark Toxicity Values - Sediment COCs

This assessment used a separate approach for estimating exposure concentrations of inorganic and organics sediment COCs. The

# **TABLE 7–14**

# BENCHMARK TOXICITY VALUES FOR SURFACE WATER CHEMICALS OF CONCERN NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

	FRESHWATER AQUATIC ANIMAL	FRESHWAT AQUATIC PLANT	Γ <b>ΕR</b> 	AVIAN/ RRESTRIAL ANIMAL	•	
	TOXICITY VALUE	TOXICITY VALUE		TOXICITY VALUE		
	ug/L RE	F* ug/L	REF*	ug/L	REF*	
A! '	87	1 4.6	3	_	5	
1	<b>11</b> 1	1 0.2	4	50	6	
UC.	<b>3.6</b> 1	1 0.1	4	1.0	6	
LEAD	0.5 1	1 50	4	50	6	
MERCURY (INORGANIC)	<b>0.2</b> 1	1 0.5	4	10	6	
NICKEL	49 1	l 0.5	4	15.4	6	
SILVER	0.1 1	1 3	4	50	6	
ZINC	33 1	I 3	4	5	6	
BIS-(2-ETHYLHEXYL) PHTHALATE	0.3 2	2 856	3	21000	6	

* REFERENCES

- 1: CHRONIC AMBIENT WATER QUALITY CRITERIA (EPA, 1980–88) ADJUSTED FOR HARDNESS = 25 mg CaCO3 (Cu, Pb, Ni, Zn)
- 2: LOWEST OBSERVED EFFECT CONCENTRATION (LOEC) FOR INVERTEBRATES OR FISH / 10 (EPA, 1980-1988) LOEC/10 = ESTIMATED NO OBSERVED EFFECT CONCENTRATION (NOEC)
- 3: LOWEST EFFECT CONCENTRATION, 50% (EC50) FOR AQUATIC PLANTS / 100 (EPA, 1980-1988) EC50/100 = ESTIMATED NO OBSERVED EFFECT CONCENTRATION (NOEC)
- 4: LOWEST LOEC FOR AQUATIC PLANTS / 10 (EPA, 1980-1988) LOEC/10 = ESTIMATED NO OBSERVED EFFECT CONCENTRATION (NOEC)
- 5: TOXICITY DATA FOR WATER INGESTION FOR TERRESTRIAL/AVIAN SPECIES UNAVAILABLE
- 6: AMBIENT WATER QUALITY CRITERIA FOR THE PROTECTION OF HUMAN HEALTH (EPA, 1986d) ADJUSTED FOR DRINKING WATER ONLY

separate exposure approaches required distinct methods for establishing benchmark toxicity values for these COCs.

For inorganic COCs in sediment, biological effects data, based on ug chemical/kg sediment, dry weight, were used to establish benchmark toxicity values. At this ti-e, sediment quality criteria values have been established for only a few organic chemicals. Therefore, sediment COC concentrations cannot be compared directly to criteria values. In a manner similar to that used for the selection of sediment COCs, biological effects data from Long and Morgan (1989) were used as surrogate criteria values for sediment COCs.

Long and Morgan derived toxicant concentrations in sediments that are associated with observed adverse biological effects, and sorted, in ascending order, the concentrations in sediment of a specific toxicant associated with observed adverse biological effects. The 10th and 50th percentile concentrations, described as the Effects Range-Low and Effects Range-Median, respectively, were derived for each chemical evaluated. ER-L values represent concentrations of a chemical in sediment (dry weight) that is equivalent to the lower 10th percentile of the screened available data. Similarly, ER-M values are equivalent to the median (50th percentile) of the screened available data.

Because both ER-L and ER-M values were expected to provide valuable information about the range of potentially toxic concentrations, both values were selected as benchmark toxicity values for inorganic sediment COCs. ER-L values were unavailable for two inorganic sediment COCs, aluminum and selenium. For these two COCs, surrogate values were used as benchmark toxicity values. To derive the surrogate values, the relationship between CAWQC and ER-L values was investigated. The median value of the ratio of CAWQC/ER-L computed for all chemicals for which ER-L values and CAWQC are available equals approximately 0.0005.

That is, dividing the chemical-specific CAWQC value by 0.0005 provides a reasonable estimate of ER-L for that chemical. Similarly, for aluminum and selenium, the estimated ER-M value was based on the relationship between ER-L and ER-M for all chemicals for which both values have been derived. This relationship can be described generally as

# ER-M is approximately equal to ER-L * 3

This relationship was used to estimate ER-M values for both aluminum and selenium.

Organic COCs in sediment were partitioned to interstitial water using equilibrium partitioning in the exposure assessment.

- W92194F

Concentrations .f sediment COCs in interstitial water are most appropriately .ompared to aqueous toxicity values. CAWQC or surrogate values were selected to represent the aqueous benchmark toxicity values for organic sediment COCs. In contrast to what was done with inorganic COCs in sediment, only a single benchmark toxicity value was established for each chemical. Surrogate values were used for organic COCs in sediment for which CAWQC have not been derived; these values include final chronic values (FCVs), LC50/100, and EC50/10.

Toxicity data are unavailable for an assessment of potential adverse effects of contaminated sediments on terrestrial or avian species. The potential for contaminated sediments to produce toxic adverse effects on terrestrial and avian species is considered minimal compared to potential food chain effects on terrestrial and avian species due to biomagnifiable COCs in sediment. Section 7.5.3 describes potential food chain effects.

Table 7-15 lists ER-L and ER-M values for inorganic sediment COCs and CAWQC (and surrogate values) for organic COCs in sediment.

## 7.5.3 Benchmark Toxicity Values - Biota COCs

The hazard assessment component of an ecological risk assessment is commonly referred to as a "toxicity assessment". Although this terminology is appropriate for many assessments, certain situations exist for which the term "hazard assessment" is more appropriate. This risk assessment considers the risk of biomagnification of certain COCs (e.g., mercury, DDT, chlordane) in food chains to be a critical component requiring assessment beyond that associated with acute or chronic toxicity. Therefore, the selection of biota COCs was based on bioconcentration potential, which is directly related to biomagnification, rather than on toxicity.

Chemicals that tend to biomagnify within food chains can produce severe adverse effects in top-level predators, while organisms occupying lower trophic levels exhibit little or no adverse effects even though their tissues contain measurable chemical residues. Clearly, potential risks associated with biomagnifiable COCs are best evaluated by considering both toxicity and bioconcentration/biomagnification potential.

Benchmark toxicity values for biota COCs were based on the maximum permissible tissue concentration (MPTC) for fish. The MPTC is the most sensitive (lowest value) of either the Food and Drug Administration (FDA) Action Level or the lowest value derived from an appropriate dietary intake study (EPA, 1985h). Although FDA Action Levels are not applicable to criteria which are protective of the ecology, the number is utilized as a reference point due to the lack of other relevant data. MPTC values are expressed as

W92194F

7-54

FINAL

# **TABLE 7-15**

	- <u> </u>	MIDE	DLESEX CO	UNTY, I	MASSACHUSETTS		
INORGANIC COC	ER-L VALUE		ER-M VALUE		ORGANIC COC	TOXICITY VALUE	
IN SEDIMENT	ug/kg	REF*	ug/kg	REF*	IN SEDIMENT	ug/L	REF*
ALUMINUM	174000	1	522000	3	4.4'-DDD	0.001	5
ARSENIC (III)	33000	2	85000	4	4.4'-DDE	0.001	5
CADMIUM	5000	2	9000	4	4.4'-DDT	0.001	5
CHROMIUM (VI)	80000	2	145000	4	ACENAPHTHENE	23	6
COPPER	70000	2	390000	4	ACENAPHTHYLENE	15	7
LEAD	35000	2	110000	4	ANTHRACENE	15	7
MERCURY (II)	150	2	1300	4	BENZO(A)ANTHRACENE	10	8
NICKEL	30000	2	50000	4	BENZO(A)PYRENE	10	9
SELENIUM	6000	1	18000	3	BENZO (B) FLUORANTHENE	15	7
ZINC	120000	2	270000	4	BENZO(G,H,I)PERYLENE	15	7
					BENZO(K)FLUORANTHENE	. 15	7
					BIS(2-EH)PHTHALATE	0.3	10
					CHRYSENE	10	9
					DIBENZOFURAN	15	7
					DIBENZO(A,H)ANTHRACEN	10	7
					DI-N-BUTYLPHTHALATE	0.3	11
					FLUORANTHENE	8.1	6
					FLUORENE	3.2	9
					INDENO(1,2,3-CD)PYRENE	15	7
					METHYLMERCURY	0.012	5
					NAPHTHALENE	20	9
					PHENANTHRENE	6.3	6
					PYRENE	15	7

## BENCHMARK TOXICITY VALUES FOR SEDIMENT CHEMICALS OF CONCERN NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY MASSACHUSETTS

INORGANIC TOXICITY VALUES: ug/kg sediment, dry wt. ORGANIC TOXICITY VALUES: ug/L surface or interstitial water

# * REFERENCES

1: (CAWQC / 0.0005) (EPA 1980-88)

0.0005 = MEDIAN CAWQC/ER-L FOR ALL CHEMICALS WITH BOTH CAWQC AND ER-L

- 2: EFFECTS RANGE-LOW (ER-L) (LONG AND MORGAN, 1989)
- 3: ESTIMATED ER-L * 3 (LONG AND MORGAN. 1989)
  - 3 = APPROXIMATE MEAN RATIO OF ER-M/ER-L VALUES FOR INORGANIC CHEMICALS
- 4: EFFECTS RANGE-MEDIAN (ER-M) (LONG AND MORGAN, 1989)
- 5: CAWQC (EPA, 1980-1988)
- 6: FINAL CHRONIC VALUE (FCV) (EPA. 1991d.e.f)
- 7: LOWEST ACUTE 96H LC50 FOR PAHS / 100 (USFWS, 1987b)
- 8: LC87/100 (USFWS, 1987b)
- 9: LOWEST ACUTE 96H LC50 / 100 (USFWS, 1987b)
- 10: LOWEST CHRONIC EC50/10 (EPA, 1980i)

11: LOWEST CHRONIC EC50 FOR PHTHALATE ESTERS / 10 (EPA, 1980i)

ug/kg fish tissue, wet weight. The endpoint of appropriate dietary intake studies can vary with each study, but in general any ecologically significant endpoint is considered to be critical. Such endpoints can include adverse reproductive effects or decreased survival of sensitive species. MPTC values are generally available for classes of chemicals, and a single MPTC value can represent the MPTC for several closely related chemicals.

Therefore, to establish benchmark toxicity values for biota COCs, fish tissue concentrations of some biota COCs were combined. DDD and DDT were assessed as total DDT, Aroclor 1254 and 1260 were assessed as total PCBS, alpha and gamma chlordane were assessed as total chlordane, and inorganic mercury and methylmercury were assessed as total mercury. Table 7-16 lists the benchmark toxicity values for biota COCs.

# 7.5.4 Uncertainty Analysis, Hazard Assessment

Because complete toxicological databases do not exist for the vast majority of chemicals, there are many opportunities for uncertainty to impact the hazard assessment. Specific areas of concern that can indicate sources of potential uncertainty include the following:

1. Species tested should be representative of site species.

EPA (1985h) considers tested aquatic species used to derive water quality criteria to be representative of species found throughout the nation. Site-specific species are not expected to be more or less sensitive to contaminants than those comprising the national aquatic database.

2. Laboratory toxicity tests should be representative of the Study Area environment.

laboratory-to-field extrapolations In general, of toxicity data are considered to be acceptable under most conditions. Recent microcosm studies appear to confirm acceptability laboratory-to-field the general of extrapolations for toxicity testing. While laboratory conditions allow more control over test variables, field depict actual conditions more accurately. studies Unfortunately, acceptable field studies are rare for most organisms and most chemicals, and the majority of accepted toxicity data are based on laboratory studies.

3. Extrapolations (acute-to-chronic, species-to-species, chemical-to-chemical) should be reasonable and appropriate for the chemicals and species selected.

# **TABLE 7-16**

# BENCHMARK TOXICITY VALUES FOR BIOTA CHEMICALS OF CONCERN NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

BIOTA COC	ASSESSED	MPTC ug/kg wet wt.	REF*	ACTION LEVEL OR MOST SENSITIVE ECOLOGICALLY SIGNIFICANT EFFECT
4,4' – DDD 4,4' – DDT AROCLOR 1254	TOTAL DDT TOTAL DDT TOTAL PCB	150	1	REDUCED REPRODUCTION IN BROWN PELICANS
AROCLOR 1260 CHLORDANE (TOTAL) DIELDRIN METHYLMERCURY	TOTAL PCB TOTAL CHLORDANE DIELDRIN TOTAL MERCURY	640 300 300 100	2 3 4 5	REDUCED SURVIVAL AND REPRODUCTION IN MINK U.S. FDA GUIDELINE FOR HUMAN CONSUMPTION OF FISH AND SHELLFISH U.S. FDA GUIDELINE FOR HUMAN CONSUMPTION OF FISH AND SHELLFISH REPRODUCTIVE EFFECTS, SENSITIVE AVIAN SPECIES

MPTC: MAXIMUM PERMISSIBLE TISSUE CONCENTRATION, FISH BASED ON LOWEST OF FDA ACTION LEVEL OR VALUE FROM APPROPRIATE DIETARY INTAKE STUDY

* REFERENCES

1: EPA, 1980d

.....

- 2: EPA, 1980g
- 3: USFWS, 1990
- 4: EPA, 1980b
- 5: USFWS, 1987a

Direct testing with local species and site-specific COCs is the best approach for defining potential toxicities of COCs at a given site. However, such data are generally unavailable, and the use of appropriate extrapolations is a reasonable approach. Extrapolations from tested to untested species and chemicals should be based on reasonable, tested relationships between species or This document provides equations or other chemicals. expressing such relationships support for where appropriate.

Deriving appropriate toxicity values for species or chemicals for which toxicity data are limited often requires the use of interspecies correlations for species-to-species extrapolations and chemical QSARs for extrapolatior ... chemical-to-chemical Interspecies correlations (e.g., Doherty, 1983; Suter and Rosen, 1986) and quantitative structure-activity relationships (QSARs) (e.g., Borman, 1990; Protic and Sabljic, 1989), are commonly used to extrapolate toxicity data from species species or from chemical to chemical. Such to extrapolations were used in a limited manner for this assessment because toxicity data, although often limited, existed for most COCs. The major exception was the lack of toxicity data for some sediment COCs (e.g., some PAHs). For limited toxicity data, chemical relationships were utilized to estimate toxicity, or the toxicity of the most hazardous chemical in a class of chemicals (e.g., PAHs) was used to estimate toxicity for related nontested chemicals.

4. Single-point toxicity values such as LC50s, compared to dose/response relationships (e.g., describing the entire toxicity curve), might or might not adequately depict potential toxicity.

The description of an entire dose/response or concentration/response curve is potentially more such as an LC50. informative than a single value, However, such data are rarely reported; most often a single value such as an LC50; median lethal dose, 50 percent (LD50), or LOEC, is reported. Although this is considered to be a limitation, the use of single-point toxicity values is still expected to provide an adequate description of the relative toxicity between chemicals and between test species.

5. Accepted toxicity data should be based on standardized, accepted test protocols.

All aquatic toxicity data used throughout this assessment were based on tests that followed acceptable experimental designs, conditions, and endpoints as defined by EPA (1985h). Terrestrial and avian toxicity test results were considered to be acceptable if selected endpoints were ecologically significant and if toxicant dosing and exposure durations and frequencies followed accepted protocols.

The decision to accept or reject toxicity data for use in the ecological risk assessment was based on best professional judgment in some cases. The justification for accepting or rejecting data was deemed to be critical, and the use of appropriate safety factors and other extrapolation factors were intended to be reasonable and scientifically defensible. Best professional judgment was used only if specific guidance or accepted procedures were unavailable. For example, safety factors were applied to toxicity values using best professional judgment if no specific guidance was available.

Various ecological risk assessment methodologies have been developed to limit the variability and uncertainty associated with toxicological data; however, ecotoxicology is in its infancy in comparison to related disciplines. The number of species and chemicals tested is only a small fraction of species and chemicals that might be of concern. Although there is a continuing effort by aquatic toxicologists, ecotoxicologists, risk assessment regulatory agencies, specialists, and others to establish methodologies for performing ecological standardized risk assessments, this goal has not yet been achieved. There has been measurable success in standardizing toxicity test protocols and in increasing both the quality and quantity of the data for toxicity and exposure assessments. At this time, however, ecological risk assessments are generally based on limited available data and on extrapolations from measured to unmeasured species and chemicals. It is often necessary to draw on experience and best professional judgment, even though confidence and reliability in the resulting judgments can be less than desired. Therefore, ecological risk assessment methodologies will be associated with a degree of uncertainty for both exposure and hazard assessments for some time. In this assessment, efforts have been made to limit the level of uncertainty as much as possible by using appropriate toxicity values that were considered to be adequately protective of biota. If an uncertainty could not be quantified or limited, efforts were made to err on the side of conservatism, or overprotection. While EPA has stated that neither overprotection nor underprotection is desirable (EPA, 1985h), erring on the side of overprotection of biotic resources seems prudent.

W92194F

7-59

FINAL

# 7.6 <u>Risk Characterization</u>

Risk characterization relates exposure concentrations of COCs to concentrations of COCs that potentially cause adverse effects; it is based primarily on the integration of exposure and hazard assessments. This section compares the media-s[ecofoc and reachspecific exposures (representative organisms) from Site and Study Area contaminants and potentially toxic concentrations of these contaminants.

# 7.6.1 General Approach

The quotient method was selected to characterize risk for the toxicity portion of this ecological risk assessment. In a manner similar to that used in the initial selection of COCs, the average and the maximum estimated environmental concentrations for each media and each reach were divided by the appropriate benchmark toxicity value. The average and maximum exposure concentrations for each media-specific COC were based on the values presented in Section 7.4. Toxicity values used to derive the exposure/toxicity ratios followed the guidelines described in Section 7.5.

The quotient method is probably the most common and most accepted method used for risk characterization in ecological risk assessments. However, it is associated with inherent limitations. One primary limitation is that the quotient method is a "yes/no" method for relating toxicity to exposure. That is, it utilizes single values for exposure concentrations and toxicity values, and does not account for incremental or cumulative toxicity. This limitation is addressed specifically by summing the quotients for all COCs, including those that are less than 1.0. In general, an exposure/toxicity quotient less than 1.0 is considered to be associated with insignificant risk. However, the cumulative toxicity of several COCs with exposure/toxicity quotients that approach, but are less than, 1.0 can be significant; this issue is addressed in this section. Section 7.6.5 discusses other limitations of the quotient method.

Advantages of the quotient method, according to Barnthouse et al. (1986), include the following:

- 1. The quotient method is relatively easy to implement, is generally accepted, and can be applied to any data.
- 2. The quotient method is useful when a large number of chemicals must be screened.

The following sections present estimated risks to aquatic and terrestrial biota, based on media-specific mean and maximum exposure concentrations. The individual numeric risk values do not

represent risk probability values in themselves, but are representative of the relative probability of risk; that is, the greater the estimated environmental concentration/ toxicity value quotient (EEC/TV), the greater the probability of risk.

# 7.6.2 Cumulative Toxicity

Although risk assessments generally evaluate the potential for adverse ecological effects due to individual chemicals, they should also address potential hazards associated with the cumulative effects of toxicant mixtures. Bioassays are the preferred means to determine the toxicity of complex chemical mixtures or to identify the actual toxicity of samples for which analytical methodologies are unavailable. In general, however, site-specific bioassay data are unavailable; estimates of the hazards to site biota from complex chemical mixtures can be assessed through various methods designed to address cumulative risk.

The method used in this risk assessment to evaluate cumulative risk is based generally on the method of Barnthouse et al. (1986). This method, which is based on the principle of toxicant additivity, simply sums the individual EEC/TV quotients for a given exposure. Toxicant additivity best describes the majority of toxicant interactions, and this fact supports this methodology for estimating cumulative toxicity. Other types of interactions, including synergistic and antagonistic interactions, have been insufficiently documented for most of the chemical mixtures studied.

Because this method utilizes chronic ambient water quality criteria or surrogate values, its application is intended only for aquatic systems. At present, no method exists for estimating cumulative toxicity for terrestrial or avian biota. The toxicity associated with most sites results from one primary toxicant or a few critical toxicants. Therefore, in most cases, adverse ecological impacts should not result from cumulative low toxicities associated with many toxicants.

#### 7.6.3 Risk from Surface Water COCs

This assessment divided average and maximum concentrations of each surface water COC within each reach by benchmark toxicity values. Each individual COC-specific quotient resulting from these calculations is directly comparable to the other COC-specific quotients. Therefore, the relative toxicity associated with each COC in each reach can be assessed. In addition, individual quotients for all COCs in each reach were summed, and the cumulative risk due to all surface water COCs was calculated for each reach. Table 7-17 lists the results of risk calculations based on surface water COCs.

____ W92194F

The data listed in Table 7-17 suggest that overall risk posed to avian and terrestrial organisms that drink site surface waters is minimal. For all surface water COCs, the ingestion of site water poses little risk to terrestrial and avian life when compared to other exposure pathways; concentrations of COCs in surface waters are well below those expected to pose measurable risks to such biota.

Aluminum in surface waters appears to be the primary source of risk to aquatic plants in most reaches. For Reach 1, zinc and nickel pose the primary risk to aquatic plants. Over all reaches, the risks for aquatic plants that are associated with zinc are generally less than the risks associated with aluminum. For both of these COCs, risk to aquatic plants can be somewhat overestimated because of potential complexation with organic acids. Aluminum complexed to organic (fulvic and humic) acids can be analytically indistinguishable from bioavailable forms of dissolved aluminum. Therefore, risk estimates based on aluminum and, to a lesser extent, zinc can be overestimated for waters with significant organic acid concentrations, such as swamps, bogs, marshes, and wetlands.

Risks posed to aquatic animals vary significantly from reach to reach, with the exception of aluminum. In general, aluminum contributes similarly to overall risk in Reaches 2 through 7, and in the Outfall Creek (OC) and Cold Spring Brook (CSB) locations. Reaches 1 to 6 are associated with relatively low risk estimates compared to Reach 7, Eastern Wetlands (EW), OC, and CSB. The primary surface water COCs associated with these locations are silver (R7), lead (EW and OC), and bis(2-ethylyhexyl)phthalate (OC and CSB). The risk estimate for CSB, which is based entirely on bis(2-ethylyhexyl) phthalate, greatly exceeds all other risk estimates for surface water COCs.

# 7.6.4 Risk from Sediment COCs

This assessment divided average and maximum concentrations of each sediment COC in each reach by benchmark toxicity values. As in the surface water risk characterization, each individual COC-specific quotient is directly comparable to the other COC-specific quotients. Again, the relative toxicity associated with each COC in each reach can be assessed. Individual quotients for all sediment COCs in each reach were added together, and the cumulative risk from all sediment COCs was calculated for each reach.

W92194F

7-62

# TABLE 7-17

## RISK ESTIMATES - SURFACE WATER COC NYANZA OPERABLE UNIT 3 MIDDLESEX COUNTY, MASSACHUSETTS

				FRESHWAT			FRESHWAT	ER AQUA	TIC PLANT	AVIAN/TERF	RESTRIAL	
LOCATION	CHEMICAL	MEAN SW CONC (ug/L)	MAX SW CONC (Ug/L)	TOXICITY VALUE 1/ (ug/L)	MEAN RISK VALUE	MAX RISK VALUE	TOXICITY VALUE 1/ (ug/L)	MEAN RISK VALUE	MAX RISK VALUE	TOXICITY VALUE 1/ (ug/L)	MEAN RISK VALUE	MAX RISK VALUE
BACKGROUND	NICKEL	38	7	49	01	01	0.5	7.6	14.0	15.4	0.2	0.5
BACKGROUND	ZINC	1211	21.6	33	0.4	07	3	4.0	7.2	5	2.4	4.3
BACKGROUND	SUM RISK	····•	· ···············		0.4	0.8		11.6	21.2		2.7	4.8
R2	ALUMINUM	29 28	67 3	87	03	0.8	4.6	6.4	14.6	NA		
R2	SUM RISK			,	0.3	08		6.4	14.6		0.0	0.0
83		80 8	808	87	0.9	09	4.6	17.6	17.6	NA		
R3	ZINC	18.5	18.5	33	06	06	3	62	62	5	37	37
R3	BIS(2-EH)PHTH	NV	1	03		33	856		0.0	21000	•	0.0
R3	SUM RISK				1.5	4.8		23.7	23.7		3.7	3.7
R4	ALUMINUM	107	107	87	1.2	1.2	4.6	23.3	23.3	NA		
	LEAD	15	15	05	30	30	50	0.0	0.0	50	0.0	0.0
R4	ZINC	22 8	22 8	33	07	0.7	3	7.6	7.6	5	4.6	4.6
R4	SUM RISK				4.9	4.9		30.9	30.9		4.6	4.6
<b>R</b> 6	ALUMINUM	110 57	174	87	1.3	2.0	4.6	24.0	37.8	NA		
R6	SUM RISK	_		-	1.3	2.0		24.0	37.8		0.0	0.0
R7	ALUMINUM	72 34	155	87	0.8	18	4.6	15.7	33.7	NA		
R7	SILVER	3 86	8	01	38 8	80.0	3	1.3	2.7	50	0.1	0.2
R7	SUM RISK				39 6	81.6		17.0	36.4	·	0.1	0.2

7-63

## TABLE 7-17 RISK ESTIMATES - SURFACE WATER COC NYANZA OPERABLE UNIT 3

#### MIDDLESEX COUNTY, MASSACHUSETTS PAGE 2

				FRESHWAT		C ANIMAL	FRESHWAT	ER AQUA	TIC PLANT	AVIAN/TERF	RESTRIAL	ANIMAL
LOCATION	CHEMICAL	MEAN SW CONC (ug/L)	MAX SW CONC (ug/L)	TOXICITY VALUE 1/ (ug/L)	MEAN RISK VALUE	MAX RISK VALUE	TOXICITY VALUE 1/ (ug/L)	MEAN RISK VALUE	MAX RISK VALUE	TOXICITY VALUE 1/ (ug/L)	MEAN RISK VALUE	MAX RISK VALUE
EW	LEAD	68	8.7	0.5	13.6	17.4	50	0.1	0.2	50		
EW	MERCURY	0 43	0.49	0 2	2.2	2.5	0.5	0.9	1.0	10	0.0	0.0
EW	SUM RISK			<del></del>	15.8	19 9		1.0	1.2		0.0	0.0
ос	ALUMINUM	67	67	87	08	08	4.6	14.6	14.6	NA		
OC	LEAD	24	2.4	0.5	4.8	48	50	0.0	0.0	50		
OC	MERCURY	0.42	0.42	02	2.1	21	0.5	0.8	0.8	10	0.0	0.0
OC	ZINC	46 1	46.1	33	1.4	1.4	3	15.4	15.4	5	9.2	9.2
OC	BIS(2 - 🕀 I) PHTH	1	1	0. <b>3</b>	<b>3</b> 3	33	3	0.3	0.3	5,	0.2	0.2
OC	SUM RISK				12.4	12.4		31,2	31.2	'.	9.5	9.5
CSB	BIS(2-BH)PHTH	58	58	03	19 <b>3</b> .3	193.3	З	19.3	19.3	5	11.6	11.6
CSB	SUM RISK	1			193.3	193.3		19.3	19.3	· · · •	11.6	11.6
	1	1										

_____

.....

1/ BENCHMARK TOXICITY VALUE FROM TABLE 7-14

1

1

SAMPLED LOCATIONS: R-REACH, EW-EASTERN WETLANDS, OC-OUTFALL CREEK, CSB=COLD SPRING BROOK

1

1

NV: NO VALUE - MEAN EXCEEDS MAXIMUM NA: TOXICITY VALUE NOT AVAILABLE

1

The ingestion of highly contaminated sediments can pose a real hazard to organisms that routinely ingest significant amounts of sediment. Such large-scale ingestion, which is unlikely for most biota, is best evaluated using food chain modeling that accounts for bioaccumulation. Toxic effects to individual organisms associated with significant sediment ingestion has not been fully evaluated because supporting toxicity data are lacking. However, such risks are likely to be minimal because of the low probability of significant sediment ingestion by most terrestrial and avian biota. Herons and some duck species might be exceptions; they are considered to be among the organisms at significant risk due to the ingestion of contaminated sediments.

The risk estimates for inorganic sediment COCs listed in Table 7-18 indicate that aluminum is a primary inorganic sediment contaminant for nearly all locations, including Background areas. As stated above, aluminum concentrations that form the basis for the risk estimates might not be totally bioavailable, resulting in overestimations of risk.

Based on the similar risk values for aluminum over all reaches, including Reach 1 (background), much of the aluminum measured in sediments is assumed to be naturally occurring, complexed aluminum that is generally considered to be nontoxic.

However, the presence of elevated concentrations of aluminum in surface waters in most reaches suggests that sediment-source aluminum is releasing aluminum to surface waters in concentrations that can be toxic to aquatic life. The other inorganic sediment COCs associated with significantly higher risk estimates than the other COCs are mercury and, to a lesser degree, chromium, copper, Sediment chromium probably consists primarily of and lead. trivalent chromium (Section 4.0), which is considered to have lower bioavailability and toxicity than hexavalent chromium. Sediment copper and lead might or might not be bioavailable and toxic, depending on specific sediment conditions. In contrast to aluminum, mercury is considered potentially bioavailable because the uptake and methylation of inorganic mercury by benthic biota are likely, resulting in biomagnification in the food chain.

Table 7-19 lists risk estimates for organic COCs in sediment. The primary contributors to overall risk include DDT and its metabolites, bis(2-ethylhexyl)phthalate, di-n-butylphthalate, and methylmercury. Cumulative maximum risk estimates range from 0.1 (Reach 1) to 762.3 (Reach 2, primarily due to DDT).

W92194F

7-65

#### TABLE 7-18 RISK ESTIMATES - INORGANIC COCS IN SEDIMENT NYANZA III REMEDIAL INVESTIGATION MIDDLESEX COUNTY, MASSACHUSETTS

·····				EFFE	CTS RANGE-LO	w	EFFECT	S RANGE-MEDI	N
1	1	MEAN	MAX	ER-L	MEAN	MAX	ER-M	MEAN	MAX
		SED	SED	TOXICITY	ER-L	ER-L	TOXICITY	ER-M	ER-M
. 1		CONC	CONC	VALUE	RISK	RISK	VALUE	RISK	RISK
LOCATION	CHEMICAL		(ug/kg)	(ug/kg)	VALUE	VALUE	(ug/kg)	VALUE	VALUE
B*	ALUMINUM	9200910	16600000	174000	52.9	95.4	522000	17.6	31.8
В	ARSENIC	8740	21100	33000	0.3	0.6	85000	0.1	0.2
8	CHROMIUM	22400	55200	80000	0.3	0.7	145000	0.2	0.4
B	COPPER	73020	340400	7000	10.4	48.6	390000	0.2	0.9
В	LEAD	85090	248000	35000	2.4	7.1	110000	0.8	2.3
B	MERCURY	270	1590	150	1.8	10.6	1300	0.2	1.2
В	NICKEL	11350	51000	30000	0.4	1.7	50000	0.2	1.0
8	SELENIUM	1040	3100	6000	0.2	0.5	18000	0.1	0.2
В	ZINC	133500	62900	120000	1.1	0.5	270000	0.5	0.2
в	SUM RISK VALUES 1/				69.8	165.8		19.8	38.2
В	SUM RISK VALUES 2/			· · ·	4.8	19.0		1.2	4.1
R2	ALUMINUM	7608330	16800000	174000	43.7	96.6	522000	14.6	32. <b>2</b>
R2	ARSENIC	6950	14900	33000	0.2	0.5	85000	0.1	0.2
R2	CHROMIUM	34000	216000	80000	0.4	2.7	145000	0.2	1.5
R2	COPPER	40190	184000	7000	5.7	26.3	390000	0.1	0.5
R2	LEAD	58070	295000	35000	1.7	8.4	110000	0.5	2.7
R2	MERCURY	3810	30600	150	25.4	204.0	1300	2.9	23.5
R2	NICKEL	7470	19100	30000	0.2	0.6	50000	0.1	0.4
R2	ZINC	126440	330000	120000	1.1	2.8	270000	0.5	1.2
R2	SUM RISK VALUES 1/				78.5	341.8		19.1	62.1
P2	SUM RISK VALUES 2/				27.7	215.6		3.8	27.9
R3	ALUMINUM	12210000	21500000	174000	70.2	123.6	522000	23.4	41.2
R3	ARSENIC	7610	21500	33000	0.2	0.7	85000	0.1	0.3
R3	CADMIUM	4890	19900	5000	1.0	4.0	9000	0.5	2.2
R3	CHROMIUM	292620	2620000	80000	3.7	32.8	145000	2.0	18.1
R3	COPPER	184470	454000	7000	26.4	64.9	390000	0.5	1.2
R3	LEAD	137970	285000	35000	3.9	8.1	110000	1.3	2.6
R3	MERCURY	15960	54600	150	106,5	364.0	1300	12.3	42.0
R3	NICKEL	28430	88800	30000	0.9	3.0	50000	0.6	1.8
R3	SELENIUM	810	4000	6000	0.1	0.7	18000	0.0	0.2
<b>R3</b>	ZINC	177080	435000	120000	1.5	3.6	270000	0.7	1.6
R3	SUM RISK VALUES 1/				214.4	605.2		41.3	111.1
R3	SUM RISK VALUES 2/				115.3	409.5		16.2	65.1

# TABLE 7-18 (continued) RISK ESTIMATES - INORGANIC COCS IN SEDIMENT NYANZA III REMEDIAL INVESTIGATION MIDDLESEX COUNTY, MASSACHUSETTS

PAGE 2

÷

				E	FFECTS RANGE-LC	w I	EFFECT	S RANGE-MEDIA	N N
		MEAN	MAX	ER-L	MEAN	MAX	ER-M	MEAN	MAX
		SED	SED	TOXICITY	ER-L	ER-L	TOXICITY	ER-M	ER-M
∦ ∤		CONC	CONC	VALUE	RISK	RISK	VALUE	RISK	RISK
LOCATION	CHEMICAL	(ug/kg)	(ug/kg)	(ug/kg)	VALUE	VALUE	(ug/kg)	VALUE	VALUE
R4	ALUMINUM	12821540	22900000	174000	73.7	131.6	522000	24.6	43.9
B4	ARSENIC	10680	32300	33000	0.3	1.0	85000	0.1	0.4
R4	CADMIUM	4810	14900	5000	1.0	3.0	9000	0.5	1.7
R4	CHROMIUM	71610	224000	80000	0.9	2.8	145000	0.5	1.5
R4	COPPER	115850	332000	7000	16.6	47.4	390000	0.3	0.9
R4	LEAD	93960	219000	35000	2.7	6.3	110000	0.9	2.0
P4	MERCURY	3380	7300	150	22.5	48.7	1300	2.6	5.6
<b>P4</b>	NICKEL	28690	63000	30000	1.0	2.1	50000	0.6	1.3
R4	SELENIUM	1740	4000	6000	0.3	0.7	18000	0.1	0.2
R4	ZINC	161050	327000	120000	1.3	2.7	270000	0.6	1.2
R4	SUM RISK VALUES 1/				120.2	246.2		30.7	58.6
P4	SUM RISK VALUES 2/			······	27.4	61.7		4.6	11.2
RS	ALUMINUM	7023330	20700000	174000	40.4	119.0	522000	13.5	39.7
R5	ARSENIC	3830	9200	33000	0.1	0.3	85000	0.0	0.1
R5	CHROMIUM	17950	60600	80000	0.2	0.8	145000	0.1	0.4
R5	COPPER	41670	158000	7000	6.0	22.6	390000	0.1	0.4
R5	LEAD	237990	809000	35000	6.8	23.1	110000	2.2	7.4
R5	MERCURY	980	4100	150	6.5	27.3	1300	0.8	3.2
R5	NICKEL	6790	16100	30000	0.2	0.5	50000	0.1	0.3
R5	SELENIUM	NV	440	6000	0.0	0.1	18000	0.0	0.0
R5	ZINC	199680	765000	120000	1.7	6.4	270000	0.7	2.8
R5	SUM RISK VALUES 1/				61.9	200.0		17.5	54.3
R5	SUM RISK VALUES 2/				13.7	51.5		3.1	11.0
<b>R6</b>	ALUMINUM	11636400	19300000	174000	66.9	110.9	522000	22.3	37.0
FI6	ARSENIC	11100	24000	33000	0.3	0.7	85000	0.1	0.3
P6	CADMIUM	3830	13600	5000	0.8	2.7	9000	0.4	1.5
PI6	CHROMIUM	66600	281000	80000	0.8	3.5	145000	0.5	1.9
Pi6	COPPER	113500	303000	7000	16.2	43.3	390000	0.3	0.8
Pi6	LEAD	265700	876000	35000	8.2	25.0	110000	2.6	8.0
<del>1</del> 16	MERCURY	3300	17600	150	22.0	117.3	1300	2.5	13.5
Pi6	NICKEL	20300	78400	30000	0.7	2.6	50000	0.4	1.6
Pi6	SELENIUM	1370	6100	6000	0.2	1.0	18000	0.1	0.3
R6	ZINC	292300	564000	120000	2.4	4.7	270000	1.1	2.1
Pi6	SUM RISK VALUES 1/				118.5	311.9		30.3	67.0
PI6	SUM FISK VALUES 2/				32.1	149.3		6.2	25.2

Т

W92194F

7-67

## TABLE 7-18 (continued) RISK ESTIMATES - INORGANIC COCS IN SEDIMENT NYANZA III REMEDIAL INVESTIGATION MIDDLESEX COUNTY, MASSACHUSETTS PAGE 3

W92194F

	I			EFI	ECTS RANGE-LO	W	EFFECT	S RANGE-MEDI	Ŵ
ł	}	MEAN	MAX	ER-L	MEAN	MAX	ER-M	MEAN	MAX
		SED	SED	TOXICITY	ER-L	ER-L	TOXICITY	ER-M	ER-M
		CONC	CONC	VALUE	RISK	RISK	VALUE	RISK	RISK
LOCATION	CHEMICAL	(ug/kg)	(ug/kg)	(ug/kg)	VALUE	VALUE	(ug/kg)	VALUE	VALUE
R7	ALUMINUM	7736900	16500000	174000	44.5	<del>94</del> .8	522000	14.8	31.6
87	ARSENIC	10130	40600	33000	0.3	1.2	85000	0.1	0.5
R7	CADMIUM	2210	17900	5000	0.4	3.6	9000	0.2	2.0
R7	CHROMIUM	47950	209000	80000	0.6	2.6	145000	0.3	1.4
R7	COPPER	65940	278000	7000	9.4	39.7	390000	0.2	0.7
R7	LEAD	117940	526000	35000	3.4	15.0	110000	1.1	4.8
R7	MERCURY	1520	5500	150	10.1	36.7	1300	1.2	4.2
87	NICKEL	11290	44100	30000	0.4	1.5	50000	0.2	0.9
R7	SELENIUM	1180	7200	6000	0.2	1.2	18000	0.1	0.4
R7	ZINC	170050	646000	120000	1.4	5.4	270000	0.6	2.4
87	SUM FISK VALUES 1/				70.7	201.7		18.8	48.9
R7	SUM RISK VALUES 2/		· · · • · · · • · · · · •		14.9	59.1		2.9	12.9
<b>R8</b>	ALUMINUM	6636000	8310000	174000	38.1	47.8	522000	12.7	15.9
R8	ARSENIC	12130	30000	33000	0.4	0.9	85000	0.1	0.4
R8	CADMIUM	2080	4300	5000	0.4	0.9	9000	0.2	0.5
RB	CHROMUM	22000	36400	80000	0.3	0.5	145000	0.2	0.3
R8	COPPER	67610	96600	7000	9.7	13.8	390000	0.2	0.2
R8	LEAD	66700	100000	35000	1.9	2.9	110000	0.6	0.9
RB	MERCURY	1620	2100	150	10.8	14.0	1300	1.2	1.6
R8	NICKEL	7080	13100	30000	0.2	0.4	50000	. 0.1	0.3
R8	SELENIUM	750	1800	6000	0.1	0.3	18000	0.0	0.1
R8	ZINC	167360	329000	120000	1.6	2.7	270000	0.7	1.2
R8	SUM RISK VALUES 1/				63.5	84.1		16.1	21.4
Re	SUM RISK VALUES 2/				13.8	19.1		2.4	3.6
<b>R</b> 9	ALUMINUM	9606000	11700000	174000	56.4	67.2	522000	18.8	22.4
<b>R9</b>	ARSENIC	32000	64600	33000	1.0	2.0	85000	0.4	0.8
R9	CADMIUM	7050	8600	5000	1.4	1.7	9000	0.8	1.0
<b>R</b> 9	CHROMIUM	50260	78000	80000	0.6	1.0	145000	0.3	0.5
<b>R9</b>	COPPER	63330	135000	7000	11.9	19.3	390000	0.2	0.3
<b>R9</b>	LEAD	115260	184000	35000	3.3	5.3	110000	1.0	1.7
R9	MERCURY	3150	3900	150	21.0	26.0	1300	2.4	3.0
R9	NICKEL	24950	28100	30000	0.8	0.9	50000	0.5	0.6
R9	ZINC	197130	319000	120000	1.6	2.7	270000	0.7	1.2
R9	SUM RISK VALUES 1/				98.0	126.0		25.2	31.4
<b>R9</b>	SUM RISK VALUES 2/				27.3	35.9		5.0	6.9

(

.

# TABLE 7-18 (continued) RISK ESTIMATES - INORGANIC COCS IN SEDIMENT NYANZA III REMEDIAL INVESTIGATION MIDDLESEX COUNTY, MASSACHUSETTS PAGE 4

				EFF	ECTS RANGE-LO	w	EFFECTS RANGE-MEDIAN		
1		MEAN	MAX	ER-L	MEAN	MAX	ER-M	MEAN	MAX
		SED	SED	TOXICITY	ER-L	ER-L	TOXICITY	ER~M	ER-M
		CONC	CONC	VALUE	RISK	RISK	VALUE	RISK	RISK
	CHEMICAL	(ug/kg)	(ug/kg)	(ug/kg)	VALUE	VALUE	(ug/kg)	VALUE	VALUE
							500000		
R10	ALUMINUM	6790000	17600000	174000	39.0	101.1	522000	13.0	33.7
R10	ARSENIC	5990	12200	33000	0.2	0.4	85000	0.1	0.1
R10	CHROMIUM	11240	1/100	80000	0.1	0.2	145000	0.1	0.1
R10	COPPER	14115	31000	7000	2.0	4.4	390000	0.0	0.1
R10	LEAD	22480	33800	35000	0.6	1.0	110000	0.2	0.3
R10	MERCURY	161	530	150	1.1	3.5	1300	0.1	0.4
H10	NICKEL	/3/0	11100	30000	0.2	0.4	00000	0.1	0.2
R10	ZINC	38350	53400	120000	0.3	0.4	270000	0.1	0.2
R10	SUM RISK VALUES 1/				43.6	111.5		13.8	35.2
R10	SUM RISK VALUES 2/	··			2.0	5.1		0.5	1.0
		7216000	11050000	174000	41.5	63.5	522000	13.8	21.2
	ABSENIC	6590	12700	33000	02	0.4	85000	0.1	0.1
EW EW	CHROMEIM	123680	462000	80000	1.5	5.8	145000	0.9	3.2
EW EW	COPPER	63730	120000	7000	91	17.1	390000	0.2	0.3
PW	LEAD	72580	142000	35000	2.1	4.1	110000	0.7	1.3
BW	MERCURY	35878	152000	150	239.2	1013.3	1300	27.6	116.9
BW	NICKE	10143	40400	30000	0.3	1.3	50000	0.2	0.8
BW	SELENIUM	1430	6500	6000	0.2	1.1	18000	0.1	0.4
EW	ZINC	65393	164000	120000	0.5	1.4	270000	0.2	0.6
					294 7	1108.0		43 7	144.8
	SUM DISK VALUES 1/				243.0	1023.6		29.2	121.5
E	SUM HISK VALUES Z								
СВС	ALUMINUM	3520000	3690000	174000	20.2	21.2	522000	6.7	7.1
CBC	ARSENIC	4300	6600	33000	0.1	0.2	85000	0.1	0.1
CBC	CHROMIUM	82300	135000	80000	1.0	1.7	145000	0.6	0.9
CBC	COPPER	63350	92500	7000	9.1	13.2	390000	0.2	0.2
CBC	LEAD	38250	57500	35000	1.1	1.6	110000	0.3	0.5
CBC	MERCURY	6800	7100	150	45.3	47.3	1300	5.2	5,5
CBC	ZINC	92950	103000	120000	0.8	0.9	270000	0.3	0.4
СВС	SUM RISK VALUES 1/				77.6	86.1		13.4	14.7
CBC	SUM RISK VALUES 2/				47.6	50.9		6.2	7.0

## TABLE 7~18 (continued) RISK ESTIMATES - INORGANIC COCS IN SEDIMENT NYANZA III REMEDIAL INVESTIGATION MIDDLESEX COUNTY, MASSACHUSETTS PAGE 5

				EFF	ECTS RANGE-LO	W	EFFECTS RANGE - MEDIAN			
	l	MEAN	MAX	ER-L	MEAN	MAX	ER-M	MEAN	MAX	
	]	SED	SED	TOXICITY	ER-L	ER-L	TOXICITY	ER-M	ER~M	
		CONC	CONC	VALUE	RISK	RISK	VALUE	RISK	RISK	
LOCATION	CHEMICAL	(u <b>g/kg</b> )	(ug/kg)	(ug/kg)	VALUE	VALUE	(ug/kg)	VALUE	VALUE	
œ	ALUMINUM	5876670	11500000	174000	33.8	66.1	522000	11.3	22.0	
oc	ARSENIC	2930	4100	33000	0.1	0.1	85000	0.0	0.0	
OC	CHROMIUM	341600	988000	80000	4.3	12.4	145000	2.4	6.8	
oc	COPPER	137830	359000	7000	19.7	51.3	390000	0.4	0. <b>9</b>	
OC	LEAD	105950	233000	35000	3.0	6.7	110000	1.0	2.1	
oc	MERCURY	35330	99200	150	235.5	661.3	1300	27.2	<b>76</b> .3	
oc	ZINC	166300	390000	120000	1.4	3.3	270000	0.6	1.4	
oc	SUM RISK VALUES 1/				297.8	801.1		42.8	109.7	
<b>OC</b>	SUM RISK VALUES 2/	·····	·		242.9	680.5		30.5	85.3	
RW	ALUMINUM	22850000	29200000	174000	131.3	167.8	522000	43.8	55.9	
RW	ARSENIC	24960	37200	33000	0.8	1.1	85000	0.3	0.4	
RW	CADMIUM	2280	5600	5000	0.5	1.1	9000	0.3	0.6	
FW	CHROMIUM	155500	208000	80000	1.9	2.6	145000	1.1	1.4	
RW	COPPER	217750	306000	7000	22.2	29.7	390000	0.4	0.5	
RW	LEAD	435750	758000	35000	6.2	8.7	110000	2.0	2.8	
RW	MERCURY	710	970	150	4.7	6.5	1300	0.5	0.7	
RW	NICKEL	89890	186000	30000	3.0	6.2	50000	1.8	3.7	
RW	ZINC	365880	542000	120000	3.0	4.5	270000	1,4	2.0	
RW	SUM RISK VALUES 1/				173.7	228.3		51.5	68.2	
RW	SUM RISK VALUES 2/				14.1	20.1		4,1	6.0	
CSB	ALUMINUM	9225000	10700000	174000	53.0	61.5	522000	17.7	20.5	
CSB	ARSENIC	3500	4300	33000	0.1	0.1	85000	0.0	0.1	
CSB	CHROMIUM	15100	16700	80000	0.2	0.2	145000	0.1	0.1	
£SB	COPPER	31500	32900	7000	4.5	4.7	390000	0,1	0.1	
CSB	LEAD	234000	328000	35000	6.7	9.4	110000	2.1	3.0	
CSB	ZINC	148000	166000	120000	1.2	1.4	270000	0.5	0.6	
CSB	SUM RISK VALUES 1/				65.7	77.3		20.6	24.3	
CSB	SUM RISK VALUES 2/				7.0	9.7		2.3	3.1	

1

.

(

## TABLE 7-18 (continued) RISK ESTIMATES - INORGANIC COCS IN SEDIMENT NYANZA III REMEDIAL INVESTIGATION MIDDLESEX COUNTY, MASSACHUSETTS PAGE 6

[]	· -··			[ E	FFECTS RANGE-LO	w	EFFECTS RANGE-MEDIAN		
		MEAN	MAX	ER-L	MEAN	MAX	ER~M	MEAN	MAX
		SED	SED	TOXICITY	ER-L	ER-L	TOXICITY	ER-M	ER-M
		CONC	CONC	VALUE	RISK	RISK	VALUE	RISK	RISK
	CHEMICAL	(ug/kg)	(ug/kg)	(ug/kg)	VALUE	VALUE	(ug/kg)	VALUE	VALUE
RW	ALLMINUM	12474000	19300000	174000	71 7	110.9	522000	23.9	37.0
BW	ARSENIC	6220	11900	33000	02	0.4	85000	0.1	0.1
BW	CHROMILM	24010	101000	80000	0.3	1.3	145000	0.2	0.7
BW	COPPER	18013	83400	7000	2.6	11.9	390000	0.0	0.2
BW	LEAD	71190	254000	35000	2.0	7.3	110000	0.6	2.3
BW	MERCURY	1071	7600	150	7.1	50.7	1300	0.8	5.8
BW	NICKEL	7180	21800	30000	0.2	0.7	50000	0.1	0.4
BW	SELENIUM	70	1200	6000	0.0	0.2	18000	0.0	0.1
BW	ZINC	41078	127000	120000	0.3	1.1	270000	0.2	0.5
BW	SUM RISK VALUES 1/				84.5	184.4		26.0	47.2
BW	SUM FISK VALUES 2/	<b></b> _			9.7	59.5		1.7	9.0
HP	ALUMINUM	10580000	10580000	174000	60.8	60.8	522000	20.3	20.3
HP	ARSENIC	10850	10850	33000	0.3	0.3	85000	0.1	0.1
HP	CHROMIUM	40200	40200	80000	0.5	0.5	145000	0.3	0.3
HP	COPPER	136000	136000	7000	19.4	19.4	390000	0.3	0.3
HP	LEAD	149000	149000	35000	4.3	4.3	110000	1.4	1.4
HP	MERCURY	3500	3500	150	23.3	23.3	1300	2.7	2.7
HP	NICKEL	11300	11300	30000	0.4	0.4	50000	0.2	0.2
HP	ZINC	302000	302000	120000	2.5	2.5	270000	1.1	1.1
HP	SUM RISK VALUES 1/				111.5	111.5		26.4	26.4
HP	SUM RISK VALUES 2/				28.4	28.4		4.5	4.5

* B = BACKGROUND (INCLUDES REACH 1)

SAMPLED LOCATIONS: R-REACH, EW-EASTERN WETLANDS, CBC-CHEMICAL BROOK CULVERT, OC=OUTFALL CREEK, RW-RACEWAY, CSB-COLD SPRING BROOK, BW-BORDERING WETLAND, HP=HEARD POND

1/ SUM INCLUDES SITE - RELATED AND NON - SITE - RELATED CONTAMINANTS 2/ SUM INCLUDES SITE - RELATED CONTAMINANTS ONLY (As, Cd, Cr, Po, Hg)

W92194F

## TABLE 7-19 RISK ESTIMATES-ORGANIC COC IN SEDIMENT NYANZA III REMEDIAL INVESTIGATION MIDDLESEX COUNTY, MASSACHUSETTS

2-2-2-2		MEAN	MAX	BENCHMARK		
li		IW+	IW+	TOXICITY	MEAN	MAXIMUM
		CONC	CONC	VALUE	RISK	BISK
	CHEMICAL	(ua/L)	(ug/L)	(ug/L)	VALUE	VALUE
		<u> </u>				
B*	BENZO(A) PYRENE	NV	0.00	10	-	0.0
В	BENZO(B)FLUORANTHENE	NV	0.01	15	-	0.0
В	CHRYSENE	NV	0.03	10	-	0.0
В	FLUORANTHENE	NV	0.15	8.1	-	0.0
В	PHENANTHRENE	NV	0.44	6.3	-	0.1
В	PYRENE	NV	0.24	15	· -	0.0
1						
В	SUM RISK		·····			0.1
. 82	4 4' 000	0.00	0.02	0.001	0.0	20.0
82	4,4 - DDE	0.00	0.02	0.001	13.7	20.0
<b>B</b> 2	4,4 = 001	0.01	0.02	0.001	80.0	23.3 670.4
		0.00	10.00	0.001	80.0	0/9.4
		4.04	10.55	23	0.2	0.5
F-2	AGENAFRIRILENE	2.52	2.00	15	0.2	0.2
H2 D2	ANT HACENE	1.72	0.63	15	0.1	0.4
H2	BENZO(A)ANT HACENE	0.21	0.82	10	0.0	0.1
R2	BENZO(A)PYRENE	0.03	0.12	10	0.0	0.0
R2	BENZO(B) FLUOHAN I HENE	0.04	0.14	15	0.0	0.0
R2	BENZO(G,H,I)PERYLENE	0.00	0.01	15	0.0	0.0
R2	BENZO(K)FLUOHANTHENE	0.01	0.05	15	0.0	0.0
R2	BIS(2-EH)PHTHALATE	0.51	4.65	0.3	1.7	15.5
R2	CHRYSENE	0.16	0.63	10	0.0	0.1
R2	DIBENZOFURAN	2.59	4.94	15	0.2	0.3
R2	DIBENZO(A,H)ANTHRACENE	NV	0. <b>00</b>	10	-	0.0
i R2	FLUORANTHENE	0.93	4.29	8.1	0.1	0.5
+ R2	FLUORENE	2.05	5. <b>06</b>	3.2	0.6	1.6
R2	INDENO(1,2,3-CD)PYRENE	0.00	0.00	15	0.0	0.0
R2	METHYLMERCURY	0.06	0.17	0.012	4.9	13.9
R2	NAPHTHALENE #	12.46	24.54	20	0.6	1.2
R2	PHENANTHRENE	4.54	29.57	6.3	0.7	4.7
R2	PYRENE	1.51	8.05	15	0.1	0.5
R2	SUM RISK				103.1	762.3
R3	4.4'~DDD	0.05	0.21	0.001	50.0	206.2
R3	4.4'-DDE	0.01	0.02	0.001	10.6	18.1
B3	ACENAPHTHENE	NV	0.64	23	-	0.0
R3	ACENAPHTHYLENE	3 19	3 43	15	0.2	02
R3	ANTHRACENE	NV	0.41	15	-	0.0
R1	RENZOLALANTHRACENE	010	0.41	10	00	0.0
	RENTOVALPADENE	0.10	0.00	10	0.0	0.1
n.) D2		0.00	0.11	10	0.0	0.0
n.) Do		0.05	0.13	13	0.0	0.0
n3 00		0.00	0.01	13	0.0	0.0
H3		0.01	0.04	15	0.0	0.0
H3	DIDIZ-ETIPTIPALALE	1.36	1.59	0.3	4.0	5.3
H3		0.20	0.87	10	0.0	0.1
<b>H3</b>	DIBENZO(A,H)ANTHHACENE	NV	0.01	10	-	0.0
H3	DI-N-BUITLPHIMALATE	1.55	4.90	0.3	5.2	16.3
R3	FLUORANTHENE	0.81	4.17	8.1	0.1	0.5
R3	FLUORENE	NV	0.51	3.2	-	0.2
R3	INDENO(1,2,3-CD)PYRENE	0.00	0.00	15	0.0	0.0
R3	METHYLMERCURY	NV	0.01	0.012	-	0.9
R3	NAPHTHALENE #	NV	1.87	20	-	0.1
R3	PHENANTHRENE	2.85	15.20	6.3	0.5	2.4
R3	PYRENE	1.67	7.46	15	0.1	0.5
					74 *	
R3	SUM RISK				/1.2	250.9

# TABLE 7-19 (continued) RISK ESTIMATES-ORGANIC COC IN SEDIMENT NYANZA III REMEDIAL INVESTIGATION MIDDLESEX, MASSACHUSETTS PAGE 2

IWIWTOXICITYMEANCONCCONCVALUERISKLOCATIONCHEMICAL(ug/L)(ug/L)(ug/L)R4ANTHRACENENV0.1915-R4BENZO(A)ANTHRACENENV0.0110-R4BENZO(A)PYRENE0.000.00100.0R4BENZO(B) FLUORANTHENE0.010.01150.0R4BENZO(G,H,I)PERYLENENV0.0015-	MAXIMUM RISK VALUE 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.
CONC         CONC         VALUE         RISK           LOCATION         CHEMICAL         (ug/L)         (ug/L)         (ug/L)         VALUE           R4         ANTHRACENE         NV         0.19         15         -           R4         BENZO(A)ANTHRACENE         NV         0.01         10         -           R4         BENZO(A)ANTHRACENE         0.00         0.00         10         0.0           R4         BENZO(A)PYRENE         0.01         0.01         15         0.0           R4         BENZO(B)FLUORANTHENE         0.01         0.01         15         0.0           R4         BENZO(G, H, I)PERYLENE         NV         0.00         15         -	RISK VALUE 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.
LOCATION         CHEMICAL         (ug/L)         (ug/L)         (ug/L)         VALUE           R4         ANTHRACENE         NV         0.19         15         -           R4         BENZO(A)ANTHRACENE         NV         0.01         10         -           R4         BENZO(A)ANTHRACENE         0.00         0.00         10         0.0           R4         BENZO(A)PYRENE         0.01         0.01         15         0.0           R4         BENZO(B)FLUORANTHENE         0.01         0.01         15         0.0           R4         BENZO(G,H,I)PERYLENE         NV         0.000         15         -	VALUE 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0
R4         ANTHRACENE         NV         0.19         15         -           R4         BENZO(A)ANTHRACENE         NV         0.01         10         -           R4         BENZO(A)PYRENE         0.00         0.00         10         0.0           R4         BENZO(A)PYRENE         0.01         0.01         10         0.0           R4         BENZO(B)FLUORANTHENE         0.01         0.01         15         0.0           R4         BENZO(G,H,I)PERYLENE         NV         0.00         15         -	0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0
R4         BENZO(A)ANTHRACENE         NV         0.01         10         -           R4         BENZO(A)PYRENE         0.00         0.00         10         0.0           R4         BENZO(B)FLUORANTHENE         0.01         0.01         15         0.0           R4         BENZO(G,H,I)PERYLENE         NV         0.00         15         -	0.0 0.0 0.0 0.0 0.0 0.0 0.0
R4         BENZO(A)PYRENE         0.00         0.00         10         0.0           R4         BENZO(B)FLUORANTHENE         0.01         0.01         15         0.0           R4         BENZO(G,H,I)PERYLENE         NV         0.00         15         -	0.0 0.0 0.0 0.0 0.0 0.0
R4         BENZO(B)FLUORANTHENE         0.00         0.00         10         0.00           R4         BENZO(B)FLUORANTHENE         0.01         0.01         15         0.0           R4         BENZO(G,H,I)PERYLENE         NV         0.00         15         -	0.0 0.0 0.0 0.0 0.0 0.0
R4         BENZO(B)FLOORANTHENE         0.01         15         0.01           R4         BENZO(G,H,I)PERYLENE         NV         0.00         15         -	0.0 0.0 0.0 0.0 0.0
R4 BENZO(G,H,I)FERTLENE NV 0.00 15 -	0.0 0.0 0.0 0.0
	0.0 0.0 0.0
H4 BENZO(N)FLOOHANIHENE NV 0.00 15 -	0.0 0.0 0.0
H4 CHITSENE 0.02 0.02 10 0.0	0.0
H4 FLOCHANTHENE 0.07 0.08 8.1 0.0	0.0
H4 INDENO(1,2,3-CD) PYRENE NV 0.00 15 -	
H4 METHYLMERCUHY 0.01 0.03 0.012 0.7	2.1
R4 PHENANTHRENE NV 0.28 6.3 -	0.0
R4 PYRENE 0.14 0.14 15 0.0	0.0
R4 SUM RISK 0.7	2.2
R5 ACENAPHTHYLENE 1 25 1 25 1 5 0 1	01
R5 ANTHRACENE 0.51 0.51 15 0.0	0.0
R5 RENZO(A)ANTHRACENE 0.13 0.13 10 0.0	0.0
	0.0
	0.0
	0.0
PE DIS(0, CI) DEPUISIANTE 115 0.00 0.00 2.0	0.0
R5 BIS(2-EH)PHIHALATE 1.15 1.15 0.3 3.6	3.6
H5 CHRYSENE 0.16 0.16 10 0.0	0.0
R5 FLUORANTHENE 0.76 0.76 8.1 0.1	0.1
R5 INDENO(1,2,3-CD)PYRENE 0.00 0.00 15 0.0	0.0
R5 PHENANTHRENE 2.25 2.25 6.3 0.4	0.4
R5 PYRENE 1.49 1.49 15 0.1	0.1
R5 SUM RISK 4.5	4.5
R6 4,4'~DDD 0.22 0.39 0.001 222.3	390.1
R6 ACENAPHTHENE NV 0.47 23 -	0.0
R6 ANTHRACENE NV 0.41 15 -	0.0
R6 BENZO(A)ANTHRACENE 0.14 0.21 10 0.0	0.0
R6 BENZO(A)PYRENE NV 0.03 10 -	0.0
R6 BENZO(B) FLUORANTHENE 0.05 0.10 15 0.0	0.0
R6 BENZOIG HIPPERYLENE NY 0.01 15 -	0.0
R6 BENZOKKELUORANTHENE NY 0.01 15 -	0.0
	11.0
DE CHEVENE NV 017 10 -	00
	0.0
	01
	00
	00
NO FRENANTRIE 2.44 3.33 0.3 0.4 DE DYDENE 1.41 1.07 18 0.4	05
P16 SUM RISK 222.9	402 1
R7 ACENAPHTHENE NV 0.7 23 -	0.0
H7 ANTHRACENE NV 0.36 15 -	0.0
R7 BENZO(A)ANTHRACENE 0.6 0.12 10 0.1	0.0
R7 BENZO(A) PYRENE 0.02 0.02 10 0.0	0.0
R7 BENZO(B) FLUORANTHENE 0.02 0.03 15 0.0	0.0
R7 BENZO(G,H,I)PERYLENE NV 0.00 15 -	<b>0</b> .0
R7 BENZO(K) FLUORANTHENE 0.00 0.01 15 0.0	0.0
R7 BIS(2-EH)PHTHALATE 1.47 1.92 0.3 4.9	6.4
R7 CHRYSENE 0.07 0.14 10 0.0	0.0

## TABLE 7-19 (continued) RISK ESTIMATES-ORGANIC COC IN SEDIMENT NYANZA III REMEDIAL INVESTIGATION MIDDLESEX, MASSACHUSETTS PAGE 3

-

FINAL

	<u></u>	MEAN	MAX	BENCHMARK		
		IW	IW	TOXICITY	MEAN	MAXIMUM
		CONC	CONC	VALUE	RISK	RISK
LOCATION	CHEMICAL	(ug/L)	(ug/ <b>L)</b>	( <b>ug</b> /L)	VALUE	VALUE
87	ELLODANTHENE	0.33	0.66			
H/	FLUORANTHENE	0.33	0.00	8.1	0.0	0.1
н/		N V	0.35	3.2	-	0.1
H7		NV	0.00	15		0.0
H7	PHENANIHRENE	1.23	2.50	6.3	0.2	0.4
87	PTRENE	0.59	1,15	15	0.0	0.1
R7	SUM RISK				5.2	7.1
EW	4,4'-DDD	0.17	0.26	0.001	169.4	260.8
EW	BENZO(A)PYRENE	0.08	0.08	10	0.0	0.0
EW	BENZO(B) FLUORANTHENE	0.00	0.30	15	0.0	0.0
FW	CHRYSENE	0.46	0 54	10	0.0	0.1
FW	FLUORANTHENE	1.69	2 22	81	0.2	0.3
FW	METHYLMERCURY	0.29	0.37	0.012	24.0	30 7
FW		97 46	143.51	20	49	72
EW	PHENANTHRENE	57.40 NV	5 16	63	4.5	0.8
EW	PYDENE	3 43	4.61	0.5	0.2	0.8
	FIGENE	3.43	4.01	.5	0.2	0.5
EW	SUM RISK				198.8	300.2
CBC	ACENAPHTHYLENE	NV	0.44	15	-	0.0
CBC	ANTHRACENE	0.29	0.34	15	0.0	0.0
CBC	BENZO(A)ANTHRACENE	0.08	0.09	10	0.0	0.0
CBC	BENZO(A) PYBENE	0.01	0.02	10	0.0	0.0
CBC	BENZO(B) ELLOBANTHENE	0.02	0.02	15	0.0	0.0
080	BENZO(G H I) PERM ENE	NV	0.00	15	0.0	0.0
CBC	RENZOKOELLORANTHENE	0.01	0.00	15	0.0	0.0
		0.01	0.29	03	07	0.0
	CHOYCENE	0.21	0.28	0.3	0.7	0.9
		0.07	0.07	10	0.0	0.0
CBC	DIBENZOFURAN		0.26	15	-	0.0
CBC	FLUORANTHENE	0.36	0.39	8.1	0.0	0.0
CBC	FLUORENE	NV	0.17	3.2	-	0.1
CBC	INDENO(1,2,3 - CD)PYRENE	NV	0.00	15	-	0.0
CBC	NAPHTHALENE #	7.05	7.49	20	0.4	0.4
<b>C8</b> C	PHENANTHRENE	1.27	1.32	6.3	0.2	0.2
CBC	PYRENE	0.65	0.65	15	0.0	0.0
C8C	SUM RISK				1.4	1.8
<u>~</u>		2.05	2 41	15	0 1	0.2
<u>~</u>	ANTHRACENE	NV	0.63	15	-	00
~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	RENZOVALANTHRACENE	0.22	0.00	10	<u> </u>	0.0
~	BEN7((A) DVDENE	0.22	0.27	10	0.0	0.0
~		0.07	A + 1	42	0.0	0.0
<u>.</u>	DENLOYDIFLOUTANI MENE	0.07		13	0.0	0.0
		0.01	0.01	13	0.0	0.0
OC C	BENZUKUFLUUHANIHENE	NV	0 02	15	-	0.0
00	BIS(2-EH)PHTHALATE	3.34	6.97	0.3	11.1	23.2
00	CHHYSENE	0.24	0.32	10	0.0	0.0
00	FLUORANTHENE	0.91	1.11	8.1	0.1	0.1
OC	INDENO(1,2,3 - CD) PYRENE	0.00	0.00	15	0.0	0.0
oc	NAPHTHALENE #	6.80	10.61	20	0.3	0.5
oc	PHENANTHRENE	2.62	3.87	6.3	0.4	0.6
oc	PYRENE	2.44	3.39	15	0.2	0.2
ос	SUM RISK				12.4	25.0

• -

~

~---

TABLE 7-19 (continued) RISK ESTIMATES-ORGANIC COC IN SEDIMENT NYANZA III REMEDIAL INVESTIGATION MIDDLESEX, MASSACHUSETTS PAGE 4

FINAL

		MEAN	MAX	BENCHMARK		
		IW	W	TOXICITY	MEAN	MAXIMUM
1		CONC	CONC	VALUE	RISK	RISK
LOCATION	CHEMICAL	(ug/L)	(ug/L)	(ug/L)	VALUE	VALUE
RW	4.4°-DDD	0.02	0.08	0.001	20.0	80.0
RW	4,4'-DDE	0.00	0.01	0.001	0.0	10.0
RW	ANTHRACENE	NV	7.09	15	_	0.5
RW	BENZO(A)ANTHRACENE	1.07	2.06	10	0.1	0.2
RW	BENZO(A) PYRENE	0.13	0.15	10	0.0	0.0
RW	BENZO(B) FLUORANTHENE	0.17	0.21	15	0.0	0.0
RW	BENZO(K) FLUORANTHENE	0.06	0.08	15	0.0	0.0
RW	BIS(2-EH)PHTHALATE	NV	5.81	0.3	_	19.4
RW	CHRYSENE	0.90	1.30	10	0.1	0.1
RW	FLUORANTHENE	6.61	9.27	8.1	0.8	1.1
RW	FLUORENE	NV	9.44	3.2	-	3.0
RW	NAPHTHALENE #	84.88	139.35	20	4.2	7.0
RW	PHENANTHRENE	33.63	44.43	6.3	5.3	7.1
RW	PYRENE	8.50	11.76	15	0.6	0.8
RW	SUM RISK		_		31.2	129.1

* B = BACKGROUND (INCLUDES BACKGROUND)

+ IW = INTERSTITIAL WATER

SITE-RELATED CONTAMINANT (SRC) - (NAPHTHALENE = ONLY ORGANIC SRC IN SEDIMENTS) SUM RISK = INCLUDES SITE-RELATED AND NON-SITE-RELATED CONTAMINANTS NV: NO VALUE - MEAN EXCEEDS MAXIMUM SAMPLED LOCATIONS: R=REACH, EW=EASTERN WETLANDS, CBC=CHEMICAL BROOK CULVERT,

OC=OUTFALL CREEK, RW=RACEWAY

FINAL

The high risk estimates for these primary toxicants generally result from the combination of extremely low benchmark toxicity values and slightly to moderately elevated estimated interstitial water concentrations.

7.6.5 Risk from Biota COCs

The ingestion of contaminated biota is considered to be a critical exposure pathway for aquatic, avian, and terrestrial biota that consume fish. Terrestrial and avian organisms are not limited to specific reaches of aquatic habitat, and the entire study area should be considered a single unit for the purpose of defining actual exposures for these organisms. However, individual reaches are assessed in this section so relative hazards posed to aquatic, terrestrial, and avian biota can be assessed on a location-specific basis.

divided average and maximum This assessment fish tissue concentrations of each biota COC within each reach by benchmark toxicity values. As in the surface water and sediment risk characterizations, each individual COC-specific quotient is directly comparable to the other COC-specific quotients. Again, the relative toxicity associated with each COC in each reach can be assessed. Individual quotients for all biota COCs in each reach were added together, and the cumulative risk from all biota COCs Table 7-20 lists the results of was calculated for each reach. risk calculations based on biota COCs.

As listed in Table 7-20, the biota COCs chlordane and dieldrin appear to present little or no risk to species that consume Study Area fish. DDT and its metabolites and total PCBs appear to present relatively small, yet measurable, risks to piscivorous consumers. By far, the primary risk posed to predators of Study Area fish results from the ingestion of mercury-contaminated fish.

Total mercury accounts for nearly all of the measurable risk for all locations sampled. In some cases, however, the maximum risk values associated with mercury can be overestimated. For example, estimated whole-body mercury concentrations for yellow perch in Background areas are probably overestimated because of the single suspect sample described above. This sample was associated with extremely high mercury concentrations relative to all others from Background areas. Most likely, this value is questionable, and probably not representative of yellow perch contamination in Background areas.

For other species and other locations, both mean and maximum risk values appear to be representative of true values, based on the distribution of values for each species and each reach; that is, no

extreme outliers are associated with other reaches and other species. Maximum total mercury risk estimates, not including Background yellow perch, ranged from 11.2 (Reach 2 largemouth bass) to 76.0 (Reach 3 largemouth bass).

The values listed in Table 7-20 are based on Phase I and Phase II tissue concentrations unadjusted for fish length, weight, or age. Phase II fish data included fish length, weight, and age. All three parameters were correlated positively to tissue concentration of mercury. The correlation was slightly stronger for fish age $(r^2 = 0.67)$ than for length $(r^2 = 0.65)$ or weight $(r^2 = 0.64)$. Therefore, Phase II fish tissue concentrations of mercury were normalized to fish age. Table 7-21 lists the results of normalization for mercury concentration in Phase II fish samples.

Again, the normalized data for Background yellow perch might be adversely affected by the outlier value described previously. Disregarding Reach 1 yellow perch, maximum age-normalized total mercury concentrations ranged from 155 ug/kg/yr (Reach 6 yellow perch). Normalization of fish data decreased the differences in average and maximum tissue concentrations of mercury between fish species. Largemouth bass, because they were generally older and larger, are expected to have greater exposure duration than yellow perch or bullheads; non-normalized data support this assumption.

This assessment compared mean and maximum age-normalized total mercury concentrations in largemouth bass, based on whole-body estimates, to mean and maximum total mercury concentrations in sediment. The comparisons were limited to largemouth bass because the Phase II data used to investigate the effects of agenormalization were more complete for that species. Figure 7-5 shows the results of these comparisons. In general, the relationship between age-normalized concentrations of total mercury in fish and total mercury concentrations in sediment is not strong. However, these data suggest that, in general, fish with the highest mercury concentrations are associated with reaches that contain the most contaminated sediments. Figure 7-6 shows similar data, based on fish concentrations that were not nomalized for age.

This risk assessment derived sediment/fish ratios (ug/kg Hg in sediment / ug/kg/yr in fish) for each reach and for both nomalized (Figure 7-5) and non-normalized data (Figur 7-6). These ratios are based on both mean and maximum concentrations in largemouth bass and in sediments. In general, higher sediment/fish ratios were associated with the most contaminated reaches, and mercury concentrations in largemouth bass were less variable than concentrations in sediments.

7-77
TABLE 7-20 RISK ESTIMATES-BIOTA COCS NYANZA III REMEDIAL INVESTIGATION MIDDLESEX COUNTY, MASSACHUSETTS

		· · · · · · · · · · · · · · · · · · ·	MEAN		<u></u>		
1							
			WHOLE	WHOLE	BENCHMARK		
			BODY	BODY	TOXICITY	MEAN	MAX
4 			CONC	CONC	VALUE	RISK	RISK
LOCATION	SPECIES	CHEMICAL	(ug/kg)	(ug/kg)	(ug/kg)	VALUE	VALUE
		T 007			450		
в	LM		91	190	150	0.6	1.3
в	LM	T. PCB	108	152	640	0.2	0.2
В	LM	T. CHLORDANE	1.2	1.2	300	0.0	0.0
В	LM	DIELDRIN	2.8	2.8	300	0.0	0.0
В	LM	T. MERCURY	745	1180	100	7.5	11.8
-							
<u> </u>	LM	SUM RISK			·	8.2	13.3
в	YP	T. DOT	NS	_	150	_	_
B	YP	TECR	NS	-	640	_	_
, D	VD		NC	_	200	_	_
		I. CHECHDANE	NG NG	-	300	-	-
В	TP	DIELDHIN	NS	_	300	-	-
В	YP	T. MERCURY #	3330	9600	100	33.3	96.0
в	YP	SUM RISK				33.3	96.0
		T 00T					
H2	LM		ND	-	150	-	-
FR2	LM	T. PCB	ND	-	640	-	-
R2	LM	T. CHLORDANE	ND	-	300	-	-
R2	LM	DIELDRIN	ND	_	300	_	- 1
R2	LM	T. MERCURY	1120	1120	100	11.2	11.2
	1.64						
R ∠	W	SUM NISK				11.2	11.2
R2	YP	T. DDT	60	6 6	150	0.4	0.4
' R2	YP	T. PCB	450	450	640	07	0.7
- R2	YP		ND	-	300	-	-
			ND		200	_	
, P2	TP ND		NU	_	300		
H2	٩P	T. MERCURY	1430	1990	100	14.3	19.8
R2	YP	SUM RISK				15.4	20.9
R3	LM	T. DOT	116	293	150	0.8	2.0
R 3	ŪM.	TPCB	249	708	640	04	11
82	L M		23	3.4	200	0.4	0.0
			2.5	3.4		0.0	0.0
MG	UM	DIELDHIN	0.3	0.3	300	0.0	0.0
R3	LM	T. MERCURY	2870	7600	100	28.7	76.0
P 3	LM.	SUM RISK		_		29.9	79.1
	VD	TOOT		185	150		10
2			65	100	130	U.4	1.0
NJ	47 		206	704	640	0.3	1.1
FG	YP	I. CHLOFIDANE	1.5	1.9	300	0.0	, 0.0
P 3	YP	DIELDAIN	ND	-	300	-	-
R3	YP	T. MERCURY	1560	3640	100	15.6	36.4
P 3	YP	SUM RISK				16.3	38.5
		1.007	~	47		~~	
10	611		32	4/	150	0.2	0.3
FG	BH	I. PCB	127	150	640	0.2	0.2
R3	BH	T. CHLORDANE	4.2	5.2	300	0.0	0.0
R3	BH	DIELDRIN	2.8	2.8	300	0.0	0.0
R3	BH	T. MERCURY	1990	2850	100	19.9	28.5
P 3	BH	SUM RISK				20.3	29 1
64	1 8.4	T DOT			160		~ ~ ~
PM R4	LM	TPCB	80 553	114 <u>5</u> 53	001 040	V.6 0 0	0.0 U.B
D.4	1 8.4		446	146	300	0.0	0.5 A A ¹
114	LM		14.0	14.0	300	0.0	0.0
H4	LM	UIELUHIN	0.9	0.9	300	0.0	0.0
R4	LM	T. MERCURY	1470	419 0	100	14.7	4 1.9
D4	1 84	SIM DICK				16	17
FWI	LM .					1.5	1.7

FINAL

_

_

TABLE 7-20 (continued) **RISK ESTIMATES-BIOTA COCS** NYANZA III REMEDIAL INVESTIGATION MIDDLESEX COUNTY, MASSACHUSETTS PAGE 2

·							
			MEAN	MAX			
í			WHOLE	WHOLE	BENCHMARK		
		•	BODY	BODY	TOXICITY	MEAN	MAX
	_		CONC	CONC	VALUE	RISK	RISK
LOCATION	SPECIES	CHEMICAL	(ug/kg)	(ug/kg)	(ug/kg)	VALUE	VALUE
D4	VD		55	66	150	0.4	0.4
		T DCB	135	135	640	0.4	0.4
			100 ND	135	300	0.2	0.2
P4				-	300	-	
P4	TP VD		0.0	0.0	300	0.0 7 e	0.0 7 e
174	TP	1. MERCURY	760	/60	100	7.0	7.0
R4	YP	SUM RISK				8.2	8.3
	~						
H4	BH		86	170	150	0.6	1.1
H4	BH		127	216	640	0.2	0.3
H4	BH	1. CHLOHDANE	12	20.8	300	0.0	0.1
H4	BH	DIELDHIN	ND	_	300	_	-
H4	BH	T. MEHCUHY	920	1200	100	9.2	12.0
R4	BH	SUM RISK				10.0	13.5
H6	LM		57	88	150	0.4	0.6
FI6	LM	T. PCB	177	208	640	0.3	0.3
PI6	LM	T. CHLORDANE	ND	-	300	-	-
PI6	LM	DIELÖRIN	ND	-	300	-	-
R6	LM	T. MERCURY	940	1800	100	9.4	18.0
PI6	LM	SUM RISK			<u></u>	10.1	18.9
B6	VP		29	65	150	02	04
B6	VP	T PCB	68	95	640	0.1	0.1
P6	VP		21	23	300	0.1	0.0
26	VP		2.1	2.0	300	0.0	0.0
De	VP	TAEDOURY	200	1400	100	7.0	14.0
	16	1. MERCONT	700	1400	100	7.0	14.0
Pi6	YP	SUM RISK				7.3	14.6
· P6	BH	T. DDT	65	96	150	0.4	0.6
PI6	BH	TPCB	127	190	640	0.2	0.3
86	BH	TCHLORDANE	52	502	300	00	1.7
R6	BH	DIFLORIN	0	0	300	00	0.0
R6	BH	T. MERCURY	740	880	100	7.4	8.8
De							
						0.0	
FQ	LM	T. DOT	73	137	150	0.5	0.9
Re	LM	T. PCB	405	780	640	0.6	1.2
PQ	LM	T. CHLORDANE	0.5	0.5	300	0.0	0.0
RD	LM	DIELDRIN	0.3	0.3	300	0.0	0.0
F Ø	LM	T. MERCURY	1320	3200	100	13.2	32.0
RD	LM	SUM RISK				14.3	34 1
		T 007	•		160	0.3	
		T 000		~~~	130	~~~	0.3
	17		13/			0.4	Q.Q
H9	44	I UNLUHUANE	2.3	2.3	300	0.0	Ų.0
P\$	44		NU	-	300	-	-
HQ	YP	I. MEHLUHY	720	1800	100	72	18.0
Re	YP	SUM RISK				77	190

LOCATION: R=REACH

SPECIES: LM=LARGEMOUTH BASS, YP=YELLOW PERCH, BH=BULLHEAD * B = BACKGROUND

ND = NOT DETECTED

NS = NOT SAMPLED

VALUES INCLUDE QUESTIONABLE SAMPLE

TABLE 7-21 TOTAL MERCURY CONCENTRATIONS IN WHOLE BODY FISH, NORMALIZED FOR FISH AGE NYANZA III REMEDIAL INVESTIGATION MIDDLESEX COUNTY, MASSACHUSETTS

• - --

p								MEAN	MAY	MCAN	
4					EST		NORM	NORM	NORM	MEAN	MAX
		INORG		METHNI	TOTAL		TOTAL		T MERC	T MEDC	T MEDC
		MEDCURY		MERCURY	MERCURY	EICH	MEDCURV	CONC	CONC.	CONC	CONC
1		MERCURT		CONC	MERCURY	FISH	MERCURT	DUNC.	CONC.	DY SD/LOO	
LOCATION		CUNC. 1/	0664	CONC.	DEE# (volke)	AGE	CONC.	BT SP/LUC	BT SP/LOC	BT SP/LUC	BT SP/LUC
LUCATION	SPECIES		ncr#			(Years)	(ug/kg/yr)				
D#	VD	360			360	4	90				
		500			500		30				
D	TP VD	390	•		0000	•	140		2000	2547	
в	ΥP	9600	2		9600	3	3200	1146	3200	3517	9600
_											1
В	LM	280			28 0	2	140				
• B	LM	610			610	5	122				
В	LM	890			890	5	178				
В	LM	410			410	4	103	136	178	548	890
-											
P 2	VP	720			720	5	146				1
n2 D0		730			730	3	140				
 	TP VD	370			370		53				
H2	YP VP	1981			1981	4	495				
H2	YP	881			881	5	176	218	495	991	1981
:											
R2	LM	1123			1123	4	281				
R2	LM	685	3/	420	685	3	228				
R2	LM	1076	3/	660	1076	6	179	229	281	961	1123
						5					
R 3	I M	Q18			Q1A	5	184				
. ~ ~	1 84	1040			1040	J 4	104 0e1				í.
- no 64		1042			1042	4	201				1
n .3	LM	1394			1334	5	2/9				
R3	LM	1549			1549	5	310				
R3	LM	17 8 6			17 8 6	4	447				,
R3	LM	1739			1739	4	435				
R3	LM	2570			2570	5	514				i
R3	LM	1943			1943	4	486				
R3	LM	2831			2631	5	566				
83	L M	2785			2785	e e	484				
	1.64	4463			4463	, in the second s	404				
~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	1.14	*******			4400		490				
, <b>F</b> S	LM	2252			2282		362				1
<b>R3</b>	LW	3962			3962	5	792				
R3	LM	1899			1899	7	271				
- R3	LM	3524			3524	8	441				i
R3	LM	1214			1214	4	304				
' R3	LM	1228			1228	4	307				
R3	LM	2745			2745	9	305				
R1	I M	5775			5775	10	578				
89	1.14	4173			6170		691				
~~ ~~	1.4.4	2468			0132		001				
2	LW	3400			3400	10	347				
RS	LM	3521	J	2160	3921	3	1174				
R3	LM	2853	3/	1750	2653	5	571				
R3	LM	1842	<b>3</b> /	1130	1642	5	368				
R3	LM	2478	3/	1520	2478	2	1239				
R3	LM	5591	3V	3430	5591	5	1118				
R3	LM	4792	37	2940	4792	6	799				
R1	ι.M	4515	3	2770	4515		753				
Rh	1.64	3651		2240	3651	, , , , , , , , , , , , , , , , , , ,	809				
~~ 62		0001		1460	0001		300				
~	LIN	2304	34	1450	2304	3	/ 88				
KJ		4385	JV	2000	4385	7	626				
R3	L M	1092	3/	670	1092	2	546				
R3	LM	2103	3V	1290	2103	2	1051				
R3	LM	1402	3.	880	1402	2	701				
R3	LM	3537	Ľ	2170	3537	5	707				
R3	LM	1940	Ľ	1190	1940	4	485				
R	I M	2671	, n	1640	9671	, ,	1337	487	1 237	2811	6172
		2013	-		20, 3	6			1001	2011	U I UE
87	¥P.										
	11	116/			116/	5	253				
RG .	YP	993			993	4	248				
R3	YP	990			990	4	248				
R3	YP	680			680	4	170				
R3	YP	882			882	4	221		•		i
R3	YP	1114			1114	5	223				
R3	YP	784			784	ä	261				
 	VD	2049	3/	2250	3049	3	1316				
. D2	VP	0000 0546	3/	2110	3540	ى م	996				
- no - no	17	3345	Эү	2110	3343		4700				1
HS R	14	5359	3/	3190	5359	3	1/86				1
HS .	٧P	2386	3/	1420	2386	4	596				

FINAL

~

___.

									MEAN	MAX	MEAN	MAX
	ļ					EST.		NORM.	NORM.	NORM.	NON-NORM	NON-NORM.
			INORG.		METHYL	TOTAL		TOTAL	T.MERC.	T.MERC.	T.MERC.	T.MERC.
			MERCURY	1	MEHCUHY	MERCURY	HSH	MERCURY	CONC.	CONC.	CONC.	CONC.
	1004704		CONC. 1/	DEEM	CONC.	CONC.	AGE	CONC.	BY SP/LOC	BY SP/LOC	BY SP/LOC	BY SP/LOC
	LOCATION	SPECIES		HEF#	(ug/kg)		(years)	(ug/kg/yr) :		(ua/ka/yr)		
	81	YP	958	3/	570	958	3	319				
	R3	VP	1075	3/	640	1075	3	358				1
	R3	YP	3175	3/	1890	3175	3	1058				[
	R3	YP	2738	3/	1630	2738	4	685				i
	R3	YP	874	3/	520	874	4	218				
	R3	YP	2150	3/	1280	2150	Å	538	550	6132	785	6132
				•,	1200	2.00	•			0.01	100	0.02
	R4	LM	303			303	2	152				
i	R4	LM	888			888	3	296				
i	R4	LM	539			539	2	270				ĺ
	R4	LM	1826			1826	5	365				
	R4	LM	797			797	2	399				
	R4	LM	1308			1308	3	436				[
	R4	LM	1277			1277	5	255				
	R4	LM	1313			1313	6	219				
	R4	LM	1144			1144	3	381				ĺ
	R4	LM	1164			1164	6	194				
	R4	LM	1561			1561	7	223				
	R4	LM	1211			1211	8	151				
	R4	LM	1099			1099	6	183				1
	R4	LM	1378			1378	7	197				{
	R4	LM	1660			1660	5	332				
	R4	LM	473	3/	290	473	4	118				ĺ
	R4	LM	2201	3/	1350	2201	6	367				
	R4	LM	1744	3/	1070	1744	5	349				
ļ	R4	LM	2217	3/	1360	2217	4	554				4
	R4	LM	1011	3/	620	1011	4	253				
	R4	LM	1679	3/	1030	1679	3	560				
	R4	LM	1532	3/	940	1532	3	511				i
	R4	LM	359	3/	220	359	3	120				
	R4	LM	2787	3/	1710	2787	3	929				
	R4	LM	3032	3/	1860	3032	5	606				ł
	R4	LM	2315	3/	1420	2315	4	579				i
	R4	LM	994	3/	610	994	3	331				
	R4	LM	1451	3/	890	1451	3	484				
	R4	LM	2347	3/	1440	2347	3	782				
	R4	LM	2282	3/	1400	2262	3	761				
	R4	LM	1043	3V	640	1043	3	348				1
	H4		1174	<b>3</b> V	720	1174	4	293				i
	P14		1076	37	1000	1076	3	339				i
	P4		2100		1330	2106	د ۲	723	681	~177	2442	8049
			3/48	34	2300	3/48	5	/30	005	211	2445	0048
	Ba	VP	1708	3/	1070	1708	R	100				
	Ba	YP	2218	3/	1990	2218	, , , , , , , , , , , , , , , , , , ,	170				
	R4	YP	1304		830	1304	š	279				
	B4	VP	805	3/	480	805	, j	101				
	84	YP	521		110	421	š	104				
	R4	YP	1210		720	1210	5	242				
	R4	YP	924		540	924	5	185				
	RA	YP	121		490	823	5	165				
	R4	VP	1075	3	840	1075	5	215				
	Rd	YP	1109	3	880	1100		139	210	370	1166	2218
				-			•			•••		
	R6	LM	1800			1800	10	160				
	R6	LM	1500			1500	10	150				:
	Ris	Ϊ.M	200			200	7	29				
	R6	LM	1200			1200	4	300				,
	R6	LM	760			780	8	95				
	Ris	LM	1500			1500	5	300				
	R6	LM	1800			1800	4	450		•		1
	R6	LM	200			<b>20</b> 0	6	33				1
~	R6	LM	<b>84</b> 0			840	5	168				
	FI6	LM	460			460	3	153				İ
	Pi6	LM	660			660	3	220				1
	R6	LM	710			710	3	237				
ł	R6	LM	610			610	3	203				

#### TABLE 7-21 TOTAL MERCURY CONCENTRATIONS IN WHOLE BODY FISH, NORMALIZED FOR FISH AGE NYANZA III REMEDIAL INVESTIGATION MIDDLESEX COUNTY, MASSACHUSETTS PAGE 3

FINAL

/

	;						_		MEAN	MAX	MEAN	MAX
			INORO			ES	ST.	NORM.	NORM.	NORM.	NON-NORM	NON-NORM.
			MERCURY		MEDCHOV	IUI.	AL Ev ciel		CONC	I.MEHC.	I MERC.	T.MEHC.
1			CONC 1/		CONC	CON	IC AGE	GONC	BY SP/LOC	BY SP/LOC	BY SP/LOC	BY SPILOC
÷	LOCATION	SPECIES	(ug/kg)	REF#	(ua/ka)	REF# (ug/k	g) (vears)	(ua/ka/vr)	(ug/kg/yr)	(ug/kg/yr)	(ua/ka)	(ua/ka)
4												
	R6	LM	640			6	<b>4</b> 0 3	3 213				
1	R6	LM	554	3/	340	5	54 3	185				:
j	R6	LM	782	3/	480	7	82 3	3 261				
	R6	LM	685	3/	420	6	85 4	171				
	R6	LM	766	3/	470	7	66 4	192	414	6132	563	6132
1												
	R6	YP	590			5	90 7	84				1
1	R6	ΥP	200			2	00 3	67				,
	R6	YP	740			7	40 5	i 148				
ł	R6	YP	220			2	20 4	55				
ł	R6	YP	1000			10	00 7	143				
ļ	R6	YP	700			7	DO 6	117				•
i	R6	YP	620			6	20	155				
	R6	YP	200			2	00 5	40				i
	H6	YP	504	3/	300	5	04 5	101				1
1	H6	YP	538	3/	320	5	348 5	108				
ł	H6	ΥP	672	3/	400	6	72 6	112	103	155	544	1000 ;
÷												
	R9	LM	3200			32	90 S	356				
	R9	LM	2700			27	00 10	270				
	H9	LM	900			9		150				
	H9 Da	LM	2100			21	20 B	350				
	H9 Da	LM	1200			12		1/1				
	H9 Da	LM	2900			29		483				
	H9	LM	2500			25	90 9	278				
ł	H9 Do	LM	2700			27	8 00	338				
	H9 Doc	LM	2/00			2/0	90 91 70 7	300				
	H9 D00	LM	970			¥.		139				
	HSJ Doc	LM	250			2	50 Z	125				
	19	LM	040			۰ ۵	NU 2	420				
1	<b>F</b> 19	LM	230			2	50 Z	123				1
	Pro Pro		310			3		133				
	19	1.54	200			20	50 Z	140				i i
	100	1.54	320			3	n 2	100				
	60	1.14	550					550				1
	80	LM	250			~ ~		125				
	Dig	I M	1320	37	810	130	20 4	330				
	Re	I M	1271	3	780	12	71 5	254				
ł	Re	IM	1956	- 	1200	19	56 4	480				•
ľ	Ro	IM	1320	- 	#10	130	20 5	264				
•	Re	LM	1271	v	780	17	71 5	254				
	Re	I M	1239	- v	780	12	50 A	310				
	Re	I M	1614	3	990	16	4 3	538				
	Ro	LM	1320	- J	810	13	20 5	264				
1	Re	LM	1076	Sv	660	10	76 4	269				
	Ro	LM	1108	- J	680	11	18 4	277				
	RØ	LM	962	Ĵ	590	9	12 S	192				
	RØ	ŪM.	1076	Ĵ	860	10	ns 3	359				
	Ro	LM.	782	3V	480	71	2 2	391				
	RØ	ŪM	1011	37	620	10	4	253				
	Rip	LM	1011	3v	620	10	1 4	253				
	Ro	LM	701	SV	430	71	)1 3	234				
	Ro	LM	750	S.	480	7	50 4	167				
	Ro	LM	734	37	450	7	м з	245				
	Re	LM	734	3/	450	7	<u> </u>	183	271	550	1220	6132
				-								
	<b>Rg</b>	YP	739	3/	440	7:	<b>1</b> 9 4	165				
	Rg	YP	638	3/	380	6	36 4	160				
	Rg	YP	706	3/	420	71	6 6	118				
	Rg	YP	790	3/	470	7	- 	158				
1	Rg	YP	706	3/	420	7(	6 5	141				
ļ	R9	YP	806	3/	480	8	)6 5	161				
	R9	YP	554	3/	330	5	54 Š	111				
i	R9	YP	340			3	ю 5	68				
	Rg	YP	480			4	io 5	96				
	R9	YP	510			5	10 6	85				
- L					<u> </u>							

TABLE 7-21 TOTAL MERCURY CONCENTRATIONS IN WHOLE BODY FISH, NORMALIZED FOR FISH AGE NYANZA III REMEDIAL INVESTIGATION MIDDLESEX COUNTY, MASSACHUSETTS PAGE 4

FINAL

4

. 1								1	MEAN	MAX	MEAN	MAX
						EST.		NORM.	NORM.	NORM.	NON-NORM	NON-NORM.
			INORG.	METH	Ľ	TOTAL		TOTAL	T.MERC.	T.MERC.	T.MERC.	T.MERC.
H			MERCURY	MERCUI	iv 🛛	MERCURY	FISH	MERCURY	CONC.	CONC.	CONC.	CONC.
i			CONC. 1/	CON	<b>)</b> .	CONC.	AGE	CONC.	BY SP/LOC	BY SP/LOC	BY SP/LOC	BY SP/LOC
	LOCATION	SPECIES	(ug/kg)	REF# (ug/k	) REF	# (ug/kg)	(years)	(ug/kg/yr)	(ug/kg/yr)	(ug/kg/yr)	(ua/ka)	(ug/kg)
Ĭ		· · · · · · · · · · · · · · · · · · ·										
ł	R9	YP	660			660	5	132				
	R9	YP	710			710	6	118				
	R9	YP	740			740	5	148				
1	R9	YP	180			180	6	30				
	R9	YP	330			330	5	66				
1	R9	YP	920			920	5	184	347	6132	462	6132

____

.

# REFERENCE

1/ ANALYTICAL METHOD MEASURES INORGANIC MERCURY AND METHYLMERCURY

2/ VALUE BASED ON QUESTIONABLE SAMPLE - MAY NOT BE REPRESENTATIVE

3/ ESTIMATED FROM MEASURED METHYLMERCURY CONCENTRATION . (MEAN INORG MERCURY/MEAN METHYLMERCURY RATIO)

* B = BACKGROUND





L

FINAL

W92194F

7-84



Figure 7-6. Mean and Maximum Total Mercury Concentrations in Sediment and Largemouth Bass

FINAL

7-85

FINAL

Based on the data presented in Figures 7-5 and 7-6, total mercury concentrations in fish cannot be predicted accurately from total mercury concentrations in sediments. This finding is not unexpected; total mercury in sediments is not equally bioavailable from reach to reach, and bioavailability probably depends on the site-specific characteristics of sediments and surface waters.

## 7.6.5.1 <u>Risks to Food Chain Organisms</u>

Of the five biota COCs, chlordane and dieldrin were considered to be insignificant contributors to overall risk. Therefore, chlordane and dieldrin were not evaluated for food chain effects; total DDT, total PCB, and total mercury were evaluated.

Risks were evaluated for the three selected food chain predators (blue heron, raccoon, and osprey) that occupy upper trophic levels in aquatic-based food chains. The food chain model was used to estimate tissue burdens of the three selected biota COCs in organisms that occupy lower aquatic trophic levels. In summary, the food chain model and measured COC concentrations in fish were used to assess impacts <u>down</u> the food chain, while measured tissue concentrations in fish tissue and estimated exposure (consumption) scenarios were used to assess impacts <u>up</u> the food chain.

The evaluation of the proposed site-specific food chain, shown in Figure 7-2, requires the explicit statement of assumptions used for food chain analysis.

A major assumption is that the organisms selected to represent the various trophic levels in the site food chain adequately represent organisms within each trophic level. the other Although determination of risks to specific economically or ecologically important organisms due to the ingestion of contaminated prey might be of interest, the intended goal should include estimations of risks to all organisms in the food chain. Therefore, the selection of organisms used for food chain evaluations should be based on three criteria, if possible: (1) organisms that are ecologically or economically important, (2) organisms that are appropriate and representative for the trophic level assigned to them, and (3) organisms selected include those for which sufficient toxicity data exist.

It is inappropriate to select a representative food chain organism based solely on its economic value or human interest, especially if the organism's role (Table 7-20) in the food chain is difficult to assess or if species specific toxicity data are sparse. The organisms selected for assessing upper trophic level impacts include osprey, raccoon, and blue heron.

The following sections evaluate tissue concentrations of total mercury, total DDT, and total PCB in the three trophic levels below Level 4 fish. This assessment is based on a reach-specific basis, using largemouth bass data to represent Level 4 fish for all locations. Concentrations of the same COCs in the tissues of the three representative predators that consume Level 4 fish were estimated, using both measured fish tissue concentrations and estimates of consumption. This portion of the evaluation was not based on reach-specific data; reach-specific estimates were considered to be inappropriate because top-level predators (osprey, raccoon, and heron) are not restricted to single reaches or locations for foraging.

The impacts of home range, seasonal migrations, or seasonal dietary adjustments were not evaluated; however, their inclusion would likely decrease actual risks. Therefore, the risks discussed below, which assume limited home range and restricted diet (fish), are probably overestimates for most predators.

### Total Mercury

Lower Trophic Levels, Aquatic Food Chain

Non-normalized mercury concentrations in Level 4 fish (largemouth bass), adjusted for whole-body total mercury (ug/kg, wet weight), were used to estimate BAFs using the food chain model. Table 7-22 lists mean and maximum tissue concentrations of mercury in Level 1, These estimated values were compared to the 2, and 3 organisms. benchmark toxicity values for biota COCs, which are listed in Table Of the three biota COCs assessed in this process, only 7-16. mercury exceeded benchmark toxicity values; this occurred for concentrations of total mercury in tissues of Level 3 organisms (all reaches) and Level 2 organisms (Reaches 3, 4, and 9). Although the benchmark toxicity values established were intended for chemical concentrations in fish, the assessment assumed that ingestion of other organisms, such as Level 2 organisms, can result in similar risks to predators.

#### TABLE 7-22 ESTIMATED TISSUE CONCENTRATIONS OF BIOTA COCS FOR LEVEL 1, 2, AND 3 TROPHIC LEVELS BASED ON MEAN AND MAXIMUM MEASURED TISSUE CONCENTRATIONS IN LARGEMOUTH BASS NYANZA III REMEDIAL INVESTIGATION MIDDLESEX COUNTY, MASSACHUSETTS

	·····	MEASURED	MEASURED						
		LEVEL 4	LEVEL 4	MEAN	MAX	MEAN	MAX	MEAN	MAX
		FISH	FISH	EST.	EST.	EST.	EST.	EST.	EST.
		MEAN	MAX	LEVEL 1	LEVEL 1	LEVEL 2	LEVEL 2	LEVEL 3	LEVEL 3
		CONC.	CONC.	CONC.	CONC.	CONC.	CONC.	CONC.	CONC.
REACH	CHEMICAL	(ug/kg)	(vg/kg)	(ug/kg)	(ug/kg)	(ug/kg)	(ug/kg)	(ug/kg)	(ug/kg)
		746	1100	0.6		16.0	<b>95 7</b> [	108.0	171.0
			100	2.0	4.1	10.2	23.7	14.0	171.0
	TOTAL DOT	108	152	0.4	0.8	2.2	3.2	14.0	29.2
} .									
2	TOTAL MERCURY	1120	1120	3.8	3.8	24.3	24.3 [	162.3	162.3
3	TOTAL MERCURY	2870	7600	9.9	26.1	62.4	165.2	415.9	1101.4
3	TOTAL DDT	116	293	0.5	1.2	2.8	7.1	17.8	45.1
3	TOTAL PCB	249	708	0.8	2.3	5.2	1 <b>4</b> .8	35.6	1 <b>0</b> 1.1
4	TOTAL MERCURY	1470	4190	5.1	14.4	32.0	<b>91</b> .1	213.0	607.2
4	TOTAL DDT	9	114	0.0	0.5	0.2	2.8	1.4	17.5
4	TOTAL PCB	553	553	1.8	1.8	11.5	11.5	79.0	79.0
6		0401	1900	30	62	20.4	30.1	136.2	260.9
6		57	88	5.Z 0.2	0.2	14	21	8.8	13.5
6	TOTAL PCB	177	208	0.6	0.7	3.7	4.3	25.3	29.7
							_		
9	TOTAL MERCURY	1320	3200	4.5	11.0	28.7	69.6	191.3	463.8
9	TOTAL DDT	73	137	0.3	0.6	1.8	3.3	1 <b>1.2</b>	<b>21</b> .1
9	TOTAL PCB	405	780	1.3	2.5	8.4	16.3	57.9	111.4

BOXED VALUES EXCEED BENCHMARK TOXICITY VALUES FOR BIOTA COCS

Mercury concentrations in caddis fly larvae, which were collected as part of the benthic macroinvertebrate sampling plan, were not considered sufficiently reliable for a quantitative evaluation due to sample quantity limitations. However, such data can be used to evaluate the relative differences in tissue mercury concentrations between fish and invertebrate samples. Such data can be used, with caution, to validate the food chain model. Therefore, caddis fly larvae data were considered to be acceptable for rough estimates of actual tissue concentrations of mercury for these invertebrates.

Caddis fly larvae are primarily detrital feeders; therefore, they are assumed to occupy trophic level 2, represented by zooplankton, in the food chain model. Although uptake and depuration rates are unknown for caddis fly larvae, values associated with such rates are assumed to be similar to those of other Level 2 biota. Estimated mean mercury concentrations in Level 2 biota for the food chain model are approximately 16 to 62 ug/kg, wet weight, depending on location. Similar maximum values are predicted to range from approximately 24 to 165 ug/kg, depending on location.

Mercury tissue concentrations in caddis fly larvae, which should be interpreted cautiously, ranged from 100 to 1,000 ug/kg (0.10 to 1.0 mg/kg) dry weight. Based on the average percent solids (22.2%) for sampled caddis fly larvae, wet weight mercury tissue concentrations ranged from approximately 20 to 220 ug/kg, which compare closely with predicted concentrations.

In summary, the data suggest that mercury poses a potentially significant threat at several trophic levels to aquatic predators that prey on aquatic organisms.

Upper Trophic Levels, Representative Organisms

To assess risks to piscivorous terrestrial and avian biota, the mean and maximum measured or estimated whole body COC concentration for largemouth bass was used as the representative dietary source of COCs. Because most terrestrial and avian predators are not confined to reaches, assessing reach-specific risks was considered inappropriate, and the mean and maximum concentrations in largemouth bass from all reaches were used. Using the maximum COC concentration for assessing risk is likely to overestimate the actual risk associated with ingestion of contaminated fish; however, one focus of this assessment is to determine upper bound risk estimates based on possible exposures to food chain biota. The use of mean COC concentrations measured in fish tissue should provide a reasonable estimate of average risk to piscivorous predators.

Table 7-23 lists the mean and maximum estimated daily intake for each of the three potentially hazardous biota COCs.

W92194F

## TABLE 7–23 ESTIMATED DIETARY CONCENTRATIONS OF BIOTA COCS FOR REPRESENTATIVE PREDATORS NYANZA III REMEDIAL INVESTIGATION MIDDLESEX COUNTY, MASSACHUSETTS

	· • • ·				MEAN ESTIMATED	MAX
		MEAN	MAX	ESTIMATED	DAILY	DAILY
		CONC.	CONC.	INGESTION	INTAKE	INTAKE
CHEMICAL_	SPECIES	<u>(mg/kg)</u>	(mg/kg)	(kg fish/day)	(mg/kg/day)	(mg/kg/day)
MERCURY	RACCOON	1.411	3.182	0.45	0.63	1.43
DDT	RACCOON	0.086	0.164	0.45	0.04	0.07
РСВ	RACCOON	0.298	0.48	0.45	0.13	0.22
MERCURY	HERON	1.411	3.182	0.25	0.35	0.80
DDT	HERON	0.086	0.164	0.25	0.02	0.04
PCB	HERON	0.298	0.48	0.25	0.07	0.12
MERCURY	OSPREY	1.411	3.182	0.40	0.56	1.27
DDT	OSPREY	0.086	0.164	0,40	0.03	0.07
РСВ	OSPREY	0.298	0.48	0.40	0.12	0.19

MEAN CONC. BASED ON MEAN LARGEMOUTH BASS CONCENTRATION FOR ALL REACHES MAX CONC. BASED ON MAXIMUM LARGEMOUTH BASS CONCENTRATION FOR ALL REACHES Much of the available data on mercury contamination in fish has been compiled by the U.S. Fish and Wildlife Service (FWS, 1987a). An evaluation of those data indicates that the maximum estimated whole-body concentrations from fish at this site (7.6 mg/kg, not including the questionable yellow perch value from Reach 1) generally exceed most other reported values. Using combined muscle, viscera, and whole-body data from freshwater fish from various species and sites (polluted and unpolluted) throughout the United States and Canada, mercury concentrations in fish tissue are reported to range from 0.03 (muscle, largemouth bass, Illinois) to 23.7 mg/kg (liver, striped bass, Nevada). However, of the 60 values included in this compilation, only the 23.7 mg/kg value exceeded the maximum value measured in the present study.

The maximum value determined in this study is based on an actual measured concentration of total mercury in a filet sample, which is considered to be representative of the whole-body concentration of this individual fish. As stated above, this assumption is based on the derived V/F ratio of approximately 1.0 for both largemouth bass and yellow perch.

Potential adverse effects associated with high levels of mercury contamination on fish-eating predators can be assessed by basing estimates of the dietary intake of contaminated prey on fish ingestion rates that may result in adverse effects to predators. This approach was investigated; Table 7-23 lists the results.

Estimates of daily consumption rates for large adults of each of the three representative top-level predators are based on a synthesis of information obtained from FWS (1991) for ospreys, South Carolina Department of Wildlife and Marine Resources (1991) for ospreys, great blue herons, and raccoons, and from Webster (1985) for raccoons.

The FWS recommends that total mercury concentrations in fish should probably not exceed 100 ug/kg for protection of sensitive avian species (FWS, 1987a). Sensitive avian species include chickens, mallard ducks, American black ducks, and others. Fish collected and analyzed at this site, not including the questionable yellow perch collected in Reach 1, were estimated to contain mercury at concentrations up to nearly 80 times the recommended value, suggesting serious potential for adversely affecting avian predators.

An investigation of potential impacts to terrestrial predators (e.g., raccoons) reveals similarly high potential risks for mammalian predators that ingest mercury-contaminated fish. The FWS recommendation for maximum mercury concentrations in fish for the protection of small mammalian predators is 1,100 ug/kg wet weight.

W92194F

This value is exceeded more than 7-fold by the maximum estimated whole-body concentration in fish collected in this study.

Analysis of estimated dietary intakes suggests that serious concerns for the health of top-level predators are warranted. Some representative top-level predators, such as raccoons and herons, do not eat fish exclusively. However, such predators eat other aquatic vertebrates and invertebrates (e.g., frogs and crayfish) regularly. Because mercury contamination data are lacking for such potential prey items as frogs and crayfish, tissue concentrations of mercury in these organisms were assumed to be similar to those measured in fish. The assumption of a diet consisting exclusively of fish should not underestimate risks to raccoons, ospreys, or blue herons because fish are likely to bioaccumulate mercury in greater concentrations than organisms occupying lower trophic levels, such as crayfish.

Based on daily consumption rates for periods of two to five weeks, mallard ducks fed 0.1 mg mercury/kg wet weight in the diet experienced adverse reproductive behavior; juvenile starlings developed kidney lesions at 1.1 mg/kg total mercury; and pigeons exhibited behavioral alterations at 3.0 mg inorganic mercury/kg and 1.0 mg methylmercury/kg body weight (FWS, 1987a). These dietary intakes are based on daily consumption rates for periods of 2-5 weeks.

The maximum estimated daily intake of mercury from ingestion of contaminated fish for the representative top avian predators (1.27 mg Hg/kg, osprey) exceeds the reported values that produce adverse effects in tested avian species (0.1 to 1.1). These results suggest that estimated daily intakes can produce ecologically significant effects in representative species.

Terrestrial mammals are also sensitive to mercury poisoning from diet or oral administration. Cats fed 0.25 mg Hg/kg daily for 48 days (oral route) experienced increased incidence of abnormal fetuses; rats fed 0.5 mg Hg/kg daily (oral route) experienced reduced fertility; and mink exposed to dietary concentrations of 1.1 mg mercury/kg (exposure duration unavailable) showed signs of mercury poisoning (FWS, 1987a). The results of these studies suggest that daily ingestion prey associated with the calculated maximum mercury concentrations (1.43 mg Hg/kg) is likely to produce significant adverse effects in terrestrial mammals, represented by raccoons.

In summary, concentrations of mercury measured in Study Area fish should pose a significant hazard to aquatic, terrestrial, and avian biota in the Study Area.

### Total DDT

Lower Trophic Levels, Aquatic Food Chain

Non-normalized total DDT concentrations in Level 4 fish. represented by largemouth bass, adjusted for whole-body total mercury (ug/kg, wet weight), were employed to estimate BAFs using the food chain model. Table 7-22 lists mean and maximum tissue concentrations of total DDT in Level 1, 2, and 3 organisms. These estimated values were compared to the benchmark toxicity values for biota COCs, which are presented in Table 7-16. Estimated total DDT concentrations in Level 1 to 3 organisms did not exceed the benchmark toxicity value for DDT in any reach. Therefore, total DDT does not appear to present a significant risk to predators ingesting organisms represented by Levels 1 to 3 in the food chain model.

Upper Trophic Levels, Representative Organisms

To assess risks to fish-eating terrestrial and avian biota, the mean and maximum measured or estimated whole-body COC concentration for largemouth bass was used to represent the primary dietary source of COCs. As stated above, assessing reach-specific risks was considered to be inappropriate, and the mean and maximum concentrations in largemouth bass from all reaches were used in this evaluation.

Table 7-23 lists the mean and maximum estimated daily intake for each of the three potentially hazardous biota COCs. These values are comparable to values presented in the scientific literature. An evaluation of estimated DDT intakes that produced sublethal effects to birds or mammals consuming DDT-contaminated fish reveals relatively little recent data. Data on sublethal effects of DDT exposure to mammals and birds reveal behavioral effects observed in young mice when mothers were exposed to 2.5 mg/kg DDT (exposure type and duration unavailable), while liver DDT concentrations of 1.2 mg/kg resulted in adverse reproductive effects in mice (Stickel, 1975). Thinning of egg shells in various bird species (mallards, black ducks, American kestrels, and screech owls) has been demonstrated at dietary concentrations of 3 mg DDE/kg/day, wet weight, and pelican embryos die if eggs contain more than 2.5 mg DDE/kg (Stickel, 1975). Bats and possibly other mammals or birds appear most sensitive to DDT or DDE when they have limited fat reserves (Stickel, 1975).

This finding suggests that DDT or DDE sensitivity can vary according to nutritional state or season. Based on the limited data available, it appears reasonable to assume that DDT or DDE dietary concentrations exceeding approximately 2 to 3 mg/kg can

- W92194F

#### FINAL

adversely affect behavior or reproduction in at least some mammals and birds. The maximum dietary DDT concentrations expected for representative mammals and birds, which follow, are estimated to be well below concentrations that are expected to produce adverse effects.

If 2.5 mg/kg or greater dietary DDT is considered potentially hazardous for tested species, and a safety factor of 10 is applied for the protection of untested species, the estimated lowest effect dietary concentration is approximately 0.25 mg/kg. Maximum estimated dietary concentrations of DDT in representative organisms ranged from 0.04 to 0.07 mg/kg/day; comparable mean values ranged from 0.02 to 0.04 mg/kg/day. Because a safety factor of 10 might not be fully protective of all species, DDT contamination in site fish can pose a measurable hazard to fish-eating predators that are especially sensitive.

#### Total PCBs

Lower Trophic Levels, Aquatic Food Chain

Total PCB concentrations in Level 4 fish, represented by largemouth bass, adjusted for whole-body total mercury (ug/kg, wet weight), were used to estimate BAFs and tissue concentrations of Level 1 to 3 organisms. Table 7-22 lists mean and maximum tissue concentrations of total PCBs in Level 1, 2, and 3 organisms. These estimated values were compared to the benchmark toxicity values for biota COCs, presented in Table 7-16. Estimated total PCB concentrations in Level 1 to 3 organisms did not exceed the benchmark toxicity value for PCBs in any reach. Therefore, total PCBs do not appear to present a significant risk to predators ingesting Level 1 to 3 organisms.

Upper Trophic Levels, Representative Organisms

The maximum whole-body PCB concentration measured or estimated in sampled fish equalled 780 ug/kg. Studies investigating ecologically significant sublethal effects (impaired reproduction, growth, or behavior) in sensitive mammals and birds were used to derive recommended maximum acceptable PCB concentrations in diet for birds and mammals (FWS, 1986b). Depending on the study source, the recommended maximum dietary PCB value ranges from 100 to 640 ug/kg diet for mammals (mink), while the maximum value for birds is 3,000 ug/kg (FWS, 1986b). These recommended FWS values can be compared to estimated PCB intakes by representative birds and mammals in the Study Area based on fish ingestion.

Multiplying estimated daily dietary fish intakes (kg fish/day) for representative mammalian species by mean and maximum whole-body PCB burdens in fish (mg PCB/kg fish), reveals that representative

mammals (racoon) can consume approximately 0.1 to 0.2 mg PCB/kg/day, depending on species. The estimated mean and maximum daily intake of PCBs due to fish ingestion (130 to 220 ug/kg) from mammals are below FWS recommendations (100-640 ug/kg) for some mammalian species.

The estimated maximum daily PCB intake for representative bird species (190 ug/kg/day, oprey; 120 ug/kg/day, herons) are much less than the value recommended by FWS (3,000 ug/kg), suggesting reasonably safe levels for birds. Based on FWS recommendations, it appears reasonable to assume that while birds might not be at serious risk from ingesting PCo-contaminated fish taken from study area waters, significant risks may be posed to some sensitive mammalian species, such as mink.

#### 7.6.6 Uncertainty Analysis, Risk Characterization

Because risk characterization is essentially the integration of the exposure assessment and hazard assessment, sources of uncertainty associated with either of these elements should also contribute to uncertainty in risk characterization. In addition, the risk characterization procedure itself should contribute to overall uncertainty. Except for the food chain evaluation, the quotient method was selected as the risk characterization method of choice for this assessment. The advantages of this method, and one of the primary limitations associated with this method, were previously addressed.

Additional limitations of the quotient method, according to EPA (1989b), include the following:

- 1. EPA-reviewed toxicity data are available for only a limited number of chemicals.
- 2. Chronic toxicity endpoint data can be inconsistent.
- 3. Toxicant interactions are not addressed.
- 4. Toxicity data are sparse for media other than surface waters.
- 5. Analytical detection limits commonly exceed toxicity benchmark values (i.e., criteria).
- 6. No means for estimating severity of impacts if benchmark toxicity values are exceeded.

Decreasing the level of uncertainty associated with each of the limitations described above was accomplished using a variety of processes. A brief response to each of these limitations follows:

W92194F

- 1. The use of acceptable chemical quantitative structure activity relationships should provide reasonable estimates of toxicity data for untested chemicals.
- 2. Selecting chronic toxicity test results based only on appropriate endpoints (e.g., effects on mortality, growth, and reproduction), test design, and test durations should decrease the uncertainty associated with chronic test results.
- 3. The method of Barnthouse et al. (1986), which simply sums quotients and addresses cumulative toxicity, addresses toxicant interactions in a reasonable and consistent manner, based on the generally accepted principle of chemical additivity.
- 4. Sufficient toxicity data for media other than surface water generally exist; when combined with extrapolations based on chemical structure activity relationships or interspecies correlations, reasonable estimates of required data are possible.
- 5. A reasonable, conservative, and protective approach for dealing with relatively high detection limits and low "safe" chemical concentrations includes setting the environmental concentration of the chemical to one half the detection limit. This procedure probably results in overestimations of actual environmental concentrations of chemicals of concern, but is reasonable in view of analytical limitations.
- 6. The severity of ecological impacts expected from exceedences of toxicity benchmark values (e.g., chronic ambient water quality criteria) can be estimated using the cumulative method of assessing toxicant additivity.

Several sources can contribute to total uncertainty associated with the food chain evaluation. Major and specific categories of potential uncertainty can include the following:

- 1. Fish sample data
  - a. Collection of appropriate organisms
  - Collection and analysis of sufficient quantity of organisms
  - c. Adherence to approved sample handling and storage protocols
  - d. Accurate analysis of samples

#### 2. Food chain model

- a. Use of appropriate chemical-specific Kow values
- b. Use of appropriate biota weights and lipid content
- c. Identification of appropriate trophic levels
- d. Application of appropriate model for this study site
- 3. Representative species
  - a. Use of appropriate representative species
  - b. Use of reasonable estimates of fish ingestion
  - c. Applicability of limited toxicity data to representative species

Every effort was taken to ensure that fish sampling, food chain modeling, and the selection of appropriate representative species were performed in the most appropriate manner for this risk assessment. Although all of the above-mentioned categories probably contribute to total uncertainty to some extent, items relating to fish sample data are potentially the most significant. Because this particular study area encompassed a large geographical area, relatively limited fish sampling was performed; and none of the analyses consisted whole-body analysis for any given fish. However, the species selected and the quantity of each species collected appear to be adequate for site characterization. With the possible exception of compromise to the integrity of some Phase II samples, which was discussed previously, the level of uncertainty was considered acceptable for this phase of the ecological risk assessment.

Data collection components that can be useful for some ecological risk assessments, but were not performed for this assessment, include (1) detailed macroscopic and microscopic tissue analysis of aquatic and terrestrial biota, and (2) toxicity testing using study area waters and sediments. However, based on the extensive biological and chemical sampling incorporated into this assessment, it is considered that such additional procedures were unnecessary at this time.

In summary, several sources of uncertainty might contribute to overall uncertainty in the final risk estimates, including those sources discussed in the exposure and hazard sections of this assessment. Throughout this assessment, if levels of uncertainty were unknown, or if impacts associated with uncertainty could not be estimated accurately, a conservative approach was taken. The consistent use of conservative assumptions probably overestimated actual risk to biota in nearly all cases, but no appropriate or reasonable alternative to conservatism has been identified.

- W92194F

## 7.7 Ecological Risk Assessment Summary and Conclusions

Although many inorganic and organic chemicals were detected in various media at study area locations, only a few chemicals were found at concentrations that are a cause for concern. The primary contaminants determined in this study are chemicals associated with sediments and biota. Risks to biota from surface water COCs are minimal in comparison to those from COCs associated with sediments and biota.

FINAL

The primary Site related and Study Area contaminant is mercury, which is considered to be a site-related contaminant. This chemical contributed the major portion of the estimated risk from both contaminated sediments (except for aluminum) and contaminated biota. The risk estimates for sediment-associated aluminum which is not considered to be a site-related contaminant are probably overestimated, based on bioavailability. DDT and its metabolites also contributed significantly to the overall risk estimates associated with contaminated sediments. In some locations, phthalate esters (non-site related contaminants) also were important contributors to sediment toxicity.

The predominant contaminant for biota was mercury, followed by PCBs and DDT and its degradation products (DDD and DDE) which are not considered to be sit-related contaminants. This Study Area is somewhat unique because the most hazardous chemicals (mercury, PCBs, DDT) are those that have the greatest effects on food chains/webs. The toxicity hazards associated with these primary study area contaminants are minimal compared to hazards associated with food chain effects. Although some exceptions to this pattern were noted, in general, the least hazardous concentrations of sediment COCs were associated with Background areas, including Reach 1, while the most hazardous concentrations were associated with Reach 2 and, especially, Reach 3; the Eastern Wetlands, Outfall Creek; and the Raceway. Risk estimates based on biota COCs were more evenly distributed throughout sampled reaches in comparison to risk estimates based on sediment COCs. Cumulative risks associated with biota COCs were highest in Reach 3; the risks estimated for the other reaches were considerably lower than those of Reach 3.

In conclusion, the predominant hazards identified by this Ecological Risk Assessment in the Study Area are the adverse effects to upper trophic level predators that ingest contaminated fish and invertebrates and, to a lesser extent, the aqueous toxicity of such contaminants. The sediments that are contaminated with mercury, PCBs, and total DDT should be a source of contamination for some time, due to the persistent nature of these contaminants.

#### 8.0 CONCLUSION

## 8.1 <u>Contaminants in Water and Sediment</u>

Results of this Remedial Investigation indicate that past process wastewater and chemical disposal practices at the Nyanza Site have resulted in surface water, sediment and soil contamination in the wetlands adjacent to the Nyanza Site and in the Sudbury River.

Mercury and chromium contamination is extensive in the sediments and soils of the Eastern Wetlands, a receptor of Site runoff. Other organic Site related contaminants are present in Wetlands sediments and underlying soils including chlorobenzene, dichlorobenzene, trichloroethene, and dichloroethene. The highest concentrations of contaminants in the Study Area by one to three orders of magnitude occur in the Eastern Wetlands. Site related contaminants are also present in the Wetlands surface waters.

Trolley Brook drains the Wetlands and surface water mixes with Chemical Brook before flowing into the Chemical Brook Culvert, and finally outfalls into the Raceway. The Raceway carries contaminated sediments into the Sudbury River.

Mercury and chromium contamination is pervasive in sediments deposited by the streams draining the Eastern Wetlands. These and other inorganic contaminants are generally adsorbed to sediment particles and move throughout the Study Area by sediment transport mechanisms. In that manner, the contaminated sediments are retained in the depositional areas, until resuspension occurs during periods of high water, causing further downstream transport. Samples collected in the downgradient Reach 10 of the Study Area are comparable to background levels detected upstream of the Nyanza Site.

Dissolution of the adsorbed inorganics to the water is not expected to occur under water quality conditions which were found to occur throughout the River. Relatively low concentrations of Siterelated contaminants in surface waters compared to high concentrations in sediments indicated that the primary contaminant transport mechanism is sediment transport.

Volatile organic contaminants noted above are present in a pattern indicating the location of a possible ancestral channel through the Eastern Wetlands. Highest contaminant concentrations are present in the northern pond, and the concentrations quickly drop near the outlet of this pond. It is expected that as turbulence increases in the Chemical Brook Culvert, these compounds are volatilized. In this manner they are not persistent in the River. Volatile organic compound contamination is not widespread in the river.

Organic compounds which were noted in the Work Plan to be compounds specific to the Nyanza dye manufacturing process ("Site specific organics": aniline, napthalamines and benzidines) were not detected in the Study Area.

The Sudbury River has an extensive history of industrialization and urbanization and has been impacted by numerous chemical and physical processes. Numerous chemicals were detected in the river system, including PAH compounds, phthalates, pesticides, PCBs, volatile organic compounds, and several metals. These chemicals are generally understood to be non-site related compounds attributable to other point and non point sources.

## 8.2 <u>Contaminants in Fish</u>

Fish samples revealed the presence of bioaccumulated metals and organic compounds. Most notable was mercury, which is also present in fish tissue in the form of methylmercury. The other organic compounds detected were primarily pesticides and PCBs, which are compounds readily accumulated by predatory fish. Concentrations of mercury and pesticides in fish tissues generally increase with age and size, and are generally higher in the predator group than in the scavenger group.

Sample location was also a factor in the concentrations of mercury in fish. Reservoir 2, which exhibits the highest concentrations of mercury in sediment, also showed the highest concentrations of mercury in fish tissue. Those fish collected downstream contained gradually decreasing concentrations of mercury. It is voected that the dams and impoundments are preventing the "eam migration of fish and their prey, and therefore ry concentrations in fish should mirror bioavailable mercury 1. ...e fish habitat.

## 8.3 <u>Risk Assessment</u>

The human health risk assessment concludes that there are two categories of individuals which are susceptible to health effects of contaminants found in the Study Area. These individuals are those who; regularly eat fish captured from the Study Area; and use parts of the Study Area for swimming, wading or other intrusive recreational activities.

With few exceptions, hazard quotients (HQs) and hazard (HIs) indices calculated for the recreational and residential sediment exposure scenarios do not exceed unity. However, hazard indices calculated for the COC concentrations detected in the Eastern Wetlands sediments (recreational exposure scenarios) exceed unity when maximum contaminant concentrations are evaluated and a small child is considered the receptor of concern. (In this case, the

W92194F

8-2

hazard index calculated for chemicals affecting the kidney approaches unity (>0.95.)).

In all cases presented for the sediment exposure scenarios, HQs and HIs calculated for the accidental-ingestion exposure route exceed those calculated for the dermal-absorption exposure route. The hazard index calculated for the bordering wetlands (residential exposure scenario) exceeds unity when maximum concentrations are evaluated and a child is considered as the receptor of concern. However, hazard indices calculated on a target organ-specific basis do not exceed unity.

With the exception of fish sampled in Southville Pond (in Reach 1), maximum and/or average mercury concentrations detected in fish tissue samples collected in the Study Area exceed the FDA Action Level for mercury in fish. The fish-ingestion exposure scenarios presented in the risk assessment considered a sports fisherman and subsistence fisherman as receptors of concern. Hazard indices calculated for all COCs detected in fish tissue samples collected during the RI exceed unity in at least one of the cases presented for each surface water body evaluated.

The fact that HQs and HIs exceed unity and the fact that mercury concentrations exceed FDA Action Levels in one or more cases presented for each surface water body evaluated suggests that adverse noncarcinogenic health effects are anticipated for the sports fisherman and subsistence fisherman under the condition of the exposure scenarios considered in the risk assessment. With the exception of the Saxonville impoundment, hazard indices calculated for Nyanza Site contaminants exceed those calculated for "other Sudbury River contaminants."

Cancer risk estimated for the fish-ingestion exposure scenarios range from 0 (Cedar Swamp Pond and Southville Pond, no risk calculated; CSFs not available for COCs) to 5.5 x  $10^3$  (Reservoir No. 2, maximum COC concentrations, subsistence fisherman). As a point of reference, cancer risks for samples collected in the Sudbury Reservoir are 5 x 10⁴ and 3.6 x 10⁴ when maximum and average COC concentrations are evaluated and the subsistence fisherman is considered the receptor of concern. Cancer risks estimated for surface water bodies upstream of the Nyanza Site do not exceed 1 x 10⁹ in any case presented. Cancer risks estimated for Mill Pond, Reservoir No. 2, and Fairhaven Bay exceed 1 x 10⁹ in at least one case presented. The principal COCs contributing to the estimated excess lifetime cancer risks are arsenic, several pesticides such as 4,4-DDT, and the PCBs. In all cases presented, risks associated with "other Sudbury River contaminants" exceed those estimated for Nyanza Site contaminants.

W92194F

The predominant contaminant in biota is mercury, followed by PCBs and DDT and its degradation products DDD and DDE. The toxicity hazards associated with these contaminants are probably minimal compared to hazards associated with food chain effects. In general, mean and maximum estimated whole body concentrations of total mercury exceed values that are considered to be protective of humans and nonhuman predators that consume fish. Most fish collected from all Reaches, except background (Reach 1), are associated with mercury concentrations which exceed Food and Drug Administration action levels and Fish and Wildlife Service guidelines. The risks to fish consumers is expected to be greatest for fish collected from Reach 3, followed by Reach 4 and Reach 9.

In general, the surface water contaminants associated with the greatest risks to aquatic animals in most reaches are aluminum, lead, zinc, and bis(2-ethylhexyl)phthalate. Silver appears to pose a significant risk to aquatic biota only in Reach 7. A risk to birds and terrestrial animals is associated with drinking surface waters, from the Outfall Creek and Cold Spring Brook due to elevated concentrations of zinc and bis(2-ethylhexyl)phthalate. This risk is minimal through the other Reaches. For most Reaches, aluminum poses the greatest risk to aquatic plants.

#### FINAL

8-4

.

REFERENCES

## References

Barnthouse, L.W., G.W. Suter, S.M. Bartell, J.J. Beauchamp, R.H. Gardner, E. Linder, R.V. O'Neill, and A.E. Rosen, 1986. <u>User's Manual for Ecological Risk Assessment</u>. Environmental Sciences Division Publication No. 2679. Oak Ridge National Laboratory, Oak Ridge, TN.

Bolton, S.H, 1985. National Perspective on Sediment Quality. Battelle, Washington, D.C.

Borman, S, 1990. "New QSAR Techniques Eyed for Environmental Assessments". Science and Technology, February 19, 1990. Chemistry and Environment.

Camp, Dresser, and McKee, Inc. (CDM), 1982. Remedial Action Master Plan Draft. June 23, 1982.

Camp, Dresser, and McKee, Inc. (CDM), 1987. Report of Field Work Completed July 16 through August 22, 1986: Remedial Design for Nyanza Chemical Site. EPA Contract No. 68-01-6939, Work Assignment No. 84-1L15.4; Final Draft March 23, 1987.

Cardwell, R.D., 1989. An Overview of Aquatic Ecological Risk Assessment Methodologies. OCEANS 89 Proceedings. Marine Technological Society of Washington DC. Volume 2. 1989.

Corbitt, R.A., 1990. <u>Standard Handbook of Environmental</u> <u>Engineering</u>. McGraw-Hill Publishing Company. New York, NY.

DiToro, D.M., J.D. Mahony, P.R. Kirchgraber, A.L. O'Byrne, L.R. Pasquale, and D.C. Piccirilli, 1986. "Effects of Nonreversibility, Particle Concentration, and Ionic Strength on Heavy Metal Sorption". Environ. Sci. Technol. 20:55-61.

DiToro, D.M., J.D. Mahony, K.J. Scott, M.B. Hicks, S.M. Mayer, and D.J. Hansen, 1989. Toxicity of Cadmium in Sediments: The Role of Acid Volatile Sulfide. Proceedings of the Tenth Annual Meeting of the Society of Environmental Toxicology and Chemistry. Toronto, Ontario, Canada. October 28-November 2, 1989.

DiToro, D.M., J.D. Mahony, D.J. Hansen, K.J. Scott, A.R. Carlson, and G.T. Ankley, 1990. "Acid Volatile Sulfide Predicts the Acute Toxicity of Cadmium and Nickel in Sediments". Submitted to Science, May 16, 1990.

Doherty, F.G, 1983. "Interspecies Correlations of Acute Aquatic Median Lethal Concentrations for Four Standard Testing Species". Environ. Sci. Technol. 17:661-665.

Duxbury, A.S., 1988. The Development of Sediment Toxicant Criteria and Their Impact on Human Health. Independent Study. Medical Science.

Ebasco, 1989. Draft Final Work Plan, Nyanza III - Subbury River Study. Middlesex County, Massachusetts. EPA Work Assignment No. 151-1L15, EPA Contract No. 68-01-7250. May, 1989.

Ebasco, 1991a. Draft Final Remedial Investigation Report, Nyanza II - Groundwater Study. Ashland, MA. EPA Work Assignment No. 006-1L15, EPA Contract No. 68-W9-0034. April, 1991.

Ebasco, 1991b. Draft Final Feasibility Study Report, Nyanza II -Groundwater Study. Ashland, MA. EPA Work Assignment No. 006-1L15, EPA Contract No. 68-W9-0034. June, 1991.

Ecological Analysts, Inc., 1981. <u>The Sources, Chemistry, Fate and</u> <u>Effects of Chromium in Aquatic Environments</u>. Avail. from American Petroleum Institute. Washington, D.C.

Federal Register. Vol. 54, No. 97, Monday, May 27, 1989.

Friedman, G.F. and J.E. Sanders, 1978. <u>Principles of Sedimen-</u> tology. John Wiley & Sons, Inc. New York, NY.

Hardisty, M.W., R. J. Huggins, S. Kartar, and M. Sainsbury, 1974. "Ecological Implications of Heavy Metal in Fish from the Severn Estuary". Mar. Poll Bulll.,5 (1):12-15.

Howard, P.H., 1989. <u>Handbook of Environmental Fate and Exposure</u> <u>Data for Organic Chemicals - Volume I Large Production and Priority</u> <u>Pollutants</u>. Lewis Publishers, Inc. Chelsea, MI.

Howard, P.H., 1990. <u>Handbook of Environmental Fate and Exposure</u> <u>Data for Organic Chemicals - Volume II Solvents</u>. Lewis Publishers, Inc. Chelsea, MI.

Howard, P.H. R.S. Boethling, W.F. Jarvis, W.M. Meylan, E.M. Michalenko, 1991. <u>Handbook of Environmental Degradation Rates</u>. Lewis Publishers, Inc. Chelsea, MI.

Hynes, H.B.N., 1970. The Ecology of Running Water. Liverpool University Press, Liverpool, United Kingdom.

Integrated Risk Information system. July 1989.

Jerneloev, A., R. Hartung, P.B. Trost, and R.E. Bisque, 1972. "Environmental Dynamics of Mercury". Pages 167-201 in R. Hartung and B.D. Dinman (eds.). Environmental mercury contamination. Ann Arbor Science Publ. Ann Arbor, Michigan. Lick, W., 1984. The Transport of Sediments in Aquatic Systems. In Fate and Effects of Sediment-Bound Chemicals in Aquatic Systems. Eds. K.L. Dickson, A.W. Maki, and W.A. Brungs. Proceedings of the Sixth Pellston Workshop, Florissant, CO. August 12-17, 1984. SETAC Special Publication Series, Pergamon Press New York, NY.

Long, E.R. and L.G. Morgan, 1989. The Potential for Biological Effects of Sediment-Sorbed Contaminants Tested in the National Status and Trends Program. NOAA Technical Memorandum NOS OMA 52. National Oceanic and Atmospheric Administration. Seattle, WA.

Lyman, W.J., W.F. Reehl, and D.H. Rosenblatt, 1990. Handbook of Chemical Property Estimation Methods. Environmental Behavior of Organic Compounds. American Chemical Society. Washington, DC.

MDC (Metropolitan District Commission, Water Division), 1982. Sudbury Reservoir Water Treatment Plant, Draft Environmental Impact Report. By Parsons, Brinckerhoff, Quade, and Douglas, Inc. February 1982.

Massachusetts Department of Environmental Quality Engineering. (DEQE), 1980. Nyanza Preliminary Site Assessment Report. October 23, 1980.

Massachusetts Department of Environmental Quality Engineering, 1987. Sudbury River Basin, 1986 Water Quality Survey and Wastewater Discharge Data. Division of Water Pollution Control, Technical Services Branch. Publication No. 15, 062-50-50-10-87-CR. September 1987.

MWPC (Massachussetts Department of Environmental Quality Engineering, Division of Water Pollution Control, Technical Services Branch), 1982. The Sudbury-Assabet-Concord Basin Water Quality Management Plan. Westborough, MA. January, 1982.

Novotny, V., and G. Chesters, 1981. Handbook of Nonpoint Pollution. Van Nostrand Reinhold Company, New York, New York.

NUS Corporation, 1985a. Draft Phase I Remedial Investigation/ Feasibility Study Report, Volumes I and II, Phase I Remedial Investigation. Nyanza Chemical Site, Middlesex County.

NUS Corporation, 1985b. Draft Phase I Remedial Investigation/ Feasibility Study Report, Volumes I, Phase I Remedial Investigation. Nyanza Chemical Site, Middlesex county Massachusetts. EPA Work Assignment No. 11-1115.1, Contract No. 68-01-6699. March 1985.

NUS Corporation, 1985c. Draft Phase I Remedial Investigation/ Feasibility Study Report, Volumes II, Phase I Feasibility Study Nyanza Chemical Site, Middlesex county Massachusetts. EPA Work Assignment No. 11-1115.1, Contract No. 68-01-6699. March 1985.

NUS Corporation, 1989a. Technical Assistance: Residential Sampling, (Wet Basement Report). TDD No. F1-86611-40, Reference No. 375MA09; CERCLA No. MAD990685422. January 26, 1987.

NUS Corporation, 1989b. Final Work Plan for Operable Unit III -The Sudbury River Study. August 1989.

NUS Corporation, 1989c. Field Sampling and Analysis Plan for Operable Unit III - The Sudbury River Study. August 1989.

NUS Corporation, 1990a. Final Sampling and Analysis Plan for Operable Unit III - The Sudbury River Study. October 1990.

NUS (Halliburton NUS Environmental Corporation), 1990b. Final Technical Directive Memorandum for Phase I Activities of the Remedial Investigation/Feasibility Study, Nyanza Operable Unit III - Sudbury River Study. Middlesex County, MA. EPA Work Assignment No. 02-1L15, Contract No. 68-W8-0117.

Parker, G.W. and Izbicki, J.A. Bathymetry and Sediment Distribution in Reservoirs Nos. 1 and 2. Framingham MA. United States Geological Survey. Report in preparation.

Powell, John D, 1987. Nyanza: A Small American Town and its Superfund Hazardous Waste Site. Department of Political Science, Tufts University. Medford, Massachusetts. July, 1987.

Protic. M. and A. Sabljic, 1989. "Quantitative Structure-Activity Relationships of Acute Toxicity of Commercial Chemicals on Fathead Minnows: Effect of Molecular Size". Aquatic Toxicology. 14:47-64.

Sax, RI and RJ Lewis, 1991, Dangerous Properties of Industrial Materials - Seventh Edition.

SCS (Soil Conservation Service and US Department of Agriculture), 1973. Flood Hazard Analyses Upper Sudbury River. Acton, MA. June, 1973.

Schuurman, G. and W. Klein, 1988. Advances in Bioconcentration Prediction. Chemosphere. Vol. 17, No. 8.

Sidwell, V., 1981. Chemical and Nutritional Composition of Finfishes, Whales, Crustaceans, Mollusks, and Their Products. NOAA Technical Memorandum NMFS F/SEC-11. U.S. Dept. of Commerce. National Oceanic and Atmospheric Administration. National Marine Fisheries Service. Springfield, VA.

South Carolina Wildlife and Marine Resources Department, 1991. Columbia, SC. Personal communication, Pat Ferrel, Wildlife Biologist. Stickel, W.H., 1975. Some Effects of Pollutants in Terrestrial Ecosystems. In Ecological Toxicology Research. Effects of Heavy Metal and Oganohalogen Compounds. A.D. McIntyre and C.F. Mills, eds. Plenum Press. New York, NY.

SRC (Syracuse Research Corporation), 1990. CHEMFATE Database. Syracuse, New York.

Suter, G.W. and A.E. Rosen, 1986. Comparative Toxicology of Marine Fishes and Crustaceans. Publication No. 2804. Environmental Sciences Division. Oak Ridge National Laboratories.

Thomann, R.V., 1989. "Bioaccumulation Model of Organic Chemical Distribution in Aquatic Food Chains". Environmental Science and Technology. Vol. 23, page 699. June 1989.

U.S. Department of Forestry and Wildlife Management, 1981. Massachusetts Map Down: 1971 Land Use and Vegetative Cover Mapping.

U.S. Environmental Protection Agency, 1979a. Water-Related Environmental Fate of 129 Priority Pollutants. EPA 440/4-79-029a and b. Volume I and Volume II. Office of Water Planning and Standards. Office of Water and Waste Management. Washington, DC.

U.S. Environmental Protection Agency, 1979b. Reviews of the Environmental Effects of Pollutants: VII Lead. EPA 600/1-78-029. USEPA Office of Research and Development. Washington, D.C.

U.S. Environmental Protection Agency, 1980a. Ambient Water Quality Criteria for Acenaphthene. EPA 440/5-80-015. Office of Water Regulations. Criteria and Standards Division. Washington, DC.

U.S. Environmental Protection Agency, 1980b. Ambient Water Quality Criteria for Aldrin/Dieldrin. EPA 440/5-80-019. Office of Water Regulations. Criteria and Standards Division. Washington, DC.

U.S. Environmental Protection Agency, 1980c. Ambient Water Quality Criteria for Chlorinated Benzenes. EPA 440/5-80-028. Office of Water Regulations. Criteria and Standards Division. Washington, DC.

U.S. Environmental Protection Agency, 1980d. Ambient Water Quality Criteria for DDT. EPA 440/5-80-038. Office of Water Regulations. Criteria and Standards Division. Washington, DC.

U.S. Environmental Protection Agency, 1980e. Ambient Water Quality Criteria for Dichlorobenzenes. EPA 440/5-80-039. Office of Water Regulations. Criteria and Standards Division. Washington, DC.

U.S. Environmental Protection Agency, 1980f. Ambient Water Quality Criteria for Endrin. EPA 440/5-80-047. Office of Water Regulations. Criteria and Standards Division. Washington, DC.

U.S. Environmental Protection Agency, 1980g. Ambient Water Quality Criteria for Polychlorinated Biphenyls. EPA 440/5-80-068. Office of Water Regulations. Criteria and Standards Division. Washington, DC.

U.S. Environmental Protection Agency, 1980h. Ambient Water Quality Criteria for Polynuclear Aromatic Hydrocarbons. EPA 440/5-80-069. Office of Water Regulations. Criteria and Standards Division. Washington, DC.

U.S. Environmental Protection Agency, 1980i. Ambient Water Quality Criteria for Phthalate Esters. EPA-440-5-80-067. Office of Water Regulations and Standards. Criteria and Standards Division. Washington, DC 20460.

U.S. Environmental Protection Agency, 1982. Nyanza Hazardous Waste Site, Ashland, Massachusetts. Aerial Photography Survey Office of Research and Development, Environmental Monitoring Systems Laboratory, TS-PIC-82063; by Peter Stokely, Bionetics Corporation, Contract No. 68-03-3161. November 1982.

U.S. Environmental Protection Agency, 1984. Mercury Health Effects Update. EPA 600/8-84-019F. USEPA Office of Health and Environmental Assessment. Washington, D.C.

U.S. Environmental Protection Agency, 1985a. Ambient Water Quality Criteria for Arsenic - 1984. EPA 440/5-84-033. Office of Water Regulations. Criteria and Standards Division. Washington, DC.

U.S. Environmental Protection Agency, 1985b. Ambient Water Quality Criteria for Cadmium - 1984. EPA 440/5-84-032. Office of Water Regulations. Criteria and Standards Division. Washington, DC.

U.S. Environmental Protection Agency, 1985c. Ambient Water Quality Criteria for Chromium - 1984. EPA 440/5-84-029. Office of Water Regulations. Criteria and Standards Division. Washington, DC.

U.S. Environmental Protection Agency, 1985d. Ambient Water Quality Criteria for Copper - 1984. EPA 440/5-84-031. Office of Water Regulations. Criteria and Standards Division. Washington, DC.

U.S. Environmental Protection Agency, 1985e. Ambient Water Quality Criteria for Cyanide - 1984. EPA 440/5-84-028. Office of Water Regulations. Criteria and Standards Division. Washington, DC.

U.S. Environmental Protection Agency, 1985f. Ambient Water Quality Criteria for Lead - 1984. EPA 440/5-84-027. Office of Water Regulations. Criteria and Standards Division. Washington, DC.

U.S. Environmental Protection Agency, 1985g. Ambient Water Quality Criteria for Mercury - 1984. EPA 440/5-84-026. Office of Water Regulations. Criteria and Standards Division. Washington, DC.

U.S. Environmental Protection Agency, 1985h. Guidelines for Deriving Numerical National Water Quality Criteria for the Protection of Aquatic Organisms and Their Uses. Office of Water Regulations and Standards. Criteria and Standards Division. Washington, DC.

U.S. Environmental Protection Agency, 1985i. Technical Support Document for Water Quality-based Toxics Control. Office of Water Regulations and Standards. Washington, DC.

U.S. Environmental Protection Agency, 1986a. Ambient Water Quality Criteria for Parathion. EPA 440/5-86-007. Office of Water Regulations. Criteria and Standards Division. Washington, DC.

U.S. Environmental Protection Agency, 1986b. Ambient Water Quality Criteria for Pentachlorophenol. EPA 440/5-86-009. Office of Water Regulations. Criteria and Standards Division. Washington, DC.

U.S. Environmental Protection Agency, 1986c. Ambient Water Quality Criteria for Toxaphene. EPA 440/5-86-006. Office of Water Regulations. Criteria and Standards Division. Washington, DC.

U.S. Environmental Protection Agency, 1986d. Superfund Public Health Evaluation Manual. EPA/590/1-86/060. USEPA Office of Emergency and Remedial Response. Washington. D.C.

U.S. Environmental Protection Agency, 1986, 1987. Quality Criteria for Water. EPA 440/5-86-001.

U.S. Environmental Protection Agency, 1987a. Data Quality Objectives for Remedial Response Activities, Volume I - Development Process. Office of Emergency and Remedial Response and Office of Waste Programs Enforcement: EPA 540/G-87/003A (OSWER Directive 9335.0-7B). March 1987.

U.S. Environmental Protection Agency, 1987b. Record of Decision, Remedial Alternative Selection; Nyanza Chemical Waste Dump Site.

U.S. Environmental Protection Agency, 1987c. Data Quality Objectives for Remedial Response Activities. OSWER Directive 9355.0-78.

U.S. Environmental Protection Agency, 1987d. Ambient Water Quality Criteria for Selenium. EPA 440/5-87-006. Office of Water Regulations. Criteria and Standards Division. Washington, DC.

U.S. Environmental Protection Agency, 1988a. Ambient Water Quality Criteria for Aluminum. EPA 440/5-88-008. Office of Water Regulations. Criteria and Standards Division. Washington, DC. U.S. Environmental Protection Agency, 1988b. Superfund Exposure Assessment Manual. EPA/540/1-88/001. Office of Remedial Response. Washington DC.

U.S. Environmental Protection Agency, 1988c. Draft Guidance for Conducting Remedial Investigations and Feasibility Studies Under CERCLA. Office of Emergency and Remedial Response and Office of Solid Waste and Emergency Response (OSWER Directive 9335.3-01, Contract 68-01-7090). March 1988.

U.S. Environmental Protection Agency, 1988d. Draft Guidance for Conducting Remedial Investigations and Feasibility Studies Under CERCLA. Draft (OSWER Directive 9335.3-01). March 1988.

U.S. Environmental Protection Agency, 1988e. Review of Ecological Risk Assessment Methods. Office of Policy Analysis; EPA/230-10-88-041. November 1988.

U.S. Environmental Protection Agency, 1989a. Risk Assessment Guidance for Superfund: Vol. II Environmental Evaluation Manual Interim Final. EPA/540/1-89/001. USEPA Office Emergency and Remedial Response. Washington, D.C.

U.S. Environmental Protection Agency, 1989b. Supplemental Risk Assessment Guidance for the Superfund Program, Draft Final. EPA/901/5-89-001. USEPA Regional Risk Assessment Work Group. Boston, Ma.

U.S. Environmental Protection Agency, 1989c. Health Effects Assessment Summary Tables, Third Quarter FY 1989. OERR 9200.6-303-(89-3). USEPA Office of Research and Development. Washington, D.C.

U.S. Environmental Protection Agency, 1989d. Drinking Water Standards and Health Advisories Memorandum, April 5, 1989. USEPA Office of Drinking Water, Region III. Philadelphia, PA.

U.S. Environmental Protection Agency, 1989e. Guidance for Superfund Ecological Risk Assessments in Region I Draft Final. USEPA Region I. February, 1989.

U.S. Environmental Protection Agency, 1989f. Risk Assessment Guidance for Superfund - Environmental Evaluation Manual. Interim Final, Office of Emergency and Remedial Response; EPA/540/1-89/001A (OSWER Directive 9285.7-01). March 1989.

U.S. Environmental Protection Agency, 1989g. Risk Assessment Guidance for Superfund, Volume I, Human Health Evaluation Manual (Part A). EPA/540-1-89/002. Office of Emergency and Remedial Response. Washington, DC. U.S. Environmental Protection Agency, 1989h. Risk Assessment Methods: A Review and Evaluation of Past Practices in the Superfund and RCRA Programs. EPA-230-03-89-044. Office of Policy Analysis. Office of Policy, Planning, and Evaluation. Washington, DC.

U.S. Environmental Protection Agency, 1989i. Photoeologic Analysis, Nyanza Chemical Company. Ashland, MA, Photogeologic analysis; Environmental Photographic Interpretation Center, Environmental Monitoring Systems Laboratory; By L. Mike Fauss, Bionetics Corporation; Contract No. 68-03-3532.

U.S. Environmental Protection Agency, 1989j. Exposure Factors Handbook. Exposure Assessment Group, Office of Health and Environmental Assessments. Washington D.C.

U.S. Environmental Protection Agency, 1989k. Assessing Human Health Risks from Chemically Contaminated Fish and Shellfish. Office of Water Regulations and Standards. Criteria and Standards Division. Washington, DC 20460.

U.S. Environmental Protection Agency, 1990. AQUIRE. Aquatic Information Retrieval Data Base. Office of Information Branch.

U.S. Environmental Protection Agency, 1991a. Assessment and Control of Bioconcentratable Contaminants in Surface Waters. Draft Document. EPA-600-XX-XX-XXX. Office of Water Enforcement and Permits. Washington, DC.

U.S. Environmental Protection Agency, 1991b. Risk Assessment Guidance for Superfund, Volume I: Human Health Evaluation Manual, Supplemental Guidance, Standard Default Exposure Factors, Interim Final". OSWER Directive: 9285.6-03. USEPA Office of Emergency and Remedial Response, Toxics Integration Branch. Washington, D.C

U.S. Environmental Protection Agency, 1991c. Health Effects Assessment Summary Tables, Annual FY-1991. OSWER (OS-230), ORD (RD-689), OERR 9200.6-303 (91-1), NTIS No. PB91-921199. Washington D.C.

U.S. Environmental Protection Agency, 1991d. Proposed Sediment Quality Criteria for the Protection of Benthic Organisms: Acenaphthene. EPA-600-XX-XX-XXX. Office of Water and Office of Research and Development. Washington, DC 20460.

U.S. Environmental Protection Agency, 1991e. Proposed Sediment Quality Criteria for the Protection of Benthic Organisms: Fluoranthene. EPA-600-XX-XX-XXX. Office of Water and Office of Research and Development. Washington, DC 20460.
U.S. Environmental Protection Agency, 1991f. Proposed Sediment Quality Criteria for the Protection of Benthic Organisms: Phenanthrene. EPA-600-XX-XXX. Office of Water and Office of Research and Development. Washington, DC 20460.

U.S. Environmental Protection Agency, 1991g. Water Quality Criteria Summary. Office of Science and Technology, Health and Ecological Criteria Division. Washington, D.C.

U.S. Environmental Protection Agency, April 1992. Drinking Water Regulations and Health Advisories. Office of water. Washington, D.C.

USFWS (U.S. Fish and Wildlife Service), 1986a. Chromium Hazards to Fish, Wildlife, and Invertebrates: A Synoptic Review. Report No. 6. Patuxent Wildlife Research Center. Laurel, MD.

USFWS (U.S. Fish and Wildlife Service). 1986b. "Polychlorinated Biphenyl Hazards to Fish, Wildlife, and Invertebrates: A Synoptic Review. Report No. 7". Patuxent Wildlife Research Center. Laurel, MD.

USFWS (U.S. Fish and Wildlife Service), 1987a. Mercury Hazards to Fish, Wildlife, and Invertebrates: A Synoptic Review. Report No. 10". Patuxent Wildlife Research Center. Laurel, MD.

USFWS (U.S. Fish and Wildlife Service), 1987b. Polycyclic Aromatic Hydrocarbon Hazards to Fish, Wildlife, and Invertebrates: A Synoptic Review. Report No. 11. Patuxent Wildlife Research Center. Laurel, MD.

USFWS (U.S. Fish and Wildlife Service), 1988. Lead Hazards to Fish, Wildlife, and Invertebrates: A Synoptic Review. Report No. 14. Patuxent Wildlife Research Center. Laurel, MD.

USFWS (U.S. Fish and Wildlife Serv.ce), 1990. Chlordane Hazards to Fish, Wildlife, and Invertebrates: A Synoptic Review. Report No. 21. Patuxent Wildlife Research Center. Laurel, MD.

USFWS (U.S. Fish and Wildlife Service), 1991. National Ecology Research Center. Fort Collins, CO 80525. Personal communication.

Verschueren, Karel, 1977. <u>Handbook of Environmental Data on</u> <u>Organic Chemicals</u>. Van Nostrand Reinhold Company. New York, NY.

Webster, W.D., J.F. Parnell, and W.C. Biggs, 1985. <u>Mammals of the</u> <u>Carolinas, Virginia, and Maryland</u>. The University of North Carolina Press. Chapel Hill, NC. Wehran Engineering, Inc. Massachusetts Field Investigation Team (FIT), 1987a. Draft Phase II Report, Nyanza Vault Site in Ashland, Massachusetts. DEQE TAD No. 86-01-050-66. Wehran Engineering Project No. 50050.28. November 1987.

Wehran Engineering, Inc. Massachusetts FIT, 1987. Final Phase II Report for Nyanza Wetlands Sampling Program, Ashland, Massachusetts. DEQE TAD No. 86-01-050-66. Wehran Engineering Project No. 50050.29. October 1987.

Woodwell, G.M., C.F. Worster, Jr., and P.A. Issaacson, 1967. "DDT residues in an east coast estuary: A case of biological concentration of a persistent pesticide". Science, 156:821-823. Wetterberg, L., 1966. Acute Prophyria and Lead Poisoning, Lancet 1:498.

Zaroogian, G., J.F. Heltshe, and M. Johnson, 1985. Estimation of Toxicity to Marine Species with Structure Activity Models Developed to Estimate Toxicity to Freshwater Fish. Aquatic Toxicology, 6, 251-270.

Zen, E-an, 1983. <u>Bedrock Geologic Map of Massachusetts</u>. Department of the Interior, U.S. Geological Survey. Washington D.C.